Queensland Competition Authority

Decision

Queensland Rail's Draft Access Undertaking

June 2016

Level 27, 145 Ann Street, Brisbane Q 4000 GPO Box 2257, Brisbane Q 4001 Tel (07) 3222 0555 www.qca.org.au We wish to acknowledge the contribution of the following staff to this report:

The QCA's Queensland Rail team

© Queensland Competition Authority 2016

The Queensland Competition Authority supports and encourages the dissemination and exchange of information. However, copyright protects this document.

The Queensland Competition Authority has no objection to this material being reproduced, made available online or electronically but only if it is recognised as the owner of the copyright and this material remains unaltered.

Contents

EXECUT	IVE SUMMARY	I
INTROD	DUCTION	IV
Backgro	bund	iv
Declara	tion for third party access	v
History	of this Decision	v
Submiss	sions on the Draft Decision	v
Indeper	ndent economic advice	vi
The QC	A's considerations	vii
1	APPLICATION AND SCOPE	1
1.1	Review mechanism	3
1.2	Definition of access	4
1.3	Scope of the access undertaking	4
1.4	Line diagrams	6
1.5	Non-discriminatory treatment	7
1.6	Ring-fencing	8
1.7	Maintenance	9
1.8	Term of the undertaking	10
2	NEGOTIATION PROCESS	11
	Introduction	11
2.1	Information exchange	13
2.2	Timeframes	17
2.3	Refusal to provide access	18
2.4	Competing access requests	20
2.5	Access renewal rights	22
3	PRICING RULES	25
3.1	Hierarchy of pricing rules	27
3.2	Revenue adequacy	29
3.3	Limits on price differentiation	30
3.4	Pricing and revenue limits	39
3.5	Asset valuation methodology for negotiating access charges	41
3.6	Pricing for access rights at renewal	42
3.7	Rate of return	48
3.8	Take-or-pay arrangements	51
4	OPERATING REQUIREMENTS	53
4.1	Changes to train plans	55
4.2	Minimising the adverse effects of operational constraints	61
4.3	Coordination with adjoining networks	64
4.4	Passenger services	65

4.6ORM amendment process745REPORTING805.1Quarterly reporting on performance805.2Annual reporting on the access negotiation process815.3Reporting of cost and price information (when a reference tariff applies)825.4Reporting of cost and price information (when a reference tariff does not apply)845.5Audit requirements855.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS11PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1368.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1178.9Capial expenditure assessment process (Schedule E)1779.9Capial expenditure assessment process (Schedule E) <th>4.5</th> <th>Operating Requirements Manual</th> <th>67</th>	4.5	Operating Requirements Manual	67
5.1Quarterly reporting on performance805.2Annual reporting on the access negotiation process815.3Reporting of cost and price information (when a reference tariff applies)825.4Reporting of cost and price information (when a reference tariff does not apply)845.5Audit requirements855.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incretives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PARTB-TARIFF BUILDING BLOCKS1898.10Volumes189 <t< td=""><td>4.6</td><td>ORM amendment process</td><td>74</td></t<>	4.6	ORM amendment process	74
5.1Quarterly reporting on performance805.2Annual reporting on the access negotiation process815.3Reporting of cost and price information (when a reference tariff applies)825.4Reporting of cost and price information (when a reference tariff does not apply)845.5Audit requirements855.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incretives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PARTB-TARIFF BUILDING BLOCKS1898.10Volumes189 <t< td=""><td>_</td><td></td><td></td></t<>	_		
5.2Annual reporting on the access negotiation process815.3Reporting of cost and price information (when a reference tariff applies)825.4Reporting of cost and price information (when a reference tariff does not apply)845.5Audit requirements855.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART AREGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART BTARIFF BUILDING BLOCKS1838.10Volumes184 <td></td> <td></td> <td></td>			
5.3Reporting of cost and price information (when a reference tariff applies)825.4Reporting of cost and price information (when a reference tariff does not apply)845.5Audit requirements855.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of cormon network costs1198.4Capacity assessment1668.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PARTB-TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast base1918.13Opening asset base			
5.4Reporting of cost and price information (when a reference tariff does not apply)845.5Audit requirements855.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1668.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff aproach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B-TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast base1918.13Opening asset base193			
5.5Audit requirements855.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B-TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193			
5.6Regulatory accounts and cost allocation manual866ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART AREGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART BTARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193			
6ADMINISTRATIVE PROVISIONS886.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B-TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193		•	
6.1Commencement date of tariff-related reporting896.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART AREGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART BTARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	5.6	Regulatory accounts and cost allocation manual	86
6.2Miscellaneous other matters907STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1778.9Capital expenditure assessment process (Schedule E)178PART B-TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	6	ADMINISTRATIVE PROVISIONS	88
7STANDARD ACCESS AGREEMENT927.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART AREGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1778.9Capital expenditure assessment process (Schedule E)178PART BTARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	6.1	Commencement date of tariff-related reporting	89
7.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART AREGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART BTARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	6.2	Miscellaneous other matters	90
7.1QCA assessment approach947.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART AREGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1778.9Capital expenditure assessment process (Schedule E)178PART BTARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	7	STANDARD ACCESS AGREEMENT	92
7.2Access principles967.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A—REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193			
7.3Tripartite structure997.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A—REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193			
7.4Balanced risk allocation1027.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A—REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193			
7.5KPIs and performance reporting regime1068REFERENCE TARIFFS111PART A—REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base191		•	
8REFERENCE TARIFFS111PART A—REGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193			
PART AREGULATORY CONTEXT AND OPERATING ASSUMPTIONS1128.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B			
8.1Queensland Rail's 2015 DAU proposal1148.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B-TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8	REFERENCE TARIFFS	111
8.2Regulatory context of our Decision1148.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B-TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	PART A-	-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS	112
8.3Allocation of common network costs1198.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS8.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8.1	Queensland Rail's 2015 DAU proposal	114
8.4Capacity assessment1468.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B	8.2	Regulatory context of our Decision	114
8.5Form of regulation, take-or-pay and tariff structure1558.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART BTARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8.3	Allocation of common network costs	119
8.6Metropolitan network tariff approach1688.7Productivity, innovation and incentives1748.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8.4	Capacity assessment	146
 8.7 Productivity, innovation and incentives 8.8 Variation of reference tariffs 8.9 Capital expenditure assessment process (Schedule E) PART B—TARIFF BUILDING BLOCKS 8.10 Volumes 8.11 Forecast maintenance costs 8.12 Forecast operating costs 8.13 Opening asset base 191 	8.5	Form of regulation, take-or-pay and tariff structure	155
8.8Variation of reference tariffs1778.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8.6	Metropolitan network tariff approach	168
8.9Capital expenditure assessment process (Schedule E)178PART B—TARIFF BUILDING BLOCKS1838.10Volumes8.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8.7	Productivity, innovation and incentives	174
PART B—TARIFF BUILDING BLOCKS1838.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8.8	Variation of reference tariffs	177
8.10Volumes1848.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	8.9	Capital expenditure assessment process (Schedule E)	178
8.11Forecast maintenance costs1898.12Forecast operating costs1918.13Opening asset base193	PART B-	-TARIFF BUILDING BLOCKS	183
8.12Forecast operating costs1918.13Opening asset base193	8.10	Volumes	184
8.13 Opening asset base 193	8.11	Forecast maintenance costs	189
	8.12	Forecast operating costs	191
	8.13	Opening asset base	193
8.14 Past capital expenditure 216	8.14	Past capital expenditure	216
8.15 Forecast capital expenditure 218	8.15	Forecast capital expenditure	218
8.16 Capital charges for the coal RAB 221	8.16	Capital charges for the coal RAB	221
8.17QCA's required reference tariffs223	8.17	QCA's required reference tariffs	223
PART C—TARIFF ADJUSTMENT 228	PART C-	-TARIFF ADJUSTMENT	228
8.18Adjustment amount expectations228			-

8.19	Adjustment methodology	247	
8.20	Tariff for 2013–14 to 2015–16	249	
9	INVESTMENT PLANNING AND COORDINATION FRAMEWORK	254	
9.1	Network extension and funding agreements	256	
9.2	Network planning provisions	264	
10	LEGAL OVERVIEW	267	
10.1	Part 5 of the QCA Act	267	
10.2	Assessment approach	267	
10.3	Section 138(2) of the QCA Act	268	
ACRONYI	MS	279	
	X A : SUMMARY OF QCA REVENUE/PRICING MODEL (2016 FOR QUEENSLAND RAIL 2015	201	
DAU)		281	
APPENDI	X B : COMMENTS ON 2009 DRAFT DECISION TARIFF	284	
APPENDI	X C : SAA DRAFT DECISION COMPARISON	286	
APPENDI	X D : CL. 1.4 AND SCHEDULE I AMENDMENTS	329	
APPENDI	X E : WEST MORETON NETWORK COSTS ALLOCATION—PREVIOUS AND CURRENT		
ASSESSM		334	
APPENDI	X F : 2015 DAU MARK-UPS	348	
APPENDI	X G : 2015 SAA MARK-UPS	349	
APPENDIX H : QUEENSLAND RAIL'S EXTENSION DAAUS			
APPENDIX I : LIST OF SUBMISSIONS 31			
		360	
REFERENCES			

EXECUTIVE SUMMARY

Our Decision is to refuse to approve Queensland Rail's 2015 draft access undertaking (DAU) for providing access to its below-rail services. We have issued a secondary undertaking notice in accordance with s. 134 of the QCA Act, which asks Queensland Rail to give the QCA a copy of the amended DAU within 60 days—that is, by 16 August 2016.

Introduction

An approved undertaking for access to Queensland Rail's below-rail services is necessary to provide certainty about the detailed terms and conditions of access.

On 5 May 2015, Queensland Rail submitted its 2015 DAU, in response to an initial undertaking notice that we issued under section 133 of the QCA Act. On 8 October 2015, we released our Draft Decision on Queensland Rail's 2015 DAU. We also invited submissions on the Draft Decision from stakeholders by 24 December 2015 and subsequently provided a period for further submissions by 14 March 2016.

2015 DAU

We have reviewed Queensland Rail's 2015 DAU afresh in accordance with our obligations under section 138(2) of the QCA Act. We have also considered all stakeholder comments we received.

On balance, we have in this Decision broadly adopted the preliminary views we expressed in our Draft Decision on the 2015 DAU.

Our view is that Queensland Rail's 2015 DAU does not appropriately balance the rights and obligations of Queensland Rail with those of access seekers/holders and end customers. The approach in Queensland Rail's 2015 DAU is not appropriate, having regard to the approval criteria in the QCA Act, both across pricing and non-pricing matters.

We require changes to Queensland Rail's 2015 DAU to address these matters, as indicated in this Decision and detailed in Appendices F and G.

Pricing

Pricing for commercial freight services on the West Moreton network is perhaps the most contentious aspect of Queensland Rail's proposal.

The 2015 DAU includes a proposed 2015–16 West Moreton reference tariff for coal-carrying train services, equivalent to \$19.41/'000 gross tonne kilometres (gtk). Queensland Rail said this was below its proposed ceiling price of \$34.92/'000 gtk, based on a depreciated optimised replacement cost (DORC) valuation, which valued the existing assets in their existing (brownfields) configuration.

Asset valuation methodology

The West Moreton network was constructed in the 19th century for regional traffic. It does not reflect the service potential of a modern engineering equivalent asset, as it was not originally designed for coal transport.

Queensland Rail has spent increasing amounts on maintenance and capital expenditure to cope with coal traffics, as volumes have grown significantly since coal services began in 1996. We have largely accepted these costs as necessary to operate a network as idiosyncratic as the West Moreton network.

The high costs that Queensland Rail incurs to provide services on the West Moreton network highlight the need to examine the age of assets and appropriateness of revaluing them for inclusion in the opening asset value.

In respect of some assets (e.g. wooden sleepers and fences), Queensland Rail has in the past recovered the value of those assets by way of maintenance costs, and the Decision continues this approach. Other assets like tunnels, cuttings and embankments require only incidental further work once they have been built; they are essentially perpetual in nature. The value of those assets, where they are beyond their expected useful lives, is reflected in the value of the network as a whole.

The QCA has broadly adopted its preliminary views on an appropriate asset valuation methodology as outlined in its Draft Decision and has made adjustments to the regulatory asset base (RAB) for the above factors.

Drop in volumes

Queensland Rail's business environment has changed substantially since it submitted its 2013 DAU—with both coal and non-coal volumes having dropped significantly.

The question of how the material spare capacity on the West Moreton network should be treated therefore arises. On this occasion the QCA has not chosen to optimise the assets; rather the QCA's view is that coal traffics should only pay for the paths they can contract to use.

Queensland Rail previously said there was a binding constraint on the number of paths that coal could contract for. It now says there is no legally binding constraint on the paths that coal can contract for. However, Queensland Rail's position does not go to the issue of whether a constraint exists in practice. The QCA's position is that such a constraint exists. The QCA has therefore capped coal traffics' share of common network fixed costs (return on, and of, assets, as well as fixed maintenance and operating costs) to take into account contracting restrictions on coal services. We consider that this appropriately balances the competing interests of access seekers/holders, who should not pay for services they cannot contract, and the interests of Queensland Rail, which seeks a return on its investments.

On this basis, the QCA requires a reference tariff equivalent to \$17.92/'000 gtk from 1 July 2016.

Adjustment amount

In its withdrawn 2013 DAU, Queensland Rail proposed an adjustment to reflect any over- or underrecovery of access charges from 1 July 2013 (given the tariffs in the 2008 access undertaking were scheduled to expire on 30 June 2013) to the date when the new tariff was approved. Moreover, Queensland Rail on a range of occasions indicated that it would make such an adjustment, including in its 2013/14 annual report. However, the 2015 DAU did not propose such an adjustment.

Our view is that approving Queensland Rail's proposal would create regulatory uncertainty, which would, among other things, adversely impact on investment.

Having regard to the relevant factors in section 138(2) of the QCA Act, we require the 2015 DAU be amended to include an adjustment amount payable by Queensland Rail for its over-recovery of access charges.

Our Draft Decision proposed an adjustment amount for the West Moreton network. However, having considered stakeholder comments on the Draft Decision and undertaken a further round of submissions, we have determined that an adjustment amount should also apply to the Metropolitan network. This is consistent with the expectation generated by Queensland Rail's earlier representations that there would be an adjustment to reflect any over- or under-recovery of access charges from 1 July 2013—which meant that there would be an adjustment over both the West Moreton and Metropolitan networks.

Stakeholders, including Queensland Rail, also raised concerns with the QCA's proposed approach to implementing the adjustment amount. Having regard to these matters, the QCA has adopted an Adjustment Amount mechanism to address overpayment, which will operate by reference to a comparison between the access charge that a particular access holder actually paid in the period from 1 July 2013 to the date of approval, and the access charge that the access holder would have paid during

that period if the new reference tariff had been in effect at that time (see cl. 7.1 of the QCA's mark-ups to Schedule D of the 2015 DAU in Appendix F of this Decision). This mechanism operates in a similar manner to that proposed by Queensland Rail in relation to adjustment charges for variations to a reference tariff or a reference tariff which becomes effective from a date prior to the QCA's approval of that reference tariff (see cl. 7.1 of Schedule D of the 2015 DAU).

Non-pricing matters

Queensland Rail's 2015 DAU covers a broad range of non-pricing matters, including negotiation processes, reporting obligations and contracting and investment frameworks. The QCA has broadly adopted its preliminary views as stated in the Draft Decision—that is, that in many respects Queensland Rail's proposals skew rights and obligations in its favour and away from access holders and seekers.

We require changes to address this imbalance, including to:

- streamline and rebalance the scope, capacity, negotiation and administrative sections of the 2015 DAU to clarify the dispute resolution process and remove any inappropriate discretionary powers in Queensland Rail's favour
- provide greater transparency in the planning, scheduling and 'day of operations' processes
- apply the operating requirements for train services on Queensland's Rail's infrastructure consistently to all relevant parties
- increase transparency of Queensland Rail's reporting and compliance processes
- enable a better balance in risk allocations across parties in the standard access agreement (SAA)
- embed the right of a customer to fund a network extension and the obligation of Queensland Rail to facilitate a network extension when it agrees terms with a user funder

Legal basis for our Decision

Our decision to refuse to approve Queensland Rail's 2015 DAU has been formed in accordance with the approval criteria in the QCA Act (s. 138(2)). A key aspect of our approach has been to have regard to each aspect of section 138(2).

In some circumstances, there may be tensions between various aspects of this section of the Act, including between the objects clause (ss. 138(2)(a) and 69E), legitimate business interests of Queensland Rail (s. 138(2)(b)), the public interest (s. 138(2)(d)), the interests of access seekers (s. 138(2)(e)) and the pricing principles (ss. 138(2)(g) and 168A). This necessarily involved considering each element before forming a view.

Further information on our approach to assessing Queensland Rail's 2015 DAU is provided in Chapter 10 of this Decision.

The way forward

We have issued Queensland Rail with a secondary undertaking notice in accordance with section 134 of the QCA Act:

- stating our reasons for refusing to approve the 2015 DAU (i.e. as contained in this Decision as well as the amended DAU and SAA); and
- asking Queensland Rail to give us a copy of the amended DAU (and SAA) within 60 days (i.e. by 16 August 2016), unless this period is extended.

If Queensland Rail does not comply with this notice, the QCA may prepare, and approve, a DAU for Queensland Rail's declared service.

INTRODUCTION

Background

Queensland Rail owns and operates a 6,500-kilometre rail network, including the commuter lines in south east Queensland, the West Moreton network, and the Mount Isa and North Coast lines (see Fig. 1). It also operates the state's suburban and long-distance passenger services.¹





Source: Queensland Rail

¹ Queensland Rail was created in 2010 when the Queensland Government split the former QR Ltd. Queensland Rail owns most of the former QR Ltd rail network in Queensland, apart from the tracks in central Queensland owned by Aurizon Network Pty Ltd (formerly QR Network Pty Ltd).

Declaration for third party access

The services provided by Queensland Rail's intrastate rail network were declared by regulation in 1997, making the services subject to the third-party access provisions of the QCA Act. As a result of that declaration, Queensland Rail, access seekers and access holders gained rights and obligations relating to the negotiation of the terms and conditions of access to Queensland Rail's rail transport infrastructure.

The below-rail (track) network is subject to the access regime established by Part 5 of the QCA Act. It follows a negotiate–arbitrate model, in which the primary responsibility is on the access provider and access seeker to negotiate on price and non-price terms. Part 5 provides for the development of an access undertaking to guide how the access regime should operate.

History of this Decision

Following its creation in 2010, Queensland Rail commenced a process to transition from the 2008 undertaking to one that better reflected its assets and business structure. Some key milestones in the course of that process are as follows:

- March 2012—Queensland Rail submitted the 2012 DAU which sought to replace its 2008 undertaking with a set of requirements more suited to a network operator which is not vertically integrated with an above-rail freight business.
- April 2012—the QCA released an Issues Paper on the 2012 DAU.
- February 2013—Queensland Rail withdrew its 2012 DAU and submitted the February 2013 DAU (the 2013 DAU). In doing so, Queensland Rail indicated that it had revised the 2012 DAU to reflect concerns raised by stakeholders.
- April and May 2013—the QCA hosted a series of workshops on issues in the February 2013 DAU, including above-rail operational issues, West Moreton network pricing, standard access agreements (SAAs), Mount Isa pricing and investment framework matters.
- June 2013—Queensland Rail resubmitted its 2013 DAU and included, for the first time, its proposed reference tariffs for the West Moreton network from 1 July 2013.
- June 2014—the QCA released its consultation paper on western system coal tariffs in the 2013 DAU along with a report on the West Moreton network prepared by its rail consultant, B&H Strategic Services (B&H).
- June 2014—the QCA conducted a workshop with stakeholders on West Moreton network coal tariffs.
- October 2014—the QCA released its 2014 Draft Decision.
- December 2014—Queensland Rail withdrew its June 2013 DAU.
- February 2015—the QCA issued an initial undertaking notice under section 133 of the QCA Act, requiring Queensland Rail to submit a draft access undertaking (DAU) to the QCA within 90 days after receiving the notice.
- May 2015—Queensland Rail submitted a DAU to the QCA, within the time specified in the section 133 notice.
- October 2015—the QCA released its 2015 Draft Decision ('the Draft Decision').

Submissions on the Draft Decision

The QCA received submissions on the Draft Decision from Aurizon, Glencore, Queensland Rail, New Hope and Yancoal.

Queensland Rail considered that large aspects of the QCA's Draft Decision would be beyond power. For instance, Queensland Rail said that we had not correctly applied our approval criteria in section 138(2) by:

- not recognising the over-riding guidance of the objects clause (s. 138(2)(a)) and the pricing principles (s. 138(2)(g))
- not giving adequate regard to Queensland Rail's legitimate business interests (s. 138(2)(b))
- 'trading off' the pricing principles against other factors
- retrospectively applying an adjustment amount.

In contrast, other stakeholders were broadly supportive of many aspects of the Draft Decision, but requested refinements in a range of areas, including on our draft positions on the SAA, the adjustment amount and the pricing principles.

In light of the large amount of new material, on 15 January 2016 the QCA invited stakeholders to make further comments on submissions received through a 'submissions on submissions' process. On 19 January 2016, QCA staff released a staff 'Request for comments' paper and subsequently made the QCA's Queensland Rail tariff model available to stakeholders upon request. Following a request from Queensland Rail on 25 January 2016, the closing date for further submissions was extended to 14 March 2016.

The QCA received seven further submissions, from Aurizon, Glencore, Queensland Rail, New Hope, Queensland Resources Council (QRC), Asciano and Yancoal. These submissions largely responded to the 'Request for comments' staff paper and, for the most part, reiterated existing positions.

Independent economic advice

Professor Flavio Menezes was engaged as an independent expert to report on economic matters related to the West Moreton and Metropolitan tariffs.

Professor Menezes is a Professor of Economics at the University of Queensland, who was previously the Foundation Director of the Australian Centre of Regulatory Economics at Australian National University. He has taught, published and consulted extensively in the areas of competition and regulatory economics.

He provided two reports, 'A regulatory economics assessment of the proposed Western System asset valuation approaches' and 'The economic impact of QR's proposal not to include an adjustment to refund or recoup differences in tariffs', which the QCA published in October 2015 along with the Draft Decision.

We subsequently engaged Professor Menezes to respond to reports by other experts, that commented on his previous reports, and to provide a report on the approach to cost allocation. On 8 April 2016, before Professor Menezes had completed these reports, the Queensland Government appointed him to the QCA Board. Professor Menezes disclosed his interest to a meeting of the QCA on 19 April 2016 and was not present when the QCA considered this Decision at its 14 June 2016 meeting, or when the QCA discussed matters related to the Decision at the April and May meetings.

Professor Menezes' opinions were obtained in his capacity as an independent expert. We also engaged Professor Stephen King of Monash University as an expert to peer review Professor Menezes' reports and conclusions. Professor King is a Professor of Economics at Monash University, a Member of the Economic Regulation Authority of Western Australia (ERA) and a Member of the National Competition Council (NCC).

Professor Menezes' reports and Professor King's peer review of Professor Menezes' reports are available on the QCA's website.

The QCA's considerations

The QCA has considered Queensland Rail's 2015 DAU and stakeholder submissions in accordance with the assessment criteria in section 138(2) of the QCA Act (see Box 1).

Box 1: The legal framework

The QCA may approve the 2015 DAU only if the QCA considers it appropriate to do so having regard to each of the matters set out in the QCA Act:

The Authority may approve a draft access undertaking only if it considers it appropriate to do so having regard to each of the following (s. 138(2)) —

- (a) the object of Part 5 of the QCA Act; which is to promote the economically efficient operation of, use of and investment in, significant infrastructure by which services are provided, with the effect of promoting effective competition in upstream and downstream markets (s. 69E);
- (b) the legitimate business interests of the owner or operator of the service (s. 138(2)(b));
- (c) if the owner and operator of the service are different entities—the legitimate business interests of the operator of the service are protected (s. 138(2)(c));
- (d) the public interest, including the public interest in having competition in markets (whether or not in Australia) (s. 138(2)(d));
- (e) the interests of persons who may seek access to the service, including whether adequate provision has been made for compensation if the rights of users of the service are adversely affected (s. 138(2)(e));
- (f) the effect of excluding existing assets for pricing purposes (s. 138(2)(f));
- (g) the pricing principles mentioned in section 168A; which in relation to the price of access to a service are that the price should:
 - (i) generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved (s. 168A(a));
 - (ii) allow for multi-part pricing and price discrimination where it aids efficiency (s. 168A(b));
 - (iii) not allow a related access provider to set terms and conditions that discriminate in favour of the downstream operations of the access provider or a related body corporate of the access provider, except to the extent the cost of providing access to other operators is higher (s. 168A(c)); and
 - (iv) provide incentives to reduce costs or otherwise improve productivity (s. 168A(d)); and
- (h) any other issues the authority considers relevant.

It is not open to the QCA to approve an access undertaking that does not include the matters required by section 137. These are:

- (1) an expiry date (s. 137(1));
- (2) provisions for identifying, preventing and remedying conduct by an access provider that provides, or proposes to provide, access to itself or a related body corporate that unfairly differentiates in a material way between access seekers (in negotiations (s. 137(1A)(a)(i)) and access holders (in providing the service (s. 137(1A)(a)(ii))); and
- (3) provisions preventing an access provider that provides, or proposes to provide, access to itself or a related body corporate recovering, through the price of access, costs that are not reasonably attributable to the provision of the service (s. 137(1A)(b)).

Sections 137(2) and 138A set out matters that may be included in an access undertaking.

Our Decision

Our Decision is to refuse to approve Queensland Rail's 2015 DAU.

In this Decision we have explained our views and have set out those amendments that we consider necessary before we can approve Queensland Rail's proposed 2015 DAU. Relevantly, where matters are not in dispute, the QCA has generally adopted the positions contained in its Draft Decision. For these matters, an elaboration of the QCA's positions is contained in the Draft Decision.

Structure

This Decision follows the structure of the 2015 DAU:

- Chapter 1: Application and scope—the extent to which the 2015 DAU applies to the entirety of Queensland Rail's declared service as well as ring-fencing and non-discriminatory treatment obligations.
- Chapter 2: Negotiation and capacity management—the negotiation framework between Queensland Rail and access seekers.
- Chapter 3: Pricing rules—the rules for setting access charges under the access undertaking.
- Chapter 4: Operating requirements—the rules for how Queensland Rail will demonstrate capacity, coordinate maintenance and schedule and operate trains.
- Chapter 5: Reporting—the approach to reporting and audit of costs, performance and compliance with the undertaking.
- Chapter 6: Administrative provisions—provisions relating to, among other things, dispute resolution and tariff reporting.
- Chapter 7: SAA—the structure and terms of the standard access agreement (SAA).
- Chapter 8: Reference tariffs—the approach to the reference tariffs for the West Moreton network and the Metropolitan network. The chapter also addresses the issue of an adjustment amount to reflect the previous over-recovery of access charges by Queensland Rail.
- Chapter 9: Investment framework, planning and coordination—Queensland Rail's obligation to permit, but not fund, an extension to the network to facilitate the execution of an access agreement.
- Chapter 10: Legislative framework—how we have applied our legislated obligations in making our Decision.

Secondary undertaking notice

On 17 June 2016, the QCA issued Queensland Rail with secondary undertaking notice under section 134 of the QCA Act. The QCA asks Queensland Rail Limited to give to the QCA a copy of the amended 2015 DAU within 60 days of receiving this Notice (i.e. by 16 August 2016) or, if the period is extended under section 134(2A) of the QCA Act, the extended period.

1 APPLICATION AND SCOPE

Part 1 of the Queensland Rail's 2015 DAU contains provisions on the scope of access, nondiscriminatory treatment of above-rail operations, and the term of the undertaking.

Our Decision accepts many aspects of Queensland Rail's proposals, but has also made changes to Part 1, including to:

- clarify the extent to which the undertaking will apply to Queensland Rail's activities
- provide that descriptions of the infrastructure to which the undertaking will apply are up to date
- provide for a separation of process between disputes to which the QCA Act apply and those to which the QCA Act does not apply
- enhance ring-fencing obligations.

Introduction

Scope and administrative matters are addressed in an approved access undertaking to provide certainty to access seekers negotiating access to a declared service, while protecting the legitimate business interests of the service provider. These matters include the scope of access covered by the undertaking, as well as provisions for non-discriminatory treatment.

On balance, we have in this section of the Decision broadly adopted the preliminary views we expressed in the Draft Decision in relation to Part 1 of Queensland Rail's 2015 DAU on application and scope.

Key issues are summarised in Table 1 below. Matters that require a more detailed explanation are discussed in Sections 1.1 to 1.8.

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision		
1. General review mech	1. General review mechanism				
No proposal on a general review mechanism of the approved undertaking was presented before the Draft Decision.	Disagreed with New Hope's proposal of a general review mechanism (see right).	New Hope considered that a review mechanism was necessary.	See Section 1.1.		
2. Definition of access	2. Definition of access				
The definition of access means the non-exclusive right to use a specified part of the network.	Accepted the Draft Decision.	New Hope and Glencore disagreed and said that the definition is too narrow.	See Section 1.2.		
3. Scope of the access undertaking					
The access undertaking applies where Queensland Rail (or its successor, assign or subsidiary) is the railway manager.	Disagreed with our definition of 'Network' and with New Hope's proposal to support connecting private infrastructure.	New Hope supported the Draft Decision; however, it suggested amendments to support connecting private infrastructure, and	See Section 1.3.		

Table 1: Summary of key positions and decision—application and scope

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision
		provisions to allow the QCA to require a standard connection agreement.	
4. Line diagrams			
Queensland Rail will notify stakeholders before making material amendments to the line diagrams, and provide a dispute process if stakeholders question the accuracy of the line diagrams.	Disagreed with the Draft Decision, and considered that the dispute process should not be extended to access holders.	New Hope supported the Draft Decision.	See Section 1.4.
5. Non-discriminatory t	reatment		
Queensland Rail to clearly set out how it will be prevented from unfairly differentiating between access seekers and access holders.	Disagreed with the Draft Decision and said that the QCA is beyond power.	New Hope supported the Draft Decision. Asciano supported the Draft Decision, however they said that it could be strengthened.	See Section 1.5.
6. Ring-fencing			
The QCA may require Queensland Rail to submit a DAAU implementing ring- fencing arrangements.	Disagreed and said that our proposed provisions are ambiguous, outside of power and uncertain.	New Hope supported the Draft Decision. Asciano disagreed with the Draft Decision.	See Section 1.6.
7. Maintenance			
The Draft Decision clarified Queensland Rail's obligation to maintain the network.	Disagreed and said that we are beyond power and in direct conflict with section 119 of the QCA Act.	No comments.	See Section 1.7.
8. Term of the Undertal	king		·
The Draft Decision proposed to accept the 2015 DAU term from the date of approval to 30 June 2020.	Accepted in principle.	Not opposed.	See Section 1.8.

1.1 Review mechanism

The Draft Decision did not discuss a general review mechanism to review the undertaking after it was approved.

Stakeholders' submissions

New Hope said that such a general review mechanism was needed to appropriately protect access seekers and access holders.²

Queensland Rail disagreed with New Hope's amendments to introduce a general review mechanism and submitted that this was beyond powers under the QCA Act.³

QCA analysis and Decision

The QCA does not accept the need for a general review mechanism.

We note New Hope's position that:

- Many provisions of the approved access undertaking will be new and untested
- The impact of small changes (e.g. traffic mix, capital expenditure requirements and train configuration) on a low-volume network will be more material compared to a high-volume network like the central Queensland coal network (CQCN)
- Access holders and access seekers bear the asymmetric risks of a longer-term access undertaking, as Queensland Rail is allowed to submit a voluntary draft amending access undertaking (DAAU) at any time, while access holders and access seekers cannot reopen an approved undertaking
- The current disagreements over adjustment amounts may lead to unanticipated consequences.⁴

That said, we consider a general review provision to open an approved access undertaking may create regulatory uncertainty for all parties. The proposed provision is also overly broad and the trigger criteria are somewhat subjective.

We note that the DBCT 2010 access undertaking contains a general review mechanism. However, that undertaking was to be for five years, while the new access undertaking for Queensland Rail has a shorter duration, with it terminating on 30 June 2020. Further, the DBCT provision was a voluntary inclusion by DBCT whereas Queensland Rail opposes such a provision. In the Draft Decision we proposed that Queensland Rail strengthen the audit provisions and proposed other checks and balances intended to mitigate against monopolistic behaviour. We have also, in this Decision (see Section 1.5 below), strengthened the unfair discrimination clauses in the 2015 DAU in order to prevent the kind of behaviour that New Hope's proposal is seeking to cure. Given this, we do not consider a general review mechanism as proposed by New Hope is necessary.

Additionally, we note that under section 139 of the QCA Act, we may require Queensland Rail to submit a DAAU if we consider the approved access undertaking needs to be amended to be consistent with a provision of the QCA Act. We consider this provision provides sufficient

² New Hope, sub. 23: 5–6.

³ Queensland Rail, sub. 33: 53.

⁴ New Hope, sub. 23: 5–6.

protection in the event of an unexpected and significant change in circumstances that would render the undertaking inconsistent with a provision of the QCA Act.

1.2 Definition of access

The Draft Decision proposed to accept Queensland Rail's 2015 DAU definition of 'access' to mean 'the non-exclusive right to use a specified part of the 'network' for the purposes of operating train services'. While the Draft Decision definition of 'network' differed from that in the 2015 DAU, both versions reflected the term 'network' as referring to 'rail transport infrastructure as defined in the Transport Infrastructure Act' (TI Act).⁵

Queensland Rail accepted the Draft Decision.⁶

New Hope and Glencore said the definition of access should be widened by generally reinstating clause 2.1(b) of the 2008 access undertaking.⁷ This would involve a detailed list of matters for which access should be provided, rather than our Draft Decision approach of defining access by reference to the TI Act.

We have adopted our Draft Decision position that 'access', through the definition for 'network', should be aligned to rail transport infrastructure as defined in the TI Act. This is how the facility of rail transport infrastructure is defined in the QCA Act for the declared service.

We consider 'access to a network' should be aligned to how this term is defined in the TI Act so that there is consistency in definitions thereby reducing uncertainty. For this reason, rail transport infrastructure should be aligned to the definition in the TI Act. By aligning the definition, access essentially includes the access to facilities necessary for operating a railway, such as railway track, bridges, communication systems, marshalling yards, overhead electrical power supply systems.

For clarification purposes, we have provided examples of these facilities in our footnote to the definition of network in clause 7.1 in Appendix F consistent with the services in the TI Act.

Summary 1.1

The 2015 DAU's definition of 'Network' must include a footnote that provides examples of rail transport infrastructure.

See the definition of 'network' in Appendix F.

1.3 Scope of the access undertaking

Queensland Rail proposed in its 2015 DAU that the undertaking apply to Queensland Rail where it is a railway manager, except in the circumstance where it was providing railway manager services to the owner of the infrastructure and the terms of the contract with the owner did not allow Queensland Rail to comply with aspects of the 2015 DAU.

In the Draft Decision we said that the 2015 DAU should apply to all rail transport infrastructure for which Queensland Rail (or its successor, assign or subsidiary) is the railway manager,

⁵ QCA, October 2015 proposed DAU, cl. 7.1.

⁶ Queensland Rail, sub. 33: 53.

⁷ New Hope, sub. 23: 4. Glencore, sub. 25: 6.

consistent with the declared service in section 250(1)(b) of the QCA Act and the definition of network.

Stakeholders' submission

Queensland Rail submitted that our definition of network is concerning, and that we have no statutory power to seek to regulate future successors, assigns, or subsidiaries of Queensland Rail through the 2015 DAU.⁸

New Hope supported the Draft Decision. However they have suggested amendments to support connecting private infrastructure, and provisions to allow the QCA to require a standard connection agreement during the term of the undertaking (should the need arise).^{9,10} Queensland Rail disagreed with New Hope's suggested amendments and said they are unnecessary and inappropriate, and beyond powers under the QCA Act.¹¹

QCA analysis and Decision

We have adopted our Draft Decision position.

We disagree with Queensland Rail's position that we have do not have power to regulate future successors, assigns, or subsidiaries of Queensland Rail. This is because the declared service under section 250(1)(b) of the QCA Act includes rail transport infrastructure for which Queensland Rail or a successor, assign or subsidiary of Queensland Rail is the railway manager. Given the Act provides this definition, we repeat our Draft Decision analysis that the access undertaking should also apply when Queensland Rail, or a successor, assign or subsidiary of Queensland Rail, is the railway manager.

We also disagree with New Hope's position, as the QCA Act and parts of our amendments to the 2015 DAU already address their concerns as follows:

- The 'Extension Access Principles' section in Schedule I of the amended DAU identifies the principles of negotiation if access seekers or access holders want to build a rail connection (a form of an extension) from the mainline to its private infrastructure
- Queensland Rail is already required to negotiate access to the network in good faith under section 100 of the QCA Act, and any dispute in relation to the connection of private infrastructure to the network in order to gain access to the network is likely to be considered either an access dispute or an extension related dispute.

We note also that in the amendments we provided pursuant to our 2015 Draft Decision clause 1.2.1(b)(i)(C)(1) in the 2015 DAU was not deleted. This was an oversight. For our required amendments to be effective, this particular subclause should also be deleted.

⁸ Queensland Rail, sub. 26: 87–88.

⁹ New Hope, sub. 32: 39.

¹⁰ New Hope, sub. 23: 4–5.

¹¹ Queensland Rail, sub. 33: 52–53.

Summary 1.2

The 2015 DAU must apply to all rail transport infrastructure for which Queensland Rail (or Queensland Rail's successor, assign or subsidiary) is the railway manager, consistent with the declared service in section 250(1)(b) of the QCA Act.

See clauses 1.2.1 and 7.1 in Appendix F.

1.4 Line diagrams

Queensland Rail proposed in its 2015 DAU to use 'reasonable endeavours' to publish line diagrams that are accurate in all material respects.

The Draft Decision accepted the proposal; however, we proposed to require Queensland Rail to notify stakeholders before making material amendments to the line diagram, and to provide a dispute resolution process to access holders and access seekers.

Stakeholders' submissions

Queensland Rail said:

- The QCA had no statutory power to extend the dispute process to access holders;
- Line diagrams did not define the scope of the declared service of the network to which the undertaking related; and
- The accuracy of the line diagrams could not reasonably be regarded as affecting competition in above-rail markets and it was unclear why access seekers or, in particular, access holders would ever need to rely on the line diagrams.¹²

New Hope supported the Draft Decision and said that the line diagrams represented, in practical terms, parts of the network which Queensland Rail acknowledged were regulated under the undertaking.^{13,14}

QCA analysis and Decision

Save for the amendment discussed below, we have adopted our Draft Decision position in relation to line diagrams.

We disagree with Queensland Rail's position for the following reasons:

- An access undertaking can include an obligation on the owner or operator to comply with the decisions of the QCA in relation to disputes about matters stated in the undertaking. This also supports the object of Part 5 as investment in the network and effective market competition is likely to be promoted by regulatory certainty arising from a stated dispute resolution process for an access holder in an access undertaking; and
- While the line diagrams may not strictly define the scope of the declared service of the network to which the undertaking relates, they are used by various parties to ascertain the railways for which Queensland Rail is the railway manager. For this reason, we consider line diagrams to be relevant to the identification of rail infrastructure which is subject to the

¹² Queensland Rail, sub. 26: 88.

¹³ New Hope, sub. 23: 3.

¹⁴ New Hope, sub. 32: 39.

declared service. This is recognised by Queensland Rail in clause 1.2.3(a) of its 2015 DAU drafting.

However, in addition to the proposed amendments in the Draft Decision which we have adopted in this Decision, we have amended clause 6.1.2(b) of the undertaking to clarify which disputes between an access holder and Queensland Rail can be determined under the undertaking. We have also clarified clause 6.1.4 to provide for a separation of process between access disputes under Part 5 Division 4 of the Act and other disputes. We consider that this decision is appropriate after having regard to all of the factors in section 138(2) of the QCA Act; including, after having regard to the interests of access seekers, access holders and the legitimate business interests of Queensland Rail (s. 138(2)(b), (e), (h)).

Summary 1.3

The 2015 DAU must provide that Queensland Rail will notify stakeholders before making material amendments to the line diagrams, and provide a dispute process if stakeholders question the accuracy of the line diagrams.

See clauses 1.2.3, 6.1.2 and 6.1.4 in Appendix F.

1.5 Non-discriminatory treatment

Queensland Rail proposed in its 2015 DAU to acknowledge its obligations under sections 100, 104, 125 and 168A(c) of the QCA Act. The 2015 DAU did not contain further provisions or obligations in this respect.

The Draft Decision proposed an amendment that would require Queensland Rail to clearly set out how it will be prevented from unfairly differentiating between access seekers and access holders, consistent with sections 100, 104, 125, 137(1a) and 168A(c) of the QCA Act.

Stakeholders' submissions

Queensland Rail disagreed with the Draft Decision and said that it was unnecessary to duplicate the provisions of the QCA Act. It also stated that we had left out parts of the QCA Act provisions when duplicating, which had the effect of distorting and extending the intent of the QCA Act provisions. Queensland Rail said that as such the proposed amendment was beyond power.¹⁵

Both New Hope and Asciano supported the Draft Decision, although New Hope suggested that the provisions could be strengthened.^{16, 17, 18}

QCA analysis and Decision

The QCA has reviewed its draft position in light of further submissions. We accept it is not necessary to duplicate the provisions in the QCA Act which apply in any event. Consistent with this approach we have deleted the priority given to the obligations in the DAU, and instead extended the acknowledgement to the permissions (in some case conditional permissions) allowed to Queensland Rail by the QCA Act.

¹⁵ Queensland Rail, sub. 26: 88.

¹⁶ New Hope, sub. 23: 3.

¹⁷ New Hope, sub. 32: 39–40.

¹⁸ Asciano, sub. 28: 10–11.

Summary 1.4

The non-discrimination provisions in the 2015 DAU must:

- (a) include an acknowledgement by Queensland Rail of the permissions (in some cases conditional permissions) allowed to Queensland Rail by the QCA Act
- (b) delete the priority given to the provisions of the undertaking.

See clause 1.3 in Appendix F.

1.6 Ring-fencing

Queensland Rail's 2015 DAU proposed that if Queensland Rail gained interests in markets upstream or downstream from the below-rail services that were in competition with third parties in those markets, then Queensland Rail would inform the QCA and submit to the QCA a DAAU setting out its ring-fencing obligations.

The Draft Decision considered Queensland Rail's 2015 DAU drafting provided insufficient certainty to access holders and access seekers. As such, we proposed amendments to set out a clear trigger for when, and the process by which, Queensland Rail would amend the undertaking to include ring-fencing arrangements.

Stakeholders' submissions

Queensland Rail said the QCA's amended 2015 DAU drafting (cl. 6.5) was ambiguous, uncertain, and outside of power.¹⁹

New Hope supported the Draft Decision.²⁰ Asciano disagreed with the Draft Decision and said that some level of separation was required to minimise the potential for cost shifting and cross-subsidisation between Queensland Rail's businesses. As a result, it considered the ring-fencing obligations from the previous access undertaking should be retained.²¹

QCA analysis and Decision

After reviewing all of the stakeholder comments in relation to the proposed ring-fencing provisions we have decided to adopt Queensland Rail's original proposal (cl. 2.2.3), with some minor amendments to address the main concerns that were outlined in the Draft Decision.

As discussed in the Draft Decision, Queensland Rail's existing operational structure means ringfencing issues are unlikely to affect competition, as Queensland Rail's passenger operations do not compete with other above-rail operators. Further, we do not consider that this is likely to change during the term of this undertaking.

However, we still have the following concerns in relation to Queensland Rail's proposed ring-fencing provisions:

- Queensland Rail has not offered to provide the ring-fencing provisions before any hypothetical entry into an above-rail market; and
- Timeframes in relation to Queensland Rail's proposed lodgement of a DAAU are not clear.

¹⁹ Queensland Rail, sub. 26: 90–91.

²⁰ New Hope, sub. 23: 3.

²¹ Asciano, sub. 28: 8–9.

Because of this, we do not consider it appropriate to approve Queensland Rail's proposed clause 2.2.3 on ring-fencing appropriate in accordance with s. 138(2), including by reference to subsections (a), (d), (e) and (h). Rather, we require Queensland Rail to amend that clause to provide that Queensland Rail will inform the QCA if the stated interests are 'likely' to arise during the term of the undertaking. Further, we have amended the provision to state that Queensland Rail must inform the QCA as soon as reasonably practicable if the stated interests are likely to arise.

We consider that our required amendments to clause 1.3, coupled with the unfair differentiation and enforcement provisions under the QCA Act, will act to offset our concerns.

With these changes, we consider Queensland Rail's ring-fencing arrangements, combined with sections 104, 150A, 150AA and 153 of the QCA Act, and clause 1.3(d) in our proposed DAU, address Asciano's concerns.

We consider that the required amendments are appropriate after having regard to section 138(2) of the QCA Act in that they provide a balance between protecting the legitimate business interest of Queensland Rail and mitigating against monopoly behaviour likely to adversely affect competition in upstream and downstream markets.

Summary 1.5

The ring-fencing arrangements in the 2015 DAU must provide that Queensland Rail will inform the QCA if the relevant interests in upstream and downstream markets are 'likely' to arise during the term of the undertaking (in addition to informing the QCA if the such interests 'do' arise); and, that Queensland Rail will inform the QCA as soon as reasonably practicable.

See clause 2.2.3 in Appendix F.

1.7 Maintenance

In its 2015 DAU submission, Queensland Rail proposed to acknowledge its obligation to maintain the network within the SAA.

The Draft Decision proposed to accept that this obligation remain in the SAA, but sought to clarify Queensland Rail's obligation to maintain the network—that is, Queensland Rail has to comply with the network management principles (NMP), operating requirements manual (ORM) and interface risk management plan (IRMP). We also proposed introducing a definition of maintenance work in the SAA, to mean 'any works involving maintenance, repairs to, renewal, replacement and associated alterations or removal of, the whole or any part of the Network and includes any inspections or investigations of the Network'.

Stakeholders' submissions

Queensland Rail disagreed with the Draft Decision to include the term 'replacement' in the definition of maintenance work and said it was beyond power and in direct conflict with section 119 of the QCA Act.²²

²² Queensland Rail, sub. 26: 71.

QCA analysis and Decision

We note that the term 'replacement' forms part of the definition of 'extension' in the QCA Act and it was not our intention for the treatment of maintenance to have an impact on Queensland Rail's extension obligations. To avoid any possible confusion, we have therefore adopted a definition of maintenance that excludes the term 'replacement'. These obligations are discussed further in Chapter 9.

Separately, given maintenance of the network is integral to access and safety, we consider it appropriate that Queensland Rail's maintenance obligation is clarified and implemented in the access undertaking itself. We have therefore included an overarching maintenance obligation on Queensland Rail in the body of the access undertaking linked to the definition of 'Below Rail Services' (the definition of 'Below Rail Services' has also been amended to clarify Queensland Rail's maintenance obligations).

We consider that, after having regard to all of the factors listed in section 138(2) of the QCA Act, it is appropriate that Queensland Rail accept a more explicit responsibility for the maintenance of the network. Queensland Rail sells access to the network and recovers access charges for the maintenance of the network. Therefore, it is a principal role of Queensland Rail to maintain the network to a level that is (at least) capable of providing the contracted access rights.

This is appropriate, having regard to Queensland Rail's legitimate business interests, the interests of access seekers, access holders and the public interest (s. 138(2)(b), (d), (e), (h)). Proper maintenance of the network is also important for the safe operation of the network. It also promotes the object of Part 5 by clarifying the delineation of rights and obligations of the various stakeholders.

Summary 1.6

The 2015 DAU must specifically provide that Queensland Rail will provide below rail services and Queensland Rail's maintenance obligations in the definition of 'below rail services' must be clarified.

See clauses 1.2.2 and 7.1 in Appendix F.

1.8 Term of the undertaking

Stakeholders did not oppose our Draft Decision to accept the term of the access undertaking proposed in the 2015 DAU. The QCA refers to and adopts section 1.5 of the Draft Decision.

Summary 1.7

Queensland Rail to maintain that the term of the undertaking commences from the date of its approval and terminates on 30 June 2020.

2 NEGOTIATION PROCESS

The third party access regime in the QCA Act is underpinned by a 'negotiate–arbitrate' approach to regulation. A negotiation framework that promotes successful negotiation, and hence facilitates access, is a key component of an access undertaking.

An effective negotiation framework enables appropriate information exchange between parties, enables parties to negotiate in a timely manner and on reasonable terms, provides a transparent and predictable process for allocating limited available capacity, and protects an access provider from negotiating with parties that have no genuine interest in gaining access.

The framework seeks to balance the interests of access seekers and access holders, and the legitimate business interests of an access provider so as to promote the efficient operation of, use of, and investment in, the relevant declared infrastructure, and the public interest.

On this basis, we have proposed amendments to the negotiation process which we require before approving the 2015 DAU. Key areas where this Decision differs from the Draft Decision include:

- The requirement for an end user to provide train operations information when making an access application has been removed
- The definition of a 'renewal' now allows for an access seeker to reduce their capacity requirements.

Introduction

Part 2 of Queensland Rail's 2015 DAU proposes the framework for parties to negotiate with Queensland Rail to reach agreed terms and conditions in the form of an access agreement.²³ It also includes the process and the rules for allocating limited available capacity and outlines the responsibilities of Queensland Rail and access seekers during different stages of the negotiation process (see Figure 2).

Figure 2 Overview of the proposed negotiation framework



The key issues are summarised below in Table 2. Matters that require detailed explanation are discussed in Sections 2.1 to 2.5.

²³ The access agreement has been typically executed by Queensland Rail and a rail operator. However, the 2015 DAU proposes to allow an access agreement also be executed in a tripartite form by Queensland Rail, a rail operator and, if required, a rail operator's end customer (see Chapter 7 of this Decision).

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision
1. Information exchang	e		
Queensland Rail's information requirements should be more flexible and consistent with the QCA Act.	Accepted.	Asciano and New Hope supported the proposed amendments. New Hope and Yancoal suggested some amendments.	See Section 2.1.1.
Obligations to provide cost and pricing information to access seekers during a negotiation must be consistent with the QCA Act, including the obligation to provide appropriate capacity information.	Accepted its obligation to provide information but said that elements of that information should only be upon request, as many access applications do not require all information listed in the QCA Act.	New Hope and Glencore supported the proposed amendments but wanted greater disclosure.	See Section 2.1.2.
2. Timeframe			
A party seeking an extension to a timeframe may reasonably justify it and the other party may not unreasonably withhold its consent.	Accepted in principle.	New Hope supported the proposed amendments but suggested amendments.	See Section 2.2.
3. Refusal to provide ac	cess	1	1
Access can be refused on the grounds of concurrent requests only in the case of duplicate access requests.	Accepted in principle.	New Hope supported the proposed amendments but requested additional amendments.	See Section 2.3.
Access can be refused for passenger safety, only if Queensland Rail acts reasonably in assessing the impact and complies with non-discrimination provisions.	Disagreed on the basis that Queensland Rail's safety requirements should not be subject to dispute.	No comments.	See Section 2.3.
An access seeker can seek to extend the time it has to demonstrate it satisfies prudential requirements.	Accepted in principle.	New Hope supported the proposed amendments.	See Section 2.3.
Proposed to not approve Queensland Rail's proposal to recover costs when negotiations do not end up in an access agreement.	The DAU should reflect the principle of cost recovery.	New Hope considered Queensland Rail's right to recover costs unwarranted.	See Section 2.3.

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision
4. Competing access red	quests		
If a customer does not nominate a train operator, Queensland Rail should negotiate with all operators who are negotiating with that customer.	No comment.	New Hope supported the proposed amendments.	See Section 2.4.1.
The queuing mechanism should be consistent with 2008 and 2010 access undertaking principles; transitional provisions should address mutually exclusive applications received before approval date of this undertaking.	No comment.	New Hope supported the proposed amendments but suggested some additional amendments.	See Section 2.4.2.
5. Access renewal right	5		
Priority should be given to a renewing access holder for train services carrying coal or other bulk minerals that satisfy the conditions in the undertaking and should include transitional provisions.	Accepted in principle.	Yancoal and New Hope said renewal rights should also be granted when an access seeker chose to reduce its capacity requirement. Glencore said the definition of renewal should be broadened.	Section 2.5.
The 2015 DAU should set out the renewal application process when there is no competing access application and the access charge calculation mechanism for a non-reference access renewal.	Accepted in principle.	New Hope said the requirement for Queensland Rail to review capacity availability for an access request should be removed for renewing access seekers.	See Section 2.5.

2.1 Information exchange

The 2015 DAU sets out two forms of information exchange during the negotiation process:

- information required by Queensland Rail from an access seeker
- information provided by Queensland Rail to an access seeker.

2.1.1 Information required by Queensland Rail

Schedule B of the 2015 DAU specifies the information requirements for an access application, by either a rail operator or an end customer.

The Draft Decision proposed amendments to the information requirements in the 2015 DAU to enable a customer to apply for access rights independently of a rail operator. We also required

amendments to provide that the information requirements are within the bounds of the approved undertaking.

Stakeholders' submissions

Queensland Rail accepted the proposed amendments to Schedule B 'in principle' and gave no specific comments or amendments.²⁴ Glencore²⁵, Asciano²⁶ and New Hope²⁷ supported the Draft Decision. Yancoal²⁸, Glencore²⁹ and New Hope also proposed a number of amendments aimed primarily at distinguishing the information that should be required from an end user access seeker from that to be required from an operator access seeker.

QCA analysis and Decision

We have adopted those parts of the Draft Decision regarding Queensland Rail's proposed information requirements from an access seeker during a negotiation process, but have made some drafting amendments in Schedule B of the amended DAU in response to Yancoal's, Glencore's and New Hope's submissions.

Specifically, we agree that detailed rail operation information such as section run times, minimum dwell times and short-term storage requirements (cls. 5.1(g), (h) and (j) of Schedule B)) may not be available to an end user access seeker until an above-rail operator is contracted. For this reason, we have moved these requirements to clause 5.3 in Schedule B which applies to the information required from operator access seekers.

²⁴ Queensland Rail, sub. 26, Annexure 7: 1.

²⁵ Glencore sub. 25: 5.

²⁶ Asciano, sub. 28: 7.

²⁷ New Hope, sub. 23: 7.

²⁸ Yancoal, sub. 27: 4.

²⁹ Glencore, sub. 25: 6.

Summary 2.1

The 2015 DAU's information requirements relating to an access seeker during the negotiation process must be as follows:

(a) The information requirements and the negotiation process should enable an end customer to apply for access rights and to execute an access agreement independently of a rail operator.

See clause 2.7 and the definition of 'end user access seeker' in clause 7.1, both of which are in Appendix F. See also Schedule B to Appendix F.

(b) The information requirements for access applications should be in accordance with the undertaking.

See clauses 2.1.1 and 7.1 in Appendix F.

(c) The information that Queensland Rail requires regarding an access seeker's ability to use the access rights the access seeker is seeking, should be narrowed down.

See clauses 3 and 5.3 in Schedule B to Appendix F.

(d) Rail operation information in relation to sectional run times, minimum dwell times and short-term storage requirements should be provided by access seekers who are rail transport operators, rather than end user access seekers.

See clauses 5.1 and 5.3 in Schedule B to Appendix F.

2.1.2 Information provided by Queensland Rail

The QCA Act lists the information Queensland Rail must give an access seeker, including information about the access price (and the pricing methodology), costs (including capital, operating and maintenance) and asset values (and the asset valuation methodology). Such information could alternatively be given in the form of a reference tariff.

Queensland Rail proposed providing technical, operating and commercial information to an access seeker at different stages of the negotiation process, subject to confidentiality obligations.

In the Draft Decision we accepted Queensland Rail's proposal to specify in the undertaking the technical and operating information it will provide to an access seeker. However, we considered Queensland Rail's proposal created significant uncertainty about the provision of cost, pricing and capacity information and was inconsistent with its obligations under section 101 of the QCA Act. We therefore proposed amendments requiring Queensland Rail to provide such information to facilitate balanced negotiations.

Stakeholders' submissions

Queensland Rail³⁰ accepted the proposed amendments in principle but asked that elements of the information specified in section 101(2) of the QCA Act only be provided on request. Asciano³¹ and New Hope³² supported the proposed amendments but New Hope said that the drafting of clause 2.7.2(a)(i)(A), which specifies that Queensland Rail must supply information

³⁰ Queensland Rail, sub. 26, Annexure 7: 2.

³¹ Asciano, sub. 28: 7.

³² New Hope, sub. 23: 8.

that is reasonably required by the access seeker in accordance with section 101(1) of the QCA Act, unless that information is available elsewhere, should be amended. Glencore³³ asked that Queensland Rail's disclosure requirements be expanded.

QCA analysis and Decision

We have adopted our Draft Decision position in regard to Queensland Rail's obligation to provide information to an access seeker, but have made some drafting amendments to the DAU in response to New Hope's submission.

Specifically, we have amended clause 2.7.2(a)(i)(A) to clarify that Queensland Rail can only expect an access seeker to obtain relevant information from other sources if that information is accessible free of charge and without restriction.

We note Glencore's view that the QCA should be more specific about the information that Queensland Rail should provide to an access seeker. Given that Queensland Rail's obligations under section 101(2) include a number of the items requested by Glencore (e.g. information about the way in which the price is calculated, asset value and system capacity), and given that clause 2.7.2(a)(i)(A) (which allows an access seeker to request additional information) is subject to a reasonableness test, we do not consider such detail is necessary.

We have also rejected Queensland Rail's proposal that it only be required to supply elements of the information listed in section 101(2) on request on the basis that:

- it was unclear which aspects of the information requirement Queensland Rail was referring to; and
- Queensland Rail has not provided evidence that the information would not be required by an access seeker (or a particular type of access seeker) in order to prepare and negotiate their access request.

Restrictions on disclosure and disputes

The QCA required in QR Network's 2010 undertaking that it include a provision that prevented QR Network from restricting an access seeker or its customer from disclosing contract terms to the QCA (cl. 6.5.5(a)).

We consider that a similar restriction on Queensland Rail would promote the efficient operation and use of Queensland Rail's network and be in the interests of access seekers, access holders and their customers (s. 138(2)(a), (e), (h)). So, while such a restriction may not advance the legitimate business of Queensland Rail, we consider it is appropriate to require such a provision having regard to all the approval criteria in section 138(2).

We have therefore required amendments in clause 2.2.2(c) of Appendix F to prevent Queensland Rail from restricting the ability of access seekers, access holders or their customers to raise disputes with the QCA, or to disclose to the QCA the terms of an access agreement or a change to the number of contracted train services the description of which accords with the reference train service.

³³Glencore, sub. 25: 5.

Summary 2.2

The 2015 DAU must provide that:

(a) Queensland Rail provide cost and pricing information in an indicative access proposal, consistent with its obligations under section 101(2) of the QCA Act and an initial estimate of access charges.

See clauses 2.4.2 and 2.7.2 in Appendix F.

(b) Confidentiality obligations do not apply where parties are required to disclose information under the QCA Act.

See clauses 2.2.2 and 2.7.2(a) in Appendix F.

(c) Appropriate capacity information be provided to an access seeker. This includes providing a DTP which contains sufficient information about possibly relevant train services and also an MTP.

See clause 2.4.2 and Schedule A to Appendix F.

(d) Queensland Rail can only expect an access seeker to obtain information, which Queensland Rail can otherwise provide, if that information is obtainable by the access seeker at no cost and without restriction.

See clause 2.7.2(a)(i)(A) in Appendix F.

(e) Queensland Rail may not restrict the ability of access seekers, access holders or their customers to raise disputes with the QCA, or to disclose to the QCA the terms of an access agreement or a change to the number of contracted train services the description of which accords with the reference train service. See clause 2.2.2(c) in Appendix F.

2.2 Timeframes

Queensland Rail proposed in the 2015 DAU timeframes for different stages of the negotiation process, and provided for departures from those timeframes in certain circumstances. It also proposed penalties for access seekers that failed to meet key timeframes in the negotiation process.

In the Draft Decision we accepted Queensland Rail's proposed timeframes for the different stages of the negotiation process. However, we considered Queensland Rail's proposal of allowing an extension to some of those timeframes did not provide an appropriate balance between the interests of Queensland Rail and access seekers. We therefore proposed amendments in Part 2 of the 2015 DAU to adequately balance the interests of all parties.

Stakeholders' submissions

Queensland Rail³⁴ accepted the amendments proposed in the Draft Decision in principle and did not propose any further changes. New Hope³⁵ suggested that a clause be added to oblige Queensland Rail to justify to other access seekers in a queue how long an access negotiation was taking, if a queuing access seeker considered the negotiation period excessive.

³⁴ Queensland Rail, sub. 26, Annexure 7: 3.

³⁵ New Hope, sub. 23: 8.

QCA analysis and Decision

We have adopted our Draft Decision in regard to the timeframes relevant to each different stage of the negotiation process (see section 2.2 of the Draft Decision).

We do not accept New Hope's suggestion that Queensland Rail should be obliged to justify to other access seekers in a queue how long a negotiation is taking. The undertaking contains a number of provisions that require negotiations to be undertaken within a given timeframe, including clause 2.5.1 that allows Queensland Rail to terminate negotiations with an access seeker if it has not responded to an indicative access proposal within three months. It may not advance the legitimate business interests of Queensland Rail to extend negotiations unless it considers a successful outcome highly likely. Moreover, we do not consider it in Queensland Rail's legitimate business interests for such a burden to be imposed (s. 138(2)(b)).

Summary 2.3

The 2015 DAU must provide that if the party seeking an extension to a timeframe (in the contexts of providing an indicative access proposal; an intent to negotiate; a negotiation period; and the execution of an access agreement) the other party cannot unreasonably withhold its consent to the extension request.

See clauses 2.3.2 (deletion), 2.4.1, 2.5.2, 2.5.3, 2.7.1(b)(ii)(c) and clause 2.9.5 in Appendix F.

2.3 Refusal to provide access

In the 2015 DAU, Queensland Rail proposed that it be able to:

- cease negotiations in certain circumstances
- recover its costs where an access application did not result in an access agreement.

The Draft Decision accepted that Queensland Rail should be entitled to refuse access in certain circumstances to protect its legitimate business interests. However, we had concerns with the discretion the 2015 DAU gave Queensland Rail in refusing access in some of the proposed circumstances. We proposed amendments accordingly, to adequately balance Queensland Rail's legitimate business interests and the interests of access seekers.

With respect to cost recovery, we found Queensland Rail's proposal deficient, as it was not transparent and it was unclear if the costs recovered would be limited to the efficient incremental costs of providing the service. We invited Queensland Rail to submit an alternative proposal that sought to recover efficient incremental costs and avoids double dipping.

Stakeholders' submissions

Queensland Rail accepted in principle that its ability to refuse to provide access on the grounds of concurrent access requests should be limited to duplicate access requests³⁶, and the requirement that an access seeker be able to extend the period to demonstrate it has satisfied the prudential requirements.³⁷ However, Queensland Rail rejected the amendment which

³⁶ Queensland Rail, sub. 26, Annexure 7: 3.

³⁷ Queensland Rail, sub. 26, Annexure 7: 2.

required it to 'act reasonably' when rejecting an access request on passenger safety grounds.³⁸ Queensland Rail did not provide an alternative cost recovery proposal.³⁹

New Hope⁴⁰ said that an access seeker should be able to justify why the separate requests were not duplicates before Queensland Rail ceased negotiations on one of them and that Queensland Rail should be required to explain why it considered a request to be a duplicate.

QCA analysis and Decision

We have adopted our Draft Decision position in regard to the 2015 DAU's refusal to provide access provisions (see section 2.3 of the Draft Decision).

The QCA does not accept Queensland Rail's submission that including the words 'acting reasonably' may result in Queensland Rail's rail safety requirements being 'watered down, disputed or replaced'. Including these words is intended to provide access seekers with clarity on how safety regulations are imposed, given their application can materially affect the access seeker's operations.

We note that Queensland Rail has not provided the redrafting of its cost recovery provisions as requested in our Draft Decision, but has simply stated that the QCA has accepted the principle that when access negotiations do not result in an access agreement being finalised, Queensland Rail has the right to recover its costs from an access seeker. This statement by Queensland Rail is not correct as it presumes that we would automatically accept cost recovery in these cases.

The Draft Decision deleted the cost recovery provisions that had been drafted by Queensland Rail on the basis that they did not make clear that only the efficient incremental cost of the negotiations could be recovered, and that those provisions could potentially allow double dipping.

Queensland Rail has not made a compelling case on the matters raised in the Draft Decision. Moreover, there are likely to be significant difficulties in establishing a cost-recovery framework that transparently demonstrates that Queensland Rail is only recovering efficient costs and which provides a robust and transparent mechanism to address concerns about potential double-dipping.

An inappropriately constructed framework carries risks, including that it would impact on the ability of access seekers to seek access to the declared service, which would be inappropriate having regard to section 138(2)(a), (d), (e) and (h). The QCA is therefore not prepared to approve a cost recovery mechanism.

We have not applied New Hope's proposed amendment that would require Queensland Rail to explain why it considers a request to be duplicate when it ceases to negotiate with an access seeker on this basis. We note that clause 2.8.1(a)(iv)(A) already provides that Queensland Rail must provide reasons to an access seeker if it intends to cease negotiations on this basis, and clause 2.8.1(a)(iv)(B) provides an opportunity for the access seeker to respond.

We consider that this gives the access seeker sufficient information before negotiations actually cease with regard to duplicate requests. We have, however, amended clause 2.8.1(a)(iv)(B) to make it clear that if the access seeker is able to demonstrate that the requests are not duplicate, Queensland Rail should proceed with the concurrent requests.

³⁸ Queensland Rail, sub. 26, Annexure 7: 3.

³⁹ Queensland Rail, sub. 26, Annexure 7: 4.

⁴⁰ New Hope, sub. 23: 8.

Summary 2.4

The 2015 DAU must provide as follows:

(a) Refusal to provide access on the grounds of concurrent requests should be limited to duplicate access requests, provided the access seeker is given a reasonable explanation and a reasonable opportunity to respond before Queensland Rail refuses to deal with the access seeker in respect of those duplicate requests.

See clause 2.8.1 in Appendix F.

- (b) Refusal to provide access on the grounds of passenger safety should be subject to Queensland Rail acting reasonably in assessing the impact on passenger safety and complying with the non-discriminatory provisions. See clause 2.8.2 in Appendix F.
- (c) An access seeker should be able to seek to extend the time it has to demonstrate satisfaction of the prudential requirements by reasonably justifying the extension, and Queensland Rail, should act reasonably in considering whether to agree to such an extension. Queensland Rail must also act reasonably in assessing whether an access seeker satisfies the prudential requirements.

See clause 2.8.3 in Appendix F.

(d) The cost recovery proposal should be deleted.

2.4 Competing access requests

Generally, competing access seekers are categorised into two groups—that is, access seekers:

- seeking access rights in respect of competitive tendering—such as train operators seeking access rights to serve the same customer for the same haulage task; or
- seeking access rights in respect of mutually exclusive paths—such as train operators seeking
 access rights to provide different haulage tasks when there is insufficient capacity to meet
 their access requirements.

These two categories of access seekers are considered in turn below.

2.4.1 Competitive tendering

The 2015 DAU sets out how Queensland Rail will deal with competing access seekers that seek access rights for the same traffic task.

The Draft Decision accepted Queensland Rail's proposal to explicitly outline the process it will follow for dealing with access seekers that compete for the same haulage task, but required that the process be amended to allow for potential train operators to negotiate the price and other terms of access with Queensland Rail, to present this to the end customer, and to allow the end customer to select the operator(s) they wish to engage.

Queensland Rail made no comment and New Hope supported the Draft Decision.⁴¹

⁴¹ New Hope, sub. 23: 9.

Our analysis and position remain the same as set out in the Draft Decision. Therefore, we have adopted the Draft Decision in relation to dealing with competing access requests (see section 2.4.1 of the Draft Decision).

Summary 2.5

On dealing with competing access requests, the 2015 DAU must provide that if a customer does not nominate a train operator as its preferred operator, Queensland Rail should negotiate with all train operators who are negotiating a potential haulage agreement with that customer, and should offer each an access price and the terms and conditions of access. Queensland Rail should then negotiate and execute an access agreement with the train operator who reasonably demonstrates that it will be appointed by the relevant customer.

See clause 2.6 in Appendix F.

2.4.2 Competition for mutually exclusive paths

The 2015 DAU sets out how Queensland Rail will deal with multiple access seekers seeking access rights for different traffic tasks when there is insufficient available capacity to fulfil their access requests—that is, when access applications are mutually exclusive.

The 2015 DAU proposes a queuing mechanism to determine which access seeker will be allocated access rights. The order of a queue will initially be based on the access application date; however, Queensland Rail may change that order in various circumstances.

In our Draft Decision we accepted Queensland Rail's proposal to use a queuing mechanism for granting access rights to mutually exclusive access applications, but amended its proposal to more adequately balance the legitimate business interests of Queensland Rail and the interests of access seekers.

Queensland Rail did not comment on the proposed amendments, while New Hope⁴² supported the proposed amendments, but asked that their application to the West Moreton network be clarified to ensure that in the West Moreton system the queue could only be reordered to place an access seeker with a term of at least 10 years ahead of an application of less than 10 years.

We accept New Hope's comment that clarity is required in how the order of multiple applications for coal-carrying train services on the West Moreton network in a queue may be changed. The intent of the Draft Decision was that for such train services, a queue could only be reordered to place an application that sought a term of at least 10 years ahead of another application that sought a term of less than 10 years. We have clarified that in our amended drafting of clause 2.9.2(i)(iv) in the 2015 DAU.

⁴² New Hope, sub. 23: 10.

Summary 2.6

For mutually exclusive access applications, the 2015 DAU must provide as follows:

(a) The queuing mechanism for allocating limited available capacity to mutually exclusive access applications be consistent with the principles reflected in the 2010 access undertaking.

See clause 2.9.2 in Appendix F.

(b) Transitional provisions be included in the undertaking to deal with mutually exclusive access applications received before the approval date of this undertaking.

See clause 6.4 in Appendix F.

2.5 Access renewal rights

The 2015 DAU sets out the process for allocating access rights in situations where a queue includes an application from an existing access holder seeking to renew its access rights—that is, where a renewing access holder competes with a new access seeker for the same access rights.

The 2015 DAU enables an end customer or its nominee to apply for renewal rights. The 2015 DAU proposes that the renewing access holder 'may' get priority over a new access seeker in executing an access agreement, if the access rights being renewed are for coal-carrying or other bulk-mineral-carrying train services and satisfy certain other conditions.

In the Draft Decision we proposed amendments to the process for allocating access rights to better balance the legitimate business interests of Queensland Rail and the interests of access seekers/holders.

Stakeholders' submissions

Queensland Rail⁴³ accepted our proposed amendments in principle. New Hope⁴⁴ supported our proposed amendments but proposed a number of amendments aimed at simplifying the access renewal process and providing renewing access holders with more certainty. Glencore⁴⁵ and Yancoal⁴⁶ said that clause 2.9.3(b) and the definition of renewals were unduly narrow in their application and did not provide sufficient protection for 'renewals' for customers with sunk costs. Yancoal⁴⁷ also said that Queensland Rail should not be permitted to advise a renewing customer that there was insufficient capacity and that it should be possible to seek a renewal of part of an access right held under an existing access agreement. Glencore separately asked that renewal rights should extend to intermodal operations.⁴⁸

QCA analysis and Decision

We have broadly adopted the Draft Decision in regard to access renewal rights but we have made a number of additional amendments to those proposed in the Draft Decision.

⁴³ Queensland Rail, sub. 26, Annexure 7: 5.

⁴⁴ New Hope, sub. 23: 10.

⁴⁵ Glencore, sub. 35: 3.

⁴⁶ Yancoal, sub. 27: 4.

⁴⁷ Yancoal, sub. 35: 3.

⁴⁸ Glencore, sub. 25: 3.

New Hope has proposed three amendments to the access right renewal provisions which seek to:

- remove the ability for Queensland Rail to advise that insufficient capacity exists;
- allow a renewing access right holder to elect to renew only part of an existing access right; and
- require Queensland Rail to notify an access holder of the need to renew.

We agree that, all other things being equal, Queensland Rail should have the capacity to accommodate a renewing access request, and that therefore it is unnecessary to require Queensland Rail to provide a capacity analysis (cls. 2.4.2(b) and 2.7.2(a)(vii)) and information on whether it is willing to fund any extensions required (cl. 2.7.2(b)). However, while we have made amendments (see cl. 2.7.2(e)) to this effect, we note that, to the extent that a renewal requires asset replacements to maintain existing capacity, the cost of these asset replacements would be expected to be reflected in the pricing of a renewing access seeker, under the terms of the renewal pricing framework.

Similarly, we have amended the definition of a renewal in response to comments from New Hope, Yancoal and Glencore, to make it clear that if a renewing access holder elects to renew only a portion of its access rights, this should be treated as a renewal. However, we note that, to the extent that Queensland Rail can show that this reduction results in a material change in cost or risk, there is scope for the access price to reflect this change.

We have not applied Glencore's suggestion that an application which includes a change to an origin or destination that is not likely to impact on other access holders or access seekers should still be treated as a renewal. This is because provisions associated with renewals are primarily designed to protect an access seeker's sunk costs and these protections do not necessarily apply if an access seeker chooses to switch production from one operation to a similar operation even if that operation is nearby. In making this decision, we have had regard to our final decision on UT4 for Aurizon Network⁴⁹ which includes a clause which allows for changes to origin and/or destination when renewing an access agreement. However, we note that the clause is tailored to the specific characteristics of the Aurizon network which are not currently features of the Queensland Rail network. For example, a change in terminal at either Gladstone or Dalrymple Bay ports could be classified as a change in destination but would effectively require the same paths—this scenario does not currently occur on the Queensland Rail network.

We also do not consider it necessary to extend the renewal provisions to cover intermodal services (as requested by Glencore). We note that if an application for intermodal services was not first in the queue and other existing or renewing services were dependent on that intermodal service for their continued operation, Queensland Rail may prioritise that application with reference to clause 2.9.2(j)(ii). This is because clause 2.9.2(j)(ii) allows a queue to be reordered on the basis of the effects that an application for intermodal services may have on contribution to common cost including 'revenue that would reasonably be expected to reduce or be eliminated as a consequence of Queensland Rail not providing access for that train service'.

We have not applied New Hope's proposal that Queensland Rail be required to notify an access holder of the need to renew. We consider that the management of the access contracts would

⁴⁹ QCA 2016b, Appendix A, cl. 7.3.
be part of the normal business requirements of an access holder and we do not consider it advances Queensland Rail's legitimate business interests for such a burden to be imposed.

Summary 2.7

The 2015 DAU must provide as follows:

- (a) Queensland Rail should give priority to a renewing access holder for coalcarrying or other bulk-mineral-carrying train services that satisfy the conditions in the undertaking (i.e. those relating to contract period, nature of access rights sought and timeframes for submitting renewal application).
- (b) The access rights renewal process should reflect the amendments summarised in Table 2.7 of the Draft Decision and discussed above (e.g. Queensland Rail's obligation to offer terms consistent with the undertaking and the standard access agreement, unless the parties agree otherwise) including setting out the process that will apply to a renewal application when there is no competing access application consistent with the process noted in the Draft Decision and the provisions for the calculation of access charges for a renewal access seeker where no reference tariff applies, which requires access charges to be calculated consistently with the renewal pricing rules as required in Part 3 of Appendix F.
- (c) In relation to a renewing access request, Queensland Rail does not need to provide a capacity analysis or information regarding extensions.
- (d) The definition of 'renewal' should be amended so that a renewal access seeker can seek less train paths than its existing access agreement and, if the renewal access agreement is otherwise substantially equivalent to the expiring access agreement it will still be treated as a 'renewal'.
- (e) For the purposes of giving priority rights, transitional provisions are included in the undertaking to deal with renewal applications for coal-carrying or other bulk-mineral-carrying train services received before the approval date of this undertaking for which negotiations have not concluded.

See the definition of renewal and renewal access seeker and clauses 2.4.2, 2.7.2, 2.9.3, 2.9.4(a), 6.4 and 7.1 in Appendix F.

3 PRICING RULES

Part 3 of Queensland Rail's 2015 DAU sets out principles for developing access charges.

In considering Queensland Rail's proposed 'pricing principles' for inclusion in the access undertaking the QCA has had regard to the criteria in s. 138(2) of the QCA Act.

We consider the proposed 'pricing principles' require amendments to provide greater clarity and balance in the way Queensland Rail sets access charges. These are discussed below.

This Decision differs from the Draft Decision in a few key areas. This Decision:

- allows for changes in cost or risk, as well as differences in cost or risk, to be reflected in access charges;
- adopts a renewal pricing mechanism that allows for variations to renewing train services if variations are due to supply chain efficiency improvements;
- renames the 'pricing principles' as the 'pricing rules' to avoid confusion with the pricing principles in the QCA Act; and
- *further increases obligations on Queensland Rail to provide details of its pricing methodology.*

Introduction

The pricing rules⁵⁰ in this chapter are designed to guide the development of access charges by Queensland Rail. The pricing rules do not bind the setting of a reference tariff by the QCA, which must be set having regard to section 138(2) of the QCA Act.⁵¹

These pricing rules are also relevant if either Queensland Rail or an access seeker brings a dispute to the QCA for an access determination. Relevantly, as the pricing rules form part of the access undertaking, the QCA cannot make an access determination that is inconsistent with them.

The pricing rules are of particular use to Queensland Rail and access seekers in negotiating tariffs for non-reference train services.⁵² They also provide constraints on the setting of access charges by Queensland Rail.

For clarity, the pricing rules do not constrain the operation of the pricing principles in the QCA Act (s. 168A). Moreover, section 168A is one of a range of matters that the QCA must have regard to when deciding an access dispute. For the purposes of the 2015 DAU, the QCA considers that by providing certainty and clarity in relation to the setting of access charges, the pricing rules discussed in this chapter, and contained in Part 3 of the DAU, are appropriate

⁵⁰ Queensland Rail's 2015 DAU and our Draft Decision used the term 'pricing principles'. To avoid confusion with the pricing principles in the QCA Act (i.e. s. 168A), this Decision uses the term 'pricing rules' to reflect that these are rules within the DAU, and not requirements under the QCA Act. The distinction between s. 168A and what were previously called 'pricing principles' in the DAU is acknowledged by Queensland Rail (Queensland Rail, sub. 26: 55).

⁵¹ We have clarified this in our mark-up to the 2015 DAU (Appendix F) including by amendments to cls. 3.0, 3.5(c) and the definition of 'reference tariff'.

⁵² For example, the pricing rules would apply to train services on the Mount Isa line. They would also apply to coal-carrying train services on the West Moreton or Metropolitan networks, but mainly to reflect the cost or risk difference where those train services vary from the reference train service.

having regard to the public interest (s. 138(2)(d)) and the interests of persons who may seek access as well as access holders (s. 138(2)(e), (h)).

In our 2015 Draft Decision, we indicated a number of amendments that we required before we would accept Part 3 of the 2015 DAU as appropriate. Queensland Rail has rejected or questioned the majority of our required amendments.

We have broadly adopted our Draft Decision views on our approach to Part 3 of Queensland Rail's 2015 DAU on pricing rules.

The key issues are summarised below in Table 3. Matters that require a more detailed explanation are discussed in Sections 3.1 to 3.8.

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision
1. Hierarchy of pricing rules	5		1
Revenue adequacy should not be given primacy ahead of other pricing rules.	Disagreed; the proposed ranking potentially obliged Queensland Rail to set a price that did not achieve revenue adequacy.	Asciano and New Hope supported the Draft Decision; Aurizon supported it in principle but questioned whether the proposed ranking could conflict with the RAB review provisions.	See Section 3.1.
2. Revenue adequacy			
The definition of revenue adequacy should remove the reference to return on assets.	Disagreed.	Queensland Rail should not achieve revenue adequacy using an asset valuation approach that provided windfall gains/recovery of inefficient costs.	See Section 3.2.
3. Limits on price differenti	ation (within the same marke	et)	
Queensland Rail may differentiate prices in relation to similar train services within the same market only in certain limited circumstances.	Disagreed.	Stakeholders largely supported our Draft Decision, but some requested additional amendments.	See Section 3.3.
4. Pricing and revenue limit	ts		
Queensland Rail may set prices below the floor revenue limit subject to QCA approval.	Had concerns about the QCA's drafting.	Stakeholders agreed with our amendments except Glencore who had concerns regarding the calculation of the ceiling revenue limit.	See Section 3.4.
5. Asset valuation methodo	ology for negotiating access ch	narges	
Flexibility should be allowed on methodology for asset	Disagreed. Allowing flexibility in determining the asset valuation	New Hope, Yancoal and Aurizon supported	See Section 3.5.

Table 3 Pricing rules in the 2015 DAU

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision
valuations.	approach was inconsistent with regulatory practice.	the Draft Decision.	
6. Pricing for access rights a	t renewal (non-reference tar	iff services)	
Queensland Rail should provide more certainty to renewing access seekers in relation to pricing.	Disagreed.	Stakeholders generally agreed with our Draft Decision. Glencore had concerns about locking in existing methodologies and proposed changes to the renewal provisions.	See Section 3.6.
7. Rate of return			
Weighted average cost of capital (WACC) of 6.93%; the final dates for setting the risk free rate and debt margin to be determined.	Said risk-free rates were higher in previous periods.	Yancoal and New Hope supported the proposed methodology for setting the WACC.	See Section 3.7.
8. Take-or-pay arrangemen	ts		
Queensland Rail may apply a pre-determined take-or-pay rate for reference tariff users, but not for non- reference tariff users.	No comment on this issue.	Glencore said that there should be a pre- determined rate for non-reference tariff take-or-pay. Other stakeholders were largely concerned with the rate of take-or-pay in relation to reference tariff users (discussed in Chapter 8 of this Decision).	See Section 3.8.

3.1 Hierarchy of pricing rules

Previous access undertakings that applied to Queensland Rail's declared service included a hierarchy of pricing rules. These rules were designed to address conflicts that may arise when applying the pricing rules in an access undertaking to develop access charges. The hierarchy was explicitly included in the 2008 access undertaking and implicitly in the 2001 access undertaking.

To address any conflicts between the pricing rules in the 2015 DAU, the Draft Decision proposed the 2015 DAU include a hierarchy of pricing rules as follows:

- (1) limits on price differentiation as between access seekers/holders (within a market) in accordance with clause 3.3 of the DAU;
- (2) pricing limits (between different markets) in accordance with clause 3.2 of the DAU;
- (3) network utilisation in accordance with clause 3.1.2 of the DAU; and
- (4) revenue adequacy in accordance with clause 3.1.1 of the DAU.

Stakeholders' submissions

Queensland Rail did not support our Draft Decision, on the basis that ranking revenue adequacy last 'obliges Queensland Rail to potentially set a price that does not achieve revenue adequacy' and this was fundamentally inconsistent with the pricing principles set out in section 168A(a) of the QCA Act.⁵³ In Queensland Rail's opinion, revenue adequacy should be first in the hierarchy.

Glencore,⁵⁴ Asciano⁵⁵ and New Hope⁵⁶ supported the proposed hierarchy, while Aurizon supported elevating the 'limits on price differentiation' above revenue adequacy. However, Aurizon said that the elevation of pricing limits above revenue adequacy was potentially inconsistent with the RAB review provisions (in Schedule E). These provisions state that existing asset values cannot be reduced unless it can be shown that prices would reduce demand or the asset could be bypassed.⁵⁷

QCA analysis and Decision

We have adopted our Draft Decision position on the hierarchy of pricing rules⁵⁸ for access pricing except that we have changed the name from 'pricing principles' to 'pricing rules' as noted above (see section 3.1 of the Draft Decision).

We do not accept Queensland Rail's arguments about the pricing rules in the context of section 168A.

One of the pricing principles is that the price of access to a service should generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved. But the considerations relevant to whether to approve the 2015 DAU are broader than just the pricing principles. We do not agree that the QCA Act precludes the inclusion, in an access undertaking, of pricing rules with a hierarchy which could have the effect of requiring Queensland Rail to accept a price that would not generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved.

A hierarchy of pricing rules in the order outlined provides certainty to access seekers and access holders about which rule will prevail in the event of a conflict. Relevantly, we do not support placing revenue adequacy first in the hierarchy as it may enable Queensland Rail to achieve revenue adequacy without observing the constraints on price differentiation (i.e. by unfairly discriminating between access seekers to maximise revenues). We also consider that the price limits should take precedence over revenue adequacy. This is to preclude Queensland Rail from charging more than the stand-alone cost of a service in order to achieve revenue adequacy.

We disagree with Aurizon's point that the proposed ranking of the pricing rules was potentially inconsistent with the RAB review provisions in Schedule E. In any event we note that the RAB review provisions would only apply when there is a regulated access charge and changes to

⁵³ Queensland Rail, sub. 26: 55.

⁵⁴ Glencore, sub. 30: 4.

⁵⁵ Asciano, sub. 28: 11.

⁵⁶ New Hope, sub. 23: 11.

⁵⁷ Aurizon, sub. 20: 22.

⁵⁸ We have adopted the term 'pricing rules' to avoid confusion with the pricing principles in s. 168A of the QCA Act.

regulated access charges must be made with reference to section 138(2) of the QCA Act and not the pricing rules.

Summary 3.1

The 2015 DAU must:

- (a) rename the 'pricing principles' in the DAU to the 'pricing rules'
- (b) provide for a hierarchy of pricing rules in the following order:
 - (i) limits on price differentiation
 - (ii) pricing limits
 - (iii) network utilisation
 - (iv) revenue adequacy.

See Part 3 in Appendix F.

3.2 Revenue adequacy

In the Draft Decision we rejected Queensland Rail's definition of revenue adequacy on the basis that it was seeking a minimum return not just on investments, but also on assets, which it proposed to value universally using its preferred DORC valuation approach.⁵⁹

Stakeholders' submissions

Queensland Rail said the QCA's proposed drafting had the effect of completely rewriting and changing the meaning of what is required by section 168A(a) of the QCA Act, by:

- (a) providing Queensland Rail with a revenue that was no more than sufficient to meet the efficient costs of providing access when the wording of the Act specified that the expected revenue should be 'at least enough' to meet the efficient costs
- (b) allowing the QCA to make unspecified, unqualified and non-reviewable 'adjustments' to either Queensland Rail's revenue or its efficient costs
- (c) including the return on investment referred to in section 168A(a) of the QCA Act as part of the efficient costs and therefore, as drafted, subject to 'adjustment', rather than a separate return as required by section 168A(a).⁶⁰

New Hope supported the QCA's proposed amendment.⁶¹

QCA analysis and Decision

The language of Queensland Rail's drafting of the revenue adequacy provision departs in a number of respects from the language of section 168A(a). We do not consider these departures are appropriate in the circumstances.

The language of Queensland Rail's draft referred to Queensland Rail being 'entitled to earn' revenue. We have adopted drafting which refers to 'expected revenue'. This is more consistent with the language of 168A(a) and is appropriate because the risk of recovering less revenue

⁵⁹ QCA 2015: 50.

⁶⁰ Queensland Rail, sub. 26: 11.

⁶¹ New Hope, sub. 23: 12.

than expected is something that is factored into the calculation of the appropriate 'return on investment commensurate with the regulatory and commercial risks involved'. Further, we do not accept that Queensland Rail has an 'entitlement' to such an amount.

To address these points, we have amended the wording of the revenue adequacy provisions (cl. 3.1.1). Rather than stating that Queensland Rail is 'entitled to earn revenue' the clause now provides for Queensland Rail to set access charges based on the revenue it is expected to generate (and the TSC payments that it is expected to receive). The expected revenue should be at least enough to meet the efficient costs of providing access and should include a return on investment commensurate with the regulatory and commercial risks involved.

We have also rephrased the reference to how Queensland Rail might allocate any excess revenue so that it is consistent with the forward-looking nature of this clause.

Further, we have made amendments to clarify when Part 3 applies and when the reference tariff applies and to which train services it applies (cl. 3.0 of Appendix F) (see also Chapters 8 and 10).

Summary 3.2

The 2015 DAU must provide that the revenue adequacy pricing rule is subject to the hierarchy of pricing rules and so that access charges and TSC payments are set to generate expected revenue that is at least enough to meet the efficient costs of providing access and includes a return on investment commensurate with the regulatory and commercial risks involved. Also, where Queensland Rail is expected to earn excess revenues Queensland Rail may seek to reduce TSC payments rather than access charges. We have also clarified how and when Part 3 applies.

See Schedule D, clause 2(b) of 2015 DAU (amended and moved to clause 3.0) and clauses 3.0, 3.1.1, 3.2.3, 3.5 and the definition of 'reference tariff' in Appendix F.

3.3 Limits on price differentiation

Queensland Rail proposed to be able to vary the methodology, rates and other inputs for calculating access charges between access seekers in accordance with the price differentiation limits in clause 3.3 of the 2015 DAU.

Queensland Rail's proposed provisions allow for Queensland Rail to apply different access charge methodologies to different access seekers according to whether or not a reference tariff applies to the relevant access seeker.

We considered in the Draft Decision that the 2015 DAU provided Queensland Rail with excessively broad discretion to engage in price differentiation within markets where a reference tariff did not apply.

Stakeholders' submissions

Queensland Rail did not support the Draft Decision and maintained its original proposal.⁶²

⁶² Queensland Rail, sub. 26: 56–62.

Glencore said that it had major concerns about the lack of transparency in relation to Queensland Rail's pricing and the excessively high prices that Queensland Rail was able to achieve due to its monopoly power.⁶³

New Hope supported the Draft Decision but said that the QCA should have the power to require the development of new reference tariffs.⁶⁴ Asciano, Aurizon and Yancoal supported the Draft Decision.⁶⁵ However, Aurizon said that the limits on price differentiation should be amended slightly to provide for competitive neutrality where an access seeker funded an extension.⁶⁶

QCA analysis and Decision

The QCA considers that the 2015 DAU provides Queensland Rail with an excessively broad discretion to differentiate access charges within markets where a reference tariff does not apply. This could possibly lead to improper differentiation, which is not appropriate having regard to section 168C.⁶⁷ We consider that this level of discretion is not appropriate, with regard to each of the factors listed in section 138(2).

Queensland Rail's legitimate business interests (including the ability for efficient price differentiation) are one of the factors to be considered by the QCA pursuant to section 138(2) of the QCA Act. However, unbalanced discretion for Queensland Rail in relation to access charges may distort competition in related markets through inefficient price differentiation between different train operators or end markets (s. 138(2)(a) and (d).

We require the 2015 DAU to be amended to set out more appropriately the circumstances in which price differentiation is permitted, including when there are changes in TSC payments through each individual access agreement. The 2015 DAU should also provide that price differentiation breaches are able to be remedied when they occur.

In this regard, we have adopted our Draft Decision position, except for a number of additional changes discussed below.

Permitted access charge differentiation within a non-reference tariff market

We require Queensland Rail to amend its 2015 DAU such that differentiating access charges between access seekers within the same non-reference-tariff market is limited to situations where:

- there are differences or changes in the cost or risk to Queensland Rail of providing the below-rail service
- available capacity is demonstrably insufficient to meet all access seekers' requests, permitting the quotation of a maximum access charge (as provided by cl. 3.1.2 of the 2015 DAU).

We consider it would be inappropriate if Queensland Rail could also price discriminate within a market (as Queensland Rail proposed) on the basis of:

⁶³ Glencore, sub. 25: 1.

⁶⁴ New Hope, sub. 23: 12–13.

⁶⁵ Yancoal, sub. 27: 1; Aurizon, sub. 20: 19, Asciano, sub. 28: 11.

⁶⁶ Aurizon, sub. 20: 20.

⁶⁷ Section 168A(c) permits price discrimination when 'it aids efficiency'.

- Queensland Rail being unable to commercially provide access at the current access charges (for example, due to changes in the TSC payments)⁶⁸; or
- changes in circumstances that have a material effect on the ability of access holders to pay access charges.⁶⁹

Differences and changes in cost and risk

After reviewing the stakeholder submissions on these matters, we remain of the view that the above criteria for access charge differentiation could result in Queensland Rail being able to differentiate between access seekers (or holders) in the same market due to a broad and overly discretionary range of commercial matters which may be relevant to Queensland Rail but which may not necessarily be relevant to the provision of access. This type of price differentiation is not efficient, as anticipated in section 168A(b) of the QCA Act.

In this regard we have adopted our Draft Decision position.

However, we agree with Queensland Rail's submissions that for non-reference-tariff train services there should be a mechanism by which there can be a higher degree of flexibility to reopen access charges under access agreements, to allow for changes in cost or risk over the life of an access agreement.⁷⁰

We consider that the provisions in the 2015 DAU which allow Queensland Rail to adjust access charges during the life of an access agreement (for differences in costs or risks to Queensland Rail of providing access) are in accordance with the same principle which allows differentiation in access charges at the time of contracting (for differences in costs or risks to Queensland Rail of providing access) in non-reference tariff markets. Therefore, in this Decision we have reinstated the relevant parts of clauses 3.3 and 3.6 which allow for differentiation over time. If those costs or risks are relevant, reasonable and related to the provision of access, we consider that they form a legitimate basis for price discrimination.

Given the increased obligations on Queensland Rail to provide information in relation to its use of these provisions, our changes provide that if an access seeker or access holder feels aggrieved about the use of these clauses, they can initiate a dispute. We consider that these changes provide an appropriate balance between Queensland Rail's interests in being able to price flexibly to adjust for cost and risk, and the interests of access seekers and access holders in having more transparent access charges (s. 138(2)(a), (b), (d), (e), (g), (h)).

New Hope said that the QCA should be able to require Queensland Rail to develop a new reference tariff.⁷¹ We do not accept that it is appropriate to include this power within the 2015 DAU. This is because we have substantially increased the ability for access seekers to negotiate efficiency improvements in their relevant access agreements, both at the initial negotiation stage and during the term of the access agreement (see also Chapter 7 of this Decision).

If an access seeker or holder considers that a new reference tariff should be developed, it is open to those parties to bring an access dispute on the matter under Division 5, Part 5 of the QCA Act and the QCA may consider the appropriateness of implementing a new reference tariff. Alternatively, Queensland Rail can develop a new reference tariff which will be subject to the QCA's approval.

⁶⁸ Queensland Rail, sub. 26: 58.

⁶⁹ Queensland Rail, sub. 26: 60.

⁷⁰ Queensland Rail, sub. 26: 58, 68.

⁷¹ New Hope, sub. 23: 12, 14.

To restrict the application of the proposed clauses to appropriate cases, we have also included amendments in addition to those we proposed in the Draft Decision. These further amendments to clause 3.6 are similar to the Indicative Access Proposal information provisions,⁷² in that they require Queensland Rail to detail how it is applying a differentiation in access charges. This is appropriate having regard to Queensland Rail's legitimate business interests (s. 138(2)(b)), the pricing principles (s. 168A), as well as being in the interests of access seekers and access holders by decreasing information asymmetry (s. 138(2)(e), (h)).

Quoting maximum access charges

Queensland Rail included in its 2015 DAU a clause (cl. 3.1.2(b)(iii)) which entitles Queensland Rail to charge less than a maximum access charge despite quoting a maximum access charge in situations where capacity is constrained. The clause also states that for the purposes of determining the ceiling revenue limit, the access charge for the relevant access seeker is assumed to be the maximum access charge.

We consider that this is a substantially reasonable provision, given that, if capacity is constrained, Queensland Rail should have the right to seek to maximise its access charges (subject to the pricing limits), while also having the choice to charge less than the maximum.

However, we also agree with New Hope that the wording of clause 3.1.2 which entitles Queensland Rail to price at the maximum access charge if the network is capacity constrained should be amended.

The current drafting states that Queensland Rail can price at maximum access charge if available capacity is 'potentially' insufficient to satisfy requests for access rights. New Hope has asked that this be strengthened so that the clause will apply only when available capacity is 'demonstrably' insufficient. We consider this amendment would provide that access seekers receive information that is important to the negotiation process in a timely manner. Importantly, this information would be expected to be readily available; therefore, we consider that the strengthening of the provision would not be onerous to Queensland Rail.

Queensland Rail wanted this clause to also be applicable where a non-reference-tariff user is renewing under the renewal pricing provisions. We do not consider that this provision is appropriate in the renewal pricing circumstances. The QCA's required renewal mechanism and pricing provisions are intended to provide price and access security to access holders who have substantial sunk costs and require rail access to avoid these assets becoming stranded.

Because of the operation of the renewal mechanisms, any capacity constraint will not be caused by the renewing access seeker which (if it is a renewing access seeker) has its relevant access rights preserved. The network utilisation pricing provisions can apply to those access seekers who are seeking access to the remaining train paths in the relevant system, rather than to the renewing access seeker.

Material changes in circumstances

Queensland Rail said that the ability to differentiate access charges between access holders (during the life of an access agreement) due to the prevailing economic circumstances (cl. 3.3(b)(ii)(B)(3) of the 2015 DAU) was in the interests of access seekers and access holders on the basis that it would allow Queensland Rail to adjust access charges down if the economic

⁷² See cl. 2.4.2 of the QCA's mark-ups to the 2015 DAU in Schedule F of this Decision, and Chapter 2 of this Decision.

circumstances prevailing at the time when the access charges are being determined had materially deteriorated.⁷³

Despite Queensland Rail's commentary on the operation of the relevant clause, its drafting of the clause is sufficiently open to allow precisely the anti-competitive behaviour which we consider to be inappropriate. That is, Queensland Rail could use the provision to assess, at its discretion, an access holder or access seeker's ability to pay and then increase charges to those users that Queensland Rail considered were more profitable.⁷⁴

Although it may advance the interests of Queensland Rail to apply higher access charges to more profitable network users, the use of the clause to differentiate in this manner would be an inappropriate use of monopoly power and would be sufficiently detrimental to the efficient operation of the network to outweigh Queensland Rail's interests in retaining that ability (s. 138(2)(a), (d), (e), (g) and (h)). Therefore, having regard to all the criteria in section 138(2), we do not consider it appropriate to approve Queensland Rail's proposal. We adopt our Draft Decision in this regard.

In relation to Aurizon's submission regarding competitive neutrality in the context of extensions, we do not consider the pricing rules need to be amended to provide for competitive neutrality when an access seeker funds an extension. Pricing rules facilitate variations between tariffs provided there is a legitimate reason for the variation; they do not simply mean that users pay the same tariff. If Queensland Rail was able to demonstrate that an expansion would not be able to generate sufficient revenue (along the whole corridor) to cover the cost of funding the expansion assets, and the incremental operating and maintenance costs of providing that access, then a higher access charge might be justified. The net impact of the expansion traffic on Queensland Rail would be expected to be a key component of both the access charge and funding agreement negotiations for an expansion.

Changes in TSC Payments

We require Queensland Rail to amend its 2015 DAU to remove the ability for Queensland Rail to differentiate between access seekers or between access seekers and access holders within a non-reference tariff market on the basis of changes to Queensland Rail's TSC payments.

We adopt our Draft Decision in this regard.

We consider that it is reasonable for Queensland Rail to recover its efficient costs and return on investment in accordance with section 168A(a) and its legitimate business interests (s. 138(2)(b)), though the appropriateness of any outcome must be considered in the context of all of the factors in section 138(2).

Our key concern is that Queensland Rail's 2015 DAU provides discretion for it to differentiate access charges between users within the same market on the basis of changes in TSC payments, which affect the market uniformly. This basis for price differentiation could result in adverse competition and efficiency impacts which run counter to the public interest in having efficient and competitive markets.

However, changes in the TSC payments do affect Queensland Rail's cost of providing the service. Because of this, we said in our Draft Decision that it is appropriate to allow Queensland to adjust the access charges for changes to TSC payments via the material change provisions in

⁷³ Queensland Rail, sub. 26: 60.

⁷⁴ See also QCA 2015: 51–52.

each individual access agreement. We maintain that this is the appropriate mechanism to make adjustments following changes to TSC payments.

Queensland Rail considers that if changes in TSC payments are accounted for solely by making adjustments according to material change provisions within the standard access agreement, it will have the unintended effect of not allowing Queensland Rail to receive at least its efficient cost of providing the service.⁷⁵

We note that the QCA Act does not require section 168A(a) to be complied with in some absolute way (see Chapter 10). In any event, Queensland Rail has not substantiated its position, other than to say that the material change provision may not be completely effective because the consequences of a reduction in TSCs may not have taken effect under an existing access agreement (which is being used as a comparator) at the time Queensland Rail may be negotiating a new access agreement.⁷⁶

We do not accept this argument from Queensland Rail. The differences in cost or risk are particular to Queensland Rail; that is, if the cost of providing access has increased from the costs incurred when the first access charges were negotiated (say, due to decreased TSC payments), then it does not matter whether the material change provisions have reflected that change in a comparative access agreement—the cost to Queensland Rail of providing access is different to when the first access agreement's access charges were negotiated. Further, as noted above, we have also reinserted the changes in cost or risk mechanism and allowed for Queensland Rail to have more flexibility to adjust access charges over the life of an access agreement.

Further, in response to Queensland Rail's submission that it may not be able to cover its efficient costs and provide an appropriate return on assets when there is a reduction in TSC payments, we note the material change provisions of the standard access agreement provide that Queensland Rail has the ability to adjust the access charges to account for reduced TSC payments. This provides a check on Queensland Rail's discretion and inefficient discrimination, but still allows for Queensland Rail to adjust particular access charges if a TSC reduction materially affects access charges. We consider that this is sufficient to protect Queensland Rail's legitimate business interests (s. 138(2)(b)) as well as the interests of access seekers and holders (s. 138(2)(e)).

As noted above, we consider a more appropriate way to justify different access charges is to assess the implications in regard to each individual access holder, based on TSC payments being a material change event within the relevant access agreements.

In this way, the onus will be on Queensland Rail to justify the extent to which reductions in TSC payments have a net financial impact on the cost or risk to Queensland Rail of providing access to the affected access holder/seeker. Stakeholders other than Queensland Rail agreed with this reasoning.⁷⁷

Importantly, if TSC payments that are related to one particular system are reduced, the change affects the cost of providing service to each access holder within that system equally. We disagree with Queensland Rail and do not consider that a change which affects the costs related

⁷⁵ Queensland Rail, sub. 26: 59.

⁷⁶ Queensland Rail, sub. 26: 59; see also cl. 18 (especially cl. 18.2 (Adjustment for material change)) of the QCA's 2015 SAA, which provides that the access charges can be adjusted to cancel out the net financial effect of, amongst other things, a reduction in TSC payments.

⁷⁷ See New Hope, sub. 23: 12–13; Aurizon, sub. 20: 19–20; Glencore, sub. 25: 1; Yancoal, sub. 27: 1.

to providing a service to all access holders should be used to discriminate amongst those access holders in a relatively discretionary manner.

We consider that, in Queensland Rail's proposal, Queensland Rail's legitimate business interest in being able to discriminate between access seekers and between access seekers and access holders is outweighed by the risk of inefficient (and possibly monopolistic) price discrimination (s. 138(2)(a), (b), (d), (e), (g) and (h)). That is particularly the case, given that Queensland Rail's legitimate business interest in generating expected revenue to meet at least its efficient costs and an appropriate return on investment is capable of being addressed via the material change provisions within the SAA (s. 138(2)(b)).

Preventing or hindering access

We included in our Draft Decision a requirement that the 2015 DAU prohibit setting access charges for the purpose of preventing or hindering access by third party access seekers.

Queensland Rail and Aurizon said that the particular provision was unnecessary, given that similar restrictions are provided by the QCA Act.⁷⁸

We agree that section 104 of the QCA Act prohibits Queensland Rail from preventing or hindering access (except as otherwise provided in an access undertaking (s. 104(6)); additionally, section 168A(c) requires that prices are not used to discriminate in favour of related parties of Queensland Rail unless justified by a difference in risk or cost). Therefore, Queensland Rail cannot engage in conduct that prevents or hinders access by a third party without offending these statutory obligations (unless otherwise provided by virtue of section 104(6)).

We therefore agree with stakeholders that the provision is not necessary and have removed it from the pricing rules part of the undertaking.

Remedying price differentiation breaches

As per the Draft Decision we require that the 2015 DAU provide for an access holder to have its access charge amended in the event Queensland Rail breaches its limitations on price differentiation. In this regard we have adopted the Draft Decision. We have also added some extra remedial provisions.

In Queensland Rail's 2015 DAU a price differentiation breach was to be remedied by reliance on a 'most favoured nation'⁷⁹ clause within an access agreement. These provisions aim to provide an access holder with a mechanism to require Queensland Rail to levy the same access charges that Queensland Rail levies on another access holder, if the latter access charges were calculated in contravention of the price differentiation provisions. However, reliance on these provisions is problematic in the event that another access holder is not aware that Queensland Rail contravened its obligations. That is, information asymmetry may nullify the utility of these provisions.

To address this shortcoming, our Draft Decision added a provision to the access undertaking that provides a means by which price differentiation breaches can be identified across all access holders. Monitoring was to be achieved through the compliance audit provisions contained in Part 5 of our mark-ups to the 2015 DAU. In this Decision, we have also included a new

⁷⁸ Queensland Rail, sub. 26: 56–57; Aurizon, sub. 20: 19–20.

⁷⁹ The 'most favoured nation' clauses are commonly used provisions which provide that if a subsequent party negotiates a more advantageous outcome in its contractual negotiations, a previous contracting party can require that it receive the same outcome.

'consequences of contravention' clause (cl. 3.9 in Appendix F), which will provide a mechanism for breaches of the pricing rules, including breaches of the differentiation provisions, to be remedied more directly than by reliance just on the most favoured nation clause.

Our approach recognises the legitimate business interests of Queensland Rail in being able to differentiate access charges, but provides checks and balances on this ability, thereby protecting the interests of access seekers who may not have access to information about prices charged to other access holders. We consider that the required amendments are appropriate, as they balance Queensland Rail's ability to differentiate access charges with a direct ability of an access holder to remedy a breach of the limits on price differentiation (s. 138(2)(a), (b), (d), (e), (g)).

Summary 3.3

The limits on price differentiation in the 2015 DAU must provide follows:

- (a) Where a reference tariff does not apply, differentiating access charges between access seekers and between access seekers and holders in respect of train services for the same commodity in the same geographical area should be limited to situations where:
 - (i) there are differences or changes in the cost or risk to Queensland Rail of providing access
 - (ii) available capacity is demonstrably insufficient to meet all access seeker requests, permitting the quotation of a maximum access charge for the purposes of the ceiling revenue limit but allowing Queensland Rail to charge below the maximum access charge.
- (b) Where a reference tariff does apply, differentiating access charges between access seekers and between access seekers and holders to reasonably reflect differences in the cost or risk to Queensland Rail of providing access for that train service (the description of which does not accord with the reference train service) compared to the specified reference train service.
- (c) Queensland Rail and an access seeker should be able to require a reasonable and balanced rate review provision (in an access agreement) to adjust access charges for differences in costs or risks to Queensland Rail of providing access to different reference tariff train services or in respect of non-reference tariff train services transporting the same commodity within the same geographical area. Queensland Rail is required to detail how it is applying a differentiation in access charges in the context of the application of a rate review provision.
- (d) Information regarding price differentiation breaches should be widely disseminated to access holders.
- (e) The consequences of a breach of Part 3 of the undertaking should be made clear through the inclusion of a remediation clause as well as by allowing the QCA to require Queensland Rail to levy the same access charges as it does for another like access holder or offer an access charge that neutralises the effect of the breach.
- (f) Queensland Rail agrees to promptly provide the QCA with all information requested by the QCA to enable the QCA to determine whether any contravention of Part 3 has occurred.
- (g) Access charges can be adjusted over time to reflect changes in the relative cost or risk to Queensland Rail of providing access to a particular access holder.

See clauses 3.0, 3.1.2, 3.3, and 3.6 and 3.9 in Appendix F for these and number of other related or consequential amendments.

3.4 Pricing and revenue limits

In its 2015 DAU, Queensland Rail proposed to apply:

- a ceiling revenue limit for access charges for train services (or groups of train services) so that access seekers do not pay for the services above the stand-alone cost of providing access; and
- a floor revenue limit reflecting the incremental cost of providing access to an individual train service or combination of train services, but providing Queensland Rail with absolute discretion to charge train services less than the floor revenue limit.

Our Draft Decision agreed that a floor and a ceiling revenue limit should apply. However, we did not accept that Queensland Rail should have an absolute discretion to charge train services less than the floor revenue limit.

Stakeholders' submissions

Queensland Rail opposed several of the QCA's required amendments to the 2015 DAU and said that a number of the QCA's amendments were not explained within the 2015 Draft Decision.⁸⁰

Asciano supported the Draft Decision but said that the 'issue could be better addressed via stronger ring fencing, improved cost transparency and regulatory determined prices'.⁸¹ Other stakeholders agreed with the Draft Decision, except Glencore, which raised concerns with the ceiling limit.⁸²

QCA analysis and Decision

We consider that the 2015 DAU should provide that access charges for non-reference tariff train services are set between floor and ceiling revenue limits, but that any proposal to price below the floor level should be subject to QCA approval to avoid any inefficient price discrimination. For reference tariff train services, the approved ceiling revenue limit will apply (see chapter 8).

We adopt our Draft Decision in this regard.

Definition of pricing limits

Queensland Rail said that there was insufficient reasoning for including a definition of pricing limits (cl. 3.2.1) in the QCA's Draft Decision. This clause was intended to supplement the ceiling and floor revenue limits. However, following a review of the amended pricing limit and price differentiation clauses, we no longer consider it necessary have so have not insisted on its inclusion.

Calculation of pricing limit

Glencore⁸³ said that the ceiling limit could not realistically be calculated, and Asciano said that transparency needed to be improved. To this end, we have significantly increased the obligations on Queensland Rail to provide information in relation to asset values underpinning the calculation of access charges—in the case of both initial negotiations and renewals (See Chapter 2 of this Decision). If a stakeholder considers that Queensland Rail has breached these provisions, it remains open to them to lodge a dispute.

⁸⁰ Queensland Rail, sub. 26: 62–65.

⁸¹ Asciano, sub 28: 11.

⁸² See New Hope, sub. 23: 13; Aurizon, sub. 20: 18–22; Yancoal, sub. 27: 1; Glencore, sub. 25: 1.

⁸³ Glencore, sub. 25: 1.

Queensland Rail did not comment on our requirement to have the QCA approve any proposal to set access charges below the expected incremental cost of providing access (i.e. the floor revenue limit). Other stakeholders agreed with our amendments.⁸⁴

Definition of evaluation period

Queensland Rail said that the changes made to the definition of 'evaluation period' suggested that the QCA applied the term in setting reference tariffs.⁸⁵ This is not the intention of the change nor do we consider that our proposed amendments to the 2015 DAU allowed for the QCA to unilaterally set additional reference tariffs. However, we have had regard to, amongst other things, the point made by Queensland Rail and not maintained our proposed amendments.

We have had regard to Queensland Rail's legitimate business interests and appropriately addressed them by accepting that Queensland Rail should have the ability to price below the incremental costs of providing access if it so chooses and the price is approved by the QCA (s. 138(2)(b)).

Ceiling revenue limit equation

Queensland Rail raised concerns about the inclusion in the ceiling revenue limit equation of a variable for an adjustment amount (AAt). Queensland Rail said that the Draft Decision did not discuss the need for or purpose of AAt.⁸⁶

We note that the adjustment amount was discussed in detail in chapter 8 of the Draft Decision. Moreover, Queensland Rail and stakeholders provided submissions in response to that discussion and the QCA's final position is contained in Chapter 8 of this Decision. As part of this Decision, our technical amendments have revised how the adjustment amount will be implemented. We no longer consider a variation to the ceiling revenue limit as an appropriate mechanism for the adjustment amount and instead consider that an adjustment amount mechanism is appropriate. Therefore, we have removed 'AAt' from the ceiling revenue limit formula in Part 3 and drafted clause 7.1 of Schedule D to Appendix F.

Approved ceiling revenue limit

Queensland Rail queried the inclusion of the term 'approved ceiling revenue limit'.⁸⁷ The rationale for including this concept and the relevant calculation are discussed in Chapter 8 of this Decision (the amount is included in Schedule D to Appendix F).

Summary 3.4

The pricing and revenue limits in the 2015 DAU must provide as follows:

- (a) A clear methodology to implement floor revenue limits should be included.
- (b) If Queensland Rail proposes to price access below the floor revenue limit it should seek the QCA's approval beforehand.

See clause 3.2 in Appendix F.

⁸⁴ See, for example, New Hope, sub. 23: 13.

⁸⁵ Queensland Rail, sub. 26: 63.

⁸⁶ Queensland Rail, sub. 26: 63–64.

⁸⁷ Queensland Rail, sub. 26: 62–63.

3.5 Asset valuation methodology for negotiating access charges

Section 168(A) states that the price of access to a service should 'include a return on investment commensurate with the regulatory and commercial risks involved' (s. 168A(a)). However, the legislation does not expressly state how to calculate the amount of an 'investment'.

In order to perform our statutory functions, we consider the appropriate way to value the investment is with reference to the factors in section 138(2) of the QCA Act and the specific characteristics of the markets served by Queensland Rail. For example, in Queensland Rail's monopoly markets the value of assets should be calculated independently of the prices that are set. However, within markets in which Queensland Rail is not a monopoly provider, such as competitive intermodal transportation, market prices are more likely to shape appropriate asset prices.

The Draft Decision did not accept Queensland Rail's proposal that the asset valuation methodology for setting a ceiling revenue limit be limited to a depreciated optimised replacement cost methodology. Rather, we proposed to maintain sufficient flexibility in access negotiations on asset valuations for the appropriate asset valuation approach to be determined for a given circumstance.

Stakeholders' submissions

Queensland Rail said that 'consistency and certainty demand the setting and consistent application of a specific valuation methodology¹⁸⁸ when setting a value for an asset. They also said flexibility about the valuation methodology would be contrary to good regulatory practice and contrary to the object of Part 5 of the Act because an ever-changing asset valuation approach for assets used to provide a declared service failed to promote the economically efficient use of those assets.⁸⁹

Asciano⁹⁰ and New Hope⁹¹ supported the QCA's view that the method of asset valuation should be selected to suit the particular circumstances.

QCA analysis and Decision

We disagree with Queensland Rail's statement that a specific valuation methodology should be set, and then applied to the determination of all opening asset valuations. As noted in section 3.6 of the Draft Decision, imposing a uniform asset valuation process risks generating inappropriate valuations of Queensland Rail's infrastructure, given the varied traffic types and mixes.

We also disagree with Queensland Rail's statement that allowing flexibility in the choice of asset valuation approach will result in 'ever changing' asset valuations. For reference tariff train services the asset value has been set having regard to section 138(2) of the QCA Act (see Chapter 8). For non-reference tariff train services, the value is to be agreed between the access seeker and Queensland Rail. If a value cannot be agreed it is open for the parties to bring a dispute to the QCA, in which case the QCA can determine the appropriate value.

Accordingly, we have adopted the Draft Decision approach and we require Queensland Rail to remove the requirement in the 2015 DAU that only a DORC methodology be used for valuing

⁸⁸ Queensland Rail, sub. 26: 67.

⁸⁹ Queensland Rail, sub. 26: 67.

⁹⁰ Asciano, sub. 28: 11.

⁹¹ New Hope, sub. 23: 15.

assets when setting the ceiling revenue limit. It should instead provide for a methodology appropriate for the circumstances.

Summary 3.5

The 2015 DAU must remove the requirement that the asset value which is to be used to determine a ceiling revenue limit be set solely on the basis of a depreciated optimised replacement cost methodology.

Instead the value of the assets used to calculate the reference tariff (including where relevant the approved ceiling revenue limit) is set by reference to the regulatory asset base approved by the QCA. For the ceiling revenue limit for non-reference tariff train services, the value of the assets is to be as agreed by the access seeker and Queensland Rail or, failing agreement, as determined by the QCA.

See clause 3.2.3(c) and Schedule D, clause 4 in Appendix F.

3.6 Pricing for access rights at renewal

Queensland Rail proposed that terms for access charges for renewed access rights be based on the same methodology, rates and other inputs for calculating the access charge as were used under the existing access agreement.

However, Queensland Rail said this was subject to the following preconditions:

- There were no other competing applications for the same commodity in the same geographic area (cl. 3.3(c)(i));
- Access rights being renewed were consistent in all respects with the existing access agreement—namely, for the same commodity, same number of train services and same train characteristics and description (cl. 3.3(c)(ii)); and
- No reference tariff applied (cl. 3.3(c)(iii)).

Queensland Rail also wanted to be able to vary existing access charges based on:

- differences of, and changes to, cost or risk;
- material changes in circumstances that impacted on an access holder's ability to pay; or
- changes that resulted in Queensland Rail being unable to commercially provide access in that geographic area.⁹²

The Draft Decision accepted the need to implement a contract renewal pricing mechanism, but required amendments to Queensland Rail's approach so that the scope and application of the provisions provided greater certainty for renewing access holders and Queensland Rail.

Stakeholders' submissions

Queensland Rail raised a number of concerns in relation to the QCA's proposed amendments.⁹³

Other stakeholders largely supported the QCA's proposed amendments but said that the definition of a renewal access seeker should be less rigid;⁹⁴ Queensland Rail should be required

⁹² See QCA 2015, section 3.3.

⁹³ Queensland Rail, sub. 26: 61-2.

to provide information in relation to the calculation of its access charges;⁹⁵ and the QCA should be able to require Queensland Rail to develop a new reference tariff.⁹⁶

Glencore reiterated its concerns with an approach which may lock in an existing methodology, rate and inputs. It also said that renewal rights should extend to intermodal services and that what constituted a renewal should be more flexible.⁹⁷ New Hope asked that the network utilisation provisions be strengthened to ensure Queensland Rail was able to justify setting maximum access charges when it considered that the network was capacity constrained.⁹⁸ Asciano said it had ongoing concerns with the pricing of the Mount Isa line and that an alternative approach should be developed 'which limits Queensland Rail's monopoly power in relation to Mount Isa rail line access pricing, while ensuring an appropriate level of Queensland Rail performance'.⁹⁹

QCA analysis and Decision

We require Queensland Rail to amend the non-reference tariff renewal pricing provisions to more evenly balance the negotiating position between Queensland Rail and a renewing non-reference-tariff access holder.

Our intention is to have regard to the sunk costs of access holders whilst allowing Queensland Rail flexibility in relation to the relevant access charges with a view to generating expected revenue that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved. We have also relaxed the definition of 'renewal' in Chapter 2 of this Decision. In addition to the changes to the definition of 'renewal', we consider that a renewing access seeker should be able to change its train service description if that change supports supply chain improvements. This will allow for renewal access agreements that seek to renew a portion of the previous access rights and also make changes to their train service description due to supply chain improvements to still benefit from the renewal pricing mechanism.

Our key amendments to Queensland Rail's proposed renewal pricing mechanism are summarised in Table 4. Where the amendments have not changed from the Draft Decision, we have also adopted the relevant parts of that Draft Decision (see section 3.7 of the Draft Decision).

⁹⁴ Aurizon, sub. 20: 20–21; Yancoal, sub. 27: 4.

⁹⁵ Aurizon, sub. 20: 21; Glencore, sub. 25: 2.

⁹⁶ New Hope, sub. 23: 14.

⁹⁷ Glencore, sub. 25: 2–3.

⁹⁸ New Hope, sub. 23: 11–12.

⁹⁹ Asciano, sub. 28: 12.

2015 DAU proposal	Required amendment	Rationale
Price differentiation should be allowed in a broad range of circumstances.	Price differentiation will be limited to differences and changes in cost or risk factors between the expiring access agreement and the renewed access agreement.	Differences or changes in the cost or risk to Queensland Rail are a legitimate basis for varying access charges (see Section 3.3 above). Decreases in costs or risks to Queensland Rail should also be reflected. Queensland Rail should provide details of how it calculated the access charges; access seekers can lodge a dispute if required.
Renewal access agreements without a competing access application should be provided with the renewal pricing mechanism.	All non-reference tariff renewal access agreements are provided with the renewal pricing mechanism, not just those without a competing access application.	It is appropriate that the renewal pricing applies to all renewal access seekers where the renewal access agreement is substantially similar to protect that renewal access seeker's sunk costs.
A mechanism should apply at renewal of access agreements.	The renewal pricing mechanism will be provided on a one-off basis to provide the access seeker with an incentive to match the access agreement term with its expected payback period.	The renewal pricing mechanism is intended to provide certainty to underpin future investment. At the same time, a one-off renewal right allows scope for Queensland Rail to seek more favourable terms from alternative access seekers in the future.
Renewal access agreements should have, in all respects, the same characteristics as the existing access agreement.	Where an access application would be classed a 'renewal' but for changes to the train service description due to operational or supply chain improvements (e.g. increased payload, longer trains or reduced cycle time initiatives) the renewal pricing mechanisms will still apply, but there will be a contribution to common costs so that Queensland Rail is no worse off than under the existing access agreement.	We have broadened the definition of 'renewal' in Chapter 2 of this Decision. This will broaden the circumstances under which the renewal pricing mechanism will operate. In addition to those changes we require that if there are further changes to the train description, due to supply chain improvements, the renewal access seeker will still have the benefit of the renewal pricing mechanism. A fair contribution towards Queensland Rail's common costs is required. This amendment is required to enable a fair allocation of the benefits from operational improvements and provide for Queensland Rail's recovery of common costs (of providing the service to other access holders).

The major changes required to the renewal pricing provisions are covered in greater detail in the following discussion.

One-off application right

Queensland Rail has noted some ambiguity in relation to the QCA's drafting related to any renewal right being a one-off right.¹⁰⁰ Other stakeholders did not comment on this aspect of the Draft Decision. We acknowledge that there is scope for the clause to be interpreted incorrectly and have made amendments to address this. The intention of the QCA is that each non-reference-tariff access holder that is seeking to renew its access agreement(s) can only have the benefit of the provision once per access agreement.

¹⁰⁰ Queensland Rail, sub. 26: 61.

This 'one-off' use of the renewal pricing mechanism provides an appropriate balance between the interests of a renewal access seeker and Queensland Rail. It gives the access seeker some certainty around its renewal pricing,¹⁰¹ but encourages the access seeker to renew on a long-term basis, matching the access agreement term with its expected payback period. This should provide Queensland Rail with greater revenue certainty, and balances Queensland Rail's legitimate business interests with those of access holders (s. 138(2)(b), (e), (h)).

Renewal pricing—differences in cost or risk

Queensland Rail also said that the QCA's proposed amendments may not allow Queensland Rail to amend the access charges for a renewing access holder because exactly the same access rights are being renewed; therefore, there will be no differences in the costs or risks between the expiring and renewed access agreements.¹⁰²

We have made some amendments in addition to those proposed in the Draft Decision, to clarify that the access charges may be varied to account for differences in the nature of the costs or risks between the expiring and the renewed access agreement, as well as changes in the costs or risks between the expiring and the renewed access agreement.

Glencore said that the 'existing inputs' for the purposes of the renewal pricing should be those from the last long-term access agreement between the parties as the current inputs often reflected short term arrangements that were subject to higher prices.¹⁰³ We consider that the strengthened information provisions, coupled with the renewal pricing differentiation clauses will make negotiations fairer and more transparent. For example, the strengthened information obligations on Queensland Rail (see below) should make it clear whether or not an aspect of cost or risk priced into previous short-term access agreement is taken into account when negotiating a renewal. Any disagreement could form the basis of an access dispute.

We consider that our Decision is consistent with the negotiate-arbitrate model and is appropriate after having regard to all of the factors in section 138(2) of the QCA Act.

Renewal pricing information

Asciano¹⁰⁴ said that a new pricing approach should be developed to limit Queensland Rail's monopoly power. Glencore said that Queensland Rail should be required to disclose its existing methodology, rates and other inputs for calculating the pricing under the existing access agreement.¹⁰⁵

We note that Queensland Rail has obligations under the QCA Act to provide precisely the kind of information referred to by Glencore when negotiating access (s. 101). If Glencore considered that it had not been provided with that information in the past it was open to Glencore to bring a dispute under the QCA Act.

In the Draft Decision we also increased the access charge information requirements on Queensland Rail (see especially cls. 2.4.2(e) and 2.7.2(vi) in Appendix C of the Draft Decision). In this Decision, we consider that it is appropriate to further reinforce the information provisions by providing that, if Queensland Rail relies on clause 3.3(e),¹⁰⁶ it must also provide details of

¹⁰¹ Noting that their rights to capacity are protected as per Section 2.5 above.

¹⁰² Queensland Rail, sub. 26: 61.

¹⁰³ Glencore, sub. 25: 2.

¹⁰⁴ Asciano, sub. 28: 12.

¹⁰⁵ Glencore, sub. 25: 2; sub. 30: 2–3.

¹⁰⁶ Clause 3.3(f) in the 2015 DAU outlines how Queensland Rail can price differentiate a renewal access seeker's access charges (for non-reference-tariff trains).

how that clause has been applied. We consider that this strikes an appropriate balance between Glencore's concerns and Queensland Rail's legitimate business interests (s. 138(2)(b), (e)). That is, Queensland Rail is able to adjust for differences in risk or cost, but these adjustments must be substantiated. If a renewal access seeker disagrees with the reasons for an increase, it is open to it to bring a dispute.

Queensland Rail also said that if capacity was constrained at renewal, the renewal price should be allowed to vary to reflect limitations on capacity. We do not agree with Queensland Rail for the reasons set out in Section 3.3 above.

We consider these changes will promote the economically efficient operation of the network and are in the public interest as they promote transparency in price setting (s. 138(2)(a), (b), (d), (e), (g), (h)).

Supply chain improvements

In the Draft Decision we inserted an exception to the requirement that the renewal access agreement be identical for the renewal access pricing provisions to apply. We provided that the renewal pricing provisions could still apply if the train services varied from the expiring access agreement due to supply chain improvements. However, this was still subject to Queensland Rail not being any worse off in relation to common costs under the renewed access agreement (see cl. 3.3(g) in Appendix C of the Draft Decision and cl. 3.3(f) in Appendix F).

Queensland Rail considered that the QCA's drafting of this clause was unclear.¹⁰⁷ Aurizon and Glencore said that the Draft Decision amendments were overly restrictive by requiring that the renewed access rights be associated with an identical number of train services which were identical in all respects and asked that the definition of renewals be broadened.¹⁰⁸

Our intent under the Draft Decision was that the renewal provisions, whilst protecting an access seeker's sunk costs and allowing Queensland Rail to generate expected revenue to earn an appropriate return, should incentivise efficiency improvements in the network. This may advance Queensland Rail's legitimate business interests by increasing contracting efficiency and potentially freeing up train paths for re-contracting (s. 138(2)(b)). It also complies with the pricing principles (s. 138(2)(g)). In addition, the renewal provisions promote the interest of access seekers and access holders and the public interest in having an efficiently run network (s. 138(2)(a), (d), (e), (h)). Accordingly, we consider our required renewal provisions are appropriate with regard to all the criteria in section 138(2).

We have broadened the definition of 'renewal' in Chapter 2 of this Decision such that there will be greater scope for variations from an existing access agreement to still be considered a renewal application. In addition, we require Queensland Rail to amend the renewal pricing provisions to provide that Queensland Rail will still apply the renewal pricing rules to a nonreference tariff access holder if there are changes to the train services as a result of supply chain improvements which would otherwise have meant that the application was not within the definition of a renewal application.

In addition to these changes, we have also clarified the amendments from the Draft Decision in relation to Queensland Rail not being any worse off in relation to recovery of its common costs as a result of accommodating supply chain improvement initiatives.

¹⁰⁷ Queensland Rail, sub. 26: 62.

¹⁰⁸ Aurizon, sub. 20: 21; Glencore, sub. 25: 3.

It is likely that the introduction of supply chain improvements could result in a reduction of the incremental costs to Queensland Rail of providing the service, and a corresponding reduction in the access charges. However, Queensland Rail's risk in relation to common costs is protected and Queensland Rail also has the benefit of freed-up capacity which it can recontract.

We do not consider that Glencore's proposal is appropriate in this instance, as it tips the balance too far in favour of a renewing access holder. We are aiming to strike a balance between Queensland Rail's and the renewal access holder's interests while increasing the overall efficiency of the network. If a proposed renewal is too different from the existing access agreement, the access holder is effectively negotiating a new access agreement. In this regard, we note that we have increased the obligations on Queensland Rail to vary the SAA for demonstrable efficiency improvements (see Chapter 7 of this Decision).

See also Chapter 2 for a discussion about what the QCA considers should constitute a 'renewal' (including in relation to intermodal services) for the rest of the negotiation provisions within the 2015 DAU.

Renewal pricing appropriateness

We consider that our approach provides all parties with increased certainty. Access holders will know that Queensland Rail cannot levy access charges with undue discretion and thereby impose risk premiums on future investments. At the same time, it may advance Queensland Rail's legitimate business interest, as it increases certainty by providing the potential for future, but not open-ended, changes to access charges and the flexibility to generate expected revenue that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved (s. 138(2)(a), (b), (e), (g) and (h)).

Summary 3.6

The renewal pricing provisions in the 2015 DAU must provide that:

- (a) the renewal pricing mechanism can be used only once
- (b) access charges for renewal access applications may be varied to account for differences in the nature of costs and risks as well as actual costs or risks between an expiring and renewing access agreement
- (c) if an access seeker would be a renewal access seeker but for changes to its train services to accommodate supply chain improvements, it will still have the benefit of the renewal pricing provisions
- (d) in the case that a contribution to common costs is an input for calculating access charges arising from a renewal accommodating supply chain improvements, Queensland Rail must provide details of how it calculated that input
- (e) if Queensland Rail differentiates for cost or risk, it must provide details of how it calculated that differentiation for both reference tariff and non-reference tariff traffics and renewing access applications
- (f) price differentiation is limited to differences and changes in cost or risk or material limitations on available capacity
- (g) all non-reference-tariff renewal access agreements are subject to the renewal pricing mechanism unless the expiring access agreement provides otherwise.

See clauses 2.7.2(a)(vi) and 3.3(e), (f), (g) and the definition of 'renewal' in Appendix F.

3.7 Rate of return

The regulated rate of return is a key input into determining appropriate reference tariffs. The regulated rate of return is calculated using a weighted average cost of capital (WACC), comprising three primary components:

- cost of equity—typically estimated with reference to the Capital Asset Pricing Model (CAPM)
- cost of debt—observed or estimated from the current debt rate
- capital structure—appropriate debt and equity proportions of firm market value, typically determined by benchmarking.

Queensland Rail's WACC proposal comprises two principal parts:

- as at the approval date, an indicative, nominal post-tax, 'vanilla' WACC of 6.93 per cent, comprising a cost of equity of 8.01 per cent and a cost of debt of 6.05 per cent. Queensland Rail¹⁰⁹ proposed to update the risk-free rate and debt margin once the averaging period for determining them was agreed between the QCA and Queensland Rail
- after the approval date, a variable WACC, derived by adding a WACC margin¹¹⁰ of 4.12 per cent to the average yield on a five-year Commonwealth Government bond over a 20-day

¹⁰⁹ Queensland Rail, sub. 2: 41.

¹¹⁰ Queensland Rail's 2015 DAU defines the margin as being the difference between the WACC as at the approval date and the risk-free component of the WACC.

trading period ending as close as practicable to, but not later than, the date that Queensland Rail offers an access agreement to an access seeker.

The Draft Decision accepted Queensland Rail's proposed WACC of 6.93 per cent, subject to updating the time-variant parameters (i.e. the risk-free rate and debt margin), using a term and methodology consistent with the QCA's standard approach.

Stakeholders' submissions

Yancoal¹¹¹ and New Hope¹¹² supported the methodology applied by the QCA. Queensland Rail said the risk-free rates for periods before 2013–14 and 2014–15 were higher.¹¹³

QCA analysis and Decision

We have adopted our Draft Decision position on an appropriate non-time variant WACC parameters for Queensland Rail (see section 3.8 of the Draft Decision).¹¹⁴

Two different WACCs (see Table 5) have been determined:

- a WACC of 6.93 per cent which is consistent with that proposed by Queensland Rail in its 2013 and 2015 DAUs but which is based on the parameters in our 2014 Draft Decision. This WACC is applicable to reference tariffs relating to the period from 1 July 2013 to 30 June 2016
- an updated WACC of 5.73 per cent derived using a risk free rate and debt margin calculated over a 20-business-day period beginning Monday, 14 March 2016. This WACC is applicable to reference tariffs relating to the period from 1 July 2016 onwards.

Both WACCs use the non-time-variant parameters used by the QCA in its decisions on Aurizon Network's 2013 and 2014 DAUs, that were proposed by Queensland Rail in its 2015 DAU (see Table 5).

The difference in the two WACCs comes from the time-variant parameters (the risk-free rate and debt margin), assessed by Incenta Economic Consulting. Incenta's report for the June 2013 parameters was published with our October 2014 Draft Decision, while its report for the March–April 2016 parameters is published with this Decision.¹¹⁵

Queensland Rail said in its December 2015 submission that the WACC for the tariff that would have applied in relation to the period from 1 July 2013 to 30 June 2016 was based on market data from a different period (i.e. the 20 business days before 1 July 2013). It added that the risk-free rate so derived was higher in this previous period, and included a graph illustrating the decline in risk-free rate between July 2013 and February 2016.

¹¹¹ Yancoal, sub. 27: 3.

¹¹² New Hope, sub. 23: 14.

¹¹³ Queensland Rail, sub. 26: 30.

¹¹⁴ We note that the Australian Competition Tribunal (the Tribunal) has recently decided to set aside and remit the AER's 2015 determinations for each of the appeals brought by the NSW and ACT electricity distributors (see Applications by Public Interest Advocacy Centre Ltd and Ausgrid [2016] ACompT 1). Among other matters, the AER is required to remake its decision by reference to a gamma of 0.25. We have considered the Tribunal's decision in relation to gamma and find that there is nothing in the Tribunal's reasoning that demonstrates that our approach to estimating gamma is inappropriate. See the QCA 2016 decisions on Aurizon Network and DBCT for further details.

¹¹⁵ Incenta Economic Consulting 2016.

We agree with Queensland Rail's comment. The risk-free rate has declined since 1 July 2013. However, the WACC averaging period used to derive the WACC proposed in our 2014 and 2015 draft decisions is consistent with past practice, and the resulting WACC of 6.93 per cent is consistent with the expectations of Queensland Rail and other stakeholders.

We do not consider it would advance Queensland Rail's legitimate business interest to reduce the WACC by applying a new, later averaging period, in order to derive the tariff that would have applied in relation to the period from 1 July 2013 to 30 June 2016 (s. 138(2)(b). Changing the WACC would also be against the expectations of access seekers, access holders and their customers (s. 138(2)(e), (h)). Therefore, having regard to all the approval criteria in section 138(2), we consider it appropriate to adopt the 6.93 per cent WACC from the Draft Decision to derive the tariff that would have applied in relation to the period from 1 July 2013 to 30 June 2016.

Parameter	Applicable to the period from 1 July 2013	Applicable to the period from 1 July 2016	
Credit rating	BBB+	BBB+	
Risk-free rate	2.81%	2.00%	
Market risk premium	6.50%	6.50%	
Asset beta	0.45	0.45	
Gearing (debt to total enterprise value)	55%	55%	
Equity beta	0.80	0.80	
Gamma	0.47	0.47	
Equity margin	5.20%	5.20%	
Cost of equity	8.01%	7.20%	
Debt margin (including financing allowance)	3.24%	2.52% ¹¹⁶	
Cost of debt	6.05%	4.52%	
WACC margin	4.12%	3.73%	
WACC	6.93%	5.73%	

Table 5	Cost of capital	parameters for the 2015 access undertaking
---------	-----------------	--

¹¹⁶ The 'financing allowance' of 22 basis points covers refinancing costs (interest rate swaps) and transaction costs. This is on top of the debt margin of 2.30 per cent assessed by Incenta for the 2016–20 WACC. For more information on the allowance, see QCA 2016b: 207, Table 107.

Summary 3.7

The 2015 DAU must:

- (a) include a WACC of 6.93 per cent per annum in relation to the period 1 July 2013 to 30 June 2016
- (b) include a WACC of 5.73 per cent per annum in relation to the period from 1 July 2016, consistent with the QCA's approved WACC parameters.

3.8 Take-or-pay arrangements

Queensland Rail's 2015 DAU proposal

Queensland Rail proposed to maintain take-or-pay arrangements within its 2015 DAU. This means that access holders would pay a proportion of their contracted access charge, even in the event that they do not actually use train service entitlements (TSEs).

In particular, in relation to West Moreton network coal traffic, Queensland Rail proposed that:

- access holders be liable for a pre-determined proportion of the total access charge¹¹⁷
- Queensland Rail provide take-or-pay relief where services are not provided due to a Queensland Rail Cause (Schedule D).¹¹⁸

These arrangements, amongst other things, provide revenue certainty for Queensland Rail by transferring a degree of volume risk to access holders.

The Draft Decision was to accept Queensland Rail's proposal to apply take-or-pay for West Moreton network coal traffic at a predetermined rate where a reference tariff applies, excluding instances where access is not available due to a Queensland Rail cause (refer to chapter 8 of the Draft Decision). However, we did not require a prescribed amount of take-or-pay where no reference tariff applied.

Stakeholders' comments

Aurizon agreed that, where a reference tariff was not payable, take-or-pay charges should be a matter for negotiation between the parties.¹¹⁹ Comments by other stakeholders, except Glencore, were in relation to reference tariff services and largely concerned with the prescribed rate of take-or-pay.¹²⁰ Glencore said that, if no rate of take-or-pay was prescribed for non-reference-tariff services, Queensland Rail would simply insist on 100 per cent take-or-pay.¹²¹

QCA analysis and Decision

We accept that prescribed take-or-pay arrangements are appropriate for reference tariff train services, but we do not accept that prescription of such arrangements is appropriate in the

¹¹⁷ The rate of take-or-pay for West Moreton network coal traffics is considered in Chapter 8 of this Decision in the context of form of regulation matters.

¹¹⁸ 'Queensland Rail Cause' is a defined term in the 2015 DAU, which reflects Queensland Rail's inability to make the network available to provide train service entitlements as a result of defined events.

¹¹⁹ See Aurizon, sub. 20: 49.

¹²⁰ Aurizon, sub. 20: 49; 24, 49; Yancoal, sub. 27: 2; New Hope, sub. 24: 13.

¹²¹ Glencore, sub. 25: 2.

context of non-reference-tariff services. We have adopted our Draft Decision in this regard (see section 3.5 of the Draft Decision).

Reference tariff train services

Prescribed take-or-pay arrangements for reference tariff train services are appropriate given the allocation of risks, rewards and costs have already been determined.

We accept Queensland Rail's proposal to apply take-or-pay for West Moreton network coal traffic at a predetermined rate where a reference tariff applies, excluding instances where access is not available due to a Queensland Rail cause (refer to Chapter 8 of this Decision).

This is a reasonable approach which is in Queensland Rail's legitimate business interests, while at the same time being in the interests of access seekers and access holders, consistent with the QCA Act (s. 138(2)(b), (e), (h)). It should also promote the efficient use of capacity (s. 138(2)(a)).

Non-reference-tariff train services

Glencore said that, at a minimum, we should impose a cap on take-or-pay arrangements for non-reference tariff services. We do not agree with Glencore's submissions. It is appropriate to include regulation of take-or-pay in a reference tariff environment, because the QCA has regulated the cost, risks and returns of the access charges. Take-or-pay provides some certainty to the access provider where their upside is capped.

For a non-reference-tariff system, there is no similar regulation of costs and returns and because of this, it would be inappropriate to regulate take-or-pay. It is not uncommon for take-or-pay terms to form part of negotiations in unregulated market environments. We maintain the view that, for non-reference-tariff systems, the negotiation stage remains the appropriate forum for the parties to consider take-or-pay arrangements and that its regulation could inappropriately distort the negotiations between two sophisticated commercial parties.

Glencore also said that referring disputes for arbitration may not be a realistic option, but it did not elaborate on this point. We consider that the negotiate—arbitrate model is appropriate and cost-effective, as the reasoning behind it reflects that the possibility of dispute resolution is likely to encourage the parties to act more reasonably in negotiating take-or-pay.

We consider that this decision is appropriate after having regard to all submissions on the matter and each of the factors listed in section 138(2) of the QCA Act. It strikes a balance between Queensland Rail's interests, access seekers' interests and the public interest by allowing the negotiate-arbitrate model to be used (s. 138(2)(b), (d), (e), (g), (h)).

Summary 3.8

Queensland Rail's proposal to request take-or-pay from access holders at a predetermined proportion for reference tariff train services is accepted.

See Schedule 3 of the 2015 SAA (Appendix G).

4 OPERATING REQUIREMENTS

Part 4 of the 2015 DAU provides for the operating requirements that govern how Queensland Rail will deliver train service entitlements (TSEs). These include the network management principles (NMPs) for Queensland Rail to schedule, manage, and demonstrate capacity for train services (Schedule F). They also include the Operating Requirements Manual (ORM), which prescribes rules for use of the network by train operators (Schedule G).

This Decision accepts a substantial part of Queensland Rail's 2015 DAU proposal but requires amendments to clarify how the NMPs and ORM will operate.

The Decision in this chapter differs from the Draft Decision mainly as follows:

- In relation to scheduling changes, Queensland Rail is only required to notify those stakeholders who are non-access holders that opt in
- Queensland Rail's notification requirement for changes to planned possession has been extended to three months, from 20 business days
- Queensland Rail's obligations in relation to the Environmental Impact and Risk Management Report (EIRMR) process have been made less prescriptive
- Compensation provisions for changes to the ORM via a DAAU process have been removed
- Provisions for amending the ORM for minor matters and safety matters have been removed.

Introduction

The safe and efficient operation and use of Queensland Rail's network will be guided by the NMPs (Schedule F) and ORM (Schedule G) in the 2015 DAU. The NMPs set out how Queensland Rail will coordinate maintenance and other track restrictions, schedule and operate trains and demonstrate available capacity. The proposed ORM governs a variety of other procedures for operating trains and addressing matters such as safety and emergency responses.

In our Draft Decision, we indicated a number of particular provisions which we required to be amended or varied before we would consider the relevant Part and Schedules to be appropriate. Queensland Rail has accepted some of these amendments (in relation to the NMP) but has also not accepted a substantial number of required amendments. Key issues are summarised in Table 6 below. Matters that require a more detailed explanation are discussed in Sections 4.1 to 4.4.

	,	operating requirements, runn and eritin	
Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' positions	QCA Decision
1. Changes to train plar	IS		
A broader range of parties should be notified about changes to Queensland Rail's train plans.	Accepted in part.	Largely agreed with the Draft Decision. New Hope and Yancoal said notice for amending the MTP and DTP was short.	See Section 4.1.
Access holders should be consulted about all non- emergency operational	Did not accept.	Agreed with the Draft Decision. New Hope suggested some	See Section 4.1.

Table 6: Summary of key positions and decision—operating requirements, NMP and ORM

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' positions	QCA Decision
constraints that affect scheduled paths on the DTP.		additional amendments.	
Queensland Rail should seek agreement from access holders to vary the DTP from the MTP, except for emergencies, and report on timing of planned possessions.	Did not accept.	Agreed with the Draft Decision.	See Section 4.1.
Queensland Rail should delay changes to the MTP until related disputes are resolved.	Did not accept.	Agreed with the Draft Decision.	See Section 4.1.
Queensland Rail should make reasonable endeavours to minimise adverse effects of constraints, offer useable replacement train paths.	Did not accept.	Agreed with the Draft Decision.	See Section 4.2.
2. Coordination with ad	ljoining networks	·	
Queensland Rail should consult with other railway managers on matters affecting both networks and minimise effects on through- running trains.	Did not accept.	Agreed with the Draft Decision.	See Section 4.3.
3. Passenger services			
Network controllers should 'act reasonably' when forming a belief that it is necessary to give priority to passenger train services.	Did not accept.	Agreed with the Draft Decision	See Section 4.4.
4. Operating Requireme	ents Manual		
The ORM should balance obligations and requirements, clarify procedures and link with other provisions.	Did not accept.	Agreed with Draft Decision. Aurizon suggested some additional amendments.	See Section 4.5.
The ORM should not be amended from the SAA but (except for certain minor matters) via the DAAU process. Compensation addressed in the SAA.	Did not accept.	Agreed with Draft Decision.	See Section 4.6.

4.1 Changes to train plans

In the 2015 DAU, Queensland Rail did not propose to consult with stakeholders when it varied the daily train plan (DTP) from the schedule set out in the master train plan (MTP) if an access holder's train services were not affected (Schedule F). If an access holder's train services were affected, the 2015 DAU's proposed process for changing the MTP and DTP required that Queensland Rail (except in cases of urgent or emergency possessions):

- notify access holders whose activities were affected by any modifications, of changes to the MTP at least 20 business days in advance
- consult with relevant access holders where a proposed change to the MTP would result in those access holders' scheduled train services not being met
- agree modifications to the MTP with relevant access holders where the change was not within the scope of those access holders' TSEs.

The proposed NMPs provided that if Queensland Rail wished to make a short-term change to a DTP at least two business days before the DTP was scheduled because of an operational constraint, and the variation would result in an access holder's scheduled train services not being met, Queensland Rail first has to consult with that access holder. Queensland Rail also proposed that it may alter the DTP before the DTP is scheduled if Queensland Rail invites affected access holders to consider and agree to the changes at least 36 hours before the day of operation.¹²² Queensland Rail said:

Once scheduled, Queensland Rail cannot vary the DTP so as to adversely affect the access holder except where an Emergency Possession is required.¹²³

Queensland Rail proposed broader notification, but only by publishing an unredacted MTP on its website every six months.¹²⁴

Our Draft Decision proposed to increase the notification, consultation and agreement obligations on Queensland Rail in relation to scheduling changes.

Stakeholders' submissions

Queensland Rail largely rejected the QCA's proposed amendments.¹²⁵

Yancoal and New Hope agreed with the QCA's Draft Decision¹²⁶; however, both remained concerned about the amount of discretion Queensland Rail had to change the MTP and DTP without the consent of access holders. New Hope suggested a number of additional amendments in this regard.¹²⁷

Glencore and Aurizon supported the QCA's Draft Decision.¹²⁸ However, Aurizon said that there should be some additional amendments which allowed for an access holder to suggest changes to the MTP which did not otherwise impact on the running of the system.¹²⁹

¹²⁷ New Hope, sub. 23: 15–17.

¹²² Queensland Rail, 2015 DAU, Schedule F, cls. 2.2(e)–(h).

¹²³ Queensland Rail, sub. 1: 28.

¹²⁴ Queensland Rail, 2015 DAU, Schedule F, cls. 2.1(h), (j).

¹²⁵ See Queensland Rail, sub. 26: Schedule 7.

¹²⁶ Yancoal, sub. 27: 4.

¹²⁸ Glencore, sub. 25: 1; Aurizon, sub. 20: 5–6.

¹²⁹ Aurizon, sub. 20: 34–35.

QCA analysis and Decision

We do not consider that Queensland Rail's proposed Schedule F is appropriate. Instead, we require Queensland Rail to make amendments to Schedule F to impose stricter obligations on Queensland Rail to notify, consult and more appropriately seek agreement with access holders before making changes to the MTP and DTP.

We have largely adopted our Draft Decision position (see section 4.1 of the Draft Decision) in this regard, except for some additional amendments discussed below.

The QCA considers Queensland Rail should amend its proposed 2015 DAU, so that before changing the MTP or DTP, Queensland Rail is required to:

- notify an expanded range of affected parties about changes to its train and maintenance scheduling and planning documents
- consult with affected access holders in relation to all changes that affect access holders' scheduling
- seek agreement from access holders in more circumstances
- *delay implementation of disputed changes,* until the matter is resolved, except for urgent safety-related issues.

Notify

In our Draft Decision we proposed that Queensland Rail amend its proposed Schedule F so that Queensland Rail was obliged to notify affected parties of changes to the train services and other activities detailed in its planning and scheduling documents as early as possible and as often as necessary. This included notifying supply chain participants that are affected by those changes.

To do otherwise is inconsistent with an efficiently run network and contrary to the interests of access holders and access seekers.

Notifying affected parties is not, in our view, an onerous requirement. Queensland Rail is a sophisticated organisation that already actively notifies affected parties other than access holders. For example, Queensland Rail already publishes service updates and planned closures on its website for its related-party operations.¹³⁰ We anticipate that Queensland Rail could use a similar website approach for notifying parties about changes to train plans across its network that affect non-passenger access holders and other parties. This would limit any administrative costs, and thereby limit any negative impact on Queensland Rail's legitimate business interests (s. 138(2)(b)).

We accept Queensland Rail's position that it should not have to notify affected parties who do not opt in and that its obligation to notify should be able to be satisfied either by its portal or website, or in meetings (at Queensland Rail's discretion). This would have regard to Queensland Rail's legitimate business interests by minimising the administrative costs on Queensland Rail, while allowing notices to be available to those parties who wish to be notified. This particular suggestion was raised by the QCA in the Draft Decision.¹³¹

¹³⁰ This includes upcoming closures for the next 14 days and those planned over the next 12 months. In addition, Queensland Rail announces delays to services and provides clear, regular updates for its passenger operations. See http://www.queenslandrail.com.au/RailServices/City/Pages/Plannedclosures.aspx.

¹³¹ See QCA 2015: 74.

New Hope said that Queensland Rail should be required to provide six months' notice of proposed changes to the planned possessions which are expected to have a duration of greater than four days, and three months' notice of all other planned possessions (except emergency possessions). New Hope said that changes to planned possessions can have significant impacts on end users' logistics and impose material adverse cost impacts.

We agree with New Hope in relation to the effect of changes to planned possessions. Planned possessions are, and should be, part of a long-term planning schedule. If these stoppages occur without sufficient notice, users of the network can face substantial consequences upstream or downstream of the network. We consider that three months' notice is not a particularly long period when considering the planning that initially goes into an MTP. Accordingly, we require that the notice period for Queensland Rail changes to planned possessions should be three months.

We do not agree that the notice period for changes to planned possessions that will be longer than four days should be six months. We consider that a three-month notice period strikes a reasonable balance between flexibility and appropriate notice. This is also consistent with the amount of time which Queensland Rail requires from an access holder if it requests a change to an MTP.

Queensland Rail should be required to give three months' notice of modifications to an MTP (except for urgent and emergency possession). We consider that this period of notice is balanced and symmetrical with an access holder's notice requirements if it wishes to change the MTP. It also promotes upstream and downstream efficiency, as well as planning efficiency, and is appropriate in regard to section 138(2), most notably section 138(2)(a), (d), (e) and (h).

We remain of the view that an approach to notify a broader range of stakeholders of changes to the train plans when they are proposed and implemented will enable the efficient operation and use of significant infrastructure by providing affected parties with timely information that they could use to manage and mitigate the impact (ss. 138(2)(a) and 69E of the QCA Act). As such, we also consider this requirement to be in the public interest, and in the interests of access holders and seekers (s. 138(2)(d), (g), (e), (h)). Therefore, we consider it appropriate to require the 'opt-in' notification of a wide range of supply chain participants, having regard to all the criteria in section 138(2).

Summary 4.1

The NMPs in Schedule F of the 2015 DAU must provide that:

- (a) Queensland Rail is obliged to notify a broader range of parties about changes to its train plans. However, Queensland Rail is only required to notify parties (other than access holders) who have opted to be notified
- (b) Queensland Rail be required to give three months' notice of modifications to the MTP (except for urgent and emergency possessions).

See clauses 2.1(d), 2.1(m) and 2.2(c) in Schedule F to Appendix F.

Consult

In our Draft Decision, we required Queensland Rail to consult in a wider range of circumstances with access holders who were affected by changes to an MTP or DTP.

We do not accept Queensland Rail's position that our proposed amendments are impracticable, given the large number of changes and short planning windows leading up to the day of operation.¹³²

While it may advance Queensland Rail's legitimate business interests not to consult on operational constraints that affect the DTP, we consider that the absence of such consultation does not promote the efficient use and operation of rail infrastructure (s. 138(2)(a), (b)). Nor is it in the interests of access seekers or holders (s. 138(2)(e), (h). We therefore are of the opinion that the amendments outlined in our Draft Decision are appropriate, having regard to all the criteria in section 138(2).

We consider Queensland Rail should consult with access holders on all operational constraints that affect the access holder's scheduled paths on the DTP. Queensland Rail need only make reasonable endeavours to consult in the case of urgent or emergency possessions and pressing safety issues.¹³³ We understand that Queensland Rail consults on most operational constraints in practice, as consulting about changes to its DTP is an essential part of providing a service to its customers.

We also do not accept Queensland Rail's position that being obliged to consult with relevant access holders in relation to operational constraints which affect those access holders' scheduled paths on a DTP could put Queensland Rail in a position where it is unable to comply with its statutory obligations or incurs additional liability because third parties dictate safety requirements regarding its network.¹³⁴

This is because the obligation is merely to consult (and as discussed below, make reasonable endeavours to agree about changes). The effect of the amendments is to provide for some cooperation between parties who both have an interest in, and are affected by, the scheduling. Plainly, any statutory obligations or additional liabilities on Queensland Rail would be relevant to an assessment of the reasonableness of its action.

Given this, a broad obligation to consult is not an onerous requirement on Queensland Rail. To a large extent, our Decision formalises what occurs in practice anyway.

It also promotes the efficient use and operation of the rail network, as it provides access seekers and access holders with greater certainty they will receive a standard of service consistent with their TSEs (s. 138(2)(a), (e), (h)).

We agree with New Hope's suggested amendments to create an obligation to make access available based on the DTP and create a link between the MTP and DTP.¹³⁵ These amendments provide that a DTP is developed from, and consistent with, the applicable MTP. We consider that these suggested amendments provide clarity around Queensland Rail's obligations in relation to DTPs. We consider that this clarity further helps to promote the efficient use of and operation of the rail infrastructure, while clearly delineating each party's responsibilities in

¹³² Queensland Rail, sub. 26, Schedule 7: 6.

¹³³ Many safety issues will be addressed by urgent or emergency possessions, but some result in other measures such as speed restrictions.

¹³⁴ Queensland Rail, sub. 26, Schedule 7: 7.

¹³⁵ New Hope, sub. 23: 16.

relation to train scheduling (s. 138(2)(a), (b), (e), (h)). It may also advance Queensland Rail's legitimate business interests for obligations to access holders to be clearly outlined, so that each party knows where it stands in relation to each other (s. 138(2)(b)).

Summary 4.2

The NMPs in Schedule F of the 2015 DAU must:

- (a) provide for Queensland Rail to consult access holders in relation to all operational constraints that affect the access holder's scheduled paths on the DTP, except in the case of urgent or emergency possessions, when Queensland Rail need only make reasonable endeavours to consult.
- (b) clarify Queensland Rail obligations in relation to making access available based on the relevant DTPs.

See clauses 2.2(b), 2.2(e), 2.2(f), 2.2(j) and 2.2(j)(iii) in Schedule F to Appendix F.

Seek agreement

The 2015 DAU provided that Queensland Rail would seek agreement from access holders for all changes to the MTP that were not consistent with contracted TSEs.

In our Draft Decision we also required that Queensland Rail make reasonable endeavours to agree with access holders about changes to the DTPs that affected those access holders' TSEs.

In doing so, we accepted that Queensland Rail could make an exception for urgent and emergency possessions and pressing safety matters such as speed restrictions, but we still proposed that its planners and controllers should make reasonable endeavours to consult about changes to the DTP.

We do not agree with New Hope's suggestion to delete the reference to urgent possessions. We believe that Queensland Rail is currently in the best position to understand its own requirements in relation to emergency and urgent possessions.

Queensland Rail said:

Seeking agreement ... provides no perceivable benefit ... as Network manager, Queensland Rail already ensures that it consults with affected parties on changes.¹³⁶

Queensland Rail did not accept our Draft Decision, but considered that the obligation to seek agreement, despite the fact that it already consulted with affected parties, would impose a significant administrative burden.¹³⁷

As Queensland Rail already consults with affected parties, we cannot accept the assertion of an added significant administrative burden. Our required amendment to seek agreement does not impose obligations on Queensland Rail in relation to non-affected parties, only relevant access holders. Further, our amendments only require Queensland Rail to use reasonable endeavours to seek agreement. Our required amendments would therefore not create an oppressive administrative burden.

Timeliness in respect of the start and finish of planned possessions is a fundamental indicator of the efficiency of the management of a rail network. As the name implies, planned possessions

¹³⁶ Queensland Rail, sub. 26, Annexure 7: 7.

¹³⁷ Queensland Rail, sub. 26, Annexure 7: 7.
are typically settled well in advance—sometimes two years or more before the day of operation. If planned possessions are changed at a late stage, a range of parties that have relied on those long-established plans will be adversely affected. Therefore, varying the DTP from the MTP for changes to planned possessions should be an unusual event, particularly as all traffics on Queensland Rail's network are timetabled.

We therefore consider that Queensland Rail's proposed requirements for making such changes are not appropriate having regard to the interests of access seekers and holders in terms of giving them certainty about receiving their TSEs (s. 138(2)(e), (h)). They also hinder the efficient use and operation of the rail infrastructure (s. 138(2)(a)), as they place too little onus on Queensland Rail to adhere to the timing of its planned possessions. Therefore, while it may be in Queensland Rail's legitimate business interest to change the time of possessions (s. 138(2)(b), we consider that Queensland Rail's proposal is not appropriate.

As mentioned above, we do not consider it appropriate that Queensland Rail should be required to agree with affected access holder in relation to changes to the DTP. Rather, we require that Queensland Rail make reasonable endeavours to agree any changes to planned possessions in the DTP compared with the MTP, where those changes affect TSEs. Also, we consider that Queensland Rail should be able to vary the DTP from the MTP without seeking agreement from affected access holders in cases of emergency possessions and pressing safety issues.

We would be concerned, however, if Queensland Rail consistently failed to adhere to the timing of planned possessions. We therefore require Queensland Rail to report on whether it has adhered to the timings of the planned possessions in its MTP.

This reasonable-endeavours regime for changes to planned possessions in the DTP supports the efficient use and operation of the rail network, and the interests of access seekers and holders, in giving them certainty about receiving their TSEs (s. 138(2)(a), (e), (h)).

Summary 4.3

The NMPs and reporting requirements in the 2015 DAU must provide that Queensland Rail:

- (a) make reasonable endeavours to seek agreement from affected access holders where it varies the DTP from the MTP, except for emergency possessions and pressing safety issues
- (b) report on its adherence to timings of planned possessions in the MTP.

See clause 5.1.2(a)(x) and Schedule F, clause 2.2(f) in Appendix F.

Delaying disputed changes until resolution

We require Queensland Rail to amend its proposed Schedule F, so that where an access holder disputes a change to the MTP, other than in cases of an emergency or urgent possession or a pressing safety matter, then the change should take effect once the dispute is resolved via the dispute resolution mechanisms in the 2015 DAU.

We consider that allowing changes to the MTP to go ahead while they are subject to a dispute would be detrimental to the interests of access seekers and holders (s. 138(2)(e), (h)) and that this outweighs any legitimate business interest Queensland Rail may have in going ahead with changes it has decided are desirable.

Asciano said that either truncated timeframes or an alternative to the DAU dispute resolution provisions were required to allow a timely resolution of disputes over train plan changes.¹³⁸ We are not convinced of the merits of a separate truncated process applying in this circumstance. We consider it sufficient that the parties should act reasonably in any event, and consider that this should provide for timely resolution of disputes.

We do not agree with Queensland Rail's (and implicitly Asciano's) position that our proposed amendment will lead to inefficiencies and disruptions to the running of the network in some cases. Nor do we agree with Queensland Rail's proposition that stakeholders could use the process frivolously, as there is no evidence to date that stakeholders have acted in such a manner.¹³⁹

Relevantly, we note that for changes to the MTP an access holder's agreement is 'not to be unreasonably withheld'. We have, in addition to the changes required in the Draft Decision, also included that the dispute provision operates in relation to 'bona fide' disputes. We consider that these factors should mitigate against any fears Queensland Rail has about 'frivolous' disputes.

Also, given that any change to the MTP that is not an urgent or emergency possession or pressing safety matter should, as noted by Queensland Rail¹⁴⁰, occur at least three months before the day of operation, Queensland Rail and its access holders should have sufficient time to resolve disputes under the provisions in the access agreements. We consider that this reasoning also applies to Asciano's concerns.

Our required amendments promote the efficient use and operation of the rail network and the interests of access seekers and holders (s. 138(2)(a), (e), (h)).

Summary 4.4

The rules for changes to train plans in Schedule F of the 2015 DAU must provide for delaying changes to the MTP until related bona fide disputes are resolved.

See clause 2.4 in Schedule F to Appendix F.

4.2 Minimising the adverse effects of operational constraints

Queensland Rail proposed to use reasonable endeavours to minimise any material adverse effects of planned, urgent or emergency possessions, that prevented train services from operating 'substantially in accordance with the Access Holder's Train Service Entitlement'.¹⁴¹

In our Draft Decision, we proposed that Queensland Rail be required to use reasonable endeavours to minimise the material adverse effects in relation to all operational constraints. We also proposed that Queensland Rail be required, when mitigating material adverse effects caused by changes to the MTP or DTP, to use reasonable endeavours to offer substitute train paths that an access holder could actually use.¹⁴²

¹³⁸ Asciano, sub. 28: 17.

¹³⁹ Queensland Rail, sub. 26, Annexure 7: 7–8.

¹⁴⁰ Queensland Rail, sub. 26, Annexure 7: 7.

¹⁴¹ Queensland Rail, 2015 DAU, Schedule F, cl. 2.3.

¹⁴² See QCA 2015: 78–80 and the QCA's 2015 DAU cl. 2.3, which introduced the concepts of 'useable schedule time' and 'alternative schedule time'.

Stakeholders' comments

Stakeholders agreed with our required amendments, but suggested some further amendments.¹⁴³

Queensland Rail did not accept our required amendments and said they did not balance the interests of all parties appropriately.¹⁴⁴

QCA analysis and Decision

We are of the view that Queensland Rail should use reasonable endeavours to minimise any resulting material adverse effects of all operational constraints, consistent with the obligations in its access agreements. We are also of the view that where Queensland Rail intends to provide a replacement path, the replacement path should be, within reason, a path the access holder is able to use efficiently.

Our Decision has adopted the Draft Decision position in this regard (see section 4.2 of the Draft Decision).

Obligation to address adverse effects

Any operational constraint has the potential to result in an access holder's scheduled train services not being met.

But Queensland Rail's proposal that it will use reasonable endeavours to address adverse effects only in relation to possessions means that a range of disruptions including speed restrictions and other safety-related matters will not be covered.¹⁴⁵

Minimising adverse effects only in relation to operational constraints which are possessions is not in the interest of access seekers or access holders in receiving their TSEs, or consistent with the efficient use and operation of the rail network (s. 138(2)(a), (e), (h)). This concern outweighs any legitimate business Queensland Rail may have in leaving the adverse effects unaddressed (s. 138(2)(b)). We therefore cannot consider Queensland Rail's proposal to be appropriate having regard to all the criteria in section 138(2).

Accordingly, we require that Queensland Rail use reasonable endeavours to minimise the effects of operational constraints wherever a TSE is affected. This broader responsibility on Queensland Rail is reasonable because the proposed requirement is not open-ended. In particular, in using 'reasonable endeavours to minimise' the effects of the change, Queensland Rail may take into account a range of commercial and operational matters, including safe operation of the network.¹⁴⁶

We consider that this appropriately balances the interests of all parties.

We consider that this is appropriate having regard to the legitimate business interest of Queensland Rail as it provides enough flexibility in complying with the requirement (s. 138(2)(b)). It also places sufficient onus on Queensland Rail to consider the interests of access seekers and holders in receiving their TSEs (s. 138(2), (e), (h)).

¹⁴³ See New Hope, sub. 23: 16–17; Yancoal, sub. 27: 4; Glencore, sub. 25: 1; Aurizon, sub 20: 5–6.

¹⁴⁴ Queensland Rail, sub. 26, Schedule 7: 9. Queensland Rail's detailed comments were limited to those made in relation to the new definitions and obligations in relation to alternative scheduled times.

¹⁴⁵ Queensland Rail, 2015 DAU, Schedule F, cl. 2.3(a).

¹⁴⁶ Queensland Rail, 2015 DAU, Schedule F, cls. 2.3(a) and (b).

Useful train paths

In our Draft Decision, we also included two new definitions which had the effect of requiring Queensland Rail to have regard to the usefulness of the alternative schedule times which it offered in substitute for varied paths of access holders.

All stakeholders, except Queensland Rail, agreed with these required amendments. New Hope suggested some additional amendments to the definition of 'useable schedule time'.¹⁴⁷ Aurizon and Asciano also suggested that if Queensland Rail cannot offer a useable schedule time, and the offered replacement path is not useable by an operator, it should be recorded as a Queensland Rail Cause.¹⁴⁸ Queensland Rail considered that the obligations should be on each operator to accommodate the alternative paths which Queensland Rail offers in substitution for varied paths.¹⁴⁹

Despite the concerns raised by Queensland Rail in its December 2015 submission, we are of the opinion that Queensland Rail should be required to have regard to the utility of any alternative paths. We consider that this is the appropriate outcome for a number of reasons, including:

- Queensland Rail is in the best position to have an overall understanding of the operation of the network and the relevant available paths and schedules.
- Most variations to the MTP and DTP will be as a result of Queensland Rail's changes. Queensland Rail has contracted to provide access and has agreed to provide train paths in accordance with the MTP and DTP. If these are changed, there could be substantial inconvenience and cost to an operator.
- The required amendments do not impose an absolute obligation. They only require Queensland Rail to use reasonable endeavours. We consider that the reasonable endeavours test still provides some flexibility for Queensland Rail if it cannot, after consulting with the relevant access holder, provide an ideal replacement.

The current replacement path rules are not in the interests of access seekers or access holders in receiving their TSEs (s. 138(2)(e), (h)). In particular, Queensland Rail's proposal is one-sided to the extent that it is in conflict with the interests of access seekers and holders, which could make the operation and use of the network less efficient (s. 138(2)(a), (e), (h)).

We do not consider it appropriate, given the required amendments which place additional obligations on Queensland Rail regarding variations to the MTP and DTPs, to include New Hope's suggested amendments or Aurizon's and Asciano's suggestions in relation to increasing the scope of the definition of 'Queensland Rail Cause'.¹⁵⁰

We consider that our required amendments will increase the obligations on Queensland Rail sufficiently. Our amendments will have the effect of minimising interruptions to access holders' schedules, allow for disputes to be lodged with the QCA, and require Queensland Rail to consult and seek agreement where relevant. We consider that these changes strike an appropriate balance between Queensland Rail's and access holders' interests. We consider that to include New Hope's suggestions and/or to increase the scope of the definition of 'Queensland Rail Cause' would tip the balance too far in favour of one party.

¹⁴⁷ See New Hope, sub. 23: 16–17; Yancoal, sub. 27: 4; Glencore, sub. 25: 1; Aurizon, sub. 20: 5–6.

¹⁴⁸ Aurizon, sub. 20: 45; Asciano, sub. 28: 17.

¹⁴⁹ Queensland Rail, sub. 26: 9–10.

¹⁵⁰ Aurizon, sub. 20: 45.

Summary 4.5

The operational constraints provisions in Schedule F of the 2015 DAU must provide that Queensland Rail is required to use reasonable endeavours to minimise the material adverse effects of all operational constraints and also use reasonable endeavours to offer useable replacement train paths.

See clauses 2.3(a) and 2.3(c) in Schedule F, and definitions of 'alternative schedule time' and 'useable schedule time' in Appendix F.

4.3 Coordination with adjoining networks

Queensland Rail's 2015 DAU proposal

Queensland Rail proposed to use reasonable endeavours to consult with other railway managers on coordinating maintenance activities, developing MTPs and amending the ORM, and to minimise adverse effects on through-running trains.¹⁵¹

In our Draft Decision, we proposed making the obligations to consult and minimise adverse effects in relation to through-running trains more robust.¹⁵²

Stakeholders' comments

Stakeholders supported our proposed amendments and said that that coordination with other railway managers is essential to the efficient use and operation of Queensland Rail's infrastructure.¹⁵³ New Hope said 'the rail network will not operate efficiently unless Queensland Rail is properly engaged in alignment/coordination activities'.¹⁵⁴

Queensland Rail did not accept our proposed amendments. Queensland Rail said that the QCA had no authority to require Queensland Rail to consult or otherwise communicate with other railway managers.¹⁵⁵

QCA analysis and Decision

We consider Queensland Rail should always consult with adjoining network managers on scheduling and other operating matters affecting both networks rather than just using reasonable endeavours to consult 'as relevant' and 'from time to time'. We consider that rather than merely having a 'view to' minimising adverse effects, Queensland Rail should be required to use reasonable endeavours to minimise the effect of any scheduling decisions or changes on through-running trains.

We have adopted our Draft Decision position in this regard (see section 4.3 of the Draft Decision).

We consider that Queensland Rail's 2015 DAU proposal is too weak a level of obligation for such an important matter given that a large proportion of freight services contracted to use Queensland Rail's network use track managed by other operators for part of their journey. This

¹⁵¹ Queensland Rail, 2015 DAU, cl. 4.2.

¹⁵² See QCA 2015: 81–82.

¹⁵³ See New Hope, sub. 23: 17.

¹⁵⁴ New Hope, sub. 10: 39.

¹⁵⁵ Queensland Rail, sub. 26: 11.

includes all services travelling along the North Coast line between Gladstone and Rockhampton.¹⁵⁶

Consultation with other adjoining networks involves giving other parties the information about the operation of Queensland Rail's below-rail infrastructure, which is necessary to enable effective access to the declared service. Furthermore, upstream and downstream systems, and the network itself, must be coordinated at the points of overlap to avoid interruptions to train services and be delivered efficiently and in a timely manner. To do otherwise, would preclude effective access to the services provided by Queensland Rail's below-rail infrastructure.

We do not agree with Queensland Rail that we do not have the authority to impose this obligation on Queensland Rail.

We have had regard to Queensland Rail's legitimate business interests in considering its proposed rules for scheduling through-running trains¹⁵⁷ and we do not consider that this obligation will impose a significant financial burden on Queensland Rail. We consider that Queensland Rail's proposal is not appropriate, having regard to the interests of access holders and seekers in receiving their TSEs, as it is too narrow, given the importance of proper scheduling to through-running trains.

Queensland Rail's proposal, therefore, does not promote the efficient use and operation of the rail infrastructure (s. 138(2)(a), (e), (h)).

We require that Queensland Rail consult on scheduling changes that affect other railway managers and that it use reasonable endeavours to minimise the effects of these changes, rather than just having 'a view' to doing so.¹⁵⁸

Summary 4.6

The 2015 DAU must provide that Queensland Rail is required to consult with other railway managers on scheduling and other matters affecting both networks and use reasonable endeavours to minimise the effect on through-running trains.

See clause 4.2 in Appendix F.

4.4 Passenger services

We require that Queensland Rail amend the NMPs in its 2015 DAU so that a network controller must be 'acting reasonably' when forming a belief that it is necessary to give priority to passenger train services.

We have adopted our Draft Decision position in this regard except for the change discussed below (see section 4.4 of the Draft Decision).

Stakeholders' comments

Stakeholders agreed with our Draft Decision. New Hope said it:

¹⁵⁶ See also New Hope's discussion on the various possible futures for the West Moreton system: New Hope, sub. 23: 17.

¹⁵⁷ Trains from adjoining networks which interact with Queensland Rail's network.

¹⁵⁸ Queensland Rail, 2015 DAU, cl. 4.2.

appreciates the legislative requirement for passenger priority, but it is not in the interests of Access Seekers or Access Holders, the public interest or efficient use of the infrastructure for passenger priority to become a cloak for poor planning and scheduling practices.¹⁵⁹

Queensland Rail objected to our proposed amendment, saying that the QCA Act did not override the TI Act and that Queensland Rail, if it failed to comply with its passenger priority obligations, might face substantial civil penalties.¹⁶⁰

QCA analysis and Decision

We accept that Queensland Rail has legislative obligations in relation to passenger trains. However, we do not consider that our required amendments result in the QCA Act overriding the TI Act. Nor do we accept the implicit proposition that, by having a network controller act reasonably, Queensland Rail will somehow become more exposed to penalties under the TI Act.

The intention and effect of the amendments is simply to provide that a belief formed by the network controller must be reasonable. This is not a particularly onerous standard and one that a network controller should be meeting in any event. The explicit inclusion of this standard removes the unfettered discretion which, although not likely, could be used by Queensland Rail to cloak poor planning and scheduling.

For the most part, Queensland Rail's 2015 DAU provisions provide a balance between the public interest in safe and timely operation of passenger trains, the legitimate business interests of Queensland Rail as operator of those passenger trains and provider of below-rail services, and the interests of non-Queensland-Rail access seekers and holders (s. 138(2)(d), (b), (e), (h)).

We have accepted Queensland Rail's submission that the subclause (cl. 2.2(e)(i)(C) of Schedule F) which allows Queensland Rail to vary the DTP in relation to its passenger trains where no other access holder is affected should be reinstated (we proposed in the Draft Decision to delete this subclause). We consider that this is appropriate given the checks and balances we have applied to other variations to the MTP and DTP.

However, Queensland Rail has not specified that the network controller must act reasonably in determining that it is necessary to act to favour passenger services in such circumstances. Given that, in relation to passenger services, Queensland Rail is a related access provider¹⁶¹, Queensland Rail may have a potential conflict of interest and be inclined to support actions (beyond those provided in the QCA Act or envisioned by the QCA in relation to peak services in the metropolitan system) to prevent passenger trains from being late. Similarly, without a decision having to be based on relevant, objective evidence, decisions which may have a materially adverse effect on access holders could be made based on any number of irrelevant factors.

We therefore consider Queensland Rail's passenger service provisions do not appropriately balance the interests of Queensland Rail and access seekers and access holders. As such, we do not consider it appropriate to approve Queensland Rail's proposal having regard to all the approval criteria in section 138(2).

We consider that it is appropriate to require the network controller to act reasonably when forming a view about scheduling to favour passenger trains because it advances the interests of access holders and promotes the efficient use and operation of Queensland Rail's network. We

¹⁵⁹ See New Hope, sub. 23: 17.

¹⁶⁰ Queensland Rail, sub. 26, Schedule 7: 11–16.

¹⁶¹ See QCA Act, definition of 'related access provider'.

do not consider the amendments to be contrary to Queensland Rail's legitimate business interests because we do not consider that that amendments would cause Queensland Rail to fail to satisfy its obligations either under the TI Act or the pending access undertaking (s. 138(2)(a)).

We also adopt Section 4.5 of our October 2014 Draft Decision and require Queensland Rail to clearly specify the NMPs in the undertaking will apply to all services including Queensland Rail's own passenger services.

Summary 4.7

The operating requirements and NMPs in the 2015 DAU must:

- (a) clearly specify that the NMPs will apply to all services including Queensland Rail's own passenger services; and
- (b) provide that a network controller is required to be 'acting reasonably' when forming a view that it is necessary to give priority to passenger services.

See clause 4.1(d) and Schedule F, clauses 3(i)(i), 3(i)(ii) in Appendix F.

4.5 Operating Requirements Manual

Queensland Rail has proposed that a variety of rules and procedures for use of the network by train operators be set out in the ORM. These standard provisions, most of which were included in the SAA in the previous 2008 undertaking, are common across the network and not subject to individual variation between different access agreements. They address, among other things:

- interface risk management¹⁶², including environmental risk management
- safe working procedures and safety standards
- incident and emergency response procedures
- various technical requirements for train control and network planning
- requirements such as those for forecasts by the operator of expected train services and how and when safety notices will be issued.¹⁶³

Queensland Rail said in its material accompanying the 2015 DAU that the ORM reflected 'an appropriate allocation of risks for its business'.

In our Draft Decision, we proposed that the ORM be amended in a number of places to make certain requirements more reasonable, improve its operation and clarity, and enable a proper fit with the 2015 DAU and SAA.

Stakeholders' comments

Stakeholders other than Queensland Rail agreed with the amendments we proposed in our Draft Decision. Aurizon said that, in addition to the QCA's proposed amendments, Queensland

¹⁶² How different stakeholders on the network interface with each other.

¹⁶³ Queensland Rail, 2015 DAU, schedule G.

Rail should also be obliged to consult with access holders in relation to changes to the ORM due to safety matters and also in relation to changes to the interface standards.¹⁶⁴

Aurizon also noted a number of specific objections to the ORM which was submitted by Queensland Rail in December 2015.¹⁶⁵ New Hope said it supported the QCA's proposed amendments as an appropriate rebalancing but also noted some specific comments.¹⁶⁶

Asciano said that the obligation on an operator to provide contact details should be reciprocal. It also said that if the dispute resolution process under the access undertaking did not apply to the ORM then that should be stated explicitly.¹⁶⁷

Queensland Rail said that, generally, the QCA's proposed amendments made the ORM unnecessarily prescriptive and limited Queensland Rail's ability to plan and respond to demand on the network. Queensland Rail also highlighted its concerns in relation to particular provisions within the QCA's proposed ORM.¹⁶⁸

Queensland Rail submitted an amended ORM which reverted to its previous version, except for changes that Queensland Rail said it considered necessary to update the ORM to comply with the *Transport (Rail Safety) Act 2010* and Queensland Rail's systems and procedures. This ORM largely disregarded the QCA's proposed amendments.¹⁶⁹

QCA analysis and Decision

We consider that Queensland Rail should amend the ORM to balance the obligations and requirements between Queensland Rail and train operators, and to clarify how various procedures will operate and link with relevant provisions in the undertaking and SAA.

This Decision has largely adopted the Draft Decision in this regard; however, we have made some additional changes to the ORM (see section 4.5 of the Draft Decision).

We consider that much of the ORM proposed by Queensland Rail as part of its 2015 DAU represents a reasonable way of moving a variety of procedures from individual access agreements to a document that will apply to all access holders.

We also accept a number of Queensland Rail's most recent amendments as reasonable, appropriate and conforming to Queensland Rail's current practice.

However, we accept stakeholders' concerns that a variety of amendments to the ORM are also required to improve the balance of risks and responsibilities between Queensland Rail and its access holders, and make various provisions clearer or more reasonable.¹⁷⁰

As noted above, Queensland Rail only provided comments in relation to some of the amendments we required in the Draft Decision. The remainder were not raised, but were presumably not accepted, as evidenced by Queensland Rail's December 2015 version of the ORM, which does not contain the QCA's proposed amendments.

¹⁶⁴ Aurizon, sub. 20: 45–46, 54.

¹⁶⁵ Aurizon, sub. 29: 15.

¹⁶⁶ New Hope, sub. 32: 40–43.

¹⁶⁷ Asciano, sub. 28: 18.

¹⁶⁸ Queensland Rail, sub. 26: 105.

¹⁶⁹ See Queensland Rail, sub. 26, Annexure 6.

¹⁷⁰ Aurizon, sub. 6: 29, 30, 31–32; Asciano, sub. 5: 27–29. The various matters covered by the ORM are discussed in more detail in our October 2014 Draft Decision—see QCA 2014d: 71–74, 178–187.

We agree with Aurizon that Queensland Rail should be required to provide full details of any complaints (see cl. 2.1(d)(D) of the proposed ORM in the Draft Decision).¹⁷¹ Queensland Rail proposed to provide only a summary of any complaints. However, we agree that full details of any complaints will allow the most appropriate consideration of the complaints in any interface risk assessment.

Aurizon also supported the QCA's proposed amendments in relation to baseline environmental data and said that, if applicable, the requirements for environmental controls should be based on an understanding of the existing environmental conditions.¹⁷²

Moreover, while it may advance Queensland Rail's legitimate business interest to manage its network in a way that it considers safe and efficient, in some cases its proposed ORM goes beyond what is required from Queensland Rail in order to meet its regulatory and contractual obligations (s. 138(2)(b)). In particular, Queensland Rail's proposal is one-sided in areas such as emergency responses and the treatment of baseline environmental standards to the extent that it is not in the interest of access seekers and holders and is likely to make the operation and use of the network less efficient (s. 138(2)(a), (e), (h)). Therefore, while we accept much of Queensland Rail's proposed ORM, we do not consider it appropriate to approve it as submitted, having regard to all the approval criteria in section 138(2).

Table 7 provides our responses to those aspects of the QCA's proposed ORM on which Queensland Rail did comment.

QCA reference clause	Queensland Rail concern ¹⁷³	Stakeholders' comments	QCA response
2.2(b)	It cannot update its safety management systems to accommodate the requirements of individual operators.	New Hope said safety was the objective rather than avoiding changes to Queensland Rail's safety management system. ¹⁷⁴	It is appropriate that the IRMP consider the obligations of both parties in relation to the risks identified. Queensland Rail can update its safety management systems in appropriate circumstances.
2.4	The amendments in relation to the removal of assumptions regarding baseline environmental standards were unbalanced, uncommercial and made obligations reciprocal, whereas the risks were solely the preserve of the operator.	New Hope said that Queensland Rail had no reasonable basis to avoid its responsibilities for environmental risks. ¹⁷⁵	The QCA's amendment does away with the previous assumption that the network is taken to meet all environmental standards. Future environmental impacts should be assessed based on all available information, without being constrained by an assumption that is not supported by data. This should not require Queensland Rail to undertake baseline studies of the whole network. It is also more balanced than Queensland Rail's proposal

¹⁷⁴ New Hope, sub. 32: 40.

¹⁷¹ Aurizon, sub. 29: 15.

¹⁷² Aurizon, sub. 29: 15.

¹⁷³ See Queensland Rail, sub. 26: 105–106 for summaries of Queensland Rail's concerns.

¹⁷⁵ New Hope, sub. 32: 41.

QCA reference clause	Queensland Rail concern ¹⁷³	Stakeholders' comments	QCA response
			which places all the risk on the access holder.
2.6	The amendment to include the EIRMR ¹⁷⁶ process was unnecessary, as identification and management of environmental risks was specifically dealt with in the IRMP process.	New Hope said that the process proposed by the QCA was quite specific to environmental risks. ¹⁷⁷	The required amendments clarify each party's responsibility and the process in relation to the EIRMR. However, we have removed some duplication and simplified the drafting (the relevant clause is now cl. 2.5 in Schedule G to Appendix F).
2.6(j) ¹⁷⁸	Unresolved environmental matters disputes should be referred to an appropriate expert not the QCA.	New Hope said that the QCA could engage experts on matters such as this. ¹⁷⁹	The QCA, when conducting an arbitration, can have regard to suitable experts if necessary (see s. 197(1)(e) of the QCA Act).
3.1	The operation of a 'reasonableness' test is unclear and may lead to the 'watering down' of Queensland Rail's environmental and safety requirements.	New Hope said that Queensland Rail was being unnecessarily concerned with exceptions to safe- working procedures and standards. ¹⁸⁰	A requirement to act reasonably will not have the effects Queensland Rail suggests. Rather, including these words helps to minimise the risk of frivolous and opaque decision-making.
4.3	The operation of a 'reasonableness' test in practice was unclear.	New Hope said the 'reasonableness' clause provided appropriate balance. ¹⁸¹	The inclusion of the obligation to act reasonably would not be vague or unclear in practice. Rather, it provides that directions should be based on evidence. Queensland Rail's directions should be objectively reasonable.
6.5(c), 6.8, 6.9	The amendments to its communication system change notification requirements imposed unduly onerous obligations.	New Hope said that the obligation to update was reasonable given that Queensland Rail was the owner of the radio network. ¹⁸² New Hope also considered the changes to cls. 6.8 and 6.9 to be reasonable. ¹⁸³	Queensland Rail is already able to communicate with operators and stakeholders via its website, which is not onerous. A similar method of communication would suffice. Queensland Rail did not otherwise substantiate its concerns.
7.1.1	The 'reasonableness' test in relation to the provision of	New Hope said it was entirely reasonable for	See our response to Queensland Rail's comments in

¹⁷⁶ Environmental Investigation and Risk Management Report.

- ¹⁷⁸ Now 2.5(i) in Schedule G to Appendix F.
- ¹⁷⁹ New Hope, sub. 32: 41.
- ¹⁸⁰ New Hope, sub. 32: 41.
- ¹⁸¹ New Hope, sub. 32: 41.
- ¹⁸² New Hope, sub. 32: 42.
- ¹⁸³ New Hope, sub. 32: 42.

¹⁷⁷ New Hope, sub. 32: 41.

QCA reference clause	Queensland Rail concern ¹⁷³	Stakeholders' comments	QCA response
	safety updates in practice was unclear.	Queensland Rail to provide safety updates. ¹⁸⁴	relation to cl. 4.3 (above).
'Safety Standards'	The definition of safety standards should be limited to those standards relevant to the operator's activities and Queensland Rail should only be required to provide an operator with Queensland Rail's internal standards.	New Hope acknowledged that the definition was quite broad and might include information that Queensland Rail did not have. ¹⁸⁵	We accept that Queensland Rail should only be required to provide an operator with Queensland Rail's internal standards. However, the definition of safety standards should be broad enough to include standards which are applicable but not necessarily Queensland Rail's internal standards. This is because there may be standards, in addition to Queensland Rail's internal standards, to which operators must comply. To the extent that Queensland Rail is aware of other safety requirements that are relevant to trains operating on its network, it should be obliged to make the operator aware of them.
'Safeworking Procedures'	The definition of safeworking procedures should be limited to those standards relevant to the operator's activities.	New Hope said the QCA's amendments were appropriate. ¹⁸⁶	Our decision is the same as above ('Safety Standards').

Interface standards

We consider it appropriate to limit Queensland Rail's discretion to unilaterally amend or vary the interface standards¹⁸⁷ without consultation, as Queensland Rail's maintenance requirements are linked to these interface standards. If Queensland Rail has the ability to unilaterally amend the interface standards without consultation, it erodes the certainty which access seekers and holders have in relation to Queensland Rail providing the contracted level of access. While this may advance Queensland Rail's legitimate business interests (s. 138(2)(b)), it is contrary to the interests of access seekers and access holders or the public interest in having an efficient and competitive network (s. 138(2)(a), (d), (e), (h)). Accordingly, we do not consider it appropriate to approve an ORM that includes a unilateral ability to amend interface standards without consulting with access holders, having regard to all the approval criteria in section 138(2).

¹⁸⁴ New Hope, sub. 32: 42–43.

¹⁸⁵ New Hope, sub. 32: 43.

¹⁸⁶ New Hope, sub. 32: 43.

¹⁸⁷ The interface standards are Queensland Rail's minimum requirements or standards relating to the interface between a train and the network with which the applicable rolling stock and train configurations must comply in order to operate on the network.

Interface risk training and environmental matters

Some of our more significant proposed amendments relate to interface risk training, environmental standards and environmental risk management.

This includes a requirement to provide training to an access seeker/holder's staff or contractors on how to address an interface risk, where they can only obtain that training from Queensland Rail.¹⁸⁸ The requirement is appropriate, as it also provides that Queensland Rail will be able to recover a reasonable commercial charge for providing the training. Our proposed amendment reflects the drafting of an equivalent clause in the previous 2008 undertaking SAAs.¹⁸⁹

In addition, we require that the ORM include an environmental risk management process to specify how the operator will prepare an EIRMR and agree it with Queensland Rail.¹⁹⁰ Queensland Rail said that our required amendments were overly prescriptive and that the IRMP already included the obligation to undertake an EIRMR. Whilst we agree that the EIRMR provisions can be relaxed slightly (and have so amended the ORM accordingly), we consider a specified EIRMR process is more efficient and better protects the interests of access seekers and holders. Our proposed amendments are largely based on the drafting of an equivalent clause in the previous 2008 undertaking SAA.¹⁹¹

Other amendments

We also require a range of other amendments that improve the operation of the ORM and the procedures it specifies, including to:

- amend the definition of 'comparison train length' to provide for variation of parameters;¹⁹²
- specify that a sample Interface Risk Management Plan (IRMP) be published on Queensland Rail's website to give new access seekers an indication of what is addressed in the document, as it is required in order to conclude an access agreement;¹⁹³
- specify that an operator's obligations to provide emergency response and incident management plans will be subject to the terms of an access agreement;¹⁹⁴
- require that operators notify only the train control centre about contact details;¹⁹⁵
- require that Queensland Rail consult (but not agree on) the location of train crew breaks;¹⁹⁶
- require that the operator's controller (not train crew) notify Queensland Rail's controller and consult about crew changes;¹⁹⁷
- provide for Queensland Rail's controllers to use reasonable endeavours to relay messages between an operator's controllers and train crew;¹⁹⁸

¹⁸⁸ Appendix F, Schedule G, cl. 2.3.

¹⁸⁹ Queensland Rail, 2008 undertaking, Operator SAA, cl. 11(h).

¹⁹⁰ Appendix F, Schedule G, cl. 2.5.

¹⁹¹ Queensland Rail, 2008 undertaking, Operator SAA, cl. 8.1.

¹⁹² Appendix F, Schedule G, definitions.

¹⁹³ Appendix F, Schedule G, cl. 2.1(a).

¹⁹⁴ Appendix F, Schedule G, cls. 4.1 and 4.2.

¹⁹⁵ Appendix F, Schedule G, cl. 6.2(a).

¹⁹⁶ Appendix F, Schedule G, cl. 6.3(a).

¹⁹⁷ Appendix F, Schedule G, cl. 6.3(d).

¹⁹⁸ Appendix F, Schedule G, cl. 6.3(e).

- require Queensland Rail to notify operators about changes to network control radio channels 'as soon as reasonably possible'; and¹⁹⁹
- provide for Queensland Rail to notify operators about changes to online documents or the location of control centres and interface points.²⁰⁰

In addition, we require several minor changes of wording that balance the obligations so that Queensland Rail bears joint or equal responsibility for complying. These include changes to the sections on:

- the contents of the IRMP²⁰¹
- environmental risks to be considered²⁰²
- emergency responses²⁰³
- operational meetings.²⁰⁴

These changes are appropriate, as without them, Queensland Rail's ORM is imbalanced in Queensland Rail's favour, while reducing clarity and transparency in Queensland Rail's accountability as the railway manager. To do otherwise would not be appropriate with regard to section 138(2). We note that a number of these provisions differ from the ORM attached to the Draft Decision where we have accepted changes proposed by Queensland Rail in its submissions provided since the Draft Decision.

We have also included a number of the amendments to the ORM which Queensland Rail proposed in its December 2015 submission. These include amendments made by Queensland Rail to clause 2.1(d)(i) (except in relation to only providing a summary of complaints (as discussed above)), parts of clause 2.1(d)(ii), parts of clause 2.2, parts of clause 2.4, changes to clause 2.6 (previously cl. 2.5) and clause 4.4 of Schedule G to Appendix F. In response to these amendments, we have also simplified and streamlined the 'environmental risks' and process provisions and removed duplications (as noted above).

We consider that the above changes, individually and together, do not place unduly onerous requirements on Queensland Rail and they provide for Queensland Rail to improve efficiency and minimise disputes. Also, they support the efficient operation of the supply chain and are therefore in the public interest and in the interest of access seekers and holders (s. 138(2)(d), (e), (h)). In addition, they promote the efficient use and operation of Queensland Rail's infrastructure, as well as the public interest and the interests of access seekers and holders in receiving their TSEs (s. 138(2)(a), (d), (e), (h)).

¹⁹⁹ Appendix F, Schedule G, cl. 6.5(c).

²⁰⁰ Appendix F, Schedule G, cls. 6.8(b), 6.9(b).

²⁰¹ Appendix F, Schedule G, cls. 2.2(a)(i)–(vi), (b), (c).

²⁰² Appendix F, Schedule G, cls. 2.4(a)–(e).

²⁰³ Appendix F, Schedule G, cl. 4.3.

²⁰⁴ Appendix F, Schedule G, cls. 7.3.1(c), (d).

Summary 4.8

The ORM in the 2015 DAU must balance the obligations and requirements between Queensland Rail and train operators, clarify how various procedures will operate and link with relevant provisions in the undertaking and SAA.

See Schedule G to Appendix F.

4.6 ORM amendment process

Queensland Rail's 2015 DAU proposal

Queensland Rail proposed that the ORM be a schedule to the 2015 DAU.²⁰⁵ However, it proposed that the process for amending the ORM, including compensation provisions, be included in the SAA.²⁰⁶ Queensland Rail said that its ORM, included as a schedule to the DAU, was only intended to be a snapshot of the ORM at that point in time.²⁰⁷

Our Draft Decision proposed accepting that the ORM be a schedule to the 2015 DAU, but proposed a requirement that any amendments occur through a DAAU and not through the exercise of a provision in the SAA.

Stakeholders' comments

Aurizon agreed with our Draft Decision and said that it was more appropriate to have the protections supplied by the DAAU process in relation to any amendment to the ORM.²⁰⁸ Aurizon also said that Queensland Rail should be required to consult in relation to changes to the ORM due to safety matters.²⁰⁹ New Hope, Glencore and Yancoal agreed with our proposed amendments.²¹⁰ Glencore said that changes to the ORM could fundamentally alter the access holder's ability to use access rights in the manner intended at the time of contracting.²¹¹

Queensland Rail said our amendments impacted on its ability to operate the network efficiently and to deal with matters affecting safety in a timely manner, and that the QCA fundamentally misunderstood Queensland Rail's intention when it proposed having the ORM as a schedule to the DAU.²¹²

QCA analysis and Decision

We accept Queensland Rail's proposal to include the ORM as a schedule to the 2015 DAU, but require Queensland Rail to remove any mechanisms in the SAA for changing the ORM.

²⁰⁵ Queensland Rail, 2015 DAU, Schedule H.

²⁰⁶ Queensland Rail, 2015 DAU, SAA, cl. 8. The compensation provisions are in cl. 8.3.

²⁰⁷ Queensland Rail, sub. 26: 107.

²⁰⁸ Aurizon, sub. 29: 15.

²⁰⁹ Aurizon, sub. 20: 46.

²¹⁰ New Hope, sub. 23: 17; Glencore, sub. 25: 1; Yancoal, sub. 27: 1.

²¹¹ Glencore, sub. 30: 5.

²¹² Queensland Rail, sub. 26: 107.

ORM as part of the undertaking

We accept the proposal to include the ORM as a schedule to the access undertaking, as that removes the need to amend individual access agreements and also provides for consistency of operational requirements across multiple access agreements.²¹³

However, we do not accept Queensland Rail's proposal that a mechanism for amending the ORM be included in the SAA in the 2015 DAU. This is unworkable and undesirable, because amendments to the undertaking (which would include the ORM under Queensland Rail's proposal) must be effected through the process in the QCA Act for amending an approved access undertaking. Moreover, the prospect of different SAAs having different ORMs is likely to be unworkable (and potentially dangerous) as this would mean different access holders would operate according to different rules on the network. One universal ORM promotes certainty in relation to access holders' and access seekers' operating requirements.

Therefore, Queensland Rail will need to submit a DAAU to implement any changes to the ORM. We consider that the rights of access holders and other parties, including end users and Queensland Rail, will be protected, as the QCA considers the DAAU through the processes prescribed in the QCA Act. The approval process in the QCA Act provides for the QCA to seek submissions from stakeholders and apply the criteria in section 138(2) to decide whether or not to approve the DAAU.

Queensland Rail said in its December 2015 submission that the inability of Queensland Rail to amend the ORM because of a 'change of law' ²¹⁴ would impact on Queensland Rail's ability to operate the network efficiently. We do not accept this submission from Queensland Rail. Having to use the DAAU process for all amendments to the ORM may be less efficient for Queensland Rail; but, this is outweighed by the increased efficiency to the system as a whole, which is achieved by having certainty and consistency in relation to the ORM (s. 138(2)(a), (e), (h)).

Further, we note that the operator's and Queensland Rail's compliance with the ORM is predicated on that compliance not being inconsistent with all applicable laws and authorisations; and, to the extent that any applicable law is inconsistent with the ORM, the applicable law prevails (see cls. 7.3 and 8.4 of the SAA). Also, as noted by New Hope, in the event that there are genuine reasons for change, agreement from stakeholders in advance is likely to expedite the DAAU approval process.²¹⁵

Queensland Rail also said that, by removing clause 8.4(b) of the 2015 SAA (which deems changes to the ORM via a DAAU not to be changes to the ORM for the purposes of a SAA), the QCA would be improperly imposing a compensation process on Queensland Rail for the exercise of its statutory rights (namely lodgement of a DAAU).²¹⁶ The appropriateness or otherwise of any compensation is a matter that could be raised and considered in the course of the approval process for any DAAU.

We consider that our requirements provide an appropriate balance between Queensland Rail's legitimate business interests in having the ability to amend the ORM, the public interest, and access holders' and access seekers' interests in having a consistent set of operating requirements (s. 138(2)(a), (b), (d), (e), (h)). Consistency of operating requirements promotes

²¹³ We note that our October 2014 Draft Decision had accepted an alternative approach by Queensland Rail of publishing the ORM on its website, which also achieved this outcome.

²¹⁴ Queensland Rail, sub. 26: 107.

²¹⁵ New Hope, sub. 31: 40.

²¹⁶ Queensland Rail, sub. 26: 107.

efficiency and productivity, as access holders and seekers can appropriately plan and prepare their operations without having to adapt to idiosyncratic or individual variants to the requirements.

In line with the above review, we have removed the provisions in the 2015 DAU which allowed for changes to certain matters (typographical changes, people and places, and safety matters) to be made by Queensland Rail without submitting a DAAU. We have also amended the definition of the Operating Requirements Manual to remove the reference to it being amended through a process in the access agreements.

Safety issues and minor matters

Despite these changes reducing the ability to amend the ORM promptly, we still consider that they are appropriate, given the fundamental importance of every access holder being able to rely on the operating requirements. The QCA cannot give 'pre-approval' of amendments. However, we do accept that, in practice, Queensland Rail may be able to expedite amendments by discussing potential amendments with the QCA in advance of any formal submission.

In relation to safety matters, we note that the ORM still provides an ability for Queensland Rail to amend its own internal safety documents outside of the DAAU process. Further, as noted above, the terms of the ORM and each access holders' IRMP are (via provisions of the access agreements) also subject to the TRSA and all other applicable laws.

Compensation

In our Draft Decision we noted stakeholders' concerns about compensation for changes that impose costs on access holders, and the miners' desire for this to extend to end users.²¹⁷ While we accept that an effective compensation mechanism for access holders is needed, we consider that the way compensation is managed for end users who do not hold access directly is a matter to be considered at the time that any proposed amendments to the ORM are submitted for approval.

Accordingly, while we require that the 2015 DAU SAA be amended to remove the mechanism for amending the ORM without a DAAU, we also propose to remove the provisions in the SAA for compensating access holders for material adverse effects from ORM changes. We have had regard to Queensland Rail's concerns that a compensation provision in the SAA may inappropriately seek to impose obligations on Queensland Rail for exercising its rights under the QCA Act.

Queensland Rail said that our proposed amendments in the Draft Decision inappropriately disconnected the process of making amendments from the compensation process. This should no longer be a complaint of Queensland Rail, because now, as part of the process of approving a DAAU submitted by Queensland Rail (in relation to ORM amendments), the QCA will also consider the appropriateness of any compensation. That is, the compensation and amendment processes will now be connected.

We consider our amendments balance Queensland Rail's ability to amend the ORM with the appropriateness of an access holder being able to proceed on the basis that operational matters which exist at the time the access holder contracts remain consistent.

Also, if these matters change and have a material financial effect on the access holder's operations, the access holder may, if at the time of approving the DAAU it is appropriate, be

²¹⁷ Glencore, sub. 7: 10, 35–36; New Hope, sub. 10: 20, 43–44; Aurizon, sub. 6: 30; Asciano, sub. 5: 26–28.

compensated. It may not promote Queensland Rail's legitimate business interests to have to compensate operators, however, we consider that Queensland Rail's interest in being free to amend the ORM at will is outweighed by access holders' interest in Queensland Rail maintaining the operational requirements it has contracted to provide.

Similarly, Queensland Rail should, if it is appropriate, provide compensation when it deviates from these contracted requirements to the detriment of an access holder (s. 138(2)(e) and (h)). It also promotes the efficient use and operation of the network and is in the public interest if parties are held to the promises they make when contracting (s. 138(2)(a), (d)).

Summary 4.9

The SAA in the 2015 DAU must remove all processes for amending the ORM. If Queensland Rail is minded to amend the ORM in any way, it must lodge a DAAU. The QCA will consider the appropriateness of compensation to access holders at the time it considers the appropriateness of the DAAU.

See clause 4.3, the definition of Operating Requirements Manual in Appendix F, and clauses 8 and 9.13 of the ORM in Appendix G.

5 REPORTING

Ideally, a reporting regime should provide interested parties with information on how efficiently Queensland Rail has been operating and whether it is complying with certain aspects of its undertaking or access agreements, while ensuring the obligations on Queensland Rail are not excessive.

Part 5 of the 2015 DAU sets out Queensland Rail's reporting responsibilities, as well as its auditing and information obligations.

Our Decision accepts much of Queensland Rail's 2015 DAU proposal, but requires amendments to increase transparency.

This Decision differs from the Draft Decision in a few key areas. This Decision:

- requires more of the quarterly report content to be disaggregated at the system level
- for systems with a reference tariff, requires Queensland Rail to:
 - report actual costs against forecasts on a like-for-like basis; and
 - provide an explanation of any under- or overspends compared to forecasts.

Introduction

Reporting and compliance monitoring are important parts of the regulatory regime, as they place accountability on Queensland Rail and provide for greater levels of transparency.

However, it is important that there is a balance between the benefits to access seekers and users from reporting and compliance monitoring, and the regulatory burden that these processes impose on Queensland Rail.

The key issues are summarised in Table 8. Matters that require a more detailed explanation are discussed in Sections 5.1 to 5.6.

Table 8: Summary of key positions—reporting

Summary of the 2015 Draft Decision	Queensland Rail's position	Stakeholders' position	QCA Decision
1. Operational reporting	5		
Quarterly reporting on operational matters, certain complaints, and causes of significant changes in operating performance.	No comments.	Aurizon and New Hope supported the Draft Decision; however, they have proposed further amendments.	See Section 5.1.
2. Access reporting			
Annual reporting on timeframes associated with access negotiations.	Queensland Rail disagreed with our timeframe categories.	Asciano supported the Draft Decision. New Hope accepted the Draft Decision but also proposed further amendments.	See Section 5.2.
3. Cost and price report	ing		
Where a reference tariff applies, reporting of cost and price information.	Queensland Rail accepted the Draft Decision.	Aurizon and New Hope supported the Draft Decision but also proposed further amendments. Asciano supported the Draft Decision.	See Section 5.3.
Where a reference tariff does not apply, reporting of cost information.	Queensland Rail accepted the Draft Decision.	Glencore supported the Draft Decision but also proposed further amendments. Asciano supported Glencore's position.	See Section 5.4.
4. Audit requirements			
The QCA is allowed to require an audit of compliance with any aspect of the undertaking or the QCA Act.	Queensland disagreed with this position and said that we do not have the explicit right to undertake audits.	Aurizon said that the auditor should not be limited to someone who has experience in the area of costing railway activities. Asciano submitted that auditing should be undertaken at regular intervals.	See Section 5.5.
5. Financial statements	and costing manual		
Queensland Rail is required to publicly release audited financial statements.	Queensland Rail accepted the Draft Decision. ²¹⁸	No comments.	See Section 5.6.

²¹⁸ Queensland Rail, sub. 26, Annexure 7: 14.

5.1 Quarterly reporting on performance

Queensland Rail's 2015 DAU proposed quarterly reporting on operational matters, certain complaints, and causes of significant changes in operating performance.

Our Draft Decision proposed to accept Queensland Rail's proposed quarterly reporting provisions in the 2015 DAU.

Stakeholders' submissions

Aurizon and New Hope supported the Draft Decision but also proposed further amendments.²¹⁹

QCA analysis and Decision

The QCA agrees with some of Aurizon and New Hope's positions, namely that:

- Queensland Rail should be obliged to correct an error as soon as it is identified²²⁰
- 'if any', in clause 5.1.2(a)(vii) of the proposed DAU in the QCA's Draft Decision (referring to whether there is a measure of track quality), should be removed, as it is ambiguous^{221, 222}
- disaggregating the following metrics by systems is reasonable and unlikely to be onerous to Queensland Rail:
 - track kilometres under temporary speed restrictions
 - track quality
 - the number of written complaints that are verified as correct
 - the number of planned possessions that did not start and the number and percentage of planned possessions that did not finish within the time scheduled for the relevant planned possession in the MTP
 - the cause of any material changes affecting Queensland Rail's operating performance compared to the previous quarter.²²³

We consider that the above changes will improve the efficiency of the system as a whole by further decreasing information asymmetry and by helping to assist efficient pricing and negotiations (s. 138(2)(b), (c), (e), (a)). These amendments are also in the interests of access seekers and access holders and are not an onerous requirement on Queensland Rail as it will be collecting this information in any event (s. 138(2)(b), (e), (h)).

We have not implemented the following suggestions by Aurizon and New Hope:

- Queensland Rail should report on the number and percentage of train services that did not reach their destination within an allotted time slot by two groups:
 - when it is solely due to acts or omissions of Queensland Rail; and
 - when it is primarily due to delays attributed to an access holder or a nominated rolling stock operator.²²⁴

²¹⁹ Aurizon, sub. 20: 53; New Hope, sub. 23: 18.

²²⁰ Aurizon, sub. 20: 53.

²²¹ By removing 'if any', we require Queensland Rail to report on the most recent measure of track quality when the information is applicable, or explicitly state that the information is not applicable.

²²² Aurizon, sub. 20: 53.

²²³ Aurizon, sub. 20: 53; New Hope, sub. 23: 18.

We consider the use of 'solely' in relation to Queensland Rail, but 'primarily' in relation to an access holder or nominated rolling stock operator, to be unfair.

• The definition of operational constraint should be expanded to include any speed that is not the nominated speed board.²²⁵

We consider this to be unnecessary as the 2015 DAU definition already specifies speed restrictions.

• The reporting of average speed restrictions should be expanded to include the number of individual speed restrictions and the percentage below normal line speed.²²⁶

We do not consider that these particular changes strike an appropriate balance between Queensland Rail and access seekers and access holders (s. 138(2)(b), (e), (h)).

Summary 5.1

The 2015 DAU's quarterly reporting provisions must provide that:

- (a) Queensland Rail should be required to correct any errors identified in its annual or quarterly reports as soon as reasonably practicable and publish an amended report containing the correct information.
- (b) The words 'if any' should be deleted from clause 5.1.2(a)(vii).
- (c) All of the information metrics required in a quarterly report should be disaggregated by system.

See clauses 5.1.2(a)(vii), 5.1.2(b) and 5.4.1 in Appendix F.

5.2 Annual reporting on the access negotiation process

The 2015 DAU proposed annual reporting on various measures across the access negotiation process, including capacity information requests, access applications, IAPs, negotiation cessation notices and access agreements.

Our Draft Decision accepted the majority of Queensland Rail's proposals. However, we required the time taken to issue IAPs, and the time taken by access seekers to provide their intent to negotiate that was to be reported, to be disaggregated by number-of-day ranges.

Stakeholders' submissions

Queensland Rail disagreed with the timeframe categories proposed by the QCA for reporting on the intention to negotiate.²²⁷

New Hope and Asciano supported the Draft Decision.^{228, 229}

²²⁴ Aurizon, sub. 20: 53.

²²⁵ Aurizon, sub. 20: 53.

²²⁶ New Hope, sub. 23: 18.

²²⁷ Queensland Rail, sub. 26, Annexure 7: 12.

²²⁸ New Hope, sub. 23: 18.

²²⁹ Asciano, sub. 28: 12.

QCA analysis and Decision

We do not accept Queensland Rail's position that the annual reporting of intention to negotiate provided by an access seeker should be reduced to two timeframe categories (i.e. within 20 days, and more than 20 days). We disagree that our proposed timeframes (four categories) will impose a material administrative burden and lead to further downside regulatory risks to Queensland Rail, as it has claimed.

The relevant information is obviously available to Queensland Rail, so any additional administrative burden is unlikely to be significant and, in any event, this is outweighed by the increased transparency our proposal provides in terms of Queensland Rail's negotiation processes.

Further, we note that the original reason for specifying a range of time periods was to address a concern that a simple percentage of on-time and late responses did not properly reflect whether any delay was trivial or significant. Queensland Rail's proposed approach has the same flaws, while our required approach of reporting in time ranges indicates whether any delays are material.

We have adopted our Draft Decision in this regard (see section 5.1 of the Draft Decision).

Summary 5.2

The 2015 DAU must provide that Queensland Rail report annually on the time taken to issue IAPs to access seekers, and on the time taken by access seekers to provide their intent to negotiate, in the following categories:

- (1) less than 10 business days
- (2) 10 to 20 business days
- (3) 21 to 40 business days
- (4) more than 40 business days.

See clause 5.2.2(d) in Appendix F.

5.3 Reporting of cost and price information (when a reference tariff applies)

Queensland Rail proposed the annual reporting of information relevant to reference tariffs including: maintenance costs, scope of work undertaken, capital expenditure, operating expenditure and volumes.

Our Draft Decision proposed to accept the majority of Queensland Rail's proposal; however, we proposed a requirement that the actual information to be compared against the forecasts used to develop the tariffs.

This section only applies to the West Moreton and Metropolitan networks as they are the only systems for which the QCA has approved reference tariffs (see Chapter 8).

Stakeholders' submissions

Queensland Rail accepted our Draft Decision and said it intended to provide the information through its annual performance reporting (as suggested by the QCA) or in its publicly released audited below-rail financial statements.²³⁰

Asciano, New Hope and Aurizon supported our Draft Decision.²³¹ New Hope and Aurizon also proposed further amendments:

- New Hope submitted that the actual information should be reported against forecasts on a like-for-like basis; that is, in the same categories as those on which the approved forecast was based, and adopt the approved cost allocation methodology.²³²
- Aurizon said that Queensland Rail should also provide commentary explaining any under- or over-spends compared to forecasts.²³³

QCA analysis and Decision

We agree with the proposals put forward by New Hope and Aurizon, as these will increase the transparency of the tariff calculations, without imposing a material administrative burden on Queensland Rail. We consider that this is in the interests of access seekers and access holders, as it will assist both in understanding the calculation and movement of tariffs and in reducing information asymmetry (s. 138(2)(a), (b), (e), (h).

Summary 5.3

The 2015 DAU must provide that, for systems with reference tariffs, Queensland Rail report annually for the relevant financial year on:

- (a) maintenance costs and scope of maintenance, compared with the maintenance forecasts used to develop the tariff; the information is to be aggregated by the same categories as those on which the relevant forecast was based
- (b) operating expenditure, compared with the forecasts used to develop the tariff; the information is to be aggregated by the same categories as those on which the relevant forecast was based
- (c) an explanation of the main reasons for any discrepancy between actual and forecast maintenance costs and operating expenditure
- (d) capital investment and a roll-forward of its regulatory asset base
- (e) system volumes (disaggregated by system and commodity (where appropriate)).

See clause 5.2.2 in Appendix F.

²³⁰ Queensland Rail, sub. 26, Annexure 7: 13.

²³¹ New Hope, sub. 23: 19; Aurizon, sub. 20: 53; Asciano, sub. 28: 12.

²³² New Hope, sub. 23: 19.

²³³ Aurizon, sub. 20: 53.

5.4 Reporting of cost and price information (when a reference tariff does not apply)

Queensland Rail proposed to report annually on the previous financial year's maintenance and operating costs, capital expenditure and volumes when a reference tariff does not apply.

Our Draft Decision considered that reporting for systems where no reference tariff applies should be at least as comprehensive as that provided for reference train services.

Stakeholders' submissions

Queensland Rail accepted the Draft Decision.234

Glencore supported the Draft Decision; however, it submitted that Queensland Rail should be obliged to disclose more information. Glencore said that Queensland Rail should be required to report to the QCA, access holders and end users for each year of this undertaking in relation to the Mount Isa line by type of services (e.g. bulk minerals, intermodal, agricultural freight or passenger) and amongst other things, report on aggregate costs and revenues for the year, as well as any other information the QCA would want in order to calculate a Mount Isa line reference tariff.²³⁵ This included :

- detailed information on how access charges for the requested access rights had been calculated (e.g. inputs into any formula or methodology utilised, and the methodology for prices that were said to be 'market based')
- information on the aggregate current and future revenue streams for relevant parts of the network (e.g. Mount Isa line for Glencore services)
- the assumptions (and the basis of the assumptions) used to calculate future projections (e.g. escalations, forecasts or estimates of future costs or revenue).²³⁶

Asciano supported Glencore's submission.237

QCA analysis and Decision

After having regard to all submissions on this matter, we have adopted the Draft Decision in this regard (see section 5.2.2 of the Draft Decision), except as set out below.

In addition to the amendments proposed in the Draft Decision, we have shortened the timeframe for reporting expected capital investments from five years to four years. This is to accord with the term of the undertaking which will terminate less than four years after the approval date.

We note that it is open for an access seeker to bring a dispute to the QCA for determination; this should incentivise Queensland Rail to provide sufficient information relating to cost or price. Further, we have had regard to Queensland Rail's legitimate business interests in minimising its administrative burden (s. 138(a), (b), (e), (h).

Therefore, we have not implemented the additional reporting measures proposed by Glencore and Asciano as required amendments to the reporting regime in this Decision. However, we will be updating the costing manual that governs the regulatory accounts published by Queensland

²³⁴ Queensland Rail, sub. 26, Annexure 7: 13.

²³⁵ Glencore, sub. 30: 2.

²³⁶ Glencore, sub. 25: 5.

²³⁷ Asciano, sub. 28: 5.

Rail. During our review of the costing manual we will have regard to stakeholders' requests for more or better-presented information, including those already submitted by Glencore and Asciano. This matter is discussed further in Section 5.6 below.

Summary 5.4

The 2015 DAU must provide that, for non-reference-tariff train services, Queensland Rail is to include information on capital investment over the previous financial year and expected capital investments over one and four years, as well as:

- (1) maintenance costs of its system and scope of maintenance performed
- (2) operating costs of the regional network
- (3) system volumes (disaggregated by system and commodity (where appropriate)).

See clause 5.2.2(j) in Appendix F.

5.5 Audit requirements

Queensland Rail's 2015 DAU limited the QCA's audit powers to information contained in the quarterly and annual reports.

Our Draft Decision considered Queensland Rail's auditing proposal did not provide for adequate auditing of Queensland Rail's compliance with its access obligations. Our Draft Decision proposed that the QCA, acting reasonably, be allowed to require an audit of compliance with any aspect of the undertaking or the QCA Act.

Stakeholders' submissions

Queensland Rail said the QCA Act did not include an explicit right for the QCA to undertake audits; rather, the QCA had strong information gathering rights.²³⁸

Aurizon supported our Draft Decision; however, it said that the auditor should not be limited to someone who has experience in costing railway activities.²³⁹

Asciano submitted that auditing should be undertaken at regular intervals—at least every two years, or annually if there were major issues. It said that this would provide assurance that Queensland Rail was complying with the QCA Act, the access undertaking, and other associated regulatory instruments, while managing the costs. Asciano also submitted that the audit regime could be strengthened to oblige Queensland Rail to remedy breaches identified by the audit.²⁴⁰

QCA analysis and Decision

We have adopted our Draft Decision in this regard (see section 5.4 of the Draft Decision), except as set out below.

We consider that the QCA Act does allow for the QCA to require Queensland Rail to undertake audits to determine compliance with the access undertaking. Our position is not that the QCA undertake the audit itself, but rather the QCA may instruct Queensland Rail to obtain an audit of its compliance with the undertaking and provide the results to the QCA.

²³⁸ Queensland Rail, sub. 26, Annexure 7: 15.

²³⁹ Aurizon, sub. 20: 54.

²⁴⁰ Asciano, sub. 28: 15.

However, we agree with Queensland Rail that the auditing provisions should not extend to compliance with the QCA Act itself. We therefore have not adopted this aspect of the Draft Decision.

We do not accept Asciano's proposed amendments. We consider that our required auditing regime is appropriate without being unduly burdensome.

We disagree with Asciano that the audit regime needs to include an explicit obligation in relation to remedying breaches. Enforcement provisions, in relation to an approved undertaking, are already provided for in the QCA Act (see ss. 150A and 158A).

We also agree with Aurizon that the experience of the auditor should be widened to include auditors without the relevant rail costing experience.

Summary 5.5

The regulatory audit requirements in the 2015 DAU must provide that the QCA, acting reasonably, can require an audit of compliance with any aspect of the undertaking.

See clause 5.4.4 in Appendix F.

5.6 Regulatory accounts and cost allocation manual

Regulatory accounts

The QCA Act requires that an access provider keep separate accounts for its declared service in a manner approved by the QCA (s. 163). However, it does not require that Queensland Rail publish those accounts.

Our Draft Decision therefore proposed that Queensland Rail amend the 2015 DAU so that Queensland Rail was required to publicly release audited financial statements for its declared services, consistent with the requirements in the QCA Act, within six months of the end of the relevant financial year.

Neither Queensland Rail nor stakeholders were opposed to the QCA's Draft Decision on this matter. The QCA has therefore adopted section 5.3 of the Draft Decision.

Cost allocation manual

The QCA Act gives the QCA the power to require a cost allocation manual that sets out how the regulatory accounts will be prepared (s. 159).

We note the concerns raised by Asciano that historically the cost information provided by Queensland Rail has been inadequate. Asciano said this could be addressed by requiring Queensland Rail to provide consistent and transparent cost information to the QCA and users on an ongoing basis, where such costs are allocated according to the QCA-approved cost allocation manual. Asciano said that this would allow a degree of cost certainty and consistency. We also note Glencore's concerns discussed above in Section 5.4 of this Decision.

Given that the regulatory accounts are governed by the QCA Act and that the costing manual gives the QCA the ability to specify how those accounts should be prepared, the QCA considers there is no need to duplicate those requirements in the undertaking.

In this regard, the costing manual was amended in 2012 to identify the costs, revenue and assets for the Western Moreton network separately from those for the rest of Queensland Rail's

declared below-rail operations. The QCA is minded to require further amendments to the costing manual, consistent with that precedent, so that the regulatory accounts include a similar separation for each of the Mount Isa and north coast systems.

This should address stakeholders' concerns, without including provisions in the undertaking that govern how to prepare the regulatory accounts. The QCA will seek stakeholder comments on any proposed Queensland Rail costing manual once we have published this Decision.

Summary 5.6

The 2015 DAU must require Queensland Rail to publicly release audited financial statements for its declared services, consistent with the requirements in the QCA Act, within six months of the relevant financial year.

See clause 5.3.1 in Appendix F.

6 ADMINISTRATIVE PROVISIONS

Administrative provisions provide clarity on a range of miscellaneous administrative mechanisms that are designed to assist in dispute resolution, notices, QCA decision-making processes and transitional reporting arrangements.

In this Decision we have made changes to clarify that reporting of tariff related information must occur from the commencement date of the undertaking but will include information from 1 July 2013 onwards. We have also made miscellaneous amendments relating to access applications and dispute resolution.

Introduction

Part 6 of Queensland Rail's 2015 DAU provides for dispute resolution as well as transitional arrangements for reporting and negotiations started under the 2008 undertaking.

The QCA has adopted the views expressed in its Draft Decision on Part 6 of Queensland Rail's 2015 DAU relating to administrative provisions, unless otherwise indicated in amendments required by this Decision.

Key issues are summarised in Table 9 below, with matters requiring a more detailed explanation discussed further in Sections 6.1 and 6.2.

Summary of the 2015 Draft Decision	Queensland Rail's position	Stakeholders' position	QCA Decision
1. QCA decision-making	3		
Provides decision- making procedures and criteria. The provisions on QCA decision-making apply to both Queensland Rail and other relevant parties.	No comments.	No comments.	As per the Draft Decision (see Table 6.1 of the Draft Decision).
2. Notices			·
Provisions to clarify the form, means of giving, and effect of notices relating to this undertaking.	No comments.	No comments.	As per the Draft Decision (see Table 6.1 of the Draft Decision).
3. Transitional provisio	ns		
Tariff-related reporting information to be provided from the commencement date but is to include information from 1 July 2013.	Queensland Rail disagreed with our Draft Decision.	New Hope said that different information was required to achieve the QCA's intent.	See Section 6.1.
Negotiation for access— all matters and negotiations that commenced under the	Queried drafting of amendments.	No comments.	See Section 6.2.

Table 9: Summary of key positions and Decision—administrative provisions

Summary of the 2015 Draft Decision	Queensland Rail's position	Stakeholders' position	QCA Decision
2008 undertaking have to be finalised under the 2015 undertaking once it has been approved.			
4. Miscellaneous			
Miscellaneous other matters	Queensland Rail queried the need for transitional clause 6.4(b).	No comments.	See Section 6.2.

6.1 Commencement date of tariff-related reporting

For the period prior to the undertaking's approval date, Queensland Rail's 2015 DAU proposed to only provide reporting information as was required under the 2008 access undertaking.

Our Draft Decision proposed that Queensland Rail, in addition, should provide reports containing information about tariff-related matters as from 1 July 2013, rather than such reports only containing information about tariff-related matters as from the approval date of the 2015 undertaking.

Stakeholders' submissions

Queensland Rail disagreed with our position.²⁴¹ New Hope supported our position, but said different information was required to achieve the QCA's intent.²⁴²

QCA analysis and Decision

We disagree with Queensland Rail's position that we do not have the power to require tariffrelated reports to be provided containing information from the period between 1 July 2013 and 1 July 2016. While the undertaking and the reporting provisions will take effect from the undertaking's commencement date, it is appropriate to require the provision of information that preceded the commencement date, because such information is relevant to the access charges that access holders have paid and the negotiation of access charges in the future by reducing information asymmetry.

In doing so, we have made minor drafting changes to our DAU from our Draft Decision, to better effect our intent.

We note that New Hope's position was made in the context of our Draft Decision approach to calculating the adjustment amounts. This methodology has now been revised in this Decision.

²⁴¹ Queensland Rail, sub. 26, Annexure 7: 16; Queensland Rail, sub. 26: 108.

²⁴² New Hope, sub. 23: 20.

Summary 6.1

The 2015 DAU must provide that Queensland Rail provide tariff-related reports for the West Moreton network to access seekers which will include information from the period between 1 July 2013 and the commencement date of the undertaking.

See clause 6.4 in Appendix F.

6.2 Miscellaneous other matters

Transitional treatment of access and renewal applications

The 2008 access undertaking expired on 30 June 2015. Given this, we introduced clause 6.4(b) in our Draft Decision, to provide for things that happened during the period between the expiry of the 2008 access undertaking and the commencement of the new access undertaking. That is, any access applications or renewal applications made during this time period are deemed to have been done under the new undertaking to the extent that the matters are equivalent. However, in our drafting for this Decision, we have clarified that the terms 'access applications' and 'renewal applications' used in clause 6.4(b) do not refer to access applications and renewal applications commenced after the approval date of this undertaking.

Dispute resolution changes

Separately, we have adopted our Draft Decision position on dispute resolution for access seekers (see Table 6.1 of the Draft Decision). However, we have now extended the ability for access holders to avail themselves of the dispute resolution provisions in the undertaking as well in certain circumstances (see Section 1.4 of this Decision).

Moreover, we have made minor amendments to the dispute resolution provisions which provide for the scenario that the Rail Safety Regulator no longer exists or declines to determine a dispute. We have also clarified that, if a dispute is referred directly to the QCA pursuant to a provision of the undertaking, that dispute does not have to pass through each stage of the dispute resolution process before being determined by the QCA. We consider that this is appropriate to clarify and expedite disputes referred directly to the QCA and so is in the interests of Queensland Rail, access seekers, and access holders and promotes the economically efficient operation of Queensland Rail's infrastructure (s. 138(2)(a), (b), (e), (h)).

Summary 6.2

The 2015 DAU in respect of administrative provisions must provide that:

- (a) any access applications or renewal applications done between the expiry of the 2008 access undertaking and the commencement of the new undertaking are deemed to have been 'done' under the new undertaking
- (b) the dispute resolution provisions provide for the scenario that the Rail Safety Regulator no longer exists or declines to determine a dispute
- (c) if a dispute is referred directly to the QCA by a provision of the undertaking, that dispute does not need to pass through each step of the dispute resolution escalation process.

See clauses 6.1.1, 6.1.4 and 6.4 in Appendix F.

7 STANDARD ACCESS AGREEMENT

Queensland Rail's 2015 DAU includes a standard access agreement (SAA), which sets out the standard terms and conditions on which Queensland Rail will provide access to its network for all traffic types.

Access agreements are essential for the provision of access to Queensland Rail's network. A SAA facilitates the timely development of access agreements by providing a 'safe harbour' access agreement which parties can adopt without the need for further negotiation, or which parties can use as a guide when negotiating alternative terms of access.

Stakeholders have said that Queensland Rail's proposed SAA in the 2015 DAU²⁴³ is not balanced because it significantly weakens Queensland Rail's obligations to deliver its contracted access services and materially increases the contract risk held by access seekers who become access holders.

Our Decision is that Queensland Rail's proposed 2015 SAA is not appropriate as it is currently drafted. Instead, we require amendments to the proposed 2015 SAA to appropriately balance the rights, responsibilities and legitimate business interests of Queensland Rail with those of prospective access seekers.

This Decision differs from the Draft Decision in a few key areas. This Decision:

- clarifies the process for nominating multiple operators who will now enter separate (but substantially identical) tripartite agreements
- includes provisions for ad hoc train services
- removes all ORM amendment provisions and leaves amendments (and any compensation) to be dealt with via a DAAU
- clarifies that the agreement will be legally binding on operators who execute the agreement after Queensland Rail and the access holder
- includes some additional performance level reporting requirements
- amends the take-or-pay schedule to clarify its operation in accordance with the approved ceiling revenue limit
- requires Queensland Rail to negotiate variations to access agreements in good faith for efficiency improvements both at the initial negotiation stage and during the term of an undertaking.

Introduction

The Draft Decision said that Queensland Rail's proposed 2015 SAA did not appropriately balance the rights, responsibilities and legitimate business interests of Queensland Rail with those of prospective access seekers. The Draft Decision also set out our proposed amendments to the 2015 SAA.

Queensland Rail, in response to the Draft Decision maintained that the contractual form of the 2015 SAA was appropriate and largely rejected the QCA's proposed amendments.

²⁴³ For simplicity, we have referred to the proposed SAA in the 2015 DAU as the '2015 SAA'.

Queensland Rail did not provide detailed comments on the QCA's proposed amendments, other than responding to a limited number of clauses (discussed in Table 7 of Queensland Rail's submission).²⁴⁴ Queensland Rail largely took issue with the structure and legality of the QCA's proposed SAA and noted instances where Queensland Rail believed the QCA had deviated from earlier regulatory precedents, which the QCA had identified as being relevant to its considerations of the 2015 DAU.²⁴⁵

For the reasons set out in this chapter, we have adopted the view set out in the Draft Decision that the 2015 SAA is not appropriate.

This chapter broadly sets out the major themes and important issues relevant to assessing the 2015 SAA. Appendix C outlines stakeholders' and Queensland Rail's responses to specific clauses which were marked up in the QCA's proposed SAA, as well as our responses to their submissions. The amendments that we require to the 2015 SAA are reflected in Appendix G.

Key issues are summarised in Table 10 below. Matters that require a more detailed explanation are discussed in Sections 7.2 to 7.5.

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision
1. Access principles			
No separate schedule of access principles; the 2015 SAA applies to all traffic types.	No further comment on this issue.	New Hope, Glencore and Yancoal supported the Draft Decision subject to having a robust SAA. Aurizon did not accept the Draft Decision.	See Section 7.2.
2. SAA contracting fram	nework		
A tripartite structure, to allow either an end customer or an operator to hold the access rights.	Did not accept the Draft Decision; the contractual form of its proposed 2015 SAA was appropriate.	Other stakeholders, except Aurizon and Asciano, supported the Draft Decision.	See Section 7.3.
3. Balanced risk allocat	ion	·	
The 2015 SAA should provide a more balanced risk position for all parties.	Did not accept the Draft Decision.	Other stakeholders generally supported the Draft Decision, but suggested further amendments.	See Section 7.4.
4. Performance reporti	ng and KPIs		
The 2015 SAA should include a performance reporting regime and require Queensland Rail to negotiate suitable KPIs with access seekers.	Generally supportive of measures to increase transparency, but had issues with some proposed amendments.	Other stakeholders supported the Draft Decision, but said any financial outcomes should be in the SAA or the undertaking and the provisions should be subject to the dispute	See Section 7.5.

Table 10: Summary of key positions and decision—standard access agreement

²⁴⁴ Queensland Rail, sub. 26: 94.

²⁴⁵ See Queensland Rail, sub. 26: 91–94.

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision	
		resolution provisions.		
5. Standard funding agr	5. Standard funding agreement			
Access funders should be allowed to reasonably require Queensland Rail to develop a standard funding agreement.	See Chapter 9.	See Chapter 9.	See Chapter 9.	

7.1 QCA assessment approach

We consider that an appropriate SAA (and associated provisions in the undertaking) will:

- facilitate the timely development and execution of an access agreement for all access seekers; and
- balance Queensland Rail's legitimate business interests with the rights and interests of access seekers and access holders; and
- promote efficient and non-discriminatory use of the network; and
- promote effective competition in upstream and downstream markets.

Given the above, and in accordance with section 138(2)(h) of the QCA Act, we also consider it appropriate that a SAA should take account of the regulatory precedents established by:

- the standard access principles within Schedule E of Queensland Rail's previous 2008 access undertaking (access principles) (especially if the risk allocation matrix of the access principles is to be embedded within the 2015 SAA (see below))
- the SAA—coal (Schedule H) of Queensland Rail's 2008 undertaking
- the split-form SAA of the approved Aurizon Network 2010 undertaking.

That is not to say that the 2015 SAA should be drafted on the same terms as these regulatory precedents, nor that every provision which covers similar subject matter should be exactly the same as the earlier approved examples. Queensland Rail's proposed 2015 SAA has been reviewed afresh and some provisions of the previous regulatory precedents may no longer be appropriate. However, their use as guides, considering their practical application over the years, is instructive and appropriate.

Table 11: provides more detail on how we consider that the above themes should be reflected in the 2015 SAA.

Assessment considerations	QCA's assessment approach
Does the SAA facilitate the timely	We consider the SAA will facilitate the timely development and execution of access agreements if it:
development and	 allows the SAA to be used as a workable and balanced guide to negotiations as
execution of an access	well as a 'safe harbour' so that, unless otherwise agreed by Queensland Rail and
agreement for all	the access seeker, the terms of the SAA will apply as the default access
access seekers?	agreement;
(s. 138(2)(a), (d), (e),	 assists in enabling all access seekers and end customers to obtain timely access
(h))	to the declared network, including in circumstances where access seekers

Table 11: QCA's assessment of the 2015 SAA

Assessment considerations	QCA's assessment approach
	and/or end customers require a network extension to accommodate their access application;
	 establishes appropriate dispute mechanisms so that the SAA remains a relevant and effective agreement over the term of the access undertaking; and
	 clearly defines the terms and conditions of the SAA so that they are readily understood by parties and are relatively simple to negotiate and administer.
Does the SAA address Queensland Rail's	We consider the SAA is likely to advance the legitimate business interests of Queensland Rail if, among other things, it:
legitimate business interests?	 allows Queensland Rail to deliver all access services in accordance with its executed access contracts;
(s. 138(2)(b), (g))	• recognises Queensland Rail's responsibility to deliver access services consistent with its passenger priority obligations under the <i>Transport Infrastructure Act</i> ;
	 applies a commercially balanced approach to allocating risks to the contracting party best placed to manage or mitigate the risks; and
	 allows Queensland Rail to recover all efficient costs from the construction and ownership of a network extension consistent with the 2015 DAU, including Schedule E.
Does the SAA appropriately balance	We consider the SAA will appropriately balance Queensland Rail's, access seekers' and access holders' rights and interests if it:
the rights, obligations, risks, liabilities and indemnities between	 makes it possible for an end customer access seeker to become an access holder that has the flexibility to assign operational utilisation rights to different rail operators;
all the contracting parties? (s. 138(2)(a), (b), (d),	 establishes a reasonable and commercially balanced allocation of rights, obligations and risks between the parties in the provision of access services
(e), (h))	 provides certainty and security regarding the nature and quality of the access rights being sold/purchased and the ability for parties to manage their contractual risks;
	 establishes transparent and clearly defined processes through which access rights can be varied (renewed, resumed, relinquished, transferred, suspended and/or terminated), including in response to productivity and efficiency improvements; and
	 takes appropriate account of the earlier approved regulatory precedents and, where appropriate, reflects the approved workable balance contained in those documents.
Do the arrangements promote efficient and	We consider the SAA will promote the efficient and non-discriminatory use of the network if it:
non-discriminatory use of the network?	 provides an access seeker (and access holder) with flexibility in relation to the use and management of its access rights;
(s. 138(2)(a), (b), (d), (e), (h))	 provides a rail operator with the ability to operate train services in accordance with the access rights contained in the relevant access agreements;
	 clearly delineates the rights and responsibilities of all parties to the SAA, namely Queensland Rail, rail operators, and end customers; and
	 consistently applies the same arrangements across all access holders with respect to Queensland Rail's operational, safety, and environmental requirements.
Do the arrangements promote effective	We consider the SAA will promote effective competition in upstream and downstream markets if it:
competition in upstream and downstream markets?	 provides an efficient network service at an efficient cost that is commensurate with the regulatory and commercial risks held by all the parties;
(s. 138(2)(a), (b), (d),	• provides a clear separation of roles relating to the ownership and management
Assessment considerations	QCA's assessment approach
------------------------------	---
(e), (h))	of access rights and the operation of train services on the network;
	 provides opportunities for end customers (i.e. bulk commodity companies) to hold the access rights and assign operational rights to different train operators within the term of the agreement;
	 provides the ability for access seekers and end customer access seekers to obtain access to the network when a network extension is required to accommodate its access application; and
	• provides scope for variations required to adopt productivity and efficiency improvements.

7.2 Access principles

The third party access regime in the QCA Act is underpinned by a 'negotiate–arbitrate' approach to regulation, with the access regime incorporating the primacy of contractual negotiations. Access principles²⁴⁶ have previously been used as a contractual guide to establishing the core terms and conditions according to which Queensland Rail is obliged to provide access when the SAA is not suitable or relevant. Because access principles seek to allocate the main commercial and operational risks between the contracting parties in a balanced manner, they facilitate the timely negotiation of access agreements. Queensland Rail's previous 2008 undertaking outlined its access principles in Schedule E.²⁴⁷

Queensland Rail, in its proposed 2015 DAU, did not include a separate schedule of access principles. Rather, Queensland Rail intended for the contractual risk allocation matrix underpinning the access principles to be embedded in Queensland Rail's proposed 2015 SAA, which would then apply to all traffic types, thereby removing the need for separate access principles.²⁴⁸

Our Draft Decision agreed with Queensland Rail's proposed approach of not including separate access principles.

Stakeholders' submissions

Glencore and Yancoal supported the QCA's Draft Decision.²⁴⁹ New Hope also supported the QCA's Draft Decision, subject to having a robust SAA.²⁵⁰

Aurizon said a decision to remove access principles provided Queensland Rail with the ability to reject any reasonable requests for variations to the SAA that would result in productivity gains.²⁵¹

²⁴⁶ Schedule E in both Aurizon Network and Queensland Rail's approved undertakings (2001, 2006, 2008, 2010) has remained relatively consistent since the approval of the 2001 undertaking. Aurizon Network's 2010 undertaking was the last instance where we approved access principles that set out what a network provider was required to do to reflect its obligations under the QCA Act.

²⁴⁷ Schedule E was used as a guide by all contracting parties when negotiating a customised access agreement, while the SAA provided a 'safe harbour' agreement for traffics (including non-coal traffics) to fall back on if negotiations were not successful.

²⁴⁸ Queensland Rail, sub. 1: 35–39.

²⁴⁹ Glencore, sub. 25: 1; Yancoal, sub. 27: 1, 4.

²⁵⁰ New Hope, sub. 24: 3.

²⁵¹ Aurizon, sub. 20: 31–35; New Hope subsequently supported this submission in New Hope, sub. 31: 20.

Stakeholders also said that access agreements should have mechanisms in place to allow for above-rail efficiency improvements during the term of the access agreement and that Queensland Rail should be required to substantiate why it did not consider a proposed amendment suitable.²⁵²

Queensland Rail disagreed and said that it should not have an additional obligation to negotiate productivity variations during the term of an access agreement.²⁵³

Asciano said that it remained concerned about unbalanced risk management in relation to the 2015 SAA but supported an increased emphasis on flexibility and customer focus.²⁵⁴

QCA analysis and Decision

The QCA agrees with Queensland Rail's proposal that separate access principles are not required and that the 2015 SAA should instead apply to all traffic types. However, we have had regard to Aurizon's comments that removing the access principles could stifle commercial innovation and reduce flexibility by imposing the SAA terms if agreement cannot be reached on variations.

We have adopted our Draft Decision not to include separate access principles, subject to the further amendments discussed below (Draft Decision, section 7.2).

We recognise that Queensland Rail's proposal does not adequately address the legitimate concern raised by Aurizon. Because of this, we believe it is appropriate to clarify that at the negotiation stage Queensland Rail is obliged to provide reasons if it does not accept variations to the SAA proposed by an access seeker which would promote productivity or efficiency improvements.

We therefore require clause 2.9.4 of the 2015 DAU to be amended such that Queensland Rail is obliged to provide reasons if it rejects (at the negotiation stage of an access agreement) proposed variations to the SAA that an access seeker can demonstrate will promote productivity or efficiency improvements.²⁵⁵ While this will impose an additional burden on Queensland Rail, we consider the amendment is appropriate as it will promote efficiency and productivity gains, which will in turn promote competition and increase contracting capacity for Queensland Rail.

This position is largely consistent with our Draft Decision, but further clarifies the role and purpose of the SAA in relation to all traffic types, given that there will no longer be access principles to guide negotiations for non-reference-tariff traffic.

We do not accept Queensland Rail's position that it is sufficient that it already has statutory obligations, pursuant to section 101(1) of the QCA Act, to negotiate in good faith.²⁵⁶ This provision, in itself, does not provide access seekers with sufficient clarity and certainty regarding negotiations surrounding proposed variations to the SAA.

Likewise, Aurizon, New Hope and Glencore said that there should be a mechanism to allow access holders to vary the terms of their access agreements, during the term of the access

 ²⁵² Aurizon, sub. 20: 32; New Hope and Glencore subsequently supported this submission in New Hope, sub.
 31: 20 and Glencore, sub. 30: 3.

²⁵³ Queensland Rail, sub. 33: 44.

²⁵⁴ Asciano, sub. 28: 5, 19.

²⁵⁵ Clause 2.9.4 of the 2015 DAU provides for an access agreement to be consistent with the terms of the undertaking and the SAA unless otherwise agreed between the contracting parties.

²⁵⁶ Queensland Rail, sub. 26: 92.

agreement, to allow for efficiency and productivity improvements. This might also include new reference train services and relief in relation to relinquishment fees if train paths are relinquished because of these improvements.²⁵⁷ Queensland Rail said that there was no need for this mechanism because it was already obliged to negotiate in good faith under the QCA Act. Further, Queensland Rail said that the QCA had no power to require amendments to introduce a new reference train service.²⁵⁸

We agree that it is important to promote efficiency and productivity improvements and that these improvements may be conceived during the life of a long-term access agreement. Efficiency improvements benefit access holders, access seekers and Queensland Rail. They also benefit upstream and downstream markets, as well as the system as a whole.

We therefore require Queensland Rail to reasonably consider any proposed productivity or efficiency variations which arise during the term of an access agreement and negotiate in good faith as well as provide reasons for any refusal to vary the access agreement. We do not agree with Queensland Rail that the amendment is unnecessary because of section 101 of the QCA Act, as the required provision will also operate in circumstances where an access agreement has already been agreed.

The development of new reference train services is discussed in Chapter 8.

Our required amendments are more specific than the statutory obligation and seek to address any imbalance in negotiating positions—for example, where Queensland Rail is seeking to use its monopoly position as the access provider. We consider our proposal can facilitate balanced discussions between the parties on access conditions and efficiency gains. This has the potential to encourage productivity improvements and infrastructure investments that rely on Queensland Rail's infrastructure, promoting the effective and efficient utilisation of Queensland Rail's below-rail service, as well as facilitating upstream and downstream competition.

Our required amendments are also consistent with, and provides additional guidance to, Queensland Rail's and access seekers' obligations under section 101(1) of the QCA Act to negotiate in good faith.

²⁵⁷ See Aurizon, sub. 20: 32–33; New Hope, sub. 31: 20; Glencore, sub. 30: 3.

²⁵⁸ Queensland Rail, sub. 33: 44.

Summary 7.1

The QCA accepts that Queensland Rail does not need to include a separate schedule of access principles in the 2015 DAU, as the 2015 SAA applies to all traffic types.

However, the 2015 DAU and SAA must clarify that Queensland Rail has an obligation to:

- (a) provide reasons as to why Queensland Rail rejects a proposed variation to the SAA (both at the negotiation stage and during the term of the access agreement) where the relevant access seeker or access holder can demonstrate that the proposed variations will result in productivity or efficiency improvements; and
- (b) reasonably consider, and negotiate in good faith in relation to, any proposed variations to the terms of an access agreement which are proposed by an access holder during the term of an access agreement where the relevant access holder can demonstrate that the proposed variations will result in productivity or efficiency improvements.

See clause 2.9.4 of Appendix F and clause 1.3 of Appendix G.

7.3 Tripartite structure

Queensland Rail proposed a tripartite structure in its 2015 SAA which provided that both an end customer and an operator could become signatories to a single agreement. However, despite allowing three parties to be signatories to Queensland Rail's proposed 2015 SAA, the agreement only provided for an operator to be the access holder.

Our Draft Decision considered it was appropriate to amend Queensland Rail's proposed 2015 SAA to provide that either an end customer or an operator could be the access holder. We also considered that an access holder should have the flexibility to nominate an operator (or multiple operators) to use its access rights (in the case of an end customer access holder) or, if the access holder was also an accredited railway operator, nominate itself to use the access rights.

Stakeholders' submissions

Queensland Rail said that the QCA's proposed SAA was not effective in creating legal relations between Queensland Rail, an access holder, and an operator who executed the SAA later than the original parties.²⁵⁹

Aurizon and Asciano stated that the 2015 SAA should not allow for multiple operators to enter into the same agreement.²⁶⁰

New Hope, Glencore and Yancoal all supported the QCA's Draft Decision, subject to some further amendments to, amongst other things, more clearly take account of the fact that multiple operators might execute the agreement.²⁶¹

²⁵⁹ Queensland Rail, sub. 26: 91–92.

²⁶⁰ Aurizon, sub. 20: 37; Asciano, sub. 28: 8.

²⁶¹ New Hope, sub. 24: 3, sub. 32: 27; Glencore, sub. 25: 1; Yancoal, sub. 27: 4.

QCA analysis and Decision

The QCA requires Queensland Rail to amend its 2015 SAA as proposed by the QCA in its Draft Decision, to allow for either an end customer or an operator to contract as the access holder and allow for that access holder to nominate an operator (or multiple operators via multiple tripartite agreements) to use the access rights on its behalf. This is necessary so that an access holder can control its access rights and nominate one or more operators to use its access rights.

In our view, the 2015 SAA restricts contracting flexibility and thereby reduces incentives to increase and improve efficiency across the system.

We have adopted our Draft Decision in this regard except as noted in relation to the amendments discussed below (see section 7.3 of the Draft Decision).

We reject Queensland Rail's argument that even though the proposed 2015 SAA does not reflect a particular form of contractual arrangement it remains appropriate. Our view is that the form of the 2015 SAA is not appropriate. The form of the contractual arrangements provided by a SAA are integral to the operation of the network. Amending the SAA to allow for different parties to hold the access rights greatly increases competition amongst the users of the network and therefore we consider that the form of SAA required by this Decision is appropriate.

Multiple operators

Stakeholders, except Aurizon, Asciano and Queensland Rail, supported the QCA's proposed structure. Stakeholders did however propose a number of additional amendments, the majority of which related to:

- clarifying (within particular clauses) the possibility of more than one operator being a signatory to the agreement
- restricting information sharing amongst operators
- clarifying which party is to provide initial information to Queensland Rail during access negotiations (this aspect is dealt with in the body of the DAU).²⁶²

Aurizon and Asciano said that a number of issues would result from having multiple operators being parties to one access agreement.²⁶³ For example, inconsistencies between the termination provisions and the security provisions; quarantining of commercially sensitive information between operators; and, imposing unfavourable variations on an operator who may not agree to the variation.²⁶⁴

We agree with Aurizon and Asciano that there are a number of matters within a SAA where multiple operators being parties to the same contract could compromise sensitive information or have other unintended consequences (e.g. in relation to disputes and variations). Because of this, we have moved away from our proposal in the Draft Decision in this regard and consider that the 2015 SAA should remain a tripartite agreement (i.e. only one operator, one access holder and Queensland Rail per access agreement) and not allow for multiple operators to be parties to the same access agreement.

The amended SAA will still allow an access holder the flexibility to nominate multiple operators to use its access rights; however, each operator will be required to enter into a separate access

²⁶² In relation to concerns regarding who is to provide Queensland Rail with information, see Yancoal, sub. 27: 4 (other issues — 'End User Contracting').

²⁶³ Aurizon, sub. 20: 37.

²⁶⁴ Aurizon, sub. 20: 37.

agreement which will be substantially identical to that negotiated between the access holder and Queensland Rail except for variations necessary to account for a different operator. These changes should also help to improve transparency as well as clarify and improve the structure and mechanics of the QCA's required SAA. These changes are reflected in our amendments and also noted in Appendix C.

Legal relations

We accept Queensland Rail's position that the QCA's draft SAA was not clear in creating legal relations between Queensland Rail, an access holder, and an operator who executes the access agreement after Queensland Rail and the access holder.²⁶⁵ We have sought to address this by clarifying that each signatory agrees to be contractually bound to a subsequent incoming operator and vice versa. The amendments will also clarify that an access holder can execute an access agreement with Queensland Rail before having decided on its operator(s) and any subsequent operators will also be bound.

We consider this amendment is consistent with Queensland Rail's legitimate business interest as it provides contractual certainty to Queensland Rail as access rights provider, and also solidifies the liabilities and obligations between the parties.

We consider that providing for an end customer to have the ability to control the access rights is appropriate for the 2015 SAA, as it will allow an end customer to:

- hold, transfer, assign, relinquish and terminate the access rights independently of an operator
- negotiate and execute an access agreement without concurrently nominating a rail operator
- switch between rail operators within the term of the access agreement.

Our proposed amendments also allow for operating and contractual risks to be more clearly delineated between the parties, depending on which party is best placed to manage the risk that is assigned to it.

The ability to control the access rights (which are essential to an end customer's business) is important to end customers who may seek access to the service.

Further, our amendments to Queensland Rail's proposed 2015 SAA will increase operator competition (and thereby encourage efficiencies and innovation within the network), as well as upstream and downstream competition, by allowing an end customer to nominate its chosen operator (or operators) and vary its nominations during the term of the agreement (s. 138(2)(a), (d)). The amendments will also reduce negotiation and contracting duplication and other inefficiencies thereby reducing costs. These outcomes promote the object of the relevant part of the QCA Act, the legitimate business interests of Queensland Rail, and the interests of access seekers and end user access seekers who become access holders (s. 138(2)(b),(e), (h)).

²⁶⁵ Queensland Rail, sub. 26: 91, 95, 102.

Summary 7.2

The 2015 SAA must provide for the following:

- (a) It should be possible for an end customer or an operator to be the access holder and have the necessary flexibility to manage and control the use of access rights under an access agreement.
- (b) A tripartite structure should be adopted which more appropriately divides the contract responsibilities and risks of Queensland Rail, an operator and an end customer.

See clauses 2.9.4 and 2.9.7 (deleted) in Appendix F, clauses 2, 3, 4, 14, 15, 16, 27.11 and Schedule 1 to Appendix G (SAA), in addition to various consequential amendments made throughout the SAA to provide for the tripartite structure.

7.4 Balanced risk allocation

The balance between contracting parties in allocating risks, rights and obligations under a SAA is arguably the most important element of the 2015 SAA. This is because the SAA is used as a 'fallback' position if variations to the SAA cannot be agreed in the negotiation stage. If the risk allocation is not balanced, and the access provider and an access seeker cannot agree in their negotiations on rebalancing it, a party may be liable for large risks it cannot control. Equally, a party may be liable for obligations which it cannot satisfy.

A number of regulatory precedents to the SAA that have been approved by the QCA over the years have largely proved, in practice, to be workable and balanced.²⁶⁶

Queensland Rail had previously said that it has proposed a risk and liability regime in its 2015 SAA that was a revision of the regimes in the previous regulatory precedents (2008 and 2010 SAAs).²⁶⁷

In our Draft Decision we did not consider that Queensland Rail's proposed allocation of rights, obligations and risks between the parties in Queensland Rail's proposed 2015 SAA was appropriate.

Stakeholders' submissions

Stakeholders said that Queensland Rail's proposed 2015 SAA provided an unreasonable and unbalanced risk allocation between the access provider and the access seeker.²⁶⁸ In contrast, stakeholders were largely supportive of the QCA's proposed 2015 SAA.²⁶⁹ However, Asciano said it still had strong concerns about Queensland Rail's approach to risk allocation and risk management, which were not addressed in our Draft Decision.²⁷⁰

²⁶⁶ These include, relevantly, the access principles, the SAA (coal) from the Queensland Rail's 2008 access undertaking; and, the split-form SAA from the Aurizon Network 2010 access undertaking (see discussion above in relation to these regulatory precedents).

²⁶⁷ Queensland Rail, sub. 1: 35–40.

²⁶⁸ Asciano, sub. 5: 13–15; Aurizon, sub. 6: 23; Glencore, sub. 7: 9; New Hope, sub. 11: 2, 4.

²⁶⁹ New Hope, sub. 24: 3-5; Aurizon, sub. 20: 6; Yancoal, sub. 27: 4; Glencore, sub. 25: 1, 6.

²⁷⁰ Asciano, sub. 28: 13.

Queensland Rail did not make further comments on the risk profile of the QCA's proposed SAA generally. However, Queensland Rail did provide some comments on particular clauses.

QCA analysis and Decision

We do not consider that the risk allocation balance in the 2015 SAA is appropriate.

We have provided detailed amendments to Queensland Rail's proposed 2015 SAA which we consider are required to make the risk allocation matrix underlying the 2015 SAA appropriate, having regard to section 138(2) of the QCA Act.

These amendments are contained in Appendix G. The amendments to key areas are also discussed below.

The amendments are largely consistent with the Draft Decision. However, we have made a number of changes to the approach contemplated in the Draft Decision. Because of this, we refer to and repeat our Draft Decision in relation to the balanced risk allocation except where the approach in our Draft Decision is amended or varied by this Decision as set out below (see section 7.3 of the Draft Decision).

Regulatory precedents (2008 and 2010)

We are of the opinion that the number of executed access agreements that are currently held by access holders demonstrates the relevance of the 2008 Queensland Rail and the 2010 Aurizon SAAs. Stakeholders had also recommended significant amendments to Queensland Rail's proposed 2015 SAA to reflect the risk allocation matrix contained in these approved regulatory precedents.²⁷¹

In order to identify in Queensland Rail's proposed 2015 SAA any material deviations from the approved regulatory precedents which inappropriately altered the risk balance, we conducted a clause-by-clause review of Queensland Rail's proposed 2015 SAA. We then compared Queensland Rail's proposed SAA provisions with the provisions relating to the same subject matter under the earlier regulatory precedents. The results of this review were annexed to the Draft Decision.

Additionally, as we have accepted that the access principles are not required to be included in Queensland Rail's proposed DAU, we also conducted a detailed review of the previous access principles and compared these (again clause by clause) with the risk position in Queensland Rail's proposed 2015 SAA to check for consistency.²⁷² This is because part of the role of the access principles was to guide negotiations for non-reference-tariff traffic. If the access principles are no longer available to guide these negotiations, we consider it appropriate that the balance contained in those access principles should be (as appropriate) incorporated in the SAA. The results of this review were also annexed to the Draft Decision.

However, in reviewing Queensland Rail's proposed 2015 SAA, we have considered the document with 'fresh eyes'. In effect, although the regulatory precedents are instructive and relevant, we did not merely apply the old regulatory precedents—we also considered submissions made by all stakeholders (as to their concerns regarding the risk allocation in the SAA as proposed by Queensland Rail) and considered the appropriateness of the access principles in the current context.

²⁷¹ Glencore, sub. 7: 12-14; New Hope, sub. 11: 6-24; Aurizon, sub. 6: 22-31; Asciano, sub. 5: 18–22.

²⁷² This comparison of the access principles and the 2008 Queensland Rail SAA with Queensland Rail's proposed2015 SAA was documented in Appendix B to our October 2015 Draft Decision.

Amendments to key clauses

The key clauses which we have amended to re-balance the risk allocation are outlined in Table 12. The detail of these amendments and all the other amendments we require to the 2015 SAA are contained in Appendix G (the QCA's 2015 SAA). Variations from the Draft Decision in relation to particular clauses are outlined in the table in Appendix C of this Decision.

able 12: Key provisions which require amendments to re-balance the risk allocation matrix

Queensland Rail's proposed 2015 SAA provisions	QCA's amendments to Queensland Rail's proposed 2015 SAA
Network management (cls. 7, 8 and 23 of Queensland Rail's 2015 SAA)	Queensland Rail's network management obligations in the 2015 SAA should be amended to include an obligation on Queensland Rail to maintain the network consistent with objective rolling stock infrastructure standards, and consult in relation to any changes to these standards (see Chapter 4)
	This is a reasonable obligation on Queensland Rail as an access provider to facilitate access being provided to third parties.
	Provisions requiring rail operators to warrant as to the standard of the network should be removed, as this is reasonably a matter for Queensland Rail; and insofar as an incident should have been reasonably foreseeable to an operator as a result of its inspection, the general liability provisions will apply (see cl. 12.5(a), (b) and (c) in particular).
Indemnities, liabilities and limitations (cls. 12 and 13	The indemnities, liabilities and limitations applying to Queensland Rail should be amended to allow claims to be made in relation to:
of Queensland Rail's 2015 SAA)	the standard of the network
	 the non-provision of access caused, or contributed to, by Queensland Rail's negligence or breach
	third party works undertaken on behalf of Queensland Rail
	It is reasonable for Queensland Rail to be held accountable for the state of the below rail network as the access provider.
Dangerous goods (cls. 10.5 and 13 of Queensland Rail's 2015 SAA)	The dangerous goods provisions should be amended to better reflect the ability of each party to manage the risks associated with the carriage of dangerous goods on the network, including as follows:
	 Queensland Rail should be liable for claims arising out of incidents involving dangerous goods where caused or contributed to by Queensland Rail;
	 No additional indemnity should be provided by operators to Queensland Rail in relation to dangerous goods; and
	 The rail operator's insurance provisions should not require the rail operator to be insured for Queensland Rail's negligence.
	It is reasonable that risk is allocated to the party best able to manage it.
Disputes (cl. 19 of Queensland Rail's 2015	All disputes should be escalated to an independent arbiter and Queensland Rail cannot arbitrate on disputes to which it is a party.
SAA)	This is consistent with any arbitration being impartial and unbiased.
Force majeure (cl. 20 of Queensland Rail's 2015 SAA and associated definitions)	The amended force majeure provisions provide that an access holder is relieved from its obligations to pay access charges in the event of a force majeure event which affects the regulated network—this amendment reflects the position under the previous access principles in relation to the non- reference tariff network. Similarly, reference tariff traffic is also relieved from its take-or-pay obligations if a force majeure event affects the regulated network. However, if the relevant access agreement relates to the West Moreton Network and track destroyed by a force majeure event is reinstated, Queensland Rail can submit a variation to the reference tariff to recover 50%

Queensland Rail's proposed 2015 SAA provisions	QCA's amendments to Queensland Rail's proposed 2015 SAA
	of the take-or-pay foregone while force majeure applied. Force majeure is not the fault of either Queensland Rail or access holders and partial take-or-pay relief creates incentives for Queensland Rail to expedite necessary repair works, while reducing its financial exposure to forgone take-or-pay.
	We have also clarified the operation of the force majeure provisions and the definition of 'Queensland Rail Cause' to clarify that an access holder is only relieved from its obligations to pay access charges to the extent that the force majeure event affects the regulated network (and not private infrastructure). Further information on force majeure is found in Chapter 8.

We consider our proposed amendments are appropriate for the following reasons:

- The amendments promote the efficient use of Queensland Rail's network, promote the object of the relevant part of the QCA Act and, where appropriate, have regard to the risk position underlying the 2008 and 2010 regulatory precedents (including access principles) (ss. 69E and 138(2)(a), (b), (e) and (h) of the QCA Act). Specifically, Queensland Rail's proposed 2015 SAA would need to be amended to:
 - apply, in general, the risk allocation principles commonly applied in contractual negotiations, including that each party is to:
 - o carry the contract risk that they are best placed to manage
 - be held accountable for their actions, negligence or breach under the agreement
 - indemnify the other parties for loss (personal injury, death or property damage) caused by, or to the extent contributed to by, the wilful default or negligence of the indemnifying party
 - exclude the other parties from liability for consequential loss except in limited circumstances.
 - subject to appropriate exceptions, apply a risk position that is relatively consistent with the risk position underlying the 2008 and 2010 regulatory precedents (where appropriate)
 - promote commercial confidence that access seekers, access holders and rail operators can enter into long-term access contracts with Queensland Rail and hold Queensland Rail accountable for delivering the contracted services over the life of the agreement
 - facilitate access to the network to maximise the operation and use of access rights which, in turn, will improve network productivity and lower the unit cost of access.
- The amendments may advance the legitimate business interests of Queensland Rail. They also address the interests of access seekers, access holders (both rail operators and end customers) and comply with the pricing principles (s. 138(2)(b), (e), (g) and (h) of the QCA Act). This is achieved by amending Queensland Rail's proposed 2015 SAA to:
 - provide a consistent set of terms and conditions for all traffics to access the network on a non-discriminatory basis
 - provide a level playing field to underpin access negotiations between access seekers, end customers and Queensland Rail
 - align each contracting party to its relevant contractual obligations and entitlements

- provide clear and transparent assignment of the risks and accountabilities held by each party to the agreement
- leave Queensland Rail and access seekers free to negotiate the setting of access charges for non-reference-tariff traffic
- provide open and transparent communication channels in the use and delivery of contracted access services
- allow Queensland Rail and access seekers to negotiate non-standard terms and conditions
- minimise the potential for access disputes to be triggered under the 2015 DAU and, where they are triggered, provide for an independent party to arbitrate disputes
- provide a safe harbour agreement to facilitate the timely execution of access agreements.
- The amendments promote the public interest in having competition in markets (s. 138(2)(d) of the QCA Act) where participation in that market is reliant on Queensland Rail's monopoly provision of an access service. This is achieved by, amongst other things, providing a more balanced and equitable contracting regime, and allocating risks and costs efficiently.

We have made substantial amendments to Queensland Rail's proposed 2015 SAA. This is because we have sought to work within the drafting of Queensland Rail's proposed 2015 SAA in order to achieve an appropriate SAA, rather than create a separate split-form operator SAA to sit alongside the 2015 SAA as has occurred for the 2010 Aurizon Network access undertaking.

Summary 7.3

The 2015 SAA must give effect to a more balanced risk position for all parties to the agreement (see Appendix C and Appendix G).

7.5 KPIs and performance reporting regime

The QCA considers that performance indicators are an effective method for encouraging a party to comply with its obligations in an access agreement as well as encouraging improvements in efficiency and productivity. The 2008 SAA contained provisions relating to compliance with agreed operational performance levels. Queensland Rail has not included a KPI regime within its proposed 2015 SAA. Queensland Rail said in its December 2015 submission that no KPI regime had ever been agreed between the parties despite the good faith negotiation provisions in the previously approved SAAs.

We consider that for KPIs to be effective and worthwhile, an initial performance reporting regime is imperative to create baseline data against which KPIs can be measured.

In our Draft Decision we proposed that Queensland Rail commence a performance level reporting regime and provided for the parties to agree KPIs after the initial performance reporting had commenced and had provided meaningful data.

Stakeholders' submissions

Stakeholders were supportive of the inclusion of the QCA's proposed reporting and KPI regime.²⁷³ However, stakeholders said that financial incentives should be included in the DAU or the SAA and not left to be agreed between the parties and that the performance regime should be subject to the dispute resolution provisions.²⁷⁴

Queensland Rail stated that it was generally supportive of measures which improved transparency. However, Queensland Rail suggested that parts of the proposed regime were onerous and asymmetrical.²⁷⁵

QCA analysis and Decision

We require Queensland Rail to amend its proposed 2015 SAA to include a performance reporting regime.

The 2015 SAA should then allow for an access holder to use the information provided by the reporting regime to negotiate KPIs and financial incentives during the term of the relevant access agreement. The QCA considers that the absence of any reference to KPIs in Queensland Rail's proposed 2015 SAA creates a risk profile that inappropriately favours Queensland Rail.

This position is consistent with our Draft Decision. As such, we have adopted our Draft Decision position in relation to the KPI and reporting amendments, except for the additional amendments required and discussed below (see section 7.4 of the Draft Decision).

Expanding the KPIs

All stakeholders considered that including a KPI reporting regime was important to encourage improvements to performance and efficiency. Stakeholders also generally supported the QCA's Draft Decision. Nevertheless, stakeholders (other than Queensland Rail) said that the QCA's proposed provisions did not go far enough.

We agree with Aurizon and Asciano that the reporting obligations should be expanded.²⁷⁶ But, we do not accept Aurizon's suggestion to include an initial report on the track condition, with Queensland Rail then required to report on any deviation from this established baseline²⁷⁷, as this would be inappropriately onerous. Similarly, we do not consider that path availability, as suggested by Asciano, should be included. Our required amendments should provide this information indirectly.

Rather, we consider that the obligations should be expanded to the three additional matters suggested by New Hope²⁷⁸:

- the number of train services cancelled during the month
- the number of train service cancelled during the month which are not rescheduled
- a list of speed restrictions in place at the end of each month (including when such restriction was applied, the speed limit and the start and finish locations).

²⁷³ See New Hope, sub. 24: 7; Aurizon, sub. 20: 44; Glencore, sub. 24: 3–4; Yancoal, sub. 27: 4.

²⁷⁴ See Asciano, sub. 28: 19–20.

²⁷⁵ Queensland Rail, sub. 26: 95–96.

²⁷⁶ See Asciano, sub. 28: 19–20.

²⁷⁷ Aurizon, sub. 20: 54.

²⁷⁸ New Hope, sub. 24: 7–8 and schedule 5 of New Hope's amended 2015 SAA.

These obligations are less onerous than those suggested by Aurizon and Asciano, but still provide a direct indicator of whether or not Queensland Rail is satisfying its obligations under an access agreement; including acting as an indirect indicator of track condition. We do not consider that these reporting requirements are inappropriate having regard to Queensland Rail's legitimate business interests, as contracting for the provision of train paths is the core of Queensland Rail's business.

That said, the QCA agrees with Queensland Rail's submission that weekly reporting would be onerous whilst not necessarily providing accurate and useable data. We have removed the obligation for weekly reporting, consistent with Queensland Rail's submission.

We agree with New Hope's suggestion that Queensland Rail should warrant as to the accuracy of the data it provides.²⁷⁹ Without the quality control that this warranty will encourage, data may be misleading or incorrect and consequently make the provisions of limited utility.

KPIs, financial incentives and other matters

Stakeholders said that settled and approved financial incentives should be included by the QCA rather than left for negotiation between the parties.²⁸⁰ Consistent with the Draft Decision, we consider that the negotiation of incentives is a matter which should be left to the contracting parties.

The negotiation of financial incentives is something that requires a certain threshold of baseline reporting to give the parties meaningful data upon which to base their negotiations. We consider that, once the parties have access to this information, it is a matter for the two sophisticated commercial parties to decide amongst themselves. Each party's performance can be measured against the established baseline. Financial incentives can then be applied to encourage a party to improve their performance against the established baseline. The QCA's decision is intended to provide the right conditions for these negotiations and their outcomes to be effective.

We agree with Queensland Rail that the proposed reporting regime may provide a distorted view of Queensland Rail's performance by requiring weekly reporting which does not take account of seasonality, maintenance regimes or access holder operations.²⁸¹

However, we do not agree with Queensland Rail that our required performance reporting obligations are inappropriately asymmetrical. Our required drafting provides that the parties are able to agree additional performance reporting criteria as well as incentives. Parties may agree to impose performance reporting obligations on access holders in the future. However, the initial baseline data in relation to Queensland Rail's network will first be required to inform operator reporting requirements (if any). Further, our initial KPIs are necessarily focused on the service provider, being the party that sells access to the service.

Separately, we agree with stakeholders that the reporting and incentive provisions should be made subject to the dispute resolution provisions. This will increase the effectiveness of the provisions and provide that, if one party is unreasonably unwilling to negotiate or agree financial incentives, the QCA can arbitrate the dispute.

The QCA considers that it would be inappropriate not to include a KPI regime within the 2015 SAA. A KPI regime will work to keep Queensland Rail and access holders accountable to their

²⁷⁹ New Hope, sub. 4: 8.

²⁸⁰ Glencore, sub. 25: 4; Yancoal, sub. 27: 4; New Hope, sub. 24: 7–8.

²⁸¹ Queensland Rail, sub. 26: 95–96.

obligations consistent with their commercial commitments. Further, an effective KPI regime will incentivise improved performance. This will benefit all parties and increase the efficiency and productivity of the network as a whole.

We therefore consider it appropriate for Queensland Rail to amend its proposed 2015 SAA to include a performance reporting regime which will serve to establish baseline performance criteria. These reporting criteria are outlined in Box 2 and are to be included in Schedule 5 to the 2015 SAA. We also consider it appropriate to include that the parties will, after 12 months of reporting data has been provided and if requested by one of the parties, meet to negotiate (in good faith), the inclusion of additional performance criteria and incentives (financial or otherwise) based on the baseline data.

Box 2: KPIs to be included in the 2015 SAA

A new schedule has also been included in the 2015 SAA to identify the minimum monthly operational reporting obligations based on the delivery of monthly train paths on an origin-destination pairing basis consistent with the agreement.

The performance levels to be reported by Queensland Rail under the agreement include:

- contracted versus scheduled versus actual TSE consumption by the access holder;
- network availability days for the track utilised by the agreement;
- planned and unplanned network maintenance across track utilised by the agreement;
- planned and actual track closures across track utilised by the agreement and the performance of actual track closures with Queensland Rail reporting on the percentage of track closures returned to daily services within the planned timeframe;
- sectional run-time performance for the train services operated under the agreement;
- below-rail transit time performance for the train services operated under the agreement;
- forecast versus scheduled versus actual GTKs hauled under the agreement;
- the number of train services cancelled during the month;
- the number of train services cancelled during the month which are not rescheduled; and
- a list of speed restrictions in place at the end of each month (including when such restriction was applied, the speed and the start and finish locations).

Conclusion

We consider the proposed performance reporting regime in the 2015 SAA is appropriate, having regard to each of the factors in s. 138(2) of the QCA Act for the following reasons:

- The reporting regime, in conjunction with agreed KPIs and incentives, will promote the efficient operation of Queensland Rail's network and address the legitimate business interests of Queensland Rail and interests of persons seeking access to, and holding access rights in relation to, the network (s. 138(2)(a), (b), (d) of the QCA Act), by:
 - providing clarity and transparency on Queensland Rail's contractual obligations which underpin the provision of an access service;
 - enabling an access holder to monitor and hold Queensland Rail accountable for the nondelivery of access services due to Queensland Rail's non-compliance with its operational and service obligations under the agreement (and vice-versa);
 - providing all access seekers, access holders and end customers with the required level of commercial certainty to enable them to enter into long-term access contracts with Queensland Rail and be confident that sufficient contractual remedies are available, should Queensland Rail not comply with its obligations under the agreement;

- providing Queensland Rail with a level of accountability that is commensurate with the service level obligations contained in the 2015 SAA;
- facilitating access to the network to maximise the operation and use of access rights which, in turn, will improve network productivity and lower the unit cost of access;
- minimising the potential for access disputes to be triggered under the 2015 DAU; and
- providing a default reporting regime to facilitate the timely execution of access agreements.
- The KPI reporting regime promotes the public interest in having competition in markets (s. 138(2)(d) of the QCA Act) where participation in that market is reliant on Queensland Rail's monopoly provision of an access service by keeping Queensland Rail accountable for its contracted obligations.
- The performance and KPI regime is also appropriately consistent with the previous regulatory precedents approved by the QCA (s. 138(2)(h)).

Summary 7.4

The 2015 SAA must:

- (a) include a mandatory performance reporting regime which includes reporting on those matters outlined in Box 2 (above)
- (b) provide a framework for the parties to negotiate KPIs and incentives based on the information provided by the performance reporting
- (c) require Queensland Rail to warrant as to the accuracy of the information it provides to operators and access holders.

See clauses 4.6, 6.7, 23 and Schedule 5 in Appendix G.

8 **REFERENCE TARIFFS**

The 2015 DAU included West Moreton network and Metropolitan network reference tariffs for coal-carrying train services. Queensland Rail proposed a rate equivalent to \$19.41/'000 gtk (\$19.74/'000 gtk as at 1 July 2016) and sought to justify it on the basis that this tariff was below its proposed ceiling price of \$34.92/'000 gtk. These are the only reference tariffs in the 2015 DAU.

Our Decision is to refuse to approve Queensland Rail's reference tariffs and ceiling price proposal. We require reference tariffs as at 1 July 2016 equivalent to \$17.92/'000 gtk for the West Moreton network and \$16.66/'000 gtk for the Metropolitan network.

Among other things, we require that Queensland Rail:

- apply a price cap form of regulation with a take-or-pay capping mechanism and revise the allocation of costs between coal and non-coal traffics
- use efficient reference tariff building blocks, including for maintenance costs and the regulatory asset base;
- continue with a two-part tariff; and
- adopt an 'adjustment amount' mechanism to address an over-recovery of access charges since 1 July 2013.

Background

The West Moreton network was built 150 years ago for mixed freight and passenger services. Upgrades to cope with the coal services that began operating in 1996 were incremental rather than reflecting a fundamental reconfiguration of the network. The West Moreton network remains a low-volume system with low-capacity coal trains that need to travel through the passenger-focused Metropolitan network to reach the port of Brisbane.

Recently, the West Moreton network has seen declining coal and non-coal freight demand. Forecast weekly return coal train paths used for pricing purposes have fallen to 63 during the 2015 DAU period, compared to 77 contracted paths used in the 2013 DAU. Non-coal services have dropped to 3 from 29. This represents reductions of 18 per cent and 90 per cent for coal and non-coal train services respectively.

Chapter structure

This chapter sets out in three parts our approach to determining an appropriate reference tariff for Queensland Rail's West Moreton and Metropolitan networks:

- Part A—Regulatory context and operating assumptions—outlines the QCA's approach to reviewing Queensland Rail's reference tariffs and related regulatory processes;
- Part B—Tariff building blocks and calculation—outlines the QCA's position on each cost element that forms part of the approved tariffs, and the proposed tariffs over the forward regulatory period; and
- Part C—Adjustment amount—outlines the QCA's approach to addressing the expectation of an adjustment to reflect the difference between the tariffs Queensland Rail actually charged and the tariffs that would have applied in relation to the period from 1 July 2013.

PART A-REGULATORY CONTEXT AND OPERATING ASSUMPTIONS

The West Moreton and Metropolitan network tariffs in the 2015 DAU are based on a number of mechanisms, including mechanisms for allocating costs, assessing capacity and determining take-or-pay charges. The DAU also contains processes for varying the reference tariff during the term of the undertaking and accepting capital expenditure into the regulatory asset base (RAB). Key issues are summarised in Table 13 below. Matters that require a more detailed explanation are discussed in Sections 8.1 to 8.9.

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision		
1. Regulatory context of the QCA Decision					
Had regard to all of the approval criteria in s. 138(2) of the QCA Act in forming a view on Queensland Rail's proposed tariff.	Disagreed and said we failed to have regard, among other things, to the pricing principles and Queensland Rail's legitimate business interests.	Generally supported the Draft Decision.	See Section 8.2 below.		
Affordability not given a material weighting.	No comment.	New Hope and Yancoal said affordability issues should lead to a lower tariff.	See Section 8.2.1 below.		
Proposed a reference tariff based on efficient costs, and rejected Queensland Rail's de- coupling proposal.	Disagreed. Reaffirmed its proposed ceiling price and a reference tariff below this level.	Generally supported the Draft Decision.	See Section 8.2.2 below.		
2. Allocation of common costs	5		1		
Accepted Queensland Rail's position that government- imposed restrictions prevented it from contracting more than 87 coal paths through the Metropolitan network.	Claimed that the 87- path constraint was not legally binding; QCA approach was erroneous and not appropriate.	Supported our Draft Decision	See Section 8.3.1 below.		
Categorised maintenance and operating costs into fixed and variable components; treated all fixed costs (including forecast capital expenditure) as common network costs.	Disagreed and suggested re- categorising fixed costs into common and coal- triggered costs.	Supported our Draft Decision	See Section 8.3.2 below.		
Proposed that coal traffics pay common network fixed costs reflecting the proportion of West Moreton network capacity they can contract and pay variable costs reflecting their share of forecast usage.	Agreed with variable cost allocation. Disagreed with fixed cost allocation and said coal traffics should pay common network fixed costs reflecting their share of total forecast paths.	Agreed with variable cost allocation. Disagreed with fixed cost allocation and said coal traffics should pay common network fixed costs reflecting the proportion of total available paths coal services were forecast to use.	See Section 8.3.3 below.		

Table 13: Summary of key positions—regulatory context and operating assumptions

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision		
3. Capacity assessment					
Noted consultant's assessment that West Moreton capacity was 135 weekly return paths; said final views were subject to stakeholders comments.	Disagreed and said capacity was 112 return paths as proposed in the 2015 DAU.	QCA should use its consultant's capacity estimate, unless that estimate was flawed.	See Section 8.4 below.		
Metropolitan network operations reduced West Moreton capacity by 17 per cent.	Disagreed and said the Metropolitan impact was 12.1 per cent, as proposed in the DAU.	Disagreed and said Metropolitan impact was 22 per cent.	See Sections 8.4 and 8.4.1 below.		
4. Form of regulation and take	e-or-pay				
West Moreton and Metropolitan tariffs should be subject to a price cap with a reset of the West Moreton tariff if contracted volumes are above forecasts.	Queensland Rail disagreed with a volume reset.	Supported an endorsed variation event for changes in contracted volumes.	See Section 8.5.1 below.		
Take-or-pay should be 100 per cent of access charges, capped at total revenue allocated to coal services in assessing tariffs.	Supported 100 per cent take-or-pay, but opposed capping.	Opposed 100 per cent take-or-pay, but supported capping.	See Section 8.5.2 below.		
5. Tariff structure		1	1		
Accepted Queensland Rail's proposed two-part structure of train-path-based and gtk-based tariff components.	No comment.	New Hope said the QCA should confirm tariffs would not breach the pricing limits.	See Section 8.5.4 below.		
6. Metropolitan network tarif	fapproach				
Applied tariff derived on West Moreton network, modified to remove double counting of capex since 2002; separate incremental train path charge.	Disagreed and said the QCA proposal was a 'dramatic move away' from past practice.	Supported the Draft Decision, but said QCA should be clear on what would happen in future.	See Section 8.6 below.		
7. Productivity, innovation an	d incentives				
No specific productivity measures in reference tariff provisions (Schedule D).	No comment.	Aurizon said there should be incentives for innovations that provide operational efficiencies.	See Section 8.7 below.		
8. Variation of reference tariff	s				
Accepted Queensland Rail's process for varying reference tariffs.	No comment.	No comment.	See Section 8.8 below.		
9. Capital expenditure assess	nent process				
Proposed to not require a DORC valuation and to provide for consulting with stakeholders; proposed optimising assets for possibility of actual bypass.	Disagreed with removing DORC valuation and optimising assets for possible actual bypass.	Supported Draft Decision.	See Section 8.9 below.		

8.1 Queensland Rail's 2015 DAU proposal

The 2015 DAU included multi-part reference tariffs for coal-carrying train services operating on the West Moreton and Metropolitan networks. Queensland Rail said these were equivalent to \$19.41/'000 gtk (as at 1 July 2015), derived by escalating by inflation the reference tariff of \$19.14/'000 gtk that applied from 1 July 2014.²⁸²

Queensland Rail said that, as the reference tariffs were below its derived ceiling price of \$34.92/'000 gtk, we should approve them. Queensland Rail said its ceiling price was based on a building block approach. It used volume forecasts and made various allocations of common costs and assets between coal and non-coal services using different allocators for the RAB, operating costs and maintenance costs.

8.2 Regulatory context of our Decision

In assessing Queensland Rail's proposed reference tariff for coal-carrying train services, we have had regard to various matters, including:

- Queensland Rail's legitimate business interests, and the ability of Queensland Rail to recover efficient costs and a return on investments relating to coal-carrying train services;
- access seekers' and access holders' interest in not paying for network capacity they are unable to contract because of government-imposed restrictions; and
- access seekers and access holders' expectations about an adjustment to reflect the difference between the tariffs Queensland Rail actually charged and the tariffs that should have applied since 1 July 2013.

Our Decision is to refuse to approve Queensland Rail's proposed ceiling price and reference tariff, having regard to the approval criteria in the QCA Act. In particular, Queensland Rail proposed:

- an opening asset valuation that would provide windfall gains to Queensland Rail. Windfall gains do not promote the economically efficient investment in and use of the rail infrastructure and are contrary to the interests of access seekers, access holders and their customers (s. 138(2)(a), (e), (h)); and
- allocating costs to coal-carrying trains reflecting capacity they are unable to contract. This
 recovery from coal services would reflect inefficient price discrimination, which would not
 be appropriate, having regard to the pricing principles (ss. 138(2)(g) and 168A(b)). It is also
 not in the interests of access seekers, access holders and their customers to pay for services
 they cannot contract to use (s. 138(2)(e), (h)).

We have had regard to each of the approval criteria in section 138(2) of the QCA Act in forming our views on the West Moreton and Metropolitan reference tariffs. Our approach to the criteria is discussed in more detail in Chapter 10 of this Decision.

In this context, Queensland Rail and other stakeholders raised the following particular thematic issues in relation to our 2015 Draft Decision:

- affordability of the reference tariff and relative prices;
- de-coupling of the reference tariff from the ceiling price; and

²⁸² Queensland Rail, sub. 2: 5. Note that the 1 July 2014 tariff is escalated from the 2009–10 tariff of \$16.81/'000 gtk approved by the QCA in 2010.

• the limitations of the network.

These matters are discussed below.

8.2.1 Affordability and relative prices

Our October 2015 Draft Decision said that 'we have not had regard to the effects of short-term business cycles and hence affordability in considering the asset valuation for this Draft Decision'.²⁸³ We said:

While our decision has not given a material weighting to the issue of relative prices of other train services, we do not consider that we must be precluded from taking relative prices into account in the future. Indeed, such comparisons are amongst a range of factors we could give greater weight to when assessing a reference tariff under the approval criteria in the QCA Act, especially in the face of material falling demand on the West Moreton network.²⁸⁴

Additionally:

While we looked at 'relative prices' in our October 2014 Draft Decision (see p. 153) we did not take 'affordability' into account then and have not done so this time.²⁸⁵

New Hope said in its December 2015 submission that the Draft Decision appeared to lack any explanation why the QCA had:

- (a) failed to give material weight to relative prices and affordability in the face of actual (rather than hypothetical) material falling demand in the West Moreton Network; and
- (b) not acted on the 'prima facie case' that consideration should be given to reducing the value of assets to prevent a further decline in demand for access.²⁸⁶

New Hope also said that it sought a tariff which was competitive in the long term.

However, shorter term considerations such as current market conditions remain relevant as these conditions increase the risks involved in over-estimating tariffs. Current market conditions are also relevant to issues involving the timing of cashflows ...²⁸⁷

Yancoal said the cost of rail access on the West Moreton system was 'well above' that in the Bowen Basin and Hunter Valley, and the current and proposed access charges did not promote competition.

Yancoal considers that it is clearly in the public interest, the interest of access holders and access seekers and consistent with the object of Part 5 of the QCA Act to reduce the proposed tariff to reflect the affordability and competitiveness arising from the current and proposed tariffs ...²⁸⁸

However, Queensland Rail said:

it is not the role of the regulator to determine a below-rail access charge which makes coal mines 'competitive', any more than this obligation should be forced upon an above rail provider, or downstream port/terminal operator.²⁸⁹

The QCA's view

While we have had regard to affordability and market conditions to the extent that they affect matters including the utilisation of the network, we also note that there are competing

²⁸³ QCA 2015: 173.

²⁸⁴ QCA 2015: 138–139.

²⁸⁵ QCA 2015: 139, footnote 367.

²⁸⁶ New Hope, sub. 22: 6; sub. 21: 4–5.

²⁸⁷ New Hope, sub. 22: 6.

²⁸⁸ Yancoal, sub. 27: 3–4.

²⁸⁹ Queensland Rail, sub. 33: 12.

considerations. These include regulatory predictability and certainty in the regulatory process and its outcomes, achieved by applying commonly accepted regulatory methodologies consistent with the approval criteria in the QCA Act (s. 138(2)). The various considerations are outlined further in Chapter 10.

On balance, we consider that, while affordability and relative prices are relevant to assessing the West Moreton and Metropolitan tariffs, they do not on their own outweigh the other considerations, including the consideration of whether the price of access generates expected revenue that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved.

Further, setting a price based on short-term affordability considerations would have the potential to promote inefficient development of high-cost mines, which would not be in the public interest, and would not promote efficient use of the rail infrastructure.

Therefore, it would not be an appropriate balance of, among other things, the interests of Queensland Rail, the public interest, the pricing principles and the interests of access seekers and access holders, for prices to be set in a manner that gives greater weight to supporting the profitability of access holders or seekers and their customers over the regulatory predictability and certainty that will come from establishing a transparent tariff methodology to apply over the long term.

We have also considered the possibility that assets should be optimised to reflect the reduction in demand for below-rail services on the West Moreton and Metropolitan networks. While the volume forecasts provided by Queensland Rail may suggest it is appropriate to optimise the assets, we have not done so at this time, given the capacity constraints for coal services on the West Moreton network and the uncertainty about future volumes (see Sections 8.3.3 and 8.10 of this Decision).

8.2.2 De-coupling and regulatory uncertainty

QR Network's 2009 DAU proposed a 2009–10 reference tariff that it justified on the basis that it was lower than its proposed ceiling price.²⁹⁰ Our December 2009 Draft Decision rejected QR Network's tariff approach and proposed a lower ceiling price that would also apply as the reference tariff.²⁹¹

Queensland Rail has taken an approach this time that is similar to the approach followed by QR Network with the 'below-ceiling' tariff in its 2009 DAU. In Queensland Rail's 2015 DAU, it proposed a West Moreton network ceiling price of \$34.92/'000 gtk for 2015–16 and a 'decoupled' reference tariff of \$19.41/'000 gtk.²⁹²

Queensland Rail said it proposed the lower reference tariff because its calculated ceiling price was 'higher than the commercially prudent access charge'.²⁹³ It said the 'overwhelming factor

 ²⁹⁰ QR Network proposed a ceiling price of \$34.00/'000 gtk and a reference tariff of \$22.07/'000 gtk for the network west of Rosewood. See QCA, December 2009: 71; QR Network, September 2008: 118–119.
 ²⁹¹ We approach a ceiling price of \$46.01 (2000 ctk). See QCA, December 2009: 71; QR Network, September 2008: 118–119.

 ²⁹¹ We proposed a ceiling price and reference tariff of \$16.81/'000 gtk. See QCA, December 2009: 94.
 ²⁹² Queensland Rail, sub. 2: 4–6. This \$19.41/'000 gtk tariff was escalated by the CPI from the \$16.81/'000 gtk

price originally proposed for 2009–10 in the QCA's 2009 DAU Draft Decision.

²⁹³ Queensland Rail, sub. 2: 4.

contributing' to the sharp increase in the ceiling price was a changed market outlook for both coal and non-coal services.²⁹⁴

Our October 2015 Draft Decision rejected the de-coupling approach and proposed a ceiling price of \$18.88/'000 gtk for 2015–16.

Queensland Rail said in its December 2015 submission that its:

reference tariff proposal under the 2015 DAU of \$19.41/'000gtk (well below the ceiling price) was proposed at a level consistent with the status quo as a compromise to provide regulatory certainty.²⁹⁵

Queensland Rail also said:

For the West Moreton network, if the QCA accepts Queensland Rail's 'de-coupling' of the ceiling revenue limit from the reference tariff that would apply to current and future users, it negates the need for the RAB value to be adjusted.²⁹⁶

The QCA's view

We disagree with Queensland Rail's suggestion that its proposed de-coupling provides regulatory certainty. As long as the reference tariff is at some arbitrary level below a ceiling price, access seekers and holders will expect that the ceiling price will apply at some point in the future. As we said in 2009:

It would be very difficult for a non-QR party to enter the market to compete with incumbent operator QR Freight if it was believed that the access charge could be increased significantly, but with the only justification for the change being that the tariff remains lower than an estimated ceiling price.

The Authority does not accept that a process where the tariff is set on the basis that it is lower than a ceiling tariff is sufficiently transparent, robust or repeatable.²⁹⁷

While Aurizon Operations (formerly QR Freight) is no longer vertically integrated with Queensland Rail's below-rail business, the point still stands—the uncertainty about future prices in Queensland Rail's proposal would not promote competition. As Professor Menezes says:

QR's proposed tariff in its 2015 DAU implies that it is recovering less revenue than the maximum amount of revenue that it would be allowed to recover under a DORC valuation that allows QR to earn a return on assets with expired useful lives. However, as this asset valuation is rolled over to the next regulatory period, QR will still be able to charge a higher tariff, based on the higher DORC valuation, which will be associated with a risk that allocative efficiency will be negatively impacted.²⁹⁸

Accepting Queensland Rail's de-coupling proposal would implicitly mean that Queensland Rail was entitled to charge up to its proposed ceiling price. But this, in turn, would mean that the QCA would be:

 endorsing a cost build-up for Queensland Rail that went beyond approving efficient costs for Queensland Rail over the regulatory period; and

²⁹⁴ Queensland Rail, sub. 2: 6.

²⁹⁵ Queensland Rail, sub. 26: 24.

²⁹⁶ Queensland Rail, sub. 26: 27.

²⁹⁷ QCA 2009a: 73.

²⁹⁸ Menezes, F, 2016b: 11–12.

 providing for coal services to underwrite Queensland Rail's exposure to 'the changed market outlook, for ... non-coal services'²⁹⁹—these market changes being a risk that coal miners are not able to manage or control.

Neither of these outcomes would be appropriate having regard to the approval criteria in the QCA Act (s. 138(2)(a), (d), (e), (h)).

It is appropriate for the reference tariff to be based on a well-understood approach that derives the price from underlying costs and asset values. That has the benefits of:

- providing regulatory certainty to access holders, access seekers and Queensland Rail; and
- enabling all parties to assess the tariff implications of proposed measures such as expansions of the network or changes in train configuration or operation.

The QCA's reference tariff approach therefore seeks to promote the efficient operation and use of, and investment in, the rail network. It also promotes the public interest, and is in the interests of access seekers and access holders (s. 138(2)(a), (d), (e), (h)).

We note that Queensland Rail's proposed de-coupling reflects an implicit recognition that it may not be appropriate—or necessary—for Queensland Rail to recover the full ceiling price it has proposed. In other words, if Queensland Rail was correct that \$34.92/'000 gtk was the efficient ceiling price, then its proposed reference tariff would mean it was forgoing recovery of some of its sunk costs, and potentially other costs.

In practice, this means that Queensland Rail has submitted a proposal for a reference tariff that is below what it has argued is necessary to comply with section 168A(a) of the pricing principles.

Summary 8.1

The West Moreton network reference tariff in the 2015 DAU must not specify that Queensland Rail is applying a reference tariff that is below and separate from the ceiling revenue limit.

See Schedule D, clause 1.1 (deleted) in 2015 DAU.

Limitations of the network

Queensland Rail said investors knew of the limitations of the 'old, idiosyncratic' network when they sunk their capital in mines that used the West Moreton network. It said:

Queensland Rail should not be penalised for, or disadvantaged due to, the nature of the West Moreton network or because of the business challenges faced by coal mines that have freely chosen to use the West Moreton network.³⁰⁰

We agree with Queensland Rail—the miners would or should have been aware of the nature of the infrastructure. However, that awareness would not be a reason why the access holders should expect to or be required to pay both for high maintenance costs and for an asset value that delivers monopoly rents and windfall gains to Queensland Rail. To put it another way, Queensland Rail should also be aware of the limitations of its own network, and should expect a pricing regime that reflects those limitations.

²⁹⁹ Queensland Rail, sub. 2: 6.

³⁰⁰ Queensland Rail, sub. 26: 47–48.

8.3 Allocation of common network costs

As West Moreton is a mixed traffic system, a share of the common network costs (opening asset base, forecast capital expenditure, maintenance and operating costs) is allocated to coal traffics. The coal-allocated costs are used in a regulatory building block model to develop a reference tariff for coal-carrying train services.

2015 DAU context: significant spare capacity with high forward-looking costs

Unlike the 2009 DAU and the 2013 DAU, the 2015 DAU has been developed in light of a material fall in demand for below-rail services on the West Moreton network—about three-fifths of the system capacity (66³⁰¹ of the 112 paths) is forecast to be used by coal and non-coal traffics combined, resulting in about 40 per cent spare capacity (Figure 3). This spare capacity is due to reductions of 18 per cent and 90 per cent in coal and non-coal train services respectively, compared to the 2013 DAU.



Figure 3: West Moreton network capacity utilisation status

Note: a) The 2009 DAU capacity utilisation status corresponds to the last two years of the DAU period (2011–12 and 2012–13. b) The 2015 DAU forecast coal paths are 62.8 (for presentational purpose we have used 63). Source: QR Network (2009); QCA (2009b); Queensland Rail (2013a: 8; 2013d: 5; sub 2: 20).

Notwithstanding the significant reduction in forecast traffic volume, forward-looking costs in the 2015 DAU are generally higher on a per unit basis than in the 2013 DAU that was developed when the system was nearly capacity constrained. That is:

• maintenance cost per kilometre (a measure of unit cost) is slightly higher when comparing around \$67,026 (proposed in the 2015 DAU) with \$66,984 (proposed in the 2013 DAU);³⁰²

³⁰¹ The 2015 DAU total forecast paths are 65.8. For presentational purposes we use 66 paths.

 ³⁰² QCA calculation based on data reported in Queensland Rail 2013b: 3, sub. 2, Appendix 6: 22; QCA, 2014d:
 123; B&H 2015: iii, 4. Amounts are in June 2014 dollars for which the 2015 DAU data were converted to June 2014 dollars. From Queensland Rail's 2015 DAU submission it was not clear what dollar year the constant

- forecast capital expenditure per kilometre is about 26 per cent higher when comparing around \$64,000 in the 2015 DAU with \$51,000 proposed in the 2013 DAU;³⁰³ and
- operating cost is unchanged at around seven million dollars annually.³⁰⁴

Queensland Rail has predicted in its West Moreton System Asset Management Plan that the elevated levels of maintenance and capital spending on the network infrastructure will continue for at least another decade.³⁰⁵

2015 DAU cost allocation proposal and Draft Decision

Queensland Rail said that given the changed volume outlook for the 2015 DAU a different approach was required to allocate common network costs to coal traffics for pricing purposes, as compared to the approach in its 2013 DAU proposal³⁰⁶ (see Appendix E for a summary of previous cost allocation assessments).

For the opening asset base, Queensland Rail said it proposed an allocation to reflect contracting limits for coal services through the Metropolitan network. Queensland Rail stated:

In particular, the binding constraint is the maximum 87 coal paths per week, limiting the proportion of the capacity of the West Moreton Network that can potentially be contracted to coal to 87 out of 112 available paths, or 77.7%.³⁰⁷

For the forward-looking costs (maintenance and operating costs, and forecast capital expenditure), Queensland Rail proposed an allocation based on coal's share of forecast volumes. Queensland Rail proposed to recover the coal-allocated costs from the 63 forecast coal paths. Queensland Rail said its allocation approach provided it with a 'greater opportunity to recover its efficient costs including a return', even though it proposed a reference tariff 44 per cent below the ceiling price from its building block model.³⁰⁸

Our 2015 Draft Decision accepted Queensland Rail's proposed allocation that reflected the 87 path contracting restriction for coal services through the Metropolitan network. But we considered the allocation should reflect the actual contracting restriction for coal services in the West Moreton network—that is, the 87-path constraint in the Metropolitan network should be reduced further by the number of paths (10 paths at the time) contracted to coal services to operate in the Metropolitan network which did not traverse the West Moreton network. We applied the resulting allocator (77 out of 112 paths) to Queensland Rail's efficient fixed common network costs (i.e. the opening asset base, forecast capital expenditure and fixed maintenance and operating costs). Our Draft Decision allocated efficient variable common network costs (i.e.

cost referred to and for the 2015 Draft Decision we inferred it was June 2014 dollars. In its correspondence of 15 April 2016, Queensland Rail stated that 'there is an error in the calculations in Figures 8.2 & 8.3 of the Draft Decision on maintenance which results in a material overstatement of any possible difference between the 2013 DAU maintenance costs and the 2015 DAU maintenance costs. In relation to this, please note that the 2013/14 data is in 2013/14 dollars, but the \$2015/16 is not and so the figures aren't comparing like for like, resulting in an overstatement of the difference in maintenance allowance'. Therefore, we have now treated the 2015 DAU constant cost data as being in June 2015 dollars.

³⁰³ QCA calculation based on data reported in Queensland Rail, June 2013c: 12, sub. 2, Appendix 3: 22; QCA 2014d: 126–127; B&H 2015: iv & 4. Amounts do not include interest during construction and are in June 2014 dollars for which the 2015 DAU data were converted to June 2014 dollars.

³⁰⁴ Amount is in June 2014 dollars.

³⁰⁵ Queensland Rail, sub. 2, Appendix 6—West Moreton System Asset Management Plan.

³⁰⁶ Queensland Rail, sub. 2: 44–45.

³⁰⁷ Queensland Rail, sub. 2: 48.

³⁰⁸ Queensland Rail, sub. 2: 4–7.

variable maintenance and operating costs) based on coal's share of forecast volumes. Our Draft Decision proposed that coal-allocated costs be recovered from the 63 forecast coal paths.

Stakeholders' submissions

Queensland Rail and coal miners (New Hope and Yancoal) rejected the 2015 Draft Decision cost allocation approach, but they had divergent views on an appropriate cost allocation methodology. Aurizon considered the Draft Decision allocation approach was reasonable.³⁰⁹

Queensland Rail said that our Draft Decision resulted in it being unable to recover the full cost of providing the declared service. Queensland Rail stated that the QCA's rationale for the allocation of fixed costs based on the assumed 87-path constraint was 'erroneous and not appropriate', as there was no legally binding 87 path constraint.³¹⁰

Coal miners were concerned that our allocation and recovery approach would produce a 'death spiral', where the closure of a mine would increase the tariffs by such a significant extent that it would likely shut another mine. They also said that any surplus paths above the current contract levels were equally available for contracting by general freight or other commodities. Therefore, miners argued that some proportion of the fixed common network costs of the unutilised train paths up to the 77 paths should be allocated to non-coal services.³¹¹

Some of these comments highlighted concerns about an appropriate categorisation of, and allocation of, costs in the face of a material fall in demand, focusing on which party should bear the costs. Some comments focused on the existence of the 87-path constraint for coal-carrying train services.

We have broken our discussions of the matters raised by Queensland Rail and other stakeholders into the following topics:

- 87-path constraint;
- categorisation of forward-looking costs; and
- cost allocation approach.

8.3.1 87-path constraint

During the 2013 DAU assessment process, Queensland Rail and New Hope had both identified the existence of government restrictions for their inability to contract additional coal-carrying train services. For example, in the 2013 DAU coal train services were contracted to use 77 paths and there were six uncontracted paths (see Figure 3), in relation to which Queensland Rail stated:

Government have not indicated a willingness to contract additional coal services and in relation to non-coal freight, above rail operators have not shown a willingness to contract additional services.³¹²

New Hope had stated:

The level of paths which is contracted is artificially constrained (below true system capacity) by Government (QR's shareholder). NHG has been seeking to contract additional train paths for the past three years and has been unable to do so because of this constraint.³¹³

³⁰⁹ Aurizon, sub. 20: 25.

³¹⁰ Queensland Rail, sub. 26: 12, 22, 27, 30, 39.

³¹¹ New Hope, sub. 21: 4; Yancoal, sub. 27: 2.

³¹² Queensland Rail 2013d: 5.

In its 2015 DAU supporting submission, Queensland Rail reiterated that government restrictions for coal services limited its ability to contract the full amount of capacity, and it identified two constraints, being:

- preserved freight and passenger train paths from Rosewood to Toowoomba, which was 13
 paths for freight³¹⁴ and two for passenger services; and
- a constraint of 87 coal paths per week through Metropolitan network specified by Queensland Rail's 'Responsible Ministers'.³¹⁵

Queensland Rail added that 'the binding constraint is the maximum 87 coal paths per week', which limited the proportion of the capacity of the West Moreton network that could potentially be contracted to coal to 87 out of 112 available paths or 77.7 per cent. Accordingly, Queensland Rail considered it reasonable to cap the allocation of the initial asset base (or opening asset value as at 1 July 2015) to coal traffics at 77.7 per cent.³¹⁶

Our 2015 Draft Decision accepted Queensland Rail's allocation approach to reflect the 87-path constraint for coal services in the Metropolitan network. However, we reduced it by the number of paths (10 paths at the time) contracted to coal services to operate within the Metropolitan network but not traverse the West Moreton network, to reflect the contracting restriction for coal services in the West Moreton network. We applied the resulting allocator of 68.8 per cent (i.e. 77 out of 112 paths) to West Moreton efficient fixed common network costs, which included the opening asset base as well as forecast capital expenditure and fixed maintenance and operating costs. We considered that coal traffics should only pay for efficient fixed common network costs that reflected the proportion of paths available for contracting by coal services.³¹⁷

Stakeholders' submissions

In its response to the Draft Decision, Queensland Rail stated that '87 train path is not a legally binding constraint' and provided legal advice in support of its argument.³¹⁸ Queensland Rail said that all train paths in the Metropolitan network that were not allocated to existing train services were available for contracting by coal trains or other services. In this context, Queensland Rail said:

the QCA's rationale for the allocation of fixed costs based on the assumed 87 path constraint is erroneous and not appropriate³¹⁹

[a]s Queensland Rail has previously submitted and demonstrated there is no 87 train path constraint. It is unclear why the QCA continues to raise this as an issue ...³²⁰

However, West Moreton network users stated that Queensland Rail's assertion that there was no cap on contracting for coal services was completely inconsistent with Queensland Rail's practice to date. They said the cap had always been in place, and their investment and

³¹³ New Hope 2014: 3.

³¹⁴ Queensland Rail's 17 July 2015 response to the QCA's request for additional information noted that the reference to '13 paths for freight' was incorrect and that it should be '14 paths for freight'.

³¹⁵ Queensland Rail, sub. 2: 48. Queensland Rail's 17 July 2015 response to the QCA's request for additional information clarified that the restriction was advised by the Department of Transport and Main Roads.

³¹⁶ Queensland Rail, sub. 2: 48–49.

³¹⁷ QCA 2015: 143–145.

³¹⁸ Queensland Rail, sub. 26: 6 and Annexure 8.

³¹⁹ Queensland Rail, sub. 26: 30.

³²⁰ Queensland Rail, sub. 33: 5. Queensland Rail repeated essentially the same argument in a letter sent on 14 April 2016.

contracting decisions were formed in part on that basis.³²¹ Coal miners provided confidential correspondence from Queensland Rail and the Queensland Department of Transport and Main Roads (DTMR) in support of their argument.³²²

QCA analysis and Decision

We have considered Queensland Rail's 2015 DAU and reviewed our 2015 Draft Decision in light of comments and information provided by Queensland Rail and other stakeholders in accordance with the QCA Act.

Based on the considerations set out in this section and having regard to the relevant factors in section 138(2) of the QCA Act, our Decision is that 87 paths is a binding constraint for coal train services through the Metropolitan network. We require that the 2015 DAU reflect this constraint.

Queensland Rail's changed positions

Queensland Rail had made—during the 2013 DAU assessment process and in its 2015 DAU supporting submission—several representations that coal services cannot contract more than 87 paths through the Metropolitan network. Queensland Rail had made similar representations at industry forums—for example, at a rail forum organised by the Goondiwindi Regional Council on 4 December 2015, about three weeks before submissions on our 2015 Draft Decision were due (see Figure 4).



Figure 4: Queensland Rail's representation at an industry forum

Source: Queensland Rail 20151: 18.

The QCA notes that Queensland Rail has continued to change its position on this matter. Queensland Rail has variously stated:

• Queensland Rail's Responsible Ministers have specified a constraint of 87 coal paths per week through Metropolitan network.³²³

³²¹ Yancoal, sub. 35: 1; Aurizon, sub. 29: 10–14; New Hope, sub. 31: 3–5.

³²² Queensland Rail 2012, 2014b, 2014c, 2014d, 2014e, 2014f; Queensland Department of Transport and Main Roads (DTMR) 2011, 2014a, 2014b.

- The binding constraint is the maximum 87 coal paths per week.³²⁴
- Due to a constraint from the Queensland government, only 87 out of the 112 paths can be allocated to coal services.³²⁵
- 77 paths able to be contracted for coal on the Range. Coal cannot consume available capacity by contracting additional services.³²⁶
- 'The 2015 DAU allocated ... the asset base based on assumed constraints to contracting'.³²⁷
- There is no such constraint.³²⁸
- There is no legally binding 87-train-path constraint.³²⁹

The QCA is unclear about the rationale underpinning the changes in Queensland Rail's positions and finds Queensland Rail's inconsistent statements puzzling.

Is 87 paths a binding constraint?

Queensland Rail argued that the 87-path reference was contained in correspondence with the DTMR but that correspondence was not a direction from the 'responsible Ministers' and consequently was not a legally binding constraint.³³⁰

However, West Moreton network users said that a Ministerial direction was not necessary and there were other ways in which a constraint could be legally binding. For example, section 266A of the Transport Infrastructure Act 1994 (TIA) prohibited a railway manager (such as Queensland Rail) from allocating a train path that is preserved for one traffic type to a different traffic type without the approval of the chief executive of DTMR.³³¹

Indeed, Queensland Rail also said that it was required to comply with its passenger priority and preserved train path obligations under the TIA,³³² and submitted that 16 paths were preserved for non-coal-train services (see Figure 4).

The QCA's view

In any event, our view is that the relevant consideration is whether the constraint is, and will continue to be, binding after the approval date (i.e. whether it applies in practice, regardless of whether it is legally binding), which is the key concern of users and the matter that is relevant to the QCA in determining how common network costs should be allocated.

The network users argued that Queensland Rail's commercial conduct had been consistent with the existence of the 87 path constraint and that their investment and contracting decisions were based on that constraint. Aurizon stated:

the coal industry and stakeholders have incurred considerable time, cost and resources in seeking to identify options for improving the productivity of existing train paths in reliance on the 87

- ³²⁷ Queensland Rail, sub. 26: 27.
- ³²⁸ Queensland Rail, sub. 26: 30.
- ³²⁹ Queensland Rail, sub. 26: 30.
- ³³⁰ Queensland Rail, sub. 26: 30.

³³² Queensland Rail, sub. 26: 89–90.

³²³ Queensland Rail, sub. 2: 48; Queensland Rail's 17 July 2015 response to the QCA's request for additional information clarified that the restriction was advised by DTMR.

³²⁴ Queensland Rail, sub. 2: 48.

³²⁵ Queensland Rail, sub. 2, Appendix 1: 12.

³²⁶ Queensland Rail 2015I: 18.

³³¹ Aurizon, sub. 29: 11-12; New Hope, sub. 31: 4–5.

weekly train paths constraint. For example, the 87 weekly train paths constraint is also a key assumption underpinning the coal revenue projections in the Inland Rail business case. It is apparent that Queensland Rail does not have a sound appreciation of its ability to influence the decision making of other supply chain participants.³³³

Users said they were denied contracting for paths beyond the constraint at a time when they had sought additional paths. Aurizon said that:

[it] has also received correspondence from Queensland Rail in relation to the ability to contract for services above the 87 train path constraint in which the access provider confirms that it is unable to do so.³³⁴

We have been provided with copies of correspondence in which Queensland Rail asked DTMR to remove the 87-path cap for coal train services, and copies of DTMR's response which showed that the request was not approved.³³⁵ Subsequently, Queensland Rail made representations confirming that the 87-path cap was unchanged and advised coal industry participants that it (Queensland Rail) was unable to contract for coal services above that cap, despite the existence of spare capacity and demand for additional paths by coal train services. Thus, the '87-path' limit for coal services has been applied in practice through Queensland Rail not offering to contract above that level.

To date, Queensland Rail has not provided any evidence to demonstrate that it will behave differently to DTMR's stated position. Indeed, Queensland Rail's inconsistent representations would create substantial uncertainty for investment and contracting decisions by coal industry participants, which is not in the interests of access holders and access seekers, or in the public interest and would affect the efficient use of the network, having regard to section 138(2) of the QCA Act, in particular paragraphs (a), (d), (e) and (h).

Given all these considerations, we are satisfied that the information provided by Queensland Rail and users establishes that it is appropriate to proceed on the basis that the 87-path constraint has applied, and will apply, in practice, as a matter of Queensland Rail's contracting practices, for coal train services through the Metropolitan network.

If Queensland Rail were able to demonstrate that the 87-path constraint does not apply in practice, that would be a matter to which we would have regard in assessing any DAAU or a subsequent DAU. Clearly, the most compelling manner in which Queensland Rail could demonstrate that no such constraint applies in practice would be by providing evidence that it is able and willing to contract coal services above 87 paths. This would be clearest where coal services have contracted up to 87 paths and require additional paths for contracting, and DTMR has removed the 87-path cap.

We accept that, in an environment where coal trains are not contracted up to the 87-path cap, Queensland Rail could seek to demonstrate that DTMR has removed the cap on coal services, and as a result Queensland Rail is able and willing to contract coal services above that cap. In that event, the QCA would be prepared to review its approach to cost allocation within the

³³³ Aurizon, sub. 29: 13.

³³⁴ Aurizon, sub. 29: 12.

³³⁵ Queensland Rail 2012, 2014b, 2014c, 2014d, 2014e, 2014f; Queensland Department of Transport and Main Roads (DTMR) 2011, 2014a, 2014b. Queensland Rail had previously informed that coal traffics were allowed through the Metropolitan network until 2024 (Queensland Rail, November 2013(d): 12). In the 2015 DAU submission, Queensland Rail informed that the end date for coal traffics through the Metropolitan network was 2032 (Queensland Rail, sub. 2, Appendix 6: 7). Thus, although the end date over which the 87-path constraint applies has been extended, the constraint remains unchanged. See Section 8.15 of this Decision for our consideration of the end date of 2032 for coal trains.

context of the approval criteria in section 138(2), including by exploring whether there is a need for optimising the network.

For this decision, we have not assessed whether Queensland Rail's conduct of applying the 87path constraint in practice was consistent with the requirements of section 104 of the QCA Act relating to preventing or hindering access when there were previously requests for capacity above 87 paths for miners, as it is not an issue to be determined in considering the 2015 DAU.

8.3.2 Categorisation of forward-looking costs

In the context of the 2015 DAU, Queensland Rail stated that not all its planned maintenance activities were volume-dependent. For its capital program, Queensland Rail argued that its expenditure was unaffected by the decline in usage, and that its capital program was primarily aimed at 'replacing assets that have reached the end of their useful life'.³³⁶ In Queensland Rail's 2015 DAU supporting submission, PwC observed:

Without incurring any additional capital or maintenance expenditure, the Rosewood to Jondaryan (R2J) part of the network could cater for 15.7 gross million tonnes (GMT) (up from 11.5 GMT); while the Jondaryan to Columboola part of the network could cater for 3.6 GMT (up from 3 GMT).³³⁷

Our 2015 Draft Decision was that Queensland Rail's 2015 DAU maintenance cost should be categorised into fixed common network cost and variable common network cost and we accepted the assessment of our consultant (B&H), that the fixed proportion of maintenance cost was 67.4 per cent. Our Draft Decision accepted Queensland Rail's treatment of the entire forecast capital program as a common network expenditure, as it related to investments in the shared network and benefitted all users.³³⁸

Stakeholders' submissions

Queensland Rail disagreed with the categorisation of maintenance costs and the treatment of the forecast capital program.³³⁹

Queensland Rail said that if the QCA did not accept its 2015 DAU cost allocation methodology, then the Draft Decision categorisation of maintenance costs needed to be modified to include a three-way categorisation into common fixed costs, coal-triggered fixed costs, and variable costs. For the forecast capital program, Queensland Rail said it should be categorised into common fixed costs and coal fixed costs. Queensland Rail submitted reports by Everything Infrastructure and Synergies Economic Consulting (Synergies) in support of its arguments.³⁴⁰

QCA analysis and Decision

We have considered Queensland Rail's 2015 DAU and reviewed our 2015 Draft Decision in light of comments and information provided by Queensland Rail and other stakeholders in accordance with our obligations in the QCA Act.

Based on the considerations set out in this section, and having regard to the relevant factors in section 138(2) of the QCA Act, we consider that it is appropriate to categorise maintenance costs into fixed and variable costs, and that the entire forecast capital expenditure should be

³³⁶ Queensland Rail, sub. 2: 38, 52.

³³⁷ Queensland Rail, sub. 2 (Appendix 1): 13.

³³⁸ QCA (October 2015): 154–156, 185–188; Queensland Rail, sub. 2: 55.

³³⁹ Queensland Rail, sub. 26: 40–41.

³⁴⁰ Queensland Rail, sub. 33: 13–26; Attachments 2 and 3.

treated as a common network expenditure. We require Queensland Rail to amend the 2015 DAU to reflect this.

Categorisation of maintenance costs

Queensland Rail stated that 'the general standard to which the infrastructure must be maintained in order to reliably operate the (63) forecast coal services will be quite different to what would be required only to operate the (three) non-coal services'. Therefore, it argued that cost categorisation should reflect these differences in standard so that 'all costs that are incremental to each type of service are properly allocated to that group of users'.³⁴¹ Queensland Rail submitted its analysis identifying the following three categories of maintenance costs:

- common fixed costs (57.3%), which would be incurred even if only a minimal number of services were to utilise the network;
- coal fixed costs (21.7%), which are triggered as a result of the standard to which the network must be maintained to operate the forecast coal services. Queensland Rail said these costs were fixed over the 2015 DAU forecast period, but were variable over the long run; and
- variable costs (21%), which would vary with tonnages on the network within the 2015 DAU forecast period.³⁴²

We engaged B&H to review Queensland Rail's analysis, including rail-related technical aspects of the Everything Infrastructure and Synergies reports.³⁴³

B&H's assessment

B&H observed that the proportion of fixed maintenance activity is usually considered in the context of a range of tonnages, the network configuration and the condition of the assets.³⁴⁴ B&H identified a number of flaws in Queensland Rail's analysis and assertions. For example:

- The majority of the maintenance and engineering standards in the West Moreton network are driven by passenger trains, as evidenced by the rated line speed of 80 kilometres per hour (km/h) for the railway that is set, based on passenger trains. In comparison, the timetabled average speed for coal train services is 25km/h in the Toowoomba to Columboola section.
- Queensland Rail used historical costs of its various rail lines to identify the three categories of maintenance costs for the West Moreton network, but those rail lines do not have the same configuration and the same condition as the West Moreton network.
- Queensland Rail's data analysis did not establish that there were fixed maintenance costs on the West Moreton network due to the operation of the forecast coal services.³⁴⁵

³⁴¹ Queensland Rail, sub. 33: 14.

³⁴² Queensland Rail, sub. 26: 40–41; sub. 33: 13–23.

³⁴³ The review was undertaken by Mr Martin Baggott, Principal and Director, B&H. Mr Baggott has over 40 years of experience that spans track infrastructure, railway operations and commercial feasibility. He has managed train operations as well as advised on matters associated with asset valuations, operations optimisation, network capital and maintenance to Australian and overseas railways and economic regulators, and inter–governmental agencies, including the Essential Services Commission of Victoria, Economic Regulation Authority of Western Australia, Australian Rail Track Corporation, Canadian Pacific Railway Company, World Bank and Asian Development Bank.

³⁴⁴ B&H (2016, Part 4): 4.

³⁴⁵ B&H (2016, Part 4): 5–10.

B&H concluded that the characteristics of the coal task has, at best, a moderate impact on the way the network is managed for maintenance and capital expenditure, notwithstanding the effect of volume. Thus:

in respect to fixed and variable costs, the context of any determination of categorisation of fixed or variable must have a similar context to the line under analysis and the appropriate use of standards. The analysis should not attempt to identify other lines which have somewhat diverse situations.³⁴⁶

We have accepted most (84 per cent) of Queensland Rail's proposed maintenance cost in the 2015 DAU,³⁴⁷ as they are required given the age and condition of the network and regardless of the low forecast volumes. Those maintenance costs are at a similar level as the 2013 DAU proposal that was developed when the system was nearly capacity-constrained. Given these considerations, we asked B&H to estimate the fixed and variable proportion of maintenance activity based on total network capacity, rather than forecast usage, which is consistent with our approach to allocating fixed common network costs based on the relative proportion of system capacity that is available for contracting by coal services (discussed in the Section 8.3.3 of this Decision).

B&H revised its estimate of the fixed proportion of maintenance activity to account for its revision to some of the maintenance cost items in response to stakeholders' comments (see Section 8.11 of this Decision), and to provide an estimate for a scenario where the network is maintained at its total capacity. Accordingly, B&H assessed that about 57.3 per cent of the maintenance costs related to fixed maintenance activities which were on the common network. B&H said that its estimate of the fixed proportion of maintenance activity was based on Queensland Rail's configuration and condition data for the West Moreton network.³⁴⁸

Table 14 compares B&H's revised assessment with Queensland Rail's analysis.

Cost category	Revised B&H assessment	Queensland Rail analysis
Fixed cost		
Common fixed	57.3%	57.3%
Coal fixed		21.7%
Variable cost	42.7%	21.0%

 Table 14: Maintenance cost categorisation—B&H and Queensland Rail

Source: B&H (2016, Part 4): 11–12; Queensland Rail: sub. 33: 22

The QCA's view

We agree with B&H's criticism of Queensland Rail's cost categorisation approach in the post-Draft Decision submissions and consider that there are a number of flaws in Queensland Rail's proposed cost categorisation approach.

We consider B&H's approach of identifying and categorising costs by the underlying factors, usage-based (variable costs) and time-based (fixed costs), is transparent and robust and will provide appropriate signals for the efficient use of, and operation of, the network, and is in the

³⁴⁶ B&H 2016, Part 4: 10.

³⁴⁷ See Section 8.11 of this Decision for our assessment of Queensland Rail's maintenance cost proposal.

³⁴⁸ B&H 2016, Part 4: 1.

interests of access seekers and access holders, having regard to s. 138(2)(a), (e) and (h) of the QCA Act.

Accordingly, our Decision is to categorise maintenance costs into fixed and variable components and we adopt B&H's revised assessment that the proportion of fixed common network maintenance cost is 57.3 per cent.

Categorisation of forecast capital expenditure

Queensland Rail relied on its maintenance cost analysis to suggest that forecast capital expenditure should be categorised into common fixed and coal fixed costs and asserted that most of its forecast capital expenditure program was designed to support the (63) forecast coal services.³⁴⁹

However, the flaws B&H identified in Queensland Rail's maintenance analysis also apply to Queensland Rail's attempt to re-categorise forecast capital expenditure into common fixed and coal fixed costs.

Besides, our view, as stated in the context of our assessments of the 2009 and 2013 DAUs, is that incremental investment on the network shared by different traffics could benefit all traffics. For example, a project to improve the track standard will result in increased reliability and lower maintenance requirement and will benefit all traffics. Therefore, it is reasonable to consider such investment as a common network capital expenditure and to apply a pro rata allocation to such capital expenditure.³⁵⁰

In its 2015 DAU submission, Queensland Rail had proposed to include in the West Moreton common network RAB³⁵¹ all forecast capital expenditure on the basis that the capital expenditure was on the shared network and benefitted all traffics, and regardless of whether that expenditure was triggered by coal or non-coal services.³⁵² Similarly, the 2015 DAU proposal included, in the West Moreton common network RAB, capital projects on the shared network that were triggered by freight services, which provided benefit to other services. Queensland Rail referred to this capital program as the transport service contract (TSC) capital.³⁵³ We have accepted Queensland Rail's TSC capital program in the common network RAB (see Section 8.14 of this Decision for our assessment of the TSC capital).

It is unclear why Queensland Rail has on the one hand sought to include in the common network RAB a freight-triggered capital program on the shared network that benefits other (i.e. coal) services, but on the other hand has now, in response to our Draft Decision, sought to not include in the common network RAB coal-triggered capital expenditure on the shared network that benefits other (i.e. freight) services.

Given these considerations, our Decision is to treat the 2015 DAU forecast capital expenditure on the shared network as a common network capital expenditure, as we are satisfied that it benefits all traffics.

We also recognise that not all rail infrastructure on the West Moreton network is shared by different traffics. For instance, there have been certain capital programs that benefit only coal

³⁴⁹ Queensland Rail, sub. 33: 23–25.

³⁵⁰ QCA (December 2009a): 80 and (October 2014): 145–146.

³⁵¹ The RAB that is used in determining reference tariffs for coal-carrying train services comprises a share of the West Moreton common network RAB and the entire coal-specific capital expenditure (for example coal-only sidings).

³⁵² Queensland Rail, sub. 2: 49.

³⁵³ Queensland Rail, sub. 2: 35, 49.

services—the Columboola spur and balloon loop project is an example—and we have treated such capital programs as coal-specific capital expenditure.

We consider our approach to treating the forecast capital expenditure on the shared network as a common network expenditure as the underlying project benefits all traffics, provides appropriate signals for the efficient use of, operation of, and investment in, the network, and is in the interests of access seekers and access holders, having regard to section 138(2)(a), (e) and (h) of the QCA Act.

8.3.3 Cost allocation approach

Queensland Rail said that, for the 2015 DAU, in the changed-volume environment its primary concern was revenue certainty. It proposed allocating around 95 per cent of its forward-looking costs (i.e. forecast capital expenditure, maintenance cost and operating expenditure) to coal traffics based on coal's share of forecast volumes (effectively representing around 63 of 66 paths).³⁵⁴

However, Queensland Rail proposed capping coal's share of the opening asset base to around 78 per cent (representing 87 of 112 paths) to reflect government constraints on contracting capacity to coal.³⁵⁵ On its allocation of the opening asset base, Queensland Rail stated:

This approach is consistent with the overarching objectives established by the QCA ... of balancing Queensland Rail's right to recover its costs from users with mining customers' right to not be required to pay for capacity that they are not permitted to use.³⁵⁶

New Hope disagreed with Queensland Rail's approach and proposed allocating:

- fixed common network costs based on the higher of coal's forecast or contract paths as a proportion of system capacity (since the 2015 DAU forecast coal paths (63) are greater than contract paths (53), this would represent 63 of 112 paths); and
- variable costs based on coal's share of forecast usage.³⁵⁷

New Hope argued that allocating costs on the basis of available capacity rather than forecast usage will avoid coal services bearing the risk of declining demand from non-coal services.

Our 2015 Draft Decision considered that most below-rail infrastructure costs were common and fixed.³⁵⁸ These comprised the opening asset base and efficient forward looking fixed costs (that

³⁵⁴ The 2015 DAU proposed total forecast capital expenditure of \$141.9 million (coal allocation \$133.0 million), total maintenance cost allowance of \$143.0 million (coal allocation \$139.9 million) and total operating expenditure (excluding working capital) allowance of \$37.2 million (coal allocation \$34.9 million). The coal allocations reflected coal's share of forecast volumes (gtk-based) for maintenance costs, and coal's share of forecast paths for operating expenditure and forecast capital expenditure. Queensland Rail, sub. 2: 44–56.

³⁵⁵ Strictly speaking, Queensland Rail proposed to apply the 78 per cent allocator to assets in place since 1995. A lower allocator was proposed for the pre-1995 assets to reflect the impact of metropolitan passenger operations on the availability of West Moreton network paths, which is considered in Section 8.4.1 of this Decision.

³⁵⁶ Queensland Rail, sub. 2: 48.

³⁵⁷ New Hope, sub. 9: 24–27.

³⁵⁸ Common network costs comprise sunk costs (common network opening asset base) and forward-looking costs (forecast common network capital expenditure, maintenance cost and operating expenditure). Whereas opening asset base and forecast capital expenditure are considered as a fixed cost, operating and maintenance costs are categorised into fixed costs and variable costs. For example, about 57 per cent of maintenance costs and about 82 per cent of operating expenditure relate to activities that do not vary with usage (see Section 8.3.2 for the categorisation of maintenance costs and Section 8.12 for the categorisation of operating costs).

is, forecast capital expenditure and fixed maintenance and operating costs). Our Draft Decision proposed to:

- cap coal's share of efficient fixed common network costs to reflect the contracting restriction for coal services in the West Moreton network—that is, the 87 paths constraint less the paths (10 paths at the time) contracted to coal services to operate within the Metropolitan network, resulting in coal's share of 77 of 112³⁵⁹ paths; and
- allocated efficient variable common network costs (i.e. variable maintenance and operating costs) based on coal's share of forecast usage.³⁶⁰

Stakeholders' comments on 2015 Draft Decision

Stakeholders accepted our approach of allocating variable costs based on coal's share of forecast usage.³⁶¹ However, stakeholders did not accept our allocation approach for the fixed common network costs.

Queensland Rail said that our Draft Decision allocation approach resulted in it being unable to recover the 'full cost of providing the declared service' and stated that the only legitimate and rational basis for allocating costs was on the basis of forecast volumes. Queensland Rail provided advice from consultants, PwC and Synergies, in support of its argument.³⁶²

Coal miners said that our proposed allocation approach made the remaining coal producers bear the fixed common network costs of capacity formerly contracted for the now closed Wilkie Creek mine, which they said was unsustainable. They reiterated that fixed common network costs should be allocated to reflect the higher of coal forecast or contract paths as a proportion of system capacity.³⁶³

QCA analysis and Decision

We support common network costs being allocated amongst the different classes of users in the West Moreton network. However, the point of contention has been the allocation approach—in particular, the allocation of efficient fixed common network costs.

In the face of falling demand, our 2015 Draft Decision had sought to balance the conflicting objectives of:

- Queensland Rail recovering its efficient common network costs relating to coal-carrying train services in the West Moreton network; and
- coal traffics not paying for network capacity they are unable to contract to use.

Nevertheless, the submissions on our Draft Decision show that the positions of Queensland Rail and other stakeholders are far apart.

Based on the considerations set out in this section and having regard to the relevant factors in section 138(2) of the QCA Act, our decision is that:

• coal's share of efficient fixed common network costs be capped to reflect the contracting restriction for coal services in the West Moreton network;

 ³⁵⁹ The 112 paths capacity is Queensland Rail's estimate, which is reviewed in Section 8.4 of this Decision.
 ³⁶⁰ QCA 2015: 143–146.

³⁶¹ Queensland Rail, sub. 26: 41; New Hope, sub. 22: 11–12.

³⁶² Queensland Rail, sub. 26: 12, 29-31, 42-43, Annexure 4 and sub. 33, Attachment 2. Queensland Rail's comments on re-categorisation of fixed costs are considered in Section 8.3.2 of this Decision.

³⁶³ New Hope, sub. 22: 4, 9, sub. 21: 4; Yancoal, sub. 27: 2.
- efficient variable common network costs be allocated based on coal's share of forecast usage in the West Moreton network; and
- Queensland Rail be entitled to recover these coal-allocated costs from the forecast coal train services.

Cost allocation approaches

The contentious issue is about who should bear the efficient fixed common network costs of spare capacity, which comprise the opening asset base and forward-looking fixed costs (i.e. forecast capital expenditure, and fixed maintenance and operating costs). Therefore, unless otherwise specified, our consideration will primarily focus on the allocation of efficient fixed costs of the common network.

Queensland Rail and coal miners rejected our 2015 Draft Decision allocation approach to cap coal's share of efficient fixed common network costs to reflect the contracting restriction for coal services in the West Moreton network (at the time, 77 of 112 paths).

In response to our Draft Decision, Queensland Rail claimed that 87 paths was not a legally binding constraint and proposed re-categorising fixed costs into common fixed and coal fixed costs with different coal allocators for the two fixed cost categories.³⁶⁴ Nevertheless, Queensland Rail said that the allocation approach proposed in its 2015 DAU was appropriate and it was prepared to accept and comply with its 2015 DAU approach.³⁶⁵ Therefore, we have assessed the approach Queensland Rail proposed in its 2015 DAU, although we note Queensland Rail's 2015 DAU approach to allocating the opening asset base to reflect the 87-path constraint is inconsistent with its changed position that there is no such constraint. Queensland Rail's criticisms of our Draft Decision allocation approach are discussed separately in this section.

Coal miners preferred allocating fixed common network costs based on the higher of coal's forecast or contract paths as a proportion of total available paths.³⁶⁶ The miners' criticisms of our Draft Decision allocation approach are considered separately in this section.

The coal allocators reflecting the three allocation approaches are set out in Table 15.

Allocation approaches	Opening asset base allocator	Forward-looking fixed costs allocator
Queensland Rail's approach	87/112 (78%)	63/66 (95%) ³⁶⁷
Miners' approach	63/112 (56%)	63/112 (56%)
2015 Draft Decision approach	77/112 (69%) [at the time of Draft Decision]	77/112 (69%) [at the time of Draft Decision]

³⁶⁴ See Sections 8.3.1 and 8.3.2 of this Decision for our consideration of the 87-path constraint and the categorisation of costs.

³⁶⁵ Queensland Rail, sub. 26: 29–30 and 39–43 and sub. 33: 5–6.

³⁶⁶ New Hope, sub. 22: 8–9; Yancoal, sub. 27: 2 and sub. 35: 2.

³⁶⁷ Queensland Rail proposed a different allocator of 87.5 per cent for the Jondaryan to Columboola section of the West Moreton network, based on coal's share of forecast paths in that section. However, for presentational purposes we have used coal's share of total forecast paths (i.e. 63/66 paths) that Queensland Rail applied in the Rosewood to Jondaryan section. This section accounts for around 79 per cent of the total forecast volume across the West Moreton network.

Assessment of the three allocation approaches

Determining an appropriate allocation of fixed common network costs in a mixed-traffic network with significant spare capacity is a complex matter and requires a balanced consideration. We have assessed the three allocation approaches having regard to the statutory assessment criteria in section 138(2) of the QCA Act.

We engaged Professor Flavio Menezes of the University of Queensland to advise us on the economic efficiency aspects of the three allocation approaches.³⁶⁸ Following Professor Menezes being appointed to the QCA Board, we engaged Professor Stephen King to conduct an independent review of Professor Menezes' report on allocation approaches.³⁶⁹

Queensland Rail's approach

Queensland Rail argued that all users benefited from its proposed 2015 DAU cost allocation approach, as they were better off sharing costs than covering the stand-alone costs attributable to their own demand.³⁷⁰ However, Queensland Rail's approach makes coal traffics pay common network costs for an option to use up to the total available West Moreton capacity, which they cannot exercise given the contracting restriction on the network. Such an allocation approach would adversely impact the demand by coal services and result in inefficient price discrimination, which would not promote the efficient use of the network and not be in the interests of access holders and access seekers of coal services.

In this context, Professor Menezes observed that under Queensland Rail's approach, miners would pay for an option that they cannot exercise given the 80-path³⁷¹ constraint and that 'it has the highest likely adverse impact on the demand for coal transport'.³⁷²

Queensland Rail said the only legitimate and rational basis for allocating forward-looking costs was on the basis of forecast usage and argued that coal services should bear a higher proportion of those costs, given that only three non-coal services were forecast to run on the West Moreton network.³⁷³

Thus, Queensland Rail's approach seeks to transfer the risk of bearing the forward-looking costs to coal traffics and allows it the best chance to recover those costs. In that regard, it is likely to advance Queensland Rail's legitimate business interests.

However, Queensland Rail's proposal makes coal traffics pay almost all the forward-looking costs as if the network was designed to serve only coal traffics, but in practice the operation of coal services is subject to the constraints of a mixed-traffic network configuration. Therefore, Queensland Rail's approach will not send appropriate signals to promote the efficient use of the network and its approach would negatively impact the interests of access holders and access seekers of coal services.

³⁶⁸ Menezes, F, 2016a.

³⁶⁹ See King, S, 2016: 10–12.

³⁷⁰ Queensland Rail, sub. 33: 10.

³⁷¹ In our Draft Decision, the 77 paths available for contracting by coal services in the West Moreton network were derived by subtracting from the 87-path constraint, the 10 paths (at the time) contracted to coal services that operated within the Metropolitan network. In its March 2016 submission, Queensland Rail informed us that the paths contracted to coal services that operated within the Metropolitan network had decreased from 10 to 7, which means 80 paths are now available for contracting by coal services in the West Moreton network. Our consideration of this change in the number of paths available for contracting by coal services is discussed below under the heading 'Change in circumstances' in this Section 8.3.3.

³⁷² Menezes, F, 2016a: 11–12.

³⁷³ Queensland Rail, sub. 26: 30; sub. 33: 13, 26.

Professor Menezes was of the view that it was not clear that allowing Queensland Rail to recover practically all its forward-looking common costs from coal traffic through the regulated tariff aided long-run efficiency and said:

This suggests that in the absence of coal traffic, QR [Queensland Rail] would not be able to provide non-coal services. This in turn implies that in a world where QR were not constrained to offer non-coal services, it would not do so as it could not recover its costs. In this world, QR would only provide coal services and the network configuration might have been different, possibly involving different, lower forward looking efficient fixed costs. Such a network might have led to different investment decisions by coal miners.³⁷⁴

Professor King agreed with Professor Menezes' views that there were economic limitations to Queensland Rail's approach to recovering common costs from users according to their capacity to contribute to those costs, and observed:

If there is currently excess capacity on the network, but this excess capacity is 'efficient' in the sense that there is likely to be increasing demand for rail services on the network in the future, then it may be economically efficient not to fully recover the common costs in the short term. Rather, it would be recognised that these costs were in part an investment today in the future use of the network and should be allocated over time as well as over current users.³⁷⁵

Queensland Rail argued that a material decrease in non-coal-traffic levels on the West Moreton Network will rightly result in a higher proportion of costs allocated to coal services.³⁷⁶ However, given the contracting restriction for coal services on the West Moreton network, Queensland Rail's approach makes coal's share of forward-looking costs subject to the non-coal-traffic volumes. Since non-coal volumes are influenced by market and seasonal factors, and by government policy, Queensland Rail's approach will introduce significant unpredictability in setting coal reference tariffs and create regulatory uncertainty. Therefore, Queensland Rail's approach is detrimental to the efficient use of the network and to the public interest.

Queensland Rail argued that its approach sought to align with the QCA's December 2009 and October 2014 Draft Decisions.³⁷⁷ However, Queensland Rail did not accept our train path allocation based on coal's share of total available paths and proposed a fundamental shift in cost allocation by making it subject to the unpredictable non-coal volumes (see Appendix E for a summary of previous cost allocation assessments).

Previous considerations of the West Moreton network coal reference tariff were undertaken in the context of available capacity being potentially insufficient to satisfy all requests for access rights. We consider that the material reduction in demand for West Moreton network train paths and the contracting restriction for coal services in the West Moreton network necessitates an appropriate approach to allocating common network costs. However, we do not consider Queensland Rail's approach is appropriate, having regard to the factors in section 138(2) of the QCA Act as discussed above.

In summary, our view is that Queensland Rail's approach may advance Queensland Rail's legitimate business interest (s. 138(2)(b)), to the extent it allows Queensland Rail the best chance to recover forward-looking costs.

However, Queensland Rail's approach:

does not promote the efficient use of the network (s. 138(2)(a));

³⁷⁴ Menezes, F, 2016a: 11.

³⁷⁵ King, 2016: 11.

³⁷⁶ Queensland Rail, sub. 33: 26.

³⁷⁷ Queensland Rail, sub. 26: 42, sub. 33: 5–6, 14. See also Appendix E in this Decision.

- is not in the interests of access seekers and access holders of coal services (s. 138(2)(e), (h));
- is not in the public interest (s. 138(2)(d)); and
- is not appropriate, having regard to the pricing principle in section 168A(b), as it results in inefficient price discrimination.

Miners' approach

Coal miners argued that allocating fixed common network costs based on the relative proportion of network capacity forecast for use by coal services (i.e. 63/112 paths) will avoid coal services bearing the risk of declining demand from non-coal services.³⁷⁸ We consider that this would be in the interests of access holders and access seekers of coal services (s. 138(2)(e), (h)).

However, under the miners' approach, coal traffics will not pay for the common network costs reflecting the spare capacity available to coal services to contract on the West Moreton network (i.e. 14 spare coal paths at the time³⁷⁹). Professor Menezes was of the view that the miners' approach produced a 'free option' for coal traffic to contract capacity up to the 80-path constraint and said:

This may not lead to efficient outcomes in the context of a regulated firm making investment decisions under demand uncertainty.³⁸⁰

Effectively, miners' approach would make Queensland Rail bear the risk of recovering common network costs of the spare capacity available for contracting by coal services, which would discourage efficient operation of, and investment in, the network by Queensland Rail for providing coal services.

Moreover, as long as forecast coal paths (63 paths in the 2015 DAU) are below the capacity available for contracting by coal services (77 paths at the time), Queensland Rail will be unable to recover, under the miners' approach, the common network costs of providing the declared service to coal traffics. In our view, the miners' approach will hinder efficient operation of and investment in the network (s. 138(2)(a)). It is also contrary to Queensland Rail's legitimate business interests (s. 138(2)(b)), and is not appropriate, having regard to the pricing principle in section 168A(a).

To the extent the miners' approach would discourage efficient operation of and investment in the network for the provision of coal services, it would create a misalignment of the network with the operation of the rest of the supply chain and that would not be in the public interest, having regard to section 138(2)(d) of the QCA Act.

In summary, our view is that the miners' approach is in the interests of access holders and access seekers of coal services (s. 138(2)(e), (h)).

However, the miners' approach:

- will hinder the efficient operation of and investment in the network (s. 138(2)(a));
- is contrary to Queensland Rail's legitimate business interests (s. 138(2)(b));
- is not in the public interest (s. 138(2)(d)); and

³⁷⁸ New Hope, sub. 9: 24–27; Yancoal, sub. 35: 2.

³⁷⁹ 14 spare coal paths based on 77 paths maximum capacity available for coal services to contract on the West Moreton network less 63 forecast coal paths, at the time of the 2015 Draft Decision.

³⁸⁰ Menezes, F, 2016a: 12.

• is not appropriate, having regard to the pricing principle in s. 168A(a).

2015 Draft Decision approach

Our Draft Decision proposed allocating fixed common network costs to reflect the share of West Moreton network capacity available to coal services to contract.

Our Draft Decision approach signals to coal train users that they will not pay for the fixed common network costs reflecting the capacity they are unable to contract to use. Therefore, unlike Queensland Rail's approach, coal traffics do not pay for an option they cannot exercise. Our Draft Decision approach will send appropriate signals for the efficient use of the network and is in the interests of access holders and access seekers of coal services.

Furthermore, unlike Queensland Rail's approach, under our Draft Decision approach, coal's share of fixed common network costs is unaffected by the variations in non-coal volumes. That would mitigate the unpredictability in setting coal reference tariffs, as non-coal volumes are influenced by market and seasonal factors, and government policy. The resulting regulatory certainty would promote efficient use of the network and the public interest, having regard to s. 138(2)(a) and (d) of the QCA Act.

Our Draft Decision approach makes coal traffics pay for an option up to the capacity they are able to contract to use, which avoids the free option issue associated with the miners' approach. Therefore, our approach will send appropriate signals for the efficient operation of, and investment in the network for providing coal services.

In comparison, Queensland Rail's approach makes coal services pay for more than what they can contract which is not efficient. Also, the miners' approach does not give any consideration to the spare capacity available for coal services to contract, which creates a free option and is also not efficient.

Professor Menezes said that 'the QCA draft decision approach can also be supported on efficiency grounds' and concluded that:

Looking at the allocation of common cost as pricing an option to use capacity beyond expected usage suggests that the 80/112³⁸¹ rule in the QCA draft decision approach may provide a superior approach to both the Miners' and QR's proposals.³⁸²

Professor King agreed with Professor Menezes' conclusion.³⁸³

Our Draft Decision approach allows Queensland Rail to recover, from the 63 forecast coal paths, its efficient costs and investments reflecting the capacity (77 paths at the time), including spare capacity that is available for contracting by coal services in the West Moreton network. Therefore, subject to the adjustment amount (see Section 8.18 of this Decision), our allocation approach will generate expected revenue from coal-carrying train services that should be at least enough to meet the efficient costs that reflect the capacity coal services are able to contract on the West Moreton network.

We also analysed whether expected revenue across all traffics under each of the three allocation approaches would be enough to meet the efficient costs of providing access for all traffics on the West Moreton network (i.e. the costs that reflect the capacity of the network to provide all those services). That analysis is presented in Appendix E of this Decision, which

³⁸¹ The number 112 refers to Queensland Rail's capacity estimate, which is reviewed in Section 8.4 of this Decision.

³⁸² Menezes, F, 2016a: 12.

³⁸³ King, 2016: 12.

shows that each of the three allocation approaches results in a shortfall in recovering the efficient costs of providing access for all traffics on the West Moreton network. This outcome is due to the substantial decline in non-coal volumes, which is a commercial matter for Queensland Rail.

In this context, we agree with Professor Menezes' views that the key features of the West Moreton network, where coal traffic pays the regulated tariff but is subject to the 80-path constraint, and non-coal traffic does not pay the regulated tariff and is subject to both market and non-market constraints, along with the existing spare capacity imply that:

QR [Queensland Rail] may not be able to recover common or even service-specific costs from non-coal services.³⁸⁴

Professor King said that:

The inability of QR to be able to 'fully recover' its efficient forward looking costs in the short term under a specific allocation method does not, as a matter of economics, mean that either:

- (a) The allocation method is inconsistent with economic efficiency; or
- (b) An allocation method that does 'fully recover' efficient forward looking costs in the short term is preferable.³⁸⁵

The QCA's approach provides for coal services to pay their share of the efficient costs of the network reflecting the proportion of capacity they are able to contract to use the service. Thus, under the QCA's approach, the expected shortfall in recovering the efficient costs of providing access for all traffics on the West Moreton network is due to the material decline in non-coal services. To the extent that Queensland Rail does not recover, from non-coal services, the efficient costs not allocated to coal services, that is a commercial matter for Queensland Rail.

Stakeholders' concerns

Stakeholders expressed a number of concerns about our Draft Decision allocation approach. These are considered in Table 16.

Stakeholders' concerns	QCA response
Queensland Rail said the Draft Decision approach prevented it from recovering the assessed efficient costs of providing the infrastructure and failed to set an access price that was consistent with section 168A(a) of the QCA Act. ³⁸⁶	Our allocation approach on its own establishes a reference tariff that entitles Queensland Rail to recover from coal-carrying train services the efficient costs that reflect the capacity coal services are able to contract on the West Moreton network. This is because our approach makes forecast coal services pay the efficient costs that reflect the capacity coal services are able to contract, which includes spare capacity available for coal services to contract.
	However, we acknowledge that the overall access charge for coal services, given the adjustment amount mechanism (see Section 8.18 of this Decision), will not generate expected revenue in the operative period of the undertaking that is at least enough to meet the efficient costs that reflect the capacity coal services are able to contract on the West Moreton network.

Table 16: Stakeholders' concerns about the QCA's 2015 Draft Decision approach

³⁸⁴ Menezes, F, 2016a: 2–3.

³⁸⁵ King, 2016: 11.

³⁸⁶ Queensland Rail, sub. 26: 12, 29–30, 42 and Annexure 4; sub. 33: 5, 8–9, 25.

Stakeholders' concerns	QCA response
	If efficient costs of providing access for all traffics on the West Moreton network were considered (i.e. costs that reflect the overall capacity of the network to provide all those services), then none of the three allocation approaches in front of the QCA provides for Queensland Rail to recover those efficient costs (see Appendix E of this Decision).
	In any event, our Draft Decision approach balanced the conflicting objectives of Queensland Rail recovering the efficient common network costs relating to coal-carrying train services in the West Moreton network and coal traffics not having to pay for network capacity they are unable to contract to use.
	We consider, for all the reasons given in this section, that our allocation approach is appropriate having regard to section 138(2) as a whole.
Queensland Rail said the QCA cannot prioritise the interests of users over the pricing principles. It said: 'Queensland Rail's receipt of at least its efficient costs and a return, while clearly in Queensland Rail's legitimate business interests, is also a fundamental pricing principle which cannot be traded off. ¹³⁸⁷	We disagree with Queensland Rail and consider that the pricing principles are one of a list of the statutory assessment criteria that we must have regard to in considering a DAU (see Chapter 10 of this Decision). Our Draft Decision approach sought to balance the conflicting interests of Queensland Rail and network users.
Queensland Rail said that the approach forces it to bear the costs not allocated to coal services and by doing so, the QCA would be requiring Queensland Rail to subsidise coal trains. It said: 'It does not reflect the requirements that must be met to ensure cross subsidies do not occur. ¹³⁸⁸	We disagree with Queensland Rail. Under our Draft Decision approach coal services would be required to pay the full coal-specific capital expenditure, almost all (98 per cent) the variable common network costs reflecting coal's share of forecast usage, and 69 per cent of fixed common network costs reflecting the network capacity coal traffics are able to contract to use. We agree with Professor Menezes' view that coal tariffs that cover the incremental costs of providing coal services are subsidy-free independently of whether or not Queensland Rail is able to recover the fraction of efficient common network costs not assigned to coal traffic. ³⁸⁹
	Indeed, as per the analysis presented in Appendix E, Queensland Rail's approach itself is unlikely to provide full recovery of the efficient costs of providing access for all traffics on the West Moreton network (i.e. costs that reflect the overall capacity of the network to provide all those services). ³⁹⁰ This outcome is due to the substantial decline in non-coal volumes, which is a commercial matter for Queensland Rail.

³⁸⁷ Queensland Rail, sub. 26: 31

³⁸⁸ Queensland Rail, sub. 26: 30, 42.

³⁸⁹ Menezes, 2016a: 14.

³⁹⁰ This result is largely because, under Queensland Rail's approach, capital charges associated with around 22 per cent of sunk costs, which reflects the share of opening asset base not allocated to coal services (see Table 15), are effectively left for recovery from the three forecast non-coal services, which represent 4.6 per cent share of the total forecast paths.

Stakeholders' concerns	QCA response
Queensland Rail said the QCA's approach forced	We disagree with Queensland Rail.
non-coal carrying services to bear a high proportion of the costs of providing the declared service as compared to the coal carrying services, which constituted differential treatment. It said: 'The QCA cannot require from Queensland Rail an access undertaking that permits a form of differential	We consider that coal services should not be made to share the common network costs that reflect the capacity they are unable to contract to use, and our approach does not result in inappropriate differential treatment.
treatment that is prohibited by section 138A(2). ^{'391}	Rather, we consider Queensland Rail's approach of making coal services pay for an option they cannot exercise will give rise to inefficient price discrimination.
Queensland Rail said that there was no justification	We disagree with Queensland Rail.
(economic or otherwise) for the QCA to require that common costs be allocated between coal and non- coal users on the basis of their potential share of maximum installed capacity. ³⁹²	Of the three approaches in front of the QCA, we consider that Queensland Rail's approach will make coal traffics pay for fixed common network costs of capacity they cannot contract to use and the miners' approach will provide them a free option to use the spare capacity they are able to contract to use. Overall, our Draft Decision approach of capping the fixed common network costs to reflect contracting restrictions for coal services in the West Moreton network strikes an appropriate balance and provides for coal services to pay for fixed common network costs reflecting the capacity they are able to contract to use.
Queensland Rail said the approach would prevent	We disagree with Queensland Rail.
efficient signals being given to it in relation to the future maintenance and renewal of the infrastructure that is essential for the ongoing provision of coal services, as it will not have a business case that anticipates full recovery of these future costs. ³⁹³	Under our Draft Decision approach, Queensland Rail is able to recover the future costs reflecting the capacity that is available for contracting by coal services, which avoids providing coal services a free option to use spare coal capacity. This would send appropriate signals for the efficient operation of, and investment in the network for providing coal services.
	Besides, as observed by Professor Menezes, our approach provides strong incentives for Queensland Rail to make additional paths available to coal services, once the 80-path capacity limit is reached. ³⁹⁴
	To the extent Queensland Rail is unable to recover efficient costs that are not allocated to coal services, that is a commercial matter for Queensland Rail. Our Draft Decision approach sought to balance the conflicting interests of Queensland Rail and network users.
Queensland Rail argued that the declaration under section 250 of the QCA Act and the undertaking covers all services on the West Moreton Network, notwithstanding the undertaking has proposed (and the QCA previously has agreed to) setting a reference tariff only for coal-carrying services. It	Queensland Rail's argument is for Queensland Rail to be able to recover its efficient common network costs of providing access for all services on the West Moreton network. If efficient costs reflecting the overall capacity of the

³⁹¹ Queensland Rail, sub. 26: 31.

³⁹² Queensland Rail, sub. 26: 41–42; sub. 33: 6–7, 25, Attachment 2: 4.

³⁹³ Queensland Rail, sub. 26: 42; sub. 33: 25.

³⁹⁴ Menezes, F, 2016a: 12.

Stakeholders' concerns	QCA response
said: 'The QCA cannot simply ignore the residual of costs that it proposes not to allocate to coal, and the way in which any cost allocation approach impacts on non-coal services'. ³⁹⁵	network to provide all services were considered, then none of the three options in front of the QCA provides for Queensland Rail to recover the efficient costs of providing access to the West Moreton network. This outcome is due to the substantial decline in non-coal volumes.
	Our Draft Decision approach balances the conflicting objectives of providing for Queensland Rail to recover its efficient common network costs relating to coal-carrying train services in the West Moreton network and coal traffics not having to pay for network capacity they are unable to contract to use.
	While Queensland Rail's application did not seek an assessment of cost recovery from non-coal train services, we consider that any anticipated shortfall in non-coal revenue is a commercial matter for Queensland Rail and that reference tariffs for coal services should not recover costs reflecting services coal customers are unable to contract.
Queensland Rail argued that the service provider	We disagree with Queensland Rail.
should not be penalised by a cost allocation approach which denies it from recovering costs that, absent a shared network use, it would be allowed to recoup. ³⁹⁶	Queensland Rail's proposal makes coal traffics pay almost all the forward looking costs as if the network was designed to serve only coal traffics, but in practice the operation of coal services is subject to the constraints of a mixed-traffic network.
	Our approach reflects the shared nature of the West Moreton network, in particular the contracting restriction for coal services.
Queensland Rail said that the West Moreton	We disagree with Queensland Rail.
network was appropriately sized for the forecast network demand, and stated that 'the QCA's own technical advisor has confirmed' that this was the case, noting that B&H's 2015 review did not exclude assets on the basis of surplus or excess network capacity. It said: 'There is no "smaller" asset configuration that could be adopted as the basis of an optimised network configuration and valuation.' ³⁹⁷	Queensland Rail has mischaracterised B&H's 2015 review, which observed that 'Queensland Rail now has many redundant assets but in the absence of closure, these assets continue to be inspected and maintained, presumably at minimal but safe levels In fact a deep review of this network at the forecast traffic levels could conclude that it contained many redundant assets and that an entirely different RAB is constructed and a new maintenance plan conceived'. ³⁹⁸
	That said, we have not sought to optimise the network in this Decision for the purposes of determining reference tariffs, given the uncertainty about future demand. ³⁹⁹
Queensland Rail argued that the 2015 Draft Decision	We disagree with Queensland Rail.
attributed its changed methodology to the fall in demand and the resultant increase in costs to coal services. However, when the QCA's 2014 Draft Decision on the 2013 DAU was released, a Draft	Our 2014 Draft Decision reflected a capacity constrained network based on Queensland Rail's 2013 DAU proposal.

³⁹⁵ Queensland Rail, sub. 33: 6.

³⁹⁶ Queensland Rail, sub. 33: 7.

³⁹⁷ Queensland Rail, sub. 33: 12.

³⁹⁸ B&H 2015: 4.

 ³⁹⁹ Both Queensland Rail and miners expect demand to rise in future. See Yancoal, sub. 27: 3; New Hope, sub.
 22: 9; Queensland Rail, sub. 2: 38 and Appendix 6: 9; sub. 33: 26.

Stakeholders' concerns	QCA response
Decision which had the same methodology as proposed in the 2015 DAU, the network was not at full capacity, and there were paths available for contracting to coal services. ⁴⁰⁰	Indeed, the drop in volumes was a reason cited by Queensland Rail to withdraw its 2013 DAU in order for it to submit a replacement DAU. The QCA never received a revised volume proposal from Queensland Rail, rather Queensland Rail had submitted that '[it] reserves the right to resubmit volumes for the AU1 period prior to the Authority's Final Decision'. ⁴⁰¹ The 2015 DAU is based on a revised (lower) volume
	forecast as compared to the 2013 DAU and our 2015 Draft Decision took that into account in proposing an appropriate cost allocation approach.
Queensland Rail claimed that its proposed maintenance and operating cost allowance reflected the costs that were necessary to provide forecast services (paths), and argued that it was therefore appropriate that the cost allocation approach allowed Queensland Rail to recover these costs from the customers using the service. ⁴⁰²	We disagree with Queensland Rail. Our technical consultant (B&H) observed in its September 2015 review that Queensland Rail's proposed costs were higher than would be required to support the forecast services, due to, for instance maintaining assets that would be considered redundant at the forecast traffic levels. ⁴⁰³ That said, for this Decision, we have not sought to optimise the asset base for the purposes of determining reference tariffs, given the uncertainty about future demand and the capacity limit for coal services on the West Moreton network. Also, Queensland Rail's proposed costs are disproportionately high, considering the material drop in demand, because there are fixed and
	drop in demand, because there are fixed and variable components. By nature, fixed costs are time dependent and not usage dependent, so an attempt to link fixed costs to forecast usage and allocating fixed costs on that basis is inappropriate. However, variable costs are affected by usage, and our approach to categorising costs into fixed and variable components, and allocating variable costs based on coal's share of forecast usage does just that.
Coal miners argued that requiring the remaining coal mines to pay tariffs based on the maximum possible paths available for coal, rather than what they contracted for, unfairly punished them for Wilkie Creek's closure and for investment decisions Queensland Rail had made in the past. ⁴⁰⁴	We disagree with the miners. If coal services did not pay the fixed costs reflecting the spare capacity they are able to contract, that would create a free option. That would not send efficient signals for the operation of, and investment in, the network by Queensland Rail for providing coal services, and would potentially adversely affect coal miners attempt to increase future coal volumes. ⁴⁰⁵ Our Draft Decision approach sought to balance the conflicting interests of all parties.
	Besides, miners' approach would not create an incentive for coal services to contract more paths up

⁴⁰⁰ Queensland Rail, sub. 33: 26.

⁴⁰⁵ We note Yancoal's above-rail haulage contract is now 1.7 mtpa—21 per cent more than its previous
 1.4 mtpa contract (Yancoal, sub. 35: 3–4)—and that coal miners are expecting volumes to rise in future.

⁴⁰¹ Queensland Rail (July 2014a): 20.

⁴⁰² Queensland Rail (14 April 2016): 3.

⁴⁰³ B&H 2015: 4–5.

⁴⁰⁴ Yancoal, sub. 27: 2 and sub. 35: 2; New Hope, sub. 22: 8–9 and sub. 31: 6.

QCA response
capacity available to coal services he free option, whereas our Draft proach would create incentives for coal ontract more paths which would mean due to the volume trigger (see Section Decision).
Decision). onsider the alternative approaches y the miners are appropriate and n the present circumstances. AU has been developed in light of a in demand for below-rail services on the on network, compared to historical vels. Insidered the merits of optimising the iven the deterioration in demand. the have chosen not to optimise the asset purposes of determining reference al services based on the 80-path which is based on the 87-path constraint lied in practice). Were it not for the the constraint, the circumstances for the reference tariff would have been d it would have been open for the QCA fferent conclusion, including optimising we have not sought to optimise the this decision to reflect, among other 0 per cent decline in the forecast non-coal services, given the uncertainty e demand ⁴⁰⁷ and considering that constraint means coal services would exposure to the risk of a decline in non- a. If the 80/87-path constraint was d demand for below-rail services low levels, we would consider the optimising the asset base for the determining reference tariffs. o defer capital charges would be when a capacity expansion, due to its esults in capacity that is more than o provide the least cost option for icipated demand. However, this is not curred here as the current surplus ue to a material drop in demand. ternative approaches proposed by d create a free option, as those effectively seek to defer Queensland ry of efficient common network costs the spare capacity available for coal ontract. Given demand uncertainty at
te d e ch

⁴⁰⁶ Yancoal, sub. 27: 2 and sub. 35: 2; New Hope, sub. 22: 9.

 ⁴⁰⁷ Indeed, both Queensland Rail and miners expect demand to rise in future. See Yancoal, sub. 27: 3; New Hope, sub. 22: 9; Queensland Rail, sub. 2: 38 and Appendix 6: 9; sub. 33: 26.

Stakeholders' concerns	QCA response
	investment in, the network by Queensland Rail, and will not be in the public interest.
Coal miners argued that the 77 paths available for contracting by coal services were equally available for contracting by non-coal services. Therefore, some proportion of the costs of unutilised train paths within the 77 paths should be allocated to non-coal services. 'Allocating all of the costs of the spare capacity up to 77 paths to coal traffics, in a mixed use system, is inequitable, will distort competition in markets, and is not appropriate having regard to s. 138(2)(a), (d), (e) of the QCA Act'. ⁴⁰⁸	We disagree with the miners. We consider our Draft Decision approach to make coal traffics pay the efficient fixed common network costs that reflect the proportion of capacity they are able to contract, and no more, balances the interests of all parties and is appropriate having regard to the assessment criteria in section 138(2) of the QCA Act.

Change in circumstances

Our 2015 Draft Decision proposed a coal allocator of 77/112 paths for allocating efficient fixed common network costs. There have since been changes in both of those numbers.

We have not accepted Queensland Rail's estimate of West Moreton network capacity of 112 weekly return train paths. Rather, we have accepted B&H's revised estimate of West Moreton network's capacity of 113 return paths per week (see Section 8.4 of this Decision). Therefore, the relevant capacity estimate is 113 paths that applies for the purposes of determining coal reference tariffs in this Decision.

At the time of our 2015 Draft Decision, 10 paths were contracted to coal services to operate in the Metropolitan network, which did not traverse the West Moreton network. Given the 87-path constraint, this resulted in 77 paths being available for contracting by coal services in the West Moreton network. In its March 2016 submission, Queensland Rail informed us that subsequent to the release of the Draft Decision, the number of paths contracted to coal services to operate within the Metropolitan network had decreased from ten to seven⁴⁰⁹, which means 80 paths are now available for contracting by coal services in the West Moreton network.

Therefore, under our Draft Decision approach, the coal allocator for fixed common network costs would now be 80/113 paths (71 per cent) as compared to 77/113 paths (68 per cent).

An allocation of fixed common network costs based on 80/113 paths (instead of 77/113 paths) would increase coal traffics' contribution to Queensland Rail's regulatory annual revenue requirement by less than one per cent, which we consider is immaterial.

Besides, 80 weekly coal paths is within reasonable bounds of the paths that were used by coal services on the West Moreton network in the recent past (for example, coal traffics used around 79 paths during 2011–12 and 74 paths during 2012–13).⁴¹⁰

⁴⁰⁸ New Hope, sub. 22: 8–9.

⁴⁰⁹ Queensland Rail, sub. 33: 7. Queensland Rail claimed confidentiality over the number of coal contracted paths within the Metropolitan network. The release of that information is governed by section 187 of the QCA Act. Accordingly, we sought information from relevant stakeholders on why that information should not be disclosed. We considered stakeholders' responses pursuant to the requirements in section 187, and decided to release in this Decision the number of coal contracted paths within the Metropolitan network.

⁴¹⁰ QCA calculation based on West Moreton coal export data for individual mines available from the Queensland government's data portal (accessed 20 May 2016).

Therefore, we consider 80/113 paths is an appropriate coal allocator for efficient fixed common network costs.

However, the 77 paths contracting limit applied for a part of 2015–16 and the 80-path limit applies to the remainder of 2015–16 and beyond. For 2015–16, if common network fixed costs were allocated based on the ratio of 80/113 paths, that would mean coal services would pay for additional three paths that were not available for contracting during a part of the year. Conversely, if those costs were allocated based on the ratio of 77/113 paths, that would mean giving coal services a free option to contract additional three paths for a part of the year.

Given these considerations, our view is to apply an average of 77/113 and 80/113 ratios for allocating efficient fixed common network costs for 2015–16 (that is, 69.5 per cent share) and apply the allocator 80/113 for 2016–17 and beyond until that allocator is reviewed as part of any DAAU or a DAU that Queensland Rail may submit.

We consider these coal allocators will:

- provide coal miners an incentive to contract more paths, which is in the interests of access holders and access seekers (s. 138(2)(e) and (h);⁴¹¹
- generate expected revenue from coal services that is at least enough to meet the efficient fixed common network costs that reflect the capacity available for contracting by coal services, which would advance Queensland Rail's legitimate business interests to the extent that the risk of spare coal contractible capacity is borne by coal traffics (s. 138(2)(b)); and
- promote economically efficient outcomes and be in the public interest (s. 138(2)(a) and (d)).

Our Decision

Given the above considerations, we consider the approach to allocating efficient fixed common network costs to reflect the share of West Moreton network capacity available to coal services to contract (i.e. a coal allocator of 80/113 paths) and allocating efficient variable common network costs based on coal's share of forecast usage:

- will promote the economically efficient use of, operation of, and investment in the network (s. 138(2)(a))—as it signals to coal train users that they will pay for efficient fixed common network costs that reflect the share of capacity they are able to contract and efficient variable common network costs reflecting their share of usage (and no more), and provides for Queensland Rail to recover its efficient costs and investments relating to the spare capacity that is available for contracting by coal services;
- is in the interests of access seekers and access holders of coal services and their customers (s. 138(2)(e) and (h))—as they are not required to pay for the fixed common network costs of network capacity that reflects the share of capacity they are unable to contract;
- is in the public interest (s. 138(2)(d))—as it promotes the future development of the aboverail market by signalling to customers that they will not have to pay for the fixed common network costs of network capacity that reflects the share of capacity they are unable to contract; and
- will advance Queensland Rail's legitimate business interests, to the extent that the risk of recovering efficient fixed common network costs of spare capacity, reflecting the capacity

⁴¹¹ We note that the 80 paths available for contracting by coal services is close to the usage of 79.6 paths miners expect towards the end of the 2015 DAU period (New Hope, sub. 19: 3).

available for contracting by coal services, is borne by coal traffics and not by Queensland Rail (s. 138(2)(b)).

The above considerations are in favour of our cost allocation approach.

Moreover, subject to the adjustment amount (see Section 8.18 of this Decision), our allocation approach establishes a reference tariff that will generate expected revenue from coal-carrying train services that should be at least enough to meet the efficient costs that reflect the capacity coal services are able to contract on the West Moreton network.

However, under our allocation approach (as well as under the approaches proposed by Queensland Rail and the miners), Queensland Rail may not be able to recover the efficient costs of providing access for all traffics on the West Moreton network (i.e. costs that reflect the overall capacity of the network to provide all those services). This is due to the risk that Queensland Rail may not be able to recover, from non-coal services, its efficient costs that are not allocated to coal services, and that is a commercial matter for Queensland Rail.

Nonetheless, our allocation approach is appropriate having regard to section 138(2) as a whole, including section 138(2)(a), (d), (e) and (h). Thus, having regard to all of these matters, our decision is that:

- (a) coal's share of efficient fixed common network costs be capped to reflect the relative proportion of the West Moreton network capacity available for coal services to contract—that is, the 87 path constraint less the seven paths contracted to coal services to operate within the Metropolitan network, resulting in coal's share of 80 of 113 paths;
- (b) efficient variable common network costs be allocated to reflect coal's share of forecast usage in the West Moreton network; and
- (c) coal train services' share of efficient common network costs be fully allocated to the 63 forecast coal traffics to derive the reference tariff for coal-carrying train services for the purposes of this Decision.

We consider that a proposal to amend the reference tariffs arising due to a change to the 80/87 path capacity constraints and the 113 paths capacity estimate should occur by way of a DAAU.

We do not consider it is appropriate that a proposal to amend the reference tariffs due to a change to these parameters be included in an application to vary reference tariffs as per clause 5.1 of Schedule D in Appendix F. That is because these parameters reflect the allocation approach we are approving in this Decision based on the circumstances at the time of this Decision, in particular, the circumstances of significant spare capacity with uncertainty about future demand outlook in the face of capacity limit for coal services on the West Moreton network.

Therefore, any change in circumstances that require an assessment of these parameters will require us to review our allocation approach, considering the circumstances prevailing at that time, which may, among other things, require us to consider optimising the network.

Summary 8.2

The West Moreton network reference tariff in the 2015 DAU must provide that efficient common network costs of the West Moreton network are apportioned to coal train services such that:

- (a) the allocation to coal services is based on the following network capacity constraints:
 - (i) the 87 paths per week constraint for coal train services through the Metropolitan network;
 - (ii) the 80 paths per week available for contracting by coal train services in the West Moreton network; and
 - (iii) a maximum capacity of the West Moreton network of 113 paths per week;
- (b) efficient fixed common network costs approved in this Decision are allocated to coal train services in the ratio of 80/113 paths;
- (c) efficient variable common network costs approved in this Decision are allocated to coal train services based on coal train services' share of forecast usage in the West Moreton network; and
- (d) Queensland Rail is entitled to recover these coal-allocated costs from the 63 forecast coal-carrying train services.

These network capacity constraints must be included within the 2015 DAU so that any amendment to the reference tariffs arising due to a change to a network capacity constraint can only be made by way of a DAAU, and not by way of an application to vary reference tariffs.

See definition of 'network capacity constraint', 'review event', clause (a) in definition of 'endorsed variation event', and Schedule D, clause 3.1(f)(i)–(iii) and clause 5.1 in Appendix F.

8.4 Capacity assessment

West Moreton network capacity is estimated by reference to the total return train paths available per week, where a 'return train path' refers to a loaded train trip from an origin (e.g. a mine) to a destination (e.g. a port) and an empty train trip back to the origin (the mine).

At the initial step, capacity is estimated as the number of return paths available for operations after considering the time required for trains to traverse the longest section, and allowing for planned maintenance time. This initial capacity estimate is usually reduced by a factor (a 'reduction factor') to account for unplanned infrastructure events in order to determine the actual effective infrastructure capacity.

In the 2015 DAU, Queensland Rail estimated the West Moreton network capacity to be in the order of 112 return paths, based on a reduction factor that took account of, among other

things, above-rail inefficiencies (e.g. train delays, reduced train performance and other unplanned above-rail incidents).⁴¹²

Our 2015 Draft Decision noted that our consultant, B&H, estimated the West Moreton network capacity to be in the order of 135 return paths⁴¹³, based on a reduction factor that did not consider above-rail inefficiencies. Although our Draft Decision used Queensland Rail's capacity estimate, the Draft Decision stated that our final views on the West Moreton network capacity were subject to stakeholders' comments.⁴¹⁴

Stakeholders' submissions

Queensland Rail said that its capacity estimate of 112 return paths was approved by the QCA in the calculation of the approved reference tariff in 2010, and the QCA should not move away from its established precedent.⁴¹⁵

Queensland Rail rejected B&H's capacity estimate and provided an alternative derivation of its 112 paths capacity estimate. Queensland Rail claimed that the reduction in available capacity due to above-rail factors was relevant and said that its capacity estimate included the impact of Metropolitan network operations on West Moreton network capacity, which was not reflected in B&H's estimate.⁴¹⁶

Miners said the QCA should use B&H's estimate unless there was compelling evidence that B&H's higher estimate of network capacity was flawed or incorrect.⁴¹⁷

QCA analysis and Decision

Our Decision is that the West Moreton network capacity is 113 weekly return paths based on a revised assessment done by B&H.

Although our capacity estimate is closer to Queensland Rail's capacity estimate, it reflects two opposite effects in comparison to Queensland Rail's estimate, as we consider:

- Above-rail inefficiencies should be reflected in an access charge and not in calculating a capacity estimate, which has the effect of increasing our capacity estimate as compared to Queensland Rail's estimate; and
- Queensland Rail understated the impact of Metropolitan network operations on West Moreton network capacity, which has the effect of reducing our capacity estimate as compared to Queensland Rail's estimate.

Queensland Rail's reduction factor-based capacity estimate

Queensland Rail's capacity estimate of 112 return paths was based on a reduction factor of 65 per cent.⁴¹⁸

Queensland Rail variously claimed that:

⁴¹² Queensland Rail, sub. 2, Appendix 6: 8 and 2015n.

⁴¹³ B&H 2015: 65–67.

⁴¹⁴ QCA 2015: 191.

⁴¹⁵ Queensland Rail, sub. 33: 31.

⁴¹⁶ Queensland Rail, sub. 26, Annexure 1: 4–11 and Annexure 9: 3–8.

⁴¹⁷ New Hope, sub. 22: 16; Yancoal, sub. 27: 2.

⁴¹⁸ The reduction in capacity is actually 35%, and the reduction factor of 65% is applied to the initial capacity estimate derived after making an allowance for planned maintenance.

The 65% West Moreton reduction factor, and resultant 112 return paths per week capacity was approved by the QCA in the calculation of the current reference tariffs⁴¹⁹

... Queensland Rail's adjustment factor of 65% (as previously approved by the QCA).⁴²⁰

However, Queensland Rail first provided information about the 65 per cent reduction factor in the 2013 DAU process, in its supplementary submission on the QCA's information request about its maintenance expenditure.⁴²¹

The QCA had approved a reference tariff for the West Moreton network in June 2010 that was derived on the basis of, among other things, a cost allocation methodology of which the 112 return paths capacity estimate was an input. However, the QCA's view was that the derivation of that tariff, including the cost allocation methodology, had not been resolved.⁴²² Moreover, at the time of the 2010 approval, the QCA did not have information which indicated that the 112 paths capacity was based on a 65 per cent reduction factor. In that sense, it is not accurate to say that the QCA previously approved a reduction factor of 65 per cent.

Queensland Rail did not substantiate its reduction factor estimate

Queensland Rail claimed that a number of factors reduced network capacity and influenced its estimate of the 65 per cent reduction factor, which included:

- the prevailing weather conditions;
- temporary speed restrictions;
- minor signal and trackside equipment faults;
- reduced locomotive and rollingstock performance;
- individual train dynamics and driving techniques; and
- unplanned above rail incidents.⁴²³

The QCA requested Queensland Rail to provide historical data (for example, the number of minutes or the percentage of pathways lost) on each of those six factors identified by Queensland Rail, as that information was relevant to our consideration of Queensland Rail's estimate of the reduction factor-based West Moreton network capacity.

However, Queensland Rail stated that:

[it] does not have information that identifies the percentage of pathways lost in these categories.

[its] current data recording codes do not record in many of the six categories sought by the QCA. As such, Queensland Rail has had to make assumptions in relation to existing categories to roll these into the categories that are being sought in order to provide information on minutes lost compared to plan. This will lead to inaccuracies.⁴²⁴

Effectively, Queensland Rail failed to substantiate its claim of a 65 per cent reduction factor.

The QCA's view

We consider Queensland Rail's capacity estimate based on a reduction factor that has not been substantiated is not transparent and would not promote economically efficient outcomes;

⁴¹⁹ Queensland Rail, sub. 33: 31.

⁴²⁰ Queensland Rail, sub. 33: 33.

⁴²¹ Queensland Rail 2013d: 5.

⁴²² QCA, 2010a: 89; QCA 2010b. See also Appendix E of this Decision.

⁴²³ Queensland Rail 2013d: 5 and 2015n.

⁴²⁴ Queensland Rail, 19 August 2015: 3.

hence it is not appropriate, having regard to section 138(2), paragraphs (a) and (h) of the QCA Act.

We have also considered that Queensland Rail has now provided an alternative derivation of its 112 paths capacity estimate, which is assessed separately in this section.

Given these considerations, we do not accept Queensland Rail's capacity estimate based on the 65 per cent reduction factor.

Queensland Rail's alternative derivation of capacity estimate

In its December 2015 submission, Queensland Rail provided an alternative derivation of its 112 paths capacity estimate, which had three key components:

- a revised time of 30 minute for trains to traverse the longest section in the West Moreton network, to allow for variability in trains and infrastructure performance;⁴²⁵
- a reserve path allowance to accommodate unplanned infrastructure events, including events due to above-rail inefficiencies; and
- impact of Metropolitan network operations on West Moreton network capacity.⁴²⁶

We engaged B&H to review its previous estimate in the light of the alternative derivation provided by Queensland Rail.

B&H accepted as reasonable Queensland Rail's revised 30-minute time for trains to traverse the longest section on the West Moreton network. However, B&H disagreed that above-rail inefficiency was relevant in estimating network capacity, and found that Queensland Rail understated the impact of Metropolitan network operations on West Moreton network capacity.⁴²⁷

The main differences between Queensland Rail's and B&H's capacity estimates are:

- whether above-rail inefficiencies should be taken into account in estimating network capacity; and
- the impact of Metropolitan network operations on West Moreton network capacity.

Above-rail inefficiencies

Queensland Rail said that if a train was running late due to a train operator's fault, it would effectively need to run on the next available path. Queensland Rail argued it was relevant to set aside an allowance for reserve paths to accommodate such circumstances, which reduced available capacity on the network.⁴²⁸

However, B&H argued that the capacity estimate should reflect the number of paths the infrastructure could provide, and not the number of paths an operator might require for its operations. B&H stated that 'above-rail inefficiencies are a matter for the contract between the above-rail operator and Queensland Rail'.⁴²⁹

⁴²⁵ Queensland Rail's previous capacity estimate was based on a 26-minute section running time (Queensland Rail, November 2013d: 5).

⁴²⁶ Queensland Rail, sub. 26, Annexure 1: 3–10 and Annexure 9: 3–7.

⁴²⁷ B&H 2016, Part 3.

⁴²⁸ Queensland Rail, sub. 26, Annexure 9: 6–7.

⁴²⁹ B&H 2015: 66 and 2016, Part 3: 2.

B&H used Queensland Rail's 'minutes lost compared to plan' data to identify an allowance for reserve paths due to below-rail factors. That assessment resulted in a reserve path allowance of 7.8 per cent of the initial capacity estimate, compared to Queensland Rail's 12.8 per cent estimate.⁴³⁰

The QCA's view

We consider that Queensland Rail's approach to reducing available network capacity to accommodate above-rail inefficiencies does not appropriately allocate the cost of above-rail inefficiencies to the party (train operator) that is best placed to manage those inefficiencies. Therefore, Queensland Rail's approach does not create a transparent pricing framework and will not send appropriate signals for the efficient use of the network; hence it is not appropriate, having regard to sections 138(2)(a) and (h), and 168A(d) of the QCA Act.

Relevantly, our amendments to the 2015 DAU includes provisions that protect Queensland Rail's business interests from the effect of above-rail inefficiencies. For instance, the amended DAU provides for an access charge to vary from a reference tariff to reflect cost and risk differences associated with differing train operating characteristics.⁴³¹ Additionally, Queensland Rail would be entitled to a take or pay charge if a train service is cancelled due to above-rail causes.⁴³² We consider that accounting for above-rail inefficiencies through such provisions in the DAU will promote pricing transparency and send appropriate signals for the efficient use of the infrastructure, having regard to sections 138(2)(a), (d), (e) and (h) of the QCA Act.

Therefore, our view is that above-rail inefficiencies should not be considered in estimating infrastructure capacity. Accordingly, we consider B&H's estimate is reasonable, as it does not consider above-rail inefficiencies.

Impact of Metropolitan network operations

West Moreton train services travel through the Metropolitan network to port and use portions of several of the Brisbane commuter rail lines. The West Moreton network capacity is affected by the Metropolitan network operations in two ways:

- Peak passenger periods limit access for trains crossing the Metropolitan network; and
- Maintenance activities outside the peak period on the Metropolitan network, which are not aligned in time with maintenance on the West Moreton network, also limit accessibility to the Metropolitan network.⁴³³

In the 2015 DAU, Queensland Rail estimated that the Metropolitan network reduced the West Moreton network capacity by 12.1 per cent.⁴³⁴ However, B&H estimated that impact was 17 per cent, which was adopted in our 2015 Draft Decision.⁴³⁵

Stakeholders' concerns

Queensland Rail argued that B&H did not consider relevant information provided by Queensland Rail, and B&H's assessment was based on incorrect assumptions. It submitted new information in support of its proposed 12.1 per cent Metropolitan impact.⁴³⁶

⁴³⁰ B&H 2016, Part 3: 3.

⁴³¹ See clause 3.3(b) in Part 3 (Pricing Principles) in Appendix F.

⁴³² See clauses 8.6, 8.7, 13.6(c) and Schedule 5, clause 5.4 in Appendix G.

⁴³³ B&H 2015: 57; Queensland Rail, sub. 2: 50.

⁴³⁴ Queensland Rail, sub. 2: 49–52.

⁴³⁵ B&H 2015: 64.

⁴³⁶ Queensland Rail, sub. 26: 32–33 and Annexure 1.

New Hope considered that 22 per cent was a more appropriate estimate and said that B&H did not consider the impact of 'the significant number of special events on weekends, which require additional passenger services'.⁴³⁷

B&H's assessment

We engaged B&H to review its previous estimate given stakeholders' comments, including to consider new information provided by Queensland Rail and New Hope.

B&H considered that information and concluded that Queensland Rail's proposal underestimated the impact of the Metropolitan network on the capacity of the West Moreton network. B&H stated:

This is because it [Queensland Rail] only calculates a narrow part of the overall picture of works scheduling and train operations. It only deals with large possession works. A calculation must also include "weeknight closures", rescheduled works, special events additional trains and the quantity of work conducted on both systems.⁴³⁸

B&H's view on Queensland Rail's estimate and New Hope's estimate was that:

[c]learly there is a range between the narrow calculation of Queensland Rail's at 12.1% that omits additional trains for special events and maintenance non-alignment, and higher values previously estimated by B&H and other stakeholders in the range of approximately 20%-22%. This was a conservative estimate prior to the detail provided by Queensland Rail that included train charts and maintenance alignment details.⁴³⁹

B&H concluded:

Even allowing for clarification of alignment of the large possession works we are of the view that 17% impact of the Metropolitan system on the capacity of the West Moreton system is still a reasonable estimate.⁴⁴⁰

The QCA's view

We agree with B&H's assessment that Queensland Rail understated the effect of Metropolitan operations on West Moreton network capacity and that New Hope overstated that effect.

B&H's estimate of the impact of Metropolitan operations on West Moreton network capacity reflects relevant factors including misalignment of maintenance time between the West Moreton network and the Metropolitan network and the effect of unplanned events on the Metropolitan network that reduce the potential capacity of the West Moreton network.

We therefore consider that B&H's assessment that Metropolitan operations reduce the West Moreton network capacity by about 17 per cent, which we note lies between Queensland Rail's 12.1 per cent and miners' 22 per cent estimates, is appropriate to adopt.

Since West Moreton train services travel through the Metropolitan network to the port, the West Moreton network capacity, as the number of return train paths, depends on how many train services can travel through the Metropolitan network to the port and back.

Therefore, it is relevant to consider the Metropolitan impact in determining the capacity of the West Moreton network. We agree with B&H that the Metropolitan impact should be applied

⁴³⁷ New Hope, sub. 22: 16.

⁴³⁸ B&H 2016, Part 3: 8.

⁴³⁹ B&H 2016: Part 3: 8.

⁴⁴⁰ B&H 2016, Part 3: 8.

after West Moreton network potential capacity has been assessed, based on constraints from within the West Moreton network.⁴⁴¹

The effect of the Metropolitan operations means that the actual effective capacity on West Moreton is lower than the potential capacity.

Our Decision

Given the above considerations, our view is that the West Moreton network capacity is 113 weekly return paths as per the derivation summarised in Table 17.

This capacity derivation is:

- transparent, unlike Queensland Rail's reduction factor-based estimate;
- based on relevant factors and does not reflect above-rail inefficiencies; and
- based on a reasonable estimate of the factors that influence capacity, having regard to the Metropolitan operational constraints on the West Moreton network capacity.

We therefore consider our capacity estimate will promote the efficient operation of, use of, and investment in, the network. It is in the interests of access seekers and access holders, as they will be able to better assess whether capacity is available to meet their access requirements. It would also allow an access seeker and Queensland Rail to assess the requirements for an expansion. For those reasons, we consider that the above capacity estimate is appropriate, having regard to all the assessment criteria in section 138(2) of the QCA Act.

Table 17 West Moreton network capacity estimate

Parameter	Result
Longest section train scheduling interval ¹	30 minutes
Total theoretical capacity per week (total weekly time (7 days x 24 hours x 60 minutes)/scheduling interval)	336 one-way paths/week
Less planned maintenance allowance of 19 hours per week ² (planned maintenance time (19 hours x 60 minutes)/scheduling interval)	(38 one-way paths/week)
Available capacity after planned maintenance	298 one-way paths/week 149 return paths/week
Less allowance for unplanned infrastructure events ³ (7.8 per cent of available capacity after planned maintenance)	(12 return paths/week) ⁴
Potential West Moreton network capacity	137 return paths/week
Less impact of Metropolitan network operations ³ (17 per cent of potential capacity)	(24 return paths/week) ⁴
West Moreton network actual effective capacity	113 return paths/week

1 Longest section is Rangeview to Spring Bluff. Average running time over that section is 23 minutes and the additional seven minutes provides for train control allowance (Queensland Rail, sub. 26, Annexure 1: 4 and Annexure 9: 4).

2 Queensland Rail, sub. 26, Annexure 1: 4.

3 B&H 2016, Part 3: 3, 8.

4 These return path numbers are rounded up to a greater loss due to the need to return a single path train (B&H 2016, Part 3: 3).

⁴⁴¹ B&H 2016, Part 3: 8.

Summary 8.3

The West Moreton network tariff derivation in the 2015 DAU must provide that the West Moreton network capacity is 113 return paths per week.

See Schedule D, clause 3.1(f)(iii) in Appendix F.

8.4.1 Metropolitan impact and cost allocation

In the 2015 DAU, Queensland Rail adjusted its coal train path allocation by its estimate of the Metropolitan impact, and applied the resulting 68.3 per cent share to allocate to coal traffics the value of common network assets that were in place before 1995.^{442,443}

Although Queensland Rail considered that such an adjustment for pre-1995 assets was not required, it said the approach reflected 'a pragmatic way of addressing the concerns of customers around the impact of the passenger dominated Metropolitan Network on the available capacity of the West Moreton Network'.⁴⁴⁴

Our 2015 Draft Decision proposed to accept Queensland Rail's adjustment for allocating pre-1995 assets, but proposed a coal train path allocation of 57.1 per cent that reflected in part our assessment of a greater impact of the Metropolitan network on West Moreton network capacity.⁴⁴⁵

Stakeholders' submissions

Queensland Rail reiterated that, although it considered the adjustment for Metropolitan impact in allocating pre-1995 assets to coal traffics was inappropriate and resulted in it not recovering its efficient costs, it was prepared to make that adjustment. Queensland Rail also stated that the QCA had no power to impose such an adjustment or an adjustment that was more adverse to Queensland Rail.⁴⁴⁶

New Hope said the train path allocator should be adjusted to reflect the reduction in West Moreton network capacity due to the Metropolitan network operations.⁴⁴⁷

QCA analysis and Decision

Our Decision accepts Queensland Rail's proposed approach to consider the Metropolitan impact in allocating the pre-1995 asset values; however, our coal path allocation reflects our decision to accept a 17 per cent Metropolitan impact.

As assessed in Section 8.4 of this Decision, the *potential* West Moreton network capacity is 137 return paths per week. However, about 17 per cent of this potential capacity is lost due to the

⁴⁴² The 68.3 per cent share represented an adjustment to the 77.7 per cent train path allocation (representing 87 of 112 paths) by 12.1 per cent Metropolitan impact—that is, 77.7% * (1 – 12.1%)).

⁴⁴³ The pre-1995 assets refer to assets that were in place before West Moreton coal traffic began in the mid-1990s.

⁴⁴⁴ Queensland Rail, sub. 2: 49–52.

 ⁴⁴⁵ QCA 2015: 191. The 57.1 per cent allocation of pre-1995 assets represented an adjustment to the 68.8 per cent train path allocation (representing 77 of 112 paths) by 17 per cent Metropolitan impact—that is, 68.8%
 * (1 – 17%).

⁴⁴⁶ Queensland Rail, sub. 26: 32–33.

⁴⁴⁷ New Hope, sub. 9: 26.

operations of the Metropolitan network, resulting in an *actual effective* West Moreton network capacity of 113 return paths.

Stakeholders' comments relate to the treatment of the metropolitan impact in allocating common network fixed costs to coal traffics for the purposes of deriving the coal reference tariff.

We consider that, for common network fixed costs that pertain to the period before coal traffics commenced on West Moreton network (i.e. pre-1995 assets), coal traffics should bear a pro rata share based on *potential* West Moreton network capacity. This approach provides that coal traffics would not pay for sunk costs that reflect West Moreton network capacity that was unavailable due to the Metropolitan impact when coal train services commenced in 1995.

In previous assessments, the pre-1995 assets coal path allocation was derived by adjusting the general coal path allocation by the Metropolitan impact. However, the West Moreton capacity derivation is now better understood (see Table 17 in Section 8.4 of this Decision). Therefore, we consider that the coal path allocation for pre-1995 assets should be the proportion of potential West Moreton network capacity available for contracting by coal services on the West Moreton network—that is, 58.4 per cent representing 80 of 137 paths.^{448,449}

We also consider that, for common network fixed costs that pertain to the period after coal train services commenced operations on West Moreton network, coal traffics should bear a pro rata share based on the *actual effective* West Moreton network capacity. This approach provides that coal traffics are allocated common network fixed costs relating to investment decisions that were made after coal train services commenced operations in 1995, in the knowledge of the Metropolitan impact. None of those investments changed the capacity of the West Moreton network in terms of the number of paths. Rather they were driven by the increased traffic on the network and undertaken to strengthen the network for all services. Therefore, they were on the common network and should be allocated as per the approach recommended in Section 8.3.3 of this Decision.⁴⁵⁰

We consider our approach is appropriate having regard to all the assessment criteria in section 138(2) of the QCA Act.

⁴⁴⁸ The 58.4 per cent share for pre-1995 assets is very close to the 58.8 per cent share that results from applying the previous method that had been used until now of adjusting the general coal path allocation (now 70.8 per cent representing 80 of 113 paths) by the Metropolitan impact (now 17 per cent) – that is, 70.8% * (1 - 17%).

⁴⁴⁹ As discussed in Section 8.3.3 of this Decision, the 77-path contracting limit applies for a part of 2015–16 and the 80-path contracting limit applies to the remainder of 2015–16. Therefore, for 2015–16, the allocation for pre-1995 assets is the average of 77/137 and 80/137 (that is, 57.3%).

⁴⁵⁰ See also Appendix E of this Decision that summarises our previous draft assessments.

Summary 8.4

The West Moreton network tariff derivation in the 2015 DAU must provide that:

- (a) the pre-1995 common network asset base is allocated to coal traffics based on the proportion of potential West Moreton network capacity available for contracting by coal services on the West Moreton network—that is, representing 80 of 137 paths.
- (b) The post-1995 common network asset base is allocated to coal traffics based on the proportion of actual effective West Moreton network capacity available for contracting by coal services on the West Moreton network—that is, representing 80 of 113 paths.

8.5 Form of regulation, take-or-pay and tariff structure

The form of regulation and pricing structure developed within a regulatory framework should, amongst other things, promote economic efficiency and the public interest, provide incentives for investment and allocate risks to the parties best able to manage them.

West Moreton and Metropolitan coal reference tariffs have operated using a price cap arrangement in combination with access agreements that include take-or-pay obligations.

A price cap form of regulation incentivises a monopoly infrastructure owner to provide access and achieve efficiencies, but exposes it to volume risk. A revenue cap reduces or eliminates the infrastructure provider's revenue risk, but can blunt or remove performance incentives.

An associated take-or-pay regime helps limit the revenue risk faced by the infrastructure provider, by providing that it will be paid for contracted capacity even if it is not used.

Queensland Rail's 2015 DAU proposed to continue the price cap form of regulation with takeor-pay obligations set at 80 per cent of the access charge payable.

Our Draft Decision also proposed a price cap form of regulation for Queensland Rail's West Moreton and Metropolitan networks. However, we proposed:

- requiring Queensland Rail to submit a 'volume trigger' endorsed variation event⁴⁵¹ to review reference tariffs when contracted volumes were greater than the forecast volumes used to develop reference tariffs;
- capping take-or-pay revenue from coal services at the total revenue allocated to coal services in assessing coal tariffs; and
- applying take-or-pay obligations at 100 per cent of access charges.⁴⁵²

Stakeholders' submissions

Queensland Rail said the QCA's proposed volume reset and take-or-pay regime restricted its ability to receive revenues from railing above its forecasts, while exposing Queensland Rail to

⁴⁵¹ 'Endorsed variation events' are defined in the undertaking. They set out specific circumstances including a change in law, in which Queensland Rail is required to apply to amend a reference tariff. Our proposal adds an additional endorsed variation event for changes in contracted volumes.

⁴⁵² QCA 2015: 196–201.

downside risk, including from force majeure events.⁴⁵³ Other stakeholders favoured retaining a price cap, applying a volume reset and capping take-or-pay, but said the take-or-pay percentage should reflect that some operating and maintenance costs could be avoided.⁴⁵⁴

QCA analysis and Decision

The QCA requires Queensland Rail to apply a price cap form of regulation, with a tariff reset where access holders contract above forecast volumes, and take-or-pay at 100 per cent of access charges.

Much of our decision is consistent with our October 2015 Draft Decision.

That said, we have reviewed our form of regulation and take-or-pay framework and made changes where necessary in light of a reconsideration of Queensland Rail's 2015 DAU and stakeholder submissions.

Given the complexity of the issues, this analysis:

- considers the form of regulation and tariff reset;
- considers the take-or-pay regime, including capping and force majeure; and
- sums up the balance of risks and incentives from the QCA's overall approach.

8.5.1 Form of regulation and tariff reset

A price cap gives Queensland Rail a strong incentive to provide capacity, either through contracted or ad hoc services. Given the West Moreton network has substantial spare capacity, we accept the proposal in Queensland Rail's 2015 DAU to apply a price cap.

However, one of the persistent issues with a price cap is the incentive for the regulated infrastructure provider to under-forecast demand, in order to increase the price.

Queensland Rail's tariff proposal provides a forecast by using both contracted services and expected ad hoc services to derive its volume forecasts (see Section 8.10 of this chapter).

Queensland Rail's proposal has merit to the extent that while its revenue can increase if actual volumes end up higher than its forecast, it is also exposed to a reduction in demand for ad hoc services, to which take-or-pay protection does not apply. However, if access holders contract for volumes higher than Queensland Rail's forecast, then Queensland Rail's downside risk will be removed, while the tariff remains at a level based on lower volumes.

An unchanged price when contract volumes were higher than Queensland Rail's forecast volumes would be against the interests of access holders and access seekers (s. 138(2)(e), (h)). Such a price would also be against the public interest, as it would have the potential to reduce demand for coal haulage and diminish competition in related markets (s. 138(2)(a), (d)).

Our October 2015 Draft Decision proposed to address this issue by providing for an endorsed variation event to recalculate the West Moreton and Metropolitan network tariffs where contracted volumes for any origin or in aggregate rose above the volumes used to calculate the tariff.⁴⁵⁵

⁴⁵³ Queensland Rail, sub. 26: 50–54.

⁴⁵⁴ New Hope, sub. 22: 17–19; sub. 31: 14–17; Yancoal, sub. 27: 2, Aurizon, sub. 20: 24.

⁴⁵⁵ QCA 2015: 196–198.

The volume reset would apply up to the 80-path contracting limit for coal services⁴⁵⁶ (see Sections 8.3.3 of this Decision). Since coal services cannot contract beyond the path limit, any paths used beyond 80 per week would be provided on an ad hoc basis, so would not trigger the contract volume reset. Therefore, the lowest the price could go under the contract volume reset would be the point where 80 paths of contracted capacity were paying for the 80 paths worth of costs allocated to coal.⁴⁵⁷

Queensland Rail said in its December 2015 submission that the QCA's proposed volume reset approach mimicked a revenue cap in terms of restricting above-forecast revenues, while exposing Queensland Rail to the downside risk that came with a price cap.⁴⁵⁸

We disagree with Queensland Rail's position. The contract volume reset still leaves Queensland Rail able to increase its revenue through selling any uncontracted paths on an ad hoc basis. Queensland Rail's concerns about its downside exposure, even where it benefits from take-or-pay provisions, are considered below in the discussion of take-or-pay.

Aurizon said a price cap was essential to allow Queensland Rail to 'recover any revenue deficit' by providing ad hoc coal services.⁴⁵⁹ Aurizon said that, on the other hand:

[b]y increasing contract volumes above this amount [the forecast volumes] Access Holders would obtain a reduction in their reference tariff and are provided a strong incentive to contract for available capacity.⁴⁶⁰

New Hope favoured retaining a price cap and applying a volume reset.⁴⁶¹ It said:

In the event that volumes increase beyond 63 paths to up to 77 paths, then QR (in the absence of the Endorsed Variation Event) will be paid twice for the same fixed costs. NHC can understand this proposal to the extent that the additional services are ad-hoc, as QR faces reciprocal upside and downside risk, but considers that the double recovery of these costs is not appropriate where QR is protected from downside risk by ToP.⁴⁶²

We agree with Aurizon and New Hope. The contract-based volume trigger recognises that Queensland Rail has forecast more than contracted volumes, by including expected ad hoc services as well as contracted services in its projected tonnages. At the same time, the endorsed variation event trigger protects the interests of access holders, access seekers and their customers, if there is significant contracting for new services during the term of the undertaking that largely removes Queensland Rail's exposure to volume risk.

⁴⁵⁶ For the Draft Decision, the West Moreton coal contracting limit was 77 paths, derived by subtracting 10 paths contracted to coal services with an origin in the Metropolitan network, from the government-imposed limit of 87 paths through the Metropolitan network. Queensland Rail subsequently advised that 'the contract for coal paths [with an origin in the Metropolitan network] has been renegotiated resulting in a decrease in the number of coal train paths from ten to seven' (Queensland Rail, sub. 33: 7). This leaves 80 of the 87 coal paths through the Metropolitan Network available to contract on the West Moreton network. See Section 8.3.3 of this Decision.

⁴⁵⁷ This could change if the number of paths with an origin in the Metropolitan network changed, or the 87path contracting limit changed. But in those circumstances, Queensland Rail would have an incentive to submit a DAAU to amend the reference tariff.

⁴⁵⁸ Queensland Rail, sub. 26: 50–53.

⁴⁵⁹ Aurizon, sub. 20: 24.

⁴⁶⁰ Aurizon, sub. 20: 24.

⁴⁶¹ New Hope, sub. 22: 17-18; sub. 31: 15–16.

⁴⁶² New Hope, sub. 31: 16.

We also consider it necessary to have a volume reset mechanism because of Queensland Rail's incentive to under-forecast volumes. Indeed, there is some evidence that its forecasts are less than West Moreton actual volumes for periods where that information is now available.⁴⁶³

Therefore, while accepting Queensland Rail's volume forecasts would advance Queensland Rail's legitimate business interests, it would not promote efficient use of the rail network or be in the interests of access seekers and holders to accept those forecasts without the protection of the contract volume reset (s. 138(2)(a), (b), (e), (h)). The volume forecasts, including comparisons with actual volumes and our assessment of Queensland Rail's forecasts under the section 138(2) approval criteria, are discussed in more detail in Section 8.10 below.

We acknowledge Yancoal's argument that, while the proposed endorsed variation event would give access holders an incentive to contract for more capacity, it would also give Queensland Rail an incentive to frustrate and delay access negotiations in order to provide access on an ad hoc basis at the existing reference tariff.⁴⁶⁴ However, we note that the QCA Act requires Queensland Rail to negotiate in good faith (s. 100), and that the 100 per cent take-or-pay (discussed below) gives Queensland Rail an incentive to contract.

We also note that there is potential for Queensland Rail's maintenance and operating costs to rise if coal volumes are substantially above those it has forecast. We consider, having regard to all the approval criteria in the QCA Act, including Queensland Rail's legitimate business interests, the efficient use and operation of the network, and the pricing principles (ss. 138(2)(a), (b), (g) and 168A(a)), that it is appropriate to take this potential cost increase into account when applying the volume reset.

We have therefore amended the drafting proposed in the Draft Decision in relation to the volume reset. The required drafting now explicitly provides that the review of tariffs would have regard to the possible increase in maintenance and operating costs arising from increased utilisation of the West Moreton capacity (see Schedule D, clauses 5.4(a)(ii)(C) and 5.4(g), in Appendix F of this Decision).

Volume increases for individual origins

We agree with the comments of New Hope and Yancoal that the endorsed variation event trigger should clarify that a change in volumes applies where contracts exceed forecasts for any individual origin, rather than just for the overall volumes on the West Moreton network.⁴⁶⁵

Therefore, for the avoidance of doubt, we have amended the drafting from the Draft Decision to specify that the trigger will apply if the number of contracted coal-carrying train services for any single origin or in aggregate is greater than the forecasts used to develop the West Moreton and Metropolitan reference tariffs (see definition of 'Endorsed Variation Event' in Appendix F of this Decision).

We have also amended the drafting to:

 provide for the endorsed variation event to apply to any contract above forecast volumes that was agreed before the approval date (see Schedule D, cl. 5.1(a)(ii)(b) in Appendix F, and discussion in Section 8.10 of this Decision); and

⁴⁶³ See Section 8.10 of this Decision.

⁴⁶⁴ Yancoal, sub. 27: 2.

⁴⁶⁵ New Hope, sub. 22: 17–18; Yancoal, sub. 27: 3.

• require Queensland Rail to notify the QCA as soon as reasonably practicable after it has contracted coal-carrying train services on the West Moreton and Metropolitan networks (see cl. 5.4.1(c) in Appendix F).

Provisional contracting

We note Aurizon's proposal that the volume reset would work better if access seekers were able to contract on a provisional basis until access negotiations were completed.⁴⁶⁶ We are unclear how such a measure would operate in practice. However, it would not be appropriate for provisional contracting to have the effect of reducing the tariff, without giving Queensland Rail the long-term certainty of revenue that would be provided from a finalised access contract. We do not consider that such a proposal appropriately balances the interests of Queensland Rail and access seekers and holders. Therefore, we consider that finalised, rather than provisional, contracts should trigger the volume reset.

8.5.2 Take-or-pay

Take-or-pay encourages customers to contract for capacity they are most likely to need, and provides a degree of revenue certainty to Queensland Rail. Both of these objectives are served by applying take-or-pay at 100 per cent of access charges.

However, we consider that, in providing revenue certainty to Queensland Rail, it is not reasonable for the take-or-pay regime to enable Queensland Rail to recover more than the share of network costs used to assess the reference tariff.

As New Hope said:

The primary purpose of take or pay is to mitigate QR's downside volume risk ... Upside earned from the collection of take or pay, which requires no particular level of performance from QR, is not appropriate, and is likely to involve double payment for the same path, once through take or pay and again through another party using that path on an ad-hoc basis.⁴⁶⁷

Queensland Rail has proposed that its take-or-pay revenue not be capped—that is, it could recover more than the revenue used to assess its proposed reference tariffs, by collecting take-or-pay revenue for unused paths from one origin, while also receiving ad hoc revenue for selling the same paths for services from another origin.

Increasing revenues (as opposed to helping secure revenues) through take-or-pay may serve the legitimate business interest of Queensland Rail in receiving a degree of revenue certainty from take-or-pay provisions (s. 138(2)(b)). But it would not promote the efficient use of Queensland Rail's network, as it rewards Queensland Rail for providing paths it has already contracted to provide and been paid for, rather than for providing additional capacity (s. 138(2)(a)). Such a position is also not in the interest of access seekers and holders who would be required to pay for paths which can then be re-sold by Queensland Rail to generate revenue through ad hoc services, to the point where it is receiving extra revenue without providing extra capacity (s. 138(2)(e), (h)). Therefore, on balance, we do not consider Queensland Rail's proposal to be appropriate.

Accordingly, we require a limited take-or-pay capping mechanism to avoid over-recoveries of revenue. Queensland Rail will be able to recover sufficient take-or-pay to bring its West Moreton and Metropolitan network revenue up to coal's share of the network costs used to assess the coal reference tariff—the 'approved ceiling revenue limit'. But it will not be able to

⁴⁶⁶ Aurizon, sub. 20: 24.

⁴⁶⁷ New Hope, sub. 22: 18.

charge take-or-pay once its revenues have reached that level. The capping would include both take-or-pay and the revenue from payments such as relinquishment fees, as these are the outstanding future access/take-or-pay revenue paid to Queensland Rail when train service entitlements are surrendered.

The take-or-pay cap may not advance Queensland Rail's legitimate business interests, as it places some limits on Queensland Rail's ability to charge take-or-pay to increase its revenue. However, to the extent that Queensland Rail will be able to recover more than coal's share of network costs from coal services if it outperforms by delivering more than the total tonnes or paths that it has contracted to provide, we consider that it may advance Queensland Rail's legitimate business interests (s. 138(2)(b)). For the same reason, the take-or-pay cap promotes the efficient operation and use of the network (s. 138(2)(a)). It is also in the public interest and in the interests of access seekers and holders as it provides Queensland Rail with an incentive to provide capacity, without allowing for windfall gains (s. 138(2)(d), (e), (h)). Therefore, having regard to all the approval criteria in section 138(2), we consider it appropriate to require the take-or-pay cap.

The relevant revenue amount above which take-or-pay may not be levied (the approved ceiling revenue limit) is set out in the reference tariff table in Appendix A. This limit would be subject to review if tariffs were amended through the contract volume reset provisions discussed above (see Sch. D, cl. 5.4(a)(ii)(C) in Appendix F).

The approved ceiling revenue limit approach to preventing over-recovery through take-or-pay is consistent with our Draft Decision. However, Queensland Rail and other stakeholders have raised concerns about this approach.

Queensland Rail said 100 per cent take-or-pay increased its revenue protection, but that the 'downside exposure' remained for a number of reasons, including where take-or-pay was not payable during a force majeure event. It said this was not appropriate given the price cap model and would reduce its incentive to accept ad hoc services.⁴⁶⁸

Other stakeholders (Aurizon, New Hope, Yancoal) supported capping take-or-pay but said the take-or-pay percentage should reflect that some operating and maintenance costs could be avoided if a service did not operate.⁴⁶⁹ Yancoal said 100 per cent take-or-pay would result in windfall gains to Queensland Rail if trains did not run.⁴⁷⁰

We have discussed ad hoc services, force majeure and the take-or-pay and relinquishment fee percentages in turn below. Note that the treatment of interim take or pay notices is discussed in relation to the SAAs in Appendix C.

Ad hoc services

Queensland Rail has argued that a cap on take-or-pay revenue will work against the objective of making more paths available.

Queensland Rail's strong incentive would be to remove the option of ad hoc railings, and require all shippers to agree to take-or-pay terms for the full (potential) demand they may require from the West Moreton Network. This would protect Queensland Rail's revenue (to the level permitted

⁴⁶⁸ Queensland Rail, sub. 26: 52–54.

⁴⁶⁹ Aurizon, sub. 20: 24; New Hope, sub. 22: 18; sub. 31: 3, 16; Yancoal, sub. 27: 2.

⁴⁷⁰ Yancoal, sub. 27: 2.

by the QCA), but result also in there being less available capacity to support the entry of any new users into the system.⁴⁷¹

New Hope said:

We note that the QCA's capping mechanism achieves a similar outcome to that which would occur if the DAU includes a flexible short term transfer mechanism (which NHC would welcome in the future) ... NHC does not agree with QR's claim that the capping of ToP creates an inappropriate 'hybrid' price cap/revenue cap approach. QR's revenue under the Draft Decision is uncapped; only ToP collections are capped in certain circumstances.⁴⁷²

We agree with New Hope. Our mechanism does not remove Queensland Rail's ability to increase its revenue beyond the revenue used to calculate the reference tariff (i.e. the approved ceiling revenue limit) by operating ad hoc services. Rather, our proposed approach simply does not allow Queensland Rail to use take-or-pay revenue to exceed the approved ceiling revenue limit.

But while Queensland Rail would not be able to collect take-or-pay once revenues reached this limit, it could still sell all of its unused contract (and non-contract) paths to coal as ad hoc services, and keep that revenue. In other words, the incentive to provide capacity under the price cap would continue to operate, exactly as it is supposed to.

Moreover, given Queensland Rail would have sold those ad hoc paths and been paid for them, up to and beyond the approved ceiling revenue limit, its revenue downside would be removed for that year. It is not clear why Queensland Rail would remove the option of ad hoc railings in these circumstances, given that doing so would reduce its expected revenue.

Force majeure

Queensland Rail proposed in its 2015 DAU to exclude force majeure as a 'Queensland Rail Cause' for the network not being available. Our October 2015 Draft Decision proposed not to accept that change, so that force majeure would continue to be treated as a 'Queensland Rail Cause'. This meant, among other things, that take-or-pay would not be payable where Queensland Rail did not provide access after declaring force majeure.

Queensland Rail said that, as a result of this position, it would receive little downside protection from take-or-pay. It added that the QCA Draft Decision approach was not consistent with the force majeure treatment in central Queensland, both for Aurizon Network's below-rail services, and for access to DBCT's port. It said the solution was to change the standard access agreement to remove force majeure as a 'Queensland Rail Cause'.⁴⁷³

New Hope said customers should receive take-or-pay relief during force majeure events. It said that deciding whether Queensland Rail should be able to recover those amounts in the way Aurizon Network does in Central Queensland:

requires a consideration of the package of risks and benefits which QR receives under the proposed undertaking. We consider that the package as proposed under the Draft Decision is reasonable.⁴⁷⁴

We agree with Queensland Rail that the way force majeure is treated in central Queensland is different from the QCA Draft Decision approach, both for DBCT and Aurizon Network, each of which enjoys a degree of revenue protection even when force majeure applies.

⁴⁷¹ Queensland Rail, sub. 26: 53.

⁴⁷² New Hope, sub. 31: 17.

⁴⁷³ Queensland Rail, sub. 26: 52.

⁴⁷⁴ New Hope, sub. 31: 17.

By definition, most force majeure events are beyond the control or influence of either Queensland Rail or its customers.⁴⁷⁵ Yet these events, be they flooding or some other natural disaster, have a cost that must be borne by someone. Given that Queensland Rail's revenue upside is limited, it seems appropriate that its downside exposure to events beyond its control should also be limited.

However, we do not consider it appropriate to change the contractual risk balance that has existed in the standard access agreements in relation to force majeure and take-or-pay through more than 15 years of regulation (see Section 7.4 of this Decision). In this context, we have looked to the examples in central Queensland for ways to provide Queensland Rail with a degree of revenue protection when force majeure applies.

For DBCT, the revenue protection during force majeure events is through the take-or-pay provisions, as access holders remain liable for take-or-pay when force majeure has been declared.⁴⁷⁶ DBCT's situation is different from that of Aurizon Network and Queensland Rail because the terminal is operated by a company owned by the DBCT users, not by the infrastructure owner.⁴⁷⁷ This means, among other things, that the operator's (as opposed to the owner's) incentive to get the terminal up and running quickly will not be distorted by strong take-or-pay.

Aurizon Network has force majeure as an 'Aurizon Network Cause', so that take-or-pay is not payable when force majeure is declared. However, its revenue protection for force majeure is through an 'unders' and 'overs' mechanism in the revenue cap provisions in its access undertaking, that provides another way of recovering the forgone take-or-pay.⁴⁷⁸

Given that Queensland Rail is similar to Aurizon Network, in that it both owns and operates the below-rail infrastructure, we consider that the best model for addressing the consequences of force majeure is an approach similar to the one which applies for Aurizon Network.

We therefore propose to retain force majeure as a below-rail cause in Queensland Rail's SAAs, but address the cash flow effects of force majeure through an ex post revenue and tariff adjustment. This will be implemented as a review event, where Queensland Rail can apply for an adjustment that recovers 50 per cent of its forgone take-or-pay through future tariffs, if force majeure in any financial year reduces its revenue from an origin by more than 2.5 per cent of contracted annual revenue for that origin.

We have chosen 50 per cent after considering the comments from both Queensland Rail and the customers. Given that the event will have been beyond any of the parties' control, it is reasonable that the cost to Queensland Rail in terms of forgone revenue be shared equally. The 50-50 sharing is appropriate having regard to the legitimate business interest of Queensland Rail and the interests of access seekers and access holders (s. 138(2)(b), (e), (h)).

We note that, if the damage is substantial, Queensland Rail could seek capital underwriting and/or funding from access holders or their customers, for the cost of restoring the network (see Chapter 9 of this Decision).

We accept that our approach differs from that adopted for DBCT and Aurizon Network. But our approach for Queensland Rail reflects a different package of measures, both proposed to us and

⁴⁷⁵ While force majeure mainly covers events such as natural disasters, that are entirely beyond the parties' control, it also includes events, particularly strikes, where this is not the case.

⁴⁷⁶ DBCT Management 2010, Schedule C.

⁴⁷⁷ QCA 2016a, section 1.2.2.

⁴⁷⁸ QCA 2016b, Volume III—Pricing & tariffs, section 19.5.5.

considered by us. In particular, we consider that our proposed price cap framework, coupled with 50 per cent take-or-pay recovery for force majeure events, provides incentives for Queensland Rail to expedite any repairs that are necessary to restore the network.

We also note that the definition of force majeure covers a variety of events, not all of which relate to damage to the network that would result in trains not operating.⁴⁷⁹ We have therefore specified in the review event drafting that the take-or-pay recovery applies for a 'force majeure event which resulted in a part of the network being damaged or destroyed' (see definition of 'review event' in Appendix F).

Take-or-pay and relinquishment fee percentages

Queensland Rail's 2015 DAU proposed that West Moreton Network take-or-pay be levied at 80 per cent of access charges, which is the same as in the 2008 undertaking.

Our October 2015 Draft Decision proposed that this be increased to 100 per cent of access charges, to balance the take-or-pay capping by giving Queensland Rail more certainty about its revenue from contracted services.⁴⁸⁰

Aurizon, New Hope and Yancoal have argued that, as Queensland Rail can avoid costs when trains do not run, take-or-pay should be charged at less than 100 per cent of the associated access revenue.⁴⁸¹ Queensland Rail, on the other hand, said:

Take or pay is not a "windfall" to Queensland Rail. Queensland Rail plans for maintenance in each 12-month period, on the assumption that the Operator will operate 100% of contracted services. This is obviously necessary for safety and operational reasons. Maintenance costs are therefore fixed in each 12-month period.⁴⁸²

On Aurizon Network's central Queensland rail network, take or pay is levied at 100 per cent, but only for four out of five components of the access tariff. Take or pay does not apply to the 'AT1' tariff component, which is designed to represent variable maintenance costs that can be avoided if a train service does not run. The 'AT1' component represents less than 10 per cent of overall tariff revenue in central Queensland, so take or pay covers more than 90 per cent of the access charges.⁴⁸³

Given that the West Moreton and Metropolitan tariffs are two-part prices that are not suited to the Central Queensland approach (i.e. there is no West Moreton/Metropolitan tariff component that is intended to signal short-term variable costs alone), we consider that a different compromise is appropriate.

We agree with Queensland Rail that it has only limited ability to vary planned maintenance tasks to respond to temporary fluctuations in usage. However, if capacity is unused over an extended period of time, then it is reasonable to expect Queensland Rail to adjust its maintenance planning accordingly.

We therefore adopt the 100 per cent take-or-pay proposed in the Draft Decision, but require that the lower maintenance operating and costs for unused capacity be recognised in the relinquishment fees, by making them 80 per cent of the present value of remaining future take-or-pay obligations.

⁴⁷⁹ Queensland Rail, 2015 DAU, definition of 'Force Majeure Event'.

⁴⁸⁰ QCA 2015:200–201.

⁴⁸¹ Aurizon, sub. 20: 24; New Hope, sub. 22: 18; sub. 31: 3, 16; Yancoal, sub. 27: 2.

⁴⁸² Queensland Rail, sub. 33, Att. 5: 66.

⁴⁸³ QCA, 2016b, Volume III—Pricing & tariffs and Volume IV—Maximum Allowable Revenue.

This will have the benefits of:

- encouraging access holders to relinquish unused capacity quickly, making it available for access seekers to contract (in contrast, 100 per cent relinquishment fees give some incentive to hoard capacity, as the cost will be the same as 100 per cent take or pay regardless of whether the contracted capacity is handed back or not used); and
- giving Queensland Rail an incentive to re-contract those paths, to recover the higher take-orpay proportion.

Applying relinquishment fees at 80 per cent therefore promotes the efficient use of and investment in the rail network, and may advance the legitimate business interest of Queensland Rail (s. 138(2)(a), (b)). It is also in the interest of access seekers and access holders (s. 138(2)(e), (h).

8.5.3 Balanced proposal

We have developed a form of regulation and pricing structure, including a take-or-pay regime that, among other things, promotes economic efficiency and the public interest, and provides incentives for investment. In doing so, it allocates risks to the parties best able to manage them.

Queensland Rail has argued that the QCA's approach is like a revenue cap, in that it restricts above-forecast revenues, yet exposes Queensland Rail to downside risk.⁴⁸⁴ Queensland Rail said:

While Queensland Rail theoretically receives some increase in revenue protection due to take or pay, in actuality the benefit is negligible—given the gap between forecast demand and the level of contracted demand for which take or pay applies, Queensland Rail's "losses" from a reduction in ad hoc demand would far outweigh any incremental value offered by the increase in take-or-pay from 80% to 100%.⁴⁸⁵

It also said:

Where demand falls short of Queensland Rail's forecast, then the business' revenue will be reduced. However, where demand is higher, Queensland Rail will need to adjust Reference Tariffs and/or cap the application of take-or-pay. Given any probability of distribution around demand, this means that Queensland Rail's expected revenue must be less than that determined as efficient by the QCA.⁴⁸⁶

Other stakeholders favoured the QCA approach. New Hope said:

[T]he form of regulation and the ancillary arrangements proposed in the Draft Decision provide a reasonable balance and a fair allocation of risks.⁴⁸⁷

The evidence from Queensland Rail's revenues over recent years is that the combination of a price cap and take-or-pay has served Queensland Rail well. The data provided by Queensland Rail for the 2009–13 undertaking period shows that Queensland Rail's revenue was at or close to the revenue stream modelled by the QCA in three of those years. In the fourth year (2010–11), flood damage after extreme weather shut the range crossing for several months.⁴⁸⁸

Importantly, if the QCA's approved ceiling revenue limit approach had been applied during that period, it would not have restricted Queensland Rail's revenue in any of the four years. Further,

⁴⁸⁴ Queensland Rail, sub. 26: 36–37; 50–54.

⁴⁸⁵ Queensland Rail, sub. 33: 37.

⁴⁸⁶ Queensland Rail, sub. 26: 54.

⁴⁸⁷ New Hope, sub. 31: 15.

⁴⁸⁸ Queensland Rail, sub. 26: 51, Figure 1.

the take-or-pay approach we propose for force majeure events would have allowed Queensland Rail to recover half of the take-or-pay revenue forgone during the 2011 floods that closed the Toowoomba Range crossing, if it had been in place.

Queensland Rail is also wrong to assert that, given the probable range of demand outcomes, the QCA's volume reset and take-or-pay approach means its expected revenue will be less than that determined as efficient.⁴⁸⁹ The fact that the forecast volumes include some ad hoc paths leaves Queensland Rail exposed to some fluctuations in demand—both upwards and downwards. However, in the circumstances where the contract volume endorsed variation event is triggered, Queensland Rail will have contracted volumes for at least as much as its forecast volumes, and will therefore not depend on ad hoc services to achieve its forecasts. For example, if Queensland Rail contracted 10 more paths, and its contracted volumes rose to 63 paths, it would no longer be exposed to downwards fluctuations below 63 paths.⁴⁹⁰ So, by definition, Queensland Rail's exposure to downwards fluctuations will have been removed where the contract volume reset is applied.

We therefore consider that the combination of a contract volume reset with take-or-pay protection provides an effective balance of all parties' interests. Key features of the approach include:

- exposing Queensland Rail to volume risk to the extent that actual volumes are less than its forecasts, while providing Queensland Rail with an increasing share of take-or-pay protection when contracted volumes are greater than forecasts;
- providing Queensland Rail with an incentive to sell ad hoc paths above the 80-path contracting limit, as this will not reduce its price and will increase its revenue;
- enabling access holders to use capacity on an ad hoc basis above the 80-path limit, while paying a cost per path that represents an efficient share of fixed costs;
- providing 100 per cent take-or-pay to Queensland Rail to promote revenue certainty and drive access seekers and holders to contract only for the amount of capacity that they expect to use;
- protecting Queensland Rail by providing for 50 per cent of take or pay forgone during a force majeure event to be recovered from access holders if services are restored;
- providing incentives for users to contract capacity as, to the extent the contracted volumes are greater than those forecast to develop reference tariffs, the variation approach results in a lower approved reference tariff;
- mitigating incentives for access holders and access seekers to use ad hoc services in preference to contracting capacity;
- encouraging users to relinquish, rather than hoard, unused capacity by applying relinquishment fees at 80 per cent of the present value of remaining future take-or-pay obligations; and
- protecting Queensland Rail by adjusting the approved ceiling revenue limit, if necessary, to reflect amendments to the reference tariffs through the contract volume reset process.

⁴⁸⁹ Queensland Rail, sub. 26: 54.

⁴⁹⁰ Queensland Rail's forecast volume is actually 62.8 weekly paths, although we have used 63 paths for presentational purposes. So the contract volume reset would apply for 63 or more contracted paths.

The volume trigger may not advance Queensland Rail's legitimate business interest, as it will reduce the reference tariff if contracted volumes from an origin or in aggregate increase above Queensland Rail's forecasts. However, having regard to Queensland Rail's legitimate business interests, the trigger only applies to contracted capacity (i.e. it does not apply to ad hoc paths), and it provides an incentive for access seekers to contract capacity (s. 138(2)(b)).

At the same time, the trigger is in the interests of access seekers/holders, as they benefit when contracting access rights above forecast volumes by means of lower prices (s. 138(2)(e), (h)). It also promotes efficient use of capacity (s. 138(2)(a)). Therefore, having regard to all the approval criteria in section 138(2), we consider it appropriate to require the volume trigger.

The take-or-pay cap may not advance Queensland Rail's legitimate business interests, as it places some limits on Queensland Rail's ability to charge take-or-pay to increase its revenue. (s. 138(2)(b). However, to the extent that Queensland Rail will be able to recover more than coal's share of network costs from coal services if it outperforms by delivering more than the total tonnes or paths that it has contracted to provide we consider that it may advance Queensland Rail's legitimate business interests (s. 138(2)(b)). For the same reason, the take-or-pay cap promotes the efficient operation and use of the network (s. 138(2)(a)). It is also in the public interest and in the interests of access seekers and holders as it provides Queensland Rail with an incentive to provide capacity, without allowing for windfall gains (s. 138(2)(d), (e), (h)). Therefore, having regard to all the approval criteria in section 138(2), we consider it appropriate to require the take-or-pay cap.

The overall take-or-pay and volume trigger approach may not advance Queensland Rail's legitimate business interest as it places some limits on what Queensland Rail may charge access holders (s. 138(2)(b)). However, it supports efficient use of the network by encouraging access seekers and Queensland Rail to sign contracts (s. 138(2)(a)). It also advances the interests of access seekers and holders as they will benefit from lower prices when they contract for access rights above forecast volumes and from Queensland Rail's incentive to provide capacity without allowing for windfall gains (s. 138(2)(e), (h)). Therefore, having regard to all the approval criteria in section 138(2), we consider it appropriate to require that Queensland Rail's tariff regime include the take-or-pay and volume trigger mechanisms.

Summary 8.5

The form of regulation and take-or pay-approach in the 2015 DAU must provide as follows:

(a) An endorsed variation event be included for the reference tariff to be adjusted if contracted train services, the description of which accords with the reference train service, exceed the forecasts used to develop reference tariffs for any origin-destination pair or in aggregate, including having regard to possible increases in operating and maintenance costs.

See the definition of 'endorsed variation event' and Schedule D, clause 5.4(a)(ii)(C) in Appendix F.

(b) The endorsed variation event also apply to any contract for above-forecast volumes that is agreed before the approval date.

See Schedule D, clause 5.1(a)(ii)(b) in Appendix F.

(c) Queensland Rail is required to notify the QCA as soon as reasonably practicable after it has contracted coal train services on the West Moreton or Metropolitan networks.

See clause 5.4.1(c) in Appendix F.

(d) Queensland Rail will be able to recover take-or-pay sufficient to bring its West Moreton Network and Metropolitan network revenue from contracted train services, the description of which accords with the reference train service, (including access charges, take-or-pay and the revenue from payments such as relinquishment fees, but not including that portion of access charges that relate to a difference in cost or risk) up to the approved ceiling revenue limit (i.e. coal's share of the network costs used to assess the coal reference tariff), and be able to retain ad hoc revenues above that level.

See clause 3.2.2, the definition of 'approved ceiling revenue limit' and Schedule D, clause 4(c), (d)(vi), (d)(vii), (e), (f), (g) and (h) in Appendix F.

(e) Take-or-pay will be applied at 100 per cent for train services the description of which accords with the reference train service.

See Schedule D, clause 4(d) in Appendix F and Schedule 3, clause 4.1 in Appendix G.

(f) Force majeure will be retained as a Queensland Rail cause in the SAAs insofar as the event relates to Queensland Rail's network, but Queensland Rail will be able to recover 50 per cent of the forgone take-or-pay through a review event if track damaged or destroyed by a force majeure event is restored such that access holders can recommence their train services.

See the definition of 'review event' and 'Queensland Rail cause' and Schedule D, clause 5.4(a)(iii)(B) in Appendix F.

(g) Relinquishment fees will be set at 80 per cent of the present value of remaining future take-or-pay obligations.

See the definition of 'relinquishment fee' in Appendix G.

8.5.4 Tariff structure

The structure of West Moreton coal reference tariffs has varied over time. The first approved reference tariff was a single charge, levied based on weight and distance (i.e. per thousand gtk),
that applied from July 2005 to June 2009. Since July 2009, a two-part tariff has applied, with a charge per thousand gtk and an additional train-path tariff component.

Our Draft Decision proposed to accept Queensland Rail's proposal for a two-part tariff.

New Hope said in its December 2015 submission that recovering half of the tariff through a train path charge gave a 'strong discount, in terms of cost per gtk, to the Cameby Downs mine'. New Hope said it was not seeking to alter the tariff structure at this time, but it was important for the QCA to confirm that the proposed reference tariffs complied with the pricing limits in the undertaking, such that there was no cross-subsidy.⁴⁹¹

We note that a cross-subsidy requires that one access holder pay less than its incremental costs. We have assessed whether the Cameby Downs tariff covers the incremental cost of providing services from that origin, and confirm that this is the case. The assessment is discussed below in Section 8.17, on the QCA's required reference tariff.

As no stakeholder has opposed the two-part tariff, and we have assessed that it complies with the pricing limits, we have adopted the Draft Decision approach, and approved Queensland Rail's proposed tariff structure. We consider this appropriate having regard to all the criteria in section 138(2). The tariff structure is discussed in greater detail in Section 8.9.2 of our Draft Decision.

8.6 Metropolitan network tariff approach

Surat Basin coal trains travel through the Metropolitan network for more than one-quarter of their journey from mine to port. They follow a complicated route that uses portions of several of the Brisbane commuter rail lines, as well as some sections of track that are largely or entirely dedicated to freight traffic.

The 2009–13 tariffs approved for QR Network applied the West Moreton network tariff across the Metropolitan network—that is, the same tariff calculated for the sections west of Rosewood was applied to the track east from Rosewood to the port. This avoided the complicated and potentially inconclusive task of determining specific costs and asset values for the metropolitan section. The tariff included an allowance for incremental capital spending for freight services on the Metropolitan network, but there was no separate Metropolitan component of the AT2 train path charge.⁴⁹² The 2009 Draft Decision approach also had the effect of double-counting the asset return, as the west of Rosewood tariff applied as a proxy for the Metropolitan tariff included capital investments from the same period covered by the Metropolitan incremental capital spending.

Queensland Rail said in the material accompanying the 2015 DAU that it would continue the approach of applying a tariff developed for the West Moreton network to the Metropolitan network.

It said it would also recover 'incremental capacity expansion capex' on the Metropolitan network through a charge added to the AT2-path-based tariff component, adding:

While this results in a small increase in the total reference tariff, it also enables the payment of rebates to mining companies in relation to user funded capex within the Metropolitan Network.

⁴⁹¹ New Hope, sub. 22: 19.

⁴⁹² See QCA, 2009a: 92–93.

This is consistent with the approach recommended by the QCA in its Draft Decision on the 2013 DAU. $^{\rm 493}$

Our October 2015 Draft Decision proposed to accept Queensland Rail's overall approach of extending the tariff developed for the West Moreton network to apply for the Metropolitan network. We also proposed to accept the principle of having a Metropolitan incremental capacity charge as it provided incentives for efficient future investment.⁴⁹⁴

However, we proposed rejecting Queensland Rail's proposed treatment of incremental investment on the West Moreton network, where it was applied across the Metropolitan network. We said Queensland Rail's proposal to escalate the tariff at the approval date, and also charge access holders the Metropolitan incremental capacity charge for investments dating back to 2002, would result in double-counting of returns for the period between 2002 and the approval date.

Accordingly, the QCA proposed to set the metropolitan tariff so that it gave Queensland Rail an incentive to make efficient investments in the Metropolitan network by:

- (a) maintaining a RAB for investments made since 2002 and into the future to support coal and freight traffic in the metropolitan system (the incremental RAB) and using it to derive an annual revenue requirement; and
- (b) adjusting the West Moreton network asset base used for deriving the Metropolitan network tariff to exclude incremental capital investment from 2002 onwards, then increasing the resulting Metropolitan tariff annually by CPI.

Stakeholders' submissions

Queensland Rail said the QCA's proposal to use the West Moreton asset base only up to 2002 for deriving the Metropolitan tariff was a 'dramatic move away' from the QCA's past practice. Queensland Rail proposed to resolve double-counting issues by removing the Metropolitan incremental RAB.⁴⁹⁵ New Hope and Yancoal supported the QCA's Draft Decision approach.⁴⁹⁶ New Hope also said the QCA should be clear about what would happen for future undertakings.⁴⁹⁷

QCA analysis and Decision

The QCA requires Queensland Rail to extend the West Moreton network tariff so that it applies to coal services as they cross through the Metropolitan network, and to apply a separate Metropolitan incremental capacity charge to recover coal-specific investment and a share of relevant freight-specific investment on the network after 1 July 2013.

Much of our Decision on the Metropolitan tariff has adopted the positions in our October 2015 Draft Decision. That said, we have reviewed our Metropolitan network tariff framework and made changes where necessary in light of a reconsideration of Queensland Rail's 2015 DAU and stakeholder submissions. As such, this Decision also relies on and incorporates our analysis in section 8.9.3 of our Draft Decision, unless revised in this section.

Given the complexity of the issues, this analysis:

⁴⁹³ Queensland Rail, sub. 2: 5.

⁴⁹⁴ See explanation in QCA 2014b: 38–40.

⁴⁹⁵ Queensland Rail, sub. 26: 46–47; sub. 33: 38–39.

⁴⁹⁶ New Hope, sub. 22: 19; sub. 31: 18–19; Yancoal, sub. 35: 3.

⁴⁹⁷ New Hope, sub. 22: 19–20.

- considers the overall Metropolitan network tariff approach and sets out the QCA's required approach to extending the West Moreton tariff;
- considers incentives and the treatment of the incremental RAB and capacity charge; and
- provides some more general comments on the Metropolitan tariff approach.

8.6.1 Overall Metropolitan network tariff approach

Applying the West Moreton network tariff across the Metropolitan network through a proxy approach is a reasonable methodology for determining how much coal trains should contribute to the common costs of providing Metropolitan below-rail services.

The proxy approach means that the coal contribution is based on costs assessed on a network where the specific costs that apply to coal services are easier to identify. This promotes the efficient use of the network, and is in the interests of Queensland Rail, access holders and access seekers (s. 138(2)(a), (b), (e), (h)).

Queensland Rail proposed in its 2015 DAU to apply the same tariff derived for the West Moreton network for access to the Metropolitan network. It also proposed a separate train path charge to recover its investment in coal- and freight-specific infrastructure in the Metropolitan network since 2002.

As we said in our October 2015 Draft Decision, the proposal to recover Metropolitan capital expenditure since 2002, and also use a tariff extended from the West Moreton network that includes capital expenditure over the same period, leads to double-counting.

Queensland Rail's proposal does not promote efficient investment in the rail network as it provides for double-counting of the capital expenditure since 2002 on the West Moreton and Metropolitan networks (s. 138(2)(a)) For similar reasons, and because it would provide a return on investment that exceeded the regulatory and commercial risks of providing access, it is not in the public interest and does not advance the interests of access holders and access seekers (s. 138(2)(d), (e) and (h)).

So, while such a proposal may advance the legitimate business interest of Queensland Rail (s. 138(2)(b)), having regard to all the approval criteria, we do not consider it appropriate to approve Queensland Rail's proposal to both escalate the Metropolitan tariff that applies at the approval date, and apply a Metropolitan incremental capacity charge for investments dating back to 2002.

Queensland Rail, in responding to our Draft Decision, said the QCA's 2014 Consultation Paper and Draft Decision were in favour of 'effectively using the West Moreton tariff as a "proxy" for the Metropolitan tariff'.⁴⁹⁸ It said the QCA's 2015 Draft Decision approach meant 'the Metropolitan Network tariff will no longer bear any real resemblance to a verified efficient cost structure for providing West Moreton coal services'.⁴⁹⁹ Queensland Rail said:

The purpose of using the West Moreton Network as a proxy for the Metropolitan Network was to avoid the difficulties in determining a reference tariff that is based on a more accurate building block approach. This will result in swings and roundabouts in terms of individual elements. As such, it is not appropriate to cherry pick individual items in the way the QCA has sought to do, and which, for example, did not include renewal capital.⁵⁰⁰

⁴⁹⁸ Queensland Rail, sub. 33: 38.

⁴⁹⁹ Queensland Rail, sub. 26: 46.

⁵⁰⁰ Queensland Rail, sub. 26: 47.

Queensland Rail said the asset renewal costs on the West Moreton network since 2002 had been used to improve the asset quality on the network to allow the continued operation of coal services.⁵⁰¹ It said that the effect of the QCA's October 2015 Draft Decision proposal was to 'strip out' \$301.8 million of capital expenditure between 2002 and 2020 on the West Moreton network, to address \$21.7 million of Metropolitan capital expenditure over the same period.⁵⁰²

It proposed that, instead of removing the West Moreton assets since 2002 in deriving the Metropolitan network tariff, it would 'remove' the Metropolitan network incremental RAB and 'retain the other elements of the 2014 Draft Decision requirements'.⁵⁰³

New Hope said both the QCA's Draft Decision approach and Queensland Rail's revised approach had the potential to resolve the double-counting issue. It questioned Queensland Rail's statements that the QCA approach failed to recognise the work done to improve the standard of the West Moreton network since 2002. New Hope said it had 'seen no evidence of substantial reductions in QR's future maintenance costs arising from the claimed improvement in asset condition¹⁵⁰⁴.

New Hope said that if the QCA agreed with Queensland Rail that the 2015 Draft Decision proxy approach reflected the lower maintenance cost that came from renewing assets, while excluding the cost of renewing those assets, the solution was to include an 'appropriate proportion of the renewals cost within the proxy' costs from West Moreton that were applied for reference tariffs across the Metropolitan network.⁵⁰⁵

We are of the view that using forward-looking efficient costs and an appropriate allocation of sunk costs determined for the West Moreton Network as a proxy for forward-looking efficient costs of providing below rail services for coal traffic on the Metropolitan network is appropriate, having regard to the approval criteria in the QCA Act (s. 138(2), including paragraphs (a), (b), (e) and (h)).

The proxy methodology means that the coal contribution to common costs on the Metropolitan network is based on costs assessed on a network where the specific costs that apply to coal services are easier to identify and assess (i.e. the West Moreton network). This is because, among other things, the West Moreton costs reflect coal's share of fixed costs and a share of the wear and tear (i.e. variable costs) that the coal trains originating in the West Moreton network impose on rail infrastructure (see Sections 8.11 and 8.17).

Coal trains traverse portions of a variety of different lines within the Metropolitan network as they move through the network, so identifying and allocating the relevant costs in an efficient and representative way would involve a large number of subjective judgements, that would be unlikely to result in a transparent and appropriate outcome. As Queensland Rail said in the submission accompanying its June 2013 DAU:

Assessing a cost for coal-carrying train services for this section of track would be a sizeable task requiring a valuation, optimisation (in relation to track quality) and allocation (in relation to track type). ... Subsequent optimisation and allocation processes ... would be complex and difficult to carry out.⁵⁰⁶

⁵⁰¹ Queensland Rail, sub. 26: 46.

⁵⁰² Queensland Rail, sub. 33: 39.

⁵⁰³ Queensland Rail, sub. 26: 47.

⁵⁰⁴ New Hope, sub. 31: 18–19.

⁵⁰⁵ New Hope, sub. 31: 19.

⁵⁰⁶ Queensland Rail, 2013a: 7.

In contrast, the proxy approach means the coal trains travelling through the Metropolitan network are making a contribution to common costs as they are covering more than the incremental cost of using a network that carries similar traffic, over similar terrain, in a similar climate (i.e. the West Moreton network).⁵⁰⁷

The proxy approach may impact negatively on Queensland Rail's legitimate business interests as it involves a degree of estimation of efficient costs, so the amount that Queensland Rail is permitted to recover may be less than what those costs are (s. 138(2)(b)). Equally, the estimation may be contrary to the interests of access holders and access seekers if the amount Queensland Rail is permitted to recover is higher than those costs (s. 138(2)(e) and (h)).

However, we note that Queensland Rail continues to favour the proxy methodology, as do New Hope and Yancoal.⁵⁰⁸ We infer from this that the parties consider that, on balance, the proxy approach better serves their interests than the alternative approaches that might be used. Therefore, we consider that using the tariff derived from efficient West Moreton costs to apply when the trains that have an origin in the West Moreton network traverse the Metropolitan network is appropriate, having regard to all the criteria in section 138(2).

We have had regard to the 2015 DAU and various submissions, including those in response to our October 2015 Draft Decision, in considering the best approach to applying the West Moreton tariff as a 'proxy' for the Metropolitan tariff, while avoiding double-counting of capital expenditure across the two networks.

We note Queensland Rail's submission that the West Moreton tariff that is used as a proxy for Metropolitan costs (i.e. instead of separately deriving a tariff from costs assessed for the Metropolitan network) should bear a 'resemblance to a verified efficient cost structure'.⁵⁰⁹ And we consider that the West Moreton building blocks discussed elsewhere in this chapter, including the maintenance and asset value, achieve that goal (see Sections 8.11 and 8.17).

As such, the tariff derived from those building blocks also provides an appropriate representation of asset value and other costs in determining a contribution to common costs for West Moreton coal trains traversing the Metropolitan network.

Therefore, we require that Queensland Rail extend the West Moreton network tariff that would have applied in relation to the period from 1 July 2013 (see Section 8.20 of this Decision), and apply it for the Metropolitan network as well. This is consistent with our October 2014 Draft Decision.

8.6.2 Incentives and Metropolitan incremental capacity charge

We consider that, when using the West Moreton tariff as a proxy, it is also desirable to provide Queensland Rail with an incentive to make efficient investments in the Metropolitan network that support West Moreton coal services. However, we seek to avoid double-counting of the returns on those Metropolitan investments.

Our October 2015 Draft Decision proposed to achieve this by including the Metropolitan incremental capacity charge for assets dating back to 2002, and calculating the tariff that would be 'extended' across the Metropolitan network with an asset base that excluded West Moreton assets built since 2002.

⁵⁰⁷ See QCA 2009a: 92–93.

⁵⁰⁸ Queensland Rail, sub. 26: 47; New Hope, sub. 31: 18; Yancoal, sub. 35: 3.

⁵⁰⁹ Queensland Rail, sub. 26: 46.

Queensland Rail, in its December 2015 submission, has proposed an alternative way to avoid double-counting—by extending the West Moreton tariff without adjusting the asset base, but not applying the Metropolitan incremental capacity charge for investments dating back to 2002.⁵¹⁰

We acknowledge the proposal by Queensland Rail in its December 2015 submission achieves the objective of avoiding double-counting. However, we note that removing the Metropolitan incremental capacity charge entirely would also reduce Queensland Rail's incentive to make efficient investments for coal and freight services in the Metropolitan network, as opposed to the West Moreton network.

Omitting the incremental capacity charge would not promote efficient investment in the infrastructure required to provide below-rail access to coal trains as they cross the Metropolitan network or be in the public interest or the interests of access seekers and holders (s. 138(2)(a), (d), (e), (h)).

We require that Queensland Rail apply a Metropolitan incremental capacity charge to recover efficient investments made on the Metropolitan network to serve coal and freight traffic after 1 July 2013. This promotes efficient investment in the infrastructure required to provide below-rail access to coal trains as they cross the Metropolitan network (s.138(2)(a). The incremental capacity charge may advance the legitimate business interest of Queensland Rail as it receives a return on the necessary investment and it may be in the interest of access seekers and access holders because Queensland Rail has an incentive to make those efficient investments (s. 138(2)(b), (e), (h)).

We note that, while the metropolitan investments since 2002 will not explicitly be covered by a separate tariff component, they are covered by the extended West Moreton tariff, that includes incremental investments since 2002. We therefore consider that Queensland Rail will be receiving revenue from the Metropolitan tariff that should be used to pay rebates on assets covered by capital underwriting through access facilitation deeds.

Our Decision is that using a proxy is an appropriate methodology for providing Queensland Rail with its efficient costs of operating coal services on the Metropolitan network, and a return on investment consistent with the regulatory and commercial risks of providing access (ss. 138(2)(b) and (g) and 168A(a)). The proxy methodology means that the coal contribution to common costs on the Metropolitan network is based on costs assessed on a network where the specific costs that apply to coal services are easier to identify and assess. For the same reasons, the extension or proxy methodology is in the interests of access seekers, access holders and their customers (s. 138(2)(e) and (h)).

The Metropolitan incremental capacity charge from 1 July 2013 provides Queensland Rail with a return on investment commensurate with the regulatory and commercial risks of providing access and promotes the economically efficient investment in infrastructure by which services are provided (ss. 138(2)(a), (b) and (g), 69E and 168A(a)). It is also in the public interest and in the interest of access seekers and access holders, given it provides Queensland Rail with an incentive to make efficient investments in rail infrastructure (s. 138(2)(d), (e), (h)).

8.6.3 General observations

The Metropolitan tariff will apply for the term of the undertaking. However, we also seek to give some guidance on the QCA's approach to the Metropolitan tariff more generally. What is

⁵¹⁰ Queensland Rail, sub. 26: 47.

said below does not predetermine our decision on any future DAU. Any such DAU will need to be (and will be) considered in accordance with the requirements of the QCA Act at the time it is submitted. However, a relevant matter for the purposes of our Decision is its impact on certainty. We consider that the following comments provide appropriate general guidance in that regard.

New Hope said that the Draft Decision was unclear on whether the QCA proposed in future undertakings to escalate the Metropolitan tariff derived for the 2015 DAU regulatory period, or use a cost-based assessment. New Hope said it preferred for the QCA:

to re-calculate this portion of the Metropolitan revenue requirement at each regulatory reset, so that this portion reflects the capital charges (excluding post 2002 assets), operating costs and maintenance of the West Moreton network, updated based on the assessment of costs for the next undertaking period.⁵¹¹

We envisage that a proxy methodology will continue to be appropriate, having regard to section 138(2), for deriving the Metropolitan tariff. Further, we acknowledge the broad support for the proxy or extension methodology and we anticipate it will continue to apply. As Queensland Rail said in its March 2016 submission, a proxy 'reflects an efficient and reasonable approach, particularly given the challenges in separately building up the cost structure for the Metropolitan Network'.⁵¹²

We note that our required amendments to the undertaking include tariff reviews if the reference train configuration changes, the contracted volumes change substantially, or the proportion of paths available for coal to contract changes. The resulting reviews would all include consideration of any required changes to the Metropolitan tariff.

Summary 8.6

The Metropolitan network tariff approach in the 2015 DAU must provide as follows:

- (a) The West Moreton network tariff is extended so that it applies to coal services as they cross through the Metropolitan network.
- (b) A separate Metropolitan incremental capacity charge is applied to recover coal-specific investment and a share of relevant freight-specific investment on the network after 1 July 2013, but not before.

See Schedule D, clause 1.2 (deleted) in 2015 DAU and Schedule D, clauses 1.1 and 3.1(e) in Appendix F.

8.7 Productivity, innovation and incentives

The QCA Act provides that an access price should, among other things, 'provide incentives to reduce costs or otherwise improve productivity' (s. 168A(d)).

Our Draft Decision had regard to incentives to reduce costs and improve productivity. However, the discussion in the Draft Decision document focused more on incentives for efficient operation of, use of, and investment in, capacity.

⁵¹¹ New Hope, sub. 22: 19–20.

⁵¹² Queensland Rail, sub. 33: 38.

Stakeholders' submissions

Aurizon said the Draft Decision did not provide sufficient incentives for operational efficiencies, particularly where those efficiencies resulted in a reduced number of train services entitlements to deliver the same volumes.⁵¹³

QCA analysis and Decision

The QCA requires Queensland Rail to include a review event in the 2015 DAU to provide for a reference tariff to be varied if a productivity improvement changes the amount of capacity that is used for a given haulage task.

It is not the QCA's role to prescribe train configurations or any other productivity improvements. Rather, we seek to create a regulatory framework that enables access seekers and holders, their customers, and Queensland Rail, to work together to find ways to improve the operation of the rail network.

Aurizon said it was exploring alternatives to existing operations, including longer trains and different configurations, that might 'benefit all parts of the relevant supply chain, in particular through the freeing up of capacity within existing systems'.⁵¹⁴ However:

the Draft Decision has the effect of creating significant financial disincentives in terms of rail operators pursuing productivity improvements. This is because under the proposed access undertaking, Queensland Rail is not required to participate with rail operators in negotiating fair pricing and capacity outcomes for these improvements. Aurizon Operations believes this is a fundamental flaw in the framework adopted in the proposed access undertaking and strongly recommends that appropriate changes be made to ensure that improvements in the utilisation of capacity, and not just paths, [are] incentivised.⁵¹⁵

We accept that Queensland Rail's undertaking, including any reference tariffs, should provide a framework that encourages innovations that increase productivity.

We are aware that the two-part West Moreton reference tariff structure proposed in our October 2015 Draft Decision would reduce Queensland Rail's revenue for a given annual haulage task, if train configurations were changed to carry more coal per train path (e.g. longer trains).

We note that the reference tariff applies to the train services whose description accords with the reference train service, including characteristics such as length, weight and sectional running times (see Schedule D, cl. 3.1 in the 2015 DAU). Therefore, the DAU implicitly provides for the tariff paid by access holders to differ from the reference tariff to reflect changes in cost or risk arising from operating a train with different characteristics (see Section 3.3 of this Decision).

However, a broad 'variation for cost or risk' provision does not provide the certainty to Queensland Rail, access seekers, access holders and their customers that a reference tariff for the new configuration would offer. We therefore propose that the undertaking include a review event for an altered or additional reference tariff that would apply where an innovation was introduced—by either Queensland Rail or its customers—that improved productivity or efficiency.

⁵¹³ Aurizon, sub. 20: 32–34.

⁵¹⁴ Aurizon, sub. 20: 32.

⁵¹⁵ Aurizon, sub. 20: 33.

Such a review event would provide for efficient investment in and use of rail infrastructure, provide incentives to reduce costs, or otherwise improve productivity and serve the public interest in better operation of the supply chain and development of coal mines and other investments (ss. 138(2)(a), (d), (g) and 168A(d)). Running the network more efficiently and productively would also advance Queensland Rail's legitimate business interest, and would be in the interest of access seekers, access holders and their customers (s. 138(2)(b), (e), (h)). We therefore consider it appropriate to include a productivity review event having regard to all the criteria in section 138(2).

The QCA will consider any application under this review event provision on its merits, and does not seek to pre-judge the outcome of the QCA's future deliberations.

The best approach, consistent with the negotiate-arbitrate model, is for the new tariff structure to be worked out by Queensland Rail and its customers before it is submitted for approval.

We note that whether or not the tariff approach is agreed between the parties when submitted, a key objective of the QCA in assessing the proposal will likely be to give all parties an incentive to reduce costs or otherwise improve productivity, consistent with section 168A(d) of the QCA Act.

In doing so, we may consider whether any amendment to the undertaking created 'win–win' outcomes for both Queensland Rail and access seekers/holders, namely by having regard to matters including whether:

- the train operators and their customers were able to benefit from their proposed innovation through reduced overall unit costs of rail transport (i.e. including both above- and below-rail charges); and
- Queensland Rail's revenue from providing access was at least protected and preferably increased during the term of the undertaking in place at the time, and potentially beyond that period.

A 'win-win' approach would balance the interests of Queensland Rail and its customers, while providing for efficient operation of, use of and investment in the rail network. It would also benefit competition on the basis of innovation, which is one of the strongest drivers of productivity.

Summary 8.7

The 2015 DAU must provide for incentives to improve productivity by including a review event that provides for Queensland Rail to propose a variation to the reference train service and reference tariff (or propose a new reference train service or reference tariff) to accommodate productivity or efficiency improvements to Queensland Rail's below-rail services or access seekers' or access holders' above-rail services.

See definition of 'review event' in Appendix F.

8.8 Variation of reference tariffs

The 2008 undertaking provided for reference tariffs to be varied during the term of the undertaking for a variety of reasons, including changes to laws and changes to contracted services on particular track sections, through endorsed variation events and review events.⁵¹⁶

Queensland Rail's 2015 DAU included an endorsed variation event to provide for a change in law and review event provisions to provide for a material change in circumstances. It included a process for Queensland Rail to submit proposed changes to the QCA, and for the QCA to consider those changes. It specified that the QCA may approve the changes if it is satisfied that, among other things, they are consistent with the undertaking.⁵¹⁷

Our Draft Decision left the provisions for varying reference tariffs (in respect of endorsed variation events and review events) in the 2015 DAU largely unchanged, apart from amendments related to the contract volume reset and take or pay provisions (see section 8.9 of the Draft Decision).

QCA analysis and Decision

The QCA requires Queensland Rail to amend the process for varying reference tariffs in relation to review events to explicitly state that, in considering such proposed variations, the QCA will have regard to the approval criteria in section 138(2) of the QCA Act.

QCA considers this appropriate because the provisions relating to review events can potentially give rise to wide range of possible tariff adjustments. Given that such variations provisions are to be included in an undertaking which QCA can only approve if it considers it appropriate to do so having regard to the matters in section 138(2), it is appropriate that the power to approve such variations also be conditioned by reference to the matters in section 138(2).

Omission of a provision that permits the QCA to have regard to the matters in section 138(2) would risk setting an inappropriately narrow scope of matters that QCA could take into account in that process. It would have the potential to prejudice the efficient use of and investment in rail infrastructure; the interests of Queensland Rail, its access seekers and holders, and their customers; and the public interest.

The variation process for review events gives Queensland Rail an ability to seek variations that is not dissimilar from its ability to submit a voluntary DAAU under section 142 of the QCA Act. The QCA could only approve such a voluntary DAAU if it considered it appropriate to do so, having regard to the matters in section 138(2) (see s. 143). This further supports our view that the process for varying reference tariffs in relation to review events should provide for the QCA to have regard to the matters in section 138(2).

The 2015 DAU also contains a provision which permits Queensland Rail to seek a variation which will promote efficient investment in the coal supply chain in the West Moreton or Metropolitan network. The above considerations also apply to that provision, and accordingly we also consider it appropriate that the QCA have regard to the matters in section 138(2) when considering variations which are said to promote efficient investment in the coal supply chain in the West Moreton or Metropolitan network.

⁵¹⁶ See Schedule F, Part A, cl. 2.2 and definitions of 'Endorsed Variation Event' and 'Review Event' in the 2008 undertaking.

⁵¹⁷ See Schedule D, cl. 6 and definitions of 'Endorsed Variation Event' and 'Review Event' in the 2015 DAU.

We note that there are other provisions in our amended version of Queensland Rail's 2015 DAU which provide for amendments to reference tariffs to be made without requiring QCA to have regard to the matters in section 138(2)—namely, variations in relation to an endorsed variation event. We consider this difference in approach to be appropriate because of the more restricted nature of variations that can result from an endorsed variation event. That is, the amended 2015 DAU itself specifies a method for determining the tariff amendments to be made upon the happening of an endorsed variation event (which method we have already determined to be appropriate having regard to the matters in section 138(2)), whereas review events involve a much broader scope of possible outcomes.

Summary 8.8

The mechanism for varying reference tariffs in the 2015 DAU must provide that, in the process of deciding whether to approve a variation in a reference tariff through a review event, or an application to promote efficient investment under Schedule D, clause 5.1(a)(i), the QCA will have regard to the QCA Act, including the approval criteria in section 138(2).

See Schedule D, clauses 5.4(a)(i) and 5.4(a)(iii)(A) in Appendix F.

8.9 Capital expenditure assessment process (Schedule E)

The 2008 undertaking provided detailed criteria for the QCA to assess whether a capital expenditure project was prudent, and the project should therefore be included in the RAB.⁵¹⁸ Our 2014 Draft Decision proposed a similar capital expenditure assessment process.⁵¹⁹

Queensland Rail's 2015 DAU included a capital expenditure assessment process (Schedule E) that was in many respects consistent with the process we had set out in our 2014 Draft Decision.⁵²⁰ The DAU provided for Queensland Rail to submit capital expenditure projects to the QCA for acceptance into the RAB, and for the QCA to assess those projects based on whether their scope, standard and cost were prudent.

Our October 2015 Draft Decision proposed to accept most of Queensland Rail's proposal. However, we proposed requiring changes including that Queensland Rail consult with access holders affected by a capital project and maintain a separate RAB for the Jondaryan to Macalister section of the West Moreton network. We also proposed that Schedule E not prescribe a DORC asset valuation methodology, enable an access funder to seek a prudency assessment, and provide for the possibility of actual bypass as a basis for optimising assets.

Stakeholders' submissions

Queensland Rail said it was 'generally satisfied' with the capital expenditure assessment approach in the Draft Decision, but raised concerns about some proposed changes.⁵²¹ New Hope supported the QCA's proposed amendments to Schedule E.⁵²²

⁵¹⁸ Queensland Rail, 2008 undertaking, Schedule FB.

⁵¹⁹ QCA, 2014 October, 2013 DAU Draft Decision, pp. 126–128.

⁵²⁰ The 2014 Draft Decision capex process was based on Schedule A of the QR Network 2010 undertaking which, in turn, carried over substantial parts of Schedule FB from the 2008 undertaking.

⁵²¹ Queensland Rail, sub. 26: 25-27.

⁵²² New Hope, sub. 22: 16.

QCA analysis and Decision

We require Queensland Rail to amend its proposed capital expenditure assessment process to, among other things, require it to consult with access holders affected by a capital project, remove a requirement for using a DORC valuation methodology, and include the potential for actual bypass as a reason for optimising the asset base.

Many aspects of Queensland Rail's proposed prudency assessment process in Schedule E can be expected to promote the efficient investment in rail infrastructure, and are in the interests of Queensland Rail, and its access seekers and holders and their customers.

However, we consider that some aspects are not appropriate, having regard to the approval criteria in the QCA Act, namely:

- Queensland Rail's proposal to require that the asset value for determining a ceiling revenue limit be set solely on the basis of a DORC methodology is not in the interests of access seekers, access holders or their customers (s. 138(2)(e) and (h)). We require Queensland Rail to remove this requirement and to not prescribe any asset valuation methodology (see Section 3.5 of this Decision).
- The capital expenditure approval process does not enable an access funder to seek an assessment of the prudency of scope, standard and cost of a capital expenditure project. As a result, the process does not promote economically efficient investment in the significant infrastructure Queensland Rail uses to provide its service, and is not in the interests of access seekers, access holders or their customers (s. 138(2)(a), (e) and (h)). It is also likely that providing this ability to access funders will be in the interest of Queensland Rail, where it is seeking to secure a capital contribution as a condition of access (s. 138(2)(b)). We require Queensland Rail to amend the process to provide for an access funder to seek a prudency assessment (see Chapter 9 of this Decision).
- Queensland Rail's DAU does not provide for it to consult with the access holders that may be affected by a capital project. This does not promote economically efficient investment in the significant infrastructure Queensland Rail uses to provide its service, and is not in the interests of access seekers, access holders or their customers (s. 138(2)(a), (e) and (h)). We require that the 2015 DAU be amended to provide that affected users of capital projects are consulted, as the projects will affect the assessment of the reference tariffs the users pay (see Schedule E, cl. 3.2(b) in Appendix F).
- Queensland Rail has not provided for assets to be optimised if there is a possibility of actual bypass. We consider that Queensland Rail's proposal could therefore result in a duplication of facilities, which would not be in the public interest and would not promote the efficient use and operation of the declared service (s. 138(2)(a), (d)). We require Queensland Rail to amend its 2015 DAU so that the reasons the asset base may be optimised include the possibility of actual bypass.
- Queensland Rail has proposed maintaining separate RABs for the Rosewood to Jondaryan and the Jondaryan to Columboola sections of the network. We consider it will be in the interest of all parties to also maintain a separate RAB for the Jondaryan to Macalister section (s. 138(2)(b), (e) and (h)) (see Section 8.7.3 of October 2015 Draft Decision).

Much of our Decision adopts our October 2015 Draft Decision. As such, this Decision relies on and incorporates our analysis in Section 8.7.3 of our Draft Decision, unless revised in this section.

Queensland Rail said in its December 2015 submission that it was 'generally satisfied' with the QCA's approach in the Draft Decision, but it raised concerns about:

- a DORC valuation for additional sections of the network;
- the right of an access funder to seek acceptance; and
- optimisation for actual bypass.

These matters are discussed in turn below.

DORC valuation

Queensland Rail said it had made 'specific submissions' about the treatment of the RAB and the valuation methodologies that should be used when setting a ceiling for prices.⁵²³ Consistent with those submissions, it did not support the QCA's proposal to remove the DORC approach as the valuation method when 'additional, pre-existing parts of the network' are reviewed under the capital prudency process.⁵²⁴

We note that the pricing principles in the QCA Act provide that the price should 'generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved' (ss. 138(2)(g) and 168A(a)). The QCA Act does not specify how assets should be valued.

While Queensland Rail has enumerated a number of examples where the DORC methodology has been applied in the past for regulatory asset valuations in Australia,⁵²⁵ there are also many examples where regulators have applied valuation approaches suited to the circumstances of the assets being valued.

Given the varied configuration, standard, age and condition of Queensland Rail's network, we do not consider it appropriate to specify an asset valuation approach. Rather, the QCA should consider the appropriate approach, with regard to the approval criteria in section 138(2) of the QCA Act.

We have therefore adopted our Draft Decision, and require that Queensland Rail amend the 2015 DAU to remove the requirement that additional sections of the network incorporated into the West Moreton network are reviewed under the capital prudency process using a DORC methodology (see Schedule E, cl. 1.2(a)(ii) in Appendix F).

Providing flexibility about valuation methodology promotes efficient use of the rail infrastructure and is in the interests of access seekers and access holders, as a DORC valuation may not be appropriate in all circumstances (s. 138(2)(a), (d), (e), (h)). So, while Queensland Rail has favoured a DORC valuation requirement, indicating it considers this will advance its interest (s. 138(2)(b)), we do not consider it appropriate to include such a requirement, having regard to all the criteria in section 138(2).

⁵²³ See, for example, Queensland Rail, sub. 2, App. 2: 6, 9–11; sub. 17.

⁵²⁴ Queensland Rail, sub. 26: 25–26.

⁵²⁵ See, for example, Queensland Rail, sub. 2, App. 2: 6, 9–11; sub. 17.

Access funder

Queensland Rail said it 'agrees with the spirit of the QCA's proposal' to enable an access funder to seek a prudency assessment.⁵²⁶ However it said the drafting would need to reflect that:

- any costs from the assessment would not be passed on to Queensland Rail, including through the QCA fee; and
- any decision on prudency of user-funded capital expenditure did not negatively affect Queensland Rail or alter any amount it could recover from the funder.⁵²⁷

We note that our approach to user funding provides that there is no obligation for Queensland Rail to fund an extension, including the costs of any studies required in preparing for that extension. However, it would be open for the parties to negotiate an agreement that shared the costs and risks in a way that resulted in Queensland Rail bearing some cost.

Regardless of what is agreed between Queensland Rail and its customers, the scope, standard and cost of the asset would need to be assessed for prudency if it was to be included in the RAB (see Chapter 9 of this Decision).

Queensland Rail has indicated our proposal may advance its legitimate business interest, if we address its concerns about bearing any costs from a user-funded investment and altering the amount it may recover from the funder, as we have done (See Chapter 9 of this Decision). We consider that enabling an access funder to seek a prudency assessment would promote efficient investment in rail infrastructure and the public interest, and be in the interests of access seekers and access holders (s. 138(2)(a), (d), (e), (h)). So, we consider it is appropriate to provide for an access funder to seek a prudency assessment under Schedule E, having regard to all the approval criteria in section 138(2).

We have therefore adopted our Draft Decision position, and require Queensland Rail to amend the DAU to provide for an access funder to seek a prudency assessment.

Actual bypass

The 2008 undertaking provides that the QCA can optimise the RAB if:

- it accepted capital expenditure based on false or misleading information;
- demand has deteriorated so much that regulatory prices based on an unoptimised asset value would result in a further decline in demand; or
- there is a possibility of actual bypass.⁵²⁸

Queensland Rail's 2015 DAU provided for false information and demand deterioration to be grounds for optimising, but not the possibility of actual bypass.⁵²⁹ Our Draft Decision required that all three criteria be retained, including the possibility of actual bypass.⁵³⁰

Queensland Rail said in its December 2015 submission that it did not consider that the possibility of actual bypass was a reason for optimising the RAB value. Queensland Rail quoted the QCA's draft decision on Aurizon Network's 2014 DAU, which proposed 'to accept Aurizon Network's proposal to remove the threat of actual bypass as a reason for reducing the RAB'. It

⁵²⁶ Queensland Rail, sub. 26: 26.

⁵²⁷ Queensland Rail, sub. 26: 26.

⁵²⁸ Queensland Rail, 2008 undertaking, Schedule FB, cl. 1.4.

⁵²⁹ Queensland Rail, 2015 DAU, Schedule E, cl. 1.2(b).

⁵³⁰ QCA 2015: 188.

said that if there was a 'credible alternative supply option' the parties could negotiate a price within the floor/ceiling model. Further, Queensland Rail said that its proposed 'de-coupling' of the ceiling price and reference tariff would negate the need for the RAB to be adjusted.⁵³¹

We agree with the approach of letting market forces address the threat of actual bypass. However, Queensland Rail's situation is different from that of Aurizon Network. For example:

- Queensland Rail is more likely than Aurizon Network to face actual bypass, including through the Inland Rail project in southern Queensland, or through increased use of road transport on the Mount Isa route in the north;
- the two rail companies' networks are very different in standard, capacity and traffic mix; and
- Queensland Rail's networks are not fully contracted—demand has in fact been falling.

Therefore, each network needs to be assessed on its own characteristics and in light of the regulatory requirements, and it is not automatically appropriate to apply the Aurizon Network approach for Queensland Rail.

We have therefore adopted our Draft Decision position that the possibility of actual bypass should be retained as one of the reasons for optimising assets in the context of an annual rollforward of asset values in the RAB. Then, if the market fails to address the issue of actual bypass, the QCA has effective options to achieve a workable outcome.

This will enable efficient use of the network and protect the interests of access seekers and access holders, as well the public interest (s. 138(2)(a), (d), (e), (h)). While Queensland Rail has indicated that providing for possible actual bypass as grounds for optimising assets may be contrary to its legitimate business interests (s. 138(2)(b)), we consider that it is appropriate to include that criterion, having regard to all the criteria in section 138(2).

As discussed above, we do not consider that the 'de-coupling' approach proposed by Queensland Rail is appropriate (see Section 8.2.2).

Summary 8.9

The capital expenditure assessment process in the 2015 DAU must:

- (a) Remove the requirement that the asset value for additional sections of the network incorporated into West Moreton network be set solely on the basis of a depreciated optimised replacement cost methodology.
- (b) Enable an access funder to seek an assessment of the prudency of scope, standard and cost of a capital expenditure project.
- (c) Provide for consultation with access holders that may be adversely affected by a customer-initiated capital project
- (d) Enable optimisation of assets if there is a possibility of actual bypass.
- (e) Provide for Queensland Rail to maintain separate regulatory asset bases for the three sections of Rosewood to Jondaryan, Jondaryan to Macalister and Macalister to Columboola on the West Moreton network.

See Schedule E in Appendix F.

⁵³¹ Queensland Rail, sub. 26: 26–27.

PART B-TARIFF BUILDING BLOCKS

We have assessed the efficient costs and return on investment for Queensland Rail's West Moreton and Metropolitan network tariffs. Key issues are summarised in Table 18 below. Matters that require a more detailed explanation are discussed in Sections 8.10 to 8.17.

Table 18: Summar	y of key po	ositions and c	decision—tariff	building blocks
------------------	-------------	----------------	-----------------	-----------------

Summary of the 2015	Queensland Rail's	Other stakeholders'	QCA Decision
draft decision	position	position	
1. Volumes	1	1	1
Accepted Queensland Rail forecast of 62.8 weekly paths, including ad hoc paths.	Said it had requests for more contracted volumes.	Said Queensland Rail forecast was too low.	See Section 8.10 below
2. Forecast maintenanc	e costs		
Proposed maintenance costs of \$114.6 million, and identified 67.4 per cent was fixed cost and 32.6 per cent was variable cost.	Disagreed and submitted new information in support of its maintenance cost proposal.	Miners said the QCA's estimate was likely to be excessive. However, Aurizon was concerned with our reduction in resurfacing allowance.	See Section 8.11 below.
3. Forecast operating Co	osts		
Proposed to accept operating cost allowance based on Queensland Rail's 2012–13 below-rail financial statements with lower train control costs, and identified 82 per cent was fixed cost and 18 per cent was variable cost.	Accepted categorisation of operating costs into fixed and variable cost.	Miners said the proposed operating costs were not efficient and exceeded comparable benchmarks.	See Section 8.12 below.
4. Regulatory asset bas	e		
Recognised the value of assets that have reached the end of their useful lives through the cost allowance given for maintenance, or through the value given to the network as a whole.	nised the value of that have reached d of their useful nrough the cost unce given for enance, or through lue given to the		See Section 8.13 below.
5. Past capital expendit	ure		
Did not consider historical (TSC) capex for the purposes of coal reference tariff and found that the capex incurred during the two years 2013–15 was excessive.	Disagreed with our treatment of TSC capex.	Generally supported our draft decision.	See Section 8.14 below.

Summary of the 2015 Queensland Rail's draft decision position		Other stakeholders' position	QCA Decision		
6. Forecast capital expe	6. Forecast capital expenditure				
Proposed a capital indicator of \$144.2 million.	Disagreed and provided new information in supports of its forecast capex proposal.	Supported our Draft Decision.	See Section 8.15 below.		
7. Capital charges for the coal RAB					
Applied an indicativeSaid risk-free rates wereWACC of 6.93 per centhigher in previousand specified asset livesperiods.for the purposes ofstraight-linedepreciation.depreciation.		New Hope said an alternative depreciation profile might address issues with a tariff derived from building blocks.	See Section 8.16 below.		
8. QCA-required ceiling price					
Indicative draft ceiling price for West Moreton network coal services of \$18.88/'000 gtk, calculated as if it were to apply from 1 July 2015.			See Section 8.17 below.		

8.10 Volumes

Traffic forecasts are important, because any variation from forecast will result in a corresponding variation to revenues from access charges. Minimising volume risk in the first instance by using the best forecast information means the parties best placed to manage volume risk are able to do so.

Queensland Rail proposed to use forecast coal and non-coal volumes that it said were based on actual, rather than contracted use of the network in the period before it submitted the 2015 DAU. It said this was the best approach as both coal and non-coal services were operating above contracted levels.

In the October 2015 Draft Decision, we proposed to accept Queensland Rail's forecasts as, among other things:

- Queensland Rail had not proposed reference tariffs for non-coal services;
- there was significant uncertainty about coal and non-coal volumes on the West Moreton network; and
- the effects of volume uncertainty would be addressed, in part, by the endorsed variation event for changes in contracted volumes (see Section 8.5.1 of this Decision).

Stakeholders' submissions

Queensland Rail said that if it had known the QCA would change its risk allocation framework for the West Moreton tariff, 'a forecast based on contracted tonnes would be appropriate'.⁵³² It

⁵³² Queensland Rail, sub. 26: 54.

also said that while overall tonnages were down, 'Queensland Rail currently has access requests in excess of current capacity'.⁵³³

New Hope said it accepted the Draft Decision for non-coal volumes, but that the QCA needed to re-examine the coal volumes.⁵³⁴ Yancoal said the coal volume forecast did not allow for sustained ad hoc demand above forecast levels.⁵³⁵ It also said the QCA should take into account that Yancoal had increased its rail haulage contract from 1.4 to 1.7 million tonnes a year.⁵³⁶

QCA analysis and Decision

The QCA accepts Queensland Rail's proposed coal and non-coal volume forecasts for services operating on the West Moreton network.

Much of our Decision adopts our October 2015 Draft Decision. As such, this Decision relies on and incorporates our analysis in Sections 8.4.2, 8.4.3 and 8.4.4 of our Draft Decision, unless revised in this section.

Substantial uncertainty remains about volumes on the West Moreton network. We address stakeholders' comments on volumes and demand below, discussing both non-coal and coal volumes.

Non-coal volumes

Queensland Rail's 2015 DAU proposed weekly non-coal volumes of two return passenger paths and one ad hoc return freight path. It said all of the access agreements for non-coal services had expired since 2013 and there had been a 'dramatic move from rail to road transport for agricultural products'.⁵³⁷

We note that the Queensland Government in April 2016 said it was investing in infrastructure on the West Moreton network to increase the number of cattle train services.⁵³⁸ Notably too, the 2015 southern Queensland grain harvest was more than double that in 2014, increasing pressure to move grain haulage off the roads.⁵³⁹

These facts indicate that non-coal freight volumes over the undertaking period are likely to be higher than the one path a week forecast by Queensland Rail. Stakeholders have also suggested this is likely to be the case. New Hope said that, while the recent reduction in non-coal traffic had been substantial, this was likely to underestimate the volumes over the longer term.⁵⁴⁰ New Hope also said:

This [the fall in non-coal volume] is not a risk which coal producers are able to manage or to which coal producers should reasonably have expected to be exposed.⁵⁴¹

However, we have not considered the non-coal volumes in detail at this stage or required Queensland Rail to use different volumes. We have sought to minimise coal miners' exposure to

⁵⁴⁰ New Hope, sub. 22: 7, 9.

⁵³³ Queensland Rail, sub. 33: 26.

⁵³⁴ New Hope, sub. 22: 9–10; sub. 31: 13.

⁵³⁵ Yancoal, sub. 27: 3.

⁵³⁶ Yancoal, sub. 35: 2–3.

⁵³⁷ Queensland Rail, sub. 2: 20.

⁵³⁸ See http://statements.cabinet.qld.gov.au/Statement/2016/4/6/palaszczuk-government-delivers-25-million-to-upgrade-oakey-rail-facilities.

⁵³⁹ See 'Agricultural Seeds' exports on p. 2 of https://www.portbris.com.au/PortBris/media/General-Files/MTR/2016/Feb2016MonthlyTradeReport.pdf.

⁵⁴¹ New Hope, sub. 22: 7.

fluctuations in non-coal volumes by providing for pricing for coal services that is based on capacity that is available for them to contract.⁵⁴² This is appropriate, given that coal miners have no ability to control non-coal volumes, or manage the risks that might arise from fluctuations in those volumes.

Given that, among other things, coal miners' exposure to non-coal volumes will be minimised, and that Queensland Rail is not proposing reference tariffs for non-coal services, we accept Queensland Rail's proposed non-coal volumes. The allocation approach that minimises the exposure is discussed in more detail in Section 8.3.3 of this Decision.

We note that the 90 per cent decline in forecast non-coal volumes could be grounds to consider optimising the West Moreton asset base. However, we have not sought to do so at this time, given the uncertainty about future non-coal volumes, and because applying the 80-path contracting constraint means that the effect on coal reference tariffs of the collapse in non-coal demand is minimised (see Table 16 in Section 8.3.3).

Coal volumes

Queensland Rail's 2015 DAU proposed weekly West Moreton network coal volumes of 53 contracted paths, plus 9.8 forecast ad hoc paths. On this basis, Queensland Rail's 2015 DAU West Moreton pricing was developed based on a coal volume forecast of 62.8 paths a week, compared with the 80 weekly paths that are available for coal services to contract (see Section 8.3.3 of this Decision). Queensland Rail said contracted coal paths had decreased due to the closure of the Wilkie Creek mine.⁵⁴³

However, there is substantial uncertainty about the outlook for coal volumes. Queensland Rail said in its March 2016 submission that it had coal access requests in excess of current capacity.⁵⁴⁴ This contrasts with Queensland Rail's argument in its December 2015 submission that a volume forecast excluding ad hoc paths (i.e. just including the contracted volumes) would be appropriate:

Queensland Rail has provided a forecast including significant ad hoc railings, based on both these being allowable under current agreements, and resulting in a reasonable allocation of costs as between contracted and non-contracted demands. Importantly, though, Queensland Rail made these forecasts with the expectation of how changes in demand - either higher or lower ad hoc railings, for instance — would impact on its revenue profile. Had Queensland Rail known the QCA intended to change its approach, then this may have caused Queensland Rail to approach the demand forecast differently. Given the QCA's proposed approach, a forecast based on contracted tonnes would be appropriate.⁵⁴⁵

Queensland Rail's December submission also said that:

as the reference tariffs are based on a forecast greater than contract, where access holders rail solely to contract Queensland Rail will receive insufficient money to fund the maintenance specified in the build up of the reference tariff.⁵⁴⁶

New Hope and Yancoal both said that coal volumes were likely to be above those forecast by Queensland Rail.⁵⁴⁷

⁵⁴² Note that coal services are also able to use ad hoc paths above the 80-path constraint, and the tariff structure provides for them to pay the same price they pay for contracted paths.

⁵⁴³ Queensland Rail, sub. 2: 16–18.

⁵⁴⁴ Queensland Rail, sub. 33: 26.

⁵⁴⁵ Queensland Rail, sub. 26: 53–54.

⁵⁴⁶ Queensland Rail, sub. 26: 25.

⁵⁴⁷ New Hope, sub. 22: 9–10; sub. 31: 13; Yancoal, sub. 27: 3; sub. 35: 2–3.

We agree there is a significant chance that coal volumes will end up higher than Queensland Rail's forecast. Indeed, publicly available export data indicates that actual volumes in 2014–15 were higher than the volumes Queensland Rail has proposed for 2015–20 (see Table 19).

Actual coal volumes (million tonnes)					
	2011–2012	2012–2013	2013–2014	2014–2015	
Jondaryan/New Acland	4.7	4.4	4.8	5.0	
Wilkie Creek	1.6	1.6	0.9	0.0	
Columboola/Cameby Downs	1.4	1.3	1.5	1.6	
Total actual volumes (mt)	7.8	7.3	7.1	6.6	
Actual coal volumes (estimated week	ly train paths) ^a		·		
	2011–12	2012–13	2013–14	2014–15	
Jondaryan/New Acland	48.4	44.9	48.5	50.9	
Wilkie Creek	16.3	16.1	8.9 ⁵⁴⁹	0.0	
Columboola/Cameby Downs	14.6	13.0	15.2	16.8	
Total actual coal paths (estimated)	79.3	74.0	72.7	67.7	
Contracted coal paths (excluding Wilkie Creek)	53.0	53.0	53.0	53.0	
Estimated ad hoc (excluding Wilkie Creek) ^b	10.0	5.0	10.8	14.7	

Table 19: West Moreton network coal export ve	olumes 2011–2015 ⁵⁴⁸
---	---------------------------------

a Calculated by dividing actual annual volumes by 50 weeks and by 1960 tonnes per train.

b Calculated by subtracting 53 contracted paths from total paths for New Acland and Cameby Downs.

Source: Queensland Government, Export by Collieries workbook for actual volumes; Queensland Rail, sub. 2: 17 for contracted paths.

Based on this data, Queensland Rail averaged 14.7 ad hoc coal paths per week in the year ended June 2015, which is 50 per cent more than the 9.8 weekly ad hoc coal paths used in Queensland Rail's forecast. Overall, the estimate of 67.7 weekly actual coal paths, based on the 2014–15 export data, is 7.8 per cent higher than Queensland Rail's forecast (62.8) coal paths.

Further, as discussed below, Yancoal has agreed a haulage contract for a higher tonnage than Queensland Rail has used in its volume forecast. At the same time, Queensland Rail remains exposed if ad hoc volumes are below the 9.8 weekly paths it has included in its volume forecast.

However, given Queensland Rail has proposed to take on some volume risk, we consider it is appropriate having regard to the approval criteria in section 138(2) of the QCA Act to accept Queensland Rail's volume forecast. We note Queensland Rail's argument that its maintenance costs may not be covered by access revenue if coal miners only rail to contract. However, the

⁵⁴⁸ The export volumes will correlate closely with the volumes of coal hauled on the West Moreton network, as all coal is transported to the port by rail. However, the correlation will not be exact, mostly due to the timing of ship departures at the beginning and end of each reporting period.

⁵⁴⁹ As Wilkie Creek closed in December 2013, we estimate that it used 17.9 paths a week for the first six months of the financial year, and none for the remaining half of the year. The 8.9 path weekly figure for 2013–14 in Table 8.7 is an average over 12 months.

evidence to date is that actual volumes have been higher than its forecasts, and the tariff mechanism provides for Queensland Rail to retain any ad hoc revenues, whether volumes be lower or higher than its forecast.

This situation will change to further reduce Queensland Rail's risk if Queensland Rail agrees access contracts for more than its forecast volumes. As discussed in Section 8.5.1, a contract that exceeds the forecast volumes will largely eliminate Queensland Rail's exposure to lower volumes from that origin. In such a situation, it will therefore be appropriate to re-set the tariff to take account of those higher volumes, as provided for in the endorsed variation event for triggered by increases in contracted volumes.

Another approach might be to review Queensland Rail's forecasts now, with a view to approving volumes that would be set in stone over the life of the new undertaking. However, given that there are only two coal mines using the West Moreton network, any assessment of volumes would necessarily depend on reaching a firm conclusion on a small number of uncertain future events.

We have therefore decided to adopt an approach to volumes that does not require us to seek to resolve the uncertainty regarding expected demand over the next four years. It is on that basis that we have used Queensland Rail's forecasts for setting the West Moreton Network tariff, in the knowledge that the interests of access holders and access seekers are protected by the contract volume reset endorsed variation event (see Section 8.5.1).

Our decision will promote the efficient operation and use of the West Moreton network, and the interests of access seekers, access holders and their customers (s. 138(2)(a), (e), (h)). It uses the volume forecasts Queensland Rail has provided, which presumably advances its legitimate business interests (s. 138(2)(b)). Therefore, we consider that using Queensland Rail's forecast volumes, in combination with the contract volume reset, is appropriate having regard to all the criteria in section 138(2).

Volume trigger

Yancoal said that since the Draft Decision it had replaced its previous 1.4 million tonnes a year rail haulage agreement with a new one for 1.7 million tonnes.⁵⁵⁰ Aurizon Operations in January 2016 said the 1.7 million tonne haulage contract for Cameby Downs had been signed for two years starting on 1 February 2016, with an option to extend it for a further two years.⁵⁵¹

A below-rail access agreement for those volumes would trigger the contract volume reset endorsed variation event (see Section 8.5.1 of this Decision).

At the time of making this Decision we are not aware that such an access agreement has been signed. We have therefore retained Queensland Rail's forecasts for assessing tariffs in this Decision. However, if contracted volumes for Yancoal's Columboola loading loop increase above the volumes in Queensland Rail's forecast, the tariffs will be recalculated in accordance with the endorsed variation event.

We have required amendments to the DAU to provide for this trigger to work even if a contract is finalised before the new undertaking is approved (see Section 8.5.3 of this Decision, and Schedule D, cl. 5.1(a)(ii)(B) in Appendix F).

⁵⁵⁰ Yancoal, sub. 35: 2–3.

⁵⁵¹ The haulage agreement announced by Aurizon was with Yancoal subsidiary Syntech Resources. See https://www.aurizon.com.au/news/news/aurizon-announces-coal-haulage-agreement-with-syntechresources.

Summary 8.10

We retain Queensland Rail's proposed forecast volumes for the West Moreton and Metropolitan network reference tariffs for this Decision.

8.11 Forecast maintenance costs

In the 2015 DAU, Queensland Rail estimated total maintenance costs of \$143.0 million for the five-year period July 2015 to June 2020.⁵⁵²

Our 2015 Draft Decision considered as reasonable total maintenance costs of \$114.6 million, having regard to our consultant's (B&H) advice. B&H assessed that aspects of Queensland Rail's maintenance costs were excessive either in scope or cost (e.g. resurfacing allowance and resleepering costs), and that some maintenance activities were capital in nature (e.g. ballast undercutting).⁵⁵³

Stakeholders' submissions

Queensland Rail had significant concerns with B&H's maintenance cost assessment and it provided new information in support of its maintenance cost proposal.⁵⁵⁴

Aurizon had concerns with the reduction in resurfacing allowance and cautioned about treating ballast undercutting as a capital activity.⁵⁵⁵

Miners said that B&H's assessment indicated Queensland Rail's proposed maintenance costs were excessive and reflected inefficiencies; however, they also said that the QCA's estimate was likely to exceed efficient costs.⁵⁵⁶

QCA analysis and Decision

Our Decision accepts as reasonable total maintenance costs for the five-year period July 2015 to June 2020 of \$120.4 million, which is 84 per cent of Queensland Rail's proposed costs of \$143.0 million. We accept that Queensland Rail needs to maintain its ageing railway to maintain the integrity of its network.

Approach to assessing maintenance costs

We engaged B&H to review its previous assessment of maintenance costs in light of stakeholders' comments, including new information provided by Queensland Rail in its post-Draft Decision submission.

B&H retained its previous assessment relating to certain aspects of maintenance costs on which stakeholders raised concerns. For example, B&H did not accept:

• Queensland Rail's claim of a higher resleepering costs, because B&H considered that Queensland Rail's proposed standard and consequently the unit sleeper cost was high and inappropriate, given the track configuration in the sections where resleepering activity was planned; and

⁵⁵² Queensland Rail, sub. 2: 41 and Appendix 6: 22.

⁵⁵³ QCA 2015: 151–154.

⁵⁵⁴ Queensland Rail, sub. 26: 47 and Annexure 2.

⁵⁵⁵ Aurizon, sub. 20: 26–29.

⁵⁵⁶ New Hope, sub. 22: 4, 10–11 and sub. 32: 22–24; Yancoal, sub. 27: 2.

 Queensland Rail and Aurizon's arguments that ballast undercutting should not be capitalised, because B&H considered that the scope proposed by Queensland Rail was actually a reconstruction of track, which was a capital activity.⁵⁵⁷

However, B&H revised its previous assessment of certain other aspects of maintenance costs, considering stakeholders' comments. For example, B&H:

- increased its previously recommended allowance for mechanised resurfacing, as it accepted Queensland Rail's argument that resurfacing activity would increase towards the end of the 2015 DAU period due to reduced effect of related maintenance activities (e.g. resleepering) that were planned at the start of the regulatory period; and
- accepted treating rail renewal as a maintenance activity rather than as a capital item, given the smaller scope and lower cost submitted by Queensland Rail in its December 2015 submission.⁵⁵⁸

Ultimately, B&H's revised assessment resulted in a total West Moreton network maintenance cost allowance of \$120.4 million for the five-year period July 2015 to June 2020.⁵⁵⁹

QCA Decision

We accept B&H's assessment that aspects of Queensland Rail's maintenance cost proposal are excessive and unreasonable. It may be that Queensland Rail's proposal advances its legitimate business interests. However, we consider that it would allow Queensland Rail to recover inefficient costs. We do not consider this to be appropriate, having regard to the object of Part 5 of the QCA Act (s. 138(2)(a)); the public interest (s. 138(2)(d)); the interests of access seekers and access holders (s. 138(2)(e) and (h)); and the inappropriateness generally of allowing the recovery of windfall gains and monopoly profits (s. 138(2)(h)). For those reasons, we do not consider Queensland Rail's proposal to be appropriate.

Based on B&H's recommendation, we consider that total maintenance costs of \$120.4 million are reasonable for the five-year period July 2015 to June 2020. We consider these maintenance costs represent efficient costs for maintaining the West Moreton network that will promote efficient use of the network and are in the interests of access seekers and access holders, and the public interest, having regard to sections 138(2)(a), (d), (e), (g) and (h) and 168A(a) of the QCA Act.

Furthermore, as assessed in Section 8.3.2 of this Decision, we consider the proportion of fixed maintenance costs is 57.3 per cent. Thus, of the \$120.4 million maintenance costs that we consider acceptable, about \$69.0 million is fixed cost and \$51.4 million is variable cost.

As per Summary 8.2 (see Section 8.3.3 of this Decision), we have allocated:

• about 71 per cent of the fixed cost to coal train services, reflecting the ratio of 80/113 paths⁵⁶⁰, which results in a coal-allocated fixed maintenance cost of \$48.6 million; and

⁵⁵⁷ B&H 2016, Part 1: 2, 4–5.

⁵⁵⁸ B&H 2016, Part 1: 3–4.

 ⁵⁵⁹ This is equivalent to the B&H recommended total maintenance cost of \$112.4 million (in June 2015 dollars).
 B&H 2016, Part 1: 5 and 2015: iii.

⁵⁶⁰ As considered in Section 8.3.3 under the heading 'Change in circumstances' the fixed cost allocator for 2015–16 is 69.5 per cent.

 about 98 per cent of the variable cost to coal train services, reflecting coal's share of forecast usage in the West Moreton network, which results in a coal-allocated variable maintenance cost of \$50.3 million.

Consequently, we have allocated \$98.9 million to coal services for the purposes of deriving the West Moreton reference tariff, which is about 82 per cent of the \$120.4 million total efficient maintenance costs.⁵⁶¹

As discussed in Section 8.3.3 of this Decision, we consider our approach to allocating efficient costs to coal traffics is appropriate, having regard to the assessment criteria in the QCA Act (s. 138(2)).

8.12 Forecast operating costs

In the 2015 DAU, Queensland Rail proposed to establish its operating cost allowance based on its 2012–13 below-rail financial statements, but with a downward adjustment for train control costs to reflect our October 2014 Draft Decision. Queensland Rail escalated the adjusted 2012–13 costs by CPI using actual inflation for 2013–14 and forecast inflation (2.5 per cent) for the period thereafter, and proposed a total operating cost allowance of \$37.2 million for the five-year period July 2015 to June 2020.⁵⁶²

Our 2015 Draft Decision proposed to accept Queensland Rail's approach to establishing an operating cost allowance based on the 2012–13 financial statements, with lower train control costs. Additionally, we categorised operating costs into fixed and variable components, with about 82 per cent of the costs identified as being related to fixed activities. Our 2015 Draft Decision allocated the fixed costs reflecting the proportion of West Moreton network capacity available to coal services to contract and variable costs based on coal's share of forecast usage.⁵⁶³

Stakeholders' submissions

Queensland Rail accepted the QCA's categorisation of operating costs into fixed and variable components, and said that fixed operating costs were common network costs. However, Queensland Rail did not support QCA's proposal to allocate fixed operating costs to reflect the 87-path constraint, and said costs should be allocated based on forecast usage.⁵⁶⁴

Miners were concerned that the proposed operating costs were not efficient and argued that aspects of operating costs exceeded comparable benchmarks.⁵⁶⁵ In particular, New Hope suggested that Queensland Rail's allocation of its network-wide costs (for example, corporate overheads) to the West Moreton network required further consideration.

QCA analysis and Decision

Our Decision accepts Queensland Rail's proposal to establish its 2015 DAU operating cost allowance based on its 2012–13 financial statements but with a lower allowance for train control costs. We have escalated those costs by CPI using actual inflation data for 2013–14 and

⁵⁶¹ This compares to the \$139.9 million Queensland Rail considered for deriving the ceiling price for West Moreton coal services in the 2015 DAU.

⁵⁶² For presentational purposes, we have not included working capital allowance, which is set at 0.3 per cent of West Moreton network coal-related allowable revenues.

⁵⁶³ QCA, 2015: 157–159.

⁵⁶⁴ Queensland Rail, sub. 26: 40–41 and sub. 33, Attachment 2: 17.

⁵⁶⁵ Yancoal, sub. 27: 2; New Hope, sub. 22: 4, 11–12 and sub. 32: 26.

2014–15 (now available) and an assumed inflation of 2.5 per cent for the period thereafter, which results in a total operating cost allowance of \$36.9 million for the five-year period July 2015 to June 2020.

We have adopted our Draft Decision which identified that about 82 per cent of operating costs relate to fixed activities with the remainder being for variable activities. We have allocated fixed operating costs to coal train services based on the relative West Moreton network capacity available for contracting by coal services, and allocated variable costs based on coal's share of forecast volumes.

Consequently, we have allocated \$27.9 million operating costs to coal services for the purposes of determining the West Moreton reference tariff.

Approach to assessing operating costs

Queensland Rail's proposed operating costs were based on its 2012–13 below-rail financial statements with a downward adjustment for train control costs as recommended in our 2014 Draft Decision. Queensland Rail did not consider the operating costs data from its 2013–14 financial results because they were higher than the 2012–13 results.⁵⁶⁶

We consider Queensland Rail's proposed operating costs are commensurate with comparable benchmarks, as reported in B&H's assessment of the 2013 DAU proposal; therefore, we do not agree with New Hope that the proposed costs exceed comparable benchmarks.⁵⁶⁷

Queensland Rail's 2012–13 financial statements allocated its network-wide costs (e.g. corporate overheads) to the West Moreton network based on the allocators set out in the costing manual that was approved by the QCA in April 2013. New Hope suggested that the allocators in the costing manual required further consideration.

We do not consider it is relevant to review the costing manual allocators for the purpose of this Decision. That is because the costing manual provides a framework for Queensland Rail to keep accounting records for below-rail services separate from the accounting records for its other operations. In particular, the costing manual allows identifying actual operating costs relating to Queensland Rail's below-rail services in a particular year. We understand that past operating costs can be a starting point for estimating an allowance for future operating costs, but they are not relevant for determining efficient operating costs. For instance, in our 2014 Draft Decision, we accepted most operating cost allowance components based on Queensland Rail's 2012–13 financial statements, having regard to our consultant's assessment that those costs were commensurate with comparable benchmarks. However, we did not accept the train control costs based on the 2012–13 financial statements because we considered them excessive compared to other benchmarks.

Given these considerations, we accept Queensland Rail's approach of establishing its operating costs allowance based on its 2012–13 financial statements, with lower allowance for train control costs. We applied actual inflation where available to escalate the 2012–13 cost data,

⁵⁶⁶ The operating costs reported in the 2013–14 financial statements were about 35 per cent more than those reported in the 2012–13 financial statements, based on a comparison of the costs of train operations management, corporate overhead and other expenses. The 2014–15 financial results were not available at the time Queensland Rail submitted its 2015 DAU. They are now available on Queensland Rail's website, and the costs reported in the 2014–15 financial statements are about four per cent more than the costs recommended in our 2014 Draft Decision that were based on the 2012–13 financial statements with a downward adjustment for train control costs.

⁵⁶⁷ B&H 2014: 48–52.

which resulted in a total operating cost allowance of \$36.9 million for the five-year period July 2015 to June 2020. We consider these operating costs represent efficient costs for operating the West Moreton network and are appropriate having regard to section 138(2) of the QCA Act.

Furthermore, since stakeholders did not object to our 2015 Draft Decision categorisation of operating costs into fixed and variable components, we have adopted our 2015 Draft Decision and determined that about 82 per cent of operating costs relate to fixed activities and the remainder (18 per cent) display variable activities—that is, about \$30.1 million is fixed and \$6.8 million is variable cost.

As per Summary 8.2 (see Section 8.3.3 of this Decision), we have allocated:

- about 71 per cent of the fixed cost to coal train services, reflecting the ratio of 80/113 paths⁵⁶⁸, which results in a coal-allocated fixed operating cost of \$21.2 million; and
- about 98 per cent of the variable cost to coal train services, reflecting coal's share of forecast usage in the West Moreton network, which results in a coal-allocated variable operating cost of \$6.6 million.

Consequently, we have allocated around \$27.9 million⁵⁶⁹ to coal services for the purposes of deriving the West Moreton reference tariff, which is about 76 per cent of the \$36.9 million total efficient operating costs.⁵⁷⁰

As discussed in Section 8.3.3 of this Decision, we consider our approach to allocating efficient costs to coal traffics is appropriate, having regard to the assessment criteria in the QCA Act (s. 138(2)).

8.13 Opening asset base

The RAB is a key component in determining a tariff using a building block approach. This RAB needs to be established through a regulatory process when an existing asset becomes regulated. Establishing an investment value for the RAB for the coal services provided by the West Moreton network has been particularly contentious over some time.⁵⁷¹ It has been contentious because the rail line was constructed in the 19th century for regional traffic (e.g. livestock, grain and other agricultural commodities, passengers and general freight). It was not designed for the heavy-haul coal services that are now its largest source of traffic and to which the reference tariff applies. Particular attributes of the West Moreton network are:

- a maximum length of 675 metres for coal trains, with an axle load of 15.75 tonnes. In contrast, in the central Queensland coal network, coal trains are two kilometres long with axle loads of 26 tonnes or more;
- train speeds limited by sharp curves and steep grades on the range east of Toowoomba; and

⁵⁶⁸ As considered in Section 8.3.3 under the heading 'Change in circumstances' the fixed cost allocator for 2015–16 is 69.5 per cent.

⁵⁶⁹ Due to rounding to one decimal, the coal allocated operating cost of \$27.9 million would not appear to be a strict sum of the \$21.2 million of coal allocated fixed operating cost and \$6.6 million of coal allocated variable operating cost.

⁵⁷⁰ This compares to the \$34.9 million Queensland Rail considered for deriving the ceiling price for West Moreton coal services in the 2015 DAU.

⁵⁷¹ For a discussion of the history of West Moreton network tariff proposals, including asset valuations, see QCA, 2014d, Appendix C: 195–199.

• trains carrying less than 2,000 tonnes, compared with about 10,000 tonnes in central Queensland.

High ongoing maintenance and capital spending

Queensland Rail's maintenance and capital spending are high relative to the capacity of its West Moreton network, to compensate for the age, standard and configuration of the rail infrastructure (see Sections 8.11 and 8.15 of this Decision).

Our assessed 2016–17 maintenance cost for Queensland Rail is \$9.79 per thousand gtk. This is almost five times the \$2.03 per thousand gtk that we have assessed as efficient for Aurizon Network on the Goonyella system in Central Queensland for 2016–17. Put another way, Queensland Rail's proposed maintenance cost for a kilometre of track carrying 6.2 million tonnes a year is about 68 per cent of the per-kilometre maintenance cost for 106 million tonnes a year in Goonyella.⁵⁷² This high level of spending on a low-volume network does not reflect a major difference in terrain—Goonyella has many similar natural challenges. Rather, it reflects, in large part, the obsolete infrastructure and alignment of the West Moreton network.

Queensland Rail has also proposed \$141.9 million of capital spending to renew its network over the five years covered by its tariff proposal, even though it has forecast that its volumes will not increase during the period. This is on top of the \$173.8 million post-1995 common network asset value at 1 July 2015, approved in this Decision (representing the rolled-forward value of capital spending between 1995 and 2015).

Queensland Rail said in its submission accompanying the 2015 DAU that:

[a]s the network was initially designed to cater for non-coal traffics, investment in infrastructure improvements, by both Queensland Rail and West Moreton Network end-users, has been necessary to accommodate coal carrying train services. Being built on a black soil plain and having tight radius curves down the Toowoomba and Little Liverpool Ranges has created additional challenges.⁵⁷³

Queensland Rail has predicted in its West Moreton System Asset Management Plan that the elevated levels of maintenance and capital spending on the network infrastructure will continue for at least another decade.⁵⁷⁴

West Moreton network would not be built today

The West Moreton network is an unusual regulated asset in that it was not built for the traffic for which it is largely being used. It therefore departs significantly from the sort of network that would be built today to transport coal.⁵⁷⁵

Indeed, if the network was built today it would have much greater capacity and a different configuration. Evidence from coal and rail projects proposed for the Galilee Basin in central Queensland is that a new entrant, where there was no railway line, would only develop a new

⁵⁷² Queensland Rail proposed a 2016–17 cost of \$55,991 per track kilometre for maintaining the West Moreton network (not including the cost of mechanised resleepering), while we have assessed that the efficient cost for Aurizon Network to maintain the Goonyella network in central Queensland is \$82,570 per track kilometre (Queensland Rail, sub. 2: 20; QCA 2016b, Volume IV—Maximum Allowable Revenue).

⁵⁷³ Queensland Rail, sub. 2: 9.

⁵⁷⁴ Queensland Rail, sub. 2, Appendix 6—West Moreton System Asset Management Plan.

⁵⁷⁵ Queensland Rail, sub. 33, Att. 3: 14–15.

rail system at a scale several times the size of the West Moreton network.⁵⁷⁶ The Galilee projects were planned to be served by modern rail lines with above- and below-rail scale efficiencies including trains carrying as much as 25,000 tonnes of coal—more than 12 times as much as West Moreton network trains—and lower maintenance costs than the West Moreton network.

The Australian Rail Track Corp's (ARTC's) proposed alignment for its Inland Rail project provides further evidence that there is no direct modern engineering equivalent, as required for a DORC valuation, for Queensland Rail's West Moreton network assets. The ARTC plan includes some sections that use the route of the West Moreton network track. But it would not use any of Queensland Rail's tunnels—rather it would 'include a 5km tunnel near Toowoomba to create an efficient route through the steep terrain of the Toowoomba Range'.⁵⁷⁷

Context of asset value

Queensland Rail's June 2013 DAU derived an asset valuation of \$419.6 million (in June 2013\$) for the West Moreton network, using a DORC methodology.⁵⁷⁸

In our October 2014 Draft Decision we concluded that this valuation was not consistent with the high maintenance costs and capital spending proposed by Queensland Rail. We proposed an asset value of \$246.6 million for the common network between Rosewood and Columboola, as at 1 July 2013.⁵⁷⁹

Queensland Rail's 2015 DAU proposed an asset value of \$463.6 million for the common network, as at 1 July 2013.⁵⁸⁰

The QCA's 2015 Draft Decision was to refuse to approve the DAU and to require Queensland Rail to amend the DAU to include an asset value of \$235.8 million for the common network, at 1 July 2013, that was derived by:

- giving value to assets such as rail, concrete sleepers and concrete bridges that have been replaced as capital items and have not been fully depreciated;
- assigning no additional value to assets where the nature of the asset is such that it has been funded and will continue to be funded from maintenance expenditure (itself a cost taken into account separately in assessing the revenue to be recovered by Queensland Rail). These include items such as wooden sleepers, fences, ballast and wooden bridges; and
- not giving tunnels, cuttings and embankments any additional value beyond that given to other assets and hence the network as a whole.⁵⁸¹

⁵⁷⁶ See https://publications.qld.gov.au/dataset/queensland-coal-mines-advanced-projects: 2–3. Planned annual production of Galilee Basin thermal coal export projects ranges from 15 million tonnes to 60 million tonnes, with an average of 35 million tonnes. Queensland Rail forecasts that the West Moreton network will have annual volumes of 6.2 million tonnes of thermal coal during the 2015–20 regulatory period (Queensland Rail, sub. 2: 17).

⁵⁷⁷ ARTC 2015.

⁵⁷⁸ Queensland Rail, 2013a: 3.

⁵⁷⁹ QCA 2014d: 140.

⁵⁸⁰ Queensland Rail, sub. 2: 35. Queensland Rail proposed an asset value at 1 July 2013, then rolled it forward to derive the regulatory asset base value for the start of its proposed regulatory period. We have adopted the same approach for consistency in this analysis. The roll-forward is discussed below in Section 8.17.

⁵⁸¹ QCA 2015: 168–169, 176.

We said:

The QCA's valuation approach addresses the imbalance between Queensland Rail's need to spend money on the network and its proposed asset value, and the inefficiency consequences of that imbalance.⁵⁸²

Stakeholders' submissions in response to the Draft Decision

Queensland Rail said the QCA's proposed asset valuation broke with regulatory precedent and that the updated Menezes Valuation Report had serious flaws.⁵⁸³ New Hope, Yancoal and Aurizon supported the QCA's proposed valuation, as it reflected the high maintenance costs for the network.⁵⁸⁴

Menezes valuation report

We engaged Professor Menezes as an independent expert to provide opinions on the economic issues relating to the West Moreton network asset valuation. Professor Menezes was subsequently appointed to the QCA Board. This appointment is discussed in more detail in the introduction to this Decision, under the heading 'Independent economic advice'.

A key aspect of his expert advice was to provide an opinion on which asset valuation approaches were most appropriate for the network, with regard to the economic principles of allocative, productive and dynamic efficiency that are relevant to the matters mentioned in the approval criteria in the QCA Act, including sections 138(2)(a) and (d).

Professor Menezes provided a preliminary report that we published in May 2015, to allow stakeholders to comment before our Draft Decision. The Menezes Valuation Report published with the Draft Decision gave regard to matters raised by Queensland Rail (including reports prepared for Queensland Rail by PwC and Frontier Economics), and by stakeholders, including New Hope, Yancoal and Aurizon, in response to the preliminary report.

The Menezes Valuation Report found that valuing assets whose lives have exceeded their expected useful lives increases the chances of setting a price that:

- (a) includes a component of monopoly rent;
- (b) would yield windfall gains to Queensland Rail; and
- (c) is sufficiently high to adversely impact competition in relevant markets.⁵⁸⁵

Queensland Rail responded in its December 2015 submission with a further report by PwC that argued Professor Menezes had:

- (a) not demonstrated windfall gains would arise, as that required knowing the expectations of the original investors,⁵⁸⁶
- (b) not addressed that excluding assets from a DORC valuation would create regulatory risk;⁵⁸⁷

⁵⁸² QCA 2015: 177.

⁵⁸³ Queensland Rail, sub. 26: 6, 15, 20–25, 27; sub. 26, Annexure 3: 1–8; Queensland Rail, sub. 33: 4, 8, 27; sub. 33, Att. 4.

 ⁵⁸⁴ Aurizon, sub. 20: 22; sub. 29: 4–5; New Hope, sub. 22: 14–15; sub. 32: 18–-22; Yancoal, sub. 27: 2; sub. 35: 5.

⁵⁸⁵ Menezes, F 2015b: 32–35.

⁵⁸⁶ Queensland Rail, sub. 26, Annexure 3: 4–5.

⁵⁸⁷ Queensland Rail, sub. 26, Annexure 3: 6–7.

- (c) not shown Queensland Rail's proposal would affect allocative efficiency or competition in other markets;⁵⁸⁸
- (d) used an inconsistent approach to Queensland Rail's and access holders' expectations in the treatment of the asset valuation and adjustment amount; ⁵⁸⁹and
- (e) ignored Queensland Rail's proposal to 'de-couple' the reference tariffs on the West Moreton Network from the ceiling revenue limit.⁵⁹⁰

Professor Menezes provided a supplementary report titled 'Response to Stakeholder comments on comments', that addressed PwC's comments. He advised that:

- (a) allowing a firm to earn a return on assets with expired expected useful lives increased the risk of recovering windfall gains regardless of the motivation that underpinned the original investment;⁵⁹¹
- (b) not allowing a regulated entity to make a return on assets beyond the expected useful lives of those assets does not increase regulatory risk;⁵⁹²
- (c) there are no gains to allocative efficiency but there are potential losses from allowing Queensland Rail to earn a return on assets with expired expected useful lives;⁵⁹³
- (d) there is no inconsistency between the approach to the treatment of the asset valuation and adjustment amount;⁵⁹⁴ and
- (e) Queensland Rail's 'de-coupling' proposal to price below a 'ceiling price' but retain the discretion to raise that price does not prevent allocative efficiency from being adversely impacted in the future.⁵⁹⁵

We also engaged Professor Stephen King of Monash University to conduct a peer review of Professor Menezes reports, including the valuation reports. Professor King found that Professor Menezes' reports were 'based on a high standard of rigorous economic analysis' and the conclusions in the reports were reasonable as a matter of economics.⁵⁹⁶

Professor King's report and Professor Menezes supplementary report are published with this Decision.

QCA analysis and Decision

We have considered Queensland Rail's 2015 DAU and reviewed our Draft Decision in light of stakeholder comments and further information provided by Queensland Rail.

Having regard to the assessment criteria in section 138(2) of the QCA Act, our Decision is to require Queensland Rail to amend the DAU to include a value of the regulatory asset base of \$254.5 million for the network from Rosewood to Columboola, as at 1 July 2013.

⁵⁸⁸ Queensland Rail, sub. 26, Annexure 3: 7–8.

⁵⁸⁹ Queensland Rail, sub. 26, Annexure 3: 10.

⁵⁹⁰ Queensland Rail, sub. 26, Annexure 3: 8.

⁵⁹¹ Menezes, F 2016b: 7–8.

⁵⁹² Menezes, F 2016b: 8.

⁵⁹³ Menezes, F 2016b: 11.

⁵⁹⁴ Menezes, F 2016b: 18.

⁵⁹⁵ Menezes, F 2016b: 12.

⁵⁹⁶ King, S 2016: 2.

We have considered the value of the RAB with regard to the nature of the West Moreton network and its history. The discussion below addresses a range of relevant issues, including matters raised by Queensland Rail and other stakeholders:

- the regulatory framework;
- discussion;
 - regulatory approaches can vary for the valuation of a regulatory asset base;
 - valuing a rail network;
 - regulatory certainty;
 - no settled value of the regulatory asset base;
 - o no settled valuation methodology for the West Moreton network; and
 - o no 'material departure' or impact on future investment;
 - consistency and expectations;
 - the treatment of renewal assets;
 - service potential;
 - tunnels, cuttings and embankments; and
 - assets replaced through maintenance; and
 - the mindset of investors and evidence.

The analysis then:

- considers Queensland Rail's proposal with regard to the approval criteria in the QCA Act;
- sets out our required valuation; and
- sets out our Decision.

Much of our Decision is consistent with our October 2015 Draft Decision. As such, this Decision also relies on and incorporates our analysis in section 8.7.1 of our Draft Decision, unless revised in this section.

Regulatory framework

The QCA may approve a DAU only if it considers it appropriate to do so having regard to each of the criteria in section 138(2) of the QCA Act. The criteria of most focus for the value of the regulatory asset base are:

- the object of Part 5 of the QCA Act, which is to promote the economically efficient operation
 of, use of and investment in, significant infrastructure by which services are provided, with
 the effect of promoting competition in upstream and downstream markets (ss. 138(2)(a) and
 69E);
- the pricing principles which specify, among other things, that the access price should:
 - generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved (s. 168A(a)); and
 - provide incentives to reduce costs or otherwise improve productivity (s. 168A(d));
- the legitimate business interests of the owner or operator of the service (s. 138(2)(b));

- the public interest, including the public interest in having competition in markets (whether or not in Australia) (s. 138(2)(d));
- the interests of persons who may seek access to the service (s. 138(2)(e));
- the effect of excluding existing assets for pricing purposes (s. 138(2)(f)); and
- the interests of access holders and avoiding monopoly rents and windfall gains (s. 138(2)(h)).

Chapter 10 of this Decision provides further discussion of section 138(2) of the QCA Act.

The QCA has had regard to each factor in section 138(2) in considering the value of the regulatory asset base.

Discussion

Regulatory approaches can vary for the valuation of a regulatory asset base

Australian regulators have, over time, considered and used a variety of valuation approaches for monopoly infrastructure assets. For example, the QCA's Statement of Regulatory Pricing Principles for the Water Sector endorses deprival value, which in turn is bounded on the high side by DORC. It says:

[T]he Pricing Principles should be viewed as a broad statement of regulatory intent to be applied with a discretion that reflects particular circumstances.

As a result, any particular approach cannot be considered to be definitive or binding on the Authority in a specific instance.⁵⁹⁷

For similar issues in Western Australia, Allen Consulting Group advised the Economic Regulation Authority that:

[w]hile economic principles suggest that regulated assets should not be valued at less than scrap value or more than a (correctly-determined) DORC value, the principles do not provide guidance as to whether a regulatory asset value should be set as scrap value or at DORC value, or at any particular value in between. There is no economic efficiency reason for regulated assets to be valued at a level that is commensurate with the cost structure of a hypothetical (efficient) new entrant.⁵⁹⁸

Professor Menezes said in his Valuation Report that DORC is used in particular contexts and the characteristics of the West Moreton network, that was not built for carrying heavy haul coal services, create particular challenges for a DORC valuation.⁵⁹⁹ As Professor Menezes noted, the history of initial asset valuations by regulators:

illustrates that while DORC has played a prominent role, there is a range of asset valuation methods that have been used by regulators in Australia, and highlights that determining initial asset values also entails wider considerations that are likely to vary on a case-by-case basis.⁶⁰⁰

A well-known example of applying valuations to suit the circumstances came with the five Victorian electricity distribution businesses in the 1990s, where the DORC values were adjusted up or down to promote uniform pricing across urban and rural consumers.⁶⁰¹

⁵⁹⁷ QCA 2000a: ii.

⁵⁹⁸ Allen Consulting Group 2005: 3.

⁵⁹⁹ Menezes, F 2015b: 11.

⁶⁰⁰ Menezes, F 2015b: 29.

⁶⁰¹ See Institute of Public Affairs 1999: 6.

Valuing a rail network

Railway networks have a large number of essential individual parts, including sleepers, rails and tunnels. No single element has service potential until it is combined with the other parts of the network. The service potential and valuation of individual assets is an input to the service potential and value of the network as a whole. The task is to value the network, not the sum of its parts.

In a competitive market the value of a network is based on buyers' and sellers' assumptions about the present value of expected future returns from it. However, where these returns are set by reference to the regulated asset value, the present value of those returns cannot guide the asset value. This 'circularity problem' is typically solved in two ways by Australian regulators:

- (a) For new assets built once regulation has started, the predominant valuation methodology has been depreciated actual cost (DAC).⁶⁰² This is because, among other things, the necessary cost information is available and the DAC approach gives an incentive for the asset owner to invest as it can expect to recover its efficient investments through return on and of capital over the life of the asset. However, DAC has the potential to understate or overstate service potential because it is a cost-based approach.
- (b) For the assets that are in place at the start of regulation, regulators and governments have applied a variety of methodologies to establish an opening asset value. These have included:
 - (i) DAC, although this has tended to be difficult, particularly for older assets, due to a lack of information;⁶⁰³
 - (ii) line in the sand, often imposed by a government to achieve a policy outcome; and
 - (iii) DORC, usually adjusted to suit the particular nature of the asset being valued, and requiring the identification of the modern engineering equivalent.⁶⁰⁴

The Productivity Commission did a thorough review of asset valuation techniques in its first review of the national access scheme more than a decade ago. The Commission found that there were:

two implications that the characteristics of infrastructure have for the choice of asset valuation method:

- to the extent that the assets are sunk, the facility owner will continue to supply infrastructure services so long as the regulated value exceeds scrap value (the next best alternative use); and
- if the facilities providing these services are natural monopolies, inefficient duplication is likely to occur only rarely regardless of the asset valuation method.

⁶⁰² For example, the capital expenditure approval process applied for Aurizon Network is a DAC approach. Queensland Rail has proposed a similar DAC-style mechanism for new assets in Schedule E of its 2015 DAU (see Section 8.9 of this Decision).

⁶⁰³ Queensland Rail, sub. 3, Att. D: 12–14.

⁶⁰⁴ For example, the QCA opted to use a brownfields DORC approach when establishing the asset value for the central Queensland coal region assets in 2001. Other asset valuations have departed from a 'standard' DORC approach, including those for the Victorian electricity distribution networks and Queensland's gas distribution networks.

This suggests that a range of valuation methods could reasonably be used and that no method is likely to be intrinsically superior. Indeed, at least conceptually, both DORC and DAC would appear to be equally able to deliver outcomes which are allocatively and productively efficient.⁶⁰⁵

Additionally:

Clearly, the myriad of specific issues that arise across infrastructure sectors means that regulators should not be bound to use one particular asset valuation approach in all situations. Rather, the Commission considers that the approach used should have regard to specific circumstances.⁶⁰⁶

Past practice has shown that regulators and governments have tailored the valuation approach to the situation.⁶⁰⁷

Regulatory certainty

We canvassed the appropriate valuation and valuation approach for the West Moreton network in our October 2015 Draft Decision and Queensland Rail responded by reiterating its arguments that:

- the QCA approved an initial asset value for the West Moreton network when it approved the 2010 pricing amendments to QR Network's 2008 access undertaking;⁶⁰⁸
- there is a 'longstanding application of a DORC valuation methodology ... to the assets comprising the West Moreton Network';⁶⁰⁹ and
- the QCA had made a 'material departure' with its valuation approach and this created regulatory uncertainty.⁶¹⁰

No settled value of the regulatory asset base

The valuation of the regulatory asset base for the West Moreton network has been unresolved since access regulation for Queensland's rail networks began in the 1990s.

A reference tariff for coal services on the West Moreton network was first introduced into the 2006 undertaking, however no regulatory asset base valuation was settled in so doing.

In response to QR Network's 2009 DAU, the QCA in its December 2009 Draft Decision proposed a tariff that rejected QR Network's proposed valuation and proposed an alternative. In response to that Draft Decision, in its 2010 DAU QR Network submitted a tariff at the level proposed by the QCA. However, in making that proposal QR Network indicated it did not accept the rationale that sat behind the QCA's Draft Decision.⁶¹¹

The QCA stated in its June 2010 draft pricing decision that 'the Authority has not achieved its desired objective of finalising a repeatable and transparent methodology for deriving the western system [West Moreton network] tariff'.⁶¹² The June 2010 Final Decision on QR

⁶⁰⁵ Productivity Commission, 2001: 360.

⁶⁰⁶ Productivity Commission, 2001: 366.

⁶⁰⁷ For a discussion of regulatory practice in applying DORC, see Appendix B of our October 2014 Draft Decision (QCA 2014: 188–194).

⁶⁰⁸ Queensland Rail, sub. 26: 6, 20–23; sub. 26, Annexure 3: 7.

⁶⁰⁹ Queensland Rail, sub. 26: 6, 7, 23.

⁶¹⁰ Queensland Rail, sub. 26: 23–24; sub. 26, Annexure 3: 6–7.

⁶¹¹ QR Network, 2010a:116–118; 2010b: 105, 114–115.

⁶¹² QCA, 2010 a: 89.

Network's June 2010 Extension DAAU approved new prices for the West Moreton network, but did not change the fact that the derivation of that price had not been resolved.⁶¹³

Queensland Rail said that its July 2015 further submission on RAB valuation matters raised in the 2015 DAU:

pointed to the QCA's persistent and consistent application and advocacy for a DORC valuation methodology (without the zero valuing of "life expired assets") to the assets in the West Moreton Network which culminated in the QCA approving a reference tariff based on just such a valuation by its own independent consultants.⁶¹⁴

New Hope and Yancoal said the QCA was not bound to adopt a DORC valuation and that Queensland Rail was wrong to claim that the QCA approved the asset values discussed in the December 2009 Draft Decision.⁶¹⁵ New Hope said:

QR now seeks to rewrite this history and claim the existence of an approved valuation of the initial asset base in order to limit the QCA's discretion in applying the statutory criteria to the assessment of the 2015 DAU ... In reality, no stakeholders (including Aurizon Network) supported the asset values set out in the Draft Decision, and other factors led to acceptance of the tariff (but not the asset values or the methodology).⁶¹⁶

The QCA's June 2010 decisions did not reflect a concluded view on the valuation of the regulatory asset base for the West Moreton network. To the contrary, it is apparent from these decisions that the valuation had not been settled.

Queensland Rail has previously shared this view:

[I]n past undertakings the QCA and QR Network were unable to reach agreement on exact building block parameters for the West Moreton system ... The establishment of a transparent and repeatable building blocks approach, including the creation of a Regulatory Asset Base (RAB), would provide a degree of revenue/cost certainty going forward ...⁶¹⁷

Queensland Rail has now, however, in its December 2015 submission, put the contrary position. Queensland Rail has quoted our entire discussion of the Western system (West Moreton) price from our June 2010 draft pricing decision.⁶¹⁸ It said this supported its argument that the methodology for setting the tariff was not settled, but the regulatory asset base valuation had been resolved. In particular, Queensland Rail submits that 'any disagreement related to the allocation of the value between different traffics, not the value itself'.⁶¹⁹

Whilst the extracted passage from the 2009 Draft Pricing Decision does discuss allocation issues, it is wrong to treat what is said in the 2009 Draft Pricing Decision as only relating to allocation issues. The discussion extracted in Queensland Rail's December 2015 submission sits below, and distinct from, the general point made that, in accepting the Western system tariffs, the QCA said 'there remains outstanding the question of the most appropriate way of deriving these tariffs'. As accepted by Queensland Rail in 2013, the derivation of tariffs involves the consideration of all the building blocks, including the regulatory asset base. It is apparent that neither the methodology for establishing the value of the regulatory asset base, nor that value, has ever been settled.

⁶¹³ QCA, 2010b.

⁶¹⁴ Queensland Rail, sub. 26: 23.

⁶¹⁵ New Hope, sub. 22: 13–14; sub. 32: 18–21; Yancoal, sub. 35: 5.

⁶¹⁶ New Hope, sub. 22: 13.

⁶¹⁷ Queensland Rail, June 2013: 6.

⁶¹⁸ Queensland Rail, sub. 26: 21–22.

⁶¹⁹ Queensland Rail, sub. 26: 22.

Queensland Rail also submitted:

even if there was a disagreement between the QCA and Aurizon Network over asset value issues, that disagreement is not relevant to determine the value for Queensland Rail's asset base. Indeed, Queensland Rail accepted the QCA's asset value and rolled it forward.⁶²⁰

However, this is not a matter for just the regulated party and the QCA—we also need to have regard to the views of affected stakeholders. We have therefore reviewed all the stakeholder submissions on the Western system tariff in response to our 2009 Draft Decision, leading up to our approval of the June 2010 pricing amendments. We have reprinted those comments in full in Appendix B.

The submissions from 2010 do not cover any of the details in the QCA's 25-page discussion of the Western system tariff in the December 2009 Draft Decision. But they make it clear that the lack of detailed response was because the stakeholders expected a 'proper assessment of tariffs and other issues for the Western System' for the next tariff period.⁶²¹

That 'proper assessment' with extensive comments on all aspects of the tariff, including the value of the regulatory asset base, is what is happening now. Given the importance of the matter, and the issues raised, it would not have been appropriate to reach a settled view without such a process.

Queensland Rail's submission, that any 'disagreement' from the past can be put to one side as it has now accepted the QCA's asset value and rolled it forward, involves a misunderstanding of the role of the QCA under the QCA Act. The QCA may approve a DAU only if it considers it appropriate to do so having regard to the matters in the paragraphs of section 138(2).

In any event, Queensland Rail did not accept the QCA's valuation and roll it forward, but instead sought in its June 2013 DAU to establish a different starting value for the RAB. Queensland Rail proposed changes to the asset valuation in the 2009 Draft Decision, including reinstating most of an adjustment for the present value of future capital expenditure that the QCA had proposed to deduct.⁶²² It further amended the valuation in the 2015 DAU to add additional capital spending and new assets and to provide for interest during construction.⁶²³ This too shows that at other times Queensland Rail has not regarded the RAB value as having been settled in 2009.

No settled valuation methodology for the West Moreton network or 'material departure'

Queensland Rail has, in its December 2015 and March 2016 submissions, argued that the QCA's approach in its October 2014 and October 2015 Draft Decisions departed from regulatory precedent and that we created 'regulatory uncertainty' by seeking to apply a valuation approach other than the DORC methodology it proposed.⁶²⁴ Queensland Rail said:

The QCA has given little regard to its own past regulatory decisions relating to the West Moreton Network. Queensland Rail is concerned that any investment that it and other stakeholders make will be at risk should the QCA 'change its mind' again.⁶²⁵

Queensland Rail's adviser PwC said the QCA's valuation was a 'material departure' from approaches approved in previous access undertakings. PwC said:

⁶²⁰ Queensland Rail, sub. 26: 23.

⁶²¹ QRC, 2010a: 9; QRC 2010b: 10–11; QRC 2010c: 9; Syntech, 2010: 3.

⁶²² Queensland Rail, 2013a June: 10–11.

⁶²³ Queensland Rail, sub. 2: 35.

⁶²⁴ Queensland Rail, sub. 26: 23–24; sub. 33: 4.

⁶²⁵ Queensland Rail, sub. 26: 24.
Queensland Rail might reasonably perceive that the regulatory framework has become less stable and predictable, contributing to an increase in regulatory risk.⁶²⁶

And Queensland Rail said in its March 2016 submission that:

the QCA's Draft Decision sets new precedents which raise concerning issues for other regulated services providers. It signals the regulator's willingness to reopen previously settled matters such as asset valuation ...⁶²⁷

To be clear, the QCA has not 'changed its mind' on the West Moreton network asset valuation or methodology. As shown above, the QCA has not previously formed a final considered view on the value of this network. There has not been a 'longstanding application of a DORC valuation methodology ... to the assets comprising the West Moreton Network'. In the process that has been ongoing since 2005 the QCA has explored a DORC-based methodology, but has never committed to it.

Indeed, in 2005 we said Western system tariffs should be assessed based on a 'well-accepted framework such as the DORC methodology'. However we also said the actual age of the system assets needed to be considered, as 'many of the assets included in the valuation have either reached, or are approaching the end of their economic lives'.⁶²⁸

In assessing Queensland Rail's 2013 DAU and 2015 DAU tariff proposals, the QCA has needed to address, among other things, the issues raised by a more than 80 per cent jump in annual maintenance costs, compared with those proposed by the QCA in its 2009 Draft Decision. In doing so, the QCA has canvassed different valuation methodologies in the context of the assessment criteria in section 138(2) of the QCA Act.

The difficulty of settling the matter is shown by the widely diverging views of the various parties.

As Aurizon said:

Queensland Rail's submission has not acknowledged the fundamental basis for the rejection of the prior valuation. ... The QCA has not ignored the regulatory precedent but correctly noted that the precedent needed to be reviewed in light of a material change in circumstances.⁶²⁹

The QCA does not agree with Aurizon's submission to the extent it suggests there was a settled prior valuation or a relevant regulatory precedent. However, the QCA notes Aurizon's contention that the very significant jump in annual maintenance costs and ongoing high levels of renewal capital expenditure necessitated a review of appropriate methodologies and values for the West Moreton network.

The QCA acknowledges that the review has been an important element of this regulatory process that Queensland Rail has had to engage with, and the QCA has had regard to the interests of Queensland Rail that arise from the review. However, the QCA does not accept the characterisation submitted by Queensland Rail that the QCA, or the review, has created regulatory uncertainty because a certainty from the past has been displaced.

As discussed above, DORC is neither the sole regulatory valuation approach adopted in Australia, nor the settled valuation approach for the West Moreton network.

⁶²⁶ Queensland Rail, sub. 26, Annexure 3: 6–7.

⁶²⁷ Queensland Rail, sub. 33: 4.

⁶²⁸ QCA 2005: 77.

⁶²⁹ Aurizon, sub. 29: 4.

After having regard to all of the criteria in the QCA Act (s. 138(2)), we have valued the assets using a methodology appropriate to the circumstances of the West Moreton network, including its standard, configuration and condition. Those circumstances include that the network is so far from modern engineering equivalent that a conventional DORC valuation is not appropriate. This is shown by various factors, including a near-doubling of maintenance spending from a level we previously considered reasonable, and Queensland Rail's forecast that high levels of maintenance and capital spending would continue for at least another decade. Nevertheless, our valuation approach uses aspects of a DORC methodology, particularly for recognising the value of assets dating from before coal services began in the mid-1990s, that still have remaining expected useful life.

In this context, the QCA does not consider that its approach to valuing the West Moreton network would have an impact on Queensland Rail's future investment decisions.

Consistency and expectations

Queensland Rail's consultant PwC also suggested that Professor Menezes' approach to regulatory certainty and expectations was inconsistent between the West Moreton network asset valuation and the adjustment amount. PwC said:

In the context of assessing a change in asset valuation approach for Queensland Rail, Professor Menezes is dismissive of Queensland Rail's expectations, yet for access seekers he accepts unequivocally the same premise for access seekers in the context of a tariff adjustment.⁶³⁰

We disagree with this premise. As explained in the discussion of the adjustment amount later in this chapter, Queensland Rail engaged in conduct that gives rise to an expectation that it would apply an adjustment charge provision to recoup or refund any difference between interim tariffs and those approved in a new undertaking (see Section 8.18 of this Decision).

In contrast, the West Moreton network regulatory asset base valuation was never settled and, in proposing its tariff in the 2013 DAU, Queensland Rail should have expected that the QCA would carefully review that valuation, particularly given Queensland Rail's proposed near-doubling of the maintenance costs.

For any adjustment amount, a key factor is promoting future efficient use and investment in the network and dependent markets by preserving regulatory certainty that will arise from giving effect to the expectation that an adjustment amount would be included in the approved access undertaking.

This is not analogous to the QCA's treatment of existing assets which takes into account the extent to which the assets continue to have remaining expected useful life. The assets being considered here are sunk and any valuation as at 1 July 2013 of these assets that is appropriate having regard to the approval criteria in section 138(2) should not have a forward investment impact, in a regulatory regime where efficient future investment is added to the RAB and recovered through approved access charges.

We have not signalled our 'willingness to reopen settled matters such as the asset valuation', as Queensland Rail claims.⁶³¹ Rather, we have determined an efficient asset value through the regulatory process, that is appropriate to approve having regard to the approval criteria in the QCA Act (s. 138(2)). Now that the RAB as at 1 July 2013 is settled, along with the other aspects

⁶³⁰ Queensland Rail, sub. 26, Annexure 3: 10.

⁶³¹ Queensland Rail, sub. 33: 4.

of the tariff methodology, Queensland Rail, access seekers and access holders will benefit from the regulatory certainty that this creates (see Section 8.17 of this Decision).

Treatment of renewal assets and 'zero value'

We consider assets that were installed as capital investments and have a remaining expected useful life should be included in Queensland Rail's regulatory asset base. This will provide Queensland Rail with an opportunity to recover its capital investment in those assets.

Our proposed valuation explicitly includes assets in which Queensland Rail has made capital investments and that have remaining expected useful lives. This includes \$170.8 million for track,⁶³² and \$6.7 million for signals and telecommunications assets—around 70 per cent of the QCA's total valuation at 1 July 2013. All of the value of these assets reflects asset renewal expenditure, since 1983 for rail, and more recently for signals and telecoms.⁶³³

However, where the asset was installed or replaced as a maintenance activity, its value (and the need to continue to 'maintain' it by future replacement) is accommodated by providing Queensland Rail with a substantial maintenance allowance.

If Queensland Rail wishes to capitalise some of its maintenance activities that are more capital in nature, then there may be a case to do so.⁶³⁴ But it would not then be appropriate to also include costs for those activities in the annual maintenance allowance used in the building blocks for the tariff. This applies to a range of assets, including ballast and timber sleepers.

Queensland Rail said in its December 2015 submission that it had been denied an opportunity to know which assets assessed in our asset valuation had been renewed or replaced.⁶³⁵ We have addressed this by providing Queensland Rail and other stakeholders with our West Moreton and Metropolitan network tariff model, that shows how all the assets disclosed by Queensland Rail and its predecessors were treated in the QCA's valuation.

It is misleading to suggest that we have 'zero-valued' assets as stated by Queensland Rail.⁶³⁶ The value of assets that, in the QCA's view, have no remaining expected useful lives, such as tunnels, cuttings and embankments, has been incorporated in the value given to the network as a whole. Assets such as sleepers, that have previously been installed or replaced as maintenance activities (the cost of such maintenance being recovered as part of Queensland Rail's tariff) have continued to be treated as maintenance activities. This is discussed further below.

Queensland Rail said in its March 2016 submission that the Draft Decision tariff model failed to include 'numerous post-1995 assets' and a 'significant one in four steel for wood sleeper replacement program' that was undertaken before 1995.⁶³⁷ Our consultant B&H has reviewed its asset valuation assessment in light of the new information provided by Queensland Rail. We have published B&H's report on the valuation and other matters with this Decision.⁶³⁸ The B&H assessment and our consideration of the appropriate response to Queensland Rail's submission is discussed in more detail below, in the 'QCA Approach' section.

⁶³² The 'track' category includes rails, sleepers, ballast and other related assets, as Queensland Rail does not separately record its spending on these items after 2007. The valuation for rails in place before 2007 is \$87.6 million.

⁶³³ See B&H 2015: 48-56.

⁶³⁴ See Section 8.11 of this Decision.

⁶³⁵ Queensland Rail, sub. 26: 27.

⁶³⁶ Queensland Rail, sub. 26: 27.

⁶³⁷ Queensland Rail, sub. 33: 27.

⁶³⁸ B&H 2016, Part 2.

Service potential

Conceptually all assets, viewed in isolation, can contribute to the overall service potential of the network. But that is not the determining factor of whether a stand-alone value should be given to the asset for the purposes of a valuation in a regulatory context. The regulatory context does not exactly mimic the outcomes of a competitive market—for example, regulation allows a return on and of the value of sunk investments. It follows that a valuation in a regulatory framework occurs in the context of achieving the objectives of the regulatory regime, having regard to the particular circumstances of the assets in question, and need not follow a particular theoretical construct.

Assets with remaining service potential, but expired expected useful lives, include essentially perpetual assets such as tunnels, cuttings and embankments.

Tunnels, cuttings and embankments

The rail infrastructure on the West Moreton network includes tunnels, cuttings and embankments whose characteristics, and therefore service potential, do not change materially over time.⁶³⁹

We have had regard to the remaining service potential of these essentially perpetual assets in arriving at a value for the West Moreton network. The service potential and value of the tunnels, cuttings and embankments is reflected in the value we have given to the network as a whole.

In simple terms, our approach compares the actual lives of Queensland Rail's assets against their expected useful lives. To do otherwise for tunnels, cuttings and embankments would overcompensate Queensland Rail and might provide it with a return in perpetuity.

We consider our approach is consistent with the regulatory treatment by the QCA and other regulators of assets once they have been accepted in the RAB. If we accept an asset in a RAB, the regulated entity earns a return of capital allowance each year (reflecting a loss in economic value). After the invested capital is fully returned, the asset remains in the RAB, but its value is recognised only in the value of the network as a whole. The asset is not re-valued for regulatory purposes, even if it continues to have service potential. To do otherwise would be to provide a regulated entity with windfall gains.

Queensland Rail said little about service potential in its submissions on our October 2015 Draft Decision, except to argue in a report from PwC that 'a regulator ought to ... consider the service potential of the existing assets, not the historic basis on which the relevant assets were developed'.⁶⁴⁰

We again note that the assets the service potential of which is essentially perpetual—the tunnels, cuttings and embankments—have a standard and configuration that would not be built for a modern equivalent asset.⁶⁴¹ However, we note as an aside that, while we have not applied

⁶³⁹ The tunnels, cuttings and embankments differ from assets such as bridges and culverts that, while longlived, do eventually have to be replaced. Cuttings and embankments are similar to tunnels as they receive only incidental maintenance. In this assessment cuttings and embankments are captured within the 'earthworks' asset category. Nevertheless, there are aspects of 'earthworks' that are dissimilar to tunnels, as earthworks receive some maintenance (about 7 per cent of the proposed maintenance budget for vegetation control, drain cleaning and minor earthworks maintenance) and some capital expenditure (about 12 per cent of the proposed capital expenditure budget for formation strengthening works).

⁶⁴⁰ Queensland Rail, sub. 26, Annexure 3: 6.

⁶⁴¹ See QCA 2015: 176.

a DORC valuation, if we were doing so the value of an obsolete asset (i.e. an asset that uses a technology or configuration that is not the modern engineering equivalent) would be zero.

This confirms our view that, while we have had regard to the service potential of the tunnels, cuttings and embankments, it is appropriate to reflect their value in the value we have given to the network as a whole. They have service potential only because of the ancillary infrastructure that operates.

Assets replaced through maintenance

We acknowledge that the situation for shorter-lived assets, such as wooden sleepers and ballast, is different. We are aware that they are still in service and have remaining expected useful lives. However, it is clear that this is because they have been installed or replaced through maintenance costs that have formed (and will, on the QCA's approach, continue to form) a component of the revenue that Queensland Rail has been authorised to derive from the service. Queensland Rail has provided information about past maintenance spending on the West Moreton network over the last 20 years. Over the same period, Queensland Rail has received coal tariff revenue, a large part of which has gone towards that maintenance. Queensland Rail has also proposed significant spending on maintaining these assets in its 2013 and 2015 DAU submissions.

Given these assets have been replaced through maintenance—and generally multiple times we have continued a high maintenance allowance required to keep those assets in service.

To include these kinds of assets in the asset base, while continuing to allow Queensland Rail to recover the cost of replacing these assets as maintenance costs, would result in Queensland Rail receiving a windfall by, in effect, receiving a return of its investment in those assets twice.

Treating the replacement of those assets as maintenance costs rather than including them as part of the asset base does not impose any material disadvantage on Queensland Rail because Queensland Rail receives a return of those assets by way of the maintenance cost allowance. There would generally be limited scope for Queensland Rail to generate a return on those assets because Queensland Rail's maintenance cost will be recovered at more or less the same rate that it is incurred.

The QCA accepts there may be some time difference between incurring the maintenance cost and recovering it, but that difference could apply both ways. That is, at times Queensland Rail may recover its maintenance cost before it is incurred; and at other times Queensland Rail may incur its maintenance cost before it is recovered. However, over the regulatory period, the QCA considers that these timing differences are broadly likely to cancel each other out, so that the overall effect will be that Queensland Rail recovers its maintenance cost at or about the time it is incurred.

Therefore, Queensland Rail does not face a material disadvantage from recovering the cost in the maintenance allowance rather than having the cost included in the asset base to receive a return on and of capital. Moreover, Queensland Rail receives a cashflow benefit from this treatment.

Mindset and evidence

We do not consider that it is necessary to establish the mindset of investors in 1867, in order to draw the reasonable conclusion that providing a return on their tunnels, cuttings and embankments in 2015 would result in windfall gains.

The tunnels, cuttings and embankments have now been there for almost 150 years. We have not seen accounting records to show that they have or have not been fully recovered, and

consider it unlikely that such records exist. However, as Professor Menezes says in his July 2015 report, 'it is neither possible or necessary to consider the mindset of the investor at the time of the investment'.⁶⁴²

The assessment of windfall gains and the adverse effects on allocative efficiency of Queensland Rail's proposed asset valuation is based on economic logic, rather than relying on knowing exactly what investors expected at the time the infrastructure was built. Professor Menezes' analysis shows that a return on assets that have exceeded their expected useful lives yields windfall gains and this increases the risk that allocative efficiency may be affected.⁶⁴³ Professor King emphasises this point, saying that, while including or not including a windfall gain in the opening asset base will not affect future investment by a regulated firm, that does not mean the decision is neutral.

Raising the asset base by including a windfall gain, under building block regulation, will in general lead to higher prices for access seekers and will lead to a lower level of allocative efficiency.⁶⁴⁴

We note that the 'expected useful life' of an asset does not depend on any particular person's expectation but rather is an economic concept that postulates a hypothetical investor making an investment decision based on the period of time over which such asset would deliver a return—that period being referred to as the 'expected useful life'.

Queensland Rail has, in its December 2015 submission, again argued that, because evidence of investors' intentions in the distant past is not available, it is impossible to form a view that the current owner will make a 'windfall gain'.⁶⁴⁵ Queensland Rail's adviser PwC said:

In our view, the current owner of an asset cannot and should not be held responsible for the perceived expectations of some earlier entity that made the initial investment decision.⁶⁴⁶

As Professor Menezes has advised, the exact expectation, and even nature, of the original investor is not relevant to forming a view that a 150-year-old asset is beyond any life over which an investment might be recovered. He said:

for windfall gains to exist it is not necessary that the entity making the original investment had a commercial motivation and that it expected to recover the full costs of the investment, including the return on capital, during the expected useful life of the asset. ... That is, regardless of whether the entity making the investment over a century ago expected to recover the cost of the investment over the life of the asset, it would not have expected to earn any revenue past its expected useful life of the asset.⁶⁴⁷

We consider that it is reasonable for Queensland Rail to receive a return on unrecovered investments. However, we consider that it is correct to reflect the value of assets whose expected useful life has expired in the value of the network as a whole.

Queensland Rail proposal not approved

The pricing principle in section 168A(a) of the QCA Act is that the price of access to a service should generate expected revenue that is at least enough to meet the efficient costs of providing access to the service and a return on investment commensurate with the regulatory and commercial risks involved.

⁶⁴² Menezes, F 2015b: 13.

⁶⁴³ Menezes, F 2015b: 10–11.

⁶⁴⁴ King, S 2016: 5.

⁶⁴⁵ Queensland Rail, sub. 26, Annexure 3: 4–6.

⁶⁴⁶ Queensland Rail, sub. 26, Annexure 3: 5.

⁶⁴⁷ Menezes, F 2016: 7.

However, Queensland Rail's proposed valuation would lead to a price that generates returns that exceed this amount. Whilst such an outcome is not precluded by section 168A(a) alone, it is relevant to other paragraphs in section 138(2). As Professor Menezes says:

[T]he key issue is to ensure that the asset value does not result in access prices that embody monopoly rent.

... a DORC approach that places a positive value on longstanding assets with expired expected useful lives yields a higher return than could have been anticipated by an investor undertaking the initial investment decision. This increases the risk that access prices are sufficiently high to distort competition in relevant markets and impact adversely on investment in coal exploration and production.

Moreover, allowing QR to earn a return on assets with an expired expected useful life would yield windfall gains.⁶⁴⁸

An access price calculated on the valuation proposed by Queensland Rail does not promote the economically efficient investment, operation and use of the infrastructure that provides belowrail services on the West Moreton network and therefore has the potential to adversely affect competition in upstream and downstream markets.

Such a price, encompassing windfall gains and monopoly rents, would distort competition in relevant markets and impact adversely on investment in coal exploration and production as it would be materially higher than the efficient price (ss. 138(2)(a) and (d) and 69E).

Further, it is in the interest of access seekers and holders to pay a return on and of an asset value that does not provide windfall returns (s. 138(2)(e) and (h)). Avoiding windfall gains and monopoly rents is a matter to which we have regard in considering the West Moreton network tariff, including the RAB valuation for the purposes of setting that tariff (s. 138(2)(h)). In Professor Menezes' view:

[W]indfall gains from regulation can be generated by allowing a regulated firm to recover costs associated with assets that, despite having an expired expected useful life, are still functioning.⁶⁴⁹

In particular, it is not in the interest of access seekers, access holders, or their customers, to pay a return based on the value of a modern equivalent asset, when they are also paying for forecast levels of maintenance and replacement capital expenditure forecast that reflect a network that is not modern equivalent standard or condition.

It would also be against the public interest to allow windfall gains as that would have potential to reduce competition and discourage investment in downstream markets such as coal mining and coal tenements, reducing economic growth in Queensland. An expectation that the regulatory regime would allow windfall gains would cause investors to favour projects in jurisdictions where this was not the case, and reduce the chance they would invest their capital in Queensland (s. 138(2)(d)).

Indeed, an excessive price that reduced the incentive for investment in dependent markets that rely on access to Queensland Rail's below rail services will increase the risk to Queensland Rail that the assets in which it has invested to provide the service will be stranded (s. 138(2)(b)).

Therefore, having regard to all the approval criteria in the QCA Act (s. 138(2)), we do not consider it is appropriate to approve the 2015 DAU which contains the asset valuation proposed by Queensland Rail.

⁶⁴⁸ Menezes, F, 2015b: 33–34.

⁶⁴⁹ Menezes, F, 2015b: 31.

QCA required valuation

Our analysis has been informed by the future maintenance and capital expenditure requirement. The QCA has proposed a maintenance allowance and capital expenditure our consultant considers necessary for the aging network to remain operating (see Sections 8.11 and 8.15 of this chapter).

The economic value of a railway network will reflect factors including the expected cost of maintaining and replacing parts of the network to make it fit for purpose and keep it that way. Queensland Rail's proposed near-doubling of its maintenance spending on the West Moreton network in the forthcoming regulatory period, compared with the 2009–13 spending, has been a material matter in our consideration of the 2015 DAU. The ongoing high capital spending, without any increase in capacity, is also relevant to the value of the network that is being replaced through this investment.

New Hope, Yancoal and Aurizon have again pointed to the relationship between asset value and ongoing costs in their December 2015 and March 2016 submissions.⁶⁵⁰ New Hope said that:

it is self-evident that the value of an asset is influenced by its condition, and that high ongoing maintenance and replacement expenditure is clearly indicative of the age and technical obsolescence of the relevant infrastructure and its unsuitability for its current use.⁶⁵¹.

Coal trains are able to use the West Moreton network only because of the high maintenance spend. The QCA has recognised the critical importance of the very high maintenance costs to the viability of the network by including substantial compensation for these costs in the reference tariff.⁶⁵² Aurizon said:

Aurizon considers that in light of the material variation in infrastructure management costs that the QCA has correctly identified two options:

- (1) Retain the original valuation and approve efficient infrastructure management costs consistent with the assumed infrastructure standards; or
- (2) Make appropriate adjustments to the DORC valuation and approve the actual efficient infrastructure management costs which reflect the actual infrastructure standards.

Aurizon supports the QCA's position to take the latter approach to improve the robustness and reliability of the building blocks estimates due to their closer proximity to Queensland Rail's actual costs.⁶⁵³

We broadly agree with Aurizon's assessment. The imbalance between the maintenance and capital costs⁶⁵⁴ and the network valuation might be addressed by reducing those ongoing costs to be consistent with the standard of the network implied by Queensland Rail's proposed value. But in this case, we consider the high maintenance and capital costs are necessary to keep the network operating.⁶⁵⁵

 ⁶⁵⁰ New Hope, sub. 22: 13–14; sub. 32: 21–22; Yancoal, sub. 27: 2; sub. 35: 5; Aurizon, sub. 29: 4–5.
 ⁶⁵¹ New Hope, sub. 22: 14.

⁶⁵² The combined maintenance and forecast capital costs we are minded to accept are about 91 per cent of Queensland Rail's proposed amount (see Sections 8.11 and 8.15 of this Decision).

⁶⁵³ Aurizon, sub. 29: 4-5.

⁶⁵⁴ Queensland Rail's forecast capital expenditure in the 2015 DAU of \$141.9 million is about the same level as its proposed maintenance costs of \$143.0 million (see Sections 8.15 and 8.11 of this Decision).

⁶⁵⁵ We note that B&H has assessed that the efficient maintenance costs for a 'mature' modern equivalent railway facing the same 'severe topography' as the West Moreton network would be approximately \$30,000 per kilometre per year. See B&H 2015: 28.

If the maintenance spending were to be reduced, trains would no longer be able to use the West Moreton network and therefore the assets could not be used. The value of the network is therefore dependent upon the high maintenance costs. Without such high levels of maintenance, Queensland Rail could not generate revenue from the network.

Some of the assets that are particularly maintenance-intensive are wooden sleepers and ballast, each of which require regular replacement.⁶⁵⁶ The cost of replacing those assets has previously been allowed as a maintenance cost rather than as part of the asset base, and the QCA has continued that approach in this Decision.

Other assets require only incidental further work once they have been built. Key among these on the West Moreton network are the tunnels, cuttings and embankments, all of which were completed by the time rail services to Toowoomba began in 1867. The tunnels, cuttings and embankments are essentially perpetual, and do not depend on significant maintenance to remain in service. Indeed, they are now akin to a natural feature of the landscape and will most likely have the same remaining service potential in 100 years as they have now, with no further capital expenditure and subject to them continuing to form part of a viable rail network.

Tunnels, cuttings and embankments depend on the other assets that are integral to a rail network, such as rail and sleepers, to have any value at all. Without tracks going through them and trains carrying goods or passengers on those tracks, the tunnels, cuttings and embankments of and by themselves have no value. The question for an economic regulator (using a building block approach to setting a tariff) revolves around the value of the allowance that should be given in the tariff for investment in tunnels, cuttings and embankments.

The tunnels, cuttings and embankments on the West Moreton network were built almost 150 years ago and the accounting records to enable a DAC valuation are not available. We have had to form a judgment about the valuation of the tunnels, cuttings and embankments having regard to how long ago they were built, the fact that they have exceeded their expected useful lives and the interests of stakeholders including Queensland Rail. In these circumstances, the tunnels, cuttings and embankments are reflected in the value of the network as a whole.

We have therefore developed a RAB valuation that recognises the investments Queensland Rail has made in its network, both before and after coal services began in the mid-1990s. Similar to a DAC approach, we have assessed the value based on actual costs reported by Queensland Rail where possible. In other cases, our valuation approach uses aspects of a DORC methodology, particularly for older assets such as steel rails, that still have remaining expected useful life, but for which actual costs are not available. Accordingly, our valuation approach:

- (a) gives separate value to assets such as rail, concrete sleepers and concrete bridges that have been replaced as capital items and have not reached the end of their expected useful lives;
- (b) takes account of and gives value to assets such as wooden sleepers, fences, ballast and wooden bridges through the allowed maintenance expenditure; and
- (c) gives value to tunnels, cuttings and embankments through the value given to the network as a whole.

This approach provides a valuation that is appropriate having regard to all the criteria in section 138(2).

⁶⁵⁶ B&H 2015: 23–25 for sleepers and 20 for ballast.

In forming our view for this Decision, we have had regard to all the submissions and comments from Queensland Rail and other stakeholders, including those provided in December 2015 and March 2016 after our October 2015 Draft Decision.

Among these, we have considered Queensland Rail's March 2016 submission on renewal assets that Queensland Rail suggests the QCA failed to include in its Draft Decision valuation.⁶⁵⁷

The QCA has, during the course of its detailed assessment of the asset base proposals in Queensland Rail's June 2013 DAU and 2015 DAU, considered the information provided in material accompanying the DAUs. We also asked Queensland Rail (through requests during the assessment of both DAUs) to provide further information on the capital expenditure that Queensland Rail claimed for inclusion in the opening asset valuation.

In general, the material considered by the QCA, including Queensland Rail's information provided in its initial submissions and responses to information requests, was detailed in respect of the years before 2007, but less detailed in respect of the period after 2007, in that the post-2007 capital projects were described in terms of categories of assets, rather than specific assets.⁶⁵⁸ The QCA, in its Draft Decision valuation, largely adopted Queensland Rail's figures for capital expenditure incurred after 1995 (save in respect of assets with expired expected useful lives and assets which are recovered by way of maintenance costs as discussed above).⁶⁵⁹

Following the issue of the Draft Decision, Queensland Rail argued that there were a number of assets which should have been included in the opening asset base.⁶⁶⁰ However, these were not specifically identified in its earlier submissions and responses to information requests.

We engaged our rail technical consultant B&H to advise us to the extent possible, based on the evidence provided and other relevant information that was available, whether those assets had been included in our Draft Decision valuation.⁶⁶¹

In particular, Queensland Rail referred to a program of steel re-sleepering from before 1995, but provided no further evidence as to the timing, scope, standard or cost of the work, beyond saying the re-sleepering had taken place before 1995.⁶⁶²

B&H considered that the pre-1995 steel sleepers cited by Queensland Rail were 'maintenance and opex funded'.⁶⁶³ The QCA notes that Queensland Rail will receive and has received an allowance that provides for such maintenance and operating costs.

Queensland Rail also submitted to the QCA a list of assorted assets from after 1995, including fences, telephone equipment, three concrete culverts, a steel culvert, six axle counters and a concrete bridge, that it said we had failed to include in our Draft Decision valuation.⁶⁶⁴ Queensland Rail did not provide any further evidence in its March 2016 submission that would enable the QCA to assess whether or not these specific assets were included in aggregate

⁶⁵⁷ Queensland Rail, sub. 33: 27; sub. 33, Att. 4.

⁶⁵⁸ This detailed information on the pre-2007 assets was from an asset register provided to the QCA as part of QR Network's western system tariff proposal in 2008.

⁶⁵⁹ B&H was able to use the information from the 2008 asset register to identify specific assets from before 1995, that Queensland Rail did not identify in its submissions, but which we included in our valuation.

⁶⁶⁰ Queensland Rail, sub. 33: 27.

⁶⁶¹ For further details, please see B&H 2016, Part 2.

⁶⁶² Queensland Rail, sub. 33: 27.

⁶⁶³ B&H 2016, Part 2: 11–12.

⁶⁶⁴ Queensland Rail, sub. 33, Att. 4.

capital spending information it had already submitted (and which we had largely adopted in the Draft Decision valuation).

B&H assessed in some detail the post-1995 assets listed by Queensland Rail—this assessment is set out in Table 2 in Part 2 of B&H's Supplementary Report, with specific responses for all the assets cited by Queensland Rail in Attachment 4 to its March 2016 submission.⁶⁶⁵

B&H found that, on the balance of probability, all assets cited in Queensland Rail's Attachment 4 except one had been included in the QCA's opening asset valuation.⁶⁶⁶ B&H was unable to undertake a further assessment because Queensland Rail's earlier information was aggregated into categories whereas its subsequent information provided in March 2016 was about specific assets (and Queensland Rail had not confirmed whether these particular assets were or were not contained in the earlier, less-specific information it provided). Queensland Rail has provided no explanation for why it failed to identify the value of these assets in its earlier submissions and responses to information requests.

In the circumstances, the QCA is satisfied that it has made a proper allowance and in particular is not satisfied that it has not allowed a value for those assets specified in Attachment 4 to Queensland Rail's March 2016 submission.

The one exception is a 'concrete rail bridge' listed by Queensland Rail as being installed in April 2011. B&H determined that the bridge was installed as part of repairs to the track east of Toowoomba after major flooding in 2011, and provided an initial estimate that the bridge had a value of about \$1 million (in \$2013).⁶⁶⁷

In relation to the concrete bridge installed in 2011, the QCA accepts that if the bridge was not part of the earlier information that Queensland Rail provided, it is entitled to an allowance for this asset. However, at this stage, Queensland Rail has not provided sufficient information for the QCA to properly assess the value of the bridge.

The QCA accepts that one option is for us to simply accept a value for the concrete bridge based on the limited information provided to us. However, we consider that a better option is for Queensland Rail to make a submission under the capital expenditure approval process in Schedule E of the undertaking, once approved, for the bridge to be included in the regulatory asset base.

This submission would need to give sufficient information for the QCA to fully assess the value of the asset, and to determine whether it was appropriate to include in the asset base. Relevantly, the capital expenditure process provides that, if the QCA finds that the scope, standard and cost of the capital project is prudent, the asset will be accepted into the regulatory asset base.

Accordingly, our proposed value for the West Moreton common network between Rosewood and Columboola, at 1 July 2013, is \$254.5 million.

Decision

The review of the RAB valuation in light of the ongoing high maintenance and capital spending has been an important element of this regulatory process and the QCA has been alert to and had regard to the interests of Queensland Rail that arise from the review and resulting asset valuation (s. 138(2)(b)). In particular, the QCA is satisfied its required valuation will allow a

⁶⁶⁵ B&H 2016, Part 2: 13–17.

⁶⁶⁶ B&H 2016, Part 2: 13–17.

⁶⁶⁷ B&H 2016, Part 2: 11.

return on investment that is commensurate with the regulatory and commercial risks of providing access for West Moreton network coal services (s. 168A(a)). However, the QCA is not satisfied that the review has given rise to regulatory uncertainty for Queensland Rail that arises for consideration.

To allow a return on investment by which access holders and end users were paying for both high maintenance spending and for assets that were renewed through that maintenance spending, or paying for assets that had reached the end of their effective useful lives, would overcompensate Queensland Rail for the regulatory and commercial risks of providing access. This would not promote economically efficient investment in the West Moreton network or competition in relevant markets, including those for above-rail haulage, coal production and coal tenements (ss. 138(2)(a) and 69E). Such a return will increase the risk to Queensland Rail that the assets in which it has invested to provide the service will be stranded (s. 138(2)(b)).

Taking account of and giving value to assets (that have been funded through maintenance allowances) by the future maintenance allowances, and reflecting the value for tunnels, cuttings and embankments in the value given to the network as a whole will provide incentives for Queensland Rail to efficiently invest in its network and promote competition in relevant markets, including those mentioned above. The required valuation is in the public interest as it will not discourage investment in relevant markets which would reduce economic growth in Queensland (s. 138(2)(a) and (d)).

The required valuation is in the interests of access seekers and holders in paying a return on and of an efficient investment value that is commensurate with Queensland Rail's regulatory and commercial risks of providing access (s. 138(2)(e) and (h)).

We have also had regard to the effect of excluding existing assets for pricing purposes. The QCA's view is that its approach, in valuing the network as a whole, does not exclude existing assets for pricing purposes, which have been taken into account in the way discussed above. However, even if the proper reading of the words 'the effect of excluding existing assets for pricing purposes' in section 138(2)(f) would mean that the approach we have taken does involve excluding existing assets for pricing purposes, that is a matter to which we have had regard, and having done so, we would not come to a different decision.

Avoiding windfall gains and monopoly rents, as the valuation does, is a matter the QCA considers relevant in considering the regulatory asset base valuation (s. 138(2)(h)). On this basis and having regard to the criteria in section 138(2), the QCA's Decision is that it is not appropriate to approve the 2015 DAU proposed by Queensland Rail and the undertaking should be amended to apply a regulatory asset base value of \$254.5 million for the West Moreton common network between Columboola and Rosewood, at 1 July 2013.⁶⁶⁸

Summary 8.11

The 2015 DAU is to provide for an opening asset value of \$254.5 million for the West Moreton common network between Columboola and Rosewood, as at 1 July 2013.

⁶⁶⁸ This figure is rolled forward to derive the opening asset value for the tariff period beginning 1 July 2016 (see Section 8.17 of this Draft Decision).

8.14 Past capital expenditure

Queensland Rail's 2015 DAU proposed opening asset value at 1 July 2015 included claims for:

- historical capital expenditure incurred during 2007–08 to 2012–13 (historical capital expenditure); and
- capital expenditure incurred during 2013–14 and 2014–15 (pre-2015 DAU capital expenditure).

Historical capital expenditure

In the 2015 DAU, Queensland Rail included a new claim for capital expenditure it incurred during 2007–08 to 2012–13 on the West Moreton network. Queensland Rail said the underlying capital projects were triggered by freight (i.e. non-coal) services, but were on the common network, and referred to it as transport service contract (TSC) capital. Queensland Rail said the government funded the return on, and of, the TSC capital. Queensland Rail's rolled forward value of TSC capital at 1 July 2013 was \$17.9 million.⁶⁶⁹

Our 2015 Draft Decision said that it was reasonable to consider investments on the shared network that benefitted all traffics as a common network capital expenditure. However, our Draft Decision also stated that investors need only receive a return on, and of, their investments once to avoid windfall gains. Since Queensland Rail was receiving a return on, and of, the TSC capital from the government, we did not consider the TSC capital for the purposes of deriving the reference tariff for coal-carrying train services in the West Moreton network.⁶⁷⁰

Stakeholders' submissions

Queensland Rail rejected our treatment of the TSC capital.⁶⁷¹ Aurizon suggested exercising due diligence when considering capital expenditure as common network costs.⁶⁷²

QCA analysis and Decision

Our decision is to consider the TSC capital in the West Moreton common network regulatory asset base, as nearly all of the TSC capital benefits all traffics.

However, our Decision is not to include the returns on, and of, the TSC capital for the purposes of deriving the coal reference tariffs. This is because Queensland Rail is receiving a return on, and of, the TSC capital from the government and including the TSC capital returns in the building block revenue requirements for recovery from coal services will amount to double recovery by Queensland Rail of those investment costs. We do not consider it appropriate for Queensland Rail to receive more than once a return on, and of, its investments.

Our assessment and treatment of TSC capital

Queensland Rail said that the 2015 Draft Decision treatment of not considering the TSC capital in the West Moreton common network asset base meant that common network investments

⁶⁶⁹ Queensland Rail, sub. 2: 35, 49; 2015g: 2.

⁶⁷⁰ QCA 2015: 181–182.

⁶⁷¹ Queensland Rail, sub. 26: 34–39. In a letter of 29 April 2016, Queensland Rail stated it was concerned that aspects of the TSC arrangement had been treated incorrectly in our 2015 Draft Decision and informed that it will provide a further submission to the QCA 'as soon as possible'. However, on 16 May 2016, Queensland Rail advised that it was not putting in a further submission. It is unknown what aspects of our Draft Decision did Queensland Rail have concerns with. In any event, in its December 2015 submission, Queensland Rail raised concerns on the TSC capital matter and we have considered them in making this Decision.

⁶⁷² Aurizon, sub. 20: 25.

triggered by non-coal users were allocated solely to those users. Queensland Rail argued this approach was inconsistent with the QCA's earlier view of allocating all common network investments between coal and non-coal users, regardless of who underwrote the investment. Queensland Rail said that the Draft Decision approach would prevent it from fully recovering the costs of its investments.⁶⁷³

We have considered Queensland Rail's 2015 DAU and reviewed our 2015 Draft Decision treatment of the TSC capital in light of Queensland Rail's and other stakeholders' comments. There are two issues:

- Is TSC capital expenditure a common network expenditure?
- How to treat government support for TSC capital?

We reiterate our earlier view that incremental investment on the network shared by different traffics could benefit all traffics. For example, a project to improve the track standard will result in increased reliability and lower maintenance requirement and will benefit all traffics. Therefore, it is reasonable to consider such investment as a common network capital expenditure and to allocate it amongst the different classes of users.⁶⁷⁴ We have also considered Aurizon's comments that not all investments on the West Moreton network may benefit coal and non-coal services.⁶⁷⁵

Given these considerations, to the extent the TSC capital is on the shared network and benefits all traffics it should be considered in the West Moreton common network RAB.

We engaged B&H to independently assess Queensland Rail's TSC capital expenditure claim. B&H considered that nearly all of the TSC capital was on the shared network and benefitted all traffics.⁶⁷⁶ We are satisfied with B&H's assessment and we have considered the TSC capital in the West Moreton common network RAB.⁶⁷⁷ Furthermore, we have allocated a share of the TSC capital to coal services for the purposes of determining coal reference tariffs in this Decision (see Summary 8.2 for our consideration of coal allocation of fixed common network costs).

We also reiterate our earlier view that investors need only receive a return on, and of, their investments once to avoid windfall gains.⁶⁷⁸ Queensland Rail is receiving a return on, and of, the TSC capital from the government, and it should not seek to also receive the TSC capital returns from access holders/seekers. On that basis and for the purposes of determining coal reference tariffs in this Decision, if returns related to the part of the TSC capital allocated to coal services were included in the building blocks revenue requirements for recovery from coal traffics, it will amount to double recovery of those investment costs. A double recovery will not promote efficient investments in the network, will not be in the public interest and the interests of access seekers and access holders, and it is not appropriate having regard to the assessment criteria in section 138(2) of the QCA Act.

⁶⁷³ Queensland Rail, sub. 26: 34–38.

⁶⁷⁴ QCA 2015: 181–182.

⁶⁷⁵ Aurizon, sub. 20: 25. Aurizon stated: 'For example, tunnel deepening may not provide any direct benefit to coal carrying train services and is therefore not a common cost.'

⁶⁷⁶ B&H identified that around 99 per cent of the TSC capital was on the common network (B&H 2016, Part 1: 9–13).

⁶⁷⁷ The rolled forward value of the TSC capital at 1 July 2013 is \$18.4 million. This is more than Queensland Rail's July 2013 value of \$17.9 million because we have considered higher asset lives for certain asset classes (see Section 8.16 of this Decision).

⁶⁷⁸ QCA 2015: 182.

Our assessment shows that the regulatory returns related to the TSC capital, based on regulatory WACC and regulatory asset lives, are less than the returns Queensland Rail receives from the government. Therefore, for the purposes of determining coal reference tariffs, we have deducted from coal revenue requirements, government-funded returns related to the part of the TSC capital allocated to coal services, subject to a cap of the regulatory returns, as any additional return Queensland Rail receives is a matter for Queensland Rail.

Our Decision prevents Queensland Rail from double recovering the efficient investment costs related to the TSC capital allocated to coal services, and is appropriate having regard to section 138(2) of the QCA Act.

Pre-2015 DAU capital expenditure

In the 2015 DAU, Queensland Rail included a claim of \$39.3 million for capital expenditure carried out in the West Moreton network during 2013–14 and 2014–15.⁶⁷⁹

Our 2015 Draft Decision proposed to accept as prudent \$37.7 million of that capital expenditure, having regard to our consultant's (B&H) advice.⁶⁸⁰

New Hope supported our 2015 Draft Decision about Queensland Rail's claim for past capital expenditure.⁶⁸¹

Since stakeholders did not raise concerns regarding our Draft Decision about the pre-2015 DAU capital expenditure, we accept as prudent \$37.7 million of the pre-2015 DAU capital expenditure. We have allocated a share of this capital expenditure to coal services for the purposes of deriving the West Moreton coal reference tariff in this Decision (see Summary 8.2 for our consideration of coal allocation of fixed common network costs).

Summary 8.12

For the treatment of past capital expenditure, 2015 DAU is to provide that:

- (a) the regulatory return on, and of, the TSC capital allocated to coal services be deducted from the capital charges for the coal regulatory asset base, for the purposes of deriving the reference tariff.
- (b) \$37.7 million of the capital expenditure incurred during 2013–14 and 2014–15 is considered in the West Moreton common network regulatory asset base.

8.15 Forecast capital expenditure

In the 2015 DAU, Queensland Rail proposed a capital indicator process and made a provision for \$141.9 million in forecast capital expenditure for the five-year period July 2015 to June 2020, for reflecting in its proposed ceiling price.⁶⁸²

Our 2015 Draft Decision proposed forecast capital expenditure of \$144.2 million, which was greater than Queensland Rail's proposal due to some maintenance activities being treated as

⁶⁷⁹ Queensland Rail, sub. 2: 36. The amount reported here includes interest during construction.

⁶⁸⁰ QCA 2015: 182–184. The amount reported here includes interest during construction.

⁶⁸¹ New Hope, sub. 22: 16.

⁶⁸² Queensland Rail's proposed prudency assessment process—where prudency of capital expenditure is assessed for scope, standard and costs for the expenditure to be included in the RAB—is considered separately in Section 8.9 of this Decision.

capital works that outweighed our proposed lower expenditure for some aspects of the capital program (for example, slope stabilisation works on the Toowoomba range).

Stakeholders' submissions

Queensland Rail had significant concerns with aspects of B&H's assessment of forecast capital expenditure and provided new information in support of its 2015 DAU proposal.⁶⁸³

Miners supported B&H's assessment of Queensland Rail's forecast capital expenditure and said they relied on the QCA to assess the prudency of actual capital expenditure.⁶⁸⁴

QCA analysis and Decision

Our Decision accepts forecast capital expenditure of \$137.9 million for the five-year period July 2015 to June 2020 for the purposes of the capital indicator, which is 97 per cent of Queensland Rail's proposal of \$141.9 million.

We will assess the planned capital works in detail through the prudency assessment process in the approved undertaking. A key aspect of that assessment will be to determine whether Queensland Rail explored the feasibility of alternative solutions.

Approach to assessing forecast capital expenditure

We engaged B&H to review its previous assessment of forecast capital expenditure in light of stakeholders' comments, including new information provided by Queensland Rail in its post-Draft Decision submission.

B&H retained its previous assessment relating to aspects of forecast capital expenditure on which Queensland Rail raised concerns. For example, B&H did not accept:

- Queensland Rail's claim of a higher expenditure for:
 - slope stabilisation works on the Toowoomba range due to past works and considering that Queensland Rail was yet to commence the proposed work that was subject to further analysis by Queensland Rail; and
 - level crossing reconditioning program due to lack of information about the proposed scope and considering that a substantial expenditure was planned for a related activity (level crossing compliance).⁶⁸⁵
- Queensland Rail's argument that ballast undercutting should not be capitalised, as B&H considered that the scope proposed by Queensland Rail was actually a reconstruction of track, which was a capital activity.

However, as noted in the discussion on maintenance costs, B&H considered treating rail renewal as a maintenance activity rather than as a capital item to be appropriate, given the smaller scope and lower cost submitted by Queensland Rail in its December 2015 submission.

Ultimately, B&H's revised assessment resulted in a total forecast capital expenditure of \$137.9 million for the five-year period July 2015 to June 2020.⁶⁸⁶

⁶⁸³ Queensland Rail, sub. 26: 47 and Annexure 2.

⁶⁸⁴ New Hope, sub. 22: 16 and sub. 32: 24; Yancoal, sub. 27: 2 and sub. 35: 4–5.

⁶⁸⁵ B&H 2016, Part 1: 6–8.

⁶⁸⁶ This is equivalent to the B&H recommended total forecast capital expenditure of \$124.4 million (June 2015\$), which excluded capitalised interest and inflation adjustment.

QCA Decision

We accept that Queensland Rail needs to make capital improvements to its ageing railway to maintain the integrity of its network. We are satisfied that B&H's assessed \$137.9 million is a reasonable estimate of Queensland Rail's future capital program for the five-year period from July 2015 to June 2020.

We consider a capital indicator that reflects a reasonable assessment of the planned capital expenditure is in the interests of all parties. It would promote efficient investment in the network and prevent tariffs being mis-specified due to inappropriate projections of capital works. Therefore, we accept the \$137.9 million capital indicator, having regard to the assessment criteria in section 138(2) of the QCA Act.

However, we also note B&H's observation that Queensland Rail has proposed capital expenditure without rigorously evaluating alternative solutions. Given this, while we accept \$137.9 million for the purposes of the capital indicator, we will subsequently assess the capital works in detail through the prudency assessment process in the undertaking. A key aspect of that assessment will be to determine whether Queensland Rail has appropriately explored the feasibility of alternative solutions.

As considered in Section 8.3.2 of this Decision, we treat the entire planned capital expenditure as common network capital expenditure as it is on the shared network and would benefit all traffics. Therefore, as per Summary 8.2 (Section 8.3.3 of this Decision), we have allocated about 71 per cent of the forecast capital expenditure to coal train services, reflecting the ratio of 80/113 paths⁶⁸⁷, and used the resultant \$97.3 million for the purposes of deriving the West Moreton coal reference tariff.⁶⁸⁸

As considered in Section 8.3.3 of this Decision, our approach to allocating costs to coal traffics is appropriate, having regard to the assessment criteria in section 138(2) the QCA Act.

2032 embargo on coal trains

Our 2015 Draft Decision noted a statement in Queensland Rail's May 2015 submission which indicated that coal trains will not continue through the Metropolitan network beyond 2032. Our Draft Decision observed that Queensland Rail's capital and maintenance programs did not recognise this 2032 embargo on coal trains and noted that 'our preliminary view, subject to stakeholders' further comments, is to assess Queensland Rail's proposed capital program on the basis that coal transport will continue beyond 2032'.⁶⁸⁹

In its post-Draft Decision submission, Queensland Rail said that there was no 2032 embargo on coal trains and that its management of the West Moreton network and Metropolitan network was not based on a '2032 embargo on coal trains'.⁶⁹⁰

New Hope said it was prudent for the QCA to assess Queensland Rail's capital program on the basis of operations continuing beyond 2032. However, New Hope suggested reviewing it in the next undertaking assessment in the event that the 2032 embargo remained in place.⁶⁹¹

⁶⁸⁷ As considered in Section 8.3.3 (Change in circumstances) the fixed cost allocator for 2015–16 is 69.5 per cent.

⁶⁸⁸ This compares to the \$133.0 million Queensland Rail considered for deriving the ceiling price for West Moreton coal services in the 2015 DAU.

⁶⁸⁹ QCA 2015: 186–187.

⁶⁹⁰ Queensland Rail, sub. 33: 38.

⁶⁹¹ New Hope, sub. 31: 17–18.

We note that the 2032 end date for coal trains is an extension of the previous 2024 end date and that Queensland Rail's capital and maintenance programs are not affected by this end date. We also note that both Queensland Rail and coal miners do not consider it as a constraint and it has not affected their planned operations at this stage. Taking all this into account, we consider this matter is best left for consideration in the next undertaking period when there is likely to be greater clarity on whether the embargo will remain in place.

8.16 Capital charges for the coal RAB

One of the pricing principles is that the price of access to a service should generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved (s. 168A(a)).

The QCA and other regulators seek to achieve this by applying the financial capital maintenance (FCM) principle (also known as the NPV=0 principle). This principle means that investors in regulated monopoly infrastructure have the opportunity to receive returns on, and of, their capital investment and have an incentive to make economically efficient investments in the future. FCM achieves this by requiring that the present value of the expected future cash flows, including returns on and of capital, equals the amount invested to provide the regulated service.⁶⁹²

Following the setting of the RAB, we are able to determine the efficient capital charges to be included within approved maximum allowable revenue limit. This involves components for:

- a return on capital—based on a WACC applied to a RAB; and
- a return of capital—based on a suitable depreciation method.

Return on capital

Identifying an appropriate rate of return is important to setting ceiling prices for access charges for coal-carrying train services that operate on the West Moreton network.

As discussed in Section 3.7 of this Decision, Queensland Rail used an indicative WACC of 6.93 per cent in developing its proposed ceiling price.⁶⁹³

In this Decision, we have applied two different WACCs.

For assessing the 2013–16 tariff, we have used the WACC of 6.93 per cent that was proposed by Queensland Rail in its 2013 and 2015 DAUs, and was used to assess prices in the QCA's 2014 and 2015 Draft Decisions. This was based on time-variant WACC parameters assessed over the 20 business days immediately before 1 July 2013.⁶⁹⁴ Queensland Rail said risk-free rates were higher in previous periods.⁶⁹⁵ This matter is discussed in Section 3.7 of this Decision. The tariff relevant to the 2013–2016 period is discussed in greater detail in Part C of this chapter.

For the purpose of Queensland Rail's proposed rate of return for coal reference tariffs from 1 July 2016, we have used a WACC of 5.73 per cent per annum, based on time-variant WACC parameters assessed over the 20 business days beginning 12 March 2016. This is outlined in more detail in Section 3.7 of this Decision.

⁶⁹² For a more detailed discussion of FCM and the NPV=0 principle, see QCA, 2014a.

⁶⁹³ Queensland Rail, sub. 2: 39–41.

⁶⁹⁴ QCA, 2014d:142–143.

⁶⁹⁵ Queensland Rail, sub. 26: 30.

Return of capital

We endorse Queensland Rail's proposed approach of using straight-line depreciation based on asset lives that reflect the physical lives of the assets.

Given the age profile and estimated lives of the West Moreton network below-rail infrastructure, we have formed the view that asset lives should be measured in terms of their physical lives. This is capped at 100 years, which is the Australian standard design life for bridges,⁶⁹⁶ and the expected useful life for essentially perpetual assets such as tunnels, cuttings and embankments.

We also consider that straight-line depreciation is a reasonable approximation of the actual time profile for the asset valuation of the West Moreton network's infrastructure.

We note New Hope's suggestion that if a 'tariff arising from a pure building blocks methodology is not appropriate', one way of addressing this would be to apply an alternative depreciation profile.⁶⁹⁷ However, we do not find this is necessary as the building block tariff set out below is appropriate having regard to the approval criteria in section 138(2) of the QCA Act.

Accordingly, we have decided to approve asset lives for the purposes of calculating straight-line depreciation charges as outlined in Table 20 below.

Asset class	Asset life (years)		
Track	35		
Roads	38		
Fences	20		
Signals	20		
Bridges	100		
Tunnels	100		
Culverts	100		
Earthworks	100		
Land acquisition costs	50		
Telecommunications	20		
Other	20		

Table 20 Approved asset lives⁶⁹⁸

QCA-approved capital charges

Based on our decisions above, we approve capital charges to be included within the maximum allowable revenues used to derive the ceiling price for coal-carrying train services as outlined in Table 21 below.

⁶⁹⁶ B&H 2014: 77.

⁶⁹⁷ New Hope, sub. 22: 17.

⁶⁹⁸ These asset lives apply to capex since 2007. The asset lives that apply to assets before 2007 are the valueweighted remaining lives based on B&H's asset value assessment (B&H 2015: 49).

Table 21 Capital charges⁶⁹⁹

\$m	2016/17	2017/18	2018/19	2019/20
Capital charges	\$14.2	\$15.6	\$17.0	\$18.2

For tax depreciation, we used Queensland Rail's estimates and calculation method with two exceptions, where data was insufficient.⁷⁰⁰

Summary 8.13

The 2015 DAU is to reflect the West Moreton network capital charges as per Table 21 above and include a WACC, for assessing the 2013-2016 tariff, of 6.93%; and, for the purpose of Queensland Rail's proposed rate of return for coal reference tariffs from 1 July 2016, include a WACC of 5.73%.

See definition of 'WACC' in Appendix F.

8.17 QCA's required reference tariffs

Queensland Rail's 2015 DAU proposed a ceiling price of \$34.92/'000 gtk and a reference tariff of \$19.41/'000 gtk as at 1 July 2015.

For the reasons set out above, having regard to the approval criteria in section 138(2) of the QCA Act, we do not consider it appropriate to approve either the ceiling price or reference tariff proposed by Queensland Rail.

We have instead derived a reference tariff at a ceiling price we have assessed based on inputs for the four years beginning 1 July 2016. In applying the mechanisms discussed above in Sections 8.3 to 8.6 of this Decision to the volumes, building blocks and capital charges discussed above in Sections 8.10 to 8.16, we have considered a number of related issues, including:

- the tariff period;
- the roll-forward of the asset base; and
- corrections and changes to the tariff model

These matters are explained in turn below, before we:

- summarise the way the West Moreton reference tariff is derived;
- assess whether the tariff complies with the pricing limits we require in the undertaking; and
- set out why our required prices are appropriate with regard to section 138(2).

Tariff period

Queensland Rail's 2015 DAU proposed a tariff for the five years beginning 1 July 2015, and a ceiling price that it based on building blocks forecasts for those years. Our October 2015 Draft Decision also proposed a tariff starting on 1 July 2015 (for indicative purposes).

⁶⁹⁹ These capital charges were calculated as the sum of return on capital (based on the 1 July 2016 WACC) and return of capital, less inflationary gain based on the coal-allocated regulatory asset base roll forward, and less the capital charge based on the coal-allocated share of TSC capital (Section 8.14 of this Decision).

⁷⁰⁰ The exceptions were the tax depreciation for the Western System Asset Replacement (WSAR) project and the TSC capital.

However, in this Decision on the 2015 DAU, we have derived a tariff over the four years beginning 1 July 2016. For the purposes of the adjustment amount mechanism discussed below, we have also derived a tariff for the three years beginning 1 July 2013 that, for the 2015–16 financial year, uses building blocks inputs based on our assessment of forecasts for that year from the 2015 DAU—this is discussed in more detail in Section 8.20 below.

This approach is appropriate for a number of reasons, including:

- (a) The approval date for the new undertaking is most likely to be during the 2016–2017 financial year, which aligns with a tariff derived starting at the beginning of that period.
- (b) The WACC for the 2015 DAU was set based on time-variant parameters assessed over the 20 business days beginning 12 March 2016. We consider it appropriate to use that WACC for the forward-looking period—that is, the period starting 1 July 2016. In this regard, there is also a WACC that, based on Queensland Rail's conduct, was expected to be used for the period beginning 1 July 2013, that we have applied for that earlier period (see Sections 3.7, 8.18 and 8.20 of this Decision).
- (c) The period over which the adjustment charge process applies is expected to be mostly covered by the tariff relating to the 2013–2016 period.

We consider that a tariff derived over the period starting 1 July 2016 is likely to advance the legitimate business interests of Queensland Rail as it aligns with the period over which the WACC has been set and provides a transparent and readily understood approach for calculating the tariff (s. 138(2)(b)). For the same reasons, the 2016–20 tariff will promote the efficient investment in and use of the rail infrastructure and be in the interest of access seekers and access holders (s. 138(2)(a), (e), (h)). We therefore consider it appropriate to apply this period, having regard to all the approval criteria in section 138(2).

Asset base roll-forward

Queensland Rail's 2015 DAU proposed an asset value at 1 July 2013, then rolled it forward using forecast capital expenditure to derive an opening asset base at 1 July 2015.

We have applied the same methodology, but rolled forward the common network asset base to 1 July 2016, using:

- (a) our assessed opening asset base at 1 July 2013 of \$254.5 million (see Section 8.13 of this Decision);
- (b) our assessed prudent capital expenditure for 2013–14 and 2014–15 of \$37.7 million (see Section 8.14); and
- (c) our assessed capital indicator of \$26.0 million for 2015–16 (see Section 8.15)

This gives a common network opening asset value of \$312.4 million at 1 July 2016.

Corrections and changes to the model

We made some changes to the tariff model used in the Draft Decision to address inconsistencies and minor errors and reflect new information that became available. These included:

 (a) changing the compounding and discounting rate from the inflation rate (used by Queensland Rail in its model) to WACC for the operating expenditure and maintenance components in the building blocks calculation and the NPV formula. Using WACC reflects the opportunity cost of capital and is consistent with the discounting of other costs in the building blocks;

- (b) updating the 2014–2015 inflation rate with the actual ABS CPI data, rather than the assumed CPI of 2.5 per cent for indexing operating expenditure that is based on Queensland Rail's 2012–13 below-rail financial statements and rolling forward the asset base. This means that a consistent inflation rate is applied for both the escalation of the tariff, and the escalation of the building blocks components. The escalation by actual CPI would have been done in the post-approval tariff indexing process if the undertaking had been approved earlier;
- (c) fixing a formula error in reproducing the 1995-2007 historical developer contribution and AFD-funded capital expenditure data—this had a non-material effect on the 1995–2007 historical capital expenditure value;
- (d) addressing an inconsistency in the data source when splitting the B&H pre-2007 RAB values into pre-1995 assets and 1995–2007 capital expenditure—this had a non-material effect on the coal-allocated values of pre-1995 and 1995–2007 assets; and
- (e) addressing an inconsistency with the 2015 DAU in the 2007–08 capital expenditure roll forward (we used 11 months for 2007–08 in our Draft Decision, and are now using 12 months, which is the same as the 2015 DAU)—this had a non-material effect on the capital expenditure values.

Tariff summary

In assessing the West Moreton tariff, we have applied:

- (a) an allocation to coal traffics of:
 - (i) 70.8 per cent for fixed costs including post-1995 assets, forecast capital expenditure and fixed maintenance and operating costs (see Section 8.3.3 of this Decision);
 - (ii) 58.4 per cent for pre-1995 assets (Section 8.4.1); and
 - (iii) about 98 per cent for variable maintenance and operating costs (Section 8.3);⁷⁰¹
- (b) forecast weekly demand of 62.8 weekly train paths for coal and 3 for non-coal services (Section 8.10);
- (c) maintenance costs allocated to coal of \$71.0 million over the four years from 1 July 2016 to 30 June 2020;
- (d) operating costs allocated to coal of \$22.6 million over the four years;
- (e) a capital indicator allocated to coal of \$79.2 million over the four years;
- (f) an opening coal asset base of \$223.0 million (at 1 July 2016); and
- (g) a WACC of 5.73 per cent for the period from July 2016 to June 2020.

The above mechanisms and building blocks give a tariff for the West Moreton network of \$17.92/'000gtk, as at 1 July 2016, split into \$3,011/train path and \$8.96/'000gtk.

⁷⁰¹ Different allocators for variable costs were used for the Rosewood to Jondaryan and Jondaryan to Columboola sections to reflect coal's share of forecast gtk volume in each of those two sections.

The tariff for 1 July 2013 to 30 June 2016 is discussed below in Part C of this Chapter 8. As discussed in Section 8.6 above, the Metropolitan network tariff is escalated from the tariff that would have applied in the period starting 1 July 2013—it is therefore addressed in the discussion in Part C below. The CPI-escalated Metropolitan network tariff at 1 July 2016 is \$16.66/'000 gtk, split into \$1,149/train path and \$8.33/'000gtk.

All the tariffs are set out in Appendix A of this Decision, which also provides further information on aspects of the building block model.

Compliance with pricing limits

We have assessed the West Moreton network tariff against the pricing limits we require in the undertaking (see Section 3.4 of this Decision). We note that New Hope has asked that we confirm that services from the Cameby Downs mine (Columboola loading loop) will be expected to pay access charges that cover at least their incremental costs.⁷⁰² We confirm that this is the case.

QCA Decision

We consider it appropriate with regard to the approval criteria in section 138(2) of the QCA Act to require a West Moreton network tariff of \$17.92/'000 gtk as of 1 July 2016. Our consideration of the approval criteria includes:

- (a) the object of Part 5 (s. 138(2)(a))—our Decision promotes efficient investment in, use of, and operation of, Queensland Rail's declared infrastructure, which will have the effect of promoting effective competition in upstream and downstream markets by removing regulatory uncertainty about an appropriate access price;
- (b) the legitimate business interests of Queensland Rail (s. 138(2)(b), (c))—Queensland Rail continues to recover coal allocated efficient costs from forecast coal services, including returns reflecting the option value of coal paths not forecast to be contracted by coal traffics during the 2015 DAU period;
- (c) the public interest (s. 138(2)(d))—our Decision promotes the future development of mines and the above-rail market by signalling to customers that they will not have to pay for assets that have reached the end of their expected useful lives, or pay access charges that include the costs for capacity they are unable to contract;
- (d) the interests of access seekers and users of the West Moreton network (s. 138(2)(e),
 (h))—they are not required to pay reference tariffs that include costs reflecting capacity they are unable to contract;
- (e) the effect of excluding assets for pricing purposes (s. 138(2)(f))—we have not excluded assets for pricing purposes. We have reflected the value of assets that have value only because of maintenance in the maintenance allowance and we have reflected the value for tunnels, cuttings and embankments in the value given to the network as a whole;
- (f) the pricing principles (ss. 138(2)(g) and 168A))—although our Decision does not generate expected revenue from coal train services that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved (having regard to the adjustment discussed below), that is an appropriate outcome having regard to all of the relevant considerations; and

⁷⁰² New Hope, sub. 22: 19; see also Section 8.5.4 above.

(g) other relevant matters (s. 138(2)(h))—we are preserving regulatory certainty for stakeholders by requiring an adjustment for previous revenue over-recovery by Queensland Rail, consistent with the expectations generated by Queensland Rail's previous statements. We have also had regard to the avoidance of monopoly profits and windfall gains, and to the interests of access holders.

Summary 8.14

The 2016–2020 reference tariffs in the 2015 DAU are to provide that:

- (a) West Moreton network tariff components are consistent with our reference tariff of \$17.92/'000gtk, as at 1 July 2016; and
- (b) Metropolitan network tariff components are consistent with our reference tariff of \$16.66/'000 gtk, as at 1 July 2016

The other components of the tariffs are specified in Appendix A of this Decision.

See Schedule D, clause 3.1(e) in Appendix F.

PART C-TARIFF ADJUSTMENT

We have considered the appropriate approach to addressing the expectations of stakeholders about an adjustment to reflect the difference between the access charges that Queensland Rail has in fact received since 1 July 2013 and the access charges that it would have received if the reference tariff approved in this Decision had applied since that date.

8.18 Adjustment amount expectations

The 2008 access undertaking was extended by the QCA to 30 June 2015,⁷⁰³ as an interim position and on the basis that each of the (now withdrawn) replacement DAUs proposed by Queensland Rail provided for the new tariffs approved by the QCA to apply from 1 July 2013. In addition, the now withdrawn replacement DAUs included an adjustment charge provision which provided, in effect, for recovery or refund (as applicable) of the difference in access charges paid by access holders since 1 July 2013 and the access charges that would have been paid if calculated in accordance with the new reference tariff approved by the QCA ('adjustment amount'). The adjustment charge provision required Queensland Rail to recover or reimburse the adjustment amount by making adjustments to future access charges to be paid by access holders.⁷⁰⁴

In contrast to its previous DAUs, Queensland Rail's 2015 DAU proposes to apply the new tariff approved by the QCA from the date of approval of the new undertaking, without the kind of adjustment referred to above.

The QCA has determined that the access charges that have been received by Queensland Rail since 1 July 2013 significantly exceed the access charges that it would have received if calculated on the basis of the reference tariff now proposed to be approved. If the new access undertaking (which contains the QCA's views on the appropriate cost build up for setting regulated tariffs) did not contain an adjustment mechanism, the QCA's indicative estimate is that Queensland Rail has received approximately \$32 million more than it would have received if an adjustment were applied.⁷⁰⁵

In this section, the QCA considers the impact of the change in Queensland Rail's approach, stakeholders' submissions with respect to that issue and whether, in these circumstances and having regard to the criteria in section 138(2), it is appropriate to approve the 2015 DAU. In doing so, the QCA has reviewed its Draft Decision position on this matter in light of stakeholder comments and further information provided by Queensland Rail.⁷⁰⁶

Background

An adjustment amount has been received by Queensland Rail and its predecessors in the past through approved access undertakings through the operation of an Adjustment Charge provision. This provision operated to reflect the difference between access charges paid by reference to interim (extended) and approved tariffs (although Queensland Rail says it did not

⁷⁰³ This was given effect to by a series of extensions approved by the QCA.

⁷⁰⁴ See for example Schedule A to the February 2013 DAU.

⁷⁰⁵ Our estimate with an indicative ±10% sensitivity range.

⁷⁰⁶ Queensland Rail has submitted further information to the QCA following the issuing of a s. 185 notice under the QCA Act.

retain the benefit of the adjustment amount). Queensland Rail says it has never paid an adjustment amount in the past.⁷⁰⁷

Nevertheless, this method of addressing delays in finalising tariffs in an approved replacement undertaking has been used both when the adjustments were in rail companies' favour and when they favoured access holders. These adjustments were applied in:

- 2006, to refund to customers the difference between access charges paid from July 2005 to June 2006 and the access charges that would have been paid pursuant to the reference tariff approved for that period by the QCA in QR Network's 2006 undertaking; and
- 2010, to recoup from customers the difference between access charges paid from July 2009 to June 2010 and the access charges that would have been paid pursuant to the reference tariff approved for that period by the QCA in June 2010 amendments to QR Network's 2008 undertaking. The adjustment was applied both by QR Network for tariffs in the central Queensland coal network and Queensland Rail for the West Moreton network coal tariffs.

Queensland Rail proposed an adjustment amount (to operate as though the new reference tariff applied from 1 July 2013) in all of its (now withdrawn) voluntary DAUs to replace the 2008 undertaking—each of Queensland Rail's March 2012, February 2013 and June 2013 DAUs provided for such an adjustment.

These adjustment amounts in the voluntary DAUs were further confirmed by Queensland Rail's letters accompanying its May 2013, November 2013 and May 2014 extension DAAUs, each of which said it intended to apply its tariffs from 1 July 2013 through adjustment charge provisions in its replacement DAU. Aurizon has noted that in its 2014 Annual Report Queensland Rail said that its approved reference tariffs would be backdated to 1 July 2013.⁷⁰⁸

In October 2014, the QCA released its Draft Decision on the June 2013 DAU. On 12 December 2014, Queensland Rail withdrew the June 2013 DAU.

The 2008 access undertaking expired on 30 June 2015.

Queensland Rail's 2015 DAU proposal

Queensland Rail submitted its 2015 DAU on 5 May 2015. Queensland Rail does not propose to apply an adjustment charge provision in the 2015 DAU in relation to the difference in access charges it has received since 1 July 2013 and the access charges it would have received applying the reference tariff that would have applied from that date. In other words, Queensland Rail does not propose an adjustment amount to address any previous over or under recovery of access charges.

Stakeholders' comments on the 2015 DAU prior to the Draft Decision

On 15 May 2015, QCA staff published a staff paper inviting stakeholders to provide further information in response to a set of questions relating to Queensland Rail's proposal not to apply the tariff approved by the QCA with effect from 1 July 2013 through adjustment charge provisions in its replacement DAU.

⁷⁰⁷ Queensland Rail, sub. 26: 13–14.

 ⁷⁰⁸ Aurizon, sub. 20: 11; Aurizon, sub. 29: 6. See page 32 of the Queensland Rail 2014 Annual Report at https://www.queenslandrail.com.au/about%20us/Documents/QueenslandRail_AnnualFinancialReport_2013 -14_FINAL.pdf.

Five submissions were received on the 2015 DAU prior to the Draft Decision.⁷⁰⁹

In addition to raising procedural concerns (which the QCA addressed in the Draft Decision), Queensland Rail said that the QCA could not require an adjustment amount pursuant to the Act. Queensland Rail provided an opinion from Corrs Chambers Westgarth dated 29 May 2015. In that opinion, it is stated that the QCA cannot, as a matter of law, compel Queensland Rail to apply a retrospective reference tariff for a number of reasons, including the Transfer Notice issued under section 9(1)(j) of the Infrastructure Investment (Asset Restructuring and Disposal Act 2009 (Qld)).

In contrast, stakeholders said an adjustment amount was:

- not inconsistent with the pricing principles (s. 168A) in the QCA Act; and
- one of a range of factors that were relevant in the context of the QCA's approval criteria in section 138(2).

New Hope, Yancoal and QRC (on behalf of its Members) said they had expected that an adjustment amount would be included in the approved access undertaking and indicated that this expectation arose from the conduct of Queensland Rail.⁷¹⁰ Stakeholders referred to Queensland Rail's extension DAAUs in May 2013, November 2013 and May 2014 (which retained adjustment charge provisions) and accompanying letters. The QRC said its Members had advised that their expectations had been formed on the basis of repeated assurances from Queensland Rail's senior executives during meetings held throughout the development of Queensland Rail's DAUs. New Hope quoted extracts of correspondence from Queensland Rail indicating that the tariffs would apply from 1 July 2013.⁷¹¹ Aurizon, New Hope and QRC said adjustments were consistent with past practice in previous regulatory periods, with New Hope referring to the adjustments being applied in 2006 and 2010.

New Hope said adoption of Queensland Rail's proposal would be a departure from regulatory precedent and demonstrate that a regulated entity can manipulate the timing of an undertaking process for financial gain.⁷¹² New Hope also submitted that if the QCA considered that a lower tariff than Queensland Rail was currently charging should apply, a failure to apply the reference tariff from 1 July 2013 would result in a windfall gain to Queensland Rail, being a return in excess of that which the QCA assesses to be commensurate with the regulatory and commercial risks involved.⁷¹³ Yancoal also referred to a windfall gain arising from a monopoly service provider delaying the finalisation of its undertaking.

Stakeholders said that the impact of Queensland Rail's proposal was a lack of regulatory certainty, in Yancoal's case likely to lead to its shareholders reassessing the risks associated with investing in Queensland and being a part of the West Moreton network, and an increase in sovereign risk. Yancoal noted its view that this cannot be in the public interest. Yancoal said that should the proposal be adopted, its Cameby Downs mine would be in serious jeopardy.

The QRC said that, in the context of access holders being in the process of weighing up major investment decisions in capacity expansions, uncertainty and associated contingent costs would need to be factored into the decision making of those companies as a direct result of Queensland Rail's conduct. New Hope noted that regulatory certainty is in the public interest

⁷⁰⁹ Queensland Rail, sub. 4; Aurizon, sub. 6; New Hope, subs. 8–12; QRC, sub. 14; Yancoal, sub. 16.

⁷¹⁰ New Hope sub. 12: 4; Yancoal sub. 16: 5; QRC sub. 14:2.

⁷¹¹ New Hope, sub. 12: 4.

⁷¹² New Hope, sub. 12: 6.

⁷¹³ New Hope, sub 12. 8.

and that increased regulatory uncertainty is likely to lead to existing or potential coal miners reassessing the risks associated with new investment in the West Moreton network and a reduction in potential competition for rail haulage services with material flow on impacts.

Aurizon too said that it expected an adjustment, based on the representations given by Queensland Rail.⁷¹⁴

The consistent theme of the stakeholder submissions was that stakeholders expected an adjustment amount and if an access undertaking was now approved without the inclusion of such an adjustment amount, there would be a detrimental impact on regulatory certainty and consequential reassessment of regulatory risk in the future, with adverse investment impacts. For instance, stakeholders said:

[t]hese factors will have a significant detrimental effect on regulatory certainty and the resulting increase in sovereign risk may be sufficient to reassess the risks of business relating to the QR network and conversely the attractiveness of opportunities outside the QR network⁷¹⁵

[t]here is a lack of regulatory certainty and a significant increase in sovereign risk in Queensland. It is also worth noting that the change in QR's position from its previous representation has caused Yancoal to be sceptical of many aspects of QR's Undertaking⁷¹⁶

[c]learly all current users of the West Moreton system have now been put on notice and Queensland Rail's written commitments may be reversed at any time. Similarly, all future and prospective users of the system will now have to factor in an escalated risk premium into their calculations to reflect the apparently ephemeral nature of the commitments from Queensland Rail.⁷¹⁷

QCA's Draft Decision

The QCA's Draft Decision was to not approve the DAU and to ask Queensland Rail to amend the DAU to (amongst other things) provide for an adjustment amount for West Moreton network (i.e. west of Rosewood).

Stakeholders' comments after the Draft Decision

Queensland Rail continues to maintain that the QCA cannot require an adjustment amount pursuant to the Act, including asserting that it is retrospective and does not comply with the pricing principles.

Queensland Rail said:

- the adjustment amount would mean that the QCA would set a price that does not comply with the principle in section 168A(a);⁷¹⁸ and
- the retroactive effect of the proposed tariff cannot be avoided by stating the reference tariff will only apply from the date of the approval of the 2015 DAU as one must look at the substance, not the form, of the QCA's proposal to assess its true effect.⁷¹⁹

In contrast, stakeholders said an adjustment amount is:

• not retrospective; and

⁷¹⁴ Aurizon, sub. 6: 8–9.

⁷¹⁵ New Hope, sub. 12: 6.

⁷¹⁶ Yancoal, sub. 16: 5.

⁷¹⁷ QRC, sub. 14: 3.

⁷¹⁸ Queensland Rail, sub. 26: 12–13.

⁷¹⁹ Queensland Rail, sub. 26: 16

• consistent with the proper application of the QCA Act as it is one of a range of factors that is relevant in the context of the QCA's approval criteria in section 138(2).

New Hope, Yancoal and Aurizon continued their submissions that a failure to include an adjustment amount would lead to increased regulatory risk and uncertainty.⁷²⁰

New Hope, Yancoal and QRC all said this would lead to a reassessment of regulatory risk in the future, both in the West Moreton network and Queensland in general. For instance, stakeholders said:

[I]f an adjustment amount is not ultimately provided for that will be such a substantial and unwarranted change to the regulatory framework (and Yancoal's expectation of how it would operate based on both the provisions of the current undertaking and QR's previous representations) that the resulting regulatory uncertainty will necessarily be taken into account when Yancoal and its shareholders are considering future investment in Cameby Downs (in comparison to other mines within the Yancoal portfolio for which this issue does not exist)⁷²¹

[T]he material variation from applied regulatory precedent without adequate and reasonable substantiation of a material change in circumstances is likely to have negative implications for complimentary investment by rail operators and end customers in the future⁷²²

The fact that QR has attempted to do this has increased NHC's assessment of the risks of investing in this region ... NHC's assessment of investment opportunities in this region, including on the New Acland extension project (on which a decision has to be made over the coming year) would then be assessed on a basis akin to having a high sovereign risk rating.⁷²³

In contrast, Queensland Rail said that an adjustment amount would create regulatory uncertainty because:

[i]t shows that the QCA may change material aspects of its regulatory approach from access undertaking to access undertaking affecting the ability of Queensland Rail and other stakeholders to invest and operate; and there is no certainty in the currently regulatory process as to what the price will ultimately be⁷²⁴

Moreover, Queensland Rail submits:

stakeholders would have been aware that a voluntary draft access undertaking can be withdrawn at any time and, therefore, would have had the knowledge and understanding that any provisions in it could be changed and therefore would not rely on them in making investment decisions; similarly stakeholders would have been aware that the QCA may also refuse to approve a voluntary draft access undertaking with the result that none of the proposed provisions have any regulatory effect.⁷²⁵

Second Request for Comments

On 19 January 2016, QCA staff published a staff paper requesting that stakeholders make further comment in response to the submissions made.⁷²⁶

⁷²⁰ Aurizon, sub. 20: 11, New Hope, sub. 21: 5, 20-21, Yancoal, sub. 27: 1.

⁷²¹ Yancoal, sub. 27: 1

⁷²² Aurizon, sub. 20: 11

⁷²³ New Hope, sub no. 21: 5

⁷²⁴ Queensland Rail, sub. 26: 14.

⁷²⁵ Queensland Rail, sub. 26: 14.

⁷²⁶ QCA 2016e.

Stakeholders' comments after second Request for Comments

Queensland Rail retained its view that there is no legal, commercial or regulatory basis for the adjustment amount⁷²⁷ and said, among other things:

- Queensland Rail has never in the past received any benefit from any adjustment amount as those amounts were paid to Aurizon Network;
- The adjustment amount was only one element of an overall suite of methodologies, assumptions and forecasts used to determine reference tariffs; and
- Stakeholders would have been aware that a voluntary DAU could have been withdrawn at any time and therefore would not rely on them in making investment decisions. The 2013 DAU was only a draft document and subject to change or even withdrawal.⁷²⁸

New Hope retained its view that the adjustment amount was appropriate and reiterated that the approval of the 2015 DAU without an adjustment amount would undermine confidence in the regulatory regime and have negative impacts on investment. Among other things, New Hope said:

- While it is true that stakeholders would have been aware that a voluntary DAU could be withdrawn at any time, stakeholders were not aware that Queensland Rail would renege on its commitments;
- Queensland Rail's position that it was willing to provide an adjustment amount as a package
 of measures meant that Queensland Rail 'is opposed to an adjustment amount because it is
 seeking to 'offset' what it considers is a worsening of its position in other areas. That in itself
 should be enough to demonstrate the inappropriateness of QR's position';
- Any future mining investment on the West Moreton network would need to be considered in the context that it is reliant on a monopoly service provider that is willing to extract monopoly rents; and
- Evidence that the disincentive is real will exist only after investment decisions have been taken.⁷²⁹

New Hope also submitted legal advice from Brian O'Donnell QC which said that the adjustment amount is not retrospective. Mr O'Donnell's advice states that, among other things.

an approved access undertaking cannot commence to operate prior to the time of its approval. But once approved, the terms of the access undertaking can regulate matters between the access provider and the user by reference to events that occurred prior to the time of the approval.⁷³⁰

Like New Hope, QRC and Yancoal reiterated their earlier views that Queensland Rail reneging on its commitments to provide an adjustment amount has undermined investment.⁷³¹

Aurizon too reiterated its earlier views on the need for an adjustment amount and said, among other things:

While it may be correct that Queensland Rail did not receive the adjustment amount approved by the QCA, this is of little practical relevance to customers of the same declared service who paid

⁷²⁷ Queensland Rail, sub: 33: 27.

⁷²⁸ Queensland Rail, sub: 33: 28–29.

⁷²⁹ New Hope, sub: 31: 8-9.

⁷³⁰ New Hope, sub: 32: Schedule 1: 3.

⁷³¹ QRC, sub: 34: 2-3; Yancoal sub: 35: 2.

those amounts. The horizontal separation of Queensland Rail from the Central Queensland Coal Network (CQCN) has not resulted in a material change in circumstances ... ⁷³²

QCA analysis and Decision

We have considered Queensland Rail's 2015 DAU and stakeholder submissions in accordance with our obligations in the QCA Act.

Based on the considerations set out in this chapter and having regard to the relevant factors in section 138(2) of the QCA Act, our Decision is that:

- it is not appropriate to approve the 2015 DAU; and
- it is appropriate to amend the 2015 DAU to include an adjustment amount provision which provides for the amount that Queensland Rail receives after commencement of the undertaking to be adjusted by reference to the difference between access charges in fact paid between 1 July 2013 and the date when the adjustment amount is calculated, and the access charges that would have been paid for that period if calculated on the basis of the reference tariff that would have applied in that period (i.e. that amount being the adjustment amount).

Adjustment amount is not beyond power

The QCA considers the pricing principles, including section 168A(a), to be fundamental considerations, but that does not mean they have primacy over other considerations and it does not mean that it is necessary for them to be 'complied with'. It is open to the QCA to consider that a DAU which provides for a price that allows a service provider to recover at least the efficient costs of providing access to the service and a relevant return on investment, is, including by reference to other factors such as the object of Part 5 of the QCA Act (s.138(2)(a)), the interests of access seekers and holders (s. 138(2)(e) and (h)) and the public interest (s. 138(2)(d)), not one which is appropriate to approve.

This is discussed generally in Chapter 10 and further in relation to the adjustment amount below.

The QCA does not accept Queensland Rail's position that an adjustment amount is retrospective. The QCA accepts stakeholder views that the QCA's reference tariff is prospective and will come into effect from the date the access undertaking is approved. The QCA is of the view that an adjustment amount is to be paid by Queensland Rail after the 2015 DAU is approved and merely takes matters that have occurred in the past as the basis for calculating that amount. The fact that such a clause would operate by reference to things that have happened in the past, does not make it retrospective.

The core of Queensland Rail's argument is that the QCA has proposed an adjustment amount:

in a way that retroactively affects Queensland Rail's access price revenue earned in a previous regulatory period, where that revenue was earned in a manner that was consistent with reference tariffs previously approved by the QCA.⁷³³

It makes similar arguments in addressing the proposition that Queensland Rail will receive a windfall gain from its change in position by referring to Queensland Rail being entitled to set access charges based on QCA approved reference tariffs under the 2008 access undertaking.⁷³⁴

⁷³² Aurizon, sub: 29: 4.

⁷³³ Queensland Rail sub. 26: 5 (fourth dot point) and as elaborated at 16.

Reference tariffs for the period from 1 July 2013 to 30 June 2015 have arisen from extensions to the 2008 access undertaking granted by the QCA. The reference tariffs for those two years were not approved by the QCA following thorough regulatory consideration, but were CPI-escalated tariffs approved on the expectation, arising from the conduct of Queensland Rail, that new tariffs approved by the QCA following an investigation under the QCA Act would operate as though they applied from 1 July 2013. Since 1 July 2015 there have been no QCA-approved reference tariffs under the 2008 access undertaking.

Queensland Rail argues that the 'clear and undeniable effect of the QCA's proposal is to retroactively alter and set aside Queensland Rail's accrued rights for the provision of the declared service to users'.⁷³⁵ However, this is not correct. Queensland Rail has charged its customers under their access agreements and the QCA's decision has no impact on those contractual matters in the past. The QCA is concerned with future considerations, including promoting future efficient use and investment in the network and dependent markets by preserving regulatory certainty that will arise from giving effect to the expectation of access holders and access seekers that an adjustment amount would be included in the approved access undertaking.

Transfer Notice

Queensland Rail says the 2008 access undertaking continues to apply given the operation of the Transfer Notice and the Transfer Notice specifically refers to the reference tariffs under the 2008 access undertaking continuing to apply. Queensland Rail said it is lawfully entitled to be paid, and is obliged to set, access charges based on this access undertaking, until a replacement access undertaking is approved by the QCA. Queensland Rail submits the QCA cannot as a matter of law override, or retroactively alter, the requirements of the Transfer Notice.⁷³⁶

The QCA is not satisfied that the Transfer Notice validly indefinitely extends the term of the 2008 access undertaking. In addition, for the reasons outlined above, the QCA does not accept that an adjustment amount is retrospective.

Relevant considerations under section 138(2)

The QCA may approve Queensland Rail's 2015 DAU, only if it considers it appropriate to do so having regard to each of the factors in section 138(2) of the QCA Act.

The factors in section 138(2) which the QCA considers are of particular significance to the consideration of the issues addressed in this section of this chapter are:

- the object of Part 5 of the Act, to promote the economically efficient operation of, use of and investment in, significant infrastructure by which services are provided, with the effect of promoting effective competition in upstream and downstream markets (ss. 138((2)(a) and 69E);
- the legitimate business interests of Queensland Rail (s. 138(2)(b));
- the public interest, including through:
 - regulatory certainty for long-lived infrastructure assets; and

⁷³⁴ Queensland Rail sub. 26: 17.

⁷³⁵ Queensland Rail sub. 26: 16.

⁷³⁶ Queensland Rail, sub. 26: 17.

- preventing a regulated entity from benefiting from delays in the regulatory process to which it has contributed (s. 138(2)(d));
- the interests of persons who may seek access to the service, namely access seekers and holders who have expected, based on Queensland Rail's representations and actions, that there would be an adjustment amount in the new access undertaking (s. 138(2)(e));
- the pricing principles in section 168A (s. 138(2)(g));
- any other issues we consider relevant (s. 138(2)(h)). The QCA considers that other relevant matters here include:
 - Queensland Rail's previously stated intention to include an adjustment amount;
 - the expectation of stakeholders of the inclusion of an adjustment amount;
 - the disappointed expectations of stakeholders if an adjustment amount is not included in the approved undertaking; and
 - the impact of the change in Queensland Rail's position with respect to the inclusion of an adjustment amount in its 2015 DAU.

As in practice the owner and operator of the service are the same, the considerations in section 138(2)(c) are considered within section 138(2)(b). The QCA has also had regard to section 138(2)(f), but notes that the adjustment amount does not exclude assets for regulatory pricing purposes.

The object of Part 5

As noted above, stakeholders' submissions are that, based on Queensland Rail's previously stated commitment, they expected Queensland Rail to include an adjustment amount which had the effect of reconciling access charges paid since 1 July 2013 and that if a new DAU was approved without such a mechanism, regulatory certainty would be adversely impacted with the result that stakeholders would reassess the regulatory risk attaching to future investments.

The QCA commissioned Professor Flavio Menezes as an independent economic consultant to explore the likely impacts of the absence of an adjustment amount in the next access undertaking in circumstances where Queensland Rail had previously stated its intention to include such an amount. His view is that access seeker investment decisions can be adversely affected by regulatory risk. In particular, Professor Menezes said:

There are at least two ways in which QR's proposal may increase regulatory uncertainty. First, it may create a perception amongst access seekers that the regulatory process favours QR through a 'heads I win, tails you lose' situation. That is, access seekers have no certainty that QR will apply these arrangements symmetrically. Indeed, QR's proposal reflects an increased likelihood that the new tariff will be lower than the existing, interim tariff, and that the proposal would not have been put forward if tariffs were likely to increase instead.

Second, deviating from the expectations of including an adjustment [amount] to refund or recoup differences in the tariffs, in a way that benefits QR, can also increase the perceived risk associated with the overall regulatory framework.⁷³⁷

Professor Menezes' report is available on the QCA's website.

Following the release of the Draft Decision, Queensland Rail engaged PWC to review Professor Menezes' conclusions.

⁷³⁷ Menezes, F 2015c: 2.

PWC had, among other things, concerns with Professor Menezes' 'conceptual model', including the distinction between 'interim' and 'final' tariffs.⁷³⁸ Professor Menezes did not accept these concerns and said his model was developed to:

examine, conceptually, what would happen to the incentives for investment faced by access seekers resulting from.... the difference between the temporary (or interim) tariff and the final tariff ... [not being] recovered from or refunded to access seekers.⁷³⁹

PWC also said that Professor Menezes presented no evidence to suggest that access seekers have had investment proposals delayed or cancelled or that access seekers made investments with the expectation of a tariff adjustment.

Again, Professor Menezes did not accept these concerns. He said:

[t]he economic analysis of the impact of QR's proposal is necessarily prospective in nature ... The mechanisms ... include the creation of asymmetric risk—the perception that the regulatory process favours QR—and a perception of an unstable regulatory process. Both mechanisms may adversely impact future investment. This analysis is indeed theoretical in nature. It was aimed at identifying what the impact of the proposal will have on future investment based on what economic theory would predict.⁷⁴⁰

PWC briefly also raised several other issues, which Professor Menezes has also addressed.

After Professor Menezes was appointed to the QCA Board on 8 April 2016, the QCA commissioned Professor Stephen King to undertake an independent peer review of Professor Menezes' analysis and conclusions. Professor King said Professor Menezes' approach in his 2015 report on the impact of the absence of an adjustment amount is 'both economically rigorous and balanced'.⁷⁴¹

Professor King said:

[T]he concern highlighted by Professor Menezes is that the failure to provide a symmetric approach to tariff adjustment in the current regulatory period will create a 'concern' by investors that such a failure will also occur in future regulatory periods. It is this risk of 'regulatory opportunism' that will lead to the distortion to investment and the dynamic inefficiency. Professor Menezes correctly concludes that the economic impact of QR not making an adjustment for tariff over-recovery, in a situation where access seekers expected such an adjustment, is a potential adverse impact on future investment ... I agree with both his analysis and his conclusion.⁷⁴²

Both Professor Menezes' subsequent report and Professor King's independent peer review report are available on the QCA's website.

The QCA has reviewed PWC's report and ultimately prefers the views expressed in Professor Menezes' reports relating to economic uncertainty and the adjustment amount (as affirmed by Professor King).

Regulatory certainty for rail access is an important underpinning of investments made in longlived infrastructure investments and expenditure on exploration activities. Indeed, uncertainty about pricing can result in a lessening of competition for upstream coal tenements (limited exploration and mine development expenditure) and inefficient use of Queensland Rail's West

⁷³⁸ Queensland Rail, sub. 26, Annexure 3 (Response to Professor Menezes' reports, PWC)

⁷³⁹ Menezes, F 2016b: 15.

⁷⁴⁰ Menezes, F 2016b: 15.

⁷⁴¹ King, S 2016: 8.

⁷⁴² King, S,2016: 8–9.

Moreton network rail infrastructure (by discouraging new entrants from taking any spare rail capacity). Regulatory certainty is consistent with the object of Part 5 of the QCA Act.

Specifically, if customers cannot rely on regulatory arrangements to provide certainty, they will be less willing to make future investments in long-lived sunk investments or undertake exploration activities to develop prospective tenements Both of these have implications for economic efficiency. In this regard, New Hope has previously noted the value it places on regulatory certainty:

[I]t is essential to establish a transparent and repeatable methodology for determining reference tariffs for the Western System. We believe this will enable greater predictability of tariffs, and hence improving understanding and management of the associated cost risk.⁷⁴³

Darryl Biggar has explained this issue as follows:

[U]sers of a monopoly service must typically take some irreversible action which increases the value of the monopoly service — such as the decision of a large gas consumer to locate close to a gas-transmission pipeline, or the decision of a factory to install electrical wiring on its premises. The value of such investments is contingent on continuing to receive access to the monopoly service at reasonable prices and quality. Such an investment is therefore at risk of expropriation through an increase in the price or a decrease in the quality of the monopoly service. The fear of such expropriation has a chilling effect on such investment, reducing overall economic welfare.⁷⁴⁴

Queensland Rail does not accept that its 2015 DAU creates any pricing uncertainty.⁷⁴⁵ However, the QCA disagrees. For example, New Hope says:

NHC considers that the approval of a DAU without appropriate Adjustment Amounts would demonstrate that QR is able to manipulate the regulatory regime to extract from its customers excessive charges to which QR has no rightful claim. The fact that QR has attempted to do this has increased NHC's assessment of the risks of investing in this region. A demonstration that the regulatory arrangements can be effective in preventing such a misuse of QR's position would go some way towards restoring confidence, while a failure of regulation in this case would extinguish any confidence in the regulatory regime and would extinguish regulatory certainty.⁷⁴⁶

Queensland Rail considers that the QCA's draft decision to impose an adjustment amount creates regulatory uncertainty and refers to the QCA's Draft Decision as a 'marked change in regulatory approach to the setting of reference tariffs'.⁷⁴⁷ However, the inclusion in the 2015 access undertaking of an adjustment amount mechanism is consistent with outcomes in 2006 and 2010, which provided for an adjustment amount to address any over or under recovery of revenues. The QCA's approach promotes regulatory certainty by giving effect to the expectation of access holders and access seekers that the approved undertaking would provide for an adjustment amount consistent with Queensland Rail's representations.

Investment impacts

Regulatory certainty about Queensland Rail's actions is particularly important given market conditions and planned investments.

For example, New Hope said it:

[h]as choices about where its money is invested ... It is critical that a properly calculated adjustment charge be applied in order to avoid creating a strong disincentive to further

⁷⁴³ New Hope 2013: 10.

⁷⁴⁴ Biggar, D, 2010.

⁷⁴⁵ Queensland Rail, sub. 26: 15.

⁷⁴⁶ New Hope Coal, sub. 21: 5.

⁷⁴⁷ Queensland Rail, sub. 26: 15.

investment in the West Moreton system. Evidence of this disincentive is likely to be available only after investment decisions have been taken and investment has been lost. However, we submit that the disincentive effective is self-evident, while the counterfactual (that NHC will be no less willing to invest in a mine which depends on a monopoly service provider which can misuse the regulatory regime to extract material excess charges) is clearly implausible.⁷⁴⁸

Also, Yancoal indicated that it has plans to expand its Cameby Downs mine from 1.4 million to 4 million tonnes per annum⁷⁴⁹, but indicated that increasing regulatory uncertainty was causing its shareholders to reassess the risks of investing in Queensland. As extracted above, Yancoal noted:

if an adjustment amount is not ultimately provided for that will be such a substantial and unwarranted change to the regulatory framework (and Yancoal's expectation of how it would operate based on both the provisions of the current undertaking and QR's previous representations) that the resulting regulatory uncertainty will necessarily be taken into account when Yancoal and its shareholders are considering further investment in Cameby Downs (in comparison to other mines within the Yancoal portfolio for which this issue does not exist).⁷⁵⁰

The QCA notes that such regulatory certainty (or uncertainty) may also impact on other pending future investments.

For example, New Hope is proposing to spend \$896 million on its New Acland Stage 3 development, which is expected to extend the life of the mine from 2017 to 2029. New Hope said:

The revised Project will directly support approximately \$6.6 billion in economic output from construction/capital and operational expenditure, while indirect and induced output will contribute a further \$12 billion for a total output impact of almost \$19 billion.⁷⁵¹

Likewise, the QCA's Draft Decision noted that Sekitan Resources has made a conditional offer to acquire the Wilkie Creek mine from Peabody for US\$75 million in cash and assumed liabilities⁷⁵² with plans to resume production in 2016.⁷⁵³

Consistent with Professor Menezes' report, it is reasonable to expect that such investments could be affected by regulatory uncertainty.

Relevantly, such investment is directly relevant to section 138(2)(a) as it relates to the efficient operation of, and use of investment in the West Moreton network, which has faced substantial declines in utilisation. A lack of future investment by miners means increasingly inefficient utilisation of Queensland Rail's infrastructure as substantial spare capacity will remain, which then impacts on the viability of the remaining mines—that is, potentially creating a 'death spiral'.

This chapter further discusses the impact of reduced volumes on the proposed tariff for the West Moreton network as part of proposing a tariff methodology.

Legitimate business interests of Queensland Rail

Section 138(2)(b) of the QCA Act relates to the legitimate business interests of the owner or operator of the service.

⁷⁴⁸ New Hope, sub. 21: 5–6.

⁷⁴⁹ Yancoal sub. 16: 4.

⁷⁵⁰ Yancoal, sub. 27: 1.

⁷⁵¹ http://www.newhopegroup.com.au/files/files/NHG0040_ExecSummary_NAP_EIS_WebRes_Ex.pdf.

⁷⁵² http://www.peabodyenergy.com/investor-news-release-details.aspx?nr=893.

⁷⁵³ http://www.internationalcoalnews.com/storyview.asp?storyID=826952237.
The legitimate business interests of Queensland Rail include its commercial interest in recovering its costs of providing the service and earning a return on investment.

Queensland Rail had previously submitted the March 2012, February and June 2013 DAUs as well as successive extension DAAUs on the basis that there would be an adjustment amount for access charges back to 1 July 2013. Queensland Rail said that it 'has been prepared to offer retrospective application of tariffs as is evident from its 2012 DAU'.⁷⁵⁴

However, in the context of the release of the QCA's October 2014 Draft Decision, Queensland Rail said that an adjustment amount was not in its legitimate business interest. Queensland Rail said its:

position only changed following the QCA's foreshadowed significant change to the long-standing regulatory approach to asset valuation and after the material, negative impact of that change on Queensland Rail's legitimate business interests became evident.⁷⁵⁵

Queensland Rail repeats:

The QCA's claim that stakeholders relied on Queensland Rail's proposals to backdate tariffs ... completely discounts the legitimate business interests for Queensland Rail to not backdate the reference tariffs—namely the fundamental change by the QCA to its longstanding regulatory precedent for the method of determining the RAB value and the dramatic effect that change in methodology had on the proposed tariff⁷⁵⁶

Our October 2015 Draft Decision position was that we did not accept that the approach to asset valuation in our 2014 Draft Decision was an appropriate basis for Queensland Rail to change its approach to the adjustment amount.

The QCA does not agree that there is any longstanding regulatory precedent for the method of determining Queensland Rail's asset value for the West Moreton network. The asset valuation has been unresolved since access regulation for Queensland's rail networks began in the 1990s. This is discussed in Section 8.13 of this Decision. That section also explains why the QCA considers that the asset valuation approach adopted in this Decision is appropriate having regard to the matters in section 138(2). At the time of Queensland Rail's conduct that gave rise to the expectation that an adjustment amount would be included in the approved access undertaking, Queensland Rail was on notice that the method of determining the asset value was unresolved and was to be the subject of investigation in considering any DAU. However, Queensland Rail did not, at any time, qualify its commitment to an adjustment amount by reference to it achieving its preferred asset valuation.

The QCA accepts that an adjustment amount may not be in Queensland Rail's legitimate business interests. This is perhaps most clearly so where it involves Queensland Rail paying an adjustment amount. On the other hand, at times where the application of the adjustment would mean that Queensland Rail receives an adjustment amount, inclusion of the adjustment amount would be expected to be in Queensland Rail's legitimate business interests.

In any event, Queensland Rail's legitimate business interests must be assessed against the background of its earlier stated commitments to include an adjustment amount and the other matters relevant under section 138(2). Queensland Rail's submission to not backdate the reference tariffs (or rather to not include an adjustment amount) in its legitimate business interests is to be considered with those other relevant matters, including the regulatory

⁷⁵⁴ Queensland Rail, sub. 4: 3.

⁷⁵⁵ Queensland Rail, sub. 4: 3.

⁷⁵⁶ Queensland Rail, sub. 26: 14.

uncertainty and investment impact considerations raised by access holders and access seekers. This is discussed further below.

The QCA also notes that Queensland Rail had itself previously proposed the imposition of what it now describes as a 'retrospective tariff'. For instance, Queensland Rail's March 2012, February 2013 and June 2013 DAUs specified that:

after new Reference Tariffs are approved by the QCA ... this Undertaking will apply as though it were amended to replace the Reference Tariffs with those new Reference Tariffs with effect on and from 1 July 2013⁷⁵⁷

The fact that Queensland Rail has previously voluntarily proposed such provisions shows that they are capable of being in Queensland Rail's legitimate business interests.

Public interest

While the term 'public interest' is not defined in the QCA Act, the QCA has previously considered the following as matters relevant to a consideration of public interest.⁷⁵⁸ The application of the QCA's consideration of the public interest in discussed further in Chapter 10.

Efficient allocation of resources

The public interest in an efficient allocation of resources is best served by the QCA approving an access undertaking that facilitates the delivery of below-rail services at efficient prices and establishes a stable, certain regulatory framework. An important objective is a regulatory framework that provides confidence that in turn underpins investment.

A key issue in relation to Queensland Rail's declared service has been to develop access charges to reflect the efficient costs of delivering below-rail services, with parties acknowledging the importance of regulatory certainty. In this way, rail operators and end users are able to have confidence that access charges promote an efficient allocation of resources which is closely aligned to the public interest in promoting competition (in the above-rail market).

Given competitiveness relates to the ability of a firm to sell its products in a market, arrangements that provide certainty for access charges are consistent with an efficient allocation of resources.

Competitive conditions for Queensland business through regulatory certainty

An approved access undertaking that delivers regulatory certainty will promote stimulus of the Queensland economy and local employment which is an important public interest consideration. The QCA acknowledges that it is not necessary for the access undertaking to achieve economic and employment stimulus.⁷⁵⁹ Rather, the QCA considers that the promotion of economic and employment stimulus is a relevant aspect of the public interest.

Given that a number of consumers of rail services (i.e. coal miners in this case that engage rail operators to transport coal) are selling their products in international markets or face intense competition in their domestic markets, the ability of such consumers to pass on rail transport costs is likely to be constrained (that is, they are price takers). In the absence of certainty, this could undermine the competitiveness for rail operators (both current and future) accessing

⁷⁵⁷ Queensland Rail's March 2012, February 2013 and June 2013 DAUs, cl. 3.4.2(b)(iii)(C). See also Attachment 1 of the QCA's Final Decision on Queensland Rail's April Extension DAAU.

⁷⁵⁸ QCA 2000b: 44–45.

⁷⁵⁹ Queensland Rail, sub. 26: 18.

Queensland Rail's declared services and consumers of above-rail services provided by those rail operators.

Consequently, to improve the conditions for the competitiveness of rail operators and their customers the QCA believes that regulatory certainty, which is necessary to facilitate efficient resource allocation, is particularly relevant. Queensland Rail's submission on this point misunderstands the Draft Decision: the QCA did not (and does not now) say it is 'up to the 2015 DAU to make upstream or downstream businesses competitive'⁷⁶⁰ but rather the impact of regulatory certainty on the competitive conditions for such businesses is a relevant public interest matter.

This issue is discussed further in Section 8.2.1 of this Decision.

Regional economic development

Proposed development of new, or replacement, coal mines may be at risk if there is material pricing uncertainty for rail access. Queensland Rail submits the 2015 DAU would provide pricing certainty.⁷⁶¹ That is not true as Queensland Rail's proposal provides scope for it to increase tariffs in the future up to its proposed ceiling price. Moreover, as discussed extensively in this section the QCA is concerned to have pricing certainty without regulatory uncertainty. To the extent that regulatory uncertainty occurs, there can be flow-on effects in terms of regional economic development.

Relevantly, New Hope, Yancoal and QRC noted potential impacts of Queensland Rail's position on investment as a relevant matter.

For example, New Hope has indicated its proposed New Acland project:

will support construction jobs of up to 260 and approximately 435 operational jobs at peak, and attract construction costs of around \$896 million.⁷⁶²

Likewise, the QRC said:

QR's actions ... highlighted by QR reneging on its commitment to provide an adjustment amount for tariffs ... have undermined investment confidence, and seriously impacted on the attractiveness of investment in this region.⁷⁶³

The QCA also refers to the submissions made in response to the Draft Decision on investment impacts of regulatory uncertainty as discussed above.

Having regard to the above discussion, the QCA considers that it is in the public interest for there to be regulatory certainty with regard to the inclusion of an adjustment amount in circumstances where Queensland Rail previously stated its commitment to an adjustment amount. It is also in the public interest for there to be efficient investment in the infrastructure, which stakeholders have said may be affected as a result of lack of confidence in the regulatory process. This would also lead to a lessening of competition which is not in the public interest.

The effect of Queensland Rail's change in position is that it will have received, in a sense, a windfall gain to the extent that the access charges it has collected between 1 July 2013 the date when the adjustment amount is calculated exceed the access charges that would have been received in that period if calculated in accordance with the reference tariff that would have

⁷⁶⁰ Queensland Rail, sub. 26: 18.

⁷⁶¹ Queensland Rail, sub. 26: 18.

⁷⁶² www.newhopegroup.com.au/files/files/NHG0040_ExecSummary_NAP_EIS_WebRes_Ex.pdf.

⁷⁶³ QRC, sub. 34: 2.

applied in that period. A regulated entity benefiting from delays in the regulatory process to which it has contributed is an outcome that is not in the public interest. As Aurizon submits:

Aurizon Operations does not support the prospect of an Access Provider obtaining a financial benefit funded by Access Holders which arises principally from the Access Provider's own conduct.⁷⁶⁴

Interest of access seekers and access holders

An adjustment amount is clearly in the interests of access seekers and access holders. This is also clear from the submissions made by stakeholders summarised above.

Pricing principles

When considering a DAU, one of the factors that we must have regard to is the pricing principles in section 168A. Relevantly, the pricing principle in section 168A(a) states that the price of access to a service should generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved.

However, the pricing principles are only one of a number of factors to which we must have regard pursuant to section 138(2). Whether a DAU allows recovery of at least enough to meet efficient costs and a relevant return is of course relevant and fundamental to our assessment of the 2015 DAU. But we are not required to consider it appropriate to approve a DAU because the price contained in it will generate revenue that is at least enough to meet the efficient costs of the service and a relevant return. Nor are we precluded from considering it appropriate to approve a DAU that contains a price that is not expected to generate revenue that is at least enough to meet the efficient costs of the service and a relevant return. So the service and a relevant return, where other relevant factors in section 138(2) lead to such a conclusion.

The consideration of the pricing principles is discussed generally in Chapter 10.

We acknowledge that the adjustment amount mechanism will mean that the price of access (as adjusted) will not generate expected revenue in the operative period of the undertaking that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved for that period. Nonetheless, we consider, for all the reasons given in this chapter, that the adjustment amount mechanism is appropriate (and that an undertaking without such a mechanism is not appropriate to approve).

Other relevant issues

The QCA considers the following issues to be relevant to its consideration of whether it is appropriate to approve the 2015 DAU.

The recovery of efficient costs plus a return over the longer term

Infrastructure assets are generally long-lived assets which exist beyond the term of a regulatory undertaking. Therefore, while the QCA and other regulators approve undertakings for a fixed regulatory period, the span of regulatory assets across undertakings is taken into account.

Consistent with this, the QCA presently imposes, and in the past has imposed, a revenue cap model on regulated entities. Under this model, a revenue under- or over-recovery in one year is offset by corresponding revenue under- or over-recovery in a subsequent year, and that

⁷⁶⁴ Aurizon, sub. 20: 11.

subsequent year may be in the next regulatory period.⁷⁶⁵ This reflects the regulatory reality that a regulated entity may under- or over-recover regulatory revenues in a given year or regulatory period and this can be addressed by way of an adjustment mechanism.

This position was reflected by New Hope:

[There is a] common regulatory practice of applying many different types of 'carry-over' mechanisms from one regulatory period to another. For example, in addition to the backdating of tariffs through adjustment charges, other provisions approved by Australian economic regulators have included capital carry-over accounts, efficiency/incentive arrangements, 'unders and overs' under a revenue cap and price paths.⁷⁶⁶

Queensland Rail submits that the QCA's consideration of this point is irrelevant because the 2015 DAU regulatory model is a price cap, not a revenue cap, and that adjustments to revenue from one year to the next are not a feature of a price cap model.⁷⁶⁷ However, the point being made is that, in whatever way the approach is described in terms of models, sections 138(2) and 168A require QCA to have regard to the price for the service by reference to the expected revenue that will be generated, and there is of course a possibility that the revenue actually generated by a given price will be greater or less than the revenue expected to be generated. The adjustment amount is a mechanism for addressing that possibility.

The QCA accepts that Queensland Rail's previously stated commitment to include an adjustment amount in its replacement access undertaking generated an expectation that such commitment would be met. The QCA has approved seven extensions to the existing undertaking on the basis of Queensland Rail's inclusion of an adjustment charge provision in its previous DAUs, up to the submission of the 2015 DAU. The QCA considers that the circumstances of its various approvals, past practice and submissions from stakeholders as to their expectations, establish the basis of the QCA's approvals. The absence of an express condition to that effect⁷⁶⁸ does not materially alter the expectations generated by Queensland Rail's conduct. Further information on this is provided in the QCA's June 2015 decision, which refused to approve Queensland Rail's April 2015 extension DAAU (Appendix H of this Decision).

It is unlikely that stakeholders would have supported the previous extensions of the 2008 access undertaking if they had not been on the basis that an adjustment amount would be applied so that neither Queensland Rail nor access holders were advantaged or disadvantaged.⁷⁶⁹ Indeed, following Queensland Rail's submission of the 2015 DAU, New Hope, who would be directly affected by the absence of any adjustment amount in the 2015 access undertaking, supported the rejection of the DAAU as submitted to the QCA.⁷⁷⁰

The impact of Queensland Rail's change in approach, in circumstances where stakeholders, based on Queensland Rail's previously stated commitment, expected an adjustment amount is

⁷⁶⁵ For example, an 'unders and overs' process for the Central Queensland Coal Network was still included in the revenue cap mechanism in the 2008 undertaking (Schedule F, Part B, Section 3—although those clauses did not apply to Queensland Rail as they were an artefact of when the 2008 undertaking applied to QR Network). The same process has applied for Aurizon Network in the 2010 undertaking (also Schedule F, Part B, Section 3). Similar 'unders and overs' provisions were applied to the revenue caps for Queensland's electricity network businesses when they were regulated by the QCA.

⁷⁶⁶ New Hope, sub. 12: 8.

⁷⁶⁷ Queensland Rail, sub. 26: 13.

⁷⁶⁸ Queensland Rail sub. 26: 14.

⁷⁶⁹ New Hope April 2015 extension DAAU submission: 5, Aurizon, sub. 20: 12.

⁷⁷⁰ New Hope April 2015 extension DAAU submission.

explored above under the 'object criterion' (s. 138(2)(a)) and is also a relevant consideration pursuant to section 138(2)(h).

The integrity of the regulatory framework

The long-lived nature of infrastructure assets means that it is important that there is an effective and credible adjustment mechanism to deal with the under- or over-recovery of access charges both during and across regulatory periods. Its absence creates a 'heads I win, tails you lose' position for regulated entities. If a regulated entity thinks it will benefit from an adjustment amount, it will likely include it in its DAU. If it does not, it may choose to withdraw the offer of an adjustment amount or opt not to make the offer in the first place.

Such a mechanism is also necessary to avoid the potential for regulatory gaming. In this regard, stakeholders said:

[i]t is highly inequitable in the circumstances for QR to retain the overpayments, which have largely occurred through its own delay in submitting an appropriate replacement undertaking⁷⁷¹

[t]he approval of a DAU without appropriate Adjustment Amounts would demonstrate that QR is able to manipulate the regulatory regime to extract from its customers excessive charges to which QR has no rightful claim.⁷⁷²

The QCA notes that the integrity of the regulatory framework is essential to regulatory certainty. A key aspect of promoting integrity is preventing stakeholders from benefitting from any delays to which they have contributed.

The absence of an adjustment mechanism is an incentive for a regulated entity to delay submitting a voluntary replacement DAU until close to the expiry of its existing undertaking in the knowledge that QCA's review will take months beyond the expiry of the existing undertaking to conclude (i.e. the entity could be rewarded for delaying its submission⁷⁷³).

Other matters

Other matters relevant to section 138(2)(h) include:

- Queensland Rail's previously stated commitment to include an adjustment amount
- the expectation of stakeholders of the inclusion of an adjustment amount
- the impact of the change in Queensland Rail's position with respect to the inclusion of an adjustment amount in its 2015 DAU.

As outlined in stakeholders' comments above, the QCA's position is that it accepts that stakeholders had a clear expectation of an adjustment amount. Relevantly, this was consistent with Queensland Rail's commitments. Given this, as outlined above, the QCA considers that the change in Queensland Rail's position will increase regulatory risk, impacting on forward-looking investment.

Extension of the adjustment amount to east of Rosewood

Our Draft Decision estimated an adjustment amount for the West Moreton network (i.e. west of Rosewood).

⁷⁷¹ Yancoal, sub. 27: 1.

⁷⁷² New Hope, sub. 21: 5.

⁷⁷³ Indeed, Queensland Rail submitted its West Moreton tariff proposal in June 2013 a few days before the 30 June 2013 expiry of 2008/10 approved tariffs.

Stakeholder submissions

New Hope⁷⁷⁴, Yancoal⁷⁷⁵ and Aurizon⁷⁷⁶ said that the adjustment amount should also be extended to the Metropolitan network (i.e. east of Rosewood).

In contrast, Queensland Rail opposed the adjustment amount for the east of Rosewood.777

QCA analysis and Decision

We accept that an adjustment amount should be extended to the Metropolitan network. This means that the adjustment amount would apply across both the West Moreton and Metropolitan networks and so cover all aspects of the below-rail service that coal services use.

To date, the tariff regimes for both networks have been inextricably linked. Indeed, as outlined in our Draft Decision, the tariffs that have applied since 2009 for the West Moreton network have been extended across the Metropolitan network (see Section 8.6 of this Decision).

Moreover, previous adjustment amounts for coal services have operated across all aspects of the below rail infrastructure (i.e. both the West Moreton and Metropolitan networks). This is what happened in 2006 and 2010. Relevantly, Queensland Rail had proposed a mechanism in its March 2012, February 2013 and June 2013 DAUs through the operation of 'Adjustment Charge' mechanisms.

For these reasons, if an adjustment amount is to apply for the West Moreton network (as we consider it should), it should also apply for the Metropolitan network.

In this regard, we note Queensland Rail's position that:

[its] submissions concerning the QCA's effective 'back dating' of pricing for the West Moreton Network coal traffics are equally applicable in the context of any consideration of an "Adjustment Amount" applying to the Metropolitan Network.⁷⁷⁸

We accept that Queensland Rail's arguments objecting to the adjustment amount apply equally to both the West Moreton network and the Metropolitan network. As we have not accepted Queensland Rail's position for the West Moreton network, it follows that, for the same reasons, we reject its argument to not extend the adjustment amount to east of Rosewood.

Indicative adjustment amount and implementation

Section 8.20 explains the derivation of reference tariff that would have applied from 1 July 2013 to 30 June 2016.

The QCA estimates that the application of the adjustment amount will result in Queensland Rail receiving indicatively \$32 million⁷⁷⁹, or 23 per cent, less than the allowable revenue that the QCA has assessed Queensland Rail would otherwise receive from coal services for the period from 1 July 2016 to 30 June 2020. The QCA has considered the magnitude of this number in reaching its conclusion as to the appropriateness of the adjustment amount.

For the purposes of this Decision, the QCA can only provide an estimate of the potential overrecovery given:

⁷⁷⁴ New Hope, sub. 21: 6; New Hope, sub. 31: 11–12.

⁷⁷⁵ Yancoal, sub. 27: 1; Yancoal, sub. 35: 2.

⁷⁷⁶ Aurizon 29: 9.

⁷⁷⁷ Queensland Rail, sub 33: 30–31.

⁷⁷⁸ Queensland Rail, sub 33: 31.

⁷⁷⁹ Our estimate with an indicative ±10% sensitivity range.

- the lag between the period when the Decision is released and when a new access undertaking is approved;
- the lag between when the access undertaking is approved and the clause applied and when the adjustment amounts are processed and paid; and
- the circumstances of individual access holders' contracts with Queensland Rail.

The final adjustment amount will be determined by reference to the operation of the Adjustment Charge mechanism in clause 7 of Schedule D. This will provide for an adjustment amount to be agreed (or determined by QCA in the event of dispute) by reference to a comparison between the amount that a particular access holder actually paid in the period from 1 July 2013 to the date of approval, and the amount that the access holder would have paid during that period if the new reference tariff had been in effect at that time.

The final adjustment amount will be determined after the approval date, according to clause 7.1 of Appendix D in Schedule F.

Consideration of factors under the QCA Act

The QCA has concluded that the adverse consequences that would arise from approving the 2015 DAU without an adjustment amount include on regulatory certainty; the future inefficient use of the West Moreton network arising from the lessening of competition in the market for upstream coal tenements; and the public interest in economic development and regulatory certainty and other matters (s. 138(2)(a), (d), (e), (h)).

The QCA considers that these considerations are outweighed by Queensland Rail's legitimate business interest in obtaining such a reference tariff (s. 138(2)(b)) and the pricing principles (s. 138(2)(g)).

On that basis, and having regard to the criteria in section 138(2), the QCA's Decision is that:

- it is not appropriate to approve the 2015 DAU proposed by Queensland Rail; and
- it is appropriate to amend the undertaking to provide for an adjustment amount.

8.19 Adjustment methodology

The QCA's Draft Decision provided for a prospective adjustment amount reflected in a reduction in the forward year tariffs for the regulatory period. For instance, the QCA's Draft Decision noted that while the indicative ceiling tariff as at 1 July 2015 would be \$18.88/'000 gtk, the reference tariff would be \$15.88/'000 gtk, reflecting an adjustment amount of \$3.00/'000 gtk in that year. Staff subsequently published additional material that demonstrated how a prospective adjustment amount could be calculated.⁷⁸⁰

Stakeholders' comments

In addition to being opposed to the principle of an adjustment amount, Queensland Rail had concerns about the methodology for its calculation. Queensland Rail said:

⁷⁸⁰ http://www.qca.org.au/getattachment/8bed340c-84c8-492f-8170-2b6ff8b9484d/QCA-Request-forcomment-on-submissions-on-QR-201.aspx; http://www.qca.org.au/getattachment/487df3b0-5358-447d-8f12-3dd41b7d746c/QCA---Addendum-to-request-for-comment-on-submissio.aspx.

- The proposed adjustment methodology was unclear, lacked transparency and was incapable of objective assessment;⁷⁸¹
- The QCA's methodologies should follow applicable regulatory precedents;
- The QCA should use comprehensive and detailed data;
- It was not possible to test the quantum of the proposed adjustment;⁷⁸²
- The proposed calculation was a mix of actual, forecast, current and past data, meaning it did not represent any reliable estimate of over-recovery; and
- The way the adjustment amount was proposed to be calculated fundamentally altered the risk allocation that otherwise would have applied, were access charges reset on 1 July 2013.⁷⁸³

Aurizon supported the principle of an adjustment amount but said, among other things, the QCA's Draft Decision approach:

- involved the transfer between current and previous access holders in a way that was disproportionate to the individual parties' respective contributions to any determined adjustment amounts;
- was inefficient as it established a reference tariff which was not reflective of forward-looking efficient costs and might create perverse incentives; and
- did not adequately account for take or pay or relinquishment fees during the adjustment period⁷⁸⁴

New Hope and Yancoal supported the Draft Decision's methodology for calculating an adjustment amount.⁷⁸⁵

QCA analysis and Decision

The QCA has reconsidered the approach set out in the Draft Decision as to the methodology for implementing an adjustment amount.

The QCA accepts many of the concerns of both Queensland Rail and stakeholders, including that the Draft Decision approach does not follow regulatory precedents, is not capable of objective assessment, and does not adequately account for take-or-pay or relinquishment fees.

The QCA accepts that its Draft Decision approach was an untested methodology for applying an adjustment amount and one which did not adequately consider the specific circumstances relating to Queensland Rail's overpayments with respect of each access holder agreement.

The QCA has therefore adopted an adjustment amount provision to address any difference in revenue over the relevant period. Relevantly, a similar mechanism was proposed by Queensland Rail in its:

- June 2013 DAU—Schedule A, clause 6.1; and
- May 2015 DAU—Schedule D, clause 7.1.

⁷⁸¹ Queensland Rail sub. 26: 19.

⁷⁸² Queensland Rail sub. 26: 19.

⁷⁸³ Queensland Rail, sub. 33: 29, 30.

⁷⁸⁴ Aurizon sub 20: 9, 13; Aurizon, sub. 29: 8, 9.

⁷⁸⁵ New Hope, sub. 21: 5–6; New Hope sub. 31: 11; Yancoal, sub. 35: 2.

In both cases, the DAUs provided for the possibility that a reference tariff would be applicable by reference to a date prior to the approval.

The QCA now requires that the 2015 DAU be amended to include an adjustment amount mechanism that:

- provides for Queensland Rail to calculate the aggregate of the monthly differences, due to each relevant access holder for the period between 1 July 2013 and the approval date of the DAU, between the access charges paid by that access holder from 1 July 2013 to the approval date and the access charges that would have been payable had the reference tariff been \$15.66/'000 gtk during that period (see Section 8.20 of this Decision);
- allows each relevant access holder to negotiate the payment of a lump sum of the aggregate adjustment amount; or, failing agreement, for the ongoing access charges of that access holder to be adjusted to account for the overpayment;
- provides for the calculation of interest on the adjustment amount at the rate of the bank bill swap rate which would have applied during the overpayment period, consistent with the expectations generated by Queensland Rail,⁷⁸⁶ and
- provides for any disputes in relation to the adjustment amount to be determined by the QCA.

Summary 8.15

The 2015 DAU must include a new clause 7 in Schedule D which provides for an adjustment for any over- or under-recovery of access charges by Queensland Rail during the adjustment period (namely from 1 July 2013 to the approval date of the Undertaking).

See Schedule D, clause 7 and definition of 'adjustment train services' and 'notional reference tariff' in Appendix F.

8.20 Tariff for 2013–14 to 2015–16

The operation of the adjustment charge provision requires an understanding of what reference tariffs would have applied for the period between 1 July 2013 and the approval date of the DAU. As discussed in Section 8.17 of this Decision, we have split the derivation of the tariffs beginning 1 July 2016 from the tariff for the preceding three years. Accordingly, this section discusses the derivation of the reference tariffs that would have applied from 1 July 2013 to 30 June 2016.

Three-year reference tariff for adjustment amount

The allowable revenue for the tariff that would have applied in relation to the period from 1 July 2013 to 30 June 2016 was determined based on building block data from two sources: the 2013 DAU assessed in our 2014 Draft Decision, and the 2015 DAU assessed in this Decision.

⁷⁸⁶ Clause 7.1 of Schedule D of the QCA's mark-ups (Appendix F of this Decision) largely reflects clause 2.3 of Schedule F of the 2008 access undertaking that previously applied to Queensland Rail. Clause 2.3 provides for interest to be paid on any over- or under-recovery of revenues that must be repaid by Queensland Rail or access holders. The 2013 DAU (cl. 6.1(b)) also reflected clause 2.3 of the Schedule F in the 2008 undertaking and provided for interest to be paid on any over- or under-recovery of access charges.

Queensland Rail provided forecasts for the 2015–16 data in both the 2013 DAU and the 2015 DAU. The 2013 DAU forecasts were assessed in our 2014 Draft Decision and have now been assessed in this Decision, at the same time as we consider the 2015 DAU. However, the 2015–16 forecasts between the two DAUs are materially different—for example, the proposed volume is significantly lower in the 2015 DAU.

For the purposes of deriving the tariff that would have applied in relation to the period from 1 July 2013 to 30 June 2016 in this Decision, we have used the 2015–16 data based on our assessment of the 2015 DAU, as it reflects different market environment prevailing in 2015–16 as compared to that which prevailed when Queensland Rail submitted the 2013 DAU.⁷⁸⁷ If we had approved a tariff based on our 2014 Draft Decision, it would have continued to apply (with usual indexing) in 2015–16. However, it would have been open for Queensland Rail to seek to vary that tariff on account of a material change in circumstances. We note that both the 2013 DAU and the 2015 DAU included provisions that allowed Queensland Rail to seek to vary 'a reference tariff if Queensland Rail considers that the variation will promote efficient investment in the coal transport supply chain in the West Moreton System or Metropolitan Region' or seek to vary a reference tariff if a review event that encompasses a material change in circumstances has occurred.⁷⁸⁸

The existence of those provisions would have created an expectation that Queensland Rail may have sought to vary the approved reference tariff due to, for example, a material drop in expected demand for the service. Therefore, our use of the 2015–16 data based on our assessment of the 2015 DAU would be appropriate having regard to any expectation those reference tariff variation provisions would have created. We consider our approach would advance Queensland Rail's legitimate business interests (s. 138(2)(b)), as tariffs would reflect lower forecast volumes for 2015–16. Our approach also advances the interests of access holders and access seekers in not paying fixed common network costs for 2015–16 beyond what they could have contracted, given the existence of significant spare capacity in 2015–16.

Hence, the building blocks data for the financial year 2015–16 are those assessed in this Decision based on our assessment of the 2015 DAU proposal. For financial years 2013–14 and 2014–15 we have adopted the building blocks data assessed in our 2014 Draft Decision.⁷⁸⁹ However, there are the following exceptions:

- Capital expenditure for 2013–14 and 2014–15—we considered actual capital expenditure that Queensland Rail submitted in the 2015 DAU and we assessed as part of this Decision (see Section 8.14 of this Decision). Our assessment of the 2013 DAU was based on forecast capital expenditure. However, using actual capital expenditure instead of forecast capital expenditure would not be contrary to any expectations held by parties. This is because the capital indicator process in the 2013 DAU and the 2015 DAU included provisions to assess the prudency of Queensland Rail's capital expenditure where a difference in actual and forecast expenditure is reflected in the tariffs for a subsequent undertaking period. Our derivation of the July 2013 tariff accounts for that difference rather than carrying it over to the derivation of the July 2016 tariff.
- Inflation for 2013–14 and 2014–15—we used actual ABS CPI data rather than the assumed CPI of 2.5 per cent for indexing and rolling forward the asset base and the tariff. This means that a consistent inflation rate is applied for both the escalation of the tariff and the

⁷⁸⁷ See Sections 8.10 to 8.15 in Part B of Chapter 8.

⁷⁸⁸ 2013 DAU (cl. 5.1(a), Schedule A) and 2015 DAU (cl. 6.1(a), Schedule D).

⁷⁸⁹ QCA, October 2014: 121–126.

escalation of the building blocks components. The escalation by actual CPI would have been done in the post–approval tariff indexing process if the undertaking had been approved earlier.

- Opening asset base at 1 July 2013—since the West Moreton network opening asset base is a contentious matter and had not been settled in the past, we have considered the asset base as assessed in this Decision, which includes our assessment and treatment of Queensland Rail's TSC capital that was not included in our 2014 Draft Decision assessment (see Section 8.13 of this Decision).
- Asset base coal allocator—the asset base coal allocator for financial years 2013–14 and 2014–15 reflects our assessment that the West Moreton network capacity is 113 paths, after deducting the 17 per cent reduction in West Moreton capacity arising from Metropolitan Network operations (see Section 8.4 and discussion about change in circumstances in Section 8.3.3 of this Decision).

Maintenance costs and volumes

During our assessment of the 2015 DAU, Queensland Rail submitted actual maintenance cost and coal volumes data for the financial years 2013–14 and 2014–15.

Queensland Rail's actual maintenance costs for those two years at \$39.0 million are less than the \$40.2 million our 2014 Draft Decision had assessed was efficient.⁷⁹⁰ Queensland Rail operates under a price cap regime, which incentivises it to achieve efficiencies. Therefore, it is appropriate that Queensland Rail retains those cost savings, as this is consistent with the expectations of a price cap regime. Accordingly, we have retained using forecast maintenance cost from our 2014 Draft Decision.

A price cap regime exposes Queensland Rail to volume risk, but a take or pay regime helps limit the associated volume risk. Also, Queensland Rail's revenue could increase if actual volumes are higher than the forecast used to set reference tariff (see Section 8.5 of this Decision). The actual coal railings during 2013–14 and 2014–15 were less than forecast, but Queensland Rail's revenue risk in those years was limited by the take-or-pay revenue and relinquishment fees Queensland Rail collected. Therefore, we consider it is appropriate to retain our use of forecast volumes rather than actual railings for deriving the reference tariff, since any risk (or reward) arising from a difference between actual and forecast volume is consistent with the expectations created by a price cap regime. This is consistent with our treatment of Queensland Rail's expectation of retaining any maintenance cost savings.

Moreover, Queensland Rail's expectation of a drop in coal railings is reflected in the volume forecast used for 2015–16, which would protect Queensland Rail's exposure to any further volume risk and would advance its legitimate business interest (s. 138(2)(b)).

The data sources used for the building block parameters to derive the tariffs for 2013–14 to 2015–16 are summarised in Table 22 and the corresponding building blocks data are provided in Appendix A to this Decision.

⁷⁹⁰ Queensland Rail, sub. 2: 43.

Building block parameter	2013–14	2014–15	2015–16	
Opening asset base	This Decision			
Capital expenditure	Actual expenditure as per this Decision		Capital indicator as per this Decision	
Asset base coal allocator	Reflects our capacity estimate in this Decision		This Decision	
СРІ	Actual		Assumption 2.50%	
Asset lives	2014 Draft Decision and this Decision unchanged			
WACC ⁷⁹¹	2014 Draft Decision and this Decision unchanged			
Maintenance and allocation	2014 Draft Decision	This Decision		
Operating cost and allocation	2014 Draft Decision	This Decision		
Forecast volume	2014 Draft Decision		This Decision	

Table 22 Data sources for 2013–14 to 2015–16 tariffs

Over-recovery during 2013–14 to 2015–16

The resulting reference tariffs for the financial years 2013–14 to 2015–16 are summarised in Table 23, which shows that the reference tariffs that would have applied in those years are lower than the tariffs charged by Queensland Rail.

Table 23 QCA assessed 2013–14 to 2015–16 reference tariffs compared with the actual tariffs charged by Queensland Rail

All tariffs in \$/'000 gtk, nominal	2013-14	2014-15	2015-16
Tariff QR charged	\$18.56	\$19.14	\$19.41
QCA reference tariff	\$15.66	\$16.15	\$16.38
Difference (tariff charged - reference tariff)	\$2.90	\$2.99	\$3.03

Metropolitan tariff

As per our Summary 8.6 (see Section 8.6 of this Decision), we require Queensland Rail to apply the West Moreton network tariff so that it is paid by West Moreton coal services as they traverse the Metropolitan network. Therefore, the tariffs summarised in Table 23 also apply to coal services using the Metropolitan network, and will continue to be escalated for the remainder of the term of the undertaking once approved.

Our Summary 8.6 also requires that a separate Metropolitan incremental capacity charge is calculated to recover coal-specific investment on the Metropolitan network in relation to the period after 1 July 2013. Since Queensland Rail has not claimed any incremental capital expenditure in the Metropolitan network since July 2013, the Metropolitan incremental capacity charge for the purposes of this Decision is zero.

⁷⁹¹ Queensland Rail said risk-free rates were higher in previous periods (see Queensland Rail, sub. 16: 30). This matter is discussed in Section 3.7 of this Decision.

Summary 8.16

The 2015 DAU must provide that the West Moreton network and Metropolitan network reference tariff components that should have applied for the period 2013–14 to 2015–16 are consistent with our reference tariffs of: \$15.66/'000 gtk as at 1 July 2013, \$16.15/'000 gtk as at 1 July 2014 and \$16.38/'000 gtk as at 1 July 2015 (the other components of the tariffs are specified in Appendix A of this Decision).

See definition of 'notional reference tariff' and Schedule D, clause 3.2 in Appendix F.

9 INVESTMENT PLANNING AND COORDINATION FRAMEWORK

An effective investment, planning and coordination framework should provide clarity and certainty on the rights and obligations of all contracting parties in the negotiation process when a network extension is required to accommodate an access application and Queensland Rail is not willing to fund the network extension. The framework must balance the legitimate business interests of Queensland Rail with those of access seekers and any third parties involved in funding the network extension.

The QCA retains the views expressed in its Draft Decision on Queensland Rail's 2015 SAA, except where they have been amended by this Decision. Key areas where the Decision differs from the Draft Decision include:

- Schedule I (Extension Access Principles) and clause 1.4 (Extensions Capacity Investment Framework) have been amended to remove minor inconsistencies in the drafting and provide more clarity.
- Funding agreements terms can now be varied depending on the stage of the extension study process being funded.
- Hybrid funding options are possible if Queensland Rail and the access funder consider this a preferred alternative.
- The requirement for Queensland Rail to produce a standard user funding agreement has been removed.
- The master planning process has been revised and Queensland Rail will be funded for the incremental costs of these studies.

Introduction

An investment framework in an approved access undertaking seeks to provide a high degree of certainty for all access seekers if they:

- commence negotiations with Queensland Rail for access to the declared network
- are advised that:
 - a network extension is required to provide the access sought
 - Queensland Rail is not willing to fund the work required for a network extension
- commence negotiations with Queensland Rail to fund the work required for a network extension
- indicate a willingness to fund the work required for an extension in the absence of Queensland Rail electing to do so.

Key issues are summarised in Table 24 below. Matters that require a more detailed explanation are discussed in Sections 9.1 and 9.2.

Summary of the 2015 Draft Decision	Queensland Rail's position	Other stakeholders' position	QCA Decision			
1. Network extensions						
Included a range of specific amendments to the network extension provisions that were intended to make it more transparent and accountable and to provide a more appropriate allocation of risks.	Disagreed and said that the proposed amendments are unwarranted and/or beyond powers and fail to adequately take into account Queensland Rail's legitimate business interests.	Glencore, Yancoal and New Hope supported the Draft Decision and have detailed how it could be expanded and strengthened.	See Section 9.1 below.			
2. Funding agreements	2. Funding agreements					
Extension funding provisions should provide more detail on funding agreement provisions. Funding and access principles that will underpin negotiations of funding agreements should be identified.	Disagreed and said the proposed amendments impose funding agreement terms which may not be relevant to all stages of the extension.	New Hope said the funding agreement provisions should be amended to require Queensland Rail to offer access facilitation deeds for smaller projects. The proposed amendments impose funding agreement terms which may not be relevant to all stages of the extension	See Section 9.2 below.			
3. Network extension process						
Network planning provisions should include a regulatory process to develop a master plan for each of its major rail corridors during the term of the 2015 DAU.	Disagreed; the requirement to produce master plans for the various systems is beyond powers.	Glencore and New Hope supported the Draft Decision.	See Section 9.3 below.			

Table 24: Summary of the key positions and Decisions—investment planning and coordination

9.1 Network extension and funding agreements

Consistent with section 137 of the QCA Act, we consider the 2015 DAU should outline the general funding principles that will apply to access seekers who agree to fund a network extension to accommodate their access application. In applying our assessment approach, we have considered whether there is sufficient detail of the funding principles and the form of contract that will underpin the negotiation of funding agreements between Queensland Rail and a relevant access seeker.

Consistent with section 137 of the QCA Act, we consider the 2015 DAU should also provide details on how an access seeker can obtain access to the declared service when a network extension project is required to accommodate the access rights nominated in its access application. Our assessment approach considers whether the proposed 2015 DAU provides for an extension project to be developed from identification through to operation in accordance with the object of the QCA Act—that is, to promote the economically efficient operation of, use of and investment in Queensland Rail's declared network.

Queensland Rail set out its proposed investment framework, funding principles and the process for the negotiations of access rights with an access seeker under clause 1.4 of the 2015 DAU.

Our Draft Decision proposed to reject those provisions on the basis that they resulted in significant asymmetries in the network extension process with respect to the following:

- lack of clarity on the negotiation process that contracting parties should follow to give effect to the access and project agreements required for a project to reach financial close and deliver access services consistent with ss. 69E and 138(2) of the QCA Act
- absence of mechanisms to achieve accountability, transparency and timeliness of Queensland Rail's decision-making process, including, for example:
 - unclear boundaries and conditions of what is being negotiated at each stage of the extension project
 - lack of objective and independently verifiable approval criteria
- unbalanced allocation of project risks, liabilities and indemnities due to Queensland Rail assigning all risks and liabilities for an extension project to the access seeker without sufficient consideration as to whether they can best be mitigated by the access seeker or Queensland Rail
- absence of any obligation on Queensland Rail to provide information and assistance to the access seeker to facilitate study of the extension project
- absence of any obligation on Queensland Rail to assist, study, construct and commission the extension project
- absence of a standard funding agreement to form the basis of negotiations between Queensland Rail and an access seeker
- dispute resolution provisions that would be ineffective if the access seeker's access rights are either time-constrained or aligned to a larger investment in the business operations of an access seeker.

Stakeholders' submissions

Queensland Rail did not support our Draft Decision on the basis that arrangements were unwarranted⁷⁹², would give rise to significant cost with no benefit⁷⁹³, were beyond power⁷⁹⁴, were too rigid⁷⁹⁵, and traded off the protection of Queensland Rail's legitimate business interests against other interests, without taking full regard for Queensland Rail's business interests.

Yancoal⁷⁹⁶, Glencore⁷⁹⁷ and New Hope⁷⁹⁸ supported the QCA's Draft Decision. However, Yancoal and Glencore both stated that some of the proposed amendments gave Queensland Rail too much discretion and should be made mandatory. New Hope suggested that the drafting should be amended to make it mandatory that Queensland Rail offer an access facilitation deed (AFD) arrangement for projects under \$25 million.

QCA analysis and Decision

We have generally adopted our Draft Decision position on network extensions and funding agreements, save in the respects outlined below.

Key matters raised by stakeholders are addressed below on a case-by-case basis and Appendix D summarises the proposed amendments to the investment framework.

Requirement for an investment framework

As noted in the Draft Decision, an efficient, transparent and accountable investment framework is a critical component of the 2015 DAU.

In its 2015 DAU, Queensland Rail 'proposed to include a more detailed investment framework than currently applies to it but which was still relatively light-handed'.⁷⁹⁹ Our Draft Decision outlined a number of proposed amendments to this framework and the inclusion of a new schedule to provide more clarity and certainty over the process by which an extension could be negotiated.

Our position is the 2015 DAU should be amended. This is because Queensland Rail's proposed investment framework provided insufficient clarity and certainty on the rights and obligations of all contracting parties in the negotiation (refer to section 9.2 of our Draft Decision).

The Draft Decision summarised a range of responses from stakeholders⁸⁰⁰ that supported the requirement for the framework and noted deficiencies with Queensland Rail's proposal. The responses to the Draft Decision supported the QCA's proposed amendments.⁸⁰¹

Queensland Rail's statement that the QCA's proposed investment framework arrangements are unwarranted on the basis that there are currently no extensions proposed is unjustified and short-sighted. On the contrary, the development of an efficient, transparent and accountable

⁷⁹⁸ New Hope, sub 23: 21–24.

⁷⁹² Queensland Rail, sub. 26: 73.

⁷⁹³ Queensland Rail, sub. 26: 69.

⁷⁹⁴ Queensland Rail, sub. 26: 74, 75,77.

⁷⁹⁵ Queensland Rail, sub. 26: 76.

⁷⁹⁶ Yancoal, sub. 27: 4.

⁷⁹⁷ Glencore, sub. 25: 5.

⁷⁹⁹ Queensland Rail, sub. 26: 68.

⁸⁰⁰ Asciano, sub. 5; Aurizon sub. 6; Glencore, sub. 7; New Hope, sub. 3; Glencore, sub. 7.

⁸⁰¹ New Hope, sub. 23; Glencore, sub. 25; Yancoal, sub. 27.

investment framework is perhaps now even more important in order to maximise the chance that any projects which are proposed go ahead.

Protection of Queensland Rail's legitimate business interests

The Draft Decision proposed amendments to Queensland Rail's proposed investment framework provisions to remove a number of specific references to Queensland Rail's legitimate business interests in its 2015 DAU.⁸⁰² In its response to the Draft Decision, Queensland Rail stated that the removal of these references 'may require Queensland Rail to act in a way that results in its legitimate business interests not being protected.⁸⁰³ and/or may result in Queensland Rail's legitimate business interests not being protected.⁸⁰⁴

We do not accept that a specific provision for the protection of Queensland Rail's legitimate business interests is appropriate as it purports to override other considerations in 138(2). We note we are required to 'have regard to' Queensland Rail's legitimate business interests as part of considering whether to approve or refuse to approve Queensland Rail's 2015 DAU. We are also required to consider, amongst other things, the public interest (s. 138(2)(d)), access seekers (s. 138(2)(e)) and any other issues we consider relevant (s. 138(2)(h)) (see Chapter 10).

Queensland Rail also stated that its legitimate business interests in relation to extensions are significantly wider than portrayed by the QCA.⁸⁰⁵ It makes this point with reference to the specific considerations noted in Table 9.2 of the Draft Decision.

We note Table 9.2 provided a summary of key considerations associated with the consideration of section 138(2)((b), (d), (e) and (h) in our Draft Decision. It was not intended to be an exhaustive statement of what is relevant, and for clarity, we have not sought to narrowly define Queensland Rail's legitimate business interests. We accept that Queensland Rail's interests are as stated by it. Moreover, other factors have been considered in the context of Queensland Rail's legitimate business, including those summarised in section 10.3.2 of the Draft Decision:

- the commercial interest in recovering efficient costs in providing the relevant service and in earning normal commercial returns
- a balanced risk position in the allocation of contractual risks and liabilities as between Queensland Rail and access holders
- appropriate incentives to maintain, improve and invest in the efficient provision of the facility
- incentives to improve commercial returns, where these returns are generated from, for example, innovative investments or cost-cutting measures
- operational processes and procedures within the undertaking.

Access holders' interests

Queensland Rail said 'the assessment criteria relied on by the QCA in relation to the approval or rejection of the extension provisions do not reflect those in the QCA Act'.⁸⁰⁶ Specifically, the QCA Act does not require the QCA to ensure the interests of access holders are 'protected'.⁸⁰⁷

⁸⁰² For example, cls. 1.4.1(c)(iii), 1.4.2(c)(iii)(F), 1.4.3(b)(iii).

⁸⁰³ Queensland Rail, sub. 26: 75.

⁸⁰⁴ Queensland Rail, sub. 26: 78.

⁸⁰⁵ Queensland Rail, sub. 26: 71.

Section 138(2)(e) requires the QCA to have regard to the interests of persons who may seek access to the service, including whether adequate provision has been made for compensation if the rights of users of the services are adversely affected. We consider that the rights of existing access holders are relevant under section 138(2)(h), to the extent they are not also access seekers under section 138(2)(e).⁸⁰⁸ In any event, we have not approached our consideration of the 2015 DAU on the basis that it must ensure the interests of access holders are 'protected'. Those interests are simply something to which we have regard in assessing the 2015 DAU. Further information on this issue is provided in Chapter 10.

Table 9.2 in the Draft Decision provided more detail on our preliminary view about how section 138(2) matters should be addressed in order for the investment framework to be appropriate. It should not be read as if all the potential considerations noted in the second column of the table were strictly applied to the drafting of one or more specific clauses. In this context, the rights of access holders are clearly relevant to the assessment of the appropriateness of the investment framework.

Allocation of additional capacity

Queensland Rail said that the requirement in Schedule I that additional capacity created by an extension be pro-rated between funders relative to their funding share is 'neither practical nor efficient'. They made this point on the basis that an extension may potentially provide more additional capacity than is required by funders, or because Queensland Rail may require part of that additional capacity for activities such as operational activities.⁸⁰⁹

We do not accept that the proposed allocation of additional capacity created by an extension is impractical. We consider that additional capacity means 'the additional capability of the network to accommodate train services that would result from an extension'.⁸¹⁰ Any calculation of the ability of the network to cater for train services would include an assessment of Queensland Rail's requirements for operational activities and preclude the contracting of capacity for these paths.

We have, however, amended Schedule I to clarify that any paths created over and above those required by the user funders will be available for contracting paths as per the terms of the undertaking, rather than be automatically allocated to the user funders.

The obligation for Queensland Rail to pay for any part of a user funded extension

Queensland Rail made a number of general and specific comments⁸¹¹, which argued that elements of the Draft Decision were outside power as they imposed an obligation that Queensland Rail bears some of the costs of an extension. Specifically, Queensland Rail argued that the right of the user funder to fund only those costs which can be reasonably shown to be efficient or prudent implies that Queensland Rail must risk bearing those costs which are not prudently incurred.

We accept that it is not appropriate to limit the obligation of a user funder to only those costs which can be reasonably shown to be efficient or prudent. Consistent with section 119(5)(c) (in the different but relevant context of access determinations) the costs of extending a facility, in

⁸⁰⁶ Queensland Rail, sub. 26: 72.

⁸⁰⁷ Queensland Rail, sub. 26: 72.

⁸⁰⁸ QCA 2015: 257.

⁸⁰⁹ Queensland Rail, sub. 26: 80.

⁸¹⁰ QCA 2015: 59.

⁸¹¹ Queensland Rail, sub. 26: 73, 74, 78.

circumstances where we are not approving Queensland Rail's 2015 DAU under section 136(4) or 142(2) of the QCA Act, should not be required to be paid by Queensland Rail.

Imposition of an obligation to extend

Queensland Rail stated that the effect of the QCA's proposed amendments is to oblige it to extend the network 'when it would otherwise have good legal and commercial reasons for not doing so'⁸¹², and that this obligation is beyond power.⁸¹³

The extension conditions provide that Queensland Rail is not obliged to proceed with a proposed extension stage where that stage of an extension does not meet certain criteria. These criteria include matters which, if not satisfied, would mean Queensland Rail would have good legal and commercial reasons for not agreeing to the relevant stage. These include, for example, if the proposed extension is not technically feasible or adversely impacts on existing access rights.

The proposed amendments provide certainty that when an extension stage is proposed and meets these hurdles, each stage will be progressed. Amendments have been made to clause 1.4.1(b) to clarify that this is the case.

Requirement to maintain an extension

Queensland Rail said that the QCA was seeking to impose an obligation to maintain an extension and that this was beyond power because the proposed definition of maintenance work in the SAA included replacement costs and was in direct conflict with section 119 of the QCA Act.⁸¹⁴

The definition of maintenance costs in the SAA has been amended to exclude replacement costs.

Access facilitation deeds

New Hope⁸¹⁵ has suggested a range of amendments to cl. 1.4, which would have the effect of obliging Queensland Rail to enter into AFD arrangements for extensions which have a project cost of under \$25 million.

We have not adopted this suggestion. The AFD process outlined by New Hope obliges Queensland Rail to provide up to \$25 million of funding on terms consistent with previous agreements that it has made (adjusted for changes to the cost of debt). New Hope proposed that any other variations to these terms should be subject to dispute. However, New Hope provided no details of those agreements. Given this, we cannot be confident that, simply because terms specified in a proposal were agreed to in the past, they will be agreed to in the future. In any event, we can see no reason why such an agreement could not be negotiated between a potential access funder and Queensland Rail without any amendments to clause 1.4.

Standard funding arrangements

New Hope has suggested a range of amendments to clause 2.9.4, which would have the effect of:

requiring Queensland Rail to submit and obtain approval for a standard study funding agreement

⁸¹² Queensland Rail, sub. 26: 74, 75.

⁸¹³ Queensland Rail, sub. 26: 74, 77.

⁸¹⁴ Queensland Rail, sub. 26: 71.

⁸¹⁵ New Hope, sub. 23: 23.

• providing QCA with the power to require Queensland Rail to produce a standard funding agreement for the construction and operation phase.

Queensland Rail, on the other hand, rejected the proposed amendments and said that the amendments to clause 2.9.4(b)–(d) were outside of power because 'a standard funding agreement cannot be "tacked-on" after the approval of the access undertaking by the QCA'.⁸¹⁶

We note that the Aurizon standard user funding agreement negotiations have occurred in the context of a very different network, where capacity is fully contracted and there is demand for further capacity. This is not the case for Queensland Rail's network, where there is substantial spare capacity. Given this, we have removed the standard user funding agreement provisions from the 2015 DAU and suggest that Queensland Rail reviews the outcome of the Aurizon process with a view to potentially using it as a basis for similar provisions to be included in its next undertaking. In the meantime, the amended investment framework provides access seekers with guidance on the terms and conditions and coverage expected in a funding agreement for an extension.

Building Queensland Act

Queensland Rail has noted that the recently passed *Building Queensland Act 2015* (the Building Queensland Act) applies to Queensland Rail as a 'government agency' and that the effect of this Act had not been taken into account in the drafting of clause 1.4.

The effect of the Building Queensland Act is to pass over the responsibility for preparing business cases for projects which have a total value over \$100 million from the government agency to Building Queensland.⁸¹⁷ The Act also states that Building Queensland is to lead the procurement or delivery of such an infrastructure project if directed to do so by the Minister.

We do not accept the drafting of clause 1.4 should be adjusted substantially for the effect of the Building Queensland Act. While the effect of the Act may alter the decision-making process within Queensland Rail, Queensland Rail will remain the railway manager and operator and the contracting entity that will be dealt with by the access seeker. It would, however, be expected that the impact of Building Queensland on the decision-making process would be taken into account in any assessment of reasonableness particularly if the organisation had impacted on the timeliness of decisions. However, we have added a new clause 1.4.8 which details how Queensland Rail and the access funder should respond to Building Queensland's involvement (or potential involvement) in an expansion.

Negotiation triggers and hybrid funding

Queensland Rail⁸¹⁸, New Hope⁸¹⁹ and Glencore⁸²⁰ said that the proposed drafting of clauses 1.4 and 2.7, and Schedule I could be interpreted as implying that if Queensland Rail notified that it was not willing to fund a stage of an expansion, a funding agreement for all stages of the extension process would have to be negotiated. Queensland Rail⁸²¹ also said that hybrid funding options (where Queensland Rail funded part of an extension) should be permitted.

⁸¹⁶ Queensland Rail, sub. 26: 81.

⁸¹⁷ An advisory body that has been set up to assess and manage the Queensland Government's investment in major projects. See (http://buildingqueensland.qld.gov.au/) for more detail.

⁸¹⁸ Queensland Rail, sub. 26: 81.

⁸¹⁹ New Hope, sub. 23: 23–23.

⁸²⁰ Glencore, sub. 25: 5.

⁸²¹ Queensland Rail, sub. 26: 82.

While we are of the view that the proposed drafting made it clear that separate funding agreements could be negotiated for each stage of the extension (e.g. cl. 1.4.2(a), final paragraph), we have further amended a number of clauses (e.g. cls. 1.4.1(a)(iii), 1.4.3(a), 2.7.2(b)(iii)) to make it clearer that this is the case.

Similarly, we have amended clause 1.4.1(c)(i)(B) to make it clear that a hybrid option would be consistent with the undertaking.

Schedule I

Queensland Rail said that Schedule I (the extension access principles) is not appropriate and needs to be completely revised by the QCA on the basis that it, amongst other things, is internally inconsistent and inconsistent with Part 1 to 7 of the DAU, promotes inappropriate contracting, imposes obligations which are not appropriate under the undertaking and is jumbled with respect to the operation of funding agreements.⁸²² New Hope supported the inclusion of Schedule I but said that the principles were loose and high-level and that many of the clauses were drafted to apply to funding commissioning and construction, and do not apply well to the funding of studies.⁸²³

We disagree with Queensland Rail that Schedule I is not appropriate and is inconsistent with Parts 1 to 7 of the DAU; as detailed in our Draft Decision, we consider it to be important to the establishment of a workable network extension process. Relevantly, it has been difficult to address many of Queensland Rail's concerns specifically, given it did not specify its criticisms of Schedule I by reference to particular provisions of the schedule. Instead it chose to list a set of potential issues under a generic introduction that stated that they were relevant to 'various provisions under Schedule I'.⁸²⁴

That said, following the above concerns of Queensland Rail and New Hope, we have amended Schedule I to more clearly allow for its application to the study phases of an extension. This should improve clarity with respect to the rights and responsibilities of the Queensland Rail and the access funder, and remove some errors and inconsistencies. The revised Schedule I has nine separate clauses/sections, instead of 10, as before.

Dispute resolution process

Queensland Rail said that the dispute resolution process specified in cl. 1.4.2(f) (which allowed an access funder to refer a dispute directly to the QCA if Queensland Rail unnecessarily delayed the provision of assistance),was beyond powers because it 'purports to give someone who is not an access seeker a right to have the QCA arbitrate disputes involving Queensland Rail'.⁸²⁵ It also said that the clause was inconsistent with clause 6.1 (which does not refer disputes automatically to the QCA) and unnecessary (because clause 1.4.7 deals with disputes in relation to extensions).

We assume that Queensland Rail is referring to the inclusion of an access seeker's nominee in the proposed definition of access funder when it makes reference to 'someone who is not an access seeker'. We note that we have deleted this reference from the definition on the basis that we do not consider that the additional complexity in the drafting of the expansion framework is justified.

⁸²² Queensland Rail, sub. 26: 79–81.

⁸²³ New Hope, sub. 23: 23.

⁸²⁴ Queensland Rail, sub. 26: 79.

⁸²⁵ Queensland Rail, sub. 26: 78.

We disagree with Queensland Rail's comment that the direct referral of disputes to the QCA is unnecessary. Timeliness is likely to be a critical component of negotiation of an extension for an access seeker. While rail access may only be a relatively small component of total investment required to enable an expansion in production, funding for the other elements are often dependent on the ability of the producer to show that they have an effective means of transporting their product to market. The dispute resolution mechanism that is mapped out in clause 6.1 can take up to 25 business days before it is escalated to the QCA for resolution. If there were a number of disputes negotiating a single stage extension, this could place undue pressure on the access seeker to agree to terms which they may otherwise consider unfavourable.

We also disagree with Queensland Rail's comment that the direct referral of disputes is inconsistent with clause 6.1. The provision that access seekers and Queensland Rail can agree to use a different dispute resolution process or timetable is included in clause 6.1.1; furthermore, we note that Queensland Rail has included clause 1.4.7(f) in its 2015 DAU, which allows the access seeker to refer a matter directly to the QCA under clause 6.1.4. However, to clarify the operation of dispute process in the context of clause 1.4, we have now amended clause 1.4.7 to specify that if no access funding agreement has been executed, the parties can escalate directly to the QCA. If a funding agreement has been executed, the dispute is dealt with through the parties' agreed dispute resolution provisions in the funding agreement. We have also amended clause 6.1.4 to clarify that disputes that are referred directly to the QCA do not have to have satisfied the earlier process steps before the QCA can determine the dispute.

In addition, we have made a number of amendments to clause 1.4 and Schedule I to improve the clarity and consistency of their operation. These amendments are summarised in Appendix D.

Summary 9.1

The 2015 DAU must provide as follows:

- (a) A transparent and accountable network extension process should be provided to underpin the negotiation of each stage of an extension project concurrently with an access agreement, including clarifying the threshold criteria which an extension must meet and outlining rights and responsibilities of the parties in relation to the separate stages of an extension.
- (b) Sufficient detail should be included, with clear reasoning, on the scope of an extension project.
- (c) Objective and reason-based assessment criteria to underpin the decisionmaking process of an extension project should be provided.
- (d) All necessary project information and project assistance reasonably required by an access seeker to develop an efficient scope, standard and cost of an extension should be provided.
- (e) A balanced allocation of the rights and responsibilities between Queensland Rail and an access seeker should be provided.
- (f) Standard principles to apply to all extensions should be included.
- (g) More detail should be provided on the funding agreement provisions reasonably required to satisfy the access rights sought by an access seeker.
- (h) The funding and access principles that will underpin the negotiation of a funding agreement should be identified.
- (i) A more streamlined dispute mechanism process should be provided for extension related disputes.
- (j) The parties' rights and obligations in relation to the Building Queensland Act should be clarified.

See clauses 1.4, 2.7, 2.9.2(q), 2.9.5(b), 7.1 and Schedule I in Appendix F.

9.2 Network planning provisions

Consistent with section 137 of the QCA Act, we consider the 2015 DAU should outline the network planning processes that apply to key corridors on Queensland Rail's network. In applying our assessment approach, we have considered whether there is sufficient detail on the planning processes applied by Queensland Rail to show that it is efficiently operating, using and investing in its network.

The Draft Decision proposed that Queensland Rail be required to amend its network planning provisions to include a regulatory process for developing a master plan for each of its major rail corridors during the term of the 2015 DAU.

Stakeholders' submissions

Queensland Rail argued that system planning is not part of the regulated service and is not covered by the QCA Act (and so the amendments proposed in the Draft Decision were beyond

power). Queensland Rail also made the comment that the costs of developing the master plans were unfunded.⁸²⁶

New Hope supported the requirement for Queensland Rail to develop regional network master plans and establish regional network capacity groups, noting that 'NHC has experienced significant frustration at the lack of any clear process for planning for growth and assessing various capital investment and expansion options'.⁸²⁷ Glencore noted its support for an obligation to develop a new Mount Isa line master plan, as additional transparency should assist future access negotiations.⁸²⁸ Asciano also supported the proposed amendments but said that the planning groups should also include operators (where those operators are neither access seekers or access holders).⁸²⁹

QCA analysis and Decision

The QCA considers Queensland Rail should amend its network planning provisions to include a regulatory process to develop a master plan for each of its major rail corridors during the term of the 2015 DAU in accordance with clause 1.5.⁸³⁰

We do not accept Queensland Rail's position that system planning is beyond power. Section 137(2) states that an access undertaking for a service may include details of information to be given to access seekers; information to be given to the QCA or 'another person' and timeframes for giving information in the conduct of negotiations about access to the service.

We are of the view that the 2015 DAU:

- does not provide sufficient transparency or accountability on Queensland Rail's compliance with the object of the Act (s. 138(2)(a)); and
- is not appropriate when having regard to the public interest and the interests of access seekers and access holders because it does not provide for access holders, access seekers, end customers and rail operators to receive sufficient information on Queensland Rail's management and operation of the network (s. 138(2)(d), (e) and (h)).

Queensland Rail has argued that the object of Part 5 of the Act (as detailed in s. 69E) is only intended to promote the relevant matters, not achieve or ensure them.⁸³¹ We do not suggest that an access undertaking will only be appropriate to approve if it 'ensures' a particular matter referred to in section 138(2). The matters in section 138(2) are simply matters to which we must have regard in considering a draft undertaking. We note that the proposed master plans will improve the information available to decision-makers and therefore promote the economically efficient operation of, use of and investment in the Queensland Rail network.

Our Decision is therefore to require Queensland Rail to amend its network planning provisions to include a regulatory process to develop a master plan for each of its major rail corridors during the term of the 2015 DAU.

The contents of the master plans should be based on the information requirements set out in section 101(2). The provision of a master planning document that transparently sets out Queensland Rail's current and expected cost of service provision including capital, operating

⁸²⁶ Queensland Rail, sub. 26: 84–85.

⁸²⁷ New Hope, sub. 23: 24.

⁸²⁸ Glencore, sub. 25: 1.

⁸²⁹ Asciano, sub. 28: 14.

⁸³⁰ Master planning and extension coordination.

⁸³¹ Queensland Rail, sub. 26: 83.

and maintenance costs (s. 101(2)(b)), approach to developing prices (s. 101(2)(a)) and estimating capacity (s. 101(2)(d)) would go a long way to addressing a number of the key issues raised by stakeholders.

Our position is not unusual—both the 2010 DBCT⁸³² and the 2010 QR Network⁸³³ access undertakings include provisions requiring the access provider to undertake master planning activities.

However, we agree that the costs of these studies should be funded by access holders and access seekers to the extent that any existing allowance for system planning is insufficient to cover the costs. If Queensland Rail can show that the costs of developing the required master plans are not covered by their existing operating cost allowance and that the proposed costs are reasonable, Queensland Rail should be able to request that those parties who want to participate in the planning process fund the process up-front. This will also assist in making sure that stakeholders are invested in the process and their interest is genuine. If stakeholders decide that they are not interested in system planning, then they are not obliged to fund a planning group.

We also agree with Asciano that operators should be invited to join the proposed regional network capacity groups in those situations where they are neither access seekers or access holders.

Our Decision:

- aligns with Queensland Rail's obligation to manage its network consistent with the object of the QCA Act (s. 138(2)(a)); and
- is appropriate having regard to the public interest, the interests of persons seeking access (s. 138(2)(d), (e)), and Queensland Rail's legitimate business interests (s. 138(2)(b)).

Summary 9.2

In respect of network planning provisions, the 2015 DAU must include a regulatory process to develop a master plan for each of Queensland Rail's major rail corridors during the term of the 2015 DAU if funded by participants.

See clause 1.5 in Appendix F.

⁸³² DBCT 2010.

⁸³³ QR Network, 2010 Access Undertaking, Part 11: Coordination and Planning.

10 LEGAL OVERVIEW

This chapter sets out how we have applied our legislated obligations in making our 2015 Decision.

10.1 Part 5 of the QCA Act

Part 5 of the QCA Act establishes a third party access regime to provide a legislated right for third parties to acquire services provided using significant infrastructure that is owned by a monopoly service provider. When the Bill to establish the QCA Act was introduced, the accompanying Explanatory Notes said:

The underlying rationale of creating third party access rights to significant infrastructure is to ensure that competitive forces are not unduly stifled in industries which rely upon a natural monopoly at some stage in the production process, especially where ownership or control of significant infrastructure is vertically integrated with upstream or downstream operations

A key aspect of the market system is that an infrastructure owner is entitled to choose with whom it will deal. The threat of competitors providing substitutes constrains a seller's ability to charge excessive prices or otherwise restrict supply. However, in cases where these substitutes do not exist, a seller possesses significant market power. A seller may exercise its market power to increase its profit by restricting output because doing so enables the seller to increase its price.

In cases of natural monopoly, one facility meets all of a market's demand more efficiently than a number of smaller and more specialised facilities. Accordingly, it is not socially desirable that the infrastructure comprising a natural monopoly be duplicated. At the same time, the absence of competition enables a natural monopoly infrastructure owner to extract excessive profits through exercising market power.

This is especially the case where the business which operates the natural monopoly also has a commercial interest in upstream or downstream markets (for example a rail operator who also owns the track). Such a business may discriminate against its upstream or downstream competitors by offering access on more favourable terms and conditions than is offered to competitors. In this way, an owner of a natural monopoly is able to stifle competition in upstream or downstream markets.

The purpose of third party access is therefore to provide a legislated right to use another person's infrastructure. This should prevent owners of natural monopolies charging excessive prices. It should also encourage the entry of new firms into the potentially competitive upstream and downstream markets which rely on a natural monopoly infrastructure in the production process, and thereby enable greater competition in those markets. This in turn would promote more efficient production and lower prices to consumers.⁸³⁴

10.2 Assessment approach

Queensland Rail lodged the 2015 DAU for our consideration and approval under sections 137 and 138 of the QCA Act. The 2015 DAU was lodged in response to our initial undertaking notice issued under section 133 of the QCA Act.

Section 134 of the QCA Act requires us to consider the 2015 DAU given in response to the section 133 notice and either approve or refuse to approve it.

⁸³⁴ Explanatory Notes to the *Queensland Competition Authority Bill 1997*, pp. 3–4.

We acknowledge that we are not permitted to refuse to approve the 2015 DAU simply because we consider a minor and inconsequential amendment should be made to the 2015 DAU.⁸³⁵

This Decision to refuse to approve the 2015 DAU comprises an attachment to a secondary undertaking notice under section 134(2) of the QCA Act, requiring Queensland Rail to give us an amended copy of the DAU within 60 days. The secondary undertaking notice and its attachments (including this Decision) sets out the reasons for the refusal and the amendments to the DAU that the QCA considers appropriate under section 134(2)(a) of the QCA Act.

If Queensland Rail complies with the secondary undertaking notice, we may approve the amended 2015 DAU.⁸³⁶ If Queensland Rail does not comply with secondary undertaking notice, we may prepare and approve an amended 2015 DAU to apply to Queensland Rail in relation to the provision of the declared service.⁸³⁷

We acknowledge the 2015 DAU is the culmination of a four-year regulatory process that has involved Queensland Rail submitting and then withdrawing three versions of a DAU. We also note that, in considering this Decision, we have reviewed the 2015 DAU with 'a fresh set of eyes' and assessed it on its merits applying the legislative criteria. In forming the positions expressed in this Decision, we had regard to, among other things:

- each part of the 2015 DAU;
- all of the 2015 DAU submissions provided by Queensland Rail and stakeholders;
- relevant sections of our 2014 Draft Decision that were referenced by Queensland Rail and stakeholders in their submissions;
- information provided by Queensland Rail in accordance with section 185 notices issued by the QCA; and
- information provided by Queensland Rail and stakeholders through a further round of submissions once submissions on the 2015 Draft Decision were received.

The remainder of this chapter sets out some comments on particular issues relating to the criteria listed in section 138(2) of the QCA Act that are relevant to the 2015 DAU when coming to our Decision outlined in the preceding chapters.

10.3 Section 138(2) of the QCA Act

Section 138(2) of the QCA Act states that we may approve the 2015 DAU only if we consider it appropriate to do so having regard to each of the matters set out in section 138(2) of the QCA Act (see Box 3).

⁸³⁵ Sections 138(5) and (6) of the QCA Act.

⁸³⁶ Section 134(3) of the QCA Act.

⁸³⁷ Section 135 of the QCA Act.

Box 3: Section 138(2) of the QCA Act Section 138(2) of the QCA Act provides: The Authority may approve a draft access undertaking only if it considers it appropriate to do so having regard to each of the following -(a) the object of this part; (b) the legitimate business interests of the owner or operator of the service; (c) if the owner and operator of the service are different entities-the legitimate business interests of the operator of the service are protected; (d) the public interest, including the public interest in having competition in markets (whether or not in Australia); (e) the interests of persons who may seek access to the service, including whether adequate provision has been made for compensation if the rights of users of the service are adversely affected; the effect of excluding existing assets for pricing purposes; (f) (g) the pricing principles mentioned in section 168A; (h) any other issues the Authority considers relevant. The 'object of this part' as referred to in section 138(2)(a) is set out in section 69E: The object of this part is to promote the economically efficient operation of, use of and investment in, significant infrastructure by which services are provided, with the effect of promoting effective competition in upstream and downstream markets. The pricing principles set out under section 168A are: The pricing principles in relation to the price of access to a service are that the price should generate expected revenue for the service that is at least enough to meet (a)the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved; and (b) allow for multi-part pricing and price discrimination when it aids efficiency; and not allow a related access provider to set terms and conditions that (c) discriminate in favour of the downstream operations of the access provider or a related body corporate of the access provider, except to the extent the cost of providing access to other operators is higher; and (d) provide incentives to reduce costs or otherwise improve productivity.

'Appropriate'

The QCA Act requires us to determine whether it is 'appropriate' to approve a DAU having regard to the relevant matters listed in section 138(2). The term 'appropriate' in the QCA Act is one of wide import.

Queensland Rail has submitted that section 138(2) does not require the QCA to consider an access undertaking to be the most appropriate undertaking in order to approve it—the access undertaking need only be considered to be appropriate having regard to all the criteria in section 138(2) of the QCA Act.⁸³⁸ The QCA agrees.

⁸³⁸ Queensland Rail, sub. 26: 7.

We are not assessing the 2015 DAU against an optimal or preferred standard; rather, we are considering whether the undertaking is 'appropriate' by reference to all the statutory factors.

The reason why the Draft Decision did not propose not to approve the 2015 DAU was not because we preferred a different access arrangement, which we believed would better achieve the statutory objectives. The Draft Decision proposed not to approve the 2015 DAU because QCA's draft conclusion was that it was not appropriate to do so after having regard to the statutory factors in section 138(2). Given our draft conclusion, we then set out the way in which we considered it would be appropriate to amend the undertaking (s. 136(5)(b)).

Equally, we note that in considering whether the 2015 DAU is appropriate to approve, we are not compelled to approve an undertaking that is the least onerous and restrictive, from the perspective solely of the regulated business. We are required to determine whether the 2015 DAU is 'appropriate' to approve by reference to the factors in section 138(2)—factors that have a focus which is wider than just the perspective of the regulated business.

The QCA has adopted this approach in this Decision.

'Have regard to'

In making our decision as to whether the 2015 DAU is appropriate to approve, we must have regard to the factors in section 138(2) of the QCA Act.

The phrase 'have regard to' has been interpreted by Australian courts as requiring the decisionmaker to take into account the matters to which regard is to be had as an element in making the Decision.

In response to the Draft Decision, Queensland Rail submitted that 'by not ensuring that the price for access gives Queensland Rail revenue at least sufficient for it to recover its efficient costs and the required return, the QCA is effectively disregarding the pricing principle factor referred to in section 138(2).' The QCA rejects the statement that it has 'effectively' disregarded section 168A(a). As discussed further below, the QCA regards the pricing principles as a fundamental consideration (in the sense of being a central element in our deliberative process), but other relevant considerations may warrant a particular decision being made.

Weight

The matters listed in section 138(2), considered in light of the provisions of the DAU, may, and indeed often will, give rise to competing considerations which need to be weighed in deciding whether it is appropriate to approve the undertaking. Some of the matters to which the QCA must have regard favour different conclusions.

Some examples of possible tensions are:

- between the legitimate business interests of the owner or operator of the service (s. 138(2)(b)) on the one hand, and the interests of persons who may seek access to the service (s. 138(2)(e)) on the other hand; and
- between the effects of excluding existing assets for pricing purposes (s. 138(f)), and setting a
 price that generates 'expected revenue for a service that is at least enough to meet the
 efficient costs of providing access to the service and include a return on investment
 commensurate with the regulatory and commercial risks involved' (s. 168A(a)).

In the absence of any statutory or contextual indication of the weight to be given to factors to which a decision-maker must have regard (as is the case in the QCA Act), it is generally for the

decision-maker to determine the appropriate weight to be given to them.⁸³⁹ We consider that this approach generally applies here.

Queensland Rail submits that the QCA cannot 'trade off' the factors listed in section 138(2) but instead must give primacy to paragraphs (a) and (g) (the object of Part 5 and the pricing principles).⁸⁴⁰ We do not agree, for the reasons given above. This issue is covered in more detail by the discussion below in the context of the pricing principles.

10.3.1 The object of Part 5

Section 138(2)(a) requires us to have regard to the object of Part 5 of the QCA Act when deciding whether it is appropriate to approve an access undertaking.

Promoting economically efficient outcomes

The object of Part 5 of the QCA Act (s. 69E) is to promote the economically efficient operation of, use of and investment in significant infrastructure by which services are provided, with the effect of promoting effective competition in upstream and downstream markets.

We consider these efficient outcomes include:

- efficient operation and use of the existing network—to promote economically efficient operation and use of existing infrastructure, we consider the 2015 DAU should, among other things, provide controls on Queensland Rail's ability to differentiate between access seekers; encourage and not discourage utilisation of the infrastructure; and enhance the transparency of regulatory processes
- efficient investment in the network—the 2015 DAU should enhance the transparency and processes associated with identifying and mitigating bottlenecks, identifying the need for network extension, and actioning those extensions. It should also provide controls and incentives to provide that future capital expenditure is prudent, including in terms of scope, standard and costs whilst offering appropriate regulatory certainty for network investment decision-making and including a return on investment commensurate with the regulatory and commercial risks involved
- encouraging and not discouraging upstream and downstream investment—including by ensuring appropriate regulatory certainty for such investment decision-making. A stable, certain regulatory framework provides confidence that in turn underpins investment in longlived infrastructure investments and expenditure on exploration activities. Conversely, uncertainty can result in a lessening of competition for upstream coal tenements (limited exploration and mine development expenditure) and inefficient use of Queensland Rail's West Moreton network rail infrastructure (by discouraging new entrants from taking any spare rail capacity).

Competitiveness of the access price

New Hope⁸⁴¹ and Yancoal⁸⁴² submitted that the competitiveness of the tariff and the impact this has on the competiveness of access holders and access seekers are relevant under section 138(2)(a), (d), (e) and (h) of the QCA Act. The submissions are, however, primarily expressed in the context of paragraph (a). It is argued that the level of the tariff has an effect on the efficient

⁸³⁹ Minister for Aboriginal Affairs v Peko-Wallsend Ltd (1986) 162 CLR 24, 41 (Mason J).

⁸⁴⁰ Queensland Rail, sub. 26: 9, 10.

⁸⁴¹ New Hope, sub. 21: 4, 5.

⁸⁴² Yancoal, sub. 27: 3, 4.

use of the Queensland Rail network, as high tariffs can lead to reduced utilisation of the network. The submissions also note that the medium-to long-term competitive position of the miners in global coal markets is a matter of the public interest (relevant under s. 138(2)(d)).

New Hope and Yancoal submitted that these considerations, along with the interests of access seekers and access holders (relevant matters under s. 138(2)(e) and (h)), should result in the price being below that which results from commonly accepted regulatory methodologies in order to make them competitive in coal markets.

The QCA accepts that the level of the tariff may have an effect on the utilisation of the network and that the competitive position of the miners in global coal markets is a matter of the public interest and a matter of interest to access holders and access seekers. However, regulatory predictability and certainty in the regulatory process and its outcomes, including the application of commonly accepted and previously applied regulatory methodologies, are also relevant matters for the QCA under section 138(2)(a), (b), (d), (e) and (h). How these considerations have been taken into account by the QCA in considering the level of the tariff is discussed in Section 8.2 of this Decision.

Our role in the application of the object of Part 5 of the QCA Act

In the Draft Decision, we stated 'our role as an access regulator includes the promotion of the efficiency objectives of Part 5 of the QCA Act and we are empowered by statute to set the appropriate arrangements that we consider necessary to achieve these objectives'.

Queensland Rail has made a number of comments on this statement and on the QCA's recitation of the language of the objects clause in the Draft Decision. Queensland Rail has concluded that this infects the whole of that Draft Decision.⁸⁴³ The QCA accepts that the QCA Act does not impose an obligation on Queensland Rail 'to maintain, operate, use and, if required, extend the network' or to manage the network to any particular end.

The QCA agrees that the focus of the object of Part 5 is the promotion of the outcomes prescribed in that object, rather than their achievement by either the QCA or Queensland Rail. This is how the QCA has applied that object.

10.3.2 The legitimate business interests of Queensland Rail

Section 138(2)(b) requires us to have regard to the 'legitimate business interests' of the owner or operator of the service, in this case Queensland Rail. As the owner and operator are the same entity, the QCA's consideration of section 138(2)(b) also covers section 138(2)(c).

'Legitimate business interests' is not a defined term under the QCA Act.

We consider Queensland Rail will have a legitimate business interest across a range of areas. These include:

- the commercial interest in recovering efficient costs in providing the relevant service⁸⁴⁴ and in earning a return on investment commensurate with the regulatory and commercial risks involved in supplying the declared service
- a balanced allocation of contractual risks and liabilities as between Queensland Rail and access holders

⁸⁴³ Queensland Rail, sub. 26: 7.

⁸⁴⁴ The 2015 DAU only seeks reference tariffs for coal services on the West Moreton and Metropolitan networks.

- appropriate incentives to maintain, improve and invest in the efficient provision of the facility to provide the declared service
- incentives to improve commercial returns, where these returns are generated from, for example, innovative investments or cost-cutting measures
- transparent and effective operational processes and procedures such as the network management principles and operating requirements manual within the undertaking
- the safe operation of the facility.

The legitimate business interest of Queensland Rail is one of the factors to which the QCA is to have regard pursuant to section 138(2).

10.3.3 The public interest

Section 138(2)(d) requires the QCA to have regard to the public interest, including the public interest in having competition in markets (whether or not in Australia).

The term 'public interest' is not defined in the QCA Act.

We also note that any assessment of the public interest will be shaped by the context of the particular assessment.

We consider it to be in the public interest for there to be an access undertaking that establishes a stable, certain regulatory framework that facilitates the delivery of below-rail services at prices that are appropriate having regard to the relevant matters referred to in the QCA Act.

In this way, rail operators and end users are able to have confidence that access charges promote an efficient allocation of resources, consistent with the public interest in having competition in the above-rail market.

Where consumers of rail services sell their products in international markets or face intense competition in their domestic markets, the ability of such consumers to pass on rail transport costs is likely to be constrained. If the costs of providing the service are not efficient, this could undermine the competitiveness of rail operators accessing Queensland Rail's declared services and consumers of above-rail services provided by those rail operators (particularly the miners in global coal markets).

In addition, we consider that an undertaking that delivers regulatory certainty provides a major stimulus to the Queensland economy and local employment which is an important public interest consideration.⁸⁴⁵ The development of new, or replacement, upstream producers may be at risk if there is material pricing uncertainty for rail access. This can have flow-on impacts on regional economic development.

Having effective competition in upstream and downstream markets

Efficient access to the Queensland Rail network is of significance for competition in the market for freight services, including through contracting and operating requirements embodied in Queensland Rail's proposal. We consider that Queensland Rail continues to have the ability and incentive to use its market power to adversely affect competition in a number of dependent markets including:

• the market for above-rail services;

⁸⁴⁵ QCA 2001b: 45.

- the market for certain products that are transported on Queensland Rail's network; and
- the market for resource tenements from which those products are produced.

Competition in the above mentioned markets can be affected by the operation of the contracting and operating requirements embodied in the 2015 DAU.

We therefore consider that an access undertaking should seek to:

- minimise any barriers for access to the declared service
- improve the conditions for competition in upstream and downstream markets by promoting regulatory certainty to enable confidence in decision-making.

Supply chain coordination

We consider there to be a strong alignment between effective supply chain coordination and the public interest. There is a clear link between allowing for the coal supply chain to operate in the most effective and efficient way possible and the public interest in maintaining an internationally competitive Queensland coal sector.

10.3.4 The interests of persons who may seek access

Section 138(2)(e) requires the QCA to have regard to the interests of persons who may seek access to the service, including whether adequate provision has been made for compensation if the rights of users of the services are adversely affected. We consider that the rights of existing access holders are relevant under section 138(2)(h), to the extent they are not also access seekers under section 138(2)(e).

We consider that section 138(2)(e) encompasses the interests of train operators as access seekers or potential access seekers. Indeed, we have received submissions from Aurizon and Asciano, as train operators on the Queensland Rail network, and their views are considered throughout this Decision.⁸⁴⁶

We consider the interests of access seekers include an effective negotiation framework; transparent and public information about access to and use of the network; adequate reporting; certain and effective transitional arrangements as one undertaking replaces another; access principles that are effective for a balanced negotiation or renewal of an access agreement; standard access agreements that represent a fair risk allocation; effective obligations to maintain the network; and a workable and effective extension and expansion framework.

Finally, in having regard to the interests of persons who may seek access, our Decision on the 2015 DAU primarily references the predominant use in each of the major systems—that is, coal in the West Moreton network, bulk freight on the Mount Isa line and containerised freight on the North Coast line. We acknowledge, however, that these are not the only commodities on each system and we have had regard to interests of other access seekers.

10.3.5 The effect of excluding existing assets for pricing purposes

Section 138(2)(f) requires the QCA to have regard to the effect of excluding existing assets for pricing purposes.

We have had regard to this criterion as part of our assessment of the revenues and tariffs appropriate for the declared service.

⁸⁴⁶ Asciano, sub. 5; Aurizon, sub. 6.

Submissions have been made in relation to this criterion in the context of various assets discussed in Chapter 8.

As it is explained further in Chapter 8, the QCA's view is that its approach does not involve 'excluding existing assets for pricing purposes' as the relevant assets have been taken into account for pricing purposes in the way discussed in that chapter. However, even if the proper reading of the words 'the effect of excluding existing assets for pricing purposes' in section 138(2)(f) would mean that the approach we have taken does involve excluding existing assets for pricing purposes, that is a matter to which the QCA has had regard.

10.3.6 The pricing principles in section 168A

Section 138(2)(g) requires the QCA to have regard to the pricing principles in section 168A.

The pricing principles in relation to the price of access to a service are that the price should:

- (a) generate expected revenue for the service that is at least enough to meet the efficient costs of providing access to the service and include a return on investment commensurate with the regulatory and commercial risks involved; and
- (b) allow for multi-part pricing and price discrimination when it aids efficiency; and
- (c) not allow a related access provider to set terms and conditions that discriminate in favour of the downstream operations of the access provider or a related body corporate of the access provider, except to the extent the cost of providing access to other operators is higher; and
- (d) provide incentives to reduce costs or otherwise improve productivity.

Queensland Rail submits that the pricing principles have primacy over other considerations in section 138(2).

It previously said:

[The] reference tariff must deliver to the access provider at least its efficient costs and a return as required by section 168A(a). Anything less should not be approved and cannot be imposed.⁸⁴⁷

It has maintained this position in its response to the Draft Decision:

In the case of the QCA Act there is an unequivocal statutory and contextual indication that the pricing principles ... are to be given priority.⁸⁴⁸

To support this view, Queensland Rail referred to the Explanatory Notes to the Queensland *Competition Authority Amendment Bill 2008*, and various sections of the QCA Act. It also considered that the pricing principles are in a stand-alone section and that the QCA cannot approve an access undertaking that is inconsistent with the QCA Act. Queensland Rail submitted the language of section 168A(a) does not support Queensland Rail providing the 'declared service at a loss'.

The QCA considers the submission being made by Queensland Rail is that these matters displace the general approach that it is for the QCA to determine the weight to be given to the matters required to be taken into account by section 138(2). Queensland Rail submitted that the QCA Act requires that section 168A, and section 168A(a) in particular, be given greater weight than other considerations.

⁸⁴⁷ Queensland Rail sub. 1: 5.

⁸⁴⁸ Queensland Rail sub. 26: 9.
The QCA does not agree. Section 138(2) is drafted as a simple list, with the language of the section imposing no requirement that any particular item be viewed necessarily more significant than the others. The QCA is not persuaded the other contextual matters in the QCA Act which Queensland Rail referred to are such as to conclude otherwise.

The QCA considers the pricing principles, including section 168A(a), to be fundamental considerations (in the sense of being a central element in the deliberative process), but that does not mean they have primacy over other considerations or that it is necessary for them to be 'complied with' in some absolute way. It is open to the QCA to consider that a DAU which provides for a price that allows a service provider to recover at least the efficient costs of providing access to the service and a relevant return on investment, is nonetheless not one which is appropriate to approve, including by reference to other factors such as the object of Part 5 of the QCA Act (s. 138(2)(a)), the interests of access seekers and holders (s. 138(2)(e), (h)) and the public interest (s. 138(2)(d)).

The pricing principles are only one of a number of factors to which we must have regard pursuant to section 138(2). Whether a DAU allows recovery of at least enough to meet efficient costs and a relevant return is of course relevant and fundamental to our assessment of the 2015 DAU. However, we are not required to consider it appropriate to approve a DAU because the price contained in it will generate revenue that is at least enough to meet the efficient costs of the service and a relevant return. Nor are we precluded from considering it appropriate to approve a DAU that contains a price that is not expected to generate revenue that is at least enough to meet the efficient costs of the service and a relevant return. So the service and a relevant return, where other relevant factors in section 138(2) lead to a conclusion that it is appropriate to approve the DAU.

Queensland Rail also takes issue with the QCA saying that '[a]lthough section 168A(a) states that prices should generate revenue to at least meet the efficient costs of providing access, it is also true that prices above the efficient cost would not be in the interests of access seekers and holders, nor in the public interest'.⁸⁴⁹ Queensland Rail says that '[b]y giving paramount effect to the purported interests of access seekers, access holders and the public, the QCA is purporting to re-write the QCA Act and expressly stating contrary to the language of section 168A(a), that Queensland Rail should never get more than its efficient costs'.⁸⁵⁰

Queensland Rail's conclusion is erroneous. The QCA merely makes the point that prices above the efficient cost would not be in the interests of access seekers and holders or in the public interest. It does not say these interests necessarily prevail as a matter of statutory construction.

10.3.7 Any other matters the QCA considers are relevant

Section 138(2)(h) allows the QCA to have regard to any other issues it considers relevant.

This paragraph is expressed in broad terms. We consider the interests of access holders and end users are relevant under this paragraph. The interests of these stakeholders broadly coincide with the interests of access seekers, as all access seekers who sign contracts will become access holders. However, as discussed in various parts of Chapter 2, there are some issues of particular relevance to access holders and end users.

In addition to the above, we consider the following matters, among other things, to be relevant:

• the terms of the 2008 AU and the reasons for proposed changes to it

⁸⁴⁹ Queensland Rail sub. 26: 10.

⁸⁵⁰ Queensland Rail sub. 26: 10.

- the previously approved split-form access agreement in Aurizon Network's 2010 undertaking
- certainty through maintenance and operational integrity
- the extent to which commercially negotiated outcomes should be recognised under the negotiate-arbitrate principle
- the extent to which a regulated entity earns windfall gains and monopoly profits.

Role of the 2008 AU

We consider the previously approved standard access principles in the 2008 undertaking and the 2008 SAA (2008 regulatory precedents) to be relevant in our consideration of the 2015 DAU. We are of the view that the 2008 regulatory precedents have been effective in providing Queensland Rail, access seekers, access holders and end customers with:

- a considered and balanced approach to contract risk management
- a contracting framework that is well understood by all freight traffics operating on the network
- a level playing field in negotiating access rights with a monopoly service provider.

On this basis, we have considered afresh the 2008 regulatory precedents and are of the view that the balanced risk position underlying the 2008 regulatory precedents are relevant in considering the stated matters in section 138(2) of the QCA Act. In particular, we consider that the 2008 SAA provides a good example of an allocation of risks and obligations to the contracting parties best able to manage them, that facilitates access to the network and an efficient total cost of access.

Aurizon Network 2010 undertaking

We consider the previously approved split-form access agreement in Aurizon Network's 2010 undertaking (2010 regulatory precedents)⁸⁵¹ to also be relevant in our consideration of the 2015 DAU. We are of the view that the 2010 regulatory precedents provide Queensland Rail, access seekers, access holders and end customers with relevant guidance on a:

- considered and balanced approach to contract risk management
- contracting framework that is well understood by all freight traffics operating on the network
- level playing field in negotiating access rights with a monopoly service provider.

On this basis, we have considered afresh the split-form SAA and are of the view that the balanced risk position underlying the split-form SAA is current and relevant in considering the matters stated in section 138(2). In particular, we consider that the 2010 undertaking SAAs provide a good example of an allocation of risks and obligations to the contracting parties best able to manage them, that will facilitate access to the network and an efficient total cost of access.

Certainty through maintenance and operational integrity

Within the context of section 138(2)(h) we have also considered the role of certainty from regulatory arrangements. That is, the extent to which stability and certainty promote

⁸⁵¹ Stakeholders have all identified the importance of ensuring the 2015 SAA contains the contractual flexibility provided by the split-form SAA in the Aurizon Network's 2010 undertaking.

confidence in the regulatory arrangements and therefore economic efficiency by reducing uncertainty associated with long-term investment decisions.

This may be considered in the context of operational integrity of the network with regard to maintenance, as well as the performance of the network over the longer term. For example, further certainty may be provided through system master plans by providing for access rights to be renewed on expiry.

Negotiate-arbitrate model and primacy of commercial negotiations

The third party access regime in the QCA Act is underpinned by the 'negotiate–arbitrate' approach to regulation, with the regime incorporating the principle of primacy of contractual negotiations.

We consider that parties should endeavour to negotiate a mutually beneficial outcome before they resort to arbitration. When parties are unable to agree, arbitration is an appropriate means of resolving disputes. Indeed, if the dispute resolution process is not credible, the negotiation process can be unduly biased in favour of Queensland Rail by not sufficiently addressing the asymmetry in bargaining power.

An access undertaking is a means to achieve ex ante certainty by providing the terms and conditions on which Queensland Rail will provide access, to avoid developing these arrangements separately with each access seeker.

We have therefore considered how the 2015 DAU affects the role of customer engagement, the balance of negotiation strength, barriers to participation, the flow of relevant and timely information, and whether it provides for effective dispute mechanisms, accountability and transparency.

Windfall gains, monopoly profits and other matters

A return that is materially above that which is necessary for an entity to meet the efficient costs of providing access to the service, including a return on investment commensurate with the regulatory and commercial risks of providing access, may generate windfall gains and monopoly profits.

This would be inconsistent with economically efficient investment, operation and use of a regulated network and has the potential to have both upstream and downstream investment impacts.

Likewise, the QCA can take account of whether the approved tariff is consistent with the recovery of efficient costs over the long term, such a matter being relevant to the object of Part 5 of the Act (s. 138(2)(a)), the public interest including in having competition in markets (s. 138(2)(d)), and other issues the QCA considers relevant (s. 138(2)(h)).

ACRONYMS

Α	
AFD	Access facilitation deed
В	
С	
clause, clauses	cl., cls.
СРІ	consumer price index
D	
DAAU	Draft Amending Access Undertaking
DAC	Depreciated actual cost
DAU	Draft Access Undertaking
DORC	Depreciated optimised replacement cost
DTMR	Queensland Department of Transport and Main Roads
DTP	Daily Train Plan
E	
EIRMP	Environmental Investigation and Risk Management Report
F	
FCM	Financial Capital Maintenance
G	
gtk	Gross tonne kilometre
н	

1	
IAP	Indicative Access Proposal
IRMP	Interface Risk Management Plan
J	
К	

L.	
Μ	
МТР	Master Train Plan
Ν	
NMPs	Network management principles

NPV	Net present value
0	
ORM	Operating Requirements Manual
Ρ	
Q	
R	
RAB	Regulatory Asset Base
S	
section, sections	S., SS.
Т	
TIA	Transport Infrastructure Act 1994
TSC	Transport Service Contract
TSE	Train Service Entitlement
TRSA	Transport (Rail Safety) Act 2010
U	
V	
W	
WACC	Weighted average cost of capital
X	
V	
Y	
-	
Ζ	

APPENDIX A: SUMMARY OF QCA REVENUE/PRICING MODEL (2016 FOR QUEENSLAND RAIL 2015 DAU)

Table 1 Summary: QCA model for coal reference tariff—West Moreton network and Metropolitan network

Coal-allocated regulatory asset base (RAB) roll-forward (F Columboola) (\$)	Rosewood to	2013-14	2014-15	2015-16	2016-17	2017-18	2018-19	2019-20
opening asset value		175,559,747	184,802,412	202,353,610	223,025,639	241,799,742	261,167,109	278,552,137
plus capex		8,807,870	16,879,521	18,084,595	20,377,204	21,285,572	19,616,005	17,895,797
less depreciation		(5,358,034)	(5,811,785)	(6,579,599)	(7,431,885)	(8,227,627)	(9,003,842)	(9,710,991)
plus inflationary gain		5,792,830	2,921,898	5,283,502	5,828,784	6,309,421	6,772,864	7,186,120
closing asset value		184,802,412	198,792,046	219,142,108	241,799,742	261,167,109	278,552,137	293,923,063
West Moreton network Coal Annual Revenue Requirement (Rosewood to Columboola)		2013-14	2014-15	2015-16	2016-17	2017-18	2018-19	2019-20
Return on capital	\$	12,476,072	13,396,610	14,655,127	13,355,130	14,456,248	15,517,525	16,463,833
Inflation	\$	(5,792,830)	(2,921,898)	(5,283,502)	(5,828,784)	(6,309,421)	(6,772,864)	(7,186,120)
Return of capital	\$	5,358,034	5,811,785	6,579,599	7,431,885	8,227,627	9,003,842	9,710,991
TSC capital charge	\$	(783,757)	(1,007,627)	(902,235)	(770,693)	(778,607)	(786,436)	(794,170)
Maintenance expenditure	\$	19,473,887	18,388,908	28,792,424	17,881,271	18,065,649	18,401,376	18,683,192
Operating expenditure	\$	4,748,652	4,820,465	5,416,737	5,600,784	5,740,804	5,884,324	6,031,432
Working capital allowance	\$	131,851	135,987	107,245	116,677	119,594	122,584	125,648
Tax allowance	\$	1,286,982	1,510,637	0	0	0	0	0
Total Building Blocks ARRs	\$	36,898,891	40,134,868	49,365,396	37,786,271	39,521,893	41,370,350	43,034,806
Net Present Value (NPV) of ARRs	\$	\$109,979,453			\$140,543,333			

		2013-14	2014-15	2015-16	2016-17	2017-18	2018-19	2019-20
Reference Tariff	\$/'000gtk	\$15.66	\$16.15	\$16.38	\$17.92	\$18.37	\$18.83	\$19.30
Inflation rate used in escalating tariff (only)		n/a	3.14%	1.43%	n/a	2.50%	2.50%	2.50%
Coal volume - West Moreton network	000 gtks	2,714,126	2,714,126	2,110,379	2,110,379	2,110,379	2,110,379	2,110,379
Smoothed Revenues (Tariff x West Moreton coal gtks, pre-tax)	\$	42,501,651	43,835,036	34,570,081	37,824,386	38,769,995	39,739,245	40,732,726
NPV of smoothed revenues	\$	\$109,979,453			\$140,543,333			
Reference tariff inputs for coal traffic using the West Moreton network	Units	2013-14	2014-15	2015-16	2016-17	2017-18	2018-19	2019-20
For train originating in West Moreton network								
Coal volume - West Moreton network	000 gtks	2,714,126	2,714,126	2,110,379	2,110,379	2,110,379	2,110,379	2,110,379
Coal volume - West Moreton network	Train paths	7,700	7,700	6,280	6,280	6,280	6,280	6,280
AT1 - W (gtk tariff)	\$/'000 gtk	\$7.83	\$8.08	\$8.19	\$8.96	1		
AT2 - W (train path tariff)	\$/train path	\$2,724.42	\$2,809.90	\$2,849.96	\$3,011.50			
Inflation rate used in escalating tariff (only)		n/a	3.14%	1.43%	n/a	_		
Reference tariff inputs for coal traffic using the Metropolitan network	Units	2013-14	2014-15	2015-16	2016-17	2017-18	2018-19	2019-20
For train originating in Metropolitan network								
AT1 - M (gtk tariff)	\$/'000 gtk	\$15.66	\$16.15	\$16.38	\$16.66			
Inflation rate used in escalating tariff (only)		n/a	3.14%	1.43%	1.69%	-		

Reference tariff inputs for coal traffic using the Metropolitan network	Units	2013-14	2014-15	2015-16	2016-17	2017-18	2018-19	2019-20
For train originating in West Moreton network								
Coal volume - Metropolitan network (West Moreton mines only)	000 gtks	1,061,984	1,061,984	866,138	866,138	866,138	866,138	866,138
Coal volume - Metropolitan network (West Moreton mines only)	Train paths	7700	7700	6280	6280	6280	6280	6280
AT1 - M (gtk tariff)	\$/'000 gtk	\$7.83	\$8.08	\$8.19	\$8.33			
AT2 - M (train path tariff)	\$/train path	\$1,079.87	\$1,113.75	\$1,129.63	\$1,148.69]		
Inflation rate used in escalating tariff (only)		n/a	3.14%	1.43%	n/a	4		

Table 2 Approved ceiling revenue limit for reference train services—Total of West Moreton and Metropolitan networks

	Units	2016-17
Approved Ceiling Revenue Limit ^a		\$53,940,246.33
Based on 2016-17 inputs:		
Reference tariff (West Moreton network)	\$/'000 gtk	\$17.92
Coal Volume (West Moreton network)	000 gtks	2,110,379
Reference tariff (Metropolitan network)	\$/'000 gtk	\$16.66
Coal volume (Metropolitan network, all mines)	000 gtks	967,494 ^b

^aApproved Ceiling Revenue Limit = Reference tariff (West Moreton) x coal volume (West Moreton) plus Reference tariff (Metropolitan) x coal volume (Metropolitan, all mines) ^bThis includes gtks relating to the seven contracted paths originating in the Metropolitan network.

APPENDIX B: COMMENTS ON 2009 DRAFT DECISION TARIFF

The stakeholder comments on the 2009–2013 western system tariff, subsequent to the detailed analysis in our December 2009 Draft Decision and the shorter discussion in our June 2010 pricing decision,⁸⁵² are quoted below. The quotes are excerpts from submissions that covered a number of other matters about the relevant draft undertakings and decisions.

QRC response to the December 2009 Draft Decision:

- Western System tariffs Industry will be considering this matter in detail and providing views to QR Network and the QCA during subsequent consultation processes. Industry is concerned with:
 - The proposed tariff structure and volume setting approach.
 - Assumptions used in the development of the reference tariffs.⁸⁵³

QRC response to QR Network's 2010 DAU:

QRC does not consider that QR Network has demonstrated that its proposed Western System tariffs are derived based on a robust methodology, nor that the methodology is suitable for "rolling forward" to the next undertaking in order to avoid subjective judgements in regard to UT4 tariffs. QR Network and the QCA failed to establish a robust approach to determining Western System tariffs for both UT1 and UT2. This matter has therefore been before the Authority since 1999. QRC is strongly opposed to a further decision which leaves this matter unresolved.

QRC considers that a reasonable process must be established for the assessment of Western System tariffs and that, in the absence of this process, no tariff increase should be approved. We note the apparent urgency to settle the tariffs for the Central Queensland region in the context of the privatisation of these assets and the natural focus of all parties on this part of the system. This focus appears set to prevent a proper assessment of tariffs and other issues for the Western System. We therefore suggest that existing tariffs be approved as the appropriate tariffs until a proper assessment of Western System issues and tariffs is completed at a later date.

Given that a new undertaking will be required upon the planned transfer of the Western System assets to a new Government Owned Corporation, and that this will raise a range of issues beyond tariffs, we suggest that the interim tariffs should remain in place until replaced by a properly assessed undertaking for the new GOC.

QRC considers that there should be no increase in Western System tariffs until the case for increases is demonstrated by a reasonable, repeatable and robust methodology.⁸⁵⁴

Syntech response to QR Network's 2010 DAU:

In regard to the quantum of tariffs, we do not consider that QR Network has demonstrated that its proposed tariffs are derived based on a robust methodology, nor that the methodology is suitable for 'rolling forward' to the next undertaking in order to avoid subjective judgements in regard to UT4 tariffs. QR and the QCA failed to establish a robust approach to determining Western System tariffs in UT2 and it appears that this outcome is set to be repeated.

Syntech considers that a reasonable process must be established for the assessment of Western System tariffs and that, in the absence of this process, no tariff increase should be approved. We note the apparent urgency to settle the tariffs for the Central Queensland region in the context of the privatisation of these assets and the natural focus of all parties on this part of the system.

⁸⁵² QCA, 2009 December: 69–94; QCA, 2010 June: 87–90.

⁸⁵³ QRC 2010a: 9.

⁸⁵⁴ QRC 2010b: 10–11.

This focus appears set to prevent a proper assessment of tariffs and other issues for the Western System. We therefore suggest that existing tariffs be approved as the appropriate tariffs until a proper assessment of Western System issues and tariffs is completed at a later date. Given that a new undertaking will be required upon the planned transfer of the Western System assets to a new Government Owned Corporation, and that this will raise a range of issues beyond tariffs, we suggest that the interim tariffs should remain in place until replaced by a properly assessed undertaking for the new GOC.⁸⁵⁵

QRC response to June 2010 extension DAAU (with amendments to pricing):

QRC maintains that there should be no increase in Western System tariffs until the case for increases is demonstrated by a reasonable, repeatable and robust methodology. QRC restates its position that existing tariffs should be approved as the appropriate tariffs until a proper assessment of Western System issues and tariffs is completed – otherwise the incentive for these matters to be resolved appears to be lost until the commencement of the next regulatory (UT4).

QR Network has not demonstrated that its proposed Western System tariffs are based on a robust methodology, nor that the methodology is suitable for "rolling forward" to the next undertaking in order to avoid subjective judgements in regard to UT4 tariffs.

QRC maintains that a reasonable process must be established for the assessment of Western System tariffs and that, in the absence of this process, no tariff increase should be approved. While the apparent urgency to settle the tariffs for the Central Queensland region in the context of the privatisation of these assets has been the focus of the QCA and QR Network, QRC maintains its position for a proper assessment of tariffs and other issues relating to the Western System.⁸⁵⁶

⁸⁵⁵ Syntech, 2010 May: 3.
⁸⁵⁶ QRC 2010c: 9.

APPENDIX C: SAA DRAFT DECISION COMPARISON

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
QCA accepted Queensland Rail's proposal to remove a separate Access Principles schedule (see Draft Decision 7.1).	The QCA, after reviewing each of the factors listed in s138(2) of the QCA Act and the stakeholder submissions, accepted QR's reasoning that, given the 2015 SAA is anticipated to apply to all traffic types, it is appropriate for the previous Access Principles to be embedded in the body of the SAA (subject to the proposed SAA's risk allocation matrix being amended (see below)).	New Hope supports Draft Decision 7.1 subject to having a robust SAA. ⁸⁵⁸ New Hope also agrees with Aurizon that QR should be required to substantiate reasons for any refusal to amend the terms of the SAA. ⁸⁵⁹ Aurizon does not support Draft Decision 7.1 and would prefer the inclusion of a list of principles for access agreements as a schedule to the undertaking to support negotiation for non-coal access agreements. ⁸⁶⁰ Aurizon also believes that the use of a SAA for all traffics, including non-Reference Tariff coal traffic, is inappropriate without additional amendments to the SAA. Yancoal supports the QCA's Draft Decision. ⁸⁶¹ Glencore said that it is supportive of all parts of the Draft Decision (in	QR does not comment further on the deletion of the access principles.	The QCA accepts QR's proposal to not include a separate schedule of access principles. The QCA requires amendments to include an obligation on QR, in cl. 2.9.4 of the 2015 DAU and cl. 1.3 of the SAA, to provide reasons if it rejects proposals by access seekers or access holders to vary the SAA where an access seeker can demonstrate the variation will improve the productivity or efficiency of its rail operations. (for further reasoning, see Summary 7.1 & associated analysis)

⁸⁵⁷ Note that the reasoning given in this table is a summary only. The table should be read in conjunction with the appropriate sections of the Decision, the Draft Decision and the drafting in the SAA provided by the QCA in Appendix G.

⁸⁵⁸ New Hope, sub. 24: 3.

⁸⁵⁹ New Hope, sub. 31: 20.

⁸⁶⁰ Aurizon, sub. 20: 41.

⁸⁶¹ Yancoal, sub. 27: 4.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		relation to the SAA) except as noted below in relation to a number of particular clauses. ⁸⁶²		
The QCA did not approve QR's proposed structure. The QCA instead required QR to amend its 2015 SAA to provide for a tripartite structure which allows an end customer or an operator to be the access holder and have the necessary flexibility to manage the use of access rights and which more appropriately divides the contract responsibilities and risk amongst the parties to the SAA (see Draft Decisions 7.3 & 7.4).	In submissions on the 2015 SAA, the majority of stakeholders reiterated to the QCA the need for the 2015 SAA to provide the contractual flexibility to allow either a rail operator or a single end customer to be the access holder. The QCA, after reviewing each of the factors in s138(2) of the QCA Act, determined that a tripartite structure (an example of which was attached as an appendix to the Draft Decision) would be appropriate. The QCA also reasoned that, after having regard to each of the factors in s.138(2), that the tripartite SAA should have a more balanced division of risks, rights and responsibilities between the parties.	New Hope supports Draft Decisions 7.3 & 7.4 but proposed further amendments (see below in relation to QCA Draft Decisions regarding particular clause). ⁸⁶³ Aurizon does not support the QCA's proposed tripartite structure. Instead, Aurizon would prefer if an access holder could allocate access rights across multiple, separate access agreements. Aurizon has a number of concerns in relation to specific provisions for example, cls. 15, 17, 24 & 27.4, of the SAA (see below). Yancoal supports the QCA's Draft Decision but submits that there needs to be some further consequential amendments to the undertaking. ⁸⁶⁴ Asciano also has concerns about multiple operators being signatories to a single access agreement. ⁸⁶⁵	QR does not accept the QCA's proposed tripartite structure. QR does not believe that allowing an end user to hold access rights is necessary to address the issues with the old form of operator access agreement. QR also submitted a number of concerns in relation to particular clauses (outlined below). ⁸⁶⁶	The QCA requires QR to amend the 2015 SAA to provide an ability for either an end customer or an operator to manage and control the use of access rights. The QCA requires QR to amend the 2015 SAA to adopt a tripartite structure which more appropriately divides the contract responsibilities and risks of each party as demonstrated in Appendix G. We have removed the ability for more than one operator to be a party to a single access agreement but rather provided that, if an access holder wants to nominate multiple operators, each operator must enter into a separate, substantially identical (unless otherwise agreed) tripartite agreement. To effect these requirements, a number of consequential amendments are required to

⁸⁶² Glencore, sub. 25: 1.

⁸⁶³ New Hope, sub. 24: 3 and New Hope, sub. 31: 27.

⁸⁶⁴ Yancoal, sub. 27: 4.

⁸⁶⁵ Asciano, sub. 28: 8.

⁸⁶⁶ QR, sub. 26: 91–93.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				particular clauses. These are outlined below. Where these are the same or differ from the same clauses in the Draft Decision this is noted. Also, to the extent that a required amendment is the same as a Draft Decision, the applicable reasoning from the Draft Decision is incorporated and relied on in this Decision (except where varied by this Decision). (for further reasoning see Summary 7.2 & associated analysis)
The QCA required QR to amend its 2015 SAA to give effect to a more balanced risk position for all parties to the agreement (see Draft Decision 7.7).	The QCA determined that the proposed QR 2015 SAA risk allocation was too imbalanced as between QR and access seekers, access holders & operators such that it was not appropriate having regard to each of the factors stated in s138(2) of the QCA Act. The QCA identified amendments to the QR 2015 SAA which would provide a more balanced risk position and therefore would be appropriate having regard to s138(2) of the QCA Act (these amendments are outlined in greater detail below).	New Hope supports Draft Decision 7.7 but proposed amendments in addition to those required by the QCA (see below in relation to particular clauses). Aurizon did not specifically comment on the QCA's proposed risk position in the Draft Decision. However, Aurizon has recommended a number of specific changes and additional amendments (see below in relation to particular clauses). Yancoal generally supports the QCA's Draft Decision subject to some further suggested amendments (outlined below). ⁸⁶⁷	QR submits that, despite the QCA considering the 2008 and 2010 regulatory precedents as relevant to its proposed consideration of the proposed 2015 SAA, the QCA has departed from these precedents in a number of instances in a manner which is adverse to QR's legitimate business interests without reason. More specific submissions from QR are noted below. ⁸⁶⁹	QCA has identified multiple amendments which it requires to make the risk balance in the 2015 SAA appropriate — see Appendix G for the particular amendments and below in relation to specific clause reasoning. The QCA, in its analysis and reasoning underlying Summary 7.3, states that the previously approved regulatory precedents (2008 & 2010 SAAs and the access principles) are relevant and indicative. However, the QCA does not consider that the regulatory precedents remain appropriate for every specific clause in the 2015

⁸⁶⁷ Yancoal, sub. 27: 4.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		Asciano said it still has strong concerns about what it considers an unbalanced risk position. ⁸⁶⁸		SAA. (for further reasoning see Summary 7.3 & associated analysis)
The QCA required QR to amend its 2015 SAA to include a KPI reporting regime (see Draft Decision 7.10).	The QCA considered that a KPI reporting regime is essential to deliver a balanced risk position for access seekers, access holders and end customers. A KPI reporting regime would be appropriate, having regard to each of the factors listed in s.138(2) of the QCA Act by, amongst other things, promoting the efficient operation of QR's network.	New Hope welcomes the inclusion of KPIs but proposes further amendments including financial incentives (see below in relation QCA Decisions regarding particular clauses). ⁸⁷⁰ Aurizon considers the QCA's Draft Decision to be a "step towards increased transparency". However, Aurizon states that it is uncertain how any financial incentives are to be determined and how financial incentives would work within the proposed tripartite structure. ⁸⁷¹ Glencore generally supports the inclusion of KPIs. However, Glencore suggests that performance reporting regime should not be left for agreement between QR and the operator (see also comments below). ⁸⁷² Yancoal supports the inclusion of a KPI regime subject to financial outcomes being included in the	QR states generally that it supports measures to improve transparency and reporting where such measures provide clear benefit to all parties to the agreement. ⁸⁷⁵ See more specifically comments in relation to clause 6.7 below.	The QCA, after reviewing all of the submissions, requires QR to include a mandatory reporting regime. Following the receipt of meaningful data, the QCA's 2015 SAA then provides for the parties to negotiate KPIs and incentives. The QCA has included some additional reporting criteria and removed the obligation on QR to report weekly. Further, the QCA has made the performance reporting regime subject to the dispute resolution provisions and requires QR to warrant as to the accuracy of the data it supplies (see cls. 4.6 & 23). (for further reasoning see Summary 7.4 & associated analysis)

⁸⁶⁹ QR, sub. 26: 93-94.

⁸⁶⁸ Asciano, sub. 28: 19.

⁸⁷⁰ New Hope, sub. 24: 7.

⁸⁷¹ Aurizon, sub. 20: 44.

⁸⁷² Glencore, sub. 25: 3-4.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		undertaking itself (see also below). ⁸⁷³ Asciano said that the reporting criteria should be expanded, performance level disagreement should be disputes and remedies should be included. ⁸⁷⁴		
 Amendments to clauses:⁸⁷⁶ 2 (Access Rights); 3 (Operational Rights); 4.1 (Changes to Operator Nominations); 4.2 (Nominations with different Train Descriptions); 4.3 (Reduction of rights resulting in an Over- Allocation); 4.4 (Information); and 4.5 (Participation in Disputes) (Draft Decisions 7.3 & 7.4). 	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to these clauses which are required to provide for a tripartite structure which allows an end customer or an operator to be the access holder and have the flexibility to manage the utilisation of access rights.	New Hope generally supports the QCA's amendments with some minor amendments. However, New Hope has also proposed a number of more detailed amendments intended to clarify that an access holder can nominate multiple operators, ensure that access rights include train movements necessary to operate on the network and allow for varying train service descriptions over the life of the agreement (see clauses 2.1, 3.1, 3.2, 4.2 and 4.5 New Hope SAA). New Hope also made a number of specific comments in response to QR's submission on particular SAA clauses. ⁸⁷⁷ Aurizon submits that the scope of access rights granted under the	QR does not accept the QCA's proposed amendments. QR suggests that having an operator execute a copy subsequent to execution by the access holder and QR will not create a legally valid contract with the operator (cl. 2.2(b) & 4.5). QR also considers the operator nomination provisions to be vague and unclear especially in regards to QR's ability to reject a nomination (cl. 2.2(e) & 4.1). QR also considers that the amendments to cl. 4.2(a)(iii) could allow an access holder to avoid the relinquishment provisions. QR further submits that an access holder should be able to participate in disputes (cl. 4.6). ⁸⁸² QR also considers New Hope's proposed amendments to be unnecessary	The QCA's amendments are largely consistent with the Draft Decision. The QCA has made amendments to these clauses to provide for a tripartite structure and allow for an access holder to have the flexibility to manage the utilisation of its access rights. In response to QR's submissions, the QCA has also made some amendments to clarify that the 2015 SAA will be legally binding on operators who execute the agreement after the original parties have executed the same document (see new cl. 27.11). Further, the QCA does not agree with stakeholder submissions that the scope of the access rights granted under the SAA should be

⁸⁷⁵ QR, sub. 26, p. 95.

⁸⁷³ Yancoal, sub. 27: 4.

874 Asciano, sub. 28: 20.

⁸⁷⁶ All references to clauses in this Appendix, unless stated to be otherwise, are references to clauses of the QR 2105 SAA as amended by the QCA and attached to this Decision as Appendix G.

⁸⁷⁷ New Hope, sub. 32: 28.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		SAA should be expanded to include ancillary train services. ⁸⁷⁸ Aurizon also considers that the SAA should include an obligation to negotiate productivity variations in good faith subject to no financial disadvantage to QR, as well as limitations on relinquishment fees which arise due to productivity improvements. Aurizon does not support the QCA's proposed SAA structure but, nevertheless, within the proposed QCA SAA structure proposes stricter controls on confidential information between parties. ⁸⁷⁹ Glencore considers that the scope of what is contracted as access rights is defined too narrowly. ⁸⁸⁰ Asciano also noted that it is seeking that timeframes be clarified. ⁸⁸¹	and ineffective. ⁸⁸³	extended to include reference to ancillary train movements. These have been the subject of further agreement or charges for 'ancillary services' in the past and we consider it appropriate that they remain so. A number of changes from the Draft Decision have been made to allow for an access holder to nominate multiple operators who will then execute a substantially identical access agreement in order to maintain the tripartite structure and also to quarantine sensitive information between operators. The mechanisms for nominating and allocating access rights and changing nominations have also been clarified. (for further reasoning see Summaries 7.1, 7.2 & 7.3 & associated analysis as well as Draft Decisions 7.3 & 7.)
Amendments to clause 4.6 (Representations and Warranties)	Pursuant to Draft Decision 7.7, the QCA identified amendments	New Hope generally agrees with the QCA's amendments except that	QR said that schedules are not the appropriate place for warranties	The QCA's required amendments are consistent with the October

⁸⁸² See Queensland Rail, sub. 26: 94–95.

⁸⁷⁸ Aurizon, sub. 20, p. 23.

⁸⁷⁹ Aurizon, sub. 20, pp. 34-35 & 43.

⁸⁸⁰ Glencore, sub. 25, p. 4.

⁸⁸¹ Asciano, sub. 28: 19.

⁸⁸³ Queensland Rail, sub. 33: 63.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
(Draft Decision 7.7)	to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	it proposes to move the QCA's clause 4.7(a)(vii) to Schedule 5.	and that it is appropriate for parties to warrant as to the correctness of information. ⁸⁸⁴	2015 Draft Decision. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decision 7.7)
Amendments to clause 5 (Accreditation) (Draft Decision 7.7)	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope agrees with the QCA's amendments but proposes some additional amendments intended to reflect the fact that there may be more than one operator (cl. 5(d) New Hope SAA).	QR does not comment specifically on the QCA's proposed amendments.	The QCA's required amendments are largely consistent with the Draft Decision. (for further reasoning see Summary 7.2 & associated analysis as well as Draft Decision 7.7)
Transferral of clause 5.7 to Schedule 3 in QR's proposed 2015 SAA (Interim Take or Pay Notices) (Draft Decision 7.7).	The QCA proposed that, in the interest of clarity and efficiency, this clause would be better placed in the Reference Tariff schedule.	New Hope raised concerns with the inclusion of interim take-or-pay notices without an adjustment to reflect the threshold take-or-pay tests. ⁸⁸⁵ Aurizon has significant concerns with inclusion of interim take-or-pay notices. Aurizon recommends that the words "conclusive evidence" are removed so that the final year-end bill can be amended to reflect an adjustment of take-or-pay amounts (if required). ⁸⁸⁶	QR said that interim take-or-pay notices provides for timely and efficient resolution of disputes rather than having to wait up to 12 months to discover disputes. ⁸⁸⁷	The QCA's required amendments are largely consistent with the Draft Decision. The QCA agrees with stakeholders that the interim take-or-pay notices should not be deemed to be accepted nor should they be conclusive evidence of a stakeholder's take-or-pay obligations. However, we consider that the interim take-or-pay provisions should remain as there are times when it may be appropriate for take-or-pay to be calculated outside of the usual 12- monthly take- or-pay period. However, as New Hope said, there

⁸⁸⁴ Queensland Rail, sub. 33: 63.

⁸⁸⁵ New Hope, sub. 31: 21.

⁸⁸⁶ Aurizon, sub. 20, pp.49-50.

⁸⁸⁷ Queensland Rail, sub. 33: 44.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				should be an adjustment provision so that the approved ceiling revenue limit reflects all take-or- pay including take-or-pay invoiced via interim take-or-pay notices. This and some other consequential amendments to accommodate interim take-or-pay notices have been included in Schedule D to the Undertaking.
Amendments to clauses 6.1 (Access Charges), 6.2 (Obligation to make payments) and 6.6 (Adjustments) (Draft Decisions 7.3 & 7.4)	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to these clauses which are required to provide for a tripartite structure. The QCA also deleted clause 6.1(c) and amended the definition of "Access Charges" to clarify and streamline the mechanics of Take or Pay Charges.	New Hope agrees with the QCA's amendments to clause 6. However New Hope also proposes an amendment to clause 6.2(a) as the New Hope has concerns that its operator has received invoices dated earlier than the date of receipt. New Hope also agrees with QR that there should be clarity in relation to which party is responsible for take-or-pay obligations. ⁸⁸⁸ Aurizon is concerned that, because of a limited definition of access, an operator may be exposed to payment of ancillary charges for stowage, shunting etc (see Decision above). ⁸⁸⁹	QR interprets the QCA's amendments as removing the requirement to pay take-or-pay charges. Further, QR submits that the agreement is unclear as to the method of calculating take-or-pay if more than one operator is nominated. ⁸⁹⁰	The QCA's required amendments are largely consistent with the Draft Decision. However, the QCA agrees with stakeholder submissions to clarify that the date for payment of an invoice is ten BD's from the date the invoice is received (noting that this may also be subject to the Notices deeming provisions). The QCA does not consider that its amendments have removed the requirement for an access holder to pay take-or-pay charges. These remain as part of the definition of "Access Charges" and as part of the calculation of Access Charges which is outlined in Schedule 3 to Appendix G. However, the QCA agrees that, to clarify matters, the access holder should, in the first

⁸⁸⁸ New Hope, sub. 31: 21.
⁸⁸⁹ Aurizon, sub. 20, p. 44.
⁸⁹⁰ QR, sub. 26, p. 95.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				instance, be liable for access charges under the SAA (the payment obligations as between the access holder and operator can be left to the relevant haulage agreement).
				(for further reasoning see Summary 7.2 & associated analysis as well as Draft Decisions 7.3 & 7.4)
Insertion of new clause 6.7 (Performance Level Reporting Regime) (Draft Decision 7.10).	See comments above in relation to KPI reporting regime.	New Hope supports the QCA's inclusion of performance reporting requirements. However, New Hope is concerned that the KPIs do not contain any financial incentives and that the dispute provisions should apply to the KPI provisions (see cl. 6.7(e) & (f) New Hope SAA). ⁸⁹¹ Aurizon supports the QCA's amendments. However, Aurizon submits that the reporting requirements should be expanded to require QR to report to access holders on the track condition for each relevant system and any deviation from a previously established baseline condition. Aurizon believes that QR should have a positive obligation to ensure that its reported data is correct. ⁸⁹²	QR considers that mandating a weekly reporting regime imposes a significant administrative and cost burden on QR and may provide a distorted view of QR's performance as it does not account for seasonality, maintenance regimes or access holder operations. QR also considers that the QCA's proposed regime is one-sided as the operator is not subject to a mandatory performance regime. Further, QR said that performance levels should not be determined by third parties. ⁸⁹⁶	The QCA's required amendments are largely consistent with the Draft Decision. However, the QCA has included some additional reporting criteria, made the provisions subject to the dispute resolution provisions and removed the weekly reporting obligations. The QCA considers that financial incentives are a matter for the parties to negotiate once meaningful data is available. (for further reasoning see Summary 7.4 & associated analysis as well as Draft Decision 7.10)

⁸⁹¹ See also New Hope, sub. 4: 8 and New Hope, sub. 32:29.

896

⁸⁹² Aurizon, sub. 20, p. 54.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		Glencore supports the inclusion of a performance reporting regime. However, Glencore's preferred approach is for Schedule 5 of the SAA to be fully settled and approved by the QCA along with the 2015 DAU. If not, Glencore submits that QR should be required to submit a performance regime for approval by the QCA within 3 months of the undertaking being approved. ⁸⁹³		
		Yancoal's submissions on the performance reporting regime under the SAA are largely similar to Glencore's. ⁸⁹⁴		
		Asciano said that a failure to agree performance levels should be grounds for a dispute, the reporting requirements should be expanded and remedies should be included. ⁸⁹⁵		
Amendments to clauses 7.1 (Maintenance), 7.2 (Network Control), 7.3 (Compliance) and new clause 8.3 (Ad Hoc Trains) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to these clauses which are required to provide a more balanced risk position between QR and access seekers, access holders, and operators including by providing that Network Control must be conducted	New Hope supports the QCA's amendments. However, New Hope has proposed further amendments which are intended to make QR ensure that third parties are contractually bound (where relevant) to meet the same network maintenance requirements as QR (cl. 7.1(d) New	QR submits that the definition of "Maintenance Work" impermissibly extends QR's obligations to be inconsistent with s. 119 of the QCA Act. ⁹⁰³ QR also submits that it should not be liable for Third Party Works as it may not have control over those Third Parties (cl. 7.1(d)). Further, QR submits that the SAA	The QCA's required amendments include changing the definition of 'Maintenance Work' from that proposed in the Draft Decision to avoid any possible overlap with the definition of 'Extension' under the QCA Act. The QCA does not accept that the

⁸⁹³ Glencore, sub. 25, p. 4.
 ⁸⁹⁴ Yancoal, sub. 27, p. 4.

⁸⁹⁵ Asciano, sub. 28: 19–20.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
	subject to the Network Management Principles, making the maintenance requirements more consistent with the earlier regulatory precedents and imposing the same Prudent Practice obligations on QR which apply to access holders and operators.	Hope SAA) as well as provide that QR will consider adjoining networks. ⁸⁹⁷ Aurizon supports the QCA's amendments. However, Aurizon remains concerned that QR has the ability to unilaterally amend Interface Standards without an obligation to consult with operators. ⁸⁹⁸ Glencore generally supports the QCA's Draft Decision. However, Glencore considers it appropriate to require maintenance of the network in accordance with a general threshold such as "good operating and maintenance practice". ⁸⁹⁹ Asciano said that, where a change to the Interface Standards has the potential to impact on Queensland Rail's maintenance obligations, Queensland Rail should be required to consult with access holders and operators. ⁹⁰⁰ Asciano also considers that Queensland Rail's	should be clear that QR does not assume any liability for acting in accordance with its obligations under the TIA. ⁹⁰⁴ Queensland Rail also said that New Hope's proposed amendments (to cl. 7.1(d)) are inappropriate and that compliance with safety standards on adjoining networks is a matter for the relevant operator. ⁹⁰⁵ QR has submitted a new clause "Ad Hoc Train Services" which allow for an operator to request and QR to accept (in its discretion) to run an ad hoc train service which is not subject to take-or-pay charges. Queensland Rail also said that it is not willing to allow an operator to dictate to it what interface standards it may adopt.	deletion of cl. 7.2(e) exposes QR to liability for acting in accordance with its TIA obligations. The QCA also requires amendments to cl. 7.1 to provide that QR will procure that any third parties with whom QR has a contractual arrangement will meet the same network maintenance requirements as QR are required to meet. This is important to maintain the integrity of Queensland Rail's maintenance obligations. Changes to the definition of 'Interface Standards' in the ORM and the DAU have also been made which impose an obligation on Queensland Rail and operators to agree changes to the Interface Standards. This is appropriate given that these standards are agreed in the first instance and also provide the benchmark by which QR must maintain the network.

⁹⁰³ Queensland Rail, sub. 26: 96.

- ⁸⁹⁷ See New Hope, sub. 32: 29 and sub. 24 (SAA).
- ⁸⁹⁸ Aurizon, sub. 20: 45.
- ⁸⁹⁹ Glencore, sub. 25: 4.
- 900 Asciano, sub. 28: 20.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		obligations in relation to train control are unclear. ⁹⁰¹ Most stakeholders raised concerns with the Ad Hoc Train Services provisions. Stakeholders said that, at a minimum, Ad Hoc Train Services should be counted toward an access holder's take-or-pay obligations and QR should not be excused from negligence. ⁹⁰²		 We have not included Glencore's suggestion because the maintenance requirements under the DAU have also been strengthened. We have not included all of Asciano's suggested amendments as we consider that some of these are overly prescriptive. However, we have made some further amendments to other clauses where we have agreed with Asciano's statements and these are noted below. We do not agree with New Hope that QR should also be specifically required to take into account adjoining networks when exercising network control. Given the above required amendments and the requirements in the DAU to consult with other railway managers, we consider that these additional amendments would be overly prescriptive. We agree with stakeholders that QR should not escape liability for negligence merely because a train service is run on an ad hoc basis. However, we do not agree that the

904 QR, sub. 26, pp. 96-97.

⁹⁰⁵ Queensland Rail, sub. 33: 63.

⁹⁰¹ Asciano, sub. 28: 20.

⁹⁰² Yancoal, sub. 35: 3; Glencore, sub. 30: 3; New Hope, sub. 31: 24–25.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				use of ad hoc paths should reduce an access holders take-or-pay liability. This is because, whether or not an ad hoc path is used, QR will have still reserved the contracted train paths with access holders and it is entitled to expect that revenue. Also, QR will be prevented from using ad hoc train paths to gain a take-or-pay windfall by the Approved Ceiling Revenue Limit.
Amendments to clause 8.2 (Additional Train Services) (Draft Decisions 7.3 & 7.4).	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to this clause which are required to provide for a tripartite structure.	New Hope supports the QCA's amendments.	QR does not comment specifically on the QCA's proposed amendments.	The QCA's required amendments are largely consistent with the Draft Decision. (for further reasoning see Summary 7.2 & associated analysis as well as Draft Decisions 7.3 & 7.4)
Amendments to clause 8.4 (Compliance) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope generally supports the QCA's amendments. However, New Hope believes that the references to 'substances' or 'things' is overly vague and should be deleted (cl. 8.3(b)(v) & 8.3(c) New Hope SAA) and the notification provisions should be reciprocal. New Hope also said that QR's previous cl. 8.3(b)(viii) was overly broad. ⁹⁰⁶ Asciano said that the obligations to	QR submits that the QCA's proposed amendment unnecessarily limits QR's contractual rights. ⁹⁰⁸	The QCA's required amendments are largely consistent with the Draft Decision. However, the QCA agrees with stakeholder submissions in regard to the deletion of the use of the terms 'substances' and 'things' as overly vague. We do not accept Queensland Rail's submission that the amendment to cl. 8.3(b)(viii)

⁹⁰⁶ New Hope, sub. 32: 29.

⁹⁰⁸ Queensland Rail, sub. 26: 97.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		notify in relation to a failure to comply with the agreement should be limited to material failures and that the obligation should be		unnecessarily limits its rights. An Operator should be allowed to operate within the scope of its agreement.
		reciprocal. ⁹⁰⁷		We accept Asciano's suggested amendment as the use of 'material' will provide a meaningful threshold to the requirement to report breaches which may otherwise be an overly onerous requirement. We also agree that the obligation to notify of breaches should be reciprocal. We have also included that certain obligations on the Operator should be limited according to prudent practices. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decision 7.7)
Amendments to clause 8.5 (Compliance Before Commencing to Operate a Train Service) (Draft Decisions 7.3 & 7.4).	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to this clause which are required to provide for a tripartite structure.	New Hope generally supports the QCA amendments and disagrees with Queensland Rail's submission that the QCA's required amendments are not specific to certain train services (cl. 8.4(a)(i) and (iv)). ⁹⁰⁹	QR submits that the right to reduce the right to operate train services may not be sufficient to remedy non-compliance with the requirements of clause 8.4(a). ⁹¹⁰	The QCA's required amendments are largely consistent with the Draft Decision. The QCA does not accept QR's proposition that cl. 8.4(c) may not be sufficient to remedy compliance. The QCA considers that the wording of the clause is sufficiently broad enough to allow, for example, the reduction of all of

⁹⁰⁷ Asciano, sub. 28: 21. ⁹⁰⁹ New Hope, sub. 32: 29.

⁹¹⁰ QR, sub. 26, p. 97.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				an operator's right to operate if the relevant "Failure" affects all of the operator's Train Services.
				We have accepted QR's drafting in relation to the preparation, by the operator, of an operating plan and EIRMR before commencing operations.
				(for further reasoning see Summary 7.2 & associated analysis as well as Draft Decisions 7.3 & 7.4)
Amendments to clause 8.7 (Alterations to Train Services) (Draft Decisions 7.3, 7.4 & 7.7).	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to this clause which are required to provide for a tripartite structure. The QCA also amended the clause to include a new definition "Alternative Schedule Time" to promote a more balanced risk position between QR and access seekers and existing access holders in accordance with Draft Decision 7.7.	New Hope supports the QCA amendments with only minor amendments for consistency (cl. 8.6(c)).	QR submits that the definition of "Alternative Schedule Time" and "Useable Schedule Time" does not balance the interests of all the parties but rather places the obligation on QR to balance the interests of individual operators against the efficiency of the supply chain. ⁹¹¹	The QCA's amendments are consistent with the Draft Decision. The QCA does not accept QR's submission that the required amendments place unreasonable obligations on QR. The operation of the provisions and the relevant definitions do not place an absolute obligation on QR to replace a scheduled Train Service rather QR is required to use reasonable endeavours. The reasonable endeavours of QR are further limited if providing a useable schedule time would trigger additional costs to QR or it would alter other train movements. ⁹¹² Also, placing these obligations on QR is appropriate given QR has information in

⁹¹¹ QR, sub. 26, p. 102.

⁹¹² See cl. 8.6(a)(ii) of the 2015 DAU.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				relation to all train services. An operator may have limited access in relation to other operators.
				(for further reasoning see Summaries 7.2 & 7.3 & associated analysis as well as Draft Decisions 7.3 7.4 & 7.7)
Amendments to clauses 8.8 (Operator to Supply Information), 8.9 (Queensland Rail may Supply Data) & 8.10 (Authorisations of Rolling Stock and Train Configurations) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to these clauses which, amongst other things, are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA's amendments. However, New Hope proposes some additional amendments to cl. 8.8 which are intended to oblige QR to collect data and, if requested by an access holder or operator, supply data to that party. New Hope also want to ensure that QR warrants as to the accuracy of the data and that requesting parties do not have access to other parties' information and that QR acts reasonably in relation to communication links. Aurizon supports the QCA's amendments. However Aurizon also proposes some additional amendments to allow operators to use information provided by QR from equipment that is provided for in the RAB as required for part of the regulated service. ⁹¹³ Asciano said that Queensland Rail's cl. 8.7 is unduly oppressive. Asciano considers that Queensland Rail should be required to consult in	QR submits that the QCA's proposed amendments are not commercially practical because it is not feasible or possible for the operator to retain the intellectual property in data collected by QR's train control systems. Further, QR contends that the intellectual property provisions may limit the use of supplied data by QR for billing and/or reporting purposes. ⁹¹⁶ QR said that the operation of an effective interface between communication systems is critical to network safety and should not be subject to a reasonableness test. QR also said that, in relation to cl. 8.9, it should not be subject to a reasonableness test. ⁹¹⁷	The QCA's amendments are largely consistent with the Draft Decision. However, we accept Asciano's submission in relation to cl. 8.8 that, at a minimum, Queensland Rail should be required to minimise cost and disruption if it seeks to upgrade communication links. Further, we consider that consultation with operators is reasonable. These obligations will not compromise network safety but rather they are likely to increase safety by providing that every stakeholder is kept abreast of communication requirements. We do not accept New Hope's submission in relation to Queensland Rail supplying data. In this regard, we have largely maintained our Draft Decision which provides that the party which supplies data retains the IP. The parties are free to substantiate the question of supply between themselves.

⁹¹³ Aurizon, sub. 20: 46.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		relation to communication system upgrades and also minimise cost and disruption to operators. ⁹¹⁴ Asciano also agreed with the QCA's amendments to cl. 8.8 but said that approval should be in writing. Further, Asciano said that Queensland Rail should not be allowed to unreasonably withhold its consent to an operator's rolling stock certification. ⁹¹⁵		 However, we have made the obligation on QR to supply data mandatory if reasonably requested by an operator. We have also included a number of amendments submitted by QR. We agree with Asciano that Queensland Rail should not unreasonably withhold its approval of an operator's certification. (for further reasoning see Summaries 7.2 & 7.3 & associated analysis as well as Draft Decisions 7.3 7.4 & 7.7)
Deletion of clause 8 in QR's proposed 2015 SAA (Operating Requirements Manual) (Draft Decisions 7.4 & 7.7).	The QCA removed the ability of QR to amend the ORM under the terms of the SAA. QR retains the ability to amend the ORM unilaterally for 'Safety Matters' or other minor matters under the terms of the QCA's proposed amended undertaking; and, with QCA approval, amend for other matters by submitting a draft amending access undertaking. This change means that, in	New Hope supports the QCA's deletion of this clause (see cl. 8.12 New Hope SAA). Aurizon supports the QCA's amendments but would like the provisions to provide additional obligations on QR to consult with operators prior to notification of a change to the ORM for safety matters. ⁹¹⁸	QR submits that the deletion of this clause and moving the ORM amendment provisions to the undertaking is inappropriate in that the QCA's reasoning for doing so is incorrect. QR believes that the ORM as part of the undertaking is only a 'snap shot' of the ORM and from then on changes should be contractual. ⁹¹⁹	The QCA's SAA amendments are largely consistent with the Draft Decision. However, there have been some further changes made to the ORM provisions within the DAU (see chapter 4 of this Decision).

⁹¹⁶ Queensland Rail, sub. 26, p. 97.

⁹¹⁷ Queensland Rail, sub. 33: 64.

⁹¹⁴ Asciano, sub. 28: 21.

⁹¹⁵ Asciano, sub. 28: 22.

⁹¹⁸ Aurizon, sub. 20, p. 46.

⁹¹⁹ QR, sub. 26, p. 97.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
	accordance with Draft Decision 7.4 and 7.7, the risk and responsibilities between QR and access seekers, access holders and operators is maintained whilst providing QR the ability to act more quickly in response to 'Safety Matters'.			
Amendments to clause 8.12 (Compensation) and 8.13 (Replacement of Operating Requirements Manual) (Draft Decisions 7.4 & 7.7).	The QCA, pursuant to Draft Decisions 7.4 & 7.7, identified amendments which appropriately balance the risks and responsibilities between the parties to the SAA by allowing the parties to the SAA to negotiate in relation to appropriate compensation for changes to the ORM which affect the parties.	New Hope supports the QCA's amendments. ⁹²⁰ Aurizon generally supports the QCA's amendments.	QR submits that the QCA's proposed amendments are not permitted by the QCA Act as QR contends that the QCA is purporting to impose a compensation process on QR for the exercise of statutory rights. QR also submits that it is inappropriate to disconnect the making of amendments from the compensation process as it potentially leads to inefficient Decision making. ⁹²¹	We have removed the compensation provisions from the ORM. QR, in order to amend the ORM will be required to submit a DAAU for all changes to the ORM. When considering the appropriateness of any DAAU, the QCA may, at that stage, also consider whether compensation is appropriate.
Amendments to clause 9 (Interface Risk Management) (Draft Decision 7.7)	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA's amendments. New Hope has also proposed some additional amendments which are intended to make certain obligations in relation to compliance with the IRMP symmetrical between all parties (cl. 9.1(c) & (e) and 9.4(c) New Hope SAA). New Hope also proposes amendments which are	QR submits that, in cl. 9.6(d)(iv), the obligation to use 'all reasonable endeavours' to mitigate any loss or damage arising from the conduct of an inspection or audit is unclear and unnecessary. ⁹²² Queensland Rail also said that arrangements for an assessment by the access holder of an operator's	The QCA's required amendments are largely consistent with the Draft Decision. The QCA also agrees with stakeholder submissions that obligations in relation to interface risks should be symmetrical. The QCA does not agree with QR that the obligation to use

⁹²⁰ New Hope, sub. 32: 30. ⁹²¹ QR, sub. 26, p. 98.

⁹²² Queensland Rail, sub. 26, p. 98.

		Queensland Rail's ("QR") Submissions	QCA Response and Decision
	intended to accommodate multiple operators.	compliance with the IRMP is a matter for a rail haulage agreement. ⁹²³	reasonable endeavours to mitigate damage is unclear. The meaning is clear on the face of the subclause.
			We have included some amendments submitted by QR in relation to changing an operating plan, ad hoc train services and access to inspect relevant property. We have also included additional mentions of environmental harm and requirements suggested by QR.
			(for further reasoning see Summary 7.3 & associate analysis as well as Draft Decision 7.7)
Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required	New Hope supports the QCA's amendments. New Hope also proposed a number of	QR is concerned that, by introducing the term "acting reasonably" rail safety, engineering	The QCA's required amendments are largely consistent with the Draft Decision.
to provide a more balanced risk position between QR and access seekers, access holders & operators.	balanced risk position including, placing obligations on QR in relation to conflicting IRMPs as QR is in the position of having access to all operators' IRMPS. Also, New Hope proposes to delete part of clause 10.2(c) which New Hope considers to negate the consultation requirements. New Hope also proposes extensive amendments to clause 10.7 to provide that operators only bear	and other operational requirements may be "watered down", disputed or replaced. QR states that it cannot be in a position where a third party dictates safety matters relating to its rail network. Further, Queensland Rail said that it should not be liable for failures in an operator's EMP. Queensland Rail also opposed New Hope's suggested deletion of cl. 10.2(b) as it considers that it may	Also, the QCA agrees with stakeholder submissions that the obstruction provisions should be symmetrical and that QR should be obliged to raise an objection to an operator's proposed EMP if the plan is inconsistent with QR's or another operator's EMP. This accords with our decision to allocate risks and obligations to the parties best able to manage the risks. QR is the only single party which has access to every
1 1 1 9	the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders &	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	operators.matter for a rail haulage agreement.923Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.New Hope supports the QCA's amendments. New Hope also proposed a number of amendments intended to provide a balanced risk position including, placing obligations on QR in relation to conflicting IRMPs as QR is in the position of having access to all operators' IRMPS. Also, New Hope proposes to delete part of clause 10.2(c) which New Hope considers to clause 10.7 to provide that operators only bear the direct cost of noise mitigationQR is concerned that, by introducing the term "acting reasonably" rail safety, engineering and other operational requirements may be "watered down", disputed or replaced. QR states that it cannot be in a position where a third party dictates safety matters relating to its rail network. Further, Queensland Rail also opposed New Hope's suggested deletion of cl. 10.2(b) as it considers that it may be required to remove obstructions

⁹²³ Queensland Rail, sub. 33: 64.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
-	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions method is on the train, or where an unusual feature of a particular operator's train triggers the need for noise mitigation. ⁹²⁴ Aurizon considers that QR, when removing obstructions, should give consideration to whether or not the proposed form of removal could cause any actual or potential environmental harm. ⁹²⁵ Asciano said that Queensland Rail should make its emergency management plan ("EMP") available to Operators and that Queensland Rail's satisfaction should not be unreasonably withheld or delayed. ⁹²⁶ Asciano also made a number of statements in relation to noise mitigation, including that Queensland Rail should act reasonably. ⁹²⁷		QCA Response and Decisionraises objections to inconsistencies.The QCA does not accept Queensland Rail's submission that the inclusion of the words 'acting reasonably' may result in Queensland Rail's safety or operational requirements may be 'watered down' The inclusion of these words is intended to prevent QR from acting whimsically or irrationally where it is given a wide discretion which could dramatically affect an operator's operations.The QCA agrees with Asciano's submission and considers that Queensland Rail's satisfaction in relation to an operator's proposed EMP should not be unreasonably withheld or delayed. We do not consider it necessary to provide that, in this clause, Queensland Rail must provide a copy of its EMP because Queensland Rail has already outlined its EMP in the
				ORM and also undertaken to consult in relation to amendments

⁹²⁴ See, in addition to cl. 10.7, New Hope, sub. 24: 7; New Hope, sub. 32: 30.

- ⁹²⁶ Asciano, sub. 28: 22.
- ⁹²⁷ Asciano, sub. 28: 22–23.
- ⁹²⁸ See Queensland Rail, sub. 33: 64.
- ⁹²⁹ Queensland Rail, sub. 33: 46.

⁹²⁵ Aurizon, sub. 20: 46-47.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				(see cl. 4.1 of ORM).
				We agree with Asciano that it is reasonable for Queensland Rail to substantiate its noise mitigation expenses. However, we have not included all amendments proposed by stakeholders as they may overcomplicate the provisions which already contain sufficient checks on QR's discretion.
				The QCA does not agree with Aurizon's suggestion as the QCA considers that QR's obligation to comply with all laws, laws which include, for example, s. 319 of the <i>Environmental Protection Act 1994</i> (Qld) which imposes a general obligation to avoid environmental harm, are sufficient.
				We accept New Hope's submission that the final paragraph of 10.2(d) may negate other parts of 10.2.
				We do not accept Queensland Rail's argument that 10.2(b) should not be removed as it may impinge on the timely removal of obstructions. The obligation to consult is tempered by a reasonableness test. If the removal of an obstruction is so pressing that full consultation is not reasonable in the circumstances, then QR would not be required to fully consult. QR should also not be allowed to cause obstructions which may result in cancellations

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				without any consequences. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decision 7.7)
Amendments to clause 11 (Inspection of Trains and Rolling Stock) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA's amendments.	QR does not comment specifically on the QCA's proposed amendments.	The QCA's amendments are consistent with the Draft Decision. (for further reasoning see Summary 7.3 & associated reasoning as well as Draft Decision 7.7)
Amendments to clause 12 (Risk and Indemnities) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators. This includes, amongst other things, requiring the risks associated with Dangerous Goods to be dealt with by the general liability provisions and removing the overly onerous requirement for QR to approve every driver employed by an operator.	New Hope supports the QCA's amendments but proposes some slight amendments to the wording of clauses 12.1(a), (b) & (c) (New Hope SAA) which are intended to limit the liabilities of the parties to the same scope as the benefits each party gains under the agreement. ⁹³⁰ Asciano said that the definition of 'Operator's Customer' is overly broad and that the clause potentially exposes operators to liability for matters which they have not caused. ⁹³¹ Asciano also said that it should not have to provide its conditions of carriage to Queensland Rail and that	QR submits that the QCA's proposed amendment to clause 12.2(c) undermines the intent of the clause by potentially exposing QR to liability to an operator's customer which is not a party to the agreement. Further, QR contends that, if the indemnity for carriage of Dangerous Goods in cl. 12.3 is deleted QR will be obliged to factor the increased risk into the access charges and that the risk of carriage of Dangerous Goods is under the primary control of the operator. ⁹³³ Queensland Rail also said that by deleting this indemnity the QCA is altering the risk profile of the SAA	The QCA's required amendments are largely consistent with the Draft Decision. We agree with Asciano that operators should not be liable to QR for matters over which they potentially have no control. This is why cl. 12.2(d) was deleted (see below); we have also amended the definition of "Operator's Customer". We also agree with Asciano that the requirement to provide its conditions of carriage to QR should be limited. The QCA considers that if QR will not be in a position to have Associates perform its obligations then cl. 12.5 will have no

⁹³⁰ New Hope, sub. 32: 30.

⁹³¹ Asciano, sub. 28: 23–24.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		Queensland Rail has the ability to insure itself for its liability. ⁹³²	in a material way, detrimental to Queensland Rail. ⁹³⁴ QR also submits that Third Party Works may not be carried out by QR's associates but by third parties engaged by the State. In these circumstances, QR believes it should not be liable for these third parties. ⁹³⁵ QR also said that it must be satisfied that agents/contractors of an operator/access holder are appropriately qualified. ⁹³⁶	application to it so does not accept QR's proposed deletion of this obligation. Further, the QCA accepts QR's statement that third parties who conduct work on or around the network on behalf of the State may not be engaged by QR and so QR should not be liable for these parties. The QCA deleted cl. 12.5(d) as the QCA agrees with stakeholders that this obligation, relating to every operator's driver, is overly onerous. In relation to cl. 12.2(d), the QCA does not believe it is appropriate for other parties to indemnify QR for losses which may occur as a result of QR's acts or omissions. Further, we agree with New Hope that the wording around the indemnities (i.e. 'or otherwise in connection with this agreement') to be overly broad and should be restricted to the scope of the benefit obtained under the agreements (Queensland Rail agreed with this amendment in its

⁹³³ Queensland Rail, sub. 26: 99.

932 Asciano, sub. 28: 24.

⁹³⁴ Queensland Rail, sub. 33: 46.

⁹³⁵ QR, sub. 26, p. 99.

⁹³⁶ Queensland Rail, sub. 26: 99.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				March 2016 submission). This is more consistent with the previous regulatory precedents. (for further reasoning see Summary 7.2, 7.3 & associated analysis as well as Draft Decisions 7.3 7.4 & 7.7)
Amendments to clause 13 (Limitations on Liability) (Draft Decisions 7.4 & 7.7).	Pursuant to Draft Decisions 7.4 & 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators. This includes, amongst other things, by limiting exceptions to the exclusion on liability for Consequential Loss and making QR's liability for failing to maintain the network more similar to the earlier regulatory precedents.	New Hope supports the QCA's amendments. However, in addition, New Hope proposes to remove the 'urgent possession' definition as New Hope considers that maintenance is either planned or an emergency (cl. 13.5 & 13.6 New Hope SAA). ⁹³⁷ New Hope also considers that the 10% threshold in relation to non-provision of access (cl. 13.6(d)) should be deleted as the clause would promote over- contracting behaviour by customers. ⁹³⁸ New Hope also supports the QCA's amendments to cls. 13.1, 13.4 & 13.6. ⁹³⁹ Glencore considers that clause 13.6(d) should be deleted as there are already significant protections for QR and it is not appropriate to pass the risk (of non-provision of	QR submits that the deletion of references to the indemnities in cls. 12.2 to 12.3 and cl. 27.18 within cl. 13.1(b), renders cl. 13.1(b) ineffective. QR contends that the QCA's proposed amendments to clause 13.4(b) increase the risk to QR in a manner which is inconsistent with the earlier regulatory precedents and that the QCA has not explained why it is appropriate. QR also considers that a cap on QR's liability should be reintroduced into cl. 13.4(c) as its deletion creates "significant downside systemic risk" to QR. In relation to QR's liability for non- provision of access, QR considers that the QCA's proposed amendment is a departure from	The QCA's required amendments are largely consistent with the Draft Decision. However, we have re-introduced the word 'directly' into clause 13.4(b) to make the provision more consistent with the same clause in the regulatory precedents. However, we remain of the opinion that maintenance of the network is of critical importance so as to warrant strong liability provisions. We also agree with stakeholders in relation to the removal of the 10% threshold in relation to claims for non-provision of access. We consider that QR is adequately protected by the remainder of the clause and the inclusion of this additional limitation

⁹³⁷ See also discussion in New Hope, sub. 23, p. 16.
⁹³⁸ New Hope, sub. 31: 22–23.

⁹³⁹ New Hope, sub. 32: 30.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		access) onto the party least able to control that risk (access holder). ⁹⁴⁰ Yancoal also submits that the closure of Wilkie Creek and reduction in non-coal traffic has produced spare capacity such that QR should not be allowed the 10% threshold in addition to its other protections regarding non- provision of access. ⁹⁴¹ Asciano said that cl. 13.6(d) allows Queensland Rail to avoid liability for non-provision of access and shifts the risk of non-performance to access holders. ⁹⁴²	the earlier regulatory precedents and is unexplained. Queensland Rail also said that the 10% threshold is making explicit a threshold which was left to be agreed in the previous regulatory precedents. ⁹⁴³ Queensland Rail also said that disruptions to business operations resulting in delays to train movements can be significant. ⁹⁴⁴	inappropriately tips the balance of risk in QR's favour. We acknowledge that the regulatory precedents provided for a threshold to be agreed. When negotiating the performance levels & KPI's, the parties will be required to also turn their minds to this threshold and, if required, negotiate amendments to this clause. The QCA considers that consistent with general contractual principles it is appropriate to limit liability for Consequential Loss between the parties. We have also made cl. 13.3 more specific as this clause was potentially very broad. Consistent with regulatory precedents we have removed the liability cap. It is unbalanced to have QR's liability capped but have an access holder or operator's liability uncapped. We agree with Queensland Rail that certain material business interruptions could cause delays to train movements. We have also included reference to certain

⁹⁴⁰ Glencore, sub. 25: 4 and sub, 30: 3.

⁹⁴¹ Yancoal, sub. 27, p. 4 and sub. 35: 3.

942 Asciano, sub. 28: 24.

⁹⁴³ Queensland Rail, sub. 33: 47–48.

⁹⁴⁴ Queensland Rail, sub. 33: 64.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				environmental harm as an emergency. In relation to QR's comments regarding departure from the regulatory precedents, as stated above, QR considers the regulatory precedents indicative and relevant but does not consider that the regulatory precedents should necessarily be carried over identically to the 2015 SAA in every instance. (for further reasoning see Summary 7.2, 7.3 & associated analysis as well as Draft Decisions 7.3 7.4 & 7.7)
Amendments to clause 14 (Suspension) and 15 (Termination) (Draft Decisions 7.3, 7.4 & 7.7).	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to this clause which are required to provide for a tripartite structure. In doing so, the QCA also identified amendments which are separate the risks and responsibilities between the parties appropriately.	New Hope supports the QCA's amendments. New Hope also suggests inserting an additional reasonableness test on QR in clause 14.1 and reiterating notice requirements in a new clause 14 so that the general notification clause is not inadvertently overlooked in relation to suspensions. New Hope also proposes some minor amendments which are in accordance with an ability to vary a train service description. New Hope also said that it is imperative that	QR takes issue with a number of amendments to these clauses including: cl. 14.1(ii) QR considers that if the rights of the access holder are suspended then the rights of the operator should be also; cl. 14.1 & 15, QR consider that the obligation to act reasonably is asymmetrical and is a deviation from the earlier regulatory precedents; 15.2, QR submits that access rights should terminate where no operator is nominated; and, cl. 15.6 QR submits that	The QCA's required amendments are largely consistent with the Draft Decision. However, the QCA agrees with QR that if the access rights of an access holder are suspended then the rights of any related operators should also be suspended. The QCA accepts that the obligation to act reasonably should be symmetrical. This is consistent with the QCA's reasoning in both the Decision and the Draft
QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
---	------------------------------	--	---	--
		an operator cannot 'lose' the end user's access rights. ⁹⁴⁵ Aurizon has noted, in keeping with its opposition to the QCA's proposed tripartite structure, that there is an inconsistency between the termination provisions and the security provisions. ⁹⁴⁶ Aurizon does not otherwise specifically comment on the QCA's amendments. Glencore considers that the references to 'substances or things' should be removed. ⁹⁴⁷ Asciano said that the words 'such consent not to be unreasonably withheld or delayed' should be added to the end of cl. 15.6(b). ⁹⁴⁸	change in control provisions should also apply to the operator. ⁹⁴⁹ Queensland Rail also said that there is no reason why an operator should not be required to remove any substance or thing brought onto the network. ⁹⁵⁰	Decision in relation to risk balance. The QCA does not agree that access rights should be terminated where no operator is nominated as there may be variations to nominations during the term of the agreement; or there may be sometime between an access holder executing the agreement and nominating an operator. The "Reduction" provisions should otherwise mitigate against idle access rights. There are a number of additional amendments which are in accordance with the tripartite structure. The QCA also requires the deletion of the reference to 'substance or thing' as overly vague and broad. We agree that Asciano's suggested amendment is reasonable, QR should not be allowed to act unreasonably. (for further reasoning see Summary 7.2, 7.3 & associated analysis as well as Draft Decisions

⁹⁴⁵ New Hope, sub. 32: 31.

- ⁹⁴⁶ Aurizon, sub. 20, p. 37.
- ⁹⁴⁷ Glencore, sub. 25, p. 4.
- 948 Asciano, sub. 28: 25.
- ⁹⁴⁹ Queensland Rail, sub. 26: 100–101.
- ⁹⁵⁰ Queensland Rail, sub. 33: 64.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				7.3 7.4 & 7.7)
Amendments to clause 16 (Insurance) (Draft Decisions 7.3 & 7.4).	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to this clause which are required to provide for a tripartite structure and balance the risks and responsibilities between the parties appropriately.	New Hope refers to Aurizon's submissions regarding the insurance provisions. New Hope also said that where the role of the end user and the operator are separated appropriately it will be clear which party carries the risk and therefore the appropriate insurance. Further, if QR is correct that joint insurance is not possible, the clause is likely to stand but be unused. ⁹⁵¹ Aurizon identified a number of concerns with the insurance provisions including that it is not acceptable for another party to determine what the acceptable exclusions from a policy are and that provision of copies of insurance rather than certificates of currency is unreasonable. ⁹⁵² Aurizon has included its proposed draft for the insurance provisions in attachment 1 to its submission.	QR submits that it is unclear how the amended cl. 16.11 would operate in practice. QR has made a range of comments in relation to Aurizon's suggested insurance clause amendments. ⁹⁵³	The QCA's required amendments are largely consistent with the Draft Decision. In relation to Aurizon's suggested amendments we have, for the most part, not included these. We consider that Queensland Rail's drafting remains largely appropriate.
Amendments to clause 17 (Security) (Draft Decision 7.7)	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk	New Hope supports the QCA amendments. However, New Hope also proposes to oblige QR to undertake a review of another	QR submits that the QCA's amendments introduce unnecessary uncertainty to the agreement.	The QCA's required amendments are largely consistent with the Draft Decision. However, the QCA also requires

⁹⁵¹ New Hope, sub. 33: 31. ⁹⁵² Aurizon, sub. 20: 47.

⁹⁵³ Queensland Rail, sub. 33: 48–51.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
	position between QR and access seekers, access holders & operators. The amendments also help to promote efficient contracting between QR and access seekers as well as being more similar to the earlier regulatory precedents	party's security if that party requests it. New Hope also notes the QR should be made to return any security after the expiry of the agreement expeditiously to limit the security provider's costs in relation to maintaining the security (see cl. 17.4(b) New Hope SAA). New Hope also said that the financial capability test should be clarified. ⁹⁵⁴ Aurizon generally supports the QCA's amendments. However, Aurizon considers that the financial capability test as to whether a party is required to provide security should be clarified so that if a party has an acceptable credit rating it should not be required to provide security. ⁹⁵⁵ Asciano also said that the financial capability test should be clarified and the limitations on Queensland Rail's ability to call on security should be imposed. ⁹⁵⁶	QR said that the provisos in relation to the provision of security are unclear and that an access holder or operator should not be able to request a review of security more than once in a 12 month period. ⁹⁵⁷	the inclusion of a clause which provides that a security provider (acting reasonably) can oblige QR to undertake a review of its security requirements but no more than once in any 12-month period. The QCA also agrees with stakeholders that QR should be made to return security expeditiously, at the end of the term of an access agreement (or if an access agreement is assigned). We do not consider it appropriate to further specify the financial capability test. We believe that the amendments made provide scope for the parties to negotiate and agree on what is suitable security in the particular circumstances. Further, we do not propose to impose further limitations on the calling of security. If there are bona fide disputes these can be brought in the usual manner. The QCA does not agree with stakeholders that a contracting party (other than QR) should not be required to provide security.
Amendments to clause 18 (Adjustment for Changes) (Draft	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified	New Hope said that if an access holder is required to take the	QR said that the material change clause is intended to ensure that	The QCA agrees with New Hope that QR should not be able to

⁹⁵⁴ New Hope, sub. 32: 31.

⁹⁵⁵ Aurizon, sub. 20, p.48.

956 Asciano, sub. 28: 25.

⁹⁵⁷ Queensland Rail, sub. 33: 65.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
Decisions 7.3 & 7.4).	amendments to this clause which are required to provide for a tripartite structure.	downside of a material change it should be entitled to any upside as well. ⁹⁵⁸ New Hope also said that the adjustment for material change clause should not operate in relation to reference tariff service as these changes are already provided for in the DAU. ⁹⁵⁹ New Hope said that if material changes result in an access agreement uneconomic then access holders should be able to terminate their agreement. ⁹⁶⁰ Aurizon does not agree that in all cases any change in government funding should automatically result in an access charge review. Aurizon proposes that the material change clause should only permit a review of access charges for a change in government. funding where the access charge is below the revenue floor limit. Also, Aurizon considers that QR should advise the access holder of the term of relevant infrastructure TSC funding and an access holder should be able to terminate the access agreement where changes to access charges	QR is kept whole during the term of an agreement and not intended to provide a windfall to access holders. Also, QR said, the endorsed variation event trigger has a 2.5% threshold which, if not breached, could result in QR bearing costs it cannot recover so should be deleted. ⁹⁶² QR said its ability to disclose information in relation to TSC payments is subject to confidentiality obligations but that it is also under an obligation in the QCA Act to provide information about how it calculates access charges. Queensland Rail accepts that an access holder should be able to terminate its access agreement if it becomes uneconomic if this right is reciprocal. ⁹⁶³	circumvent the mechanisms within the DAU for variations to reference tariffs. Further, the 2.5% has been factored into the mix of costs, risks benefits and liabilities and should remain. We do not propose to limit material changes resulting from TSC variations to those which would result in the ceiling floor limit being breached. The TSC payments have also been factored into the mix of costs, risks benefits and liabilities and the access charges determined based on this mix should remain. We have not included a right for the parties to terminate if the agreement becomes uneconomic. Given that Queensland Rail agrees with this inclusion it may be that the parties can agree this amongst themselves. (for further reasoning see Summary 7.2, 7.3 & associated analysis as well as Draft Decisions 7.3 7.4 & 7.7)

⁹⁵⁸ New Hope, sub. 24: 59 of NHC SAA. ⁹⁵⁹ New Hope, sub. 31: 23.

⁹⁶⁰ New Hope, sub. 31: 23.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		due to a material change make the agreement uneconomic. ⁹⁶¹		
Amendments to clause 19 (Disputes) (Draft Decision 7.3, 7.4 & 7.7).	Pursuant to Draft Decisions 7.3, 7.4 and 7.7, the QCA identified amendments to this clause which are required to provide for a tripartite structure and more appropriately balance the risk position under the SAA.	Stakeholders supported the QCA's amendments.	QR regards that the effect of the QCA's amendment may be that QR may incur additional liability because third parties dictate safety requirements relating to its rail network. ⁹⁶⁴ QR said that New Hope's proposed amendments to cl. 19.4(a) undermine the tripartite agreement by allowing an access holder to lodge disputes which should be the preserve of accredited rail transport operators. ⁹⁶⁵	The QCA's required amendments are consistent with the Draft Decision. The QCA does not accept QR's statement in relation to disputes being arbitrated by QR. QR said that it is a misreading of the clause to state that it is inappropriate to allow QR to determine safety disputes without reference to the Rail Safety Regulator because QR's proposed clause 19.5 provides that only disputes which are 'not otherwise resolved by the RSR' can be determined by QR. The QCA's position is that safety disputes should, unless otherwise resolved, be determined by the RSR and there should not be a fall back to a position where, in certain circumstances, QR is the arbiter in its own disputes. Further, we consider that all parties should have the ability to refer disputes to the rail safety operator. This is consistent with a

⁹⁶² Queensland Rail, sub. 33: 51, 65.

⁹⁶³ Queensland Rail, sub. 33: 51.

⁹⁶¹ Aurizon, sub. 20: 48-49.

⁹⁶⁴ QR, sub. 26, p. 101.

⁹⁶⁵ Queensland Rail, sub. 33: 65.

			tripartite agreement. (for further reasoning see
			Summary 7.2 & 7.3 & associated reasoning as well as Draft Decisions 7.4 & 7.7)
Majeure) (Draft Decisions 7.1 & 7.7).pursuant to Draft Decisions 7.1 & 7.7, under the schedule of Access Principles, which the QCA agreed should be removed (see above, Draft Decision 7.1), the obligation of an access holder to pay Access is recompleted to the pay AccessNew H QCA's the access 	e QCA's amendments. ⁹⁶⁶ w Hope and Yancoal support the A's amendments but said that e access charges schedule to the A should be amended so that it econciled with the proposed proved ceiling revenue limit. ⁹⁶⁷	QR considers that suspending payment obligations when there is a FME is inappropriate in a price cap regulatory model. Further, QR submits that it cannot be forced to make a claim on its insurance where the cost of repairing or replacing part of the network is not economic. ⁹⁶⁸	The QCA considers that it strikes an appropriate balance between who bears the risk of an FME and incentivising Queensland Rail to reinstate the relevant part of the network to allow Queensland Rail to recover 50% of expected reference tariff access charges via the review event provisions and a subsequent variation to the reference tariff if the relevant track is repaired or replaced. The amendments required for this are largely within Schedule D to Appendix F. We have also reinstated the provisions from the regulatory precedents which provide for termination of the agreement for a prolonged FME (other than an event which damages or destroys the network). We have also made some amendments to clarify the operation of the provisions. (for further reasoning see Chapter

⁹⁶⁶ See New Hope, sub. 32: 31 & sub. 31: 17; Glencore, sub. 30: 4.
 ⁹⁶⁷ Yancoal, sub. 35: 3.

⁹⁶⁸ QR, sub. 26, p. 101.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				8 of the Decision)
Amendments to clause 21 (Relinquishment of Access Rights) (Draft Decisions 7.3, 7.4 & 7.7).	Pursuant to Draft Decisions 7.3, 7.4 and 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators as well as necessary amendments to provide for a tripartite structure.	New Hope generally supports the QCA's amendments. However, New Hope has proposed some amendments which it considers better reflect the way that the Western System operates (in particular ABCD scheduling) (cl. 21.1(a)(i) New Hope SAA). New Hope has also proposed deleting the requirement for an expert to determine whether an access holder has demonstrated to QR's reasonable satisfaction an existing access holder's sustained requirement for the relevant access rights (cl. 21.1(c) New Hope SAA). New Hope also consider that any relinquishment fee which QR recovers due to QR subsequently contracting relinquished access rights should be refunded (cl. 21.2(g) New Hope SAA). Aurizon considers that where an operator is seeking to implement certain operational efficiencies, relinquishment fees associated with a variance to train service entitlements and rolling stock configurations should be capped to the variation in access revenue arising from that change. ⁹⁶⁹	QR said, amongst other things, that New Hope's proposed amendments leave the question of possible rebates of relinquishment fees open indefinitely. ⁹⁷³	The QCA's amendments are largely consistent with the Draft Decision. We do not consider it appropriate to amend the scheduling provisions as we consider that the change is not justified and would require an assessment over a 12- month period which may unduly limit the reduction provisions. We do not propose to amend the relinquishment provisions to allow for relief from fees due if the relevant relinquishment is as a result of productivity improvements. This is because we consider that QR should be incentivised to make variations to access agreements or reference train services to improve efficiency. The access holders and operators should be incentivised by their reductions in above rail costs. However, we have amended the definition of "Relinquishment Fee" to allow for the parties (each acting reasonably) to agree to vary the calculation of any relinquishment fee (note that the Relinquishment Fee has also been limited to 80% of the NPV of future access charges – discussion

⁹⁶⁹ Aurizon, sub. 20, pp. 34-35. ⁹⁷³ Queensland Rail, sub.33: 65

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		Glencore submits that relinquishment fees should be refunded to an access holder where QR subsequently contracts access rights which they would have otherwise been unable to contract but for the relinquishment and agrees that relinquishment fees should be reduced if they are incurred as a result of efficiency improvements. ⁹⁷⁰ Asciano also said that the relinquishment fee provisions allow Queensland Rail to double-dip and it is generally unbalanced. ⁹⁷¹ Asciano also said that there may be circumstances where an access holder has no scheduled paths but continues to operate ad hoc trains and that the termination provisions should be clarified to not terminate in these circumstances. ⁹⁷²		in chapter 8). We have also extended the "Relinquishment Date" to nine months. This is because the negotiation period in relation to a new access agreement is nine months and the two periods should be consistent. We agree with Asciano's statements in relation to no scheduled train paths. We have also included more reasonableness requirements on QR in relation to transfers. (for further reasoning see Summaries 7.2 & 7.3 & associated reasoning as well as Draft Decisions 7.4 & 7.7)
Amendments to clause 22 (Assignment) (Draft Decisions 7.3, 7.4 & 7.7).	Pursuant to Draft Decisions 7.3, 7.4 and 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators as well as necessary amendments to	New Hope generally supports the QCA amendments. However, New Hope does not agree with the proposition that a party may assign part of its rights and obligations under the agreement as this may have the effect of not preserving the risk profile under the SAA (cl.	QR does not comment specifically on the QCA's proposed amendments.	The QCA's amendments are consistent with the Draft Decision. The QCA considers that there are sufficient safeguards in relation to assignments to provide that an assignee will be able to discharge the obligations of QR therefore we do not accept that a partial

⁹⁷⁰ Glencore, sub. 25: 4 and sub, 30: 3.
 ⁹⁷¹ Asciano, sub. 28: 25–26.

972 Asciano, sub. 28: 26.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
	provide for a tripartite structure.	22.1 & 22.2 New Hope SAA). Asciano said that an identical approach should apply to both Queensland Rail and the access holder in that parties should be released from their obligations upon assignment. Also, Asciano said that an operator should be allowed to assign its rights. ⁹⁷⁴		 assignment of rights would increase a party's risk. Any incoming assignee also must agree to be bound by the undertaking and the SAA. We do not agree with Asciano that the release and discharge provisions should be symmetrical as the assigned obligations provision relates to assignment by an access holder to a related party In these circumstances it is appropriate that the access holder remains liable for the assigned obligations.
				Also, we do not agree that an operator should have the right to assign its rights. This is because, the access holder holds the access rights and should retain the discretion to transfer or nominate.
				Further, if the Access Holder and the Operator are the same person under the SAA then the Access Holder's rights can be assigned but any new Operator will need to be renominated under the assigned agreement.
				We have also clarified that an access holder may assign if it is not in material breach and that QR must act reasonably. (for further reasoning see

974 Asciano, sub.28: 26.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				Summaries 7.2 & 7.3 & associated analysis as well as Draft Decisions 7.3 7.4 & 7.7)
Amendments to clause 23 (Representations and Warranties) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments which are required to provide a more balanced risk position between QR and access seekers, access holders & operators. This includes removing the operator's warranty as to the standard of the network and making explicit that if an operator does inspect the network, the inspection does not necessarily restrict the operator's right to claim against QR.	New Hope supports the QCA's amendments but proposes minor amendments to accommodate multiple operators. ⁹⁷⁵	QR submits that if an operator inspects the network prior to operation it should accept some responsibility for satisfying itself as to the standard of the network. ⁹⁷⁶	The QCA's required amendments are largely consistent with the Draft Decision. However, we have also made the warranties between the Operator and Queensland Rail reciprocal. The QCA does not accept QR's statement implying that, by removing the operator's warranty, the operator has no responsibility for satisfying itself as to the standard of the network. The QCA considers that removal of the operator's warranty does not limit the responsibility of the operator in the manner described by QR. Instead deletion of this warranty removes the limitation placed on QR's responsibility for, amongst other things, matters which may not be reasonably foreseeable to an operator, despite the operator conducting an inspection of the network. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decision 7.7)

⁹⁷⁵ New Hope, sub. 32: 31.

⁹⁷⁶ QR, sub. 26, p. 102.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
Amendments to clause 25 (Notices) (Draft Decision 7.7).	Pursuant to Draft Decisions 7.3, 7.4 and 7.7, the QCA identified amendments to this clause which are required to provide for a more efficient operation of the network as well as necessary amendments to provide for a tripartite structure.	New Hope supports the QCA's amendments but proposes minor amendments in accordance with the tripartite structure. New Hope also said that it has been their experience that Queensland Rail has been willing to receive notices by email in other agreements. ⁹⁷⁷	QR states, amongst other things, that it does not accept service by email. ⁹⁷⁸	The QCA's required amendments are consistent with the Draft Decision. We consider that in modern times it is appropriate that parties be able to give and receive notices by email. This is efficient and prompt and thereby increases the efficiency of the system as a whole.
(General) (Draft Decision 7.3 & 7.4, the QCA i amendments are required t	Pursuant to Draft Decisions 7.3 & 7.4, the QCA identified amendments to this clause which are required to provide for a tripartite structure.	New Hope supports the QCA's amendments. New Hope also proposes an amendment to clause 27.1(b) which is intended to ensure that an access holder is not liable to pay any duty which QR incurs if it is incurred due to a default by QR. New Hope also said that all amendments should take effect between all parties. ⁹⁷⁹	QR submits that the clause is not legally effective in that an amendment cannot take effect only between some parties to the agreement. ⁹⁸⁰	The QCA's required amendments are largely consistent with the Draft Decision. However, the QCA considers that, if an access holder nominates multiple operators to utilise its access rights, each operator will be required to enter into a separate tripartite access agreement along with the access holder and QR. This should nullify QR's comments in relation to the variation clause.
				We also agree with New Hope that an access holder should not be liable for any duty which QR incurs as a result of QR's default.
				(for further reasoning see Summary 7.3 & associated analysis as well as Draft Decisions 7.3 &

⁹⁷⁷ New Hope, sub. 32: 31–33.

- ⁹⁷⁸ QR, sub. 26, p. 102.
- ⁹⁷⁹ New Hope, sub. 32: 32.

⁹⁸⁰ Queensland Rail, sub. 26, p. 102.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				7.4)
Amendments to clause 27.19 (Sublease) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA's amendments but also considers that, given the importance of land tenure, QR should be required to warrant that it holds, and will comply with the requisite tenure (cl. 27.18(e) New Hope SAA).	QR does not comment specifically on the QCA's proposed amendment.	The QCA's required amendments are consistent with the Draft Decision The QCA does not believe it is appropriate for QR to warrant that it holds, and will comply, with its various tenure obligations. But, if tenure is lost, an access holder's obligations to pay access charges should be reduced commensurate with that loss.
				(for further reasoning see Summary 7.3 & associated analysis as well as Draft Decisions 7.7)
Amendments to definition of "Acceptable Credit Rating" (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide for more efficient operation of the network and are more balanced as between QR and access seekers.	New Hope said that it welcomes the additional clarity and would accept a credit rating of "A" as previously proposed by Queensland Rail.	QR states that a BBB- S&P rating is not a suitable minimum and does not meet QR's board approval policies. ⁹⁸¹	We consider that a credit rating of "A" would strike an appropriate balance, as noted by New Hope this has also previously been acceptable to Queensland Rail.
Amendments to definition of "Alternative Schedule Time" (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA's amendments.	QR submits that the definition does not balance the interest of all parties but rather places an inappropriate burden on QR. ⁹⁸²	The QCA's amendments are consistent with the Draft Decision. See also above in relation to amendments to cl. 8.6 of the 2015 SAA. (for further reasoning see Summary 7.3 & associated analysis
Amendments to definition of	Pursuant to Draft Decision 7.7,	New Hope said that the QCA's	QR submits that the QCA's	as well as Draft Decision 7.7) In accordance with the maxim that

⁹⁸¹ QR, sub. 26, p. 102.

⁹⁸² QR, sub. 26, p. 102.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
"Force Majeure Event" (Draft Decision 7.7).	the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	amendment is not inconsistent as an FME is not the same as an operator failing to obtain or maintain rights to access private infrastructure. ⁹⁸³	proposed amendments allocate the risk to QR for matters which are solely within the control of the operator. ⁹⁸⁴	the party best able to manage a risk should bear that risk, we agree that access holders should bear the risk of FME's in relation to that access holder's private infrastructure or mine etc. However, we also agree with New Hope that the amendment is not inconsistent with an operator/access holder's obligation to obtain and maintain right to private infrastructure. We have therefore maintained this amendment but amended the definition of "Queensland Rail Cause" (see below). Further, we note that the clause 20.2(b) only operates in relation to the "Network" and not private infrastructure.
Amendments to definition of "Maintenance Work" (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a clarified and more balanced risk position between QR and access seekers, access holders & operators.	Stakeholders support the QCA's amendments.	QR submits that the QCA's amendments improperly impose an obligation on QR to fund Extensions. ⁹⁸⁵	The QCA's amendments are largely consistent with the Draft Decision. However, see above in relation to possible overlap with the definition of "Extension" under the QCA Act. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decisions 7.7)
Amendment to "Net Financial	Pursuant to Draft Decision 7.7,	New Hope supports the QCA's	QR does not specifically comment	We agree with New Hope that

⁹⁸³ New Hope, sub. 32: 32. ⁹⁸⁴ QR, sub. 26, p. 102.

985 QR, sub. 26, p. 102.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
Effect" (Draft Decision 7.7).	the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators. For example, if QR is able to [increase the tariff] due to an adverse financial effect, it should also be required to lower the tariff for a beneficial financial effect.	amendments. However, New Hope also proposes that any change in government funding should be relevant to particular commodities.	on the QCA's proposed amendments.	changes due to changes in government funding should be relevant to particular commodities.
Amendments to definition of "Operational Constraint" (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA's amendments but proposes deleting the term "Urgent Possessions". New Hope also said that having to act reasonably is not vague or unclear. ⁹⁸⁶	QR submits that the QCA's proposed amendments may have the effect of limiting QR's ability to impose an operational constraint, including for safety reasons. QR further submits that the amendments are an unexplained deviation from the 2010 regulatory precedent. ⁹⁸⁷	The QCA's amendments are consistent with the October 2015 Draft Decision. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decisions 7.7)
Amendment to definition of "Queensland Rail Cause" (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA amendments. ⁹⁸⁸	QR is concerned that by inserting "Force Majeure Event" into the definition allocates the risk of a force majeure event to QR when such events are outside of QR's control. QR considers that this change is a deviation from the earlier regulatory precedents. ⁹⁸⁹	The QCA's required amendments are largely consistent with the Draft Decision. However, see Chapter 8 in relation to increased incentives for QR to repair damaged track and apply for a review of the reference tariff to recover lost access charges. Also, in accordance with our stated policy on risk sharing, we have

⁹⁸⁶ New Hope, sub. 32: 32.

⁹⁸⁷ QR, sub. 26, p. 103.

⁹⁸⁸ New Hope, sub. 32: 32.

⁹⁸⁹ QR, sub. 26, p. 103.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
				amended this definition to clarify that an FME only operates in this definition to the extent that the relevant FME affects the Network or QR's ability to provide the contracted access. (See Chapter 8 for further discussion and Decision)
Amendment to definition of "Third Party Works" (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA amendments.	See above for QR's submissions in relation to "Third Party Works".	The QCA's required amendments are largely consistent with the Draft Decision. See also above in relation to the QCA's response to QR's concerns. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decisions 7.7)
Amendments to clause 28.2 (Construction) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope agrees with Queensland Rail's position.990	QR considers that the QCA's proposed change is not consistent with the QCA Act and cannot be overridden in the SAA. ⁹⁹¹	We accept Stakeholder statements and have reinserted the deleted clause.
Amendments to Schedule 1 (Reference Schedule), Item 9 (Security Amount) (Draft Decision 7.7).	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders &	New Hope supports the QCA amendments and notes that, in relation to Queensland Rail's argument that the amount is insufficient, the agreement provides for replenishment. ⁹⁹²	QR submits that the security amount in the reference schedule should be increased to account for the length of most access agreements. ⁹⁹³	The QCA's amendments are consistent with the Draft Decision. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decisions 7.7)

⁹⁹⁰ New Hope, sub. 32: 32. ⁹⁹¹ QR, sub. 26, p. 103.

⁹⁹² New Hope, sub. 32: 32.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
	operators.			
Amendments to Schedule 2 (Train Service Description – Stowage)	Pursuant to Draft Decision 7.7, the QCA identified amendments to this clause which are required to provide a more balanced risk position between QR and access seekers, access holders & operators.	New Hope supports the QCA amendments. New Hope also said that Queensland Rail should be required to provide some stowage/storage at no additional charge. ⁹⁹⁴	QR considers that the amendment is unnecessary given the provisions in relation to "Storage". ⁹⁹⁵	The QCA's amendments are consistent with the Draft Decision. (for further reasoning see Summary 7.3 & associated analysis as well as Draft Decisions 7.7)
		In addition to particular provisions which the QCA proposed be amended, New Hope also proposed amendments to cl. 8.11(b), definition of "Adjoining Network", definition of "Authority", definition of "Authority", definition of "Consequential Loss", definition of "Emergency Possession", definition of "Infrastructure", definition of "Network Interface Point", definition of "Original Train Service Take or Pay Threshold", definition of "Planned Possession", definition of "Through-Running Train" as well deleting the definition of "Train Schedule" [check whether used] and reducing take-or-pay charges	QR said that New Hope's proposed amendment to 8.11 is based on a misunderstanding of the term Network. ⁹⁹⁷ Further Queensland Rail said that, despite having an opportunity to comment on the DAU, a change in law is still outside of its control. ⁹⁹⁸ Further, Queensland Rail said that New Hope's proposed amendments to the definition of "Dangerous Goods" should not be inserted as the parties should be allowed to agree the treatment of goods to be carried. ⁹⁹⁹	We agree with New Hope that cl. 8.12 should be symmetrical. Noting Queensland Rail's concerns, the clause has been amended so that QR only needs to inform an operator in relation to damage which is relevant to that operator. We also agree with Asciano that Queensland Rail should not be allowed to unreasonably withhold its approval of accreditation (cl. 8.10). We do not accept New Hope's statements in relation to the definition of a "Change in Law" – as asserted by Queensland Rail, these changes are still outside of

⁹⁹³ QR, sub. 26: 104.

⁹⁹⁴ New Hope, sub. 32: 32.

⁹⁹⁵ QR, sub. 26: 104.

⁹⁹⁷ Queensland Rail, sub. 33: 64.

⁹⁹⁸ Queensland Rail, sub. 33: 65.

⁹⁹⁹ Queensland Rail, sub. 33: 65.

QCA October 2015 Draft Decision ("Draft Decision")	QCA Reasoning ⁸⁵⁷	Stakeholders' submissions	Queensland Rail's ("QR") Submissions	QCA Response and Decision
		from 100% of the relevant access charge in schedule 3 of NHC's amended SAA (New Hope SAA) Asciano said that cl. 8.9(a) should be amended to require that Queensland Rail should not unreasonably withhold its acceptance of certification. ⁹⁹⁶ Asciano suggested amendments to the definition of "Repeated Breach" to limit the term to breaches of material terms of the SAA.		Queensland Rail's control.We do agree with New Hope's proposed amendment to the definition of "Dangerous Goods".Queensland Rail's drafting could potentially allow for anything to be deemed a dangerous good and thereby increase costs inefficiently. There are objective codes which determine what is properly considered a dangerous good.We agree with Asciano that the definition of "Repeated Breach" should be limited by the inclusion of the words "material".Insignificant breaches of the SAA should not give Queensland Rail a right to terminate the agreement.

996 Asciano, sub. 28: 22.

APPENDIX D: CL. 1.4 AND SCHEDULE I AMENDMENTS

Other issues and proposed amendments to cl. 1.4 and 2.7.2

Clause	Issue	Response
1.4.1(a)	This clause summarises when cl. 1.4 applies. Queensland Rail said that the drafting was not fully consistent with cl. 2.7.2(d). Queensland Rail said that it potentially exposed Queensland Rail to bearing the indirect costs that it incurred.	Reference to an access seeker funding 'all costs directly incurred' has been deleted and Clause 1.4.1(a) has been simplified for consistency with cl. 2.7.2(d).
1.4.1(b)(ii)	This clause specifies that a Funding Agreement must have been executed for Queensland Rail to be obliged to complete the relevant stage of an extension, however, the signing of a funding agreement is a requirement of the Extension Conditions which are referred to in cl. 1.4.1(b)(i).	Clause deleted. A specific reference to executing Funding Agreement has been added to cl. 1.4.2(c)(vi) for clarity.
1.4.1(c)	This clause was subject to cl. 1.4.1(d)	cl. 1.4.1(d) has been deleted.
1.4.1(c)(i)(B)	Both New Hope and Queensland Rail noted that the wording potentially precluded Queensland Rail funding part of an Extension	Addressed. Words ' or part thereof' added.
1.4.1(d) Now deleted.	This clause. Queensland Rail said that it was beyond the QCA's power to impose an obligation on Queensland Rail to invest in the network only under the QCA Act. They said Queensland Rail's rights of ownership are absolute under the QCA Act as Queensland Rail is the accredited rail infrastructure manager in respect of the declared service.	Clause deleted. Clause 1.4 applies when there is an Access Funder and cl. 1.4.4 provides sufficient clarity on Queensland Rail's roles and responsibilities in this situation.
1.4.2(a)	Erroneous reference to cl. 2.7.2(c) instead of cl. 2.7.2(d)	Reference changed.
1.4.2(a)(i)	Queensland Rail stated that the drafting implied that Queensland Rail would be expected to supply information which would not be available until after the concept study before this study was completed.	The word 'available' has been added to make it clearer that the intention is to make Queensland Rail provide an Access Seeker with clarity over what is likely to be required to complete the study process.
1.4.2(a)(i)(C)	Queensland Rail objected to the phrase 'acting reasonably' in relation to safety requirements. They said safety requirements were not negotiable and therefore should not be subject to waiver.	The word 'reasonably' has been changed to 'appropriately'. The intent is that the safety standards imposed are consistent with the type of facility being built.
1.4.2(a)(i)(E) Now deleted.	Queensland Rail asked for clarification of term 'operational integrity'.	Clause deleted. The required information would be expected to derived from the application of cl. 1.4.2(a)(i)(D).
1.4.2(a)(ii)	Both Queensland Rail and New Hope noted that the wording of cl. 1.4 did not provide sufficient clarity that the study process did not have to be rigidly applied.	Introduction to cl. 1.4 amended by adding: 'by completing the required Extension Stage (or Stages).'

Clause	Issue	Response
1.4.2(a)(ii) (A)(B)(C)(D) Moved to definitions section	QR and New Hope asked for more clarity with respect to the Extension Stages.	Extension Stage is now a defined term in the definitions section.
1.4.2(a)(iii) and (iv)	Both New Hope and Queensland Rail asked for more flexibility with respect to the staging of the Extension (if both parties agreed)	Accepted in principle. Both clauses have been amended to clarify that only one extension stage and a time could be funded (rather than having to agree to fund, or not fund, them all).
1.4.2(a) - final summary	New Hope and Queensland Rail asked for more flexibility in staging the expansion by allowing for stages to be skipped if agreed by both parties.	Accepted in principle. 1.4.2(c) added. There is no requirement to complete all stages of the study process if both parties agree (acting reasonably) that an extension stage is unnecessary.'
1.4.2(c)(iii) (C) Now 1.4.2(d)(iii)(C)	This clause stated that the extension should not impact on the safety of any person maintaining, operating or using the network. New Hope suggested an edit to make it clear that positive safety benefits would be accepted.	Accepted in principle. The term 'adversely' added to make it clear that changes which had a positive impact on safety outcomes would be acceptable.
1.4.2(c)(iv) Now 1.4.2(d)(iv)	This clause stated that access agreements needed to executed for the additional capacity. Queensland Rail said the words "on terms and conditions consistent with this Undertaking unless otherwise agreed" are minor and inconsequential. Queensland Rail said the 2015 DAU is clear on the requirements for access agreements and the phrase is not necessary.	This clause has been amended to simply say that access negotiations need to be progressing as per Part 2 of the undertaking. The agreements themselves may not be executed until the extension is well progressed.
1.4.2(c)(vi) Now 1.4.2(d)(vii)		For clarity, a new cl. (1.4.2(d)(vi)) has been added with specific reference for the requirement to execute a funding agreement. The reference to funding in the original clause has been removed.
		In addition cl. 1.4.2(f) has been added to provide users with confidence that Queensland Rail will progress funding agreement negotiations.
1.4.2(e) Deleted.	Queensland Rail and New Hope noted that this clause was worded in a way that may not be applicable to the study stages of an extension	This clause has been deleted and the specific tasks listed that were listed are now detailed in clause 5 of Schedule I in relation to the relevant study stages.
1.4.2(f)	Queensland Rail said that this clause was unnecessary given the dispute cl. (1.4.7)	The clause has been deleted and amendments made to cl. 1.4.7(i) which specifies when a dispute can be referred directly to the QCA.
1.4.3(a)	Both New Hope and Queensland Rail asked for more flexibility with respect to the funding study process (if both parties agreed).	Amended to specifically allow for an individual project stage to be funded.
1.4.3(b)(iv) Now 1.4.3(b)(v)	Both New Hope and Queensland Rail asked for the contracting requirements to be allowed to vary with the requirements of the stage of the Extension being funded.	The clause has been amended to make it clear that not all tasks listed may be relevant to all stages of the extension process.

Clause	Issue	Response
1.4.3(b)(v)(B)	Queensland Rail noted that the reference to Schedule E was generic and did not specify which obligations were being referred to.	Specific reference to the prudency tests in Schedule E added.
1.4.3(b)(iv)	Included to clarify that this is an important obligation.	New clause inserted to required Queensland Rail to transfer the 'economic benefit' of an extension.
1.4.7	Queensland Rail raised issues relating to the dispute process.	Clause 1.4.7 and 6.1.4 have been amended to clarify.
1.4.8	Queensland Rail said that the investment funding provisions did not reference the Building Queensland Act.	New clause added.
2.7.2(b)(iii)	Both New Hope and Queensland Rail asked for the contracting requirements to be amended to clarify that an agreement to complete all project stages was not required in order to ask the Access Seeker to fund a particular project stage.	Words 'or Extension stage' added.
Definition of a Access Funder	Definition included reference to Access Seekers Nominee which Queensland Rail said went beyond the power of the Act.	Reference to an access seekers nominee has been deleted from the definition of an Access Funder.
Definition of Access Funder	Both New Hope and Queensland Rail noted that it was not sufficiently clear that an Access Funder for a single Extension could be more than one party.	Amended to allow for more than one party being an Access Funder for clarity.
Definition of Extension Costs	New Hope noted the definition potentially included costs which Queensland Rail might have incurred assessing an Access request that did not need an Extension.	Additional clause added so that Extension costs do not include 'any costs or expenses it would routinely incur when assessing an access
	Queensland Rail also raised concerns as to costs payable by it.	application'. Also, former (c)(iii) within the definition of 'Extension Costs' has been deleted.
Definition of Efficient Costs	Application of definition was too restrictive for the use of the term throughout the DAU.	Extended the definition of Extension Costs by clarifying that its use in the context of providing access is only an example of one application.
Asset replacement	Asset replacement was included in the definition of maintenance but is also include in the definition of Extension in the QCA Act.	Definition of maintenance in the SAA to be amended to remove the reference to replacement. See Appendix F.

Schedule I Amendments

Clause	Issue	Response
3(a)(i)(A)	The clause specified that the Access Funder was required to fund all the project stages and Queensland Rail and New Hope noted that more flexibility was required.	The requirement for the Access Funder to fund the relevant stage of the study process is set out in cl. 1.4.3 and the study processes (stages) are set out in the definition of Extension Stages.
		Clause deleted.
3(a)(i)(B)	This clause specified that the access funder should fund Queensland Rail's efficient costs.	Clause deleted.
3(a)(i)(C)	This clause specified that a Funding agreement must be executed in accordance with clause 3(b)(i).	Not clear which clause was being referred to as there was no cl. 3(b)(i). In any event the inclusion of cl. 1.4.2(c)(v) makes this clause redundant.
		Clause deleted.
3(a)(ii)	Required Queensland rail to construct. and operate and maintain an extension in accordance with the undertaking.	Issue dealt with in cl. 1.4.4. Clause deleted.
3(a)(iii)	Required that costs that Queensland Rail would have incurred without the extension could not	This clause has been moved to the definition of Extension costs.
	be included in the cost of an extension.	Clause deleted
3(a)(iv)	This clause specified that the access funder should not fund Queensland Rail's costs if they were judged to be inefficient.	The addition of cl. 1.4.3(b)(iii) now makes this unnecessary Clause deleted
3(a)(v)	Required that the concept of legitimate business interests be agreed by parties.	This process is likely to be complex and time consuming and could potentially delay the agreement / funding of projects prior to the completion any studies. Deleted due to other amendments relating to construction and costs.
3(b)	This clause stated that Queensland Rail had no obligation to bear costs in advance of a funding agreement being signed.	This is effectively a repeat of cl. 1.4.1(c)(ii) Deleted.
3(c)	This clause limited Queensland Rail's right to charge additional fees or charges to fund extension costs.	Moved to Section 4. Queensland Rail Rights and Responsibilities
3(d)(e) and (f)	This clause provided guidance on how the funding agreement terms should be amended if there were multiple and/or subsequent funders'.	A new section 'multiple and subsequent funders' has been added to the schedule.
		Moved to Section 7
4(a)(i)	This clause obliged QR to provide access right in accordance with an access agreement conditional only on an extension being commissioned.	The clause is now redundant since cl. 1.4.2(c)(iv) of the Undertaking specifies that as part of the extension conditions relevant access negotiations must be continuing in accordance Undertaking unless they are conditional only on an extension being commissioned.

Clause	Issue	Response
		Deleted.
4(a)(ii)	This clause obliged Queensland Rail to assist, construct and commission an extension consistent with the commercially balance allocation of risks.	The allocation of contract risks is detailed in cl. 8.1 while the obligation for Queensland Rail to undertake each extension stage is specified in cl. 1.4.2(c)(d) of the Undertaking.
4(a)(iii)	This clause requires Queensland Rail to pay the user funder the full economic benefit of an extension. It is dealt with in detail in the 'Full Economic Benefit Transfer' Clause.	Addressed in cl. 6. cl 1.4.3(b)(v) has also been added to the Undertaking specifying that a funding agreement must require the full economic benefit to be transferred to the access funder. Deleted.
Full Economic Benefit Transfer cl.5.		Clause moved to cl. 6
6(b) Now 3(b)	This clause required Access Funders to fund all of Queensland Rail's Extension costs. Comments were made by Queensland Rail and New Hope that there should be the option that Queensland Rail fund Extension Costs	'Unless otherwise agreed' added to the first sentence.
7(a)(iii) Now 4(a)(iii)	Comment made by Queensland Rail and New Hope that some clauses read as if all funding processes should be funded at the signing of the first funding agreement.	'as relevant to the stage of an Extension being funded' add to the end of Clause
5(c)	The clause stated that Queensland Rail should expeditiously undertake extension studies.	Edited to be specific to Concept, Pre- feasibility or Feasibility Studies
8(i) and (j) (deleted)	The clause required that an independent auditor be allowed to audit each funding agreement.	The role of the independent auditor is no longer defined in Schedule I. It is anticipated that, if required, this role will be specified within the terms of the relevant funding agreement.
9	This clause detailed the obligations of Queensland Rail and access funders when an extension was being constructed.	Clause combined into the Extension Studies Clause to improve clarity and duplication.
9(a) (deleted)	Requirement for Queensland Rail to expeditiously construct.	Deleted. This should be provided for in the relevant funding agreement.
9(b)	Requirement for Access Funders to be involved in major planning decisions.	Moved to cl. 5(d)(i)
10.13	Independent auditor clause deleted	See above.

APPENDIX E: WEST MORETON NETWORK COSTS ALLOCATION— PREVIOUS AND CURRENT ASSESSMENTS

2009 DAU

The issue of allocating West Moreton common network costs to different traffics first arose in our assessment of the 2009 DAU, submitted by then QR Network (now Aurizon Network).

2009 DAU proposal

The 2009 DAU was developed at a time when the West Moreton network's capacity was almost fully utilised—97 per cent of the capacity (109 of the 112 paths) was forecast to be used by coal and non–coal traffics combined (Figure 3 in Section 8.3 of this Decision).

The 2009 DAU proposed the following coal allocators for:

- regulatory asset base: the 2009 DAU distinguished between assets that existed before West Moreton coal traffic began in the mid–1990s and infrastructure subsequently built or replaced, and proposed allocating:
 - for capital spending since 1995 including forecast capital expenditure proposed during the 2009
 DAU period: 100 per cent to coal
 - for assets in place before 1995: 75.6 per cent to coal, to reflect the proportion of paths forecast for use by coal–carrying services (84 out of 111 forecast weekly paths¹⁰⁰⁰)
- maintenance cost: 92.7 per cent to coal, based on coal's share of forecast gtk.

QCA 2009 Draft Decision

Our 2009 Draft Decision did not approve the 2009 DAU's allocator for regulatory asset base including forecast capital expenditure, but accepted the allocator for maintenance cost.

Since the West Moreton network was then capacity constrained, our 2009 Draft Decision considered it was appropriate to determine coal reference tariff on the basis that all traffics were paying the same price and stated:

Put another way, it is not necessary for the non–coal traffics to pay the same tariffs as coal traffics. It is only necessary that the tariffs charged to the coal services not subsidise the non–coal services. So, if QR Network charges the other traffics lower tariffs, the Authority is entitled to treat those traffics as though they pay the same tariff as coal, when assessing whether QR Network is receiving sufficient revenue.¹⁰⁰¹

For determining a tariff on a capacity constrained network, our 2009 Draft Decision proposed the approach that 'each user's train service covers an equal proportion of the common cost of providing that asset base' and stated:

The share borne by coal will be based on the average proportion of available western system train paths forecast to be used by coal during the term of the 2009 DAU, or 75.6%.¹⁰⁰²,¹⁰⁰³

¹⁰⁰⁰ The path–based allocator was based on the average number of forecast paths during the undertaking period and included coal services operating through the Metropolitan network, which did not traverse the West Moreton network.

¹⁰⁰¹ QCA (December 2009a): 79.

¹⁰⁰² QCA (December 2009a): 80.

Our 2009 Draft Decision applied this allocator (75.6 per cent) to capital expenditure incurred on the West Moreton common network since 1995, including forecast capital expenditure proposed during the 2009 DAU period. In doing so, our 2009 Draft Decision rejected the 2009 DAU's proposal to allocate 100 per cent of such capital expenditure to coal. Our 2009 Draft Decision stated:

it is not apparent that all post–1995 capital expenditure has been for incremental coal infrastructure. Some of that capital expenditure has been to replace failing wooden bridges with culverts which is an investment that is common to all traffics. Other expenditure has been to support increased tonnages. Yet coal and grain trains on the western system have similar tonnages, lengths and axle loads, therefore any expenditure to handle additional tonnages is common to both coal and grain trains.¹⁰⁰⁴

While QR Network has proposed to include 100% of western system incremental capital expenditure in the regulatory asset base, the Authority considers that only 75.6% of this should be applied to calculating coal tariffs. The Authority considers this approach to be reasonable as incremental investment improves the standard of the track for both coal and non-coal services, which all benefit from the resulting increased reliability and lower maintenance requirement.¹⁰⁰⁵

For assets that were in place before 1995, our 2009 Draft Decision further reduced the 75.6 per cent allocator to reflect the impact of metropolitan peak hour passenger operations on West Moreton paths. In doing so, we accepted advice from QR Network that the 'metro blackout' reduced the available paths by about 20 per cent¹⁰⁰⁶ and determined the proportion of potential paths available to coal services to be 80 per cent of 75.6 per cent or 60.5 per cent.¹⁰⁰⁷

The pre–1995 asset base was treated separately so that Queensland Rail did not receive a return for sunk costs that related to paths that were unavailable to West Moreton network coal train services when they began in 1995 due to metropolitan operational constraints.¹⁰⁰⁸

Accordingly, our 2009 Draft Decision proposed allocating to coal traffic:

- regulatory asset base:
 - for capital spending since 1995 including forecast capital expenditure: 75.6 per cent
 - for assets in place before 1995: 60.5 per cent
- maintenance cost: 92.7 per cent, consistent with the 2009 DAU proposal.

In response, QR Network submitted the same West Moreton network reference tariff as proposed in our 2009 Draft Decision, which we approved.¹⁰⁰⁹

¹⁰⁰³ The asset base allocator (75.6 per cent) used for deriving the reference tariff reflected coal's proportion of 'total forecast paths' (not 'available paths') and included paths originating from Ebenezer, which do not traverse the West Moreton network. From Figure 1, we note that the 2009 DAU's weekly forecast paths (excluding Ebenezer paths) of 109 were slightly less than the total available paths of 112.

¹⁰⁰⁴ QCA (December 2009a): 83.

¹⁰⁰⁵ QCA (December 2009a): 87.

 ¹⁰⁰⁶ The 20 per cent metro blackout effect reflected the two three–hour periods to cater for the morning and afternoon weekday peaks of passenger traffic (QR Network, September 2008 and March 2009)
 ¹⁰⁰⁷ QCA (December 2009a): 80.

¹⁰⁰⁸ QCA (December 2009a): 84.

¹⁰⁰⁹ We proposed a West Moreton network tariff in our 2009 Draft Decision on QR Network's 2009 DAU (QCA, December 2009a: 69-94). QR Network submitted a tariff largely consistent with our 2009 Draft Decision in its 2010 DAU, in April 2010. We proposed to approve that tariff in our Draft Decision on pricing aspects of the 2010 DAU, on 2 June 2010 (QCA, June 2010a: 87-90). We gave final approval to the tariff in our 30 June 2010 Final Decision to approve an extension of the 2008 undertaking (QCA, June 2010b).

However, in making that proposal QR Network did not accept our 2009 Draft Decision methodology for assessing the tariff; in part the methodology for allocating the asset value and *forecast capital* expenditure between coal and non-coal services.¹⁰¹⁰

2013 DAU

Queensland Rail's 2013 DAU proposal

The 2013 DAU was developed at a time when the West Moreton network was capacity constrained—95 per cent of the capacity (106 of the 112 paths) was contracted for use by coal and non-coal traffics combined (Figure 3 in Section 8.3 of this Decision).¹⁰¹¹ In relation to the six uncontracted paths, Queensland Rail stated:

Government have not indicated a willingness to contract additional coal services and in relation to non-coal freight, above rail operators have not shown a willingness to contract additional services.¹⁰¹²

Queensland Rail said that since the system was capacity constrained, the 2009 Draft Decision principle for determining a ceiling price applied—that is, all traffics should be assessed as paying the same price regardless of whether they actually did.¹⁰¹³

The 2013 DAU proposed the following coal allocators for the regulatory asset base:

- for capital spending since 1995 including forecast capital expenditure proposed during the 2013 DAU period:
 - 100 per cent of capital spending on the common network that Queensland Rail required miners to underwrite or Queensland Rail determined was required solely to facilitate coal services.
 - 72.6 per cent of the remainder of capital spending, reflecting coal's proportion of total contracted paths (that is, 77 of 106 contracted paths).¹⁰¹⁴
- for assets in place before 1995: 61.7 per cent, to reflect that 15 per cent of West Moreton network paths were unavailable due to peak hour metropolitan blackout (that is, 85 per cent of 72.6 per cent).¹⁰¹⁵

In doing so, Queensland Rail did not accept the 2009 Draft Decision approach of a *pro rata* allocation of the entire common network capital expenditure between coal and non–coal services. Queensland Rail stated:

Permitting 100% of end-user funded assets to be included in reference tariff building blocks aids Queensland Rail's investment decisions as it would be unacceptable to proceed with an investment, even if it is end-user funded, in circumstances where only a partial return is included in reference tariff building blocks but a full return is rebateable to end-users.

¹⁰¹⁰ QR Network (February 2010: 99–118; April 2010: 104–115).

¹⁰¹¹ In the 2013 DAU, Queensland Rail had proposed using then contracted volumes as its forecast for the 2013 DAU period (Queensland Rail, 2013a: 15).

¹⁰¹² Queensland Rail (November 2013d): 5.

¹⁰¹³ Queensland Rail (June 2013a): 8.

¹⁰¹⁴ Queensland Rail (June 2013a): 9–10. Queensland Rail applied a similar approach to propose coal train path allocation percentage for the Macalister to Columboola section—that is, for post–1995 assets coal traffic pay 100 per cent of end–user funded and coal–specific spending and 50 per cent of the remainder of spending, reflecting coal's share of contracted paths (that is, 14 of 28 contracted paths).

¹⁰¹⁵ Queensland Rail applied a similar approach to propose coal train path allocation percentage for the Macalister to Columboola section for pre–1995 assets of 42.5 per cent (that is, 85 per cent metro adjustment of 50 per cent path allocation).

Similar to the argument above, Queensland Rail said that in circumstances where it is considering funding infrastructure that relates solely to facilitating coal, an investment is unlikely to proceed where only a partial return is included in reference tariff building blocks. This is because the shortfall in return is unlikely to be recouped from non-coal traffics in circumstances where the infrastructure is not required by them.¹⁰¹⁶

Queensland Rail accepted the 2009 Draft Decision approach to further reduce the coal train path allocator that applied to pre–1995 assets because of the impact of metropolitan peak hour period on the availability of West Moreton network paths. However, Queensland Rail did not accept the reduction percentage in the 2009 Draft Decision. Queensland Rail stated:

While Queensland Rail does have a Brisbane Metropolitan Region passenger weekday peak between the hours of 7:00AM to 9:30AM and 3:00PM to 6:30PM, a strict peak curfew for nonpassenger trains is not in place. This being said, Queensland Rail's network planners do have difficulty finding slots for non-passenger trains during these periods ... Queensland Rail proposes to apply a 15.0% reduction to the coal train path allocation percentage applied to pre-1995 assets rather than the 20.0% previously applied by the QCA.¹⁰¹⁷

The 2013 DAU retained the 2009 DAU's approach of allocating West Moreton network maintenance cost to coal traffics based on the share of forecast gtk. Additionally, the 2013 DAU proposed allocating operating costs to coal traffics based the proportion of total contracted paths.

The 2013 DAU proposed recovering the coal–allocated costs from the 77 contracted coal paths.

QCA 2014 Draft Decision

Our 2014 Draft Decision did not approve the 2013 DAU's allocators for regulatory asset base including forecast capital expenditure and operating cost, but accepted the allocator for maintenance cost.

Allocation of common network capital expenditure

On the allocation of capital expenditure, we reiterated our 2009 Draft Decision that the underlying investment, where it is on the common network, improves the track standard for all traffics that benefit from the resulting increased reliability and lower maintenance requirement. We also observed that most of the post–1995 capital expenditure had been on the shared network and said that although one business (coal) was growing and another business (non-coal) was not, that was not a reason for coal to pay for 100 per cent of the new infrastructure. Therefore, we considered it reasonable to apply a *pro rata* allocation to such capital expenditure.¹⁰¹⁸

Train path allocator

On the train path allocator for coal, our 2014 Draft Decision observed that the issue was whether coal's share should be based on 'contracted paths' or 'available paths'.

Our 2009 Draft Decision stated that coal's share should be based on the proportion of available paths.¹⁰¹⁹ However, our June 2014 consultation paper proposed to accept coal's share based on contracted (106) paths rather than available (112) paths, as we considered that contracted paths were verifiable and reflected clear evidence of customer demand.¹⁰²⁰

¹⁰¹⁶ Queensland Rail (June 2013a): 10.

¹⁰¹⁷ Queensland Rail (June 2013a): 9.

¹⁰¹⁸ QCA (October 2014): 145–146.

¹⁰¹⁹ As noted, the train path allocator that was used for deriving the reference tariff in the 2009 Draft Decision was based on coal's proportion of 'total forecast paths' (not 'available paths') and included paths originating from Ebenezer, which do not traverse the West Moreton network.

¹⁰²⁰ QCA (June 2014): 9–10.

Queensland Rail supported our consultation paper approach to use contracted paths. However, New Hope had concerns with that approach, which related to actual capacity exceeding the capacity that Queensland Rail was willing to contract. New Hope said:

The level of paths which is contracted is artificially constrained (below true system capacity) by Government (QR's shareholder). NHG has been seeking to contract additional train paths for the past three years and has been unable to do so because of this constraint. This, when combined with the use of contracted train paths in developing tariffs, has a number of implications, including:

• To the extent that capacity is withheld due to a Government requirement, which we understand is motivated by the potential future needs of passenger services, the cost of this uncontracted capacity should be allocated to 'non-coal' users when developing the notional coal asset base, rather than simply allocating the RAB between coal and non-coal on the basis of contracted paths.¹⁰²¹

Upon further consideration, in our 2014 Draft Decision we considered that there was merit in the train path allocator reflecting all available train paths, not just contracted paths. Accordingly, we proposed the train path allocator based on coal's share of available paths—that is, 77 of 112 available paths, or 68.8 per cent. We observed that our position was consistent with the 2009 Draft Decision. Our 2014 Draft Decision accepted that using total available paths will provide Queensland Rail with better incentives to increase the number of paths for coal train services, which will promote the efficient use of the network.¹⁰²²

Metro blackout sterilisation effect

On the effect of metropolitan passenger operations, our 2014 Draft Decision noted that stakeholders had divergent views on the impact of the blackout period.

In response to our June 2014 consultation paper which proposed a reduction of 22 per cent, Queensland Rail stated that it no longer supported reducing the pre–1995 asset base to reflect the metropolitan blackout arguing that it introduced asset stranding risk. Queensland Rail also said that if we applied a metro blackout, the sterilisation effect should be lower (12.1 per cent) than its original submission of 15 per cent.¹⁰²³ Conversely, other stakeholders (New Hope, Bentley and Aurizon) said the 22 per cent sterilisation effect in our consultation paper underestimated the actual impact of metropolitan blackout on West Moreton capacity.¹⁰²⁴

Our 2014 Draft Decision retained the position from our 2009 Draft Decision and stated that it was appropriate to apply the metropolitan blackout to the pre–1995 assets so that Queensland Rail did not get a return for capacity that was not available to coal trains.¹⁰²⁵ We also retained the 22 per cent sterilisation effect from our consultation paper, keeping open the option to revisit it as part of our Final Decision, subject to Queensland Rail providing compelling new information.¹⁰²⁶

Maintenance and operating cost allocation

Our 2014 Draft Decision accepted Queensland Rail's proposed gtk–based allocator for allocating total maintenance costs to coal traffic.¹⁰²⁷

¹⁰²¹ New Hope (July 2014): 3

¹⁰²² QCA (October 2014): 146.

¹⁰²³ Queensland Rail (July 2014): 13–19.

¹⁰²⁴ QCA (October 2014): 147.

¹⁰²⁵ QCA (October 2014): 148.

¹⁰²⁶ QCA (October 2014): 148.

¹⁰²⁷ QCA (October 2014): 122–123.

Our 2014 Draft Decision also accepted Queensland Rail's approach of using a train path–based allocator for operating costs. However, we used a train path allocator based on coal's share of total available paths, not total contracted paths.¹⁰²⁸

In summary, our 2014 Draft Decision proposed allocating to coal traffic:

- regulatory asset base:
 - for assets added since 1995 including forecast capital expenditure: 68.8 per cent, reflecting coal's share of total available paths
 - for assets in place before 1995: 53.6 per cent, reflecting a sterilisation effect of 22 per cent (that is, 78 per cent of 68.8 per cent)
- maintenance cost: 91 per cent, based on the share of forecast gtk consistent with Queensland Rail's proposal
- operating cost: 68.8 per cent, reflecting coal's share of total available paths.

However, the matter of the allocation of common network costs to coal traffics was not settled because Queensland Rail withdrew the 2013 DAU in December 2014. Subsequently, Queensland Rail submitted a DAU in May 2015.

2015 DAU

Queensland Rail's 2015 DAU proposal

The 2015 DAU has been developed in light of a material fall in demand for below-rail services on the West Moreton network—about three—fifths of the available capacity (66 of the 112 paths) is forecast to be used by coal and non–coal traffics combined resulting in about 40 per cent of spare capacity (Figure 3 in Section 8.3 of this Decision). This was due to reductions of 18 per cent and 90 per cent for coal and non–coal train services respectively, compared to the contracted levels used in the 2013 DAU (see Section 8.10 of this Decision for our consideration of Queensland Rail's 2015 DAU volume forecasts).

Queensland Rail said that the changed volume scenario necessitated a different approach to allocating common network costs to coal traffics for pricing purposes.

Train path allocator

In the 2015 DAU, Queensland Rail did not accept the train path allocation in the 2014 Draft Decision based on coal's share of total 'available' paths. Queensland Rail said that the QCA's allocation approach was unreasonable and inconsistent with standard regulatory precedent and stated:

With only 53 of the 112 available weekly paths currently contracted, and 62.8 expected to be utilised this regulatory period, such an approach will prevent Queensland Rail from recovering its efficient costs including a return. This is inconsistent with the pricing principles in the QCA Act.¹⁰²⁹

Queensland Rail said that in the current environment its primary concern was revenue certainty and stated:

Queensland Rail would clearly be better off if it could increase the number of paths contracted to coal (subject to the Government imposed cap on paths for coal services, discussed below), as it provides it with increased revenue certainty for services that have a higher capacity to pay than non-coal services.¹⁰³⁰

¹⁰²⁸ QCA 2014d: 126.

¹⁰²⁹ Queensland Rail, sub. 2: 7.

¹⁰³⁰ Queensland Rail, sub. 2: 46.

For the 2015 DAU, Queensland Rail considered that the most reasonable way of applying the train path allocator was based on coal's share of forecast train paths, which produced an allocation of 95.4 per cent (that is, 63 of 66 paths).¹⁰³¹

Queensland Rail considered its 2015 DAU allocation approach was more consistent with the requirements of the QCA Act and regulatory treatment more generally, noting that the pricing principles in all Australian access regimes were premised on the entitlement of the infrastructure provider to recover its efficient costs.¹⁰³²

Allocation of opening asset base

Queensland Rail said that while its proposed train path allocator reflected coal's share of forecast paths, it also acknowledged that there were a number of factors that restricted its ability to contract the full amount of the capacity created by the existing assets.¹⁰³³ Queensland Rail stated that the two main constraints were:

- preserved freight and passenger train paths from Rosewood to Toowoomba, which was 13 paths for freight¹⁰³⁴ and two for passenger services; and
- a constraint of 87 coal paths per week through Metropolitan Network specified by Queensland Rail's Responsible Ministers.¹⁰³⁵

Queensland Rail considered it was reasonable to cap the allocation of the initial asset base (or opening asset value as at 1 July 2015) to coal traffics 'to reflect Government constraints on contracting capacity to coal'¹⁰³⁶ and stated:

In particular, the binding constraint is the maximum 87 coal paths per week, limiting the proportion of the capacity of the West Moreton Network that can potentially be contracted to coal to 87 out of 112 available paths, or 77.7%. This approach is consistent with the overarching objectives established by the QCA (as set out at the beginning of this section¹⁰³⁷) of balancing Queensland Rail's right to recover its costs from users with mining customers' right to not be required to pay for capacity that they are not permitted to use. Given the asset value has been established on the basis of this constraint, it should be reviewed if there is any material change to that constraint in the future.¹⁰³⁸

¹⁰³¹ The 95.4 per cent allocator applied to the Rosewood to Jondaryan section. The corresponding allocator proposed by Queensland Rail for the Jondaryan to Columboola section was 87.5 per cent (Queensland Rail, sub. 2: 48).

¹⁰³² Queensland Rail, sub. 2: 48.

¹⁰³³ Queensland Rail, sub. 2: 48.

¹⁰³⁴ Queensland Rail's 17 July 2015 response to the QCA's request for additional information noted that the reference to '13 paths for freight' was incorrect and that it should be '14 paths for freight'.

¹⁰³⁵ Queensland Rail's 17 July 2015 response to the QCA's request for additional information clarified that the restriction was advised by the Department of Transport and Main Roads.

¹⁰³⁶ Queensland Rail, sub. 2: 45, 48.

¹⁰³⁷ Queensland Rail referenced the objectives set out in the QCA's June 2014 consultation paper for determining an appropriate allocation methodology—that is, a) Queensland Rail's reasonable desire to recover the investment it has made in the network to support the growth of coal traffic; and b) coal miners' interest in not paying for assets they are unable to use, whether that be because those paths are contracted to non-coal traffics or where a significant portion of capacity cannot be contracted because of restrictions that provide priority to passenger services on the metropolitan system.

¹⁰³⁸ Queensland Rail, sub. 2: 48.

Accordingly, for the post–1995 capital spending in the initial asset base, Queensland Rail proposed an allocator 'based on the relative share of forecast train paths, with coal's allocation capped at 77.7 per cent'.¹⁰³⁹

In doing so, Queensland Rail said that it accepted the 2014 Draft Decision approach of a *pro rata* allocation of the common network capital spending between coal and non–coal services. Queensland Rail said that its proposed opening asset base in the 2015 DAU included all capital expenditure on the common network, including projects that were triggered by freight services (referred to as transport service contract (TSC) capital), which its 2013 DAU had allocated 100 per cent to non–coal traffics.¹⁰⁴⁰

For the pre–1995 assets in the opening asset base, Queensland Rail said that consistent with previous practice it proposed reducing the allocator to reflect the impact of metropolitan passenger operations on the West Moreton network capacity and stated:

While Queensland Rail considers that this approach is not required from an economic theory perspective, it reflects a pragmatic way of addressing the concerns of customers around the impact of the passenger dominated Metropolitan Network on the available capacity of the West Moreton Network.¹⁰⁴¹

Queensland Rail proposed a capacity impact due to metropolitan passenger operations of 12.1 per cent. Queensland Rail said its adjustment percentage was based on a review of the scheduling across the Metropolitan and West Moreton networks that considered the effects of peak passenger periods and planned maintenance. Accordingly, for the pre–1995 assets, the 2015 DAU proposed an allocator of 68.3 per cent (that is, 87.9 per cent of 77.7 per cent).¹⁰⁴²

Allocation of forecast capital expenditure

Unlike the allocator for opening asset base, Queensland Rail proposed allocating forecast capital expenditure based on coal's share of forecast paths – that is, 95.4 per cent.¹⁰⁴³ Queensland Rail stated:

An infrastructure owner cannot expect to commit to what will become a sunk investment on the basis that only a portion of those costs can be allocated to existing users. Queensland Rail therefore considers that the allocation measure must be based on forecast expected usage, not total available paths ... Basing the measure on forecast paths is essential if Queensland Rail is to retain any incentive to undertake future investments in the shared network.¹⁰⁴⁴

Queensland Rail said that its forecast capital program was primarily aimed at 'replacing assets as required in order to maintain the integrity of the rail network' and stated:

... there is opportunity for coal volumes to recover in the future. Queensland Rail does not consider it appropriate to allow the network standards to deteriorate in response to a potentially short term reduction in volume, as the deteriorated state may inhibit future opportunities for traffic volumes to increase.¹⁰⁴⁵

¹⁰³⁹ Queensland Rail, sub. 2: 49.

¹⁰⁴⁰ Queensland Rail, sub. 2: 49. See Section 8.14 of this Decision for our consideration of the TSC capital expenditure.

¹⁰⁴¹ Queensland Rail, sub. 2: 49.

¹⁰⁴² Queensland Rail, sub. 2: 51.

¹⁰⁴³ The 95.4 per cent allocator applied to the Rosewood to Jondaryan section. The corresponding allocator proposed by Queensland Rail for the Jondaryan to Columboola section was 87.5 per cent (Queensland Rail, sub. 2: 55-56).

¹⁰⁴⁴ Queensland Rail, sub. 2: 55–56.

¹⁰⁴⁵ Queensland Rail, sub. 2: 38.

Coal has 87 contracted paths through the SEQ network each week. Any growth in tonnages must take advantage of these existing paths as there are no new paths available.¹⁰⁴⁶

Queensland Rail's consultant, PwC stated:

Without incurring any additional capital or maintenance expenditure, the Rosewood to Jondaryan (R2J) part of the network could cater for 15.7 gross million tonnes (GMT) (up from 11.5 GMT); while the Jondaryan to Columboola part of the network could cater for 3.6 GMT (up from 3 GMT). Queensland Rail's analysis suggests that this additional volume and decline in coal path allocation (95 per cent to 84 per cent) would result in a ceiling price of \$27.91/'000 gtk, or a 20 per cent decline.1047

Allocation of maintenance and operating cost

On allocating the 2015 DAU maintenance costs, Queensland Rail stated that not all maintenance activities were volume-dependent. Nevertheless, Queensland Rail considered that the maintenance cost allocator should be based on the expected level of activity – that is, based on relative forecast volumes. Accordingly, Queensland Rail proposed allocating 98.2 per cent of the maintenance costs to coal services, based on coal's share of forecast gtk. Queensland Rail said its proposed allocation approach was consistent with its 2013 DAU proposal.¹⁰⁴⁸

On allocating the 2015 DAU operating cost, Queensland Rail did not accept the 2014 Draft Decision approach of a train path allocator based on total available paths, stating that it will not be able to fully recover its efficient costs. Queensland Rail proposed allocating operating costs based on coal's share of forecast paths – that is, 95.4 per cent and stated:

By using Queensland Rail's general train path allocator [share of forecast train paths], Queensland Rail will have opportunity to fully recover the assessed efficient operating costs from the services that are expected to use the infrastructure.¹⁰⁴⁹

In summary, Queensland Rail proposed the following coal allocators in the 2015 DAU:

- Opening asset base:
 - for capital spending since 1995: 77.7 per cent, reflecting the cap based on maximum paths available for contracting by coal services as a proportion of total available paths
 - for assets in place before 1995: 68.3 per cent, reflecting 12.1 per cent sterilisation effect due to metropolitan passenger operations
- forecast capital expenditure and operating cost: 95.4 per cent, reflecting coal's share of total forecast paths
- maintenance cost: 98.2 per cent, reflecting coal's share of forecast gtk.

Queensland Rail said its allocation approach provided it with a greater opportunity to recover its efficient costs, even though it proposed to set reference tariffs below its ceiling price.¹⁰⁵⁰

¹⁰⁴⁶ Queensland Rail, sub. 2 (Appendix 6): 7.

¹⁰⁴⁷ Queensland Rail, sub. 2 (Appendix 1): 13.

¹⁰⁴⁸ The 98.2 per cent allocator applied to the Rosewood to Jondaryan section. The corresponding allocator proposed by Queensland Rail for the Jondaryan to Columboola section was 96.8 per cent (Queensland Rail, sub. 2: 53).

¹⁰⁴⁹ Queensland Rail, sub. 2: 54.

¹⁰⁵⁰ Queensland Rail, sub. 2: 4, 7.

Stakeholders' comments on 2015 DAU proposal

New Hope said that Queensland Rail's proposed allocation approach to certain common network costs resulted in coal services being required to compensate Queensland Rail for reductions in demand for non-coal traffics.¹⁰⁵¹

New Hope also said that Queensland Rail's inability to recover non-coal's share of costs, due to declining demand for non-coal services, was Queensland Rail's commercial risk and that Queensland Rail's proposed allocation approach had the effect of transferring Queensland Rail's risk to coal producers.¹⁰⁵²

New Hope proposed that common network costs should be allocated on the basis of:

- fixed costs based on the higher of coal's forecast or contracted paths as a share of system capacity
- costs that vary with usage (variable costs) based on coal's share of forecast usage
- recognition provided for restrictions on the number of coal services able to be contracted and metropolitan capacity constraints.¹⁰⁵³

In addition, New Hope said that recognition of operational restrictions on coal traffics operating through the Metropolitan network should also be incorporated within any proposed allocation of costs to coal traffics.¹⁰⁵⁴

2015 Draft Decision

Our 2015 Draft Decision supported the allocation of common network costs amongst the different classes of users on the West Moreton network. However, we did not approve Queensland Rail's proposed allocation approach in the 2015 DAU.

Our 2015 Draft Decision observed that previous considerations of West Moreton network pricing were undertaken in the context of available capacity being potentially insufficient to satisfy all requests for access rights. In contrast, the 2015 DAU had been developed in a fundamentally different market demand context.

We considered that the material reduction in demand for West Moreton network train paths and the constraints on the number of coal services that are able to be contracted to operate through the Metropolitan network (a maximum of 87 paths per week) necessitated an efficient approach of allocating common network costs.

We considered that most of Queensland Rail's below-rail infrastructure costs were fixed.¹⁰⁵⁵ Therefore, we proposed categorising common network costs as fixed and variable, and allocating:

- fixed costs (comprising common network assets, and fixed maintenance and operating costs) based on the relative proportion of the network capacity available to coal services to contract
- variable costs (comprising variable maintenance and operating costs) based on the relative volume forecast for all train services, as variable costs are directly affected by volumes.

In doing so, our 2015 Draft Decision accepted Queensland Rail's allocation approach to reflect the contracting restriction for coal services. But we applied that approach to all of West Moreton fixed common network costs, which included in addition to the opening asset base, forecast capital

¹⁰⁵¹ New Hope, sub. 9: 24–25.

¹⁰⁵² New Hope, sub. 9: 27.

¹⁰⁵³ New Hope, sub. 9: 25-26.

¹⁰⁵⁴ New Hope, sub. 9: 26.

¹⁰⁵⁵ For example, our 2015 Draft Decision considered that about 68 per cent of maintenance costs and about 82 per cent of operating expenditure related to activities that did not vary with usage.

expenditure and fixed maintenance and operating costs. We considered that coal traffics should only pay for the fixed common network costs of the paths they are able to contract to use.

We considered that Queensland Rail's proposed allocation approach would result in fixed common network costs associated with providing access to non-coal services being recovered from coal traffics that cannot access this capacity. We also did not propose to accept New Hope's approach to allocating fixed common network costs based on coal's forecast paths as a share of system capacity, as it would not give any consideration to the costs of spare capacity available for coal services to contract.

Therefore, we determined allocating fixed common network costs based on the relative train paths available for contracting by coal services—that is, based on 77 weekly paths (not 87 paths) to account for ten paths currently contracted to operate through the Metropolitan network which do not traverse the West Moreton network.

Our 2015 Draft Decision said that there was also a prima facie case that we consider the 2015 DAU provisions for reducing the value of assets contained in a regulatory asset base where demand for access has deteriorated to such an extent that regulated prices based on an unoptimised asset value would result in a further decline in demand for access. While optimisation could be applied with respect to coal and non-coal services' share of the common network, we were minded to approve reference tariffs for coal-carrying train services as part of our consideration of the 2015 DAU.

We also assessed Queensland Rail's proposed 12.1 per cent sterilisation effect due to metropolitan passenger operations. We were advised on this matter by B&H, which considered the new information from Queensland Rail, including that not all freight paths are affected by maintenance on the suburban lines in the Metropolitan network. B&H estimated that the sterilisation effect should be 17 per cent, which we proposed to apply for allocating the pre–1995 assets to coal traffics.

Our 2015 Draft Decision allocated the variable maintenance and operating costs based on coal's share of forecast volumes. We proposed that the coal–allocated costs should be fully recovered from the 63 forecast coal traffics.

We considered our approach provided Queensland Rail with a reasonable return on its investment in the West Moreton network for coal-carrying train services, and recovered the efficient costs of operating infrastructure assets to provide access to a regulated service, consistent with sections 138(2)(b) and 168A(a) of the QCA Act. Our approach also meant users did not pay for network capacity they were unable to contract, whether because those paths were contracted to non-coal traffics or because capacity could not be contracted due to government-imposed contracting restrictions, consistent with section 138(2)(e) and (h) of the QCA Act.

In summary, we proposed the following coal allocators for the 2015 DAU:

- Opening asset base:
 - for capital spending since 1995: 68.8 per cent, reflecting the cap based on maximum actual paths available for contracting by coal services as a proportion of available paths
 - for assets in place before 1995: 57.1 per cent, reflecting 17 per cent sterilisation effect due to metropolitan passenger operations (that is, 83 per cent of 77.7 per cent)
- forecast capital expenditure: 68.8 per cent
- maintenance and operating costs: 68.8 per cent of fixed costs and 98.2 per cent of variable costs.

Table 1 below summarises the coal–related common cost allocators that have been considered as part of the QCA's assessments of the 2009, 2013 and 2015 DAUs.

Table 1 West Moreton network cost allocation: previous and current assessments¹

DAUs	Opening asset base		Forecast capital	Maintenance	Operating costs
	Pre–1995 assets	Post–1995 assets	expenditure	costs	
2009 proposal	75.6% (84/111 paths)	100%	100%	92.7% (coal share of forecast gtk)	na²
2009 Draft Decision	60.5% (to reflect 20% adjustment for metro passenger operations)	75.6%	75.6%	92.7%	na
2013 proposal	61.7% (to reflect 15% adjustment for metro passenger operations)	88.3% ³ (user funded capex: 100% other capex: 72.6% (77/106; contracted coal paths/ total contracted paths)	94.2% ³ (coal–specific capex: 100% other capex: 72.6%)	91.0% (coal share of forecast gtk)	72.6%
2014 Draft Decision	53.6% (to reflect 22% adjustment for metro passenger operations)	68.8% (77/112 paths; contracted coal paths/ total available paths)	68.8%	91.0%	68.8%
2015 proposal	68.3% (to reflect 12.1% adjustment for metro passenger operations)	77.7% (87/112 paths; maximum potential paths available for contracting by coal/total available paths)	95.4% (63/66 paths; forecast coal paths/total forecast paths)	98.2% (coal share of forecast gtk)	95.4%
2015 Draft Decision	57.1% (to reflect 17% adjustment for metro passenger operations)	68.8% (77/112 paths; maximum actual paths available for contracting by coal/total available paths)	68.8%	78.3% ³ (fixed: 68.8% variable: coal share of forecast gtk)	74.2% ³ (fixed: 68.8% variable: coal share of forecast gtk)

1. Allocation percentages reported for 2009 and 2013 DAUs relate to the Rosewood to Macalister section and for 2015 DAU relate to the Rosewood to Jondaryan section.

2. Operating cost allocation in the 2009 DAU was based on QR Limited's cost structures and allocators which are not applicable to Queensland Rail, following the separation of QR Limited into Aurizon and Queensland Rail. Therefore, those allocators are not discussed here.

3. value weighted average percentage.

Expected revenue and efficient cost analysis under the three allocation approaches

We analysed whether expected revenue across all traffics under each of the three allocation approaches considered in this Decision (i.e. Queensland Rail's approach, the miners' approach and the QCA's approach) would be enough to meet the efficient costs of providing access for all traffics on the West Moreton network (i.e. costs that reflect the overall capacity of the network to provide all those services).

For this analysis, we considered the efficient costs for the five-year period July 2015 to June 2020, where the costs comprised:

- efficient capital charges, based on the regulatory return on and of capital (see Section 8.16 of this Decision)
- efficient maintenance costs (see Section 8.11 of this Decision)
- efficient operating costs (see Section 8.12 of this Decision).¹⁰⁵⁶

We first calculated these costs for the whole of the West Moreton network to identify the efficient total costs of providing access for all traffics on the West Moreton network.¹⁰⁵⁷ The present value¹⁰⁵⁸ of the efficient total costs for whole of the West Moreton network is estimated at \$237.7 million.

Next, we calculated the costs assigned to coal services under each of the three allocation approaches. The present value of coal-allocated costs under each approach is estimated as:

- under Queensland Rail's approach—\$211.1 million;¹⁰⁵⁹
- under the miners' approach—\$152.2 million;¹⁰⁶⁰ and
- under the QCA's approach—\$178.1 million.¹⁰⁶¹

Since the coal-allocated costs are to be recovered from forecast coal services, they also represent the present value of expected revenue from providing access for coal service on the West Moreton network.

To each of these expected coal revenue amounts, we added around \$6.8 million of the present value of expected revenue from non-coal service over the period 2015–2020, estimated on the basis of

¹⁰⁵⁶ Operating costs include an allowance for tax and working capital that were derived from the building blocks tariff model for the five-year period July 2015 to June 2020.

¹⁰⁵⁷ In the building blocks tariff model, efficient costs for the whole of the West Moreton network and return on investments were calculated by using a coal allocator of 100 per cent.

¹⁰⁵⁸ We used two WACC rates for calculating the present value—6.93 per cent for 2015–16 and 5.73 per cent for 2016–20 (see Sections 3.7 and 8.16 of this Decision).

¹⁰⁵⁹ We used coal allocators from Queensland Rail's 2015 DAU proposal, which included the fixed cost allocators (see Table 15 in this Decision) that were used for post-1995 asset base, forecast capital expenditure and operating expenditure, and included different allocators for pre-1995 assets and maintenance costs. Different allocators were used for the Rosewood to Jondaryan and the Jondaryan to Columboola sections (Queensland Rail, sub. 2: 44–56).

¹⁰⁶⁰ We used fixed cost allocators summarised in Table 15, allocated variable costs based on coal's share of forecast volumes, and used a different allocator for pre-1995 assets by reducing the opening asset base allocator in Table 15 by a 22 per cent Metro impact factor suggested by miners—see Section 8.4.1 of this Decision for our consideration of the Metro impact factor in allocating the pre-1995 assets.

¹⁰⁶¹ We used coal allocators recommended in this Decision that included the fixed cost allocators based on our Draft Decision approach summarised in Table 15 but updated to account for our estimate of West Moreton network capacity and our assessment of the number of paths available to coal services to contract in the West Moreton network—see discussion under the heading 'Change in circumstances' in Section 8.3.3 of this Decision. We allocated variable costs based on coal's share of forecast volumes, and used a different allocator for pre-1995 assets—see Section 8.4.1 of this Decision.

information submitted by Queensland Rail,¹⁰⁶² which produced an estimate of the expected access revenue across all services on the West Moreton network under each allocation approach.

The results of this analysis are summarised in Table 2 below, which shows that each of the three allocation approaches results in a shortfall in recovering the efficient total costs of providing access for all services on the West Moreton network.¹⁰⁶³ This analysis is based on Queensland Rail's forecast volumes which we have accepted (see Section 8.10 of this Decision), noting that those forecasts include one ad hoc non-coal train service. Nevertheless, as noted in Section 8.10 of this Decision, non-coal volumes are likely to be higher than Queensland Rail's forecast. If that were the case, the overall shortfall estimated in Table 2 will decrease and possibly become immaterial depending on the extent of the rise in non-coal service volumes. Effectively, the extent, and the existence, of the shortfall depends on non-coal volumes.

Allocation approaches	Efficient costs of providing access for all services	Expected coal service revenue	Expected non- coal service revenue	Estimated overall Queensland Rail's r environ	non-coal-market
	(\$ million)	(\$ million)	(\$ million)	(\$ million)	% of total costs
Queensland Rail's approach	\$237.7	\$211.1	\$6.8	(\$19.8)	(8%)
Miners' approach	\$237.7	\$152.2	\$6.8	(\$78.6)	(33%)
QCA's approach	\$237.7	\$178.1	\$6.8	(\$52.7)	(22%)

 Table 2: Efficient costs and expected revenue across all services on the West Moreton network under the three allocation approaches (present value over the five-year period 2015–2020)

¹⁰⁶² Queensland Rail expected actual revenue of around \$1.5 million from non-coal service in 2015–16 (Queensland Rail, sub. 33: 9–10), which would be from the three non-coal services forecast for 2015–16. Since forecast non-coal services remain at three services during each year of the 2015 DAU period, we expect the non-coal services revenue would be around \$1.5 million per year during the 2015 DAU period, subject to CPI escalation since all calculations in this analysis are in nominal dollars. Therefore, we escalated this noncoal revenue by the assumed inflation of 2.5 per cent and calculated the present value of the resulting noncoal revenue stream over the five-year period July 2015 to June 2020.

¹⁰⁶³ The outcome of an expected shortfall would remain unchanged if the three forecast non-coal services were expected to pay the same tariff as coal services. This could be assessed by escalating the coal service revenue amounts by the ratio 66/63 to calculate the equivalent revenue for total forecast (66) services and comparing the resulting amount with the efficient total costs of providing access for all services.
APPENDIX F: 2015 DAU MARK-UPS

[<mark>Attached separately</mark>]

APPENDIX G: 2015 SAA MARK-UPS

[Attached separately]

APPENDIX H: QUEENSLAND RAIL'S EXTENSION DAAUS

Queensland Competition Authority

File Ref: 832243 ver 2 26 June 2015

Ms Helen Gluer Chief Executive Officer Queensland Rail Ltd GPO Box 1429 Brisbane QLD 4001

Dear Ms Gluer

Final decision: Queensland Rail's draft amending access undertaking - extension of termination date

On 25 June 2015, the Queensland Competition Authority (QCA) made a final decision to refuse to approve Queensland Rail's draft amending access undertaking to extend the term of the 2008 access undertaking (the April 2015 Extension DAAU).

Context

Queensland Rail's 2008 access undertaking is due to expire on 30 June 2015.

On 9 April 2015, the QCA received from Queensland Rail, under section 142 of the *Queensland Competition Authority Act 1997* (QCA Act), a DAAU to extend its 2008 undertaking's termination date to the earlier of (i) 30 June 2016 and (ii) the date on which we approve a replacement access undertaking.

On 4 June 2015, the QCA made a draft decision to refuse to approve the April 2015 Extension DAAU and invited interested parties to make written submission on the draft decision by 12 June 2015.

Stakeholder comments and QCA Position

Five stakeholders commented on the DAAU. New Hope supported the draft decision while Asciano, Aurizon, Glencore and Queensland Rail did not. Stakeholders' comments can be grouped as below.

No lawful basis

Queensland Rail's submission says "the QCA cannot lawfully refuse to approve a DAAU on the basis that Queensland Rail has not agreed to... retrospectivity". It makes this submission having earlier said "the QCA's refusal is essentially for the reason that Queensland Rail has not given a "commitment" for the reference tariffs to be approved under the 2015 DAU to be made retrospective and backdated to 1 July 2013".

In making its decision on the April 2015 Extension DAAU, one of the factors considered by the QCA was the absence, in the April 2015 Extension DAAU, of an indication that Queensland Rail remained committed to the inclusion of an adjustment charge to recoup or refund any variations between transitional and new tariffs from 1 July 2013 (an Adjustment Charge).

Level 27, 145 Ann Street, Brisbane Q 4000 GPO Box 2257, Brisbane Q 4001 Tel (07) 3222 0555 www.qca.org.au Final decision: Queensland Rail's draft amending access undertaking - extension of termination date

This was relevant because, in the context where such an indication had previously been given in relation to earlier extensions of the 2008 undertaking which had been granted, the absence of such an indication in connection with the April 2015 Extension DAAU, if approved, would create uncertainty about the application of an Adjustment Charge; and this in turn would have a negative effect on the public interest and the promotion of the economically efficient operation of, use of and investment in, significant infrastructure (the Uncertainty Issue).

It was also relevant because the absence of such an indication meant that QCA was not satisfied that the tariff for the proposed extension period was appropriate having regard to the matters relevant under section 138(2) of the QCA Act (the Unestablished Tariff Issue).

Effect of transfer notice

Queensland Rail made a further legal submission related to the effect of the Transfer Notice. This submission was that the Transfer Notice somehow had the effect that the April 2015 Extension DAAU must be approved; that refusing to approve the extension would create uncertainty; and that the extension was "consistent with the legally binding and unequivocal provisions of the Transfer Notice".

QCA does not consider that the matters relating to the Transfer Notice raised by Queensland Rail make it appropriate to approve the April 2015 Extension DAAU.

Refusing to approve the April 2015 Extension DAAU is not inconsistent with the Transfer Notice and does not create any relevant uncertainty. Clause 5 of the Transfer Notice states that nothing in it is to be taken to preclude amendment of the access undertaking. That would include amendment of the term of the access undertaking (which has happened on a number of occasions since the Transfer Notice was issued).

Further, although (in places) the terms of the Transfer Notice appear to extend the term of the 2008 undertaking indefinitely, the QCA is not satisfied that the Transfer Notice could validly have such an effect.

Addressing 'retrospectivity' in 2015 DAU

Aurizon said the issue of 'retrospectivity' should be addressed through the consideration of the 2015 draft access undertaking (the 2015 DAU).

While the QCA expects that the issue of 'retrospectivity' (or, more accurately, the application of an Adjustment Charge) will be considered in the 2015 DAU, the QCA does not consider this to make it appropriate to approve the April 2015 Extension DAAU.

This is because simply approving the April 2015 Extension DAAU and considering the Adjustment Charge in the 2015 DAU will not avoid the Uncertainty Issue or the Unestablished Tariff Issue identified above.

Regulatory coverage and failure to approve

Stakeholders made a range of comments about regulatory coverage and a refusal to approve the extension DAAU.

Asciano and Glencore said they do not use the western system, so a rejection decision would result in their losing regulatory certainty with no offsetting gains in terms of the timing applying for the western system tariffs. The QCA acknowledges that this offsetting benefit (not as to timing of, but as to an appropriate level of, the tariff) arises only for existing and future users of the Western system.

Final decision: Queensland Rail's draft amending access undertaking - extension of termination date

Queensland Rail, Glencore and Aurizon submitted there were shortcomings in the negotiate/arbitrate and other provisions of the QCA Act (which would operate if the extension DAAU was not approved). For instance, Glencore said there are protections in the existing undertaking (like defined time periods for various stages of the access negotiation period, access conditions restrictions, the queuing framework, reporting, QCA information and audit powers) that do not exist in the QCA Act. Queensland Rail said the negotiate/arbitrate provisions lack detail, are relatively untested and that any negotiation or arbitration under those provisions would be negated or adversely affected by the approval of the 2015 DAU. Aurizon said that the 2008 access undertaking not being extended would create regulatory uncertainty, particularly for any access negotiations already commenced, and that the QCA's interest in holding Queensland Rail to its previous commitments may not achieve the desired effect if Queensland Rail avoids giving the commitment, the 2008 access undertaking lapses and negotiations have to take place under Part 5 of the QCA Act.

The QCA notes that stakeholders have processes available under the provisions of Part 5 Divisions 4 and 5 of the QCA Act which address these matters. The QCA acknowledges that there is some uncertainty about how these provisions will apply given that they are untested, however the QCA is confident that the QCA Act will operate as Parliament intended and that the procedure will be practically workable. Importantly, a certain outcome from an access application is ultimately available through the arbitration provisions of the QCA Act. In this way the QCA Act ensures an access application can be concluded with certainty without an access undertaking being in place.

The QCA is not satisfied that if a negotiation or arbitration commences under the QCA Act and the 2015 DAU is subsequently approved, then the incomplete negotiation or arbitration would be negated or adversely affected. To the contrary, the QCA considers the approved undertaking would likely assist in bringing any such processes to a close.

Aurizon did not provide examples of negotiations already commenced under the 2008 undertaking (nor did any other stakeholder), however again the QCA is satisfied the QCA Act provides an appropriate process for the resolution of these matters should they arise. Many requirements will still remain effective e.g. Queensland Rail's obligations to act in good faith, to not unfairly differentiate between access seekers and to provide relevant information to an access seeker (under ss. 100(1), 100(2) and 101 of the QCA Act, respectively).

The QCA does not accept that it has an interest in holding Queensland Rail to its previous commitments as such. Rather the QCA is concerned with the uncertainties (and related consequences) created by Queensland Rail's actions (not those actions of themselves) and it believes that those matters should be considered (where relevant) under section 138(2) of the QCA Act. Parties are of course free to agree an access contract that differs from a tariff set out in an access undertaking.

The QCA has had regard to the certainty (as to the terms of the arrangements that would apply) that would be achieved if the April 2015 Extension DAAU were approved. The QCA considers that the benefit of that certainty is outweighed by the Uncertainty Issue and the Unestablished Tariff Issue that would be produced if the April 2015 Extension DAAU were approved.

Removal of western system tariff from extension DAAU

New Hope said it preferred an undertaking to remain in place, provided that did not prejudice its ability to recoup any over-payment of tariffs compared to the tariffs which (it asserts) ought to have applied had a new undertaking been in place on 1 July 2013 and said that the QCA's draft position on how the extension DAAU could be amended would achieve that outcome. Final decision: Queensland Rail's draft amending access undertaking - extension of termination date

New Hope and Glencore suggested an alternative of removing the western system reference tariff from the undertaking for the extension period and extending the remainder of the undertaking. The QCA has had regard to these matters. However, we note Queensland Rail has not currently proposed to make such an amendment and there is insufficient time for Queensland Rail to submit such a proposal and have it considered before the expiry of the 2008 undertaking. Further, such an approach still risks creating the Uncertainty Issue (even if in some more limited form).

The QCA's approach

Queensland Rail previously sought seven extensions of its 2008 access undertaking. Between 30 March 2012 and 12 December 2014, in or in connection with its voluntary draft access undertakings, Queensland Rail had encouraged stakeholders to believe (and otherwise generally indicated) that it would give effect to new reference tariffs from 1 July 2013 by applying an Adjustment Charge (Attachment 1). The six extension requests from June 2012 to November 2014 were submitted and approved while that form of indication concerning an adjustment proposal was in place (Attachment 2). Queensland Rail's letters accompanying its extension DAAUs of May 2013, November 2013 and May 2014 also stated its intention to give effect to the new tariff from 1 July 2013 through adjustment charge provisions in its replacement DAU (Attachment 2).

Stakeholders did not object to the previous extension requests.

The QCA considered Queensland Rail's previous extension requests and stakeholder submissions in this context and approved the previous extensions.

Unlike previous extensions, the April 2015 Extension DAAU is not accompanied by a commitment from Queensland Rail to give effect to an Adjustment Charge as:

- Queensland Rail in December 2014 withdrew its June 2013 voluntary DAU that contained provisions to give
 effect to new tariff from 1 July 2013.
- Queensland Rail's cover letter accompanying the April 2015 Extension DAAU did not state its intention to apply tariff from 1 July 2013.
- The DAU Queensland Rail submitted on 5 May 2015 in response to the section 133 initial undertaking notice we issued, proposes to give effect to new reference tariff from the date we approve that DAU.

The QCA has considered stakeholder submissions in light of the above matters.

Our final decision

The QCA has considered Queensland Rail's April 2015 Extension DAAU and stakeholder comments as part of its consideration under s. 138(2) of QCA Act whether to approve or refuse to approve the April 2015 Extension DAAU.

The QCA's decision is to refuse to approve the April 2015 Extension DAAU. Having had regard to the matters mentioned in section 138(2) of the QCA Act, and weighing the competing considerations that arise, the QCA does not consider it is appropriate to approve the April 2015 Extension DAAU.

Queensland Rail's April 2015 Extension DAAU is not accompanied by a commitment from Queensland Rail to give effect to the new reference tariff from 1 July 2013 by means of an Adjustment Charge. The absence of such a commitment gives rise to the Uncertainty Issue and the Unestablished Tariff Issue.

5

Final decision: Queensland Rail's draft amending access undertaking - extension of termination date

The QCA has given careful consideration to the new material received from stakeholders concerning the regulatory uncertainty arising from not extending the 2008 access undertaking. In particular, the QCA has examined the potential uncertainty created by stakeholders having to rely on the untested process under the QCA Act for negotiating access to Queensland Rail's network rather than the provisions of an undertaking to which they are accustomed. While the QCA accepts the procedures are untested and that practical challenges may arise in respect of negotiations that commence in the interim period until a new access undertaking is approved, the QCA is not satisfied that these practical challenges are insurmountable or that the procedures under the QCA Act will be unworkable.

In the end, the QCA's task is to balance the future uncertainties flowing from on the one hand approving, and on the other hand not approving, the April 2015 Extension DAAU.

The QCA's position is that the disadvantages arising from approving the April 2015 Extension DAAU (particularly the Uncertainty Issue and the Unestablished Tariff Issue) outweigh the regulatory uncertainty, to the extent it may exist, from not approving the April 2015 Extension DAAU.

The way in which the QCA considers it is appropriate to amend the April 2015 Extension DAAU is by including the same kind of commitment as has previously been given by Queensland Rail, to give effect to new reference tariffs from 1 July 2013 by applying adjustment charge provisions to recoup or refund any variations between transitional and new tariffs.

Yours sincerely

Roy Green Chair

354

ATTACHMENT 1

DRAFT ACCESS UNDERTAKINGS SUBMITTED DURING COURSE OF QUEENSLAND RAIL'S (QR) 2008 UNDERTAKING

DAU	Date submitted	Date withdrawn	Adjustment charge provision
March 2012	30 March 2012	25 February 2013 ¹ (QR withdrew March 2012 DAU and at the same time submitted the February 2013 DAU).	The DAU specified in cl. 3.4.2(b)(iii)(C) that 'after new Reference Tariffs are approved by the QCA in accordance with this clause 3.4.2(b) this Undertaking will apply as though it were amended to replace the Reference Tariffs with those new Reference Tariffs with effect on and from 1 July 2013.'
			The DAU included adjustment charge provisions in Schedule A, cl. 6. Clause 6.1(a) specified
			"If: this Undertaking specifies that a Reference Tariff is applicable or effective from a date prior to the QCA's approval of that Reference Tariff; or Queensland Rail is entitled to recover from or will reimburse to, as applicable, each relevant Access Holder the amount (Adjustment Amount) which is the sum of:"
February 2013	25 February 2013	28 June 2013 ² (On this date, QR submitted for the first time reference tariffs for western system coal services, and proposed consequential amendments to the February 2013 DAU. This, in effect, was a withdrawał of its February 2013 DAU and submission of a new voluntary draft access undertaking (the June 2013 DAU). ³	The DAU specified in cl. 3.4.2(b)(iii)(C) the same provision as noted above in the March 2012 DAU. The DAU included adjustment charge provisions in Schedule A, cl. 6. Clause 6.1(a) specified the same provision as noted above in the March 2012 DAU.
June 2013	28 June 2013	12 December 2014 ⁴	The DAU specified in cl. 3.4.2(b)(iii)(C) the same provision as noted above in the March 2012 DAU. The DAU included adjustment charge provisions in Schedule A, cl. 6. Clause 6.1(a) specified the same provision as noted above in the March 2012 DAU.

¹ Queensland Rail, 25 February 2013.

² Queensland Rail, 28 June 2013

^a http://www.qca.org.au/Rail/Queensland-Rail/More-on-QLD-Rail/Draft-Access-Undertaking/Archive/February-2013-DAU

⁴ Queensland Rail, 12 December 2014.

ATTACHMENT 2

PREVIOUS EXTENSIONS OF QUEENSLAND RAIL'S 2008 UNDERTAKING

Extension DAAU	Queensland Rail's (QR) proposal	Stakeholder comments	QCA decision	
April 2011	Proposed extending the term to 30 June 2012. In the letter accompanying the DAAU, QR said the extension will maintain regulatory coverage of train services until the approval of a replacement DAU, which it was yet to submit then. ¹	No stakeholder objected to the extension but stakeholders raised concerns about QR taking too long to submit a replacement undertaking.	We considered it reasonable to approve the extension, given that QR was yet to submit an undertaking that better reflected its business activities, noting that the decision did not prejudice our ability to issue QR a section 133 initial undertaking notice. ²	
June 2012	Proposed extending the term to 31 December 2012. In the letter accompanying the DAAU, QR said the extension will maintain regulatory coverage of train services until the approval of its March 2012 replacement draft access undertaking (DAU). ³ The March 2012 DAU included adjustment charge provisions to backdate reference tariff to 1 July 2013 (see Attachment 1).	No stakeholder objected to the extension.	We approved the extension. ⁴	
October 2012	Proposed extending the term to 30 June 2013. In the letter accompanying the DAAU, QR said the extension will allow sufficient time for consultation with stakeholders and approval of its March 2012 DAU. ⁵	No stakeholder objected to the extension although New Hope wanted new reference tariffs to be in place by 1 July 2013 on the expectation that the tariffs for the regulatory period beginning 1 July 2013 would be lower. ⁶	We approved the extension. ⁷	
May 2013	Proposed extending the term to 31 December 2013. In the letter accompanying the DAAU, QR said the extension will provide continued regulatory certainty for stakeholders while the QCA's approval process for the	No stakeholder objected to the extension. However, stakeholders were disappointed that QR was yet to submit its western system reference tariff proposal and were concerned about rolling forward the existing tariff, as they expected the tariff to decline from 1 July	We approved the extension on the basis that it would allow time to finalise the 2013 DAU assessment process and that Queensland Rail intended to apply an adjustment charge mechanism to recover (or return) the difference between the transitional tariffs and the	

¹ Queensland Rail, April 2011. ² QCA, May 2011. ³ Queensland Rail, April 2012.

⁴ QCA, June 2012. ⁵ Queensland Rail, October 2012.

⁶ New Hope, November 2012. ⁷ QCA, November 2012.

Extension DAAU	Queensland Rail's (QR) proposal	Stakeholder comments	QCA decision
	replacement February 2013 replacement DAU is finalised. ⁸ In the letter, QR also stated that it was proposing that transitional reference tariffs will apply from 30 June 2013, being current reference tariffs escalated by CPI. QR said the adjustment charge provisions in AU1 [February 2013 DAU] will allow the reference tariff to be backdated upon the approval of AU1. QR noted that transitional tariffs were applied in the 2005 and 2010 access undertakings. ⁹	2013. ¹⁰ New Hope said 'the interim tariff applicable to the extension be set such as to minimise the risk of overpyament' and suggested adjusting QR's proposed interim tariff to reflect a lower weighted average cost of capital due to lower government bond rates and debt margins.	tariffs that would be approved as part of the 2013 DAU assessment process. ¹¹
November 2013	Proposed extending the term to the earlier of 30 June 2014 and the date on which we approve a replacement undertaking. In the letter accompanying the DAAU, QR said it intended to continue with its proposal that the transitional reference tariffs remain and continue to apply up until the approval of AU1 [June 2013 DAU ¹²]. QR also stated the adjustment charge provisions in AU1 will allow the reference tariff to be backdated to 1 July 2013 upon the approval of AU1. QR noted that transitional tariffs were applied in both the 2005 and 2010 access undertakings and is also in practice with Aurizon Network. ¹³	No stakeholder objected to the extension, although New Hope suggested we prioritise the determination of new reference tariffs as part of the 2013 DAU assessment process. ¹⁴	We approved the extension DAAU. ²⁵
May 2014	Proposed extending the term to the earlier of 31 December 2014 and the date on which we approve a replacement undertaking. In the letter accompanying the	We did not receive any submissions.	We approved the extension DAAU. ¹⁷

⁸ In February 2013, Queensland Rail withdrew its March 2012 DAU and submitted the February 2013 DAU, which retained provisions to apply reference tariffs from 1 July 2013 through an adjustment charge (see Attachment 1).

¹⁰ Aurizon, May 2013; ¹⁰ Aurizon, May 2013; ¹¹ QCA, June 2013; ¹²

QCA, June 2013. ¹² In June 2013, Queensland Rail resubmitted its February 2013 DAU and included, for the first time, its proposed reference tariffs for the western system. The June 2013 DAU retained provisions to apply reference tariffs from 1 July 2013 through an adjustment charge (see Attachment 1). ¹³ Queensland Rail, November 2013.

¹⁴ New Hope, November 2013. ¹⁵ QCA, November 2013.

Extension DAAU	Queensland Rail's (QR) proposal	Stakeholder comments	QCA decision
	DAAU, QR reiterated its intention to continue with its proposal to apply transitional reference tariffs until the approval the AU1 [June 2013 DAU] and said the adjustment charge provisions in AU1 will allow the reference tariff to be backdated to 1 July 2013. ¹⁶		
November 2014	Proposed extending the term to the earlier of 30 June 2015 and the date on which we approve a replacement undertaking. In the letter accompanying the DAAU, QR said the extension will provide continued regulatory certainty for stakeholders and allow sufficient time for the completion of the AU1 [June 2013 DAU] approval process. ¹⁸	We did not receive any submissions.	We approved the DAAU. ¹⁹ The letter accompanying the November 2014 extension DAAU did not withdraw Queensland Rail's earlier commitment to backdating tariffs and the relevant clauses were still before the QCA in the June 2013 DAU submitted by Queensland Rail.

- ¹⁷ QCA, June 2014.
 ¹⁶ Queensland Rail, May 2014.
 ¹⁸ Queensland Rail, November 2014.
 ¹⁹ QCA, December 2014.

APPENDIX I: LIST OF SUBMISSIONS

Organisation/individual	Submission number	
Asciano	5, 28	
Aurizon	6, 20, 29	
Glencore	7, 25*, 30	
New Hope	8, 9, 10, 11,12, 19*, 21, 22, 23, 24, 31*, 32*	
Port of Brisbane	13	
Queensland Rail	1, 2*, 3, 4, 17, 18*, 26*, 33*	
Queensland Resources Council	14, 34	
Sekitan Resources	15*	
Yancoal	16, 27, 35	

*Claims of confidentiality have been made for part or all of these submissions

REFERENCES

- Allen Consulting Group 2005, *Review of Asset Values, Costs and Cost Allocations of Western Australian Urban Water and Wastewater Service Providers*, March.
- Asciano 2015, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking, June. (sub. 5) 2016, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking, Request for Comment, March. (sub. 28)
- Aurizon 2015a, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking, June. (sub. 6) 2015b, Submission to the QCA—Queensland Rail 2015 Draft Access Undertaking Draft Decision, December. (sub. 20)

2016, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking Draft Decision, Request for comments on stakeholders' submissions, March. (sub. 29)

Aurizon and Queensland Rail 2013a, *South West Rail System Capacity Upgrade*, joint presentation, 22 February. [confidential]

2013b, South West Rail System Capacity Upgrade, joint presentation, 29 April. [confidential]

2013c, South West Rail System Capacity Upgrade, Joint Presentation, 27 June. [confidential]

- Australian Rail Track Corp (ARTC) 2015, *Melbourne–Brisbane Inland Rail Alignment 2015*, at http://inlandrail.artc.com.au/wp-content/uploads/2014/07/Inland-Rail-Base-Scope-Map-A3.pdf>.
- B&H Strategic Services Pty Ltd 2014, *Review of the Queensland Rail (QR) West Moreton System Maintenance Costs, Capital Costs, Operation Costs and Depreciated Optimised Replacement Cost,* report prepared for the Queensland Competition Authority, May.

2015, Review of Queensland Rail's DAU 2015, September.

2016, Submissions by Stakeholders in Response to the QCA's Draft Decision of the Queensland Rail DAU 2015, supplementary report prepared for the QCA, May.

Biggar, D 2010, Fairness in public-utility regulation: a theory, Agenda, v.17(1): 5–29.

- DBCT Management 2010, *Dalrymple Bay Coal Terminal Access Undertaking*, March. 2015, Dalrymple Bay Coal Terminal Access Undertaking, April
- Glencore 2015a, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking 1, June. (sub. 7)

2015b, Submission to the QCA—Queensland Rail 2015 Draft Access Undertaking, December (sub. 25).

2016, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking Draft Decision, March. (sub. 30)

- Houstan, A 2015, 'Explore Best Utilisation of Existing Rail Infrastructure on the South West Line', presentation at the Goondiwindi Rail Forum hosted by the Goondiwindi Regional Council, Royal Hotel Goondiwindi, 4 December.
- Incenta Economic Consulting 2016, *Benchmark BBB+ debt risk premium for 20 days to 12 April 2016*, report prepared for the QCA, May.
- Institute of Public Affairs 1999, Submission to the Office of the Regulator General (ORG)— Victoria's 2001 Electricity Distribution Price Review, Issues Paper no. 11, April.

King, S. 2016, An independent review of reports by Professor Menezes, May.

Menezes, F 2015a, A preliminary view: Regulatory economics assessment of the proposed Western System asset valuation approaches, April.

2015b, A Regulatory economics assessment of the proposed Western System asset valuation approaches, July.

2015c, The economic impact of QR's proposal not to include an adjustment to refund or recoup differences in tariffs July.

2016a, Assessing three options to allocate common costs, Draft Paper, April.

2016b, Response to Stakeholder comments on comments on "A Regulatory economics assessment of the proposed Western System asset valuation approaches" and "The economic impact of QR's proposal not to include an adjustment to refund or recoup differences in tariffs: Stage 1 Report." April.

New Hope 2013, Submission to the QCA—Queensland Rail's Proposed Tariff Reset, 31 October.

2014, Submission to the QCA—Queensland Rail's Western System Coal Tariffs, July.

2015a, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking, cover letter and vol. 1, June. (sub. 8)

2015b, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking vol. 2, June. (sub. 9)

2015c, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking vol. 3, June. (sub. 10)

2015d, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking vol. 4, June. (sub. 11)

2015e, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking vol. 5, June. (sub. 12)

2015f, Submission (additional) to the QCA—Queensland Rail's 2015 Draft Access Undertaking, August. (sub. 19)

2015g, Submission to the QCA—Queensland Rail's 2015 DAU Draft Decision, cover letter and vol. 1— Introduction to NHC Submissions & regulatory framework, December. (sub. 21)

2015h, *Submission to the QCA—Queensland Rail's 2015 DAU Draft Decision*, vol. 2—West Moreton Coal Reference Tariff, December. (sub. 22)

2015i, *Submission to the QCA—Queensland Rail's 2015 DAU Draft Decision*, vol. 3—Access Undertaking, December. (sub. 23)

2015j, Submission to the QCA—Queensland Rail's 2015 DAU Draft Decision, vol. 4—Standard Access Agreement, December. (sub. 24)

2016a, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking Draft Decision vol. 1, Request for comments, March. (sub. 31)

2016b, *Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking Draft Decision*, vol. 2, Submissions on other stakeholders' submissions, March. (sub. 32)

2016c, Submission on QCA's 2015 Draft Decision on Queensland Rail's 2015 Draft Access Undertaking ("2015 DAU"), letter to the QCA, 9 February.

Port of Brisbane 2015, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking, June. (sub. 13)

Productivity Commission 2001, Review of the National Access Regime, Inquiry Report no. 17, September.

QR Network 2008, *QR Network Access Undertaking (2009): Western System Coal Tariff Development*, September.

2009, Western System—Maximum Paths on the Toowoomba Range and Forecast Tonnage Flows, 20 March.

2010a, QR Network's Access Undertaking (2009): Response to QCA draft decision—Vol. 1—Nonpricing matters, February.

2010b, QR Network's 2010 Draft Access Undertaking, Vol. 1—Policy issues, April.

Queensland Competition Authority (QCA) 2000a, Statement of Regulatory Pricing Principles for the Water Sector, December.

2000b, Queensland Rail's Draft Undertaking, Draft Decision, vol. 2, December.

2001a, Proposed Access Arrangements for Gas Distribution Networks: Allgas Energy Limited and Envestra Limited, Draft Decision, March.

2001b, Queensland Rail's 1999 Draft Undertaking, Final Decision, July.

2001c, Proposed Access Arrangements for Gas Distribution Networks: Allgas Energy Limited and Envestra Limited, Final Decision, October.

2005, QR's 2005 Draft Access Undertaking, Decision, December.

2009a, QR Network 2009 Draft Access Undertaking, Draft Decision, December.

2009b, QR Network 2009 DAU—Western System Model, December.

2010a, QR Network's 2010 DAU—Tariffs and Schedule F, Draft Decision, June.

2010b, QR Network's 2008 Extension DAAU (Pricing), Final Approval, June.

2014a, Financial Capital Maintenance and Price Smoothing, Information Paper, February.

2014b, Queensland Rail's Western System Coal Tariffs, Consultation Paper, June.

2014c, Aurizon Network 2014 Draft Access Undertaking—Maximum Allowable Revenue, Draft Decision, September.

2014d, Queensland Rail's 2013 Draft Access Undertaking, Draft Decision, October.

2015, Queensland Rail's 2015 Draft Access Undertaking, Draft Decision, October.

2016a, Dalrymple Bay Coal Terminal's 2015 Draft Access Undertaking, Draft Decision, April.

2016b, Aurizon Network 2014 Access Undertaking, Final Decision, April.

2016c, 2015 DAU WACC averaging period, letter to Queensland Rail, 1 March.

2016d, 2015 DAU WACC averaging period, letter to Queensland Rail, 10 March.

2016e, *Queensland Rail's 2015 DAU—Request for comment—following submissions on Draft Decision*, staff paper, January.

Queensland Department of Transport and Main Roads (DTMR) 2011, letter to Queensland Rail, 8 July [confidential].

2014a, letter to Queensland Rail, 10 July. [confidential]

2014b, letter to Queensland Rail, 29 September. [confidential]

Queensland Rail 2012, letter to DTMR re: Western System Coal Paths, 16 August. [confidential] 2013, *Explanatory Submission—Queensland Rail's Draft Access Undertaking*, February.

2013a, AU1 West Moreton Reference Tariff Reset—Overall Submission (Public Release), June.

2013b, AU1 West Moreton Reference Tariff Reset—Maintenance Submission (Public Release), June.

2013c, AU1 West Moreton Reference Tariff Reset—Capital Submission (Public Release), June.

2013d, 'QCA West Moreton System Information Request (AU1 Maintenance)', Appendix A in a report prepared by B&H, *Review of the Queensland Rail (QR) West Moreton System Maintenance Costs, Capital Costs, Operation Costs and Depreciated Optimised Replacement Cost,* November. (see B&H 2014)

2014a, Submission to the QCA—West Moreton Reference Tariff Consultation Paper, July.

2014b, letter to DTMR re: West Moreton Coal Paths, 27 May. [confidential]

2014c, letter to New Hope Group re: West Moreton Coal Paths, 29 July. [confidential]

2014d, letter to Yancoal Australia Ltd, 29 July. [confidential]

2014e, letter to New Hope Group re: West Moreton Coal Paths, 10 October. [confidential]

2014f, letter to Yancoal Australia Ltd, 10 October. [confidential]

2014g, Submission to the QCA—West Moreton Reference Tariff Consultation Paper, July.

2015a, *Explanatory submission to the QCA—Queensland Rail's Draft Access Undertaking vol. 1*, letter included, May. (sub. 1)

2015b, Explanatory submission to the QCA—Queensland Rail's Draft Access Undertaking vol. 2, May. (sub. 2)

2015c, Submission to the QCA—Submissions on "A preliminary view: Regulatory economics assessment of the proposed Western System asset Valuation approaches", report by F Menezes, June. (sub. 3)

2015d, Submission to the QCA—Queensland Rail's Draft Access Undertaking 1, Request for comment, June. (sub. 4)

2015e, Submission to the QCA—DORC valuation and roll forward of initial asset base for West Moreton Network, July. (sub. 17)

2015f, Review of QR Pricing Models, Final Report, July. (sub. 18) [confidential]

2015g, West Moreton Network Information Request (2015 DAU TSC Capital), response to the QCA, 21 August. [confidential]

2015h, QCA West Moreton Network Information Request (2015 DAU Metropolitan Network Capital), response to the QCA, 21 August. [confidential]

2015i, QCA West Moreton Network Information Request (2015 DAU Capacity and Maintenance), response to the QCA, 19 August. [confidential]

2015j, Queensland Rail's 2015 Draft Access Undertaking 1—QCA notice to produce information under s. 185 of the QCA Act (2015 DAU access agreements, revenue and contracting constraints), response to the QCA, 17 July. [confidential]

2015k, West Moreton Network Information Request (2015 DAU Maintenance & Capital), response to the QCA, 20 August. [confidential]

2015I, *Goondiwindi Rail Forum—South Western Rail System*, presentation to the Goondiwindi Rail Forum hosted by the Goondiwindi Regional Council, Royal Hotel Goondiwindi , 4 December 2015, at <http://www.goondiwindirc.qld.gov.au/rail-forum> (accessed 3 March 2016).

2015m, Submission to the QCA-Queensland Rail's 2015 DAU Draft Decision, December. (sub. 26)

2015n, 'RE: Request for information - Capacity calculations for the Toowoomba Range', email to the QCA, 24 July.

2016a, letter to the QCA re: Ebenezer Contracted Paths, 14 April. [confidential]

2016b, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking Draft Decision, Request for comment, March. (sub. 33)

2016c, letter to QCA re: 2015 DAU WACC averaging period, 9 March.

2016d, letter to QCA re: QCA 2015 Draft Decision—Queensland Rail's Draft Access Undertaking 1, 29 April. [confidential]

2016e, 'Figure 8.2 & 8.3 of the Draft Decision—Maintenance', email to the QCA, 15 April.

Queensland Resources Council 2010a, Submission to the QCA—QR Network's 2009 Draft Access Undertaking Draft Decision, February.

2010b, Submission to the QCA—QR Network's 2010 Draft Access Undertaking—Reference Tariffs, May.

2010c, Submission to the QCA—Draft Amending Access Undertaking—Proposed reference tariffs and related pricing matters, June.

2015, Submission to the QCA—Queensland Rail's 2015 Access Undertaking, June. (sub. 14)

2016, Submission to the QCA—Queensland Rail's 2015 Access Undertaking Draft Decision, Request for comment, March. (sub. 34)

Sekitan Resources 2015, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking, June. (sub. 15)

Syntech Resources 2010, QR 2010 Access Undertaking, May.

Yancoal 2015a, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking, June. (sub. 16) 2015b, Submission to the QCA— Queensland Rail's 2015 DAU Draft Decision, December. (sub. 27)

2016, Submission to the QCA—Queensland Rail's 2015 Draft Access Undertaking Draft Decision, Request for comment, March. (sub. 35)